

Haroun, Omer Mohammed Osman Hamid (2015) Neuropathic Pain In Leprosy: Deep Profiling And Stratification Of Patient Groups. PhD thesis, London School of Hygiene & Tropical Medicine. DOI: https://doi.org/10.17037/PUBS.02030956

Downloaded from: http://researchonline.lshtm.ac.uk/2030956/

DOI: 10.17037/PUBS.02030956

Usage Guidelines

Please refer to usage guidelines at http://researchonline.lshtm.ac.uk/policies.html or alternatively contact researchonline@lshtm.ac.uk.

Available under license: http://creativecommons.org/licenses/by-nc-nd/2.5/



NEUROPATHIC PAIN IN LEPROSY: DEEP PROFILING AND STRATIFICATION OF PATIENT GROUPS

Omer Haroun

Thesis submitted in accordance with the requirements for the degree of Doctor of Philosophy of the University of London March 2015

Department of Clinical Research Faculty of Infectious Tropical Diseases LONDON SCHOOL OF HYGIENE & TROPICAL MEDICINE

Funded by Hospital and Homes of St. Giles

Declaration

I, Omer Haroun, confirm that the work presented in this thesis is my own. Where information has been derived from other sources, I confirm that this has been indicated in the thesis.

Signed:

(ST)

Date: 22.08.2014

Full name: Omer Mohammed Osman Hamid Haroun

Abstract

Leprosy is a chronic granulomatous infectious disease, affecting the skin and nerves. Neuropathic pain (NP), which is defined as pain caused by a lesion or disease of the somatosensory nervous system, is now being recognized as an important complication of leprosy. It occurs in 10-20% of patients as a result of persisting nerve damage. Patient with NP is associated with significant suffering, morbidity and limitation of quality of life. Thus, the accurate identification of NP in patients with pain is required. The clinical aspects of NP in leprosy patients in India were investigated using highly specialised assessment tool; quantitative sensory testing (QST). A case control study was conducted in 90 patients with and without pain. Two validation studies were conducted among healthy volunteers in London (18 participants) and Mumbai (52 participants).

Somatosensory profiles were compared in leprosy patients to healthy control subjects. The pattern revealed a novel profile of loss of cool and warm detection thresholds and also mechanical detection but with preservation of vibration detection. This is different to profiles seen in other NP conditions. The QST parameters were effective in detecting neuropathy, but were not able to distinguish between patients with and without NP. Patients with leprosy NP had a high rate of abnormal findings in almost all QST parameters in the maximum pain area over the ulnar nerve. Their sensory profiles were categorised into two subgroups. The majority of patients have spontaneous pain with evidence of sensory loss, but no signs of sensory gain. The second subgroup had profoundly impaired pain and temperature sensation, but light mechanical stimuli often produce pain. Patients with NP had a poor quality of life and psychological well-being compared to pain-free neuropathy.

For all leprosy patients who are suffering pain worldwide.

Acknowledgments

I would like to express my sincerest gratitude to my supervisors Prof Diana Lockwood and Prof Andrew Rice for generously sharing their time and expertise in support of this thesis. Their support, encouragement and guidance have provided a constant source of personal motivation. I am also grateful for the comments and advice I received from my other advisory committee members: Dr Ron Behrens and Dr Dave Bennett. I am also grateful to Dr Petr Nichols who gave me statistical advice throughout.

I would also like to thank the German Network on Neuropathic Pain Group: Prof Ch. Maier and Prof R-D Treede for their support during my visit to Bochum and Mannheim, Germany. Furthermore, I would like to thank Dr Doreen Pfau and Elena Krumova from the DFNS group for their assistance with training and valuable advice and feedback on QST.

I am grateful to the following institutes for their support: Foundation for Medical Research for providing laboratory facilities and equipment and the Bombay Leprosy Project for providing me with a good number of patients. I would like to thank the staff at these two institutions: Dr N. F. Mistry, Dr V. Shetty, Dr S. Pandya, Ms. G Capadia and Dr S. Khadilkar who gave me the initial advice and guidance to set up the project; Dr V. V. Pai who has given me invaluable advice and guidance throughout; Mrs Anju Wakade, Dr Ashish, and Dr Vikas who gave me immensely valuable support; Mr Ram who kept everything running smoothly; the administration at FMR, especially Mr. Paras, and Mrs Jaia; the administration at BLP, especially Mr. S Hoval and finally all the people I have forgotten to mention here. I am also specifically grateful for the advice and guidance on tissue cutting and processing I received from Dr Dhananjaya Saranath. Furthermore, I would also like to acknowledge all the individuals that helped me in one way or another in the completion of this thesis.

I would like to thank all the staff at the London School of Hygiene and Tropical Medicine, in particular Prof Simon Croft, Prof Robin Bailey and Prof John Porter whose encouragement has supported me through all the difficulties of writing up, especially during fasting time. I am grateful for support from the leprosy group, in particular Steve Walker who provided a source of encouragement throughout my studies and shared in all of my frustrations and achievements; Dr Ula Nour, Eleonora Staines-Urias and Dr Helen Kelly who gave me statistical advice; Dr Julia Mortimer who spent times listening to me talk about my research.; and Susan Sheedy whose administrative assistance has been invaluable.

Furthermore, I would like to thank my colleagues at Prof Rice's laboratory, in particular Phillip Tudor, Tim Pheby, and Helen Laycock who provided advice and support during my studies at the Chelsea and Westminster Hospital, Rossi, Donna and Rachel for last minute favours. I would like also to thank my colleagues in Dr Bennett's laboratory at Kings College London, and in particular, Dr Ramirez for their assistance with training on skin biopsy techniques. At the University of Khartoum, I must thank Dr Ahmed Mussa, who provided much-needed encouragement and insight.

I am grateful to Dr Aki Hietaharju and Maija Haanpaa for their training and support, as well as their hospitality during my visit to Finland. I am grateful to Dr Hdi Manji in the UCL National Hospital for Neurology and Neurosurgery in Queens Square for letting me observe neurology testing sessions during private patient care. Furthermore, I would like to thank my colleagues at TENLEP group, and in particular Dr Wim Brandsma for their advice and feedback on detecting leprosy neuropathy.

I would like to thank the Hospital and Homes of St. Giles who provided the funding for this study, not only for their generous financial support, but I am also especially grateful for the technical advice I received from the committee members during meetings.

I am incredibly grateful to my family, in particular my parents for always supporting my decisions and for being proud of me, and many thanks to my sisters and brothers who were always looking for better education and encouraged me to undertake my PhD study.

Table of Contents

Title Page		1
Declaration		2
Abstract		3
Dedication.		4
Acknowledg	ments	5
Table of Con	ntents	7
List of Table	es	11
List of Figur	res	12
List of Appe	ndices	13
List of Abbr	eviations	15
Chapter 1	Introduction	
11 PhI) approach	19
1.1 Phi 1.2 Phi) thesis timeline	20
1.2 Im 1.3 Ord	results unclude in this PhD research	.
1.5 OV	inture of the PhD thesis	20
1.4 Suc	ionale of the PhD thesis	
1.5 Rat	os of the PhD thesis	22 23
1.0 Am 1.7 Ob	actives of the DhD thesis	23
1.7 Uu	acthered of the PhD there's	23
1.0 Пуј		23
Chapter 2	Review of Literature	25
2.1 Lite	rature review of leprosy	25
2.1.1	Definition of leprosy	25
2.1.2	Leprosy transmission	25
2.1.3	Epidemiology of leprosy	25
2.1.4	Leprosy Classification	28
2.1.5	Clinical Features of Leprosy	29
2.1.6	Diagnosis of Leprosy	29
2.1.7	Treatment of Leprosy	31
2.1.8	Immunological complications of leprosy	
2.1.9	Leprosy Reactions	35
2.2 The	e central role of nerve damage in leprosy	
2.2.1	Introduction and definitions	
2.2.2	The nerve in Leprosy	41
2.2.3	Clinical presentations	44
2.2.4	Nerve function assessment	47
2.2.5	The differential diagnosis of neurological lesions of leprosy	50
2.2.6	Nerve damage consequences in leprosy	50
Chapter 3	Identification of neuropathic pain	51
3.1 Lite	rature review of pain	51
3.1.1	Definition of pain	51
3.1.2	Classification of pain	51
3.2 Lite	rature review of Neuropathic pain	52
J.2 LAC	1 1	

3.2.2	Epidemiology	53
3.2.3	Neuropathic pain classification and aetiology	53
3.2.4	Development of neuropathic pain	54
3.2.5	Neuropathic pain underling mechanisms	54
3.2.6	Clinical features of neuropathic pain	55
3.2.7	Diagnosis and assessment of neuropathic pain	57
3.2.8	Psychological co-morbidity, quality of life and neuropathic pain	58
3.2.9	Treatment of neuropathic pain	59
3.3 P	ain in leprosy	62
3.3.1	Types of pain in leprosy	62
3.3.2	Literature review of neuropathic pain in leprosy	62
3.3.3	Summary	68
Chapter 4	Instruments for assessing sensory abnormalities, pain d	isorders
and techn	iques for assessing the psychological impact of neuropathic pair	n69
4.1 In	ntroduction	69
4.2 C	Overview of the structure and function of the somatosensory system	69
4.2.1	Introduction	69
4.2.2	Types and functions of the peripheral nerve fibre	70
4.2.3	Somatosensory receptors	71
4.3 In	nstruments for assessing sensory and pain-related phenomena	72
4.3.1	Pain scales and questionnaires for symptoms	74
4.3.2	Clinical signs assessment	76
4.3.3	Skin biopsy	80
4.4 T	echniques for assessing psychological impact of neuropathic pain	
4.4.1	Introduction:	82
4.4.2	Ouality of life and psychological state questionnaires literature review	82
4.5 T	The use of DN4 and PD-O questionnaires for assessing NP	
4.5.1	Overview:	
4.5.2	Limitations of the DN4 and PD-O questionnaires	85
4.6 T	he use of OST methods for assessing sensory abnormalities and NP	
4.6.1	Introduction	86
4.6.2	History and background	86
4.6.3	Literature review of OST	
4.6.4	Overview of the OST principle	
4.6.5	OST algorithms	
4.6.6	\widetilde{OST} instruments	
4.6.7	∞ OST and standard bedside neurologic sensory testing compared	
4.7 P	revious studies on painful and painless neuropathy using OST	
4.8 C	Chapter summary	
Chapter 5	The clinical problem and aims of the thesis	96
51 T	The clinical difficulties in NP in diseases such as leprosy	96
511	Problems of defining neuropathic pain in letrosy	
512	Challonges in defining NIP in letress tratients	
513	The impact of diagnosis on treatment	103
5.1.7	The most of trafiling and stratification of later to ation to another	101
フ.1.4 5.2 ケ	The role of instruments for assessing sensory abnormality and pavel	
J.Z I	of ND in longory	1010gical
	The role of the an exection of the second se	105 105
).2.1 500	The role of Our question nutres	100
).Z.Z	The role of chin history and IENED	108
2.2.2	The role of skin biopsy and $1 E_1 N_1 D$	

5.2.	4 The role of Psychological co-morbidity and HRQoL instruments	114
5.3	Aims and objectives of the thesis	116
5.3.	1 Aims of study	116
5.3.	2 Statement of the hypothesis	116
Chanter	6 Materials and Methods	117
61	Overview	117
6.2	The search strateov and search criteria	117
63	Definitions of terms used in this thesis	118
63	1 Definitions	118
63	2 Case definitions	119
6.3	3 Pain terminology	121
64	Study design	121
64	1 An overview of the study design and research procedures	121
6.4	2 Study setting	
6.4	3 Sample sizes	132
6.4	4 Ethics and consent practice	133
6.5	Clinical and laboratory assessment	
6.5.	1 Patient interviews and examinations	133
6.6	Ouantitative Sensory Testing	139
6.6.	1 Testing sites	139
6.6.	2 Testing protocol	140
6.7	Data recording and management	146
6.7.	1 Overview	146
6.7.	2 Data preparation	147
6.8	Statistical analysis	147
6.8.	1 Strategy of analysis	147
6.8.	2 QST data analysis	151
Chanter	7 Validation studies	153
7 1	Background	153
7.1	Overview of my preparation in this PhD research	154
73	Healthy volunteers study to validate investigator	
1.0		155
73	1 Introduction	155 155
7. <i>3</i> . 7.3.	Introduction And methods	155 <i>155</i> <i>155</i>
7.3. 7.3. 7.3.	1 Introduction 2 Participants and methods 3 Data evaluation and results	155 <i>155</i> <i>155</i> <i>158</i>
7.3. 7.3. 7.3. 7.3.	Introduction Introduction Participants and methods Introduction Data evaluation and results Introduction Image: Addition of the second seco	155 <i>155</i> <i>155</i> <i>158</i> <i>161</i>
7.3. 7.3. 7.3. 7.3. 7.3	1 Introduction 2 Participants and methods 3 Data evaluation and results 4 Discussion Healthy volunteers study to validate local centre and population in India.	155 <i>155</i> <i>155</i> <i>158</i> <i>161</i> 162
7.3. 7.3. 7.3. 7.3. 7.4 7.4	 <i>Introduction</i> <i>Introduction</i> <i>Participants and methods</i> <i>Data evaluation and results</i> <i>Discussion</i> Healthy volunteers study to validate local centre and population in India <i>Introduction</i> 	155 <i>155</i> <i>155</i> <i>158</i> <i>161</i> 162 <i>162</i>
7.3. 7.3. 7.3. 7.3. 7.4 7.4. 7.4. 7.4.	 Introduction	155 <i>155</i> <i>155</i> <i>158</i> <i>161</i> 162 <i>162</i> <i>162</i>
7.3. 7.3. 7.3. 7.3. 7.4 7.4 7.4. 7.4. 7.	 Introduction	155 <i>155</i> <i>155</i> <i>158</i> <i>161</i> 162 <i>162</i> <i>162</i> <i>162</i> <i>165</i>
7.3. 7.3. 7.3. 7.3. 7.4 7.4. 7.4. 7.4. 7	 Introduction Participants and methods Data evaluation and results Discussion Healthy volunteers study to validate local centre and population in India. Introduction Participants and methods Data evaluation and results. Data evaluation and results. Data evaluation and results. 	155 155 155 158 161 162 162 162 165 165
7.3. 7.3. 7.3. 7.4 7.4 7.4. 7.4. 7.4. 7.	 Introduction	155 155 155 161 162 162 162 165 165 using
7.3. 7.3. 7.3. 7.3. 7.4 7.4. 7.4. 7.4. 7	 Introduction	155 155 155 158 161 162 162 165 using 169
7.3. 7.3. 7.3. 7.4 7.4 7.4 7.4 7.4 7.4 7.5 differe 7.5	 Introduction Participants and methods Data evaluation and results. Discussion Healthy volunteers study to validate local centre and population in India. Introduction Participants and methods Data evaluation and results. Introduction Introduction 	155 155 155 161 162 162 162 165 165 using 169 169
7.3. 7.3. 7.3. 7.4 7.4 7.4. 7.4. 7.4. 7.	1 Introduction 2 Participants and methods 3 Data evaluation and results 4 Discussion 1 Introduction and results 4 Discussion 1 Introduction in India. 1 Introduction 2 Participants study to validate local centre and population in India. 1 Introduction 2 Participants and methods 3 Data evaluation and results. 4 Discussion 3 Data evaluation and results. 4 Discussion 5 Data evaluation and results. 4 Discussion 4 Discussion 4 Discussion 5 Materials and Methods 6 Materials and Methods	155 155 155 161 162 162 162 162 165 165 using 169 169 169
7.3. 7.3. 7.3. 7.3. 7.4 7.4. 7.4. 7.4. 7	1 Introduction 2 Participants and methods 3 Data evaluation and results 4 Discussion 1 Introduction and results 2 Participants study to validate local centre and population in India. 1 Introduction 2 Participants and methods 3 Data evaluation and results 4 Discussion 2 Participants and methods 3 Data evaluation and results 4 Discussion 3 Data evaluation and results 4 Discussion 2 Comparability of mechanical detection and pain thresholds in QST 1 Introduction 2 Materials and Methods 3 Results	155 155 155 158 161 162 162 165 165 using 169 169 169 173
7.3. 7.3. 7.3. 7.4 7.4. 7.4. 7.4. 7.4. 7	1 Introduction 2 Participants and methods 3 Data evaluation and results 4 Discussion 5 Healthy volunteers study to validate local centre and population in India 6 Introduction 7 Participants and methods 8 Discussion 9 Participants and methods 10 Introduction 11 Introduction and results 12 Participants and methods 13 Data evaluation and results 14 Discussion 15 Data evaluation and results 16 Discussion 17 Introduction and results 18 Discussion 19 Introduction 10 Introduction 11 Introduction 12 Materials and Methods 13 Results 14 Discussion	155 155 155 165 161 162 162 162 165 165 using 169 169 169 169 169 173 176
7.3. 7.3. 7.3. 7.4 7.4. 7.4. 7.4. 7.4. 7	1 Introduction 2 Participants and methods 3 Data evaluation and results 4 Discussion 5 Comparability 6 Discussion 7 Introduction in India 8 Introduction 9 Participants study to validate local centre and population in India 1 Introduction 2 Participants and methods 3 Data evaluation and results 4 Discussion 5 Conclusions	155 155 155 165 161 162 162 162 165 165 169 169 169 173 176 176
7.3. 7.3. 7.3. 7.4 7.4. 7.4. 7.4. 7.4. 7	1 Introduction 2 Participants and methods 3 Data evaluation and results	155 155 155 155 161 162 162 162 165 165 169 169 169 169 169 173 176 176
7.3. 7.3. 7.3. 7.4 7.4. 7.4. 7.4. 7.4. 7	1 Introduction 2 Participants and methods 3 Data evaluation and results. 4 Discussion 4 Discussion 5 Comparability of mechanical detection and pain thresholds in QST 1 Introduction 2 Participants and Methods 3 Data evaluation and results. 4 Discussion 5 Conclusions. 8 Results Introduction	155 155 155 161 161 162 162 162 165 165 169 169 169 169 173 176 176 176
7.3. 7.3. 7.3. 7.3. 7.4 7.4. 7.4. 7.4. 7	1 Introduction 2 Participants and methods 3 Data evaluation and results 4 Discussion 5 Comparability of mechanical detection and pain thresholds in QST 7 Introduction 2 Participants and methods 3 Data evaluation and results 4 Discussion 2 Participants and methods 3 Data evaluation and results 4 Discussion 6 Comparability of mechanical detection and pain thresholds in QST 7 Introduction 2 Materials and Methods 3 Results 4 Discussion 5 Conclusions 5 Results 6 Results Introduction Descriptive analysis	155 155 155 161 162 162 162 162 165 165 169 169 169 176 176 176 177 177

8.2.1	Baseline characteristics of study population	177
8.3 Clin	nical symptoms findings	
8.3.1	Pain and sensory symptoms in leprosy patients with pain	
8.4 Clin	nical examination findings	199
8.4.1	Skin assessment	199
8.4.2	Evaluation of enlargement and tenderness of main peripheral nerves	200
8.4.3	Nerve functions assessment	
8.4.4	Disability assessment	201
8.5 Qu	antitative Sensory Testing (QST) findings in leprosy patients	
8.5.1	QST observations in healthy control participants	204
8.5.2	QST observations in patients	204
8.5.3	Distribution of QST measures	204
8.5.4	Somatosensory profiles of leprosy patients	212
8.6 Pai	n intensity and quality assessment	233
8.7 Ev:	aluation of psychological co-morbidity and HRQoL	236
8.7.1	Health related quality of life	236
8.7.2	Psychological well-being	
Chapter 9	Discussion	
9.1 Int	roduction	
9.2 Sur	nmary and discussion of the findings	
9.2.1	Sociodemographic, clinical and epidemiological characteristics of study partici	pants249
9.2.2	Diagnostic tools	252
9.2.3	Quantitative sensory testing	256
9.2.4	Pain intensity and quality assessment	261
9.2.5	Psychological co-morbidity and HRQoL	262
9.3 Ge	neral discussion	
9.3.1	Classification of pain in leprosy	266
9.3.2	Somatosensory profiling	266
9.3.3	Strengths and limitations	269
9.3.4	Practical considerations in doing research in resource-limited settings	273
9.3.5	If I had to start now, this is what I would do differently	275
9.3.6	What contributions has this study made to our knowledge?	276
9.3.7	Future perspectives	
Chapter 10	Conclusions and Recommendations	
10.1 Co	nclusions	
10.2 Red	commendations	
References		
Annendices		316
Prenances	· · · · · · · · · · · · · · · · · · ·	

List of Tables

Table 2.1. WHO-recommended MDT doses and regimes for adult with leprosy	34
Table 2.2. Physiological function of the motor, sensory and autonomic nerve fibres	48
Table 3.1. Nociceptive pain vs. Neuropathic pain	52
Table 3.2. The prevalence of NP in different conditions	53
Table 3.3. Possible neuropathic pain mechanisms	55
Table 3.4. Grading system for neuropathic pain	58
Table 3.5. Overview of studies carried out on neuropathic pain in leprosy	63
Table 3.6. NP symptoms and related mechanisms in leprosy patients*	65
Table 4.1. Peripheral nerve fibre types and classification	72
Table 4.2. Devices cleared for marketing by the FDA (1980-2003)	87
Table 4.3. QST and standard bedside neurologic sensory testing	91
Table 4.4. QST and conventional electrophysiological techniques	92
Table 4.5. QST studies on painful neuropathy and painless neuropathy	94
Table 5.1. Overview of screening tools in studies carried out on leprosy NP	.106
Table 5.2. Quantitative Sensory Testing studies in leprosy neuropathy	.110
Table 6.1 Search terms used in the study	.118
Table 6.2. Design of study groups selection	.122
Table 7.1. Sensory functions represented by different QST parameters	.161
Table 7.2. QST results performed in hands $(n=50)$ and feet $(n=4)$.167
Table 8.1. Age distribution of the sample by sex	.179
Table 8.2. Demographic and characteristics of the study population $(n=90)$.182
Table 8.3. Disability grades present in patients at diagnosis (n=90)	.184
Table 8.4. Number of participants and study groups $(n=142)$.185
Table 8.5. Demographic and clinical characteristics of participants by group $(n=142)$	186
Table 8.6. Characteristics of patients $(n=36)$ with pain	.189
Table 8.7. Pain localisation in 36 patients with leprosy (-26)	.190
Table 8.8. Association of pain with reactions at the time of examination $(n=20)$	105
Table 8.9. A comparison of symptoms detected by DN4 in pain groups $(n=32)$	107
Table 8.10. A comparison of symptoms detected by PD-Q in pain groups $(n-32)$.197
Table 8.11. A comparison of NP detected by DN4 and PD-Q ($n=32$) \uparrow	.198
Table 8.12. Distribution of skin lesions in leprosy patients $(n=90)$.199
Table 8.13. Prevalence of palpable nerve enlargement and tenderness $(n=90)$.202
Table 8.14. Number and percentage of impaired ulnar nerves $(n=90)$.202
Table 8.15. Comparability of impairment by MF test and QS1 $(n-88)$	203
Table 8.16. A comparison of QST measures between patients and controls	.209
Table 8.17. Descriptive statistics for QS1 parameters T	.215
Table 8.18. QST parameters for leprosy patients with pain \uparrow	.220
Table 8.19. QST parameters in leprosy patients with and without pain \uparrow	.222
Table 8.20. Frequency of abnormal values in NP (n=21) \uparrow	.227
Table 8.21. Comparison between symptomatic and asymptomatic sides (n=21) \uparrow	.231
Table 8.22. Distribution of abnormal findings in study population (n=142) \uparrow	.232
Table 8.23. Descriptive statistics for pain intensity measures (n=36)	.235
Table 8.24. Pain intensity between pain groups (n=36)	.235
Table 8.25. Descriptive statistics for pain interference using BPI (n=35)	.236
Table 8.26. BPI interference in pain groups	.239
Table 8.27. GHQ-12 scores [↑]	.242

List of Figures

Figure 2.1.	Global leprosy prevalence 2012	26
Figure 2.2.	Prevalence and incidence of leprosy in India	27
Figure 2.3.	R-J scale and the relation with bacterial and immune response	28
Figure 2.4.	Commonly affected peripheral nerve in leprosy	47
Figure 3.1.	Co-morbidity associated with neuropathic pain	59
Figure 4.1.	The somatosensory system	70
Figure 4.2.	Summary of the methods used for QST	90
Figure 5.1.	Dharavi, Mumbai' largest slum	99
Figure 6.1.	Study site – Mumbai	125
Figure 6.2.	Foundation for Medical Research – Mumbai	126
Figure 6.3.	Bombay Leprosy Project – Mumbai	127
Figure 6.4.	Assessment of neuropathy and NP case-definition	130
Figure 6.5.	QST Testing sites	139
Figure 6.6.	MSA Thermal Sensory Analyzer – Somedic	142
Figure 6.7.	Tuning fork	146
Figure 6.8.	Flow charts indicating appropiate statistical test	148
Figure 7.1.	Complete set of QST devices	157
Figure 7.2.	Abnormal parameters in test area compared to DFNS reference	159
Figure 7.3.	Abnormal parameters in control area compared to DFNS reference	159
Figure 7.4.	Z-score of Rt vs. Lt ulnar of the 18 subjects compared to DFNS	160
Figure 7.5.	Z-score of Rt vs. Lt radial of the 18 subjects compared to DFNS	160
Figure 7.6.	Thermal and Mechanical QST results in hands (n=50)	168
Figure 7.7.	Thermal and Mechanical QST results in feet (n=4)	168
Figure 7.8.	MDT results from the face, fingertip and hand	174
Figure 7.9.	Comparison between Pinprik, MFs and evF	175
Figure 8.1.	Flow diagram for LePaIn study	178
Figure 8.2.	Geographical origin of 90 patients presenting with leprosy	180
Figure 8.3.	Pain frequency in 36 patients with leprosy	190
Figure 8.4.	Pain grade in 36 patients with leprosy	191
Figure 8.5.	Frequency of pain descriptors	193
Figure 8.6.	DN4 scores in 32 leprosy patients with upper limb pain	194
Figure 8.7.	PD-Q scores in 32 leprosy patients with upper limb pain	196
Figure 8.8.	Patients' responses for the DN4 and PD-Q questionnaires (n=36)	199
Figure 8.9.	Distribution of the QST parameters after z-transformation	206
Figure 8.10). Frequencies of abnormal QST measures	208
Figure 8.11	1. Thermal QST measures for patients (n=88) and controls (n=50)	210
Figure 8.12	2. Mechanical QST measures for patients $(n=88)$ and controls $(n=50)$	211
Figure 8.13	3. Leprosy sensory profiling	214
Figure 8.14	4. Thermal QST measures for leprosy patients with no pain (n=54)	216
Figure 8.15	5. Mechanical QST measures for leprosy patients with no pain $(n=54)$	217
Figure 8.10	5. Sensory profiling for pain patients (n=32) †	219
Figure 8.17	7. QST findings for leprosy patients (n=88)and controls (n=50)	223
Figure 8.18	3. Leprosy neuropathic pain sensory profiling	225
Figure 8.19). Percentages of abnormal QST values in NP patients	227
Figure 8.20). Somatosensory profiles in Non-NP patients	229
Figure 8.21	1. Somatosensory profiles in NP patients	229
Figure 8.22	2. QST findings for patients with pain $(n=32)$ and controls $(n=50)$	230
Figure 8.23	3. Distribution of pain scores (n=36)	234

Figure 8.24. Pain intensity in NP and non-NP sub-groups (n=36)	235
Figure 8.25. BPI Pain interference	237
Figure 8.26. Mean BPI interference scores in patients with pain	238
Figure 8.27. Distribution of GHQ-12 scores (n=89)	241
Figure 8.28. Overall score of GHQ-12 in 89 leprosy patients	243
Figure 8.29. GHQ-12 score of 3 or more in 89 leprosy patients	243

List of Appendices

Appendix 2. Imperial College London Ethical approval. 3 Appendix 3. FMR Ethical approval. 3 Appendix 4. Healthy volunteer's recruitment invitation letter. 3 Appendix 5. Healthy volunteer's recruitment invitation in Mindi 3 Appendix 6. Healthy volunteer's recruitment invitation in Marathi 3 Appendix 7. Healthy volunteer information sheet. 3 Appendix 8. Healthy volunteer information sheet – Hindi version. 3 Appendix 10. Healthy volunteer information sheet. 3 Appendix 10. Healthy volunteer information sheet. 3 Appendix 11. Patient's recruitment in Marathi 3 Appendix 12. Patient's recruitment in Marathi 3 Appendix 13. Patient information sheet. 3 Appendix 14. Patient information sheet. 3 Appendix 15. Patient information sheet. 3 Appendix 16. Healthy volunteer consent form – German version. 3 Appendix 17. Healthy volunteer consent form – Marathi version. 3 Appendix 19. Healthy volunteer consent form – Marathi version. 3 Appendix 19. Healthy volunteer consent form – Marathi version. 3 Appendix 20. Patient consent form – Hindi version. 3 Appendix 21. Patient consent form – H	Appendix 1. LSHTM Ethical approval	317
Appendix 3. FMR Ethical approval. 3 Appendix 4. Healthy volunteer's recruitment invitation letter. 3 Appendix 5. Healthy volunteer's recruitment invitation in Marathi. 3 Appendix 7. Healthy volunteer information sheet. 3 Appendix 8. Healthy volunteer information sheet. 3 Appendix 9. Healthy volunteer information sheet – Hindi version. 3 Appendix 10. Healthy volunteer information sheet – Hindi version. 3 Appendix 10. Healthy volunteer information sheet – Marathi version. 3 Appendix 11. Patient's recruitment in Marathi. 3 Appendix 12. Patient's recruitment in Marathi. 3 Appendix 13. Patient information sheet. 3 Appendix 14. Patient information sheet – Hindi version. 3 Appendix 15. Patient information sheet – Marathi version. 3 Appendix 16. Healthy volunteer consent form – German version. 3 Appendix 17. Healthy volunteer consent form – Hindi version. 3 Appendix 19. Healthy volunteer consent form – Marathi version. 3 Appendix 20. Patient consent form – Marathi version. 3 Appendix 21. Patient consent form – Marathi version. 3 Appendix 22. Patient consent form – Marathi version. 3	Appendix 2. Imperial College London Ethical approval	318
Appendix 4. Healthy volunteer's recruitment invitation letter 3 Appendix 5. Healthy volunteer's recruitment invitation in Marathi 3 Appendix 6. Healthy volunteer information sheet 3 Appendix 7. Healthy volunteer information sheet 3 Appendix 8. Healthy information sheet – German version 3 Appendix 10. Healthy volunteer information sheet – Marathi version 3 Appendix 11. Patient's recruitment in Hindi 3 Appendix 12. Patient's recruitment in Marathi 3 Appendix 13. Patient information sheet – Hindi version 3 Appendix 14. Patient information sheet – Hindi version 3 Appendix 15. Patient information sheet – Marathi version 3 Appendix 16. Healthy volunteer consent form 3 Appendix 17. Healthy volunteer consent form – German version 3 Appendix 19. Healthy volunteer consent form – Marathi version 3 Appendix 20. Patient consent form – Marathi version 3 Appendix 21. Patient consent form – Marathi version 3 Appendix 22. Patient consent form – Marathi version 3 Appendix 23. Patients record form (PRF) 3 Appendix 24. Modified 5 - points MRC scale for muscle strength scoring 3 Appendix	Appendix 3. FMR Ethical approval	319
Appendix 5. Healthy volunteer's recruitment invitation in Hindi 3 Appendix 6. Healthy volunteer's recruitment invitation in Marathi 3 Appendix 7. Healthy volunteer information sheet. 3 Appendix 8. Healthy information sheet – German version. 3 Appendix 9. Healthy volunteer information sheet – Hindi version 3 Appendix 10. Healthy volunteer information sheet – Marathi version. 3 Appendix 11. Patient's recruitment in Marathi 3 Appendix 12. Patient's recruitment in Marathi 3 Appendix 13. Patient information sheet. 33 Appendix 14. Patient information sheet. 33 Appendix 15. Patient information sheet. 33 Appendix 16. Healthy volunteer consent form 33 Appendix 17. Healthy volunteer consent form – German version 33 Appendix 19. Healthy volunteer consent form – Marathi version 33 Appendix 19. Healthy volunteer consent form – Marathi version 33 Appendix 20. Patient consent form – Marathi version 33 Appendix 21. Patient consent form – Marathi version 33 Appendix 22. Patient consent form – Marathi version 33 Appendix 23. Patients record form (PRF) 33 Appendix 24. Modified 5 - point	Appendix 4. Healthy volunteer's recruitment invitation letter	321
Appendix 6. Healthy volunteer's recruitment invitation in Marathi 3 Appendix 7. Healthy volunteer information sheet 3 Appendix 8. Healthy information sheet – German version. 3 Appendix 9. Healthy volunteer information sheet – Hindi version. 3 Appendix 10. Healthy volunteer information sheet – Marathi version. 3 Appendix 12. Patient's recruitment in Hindi. 3 Appendix 13. Patient information sheet – Hindi version. 3 Appendix 14. Patient information sheet – Marathi version. 3 Appendix 15. Patient information sheet – Marathi version. 3 Appendix 16. Healthy volunteer consent form – German version. 3 Appendix 17. Healthy volunteer consent form – Hindi version. 3 Appendix 18. Healthy volunteer consent form – Marathi version. 3 Appendix 20. Patient consent form – Marathi version. 3 Appendix 21. Patient consent form – Hindi version. 3 Appendix 22. Patient consent form – Marathi version. 3 Appendix 23. Patients record form (PRF). 3 Appendix 24. Modified 5 - points MRC scale for muscle strength scoring. 3 Appendix 30. DN4 Questionnaire – Marathi version. 3 Appendix 30. DN4 Questionnaire – Marathi version. 3	Appendix 5. Healthy volunteer's recruitment invitation in Hindi	322
Appendix 7. Healthy volunteer information sheet – German version 3 Appendix 8. Healthy information sheet – German version 3 Appendix 9. Healthy volunteer information sheet – Hindi version 3 Appendix 10. Healthy volunteer information sheet – Marathi version 3 Appendix 11. Patient's recruitment in Marathi 3 Appendix 12. Patient's recruitment in Marathi 3 Appendix 13. Patient information sheet – Hindi version 3 Appendix 14. Patient information sheet – Marathi version 3 Appendix 15. Patient information sheet – Marathi version 3 Appendix 16. Healthy volunteer consent form 3 Appendix 17. Healthy volunteer consent form – German version 3 Appendix 19. Healthy volunteer consent form – Marathi version 3 Appendix 20. Patient consent form Marathi version 3 Appendix 21. Patient consent form – Marathi version 3 Appendix 23. Patient secord form (PRF). 3 3 Appendix 24. Modified 5 - points MRC scale for muscle strength scoring 3 Appendix 25. Sensory testing using Semmes-Weinstein monofilaments 3 Appendix 26. Body chart 3 3 Appendix 27. Dermatomes of the body 3 <t< td=""><td>Appendix 6. Healthy volunteer's recruitment invitation in Marathi</td><td>323</td></t<>	Appendix 6. Healthy volunteer's recruitment invitation in Marathi	323
Appendix 8. Healthy information sheet – German version	Appendix 7. Healthy volunteer information sheet	324
Appendix 9. Healthy volunteer information sheet – Hindi version 3 Appendix 10. Healthy volunteer information sheet – Marathi version 3 Appendix 11. Patient's recruitment in Hindi. 3 Appendix 12. Patient's recruitment in Marathi 3 Appendix 13. Patient information sheet 33 Appendix 14. Patient information sheet 33 Appendix 15. Patient information sheet Marathi version 3 Appendix 16. Healthy volunteer consent form 33 Appendix 17. Healthy volunteer consent form German version 33 Appendix 18. Healthy volunteer consent form Marathi version 33 Appendix 20. Patient consent form Marathi version 33 Appendix 21. Patient consent form Marathi version 33 Appendix 22. Patient consent form – Hindi version 33 Appendix 23. Patients record form (PRF) 33 Appendix 24. Modified 5 - points MRC scale for muscle strength scoring 33 Appendix 24. Modified 5 - points MRC scale for muscle strength scoring 33 Appendix 25. Sensory testing using Semmes-Weinstein monofilaments 33 Appendix 26. Body chart 33 Appendix 26. Body chart 33 Appendix 31. PainDETECT Questionnaire 33 <t< td=""><td>Appendix 8. Healthy information sheet - German version</td><td>327</td></t<>	Appendix 8. Healthy information sheet - German version	327
Appendix 10. Healthy volunteer information sheet – Marathi version. 3 Appendix 11. Patient's recruitment in Hindi. 3 Appendix 12. Patient's recruitment in Marathi 3 Appendix 13. Patient information sheet. 3 Appendix 14. Patient information sheet – Hindi version 3 Appendix 15. Patient information sheet – Marathi version 3 Appendix 16. Healthy volunteer consent form 3 Appendix 17. Healthy volunteer consent form – German version 3 Appendix 19. Healthy volunteer consent form – Hindi version 3 Appendix 20. Patient consent form – Marathi version 3 Appendix 21. Patient consent form – Hindi version 3 Appendix 22. Patient consent form – Marathi version 3 Appendix 23. Patients record form (PRF) 3 Appendix 24. Modified 5 - points MRC scale for muscle strength scoring 3 Appendix 25. Sensory testing using Semmes-Weinstein monofilaments 3 Appendix 26. Body chart 3 Appendix 27. Dermatomes of the body 3 Appendix 30. DN4 Questionnaire – Hindi version 3 Appendix 31. PainDETECT Questionnaire – Hindi version 3 Appendix 32. PainDETECT Questionnaire – Hindi version 3	Appendix 9. Healthy volunteer information sheet - Hindi version	330
Appendix 11. Patient's recruitment in Hindi	Appendix 10. Healthy volunteer information sheet - Marathi version	333
Appendix 12. Patient's recruitment in Marathi 3 Appendix 13. Patient information sheet 33 Appendix 14. Patient information sheet – Hindi version 33 Appendix 15. Patient information sheet – Marathi version 33 Appendix 16. Healthy volunteer consent form 36 Appendix 17. Healthy volunteer consent form – German version 33 Appendix 18. Healthy volunteer consent form – Marathi version 33 Appendix 20. Patient consent form – Marathi version 33 Appendix 21. Patient consent form – Marathi version 33 Appendix 22. Patient consent form – Marathi version 33 Appendix 23. Patients record form (PRF) 37 Appendix 24. Modified 5 - points MRC scale for muscle strength scoring 33 Appendix 25. Sensory testing using Semmes-Weinstein monofilaments 33 Appendix 26. Body chart 33 Appendix 30. DN4 Questionnaire 33 Appendix 31. PainDETECT Questionnaire 33 Appendix 32. PainDETECT Questionnaire – Marathi version 33 Appendix 33. PainDETECT Questionnaire – Hindi version 33 Appendix 34. Brief Pain Inventory Questionnaire – Marathi version 33 Appendix 35. Brief Pain Inventory Questionnaire – Hindi versio	Appendix 11. Patient's recruitment in Hindi	336
Appendix 13. Patient information sheet 33 Appendix 14. Patient information sheet – Hindi version 33 Appendix 15. Patient information sheet – Marathi version 33 Appendix 15. Patient information sheet – Marathi version 33 Appendix 16. Healthy volunteer consent form German version 3 Appendix 17. Healthy volunteer consent form – German version 33 Appendix 18. Healthy volunteer consent form – Marathi version 33 Appendix 20. Patient consent form – Marathi version 33 Appendix 21. Patient consent form – Marathi version 33 Appendix 22. Patient consent form – Marathi version 33 Appendix 23. Patients record form (PRF) 33 Appendix 24. Modified 5 - points MRC scale for muscle strength scoring 33 Appendix 25. Sensory testing using Semmes-Weinstein monofilaments 33 Appendix 27. Dermatomes of the body 34 Appendix 29. DN4 Questionnaire – Hindi version 33 Appendix 30. DN4 Questionnaire – Marathi version 33 Appendix 31. PainDETECT Questionnaire – Marathi version 33 Appendix 32. PainDETECT Questionnaire – Hindi version 33 Appendix 33. PainDETECT Questionnaire – Marathi version 33	Appendix 12. Patient's recruitment in Marathi	337
Appendix 14. Patient information sheet – Hindi version3Appendix 15. Patient information sheet – Marathi version3Appendix 16. Healthy volunteer consent form3Appendix 17. Healthy volunteer consent form – German version3Appendix 18. Healthy volunteer consent form – Hindi version3Appendix 19. Healthy volunteer consent form – Marathi version3Appendix 20. Patient consent form – Marathi version3Appendix 21. Patient consent form – Hindi version3Appendix 22. Patient consent form – Marathi version3Appendix 23. Patient consent form – Marathi version3Appendix 24. Modified 5 - points MRC scale for muscle strength scoring3Appendix 25. Sensory testing using Semmes-Weinstein monofilaments3Appendix 27. Dermatomes of the body3Appendix 29. DN4 Questionnaire3Appendix 30. DN4 Questionnaire – Hindi version3Appendix 31. PainDETECT Questionnaire – Hindi version3Appendix 33. PainDETECT Questionnaire – Marathi version3Appendix 34. Brief Pain Inventory Questionnaire – Hindi version3Appendix 35. Brief Pain Inventory Questionnaire – Hindi version3Appendix 36. Brief Pain Inventory Questionnaire – Marathi version3Appendix 37. General Health Questionnaire – Marathi version3Appendix 36. Brief Pain Inventory Questionnaire – Marathi version3Appendix 37. General Health Questionnaire – Marathi version3Appendix 39. General Health Questionnaire – Marathi version3Appendix 39. General Health Question	Appendix 13. Patient information sheet	338
Appendix 15. Patient information sheet – Marathi version 3 Appendix 16. Healthy volunteer consent form 3 Appendix 17. Healthy volunteer consent form – German version 3 Appendix 18. Healthy volunteer consent form – Hindi version 3 Appendix 19. Healthy volunteer consent form – Marathi version 3 Appendix 20. Patient consent form – Hindi version 3 Appendix 21. Patient consent form – Hindi version 3 Appendix 22. Patient consent form – Marathi version 3 Appendix 23. Patients record form (PRF) 3 Appendix 24. Modified 5 - points MRC scale for muscle strength scoring 3 Appendix 25. Sensory testing using Semmes-Weinstein monofilaments 3 Appendix 26. Body chart 3 Appendix 28. DN4 Questionnaire 3 Appendix 30. DN4 Questionnaire – Hindi version 3 Appendix 31. PainDETECT Questionnaire – Marathi version 3 Appendix 33. PainDETECT Questionnaire – Marathi version 3 Appendix 34. Brief Pain Inventory Questionnaire – Hindi version 3 Appendix 35. Brief Pain Inventory Questionnaire – Hindi version 3 Appendix 36. Brief Pain Inventory Questionnaire – Marathi version 3 Appendix 36. Brief Pain I	Appendix 14. Patient information sheet - Hindi version	341
Appendix 16. Healthy volunteer consent form 3 Appendix 17. Healthy volunteer consent form 9 Appendix 18. Healthy volunteer consent form 9 Appendix 19. Healthy volunteer consent form 9 Appendix 20. Patient consent form 9 Appendix 21. Patient consent form 9 Appendix 22. Patient consent form 9 Appendix 23. Patients record form (PRF) 3 Appendix 24. Modified 5 - points MRC scale for muscle strength scoring 3 Appendix 25. Sensory testing using Semmes-Weinstein monofilaments 3 Appendix 26. Body chart 3 Appendix 29. DN4 Questionnaire 3 Appendix 30. DN4 Questionnaire 3 Appendix 31. PainDETECT Questionnaire 3 Appendix 32. PainDETECT Questionnaire 3 Appendix 33. PainDETECT Questionnaire 3 Appendix 34. Brief Pain Inventory Questionnaire (BPI) 3 Appendix 35. Brief Pain Inventory Questionnaire 3 Appendix 36. Brief Pain Inventory Questionnaire 3 Appendix 37. General Health Questionnaire – Hindi version 3 Appendix 36. Brief Pain Inventory Questionnaire – Marathi version 3 Appendix 37. Ge	Appendix 15. Patient information sheet - Marathi version	343
Appendix 17. Healthy volunteer consent form – German version 33 Appendix 18. Healthy volunteer consent form – Hindi version 33 Appendix 19. Healthy volunteer consent form – Marathi version 33 Appendix 20. Patient consent form – Hindi version 33 Appendix 21. Patient consent form – Hindi version 33 Appendix 22. Patient consent form – Marathi version 33 Appendix 23. Patients record form (PRF) 33 Appendix 24. Modified 5 - points MRC scale for muscle strength scoring 33 Appendix 25. Sensory testing using Semmes-Weinstein monofilaments 33 Appendix 26. Body chart 33 Appendix 27. Dermatomes of the body 33 Appendix 28. DN4 Questionnaire 33 Appendix 29. DN4 Questionnaire – Hindi version 33 Appendix 30. DN4 Questionnaire – Marathi version 33 Appendix 31. PainDETECT Questionnaire – Hindi version 33 Appendix 32. PainDETECT Questionnaire – Hindi version 33 Appendix 34. Brief Pain Inventory Questionnaire – Hindi version 33 Appendix 35. Brief Pain Inventory Questionnaire – Hindi version 33 Appendix 36. Brief Pain Inventory Questionnaire – Marathi version 33 Appendix 37. General Healt	Appendix 16. Healthy volunteer consent form	345
Appendix 18. Healthy volunteer consent form – Hindi version 3 Appendix 19. Healthy volunteer consent form – Marathi version 3 Appendix 20. Patient consent form – Hindi version 3 Appendix 21. Patient consent form – Marathi version 3 Appendix 22. Patient consent form – Marathi version 3 Appendix 23. Patients record form (PRF) 3 Appendix 24. Modified 5 - points MRC scale for muscle strength scoring 3 Appendix 25. Sensory testing using Semmes-Weinstein monofilaments 3 Appendix 26. Body chart 3 Appendix 27. Dermatomes of the body 3 Appendix 29. DN4 Questionnaire 3 Appendix 30. DN4 Questionnaire – Hindi version 3 Appendix 31. PainDETECT Questionnaire – Marathi version 3 Appendix 32. PainDETECT Questionnaire – Marathi version 3 Appendix 33. PainDETECT Questionnaire – Marathi version 3 Appendix 34. Brief Pain Inventory Questionnaire – Hindi version 3 Appendix 35. Brief Pain Inventory Questionnaire – Hindi version 3 Appendix 36. Brief Pain Inventory Questionnaire – Marathi version 3 Appendix 37. General Health Questionnaire – Hindi version 3 Appendix 38. General Health Questio	Appendix 17. Healthy volunteer consent form - German version	346
Appendix 19. Healthy volunteer consent form – Marathi version 3 Appendix 20. Patient consent form – Hindi version 3 Appendix 21. Patient consent form – Marathi version 3 Appendix 22. Patient consent form – Marathi version 3 Appendix 23. Patients record form (PRF). 3 Appendix 24. Modified 5 - points MRC scale for muscle strength scoring 3 Appendix 25. Sensory testing using Semmes-Weinstein monofilaments 3 Appendix 26. Body chart 3 Appendix 27. Dermatomes of the body 3 Appendix 29. DN4 Questionnaire 3 Appendix 30. DN4 Questionnaire – Hindi version 3 Appendix 31. PainDETECT Questionnaire – Marathi version 3 Appendix 32. PainDETECT Questionnaire – Marathi version 3 Appendix 33. PainDETECT Questionnaire – Hindi version 3 Appendix 34. Brief Pain Inventory Questionnaire (BPI) 3 Appendix 35. Brief Pain Inventory Questionnaire – Marathi version 3 Appendix 37. General Health Questionnaire – Hindi version 3 Appendix 38. General Health Questionnaire – Hindi version 3 Appendix 39. General Health Questionnaire – Hindi version 3 Appendix 39. General Health Questionnaire – Hindi vers	Appendix 18. Healthy volunteer consent form - Hindi version	347
Appendix 20. Patient consent form – Hindi version 3 Appendix 21. Patient consent form – Marathi version 3 Appendix 22. Patient consent form – Marathi version 3 Appendix 23. Patients record form (PRF) 3 Appendix 24. Modified 5 - points MRC scale for muscle strength scoring 3 Appendix 25. Sensory testing using Semmes-Weinstein monofilaments 3 Appendix 26. Body chart 3 Appendix 27. Dermatomes of the body 3 Appendix 28. DN4 Questionnaire 3 Appendix 29. DN4 Questionnaire – Hindi version 3 Appendix 30. DN4 Questionnaire – Marathi version 3 Appendix 31. PainDETECT Questionnaire – Marathi version 3 Appendix 32. PainDETECT Questionnaire – Marathi version 3 Appendix 33. PainDETECT Questionnaire – Marathi version 3 Appendix 34. Brief Pain Inventory Questionnaire (BPI) 3 Appendix 35. Brief Pain Inventory Questionnaire – Marathi version 3 Appendix 37. General Health Questionnaire – Hindi version 3 Appendix 39. General Health Questionnaire – Hindi version 3 Appendix 39. General Health Questionnaire – Marathi version 3 Appendix 39. General Health Questionnaire – Marathi version	Appendix 19. Healthy volunteer consent form - Marathi version	348
Appendix 21. Patient consent form – Hindi version 3 Appendix 22. Patient consent form – Marathi version 3 Appendix 23. Patients record form (PRF) 3 Appendix 24. Modified 5 - points MRC scale for muscle strength scoring 3 Appendix 25. Sensory testing using Semmes-Weinstein monofilaments 3 Appendix 26. Body chart 3 Appendix 27. Dermatomes of the body 3 Appendix 28. DN4 Questionnaire 3 Appendix 29. DN4 Questionnaire – Hindi version 3 Appendix 30. DN4 Questionnaire – Marathi version 3 Appendix 31. PainDETECT Questionnaire – Hindi version 3 Appendix 32. PainDETECT Questionnaire – Marathi version 3 Appendix 33. PainDETECT Questionnaire – Marathi version 3 Appendix 34. Brief Pain Inventory Questionnaire – Marathi version 3 Appendix 35. Brief Pain Inventory Questionnaire – Hindi version 3 Appendix 36. Brief Pain Inventory Questionnaire – Marathi version 3 Appendix 37. General Health Questionnaire – Marathi version 3 Appendix 38. General Health Questionnaire – Hindi version 3 Appendix 39. General Health Questionnaire – Marathi version 3 Appendix 39. General Health Questionnair	Appendix 20. Patient consent form	349
Appendix 22. Patient consent form – Marathi version3Appendix 23. Patients record form (PRF)3Appendix 24. Modified 5 - points MRC scale for muscle strength scoring3Appendix 25. Sensory testing using Semmes-Weinstein monofilaments3Appendix 26. Body chart3Appendix 27. Dermatomes of the body3Appendix 28. DN4 Questionnaire3Appendix 29. DN4 Questionnaire – Hindi version3Appendix 30. DN4 Questionnaire – Marathi version3Appendix 31. PainDETECT Questionnaire3Appendix 32. PainDETECT Questionnaire – Marathi version3Appendix 33. PainDETECT Questionnaire – Marathi version3Appendix 34. Brief Pain Inventory Questionnaire (BPI)3Appendix 35. Brief Pain Inventory Questionnaire – Hindi version3Appendix 36. Brief Pain Inventory Questionnaire – Marathi version3Appendix 37. General Health Questionnaire – Marathi version3Appendix 38. General Health Questionnaire – Hindi version3Appendix 39. General Health Questionnaire – Hindi version3Appendix 34. Brief Pain Inventory Questionnaire – Marathi version3Appendix 37. General Health Questionnaire – Marathi version3Appendix 38. General Health Questionnaire – Hindi version3Appendix 39. General Health Questionnaire – Marathi version3Appendix 39. General Health Questionnaire – Marathi version3Appendix 39. Verbal instructions for performing OST-DENS protocol4	Appendix 21. Patient consent form - Hindi version	350
Appendix 23. Patients record form (PRF)	Appendix 22. Patient consent form – Marathi version	352
Appendix 24. Modified 5 - points MRC scale for muscle strength scoring3Appendix 25. Sensory testing using Semmes-Weinstein monofilaments3Appendix 26. Body chart3Appendix 27. Dermatomes of the body3Appendix 28. DN4 Questionnaire3Appendix 29. DN4 Questionnaire – Hindi version3Appendix 30. DN4 Questionnaire – Marathi version3Appendix 31. PainDETECT Questionnaire – Hindi version3Appendix 32. PainDETECT Questionnaire – Hindi version3Appendix 33. PainDETECT Questionnaire – Marathi version3Appendix 34. Brief Pain Inventory Questionnaire – Marathi version3Appendix 35. Brief Pain Inventory Questionnaire – Hindi version3Appendix 36. Brief Pain Inventory Questionnaire – Marathi version3Appendix 37. General Health Questionnaire – Hindi version3Appendix 38. General Health Questionnaire – Hindi version3Appendix 39. General Health Questionnaire – Marathi version4Appendix 39. Verbal instructions for performing OST-DENS protocol4	Appendix 23. Patients record form (PRF)	355
Appendix 25. Sensory testing using Semmes-Weinstein monofilaments3Appendix 26. Body chart3Appendix 27. Dermatomes of the body3Appendix 28. DN4 Questionnaire3Appendix 29. DN4 Questionnaire – Hindi version3Appendix 30. DN4 Questionnaire – Marathi version3Appendix 31. PainDETECT Questionnaire3Appendix 32. PainDETECT Questionnaire – Hindi version3Appendix 33. PainDETECT Questionnaire – Marathi version3Appendix 34. Brief Pain Inventory Questionnaire (BPI)3Appendix 35. Brief Pain Inventory Questionnaire – Marathi version3Appendix 36. Brief Pain Inventory Questionnaire – Marathi version3Appendix 37. General Health Questionnaire – Hindi version3Appendix 39. General Health Questionnaire – Hindi version3Appendix 39. General Health Questionnaire – Hindi version4Appendix 39. General Health Questionnaire – Marathi version4Appendix 39. General Health Questionnaire – Marathi version4Appendix 40. Verbal instructions for performing OST-DENS protocol4	Appendix 24. Modified 5 - points MRC scale for muscle strength scoring	381
Appendix 26. Body chart3Appendix 27. Dermatomes of the body3Appendix 28. DN4 Questionnaire3Appendix 29. DN4 Questionnaire – Hindi version3Appendix 30. DN4 Questionnaire – Marathi version3Appendix 31. PainDETECT Questionnaire3Appendix 32. PainDETECT Questionnaire – Hindi version3Appendix 33. PainDETECT Questionnaire – Marathi version3Appendix 34. Brief Pain Inventory Questionnaire (BPI)3Appendix 35. Brief Pain Inventory Questionnaire – Hindi version3Appendix 36. Brief Pain Inventory Questionnaire – Marathi version3Appendix 37. General Health Questionnaire (GHQ-12)3Appendix 39. General Health Questionnaire – Hindi version3Appendix 39. General Health Questionnaire – Marathi version4Appendix 40. Verbal instructions for performing OST-DENS protocol4	Appendix 25. Sensory testing using Semmes-Weinstein monofilaments	382
Appendix 27. Dermatomes of the body3Appendix 28. DN4 Questionnaire3Appendix 29. DN4 Questionnaire – Hindi vestion3Appendix 30. DN4 Questionnaire – Marathi version3Appendix 31. PainDETECT Questionnaire3Appendix 32. PainDETECT Questionnaire – Hindi version3Appendix 33. PainDETECT Questionnaire – Marathi vestion3Appendix 34. Brief Pain Inventory Questionnaire (BPI)3Appendix 35. Brief Pain Inventory Questionnaire – Hindi version3Appendix 36. Brief Pain Inventory Questionnaire – Marathi version3Appendix 37. General Health Questionnaire (GHQ-12)3Appendix 39. General Health Questionnaire – Hindi version3Appendix 39. General Health Questionnaire – Marathi version4Appendix 40. Verbal instructions for performing OST-DENS protocol4	Appendix 26. Body chart	383
Appendix 28. DN4 Questionnaire	Appendix 27. Dermatomes of the body	384
Appendix 29. DN4 Questionnaire – Hindi version3Appendix 30. DN4 Questionnaire – Marathi version3Appendix 31. PainDETECT Questionnaire – Hindi version3Appendix 32. PainDETECT Questionnaire – Hindi version3Appendix 33. PainDETECT Questionnaire – Marathi vesrion3Appendix 34. Brief Pain Inventory Questionnaire (BPI)3Appendix 35. Brief Pain Inventory Questionnaire – Hindi version3Appendix 36. Brief Pain Inventory Questionnaire – Marathi version3Appendix 37. General Health Questionnaire (GHQ-12)3Appendix 38. General Health Questionnaire – Hindi version3Appendix 39. General Health Questionnaire – Marathi version4Appendix 40. Verbal instructions for performing OST-DENS protocol4	Appendix 28. DN4 Questionnaire	386
Appendix 30. DN4 Questionnaire – Marathi version	Appendix 29. DN4 Questionnaire - Hindi vesrion	387
Appendix 31. PainDETECT Questionnaire3Appendix 32. PainDETECT Questionnaire – Hindi version3Appendix 33. PainDETECT Questionnaire – Marathi version3Appendix 34. Brief Pain Inventory Questionnaire (BPI)3Appendix 35. Brief Pain Inventory Questionnaire – Hindi version3Appendix 36. Brief Pain Inventory Questionnaire – Marathi version3Appendix 37. General Health Questionnaire (GHQ-12)3Appendix 38. General Health Questionnaire – Hindi version3Appendix 39. General Health Questionnaire – Marathi version4Appendix 40. Verbal instructions for performing OST-DENS protocol4	Appendix 30. DN4 Questionnaire - Marathi version	388
Appendix 32. PainDETECT Questionnaire – Hindi version3Appendix 33. PainDETECT Questionnaire – Marathi vesrion3Appendix 34. Brief Pain Inventory Questionnaire (BPI)3Appendix 35. Brief Pain Inventory Questionnaire – Hindi version3Appendix 36. Brief Pain Inventory Questionnaire – Marathi version3Appendix 37. General Health Questionnaire (GHQ-12)3Appendix 38. General Health Questionnaire – Hindi version3Appendix 39. General Health Questionnaire – Marathi version4Appendix 40. Verbal instructions for performing OST-DENS protocol4	Appendix 31. PainDETECT Questionnaire	389
Appendix 33. PainDETECT Questionnaire – Marathi vestion	Appendix 32. PainDETECT Questionnaire - Hindi version	390
Appendix 34. Brief Pain Inventory Questionnaire (BPI)3Appendix 35. Brief Pain Inventory Questionnaire – Hindi version3Appendix 36. Brief Pain Inventory Questionnaire – Marathi version3Appendix 37. General Health Questionnaire (GHQ-12)3Appendix 38. General Health Questionnaire – Hindi version3Appendix 39. General Health Questionnaire – Marathi version4Appendix 40. Verbal instructions for performing OST-DENS protocol4	Appendix 33. PainDETECT Questionnaire - Marathi vesrion	391
Appendix 35. Brief Pain Inventory Questionnaire – Hindi version3Appendix 36. Brief Pain Inventory Questionnaire – Marathi version3Appendix 37. General Health Questionnaire (GHQ-12)3Appendix 38. General Health Questionnaire – Hindi version3Appendix 39. General Health Questionnaire – Marathi version4Appendix 40. Verbal instructions for performing OST-DENS protocol4	Appendix 34. Brief Pain Inventory Questionnaire (BPI)	392
Appendix 36. Brief Pain Inventory Questionnaire – Marathi version 33 Appendix 37. General Health Questionnaire (GHQ-12) 33 Appendix 38. General Health Questionnaire – Hindi version 33 Appendix 39. General Health Questionnaire – Marathi version 44 Appendix 40. Verbal instructions for performing OST-DENS protocol 44	Appendix 35. Brief Pain Inventory Questionnaire - Hindi version	394
Appendix 37. General Health Questionnaire (GHQ-12)	Appendix 36. Brief Pain Inventory Questionnaire - Marathi version	396
Appendix 38. General Health Questionnaire – Hindi version	Appendix 37. General Health Questionnaire (GHQ-12)	398
Appendix 39. General Health Questionnaire – Marathi version	Appendix 38. General Health Questionnaire - Hindi version	399
Appendix 40 Verbal instructions for performing OST-DENS protocol 4	Appendix 39. General Health Questionnaire – Marathi version	400
rippendix for verbar instructions for performing QoT DT to protocommunity	Appendix 40. Verbal instructions for performing QST-DFNS protocol	402

A	100
Appendix 41. QS1-DFNS protocol – Hindi Version	
Appendix 42. QST-DFNS protocol – Marathi version	412
Appendix 43. Participant's state of health	418
Appendix 44. Recording sheet for QST	
Appendix 45. Publications and Presentations	
Appendix 46. Univaiate logistic regression - Neuropathic pain	
Appendix 47. Multivariate logistic regression - Neuropathic pain	
Appendix 48. Univalate logistic regression – Depression	
Appendix 49. Multivariate logistic regression – Depression	
Appendix 50. Pain-related definitions (IASP Taxonomy, 2014)	
Appendix 51. Neuropathic pain workshop – Mumbai	
Appendix 52. DFNS-QST translation – Mumbai	435
Appendix 53. Studies on amitriptyline for the treatment of neuropathic pain	436
Appendix 54. Thesis timeline	439

List of Abbreviations

AFB	Acid-fast bacilli
AIDS	Acquired Immunodeficiency Syndrome
ALERT	All Africa Leprosy, Tuberculosis and Rehabilitation Training Centre
AMFES	ALERT MDT Field Evaluation Study
ANOVA	Analysis of Variance
APC	Antigen-presenting cell
BAND	Bangladesh Acute Nerve Damage Study
BB	Borderline borderline
BCG	Bacille Calmette-Guerin
BI	Bacterial index
BL	Borderline lepromatous
BLP	Bombay Leprosy Project
BPI	Brief Pain Inventory
ВТ	Borderline tuberculoid
CDT	Cold Detection Threshold
СРТ	Cold Pain Threshold
CI	Confidence Interval
CMI	Cell mediated immunity
DFNS	German Research Network on Neuropathic Pain
DM	Diabetes Mellitus
DMA	Dynamic Mechanical Allodynia
DN4	Douleur Neuropathique en 4 Questions
DNA	Deoxyribonucleic acid
ENL	Erythema Nodosum Leprosum

FDA	Food and Drug Administration
FMOH	Federal Ministry of Health
FMR	Foundation for Medical Research
g, mg, ng, kg	Gram, milligram, nanogram, kilogram
GCP	Good Clinical Practice
GHQ	General Health Questionnaire
GHQ-12	General Health Questionnaire-12
НС	Healthy Control
HHSG	The Hospital and Homes of St. Giles
HIV	Human Immunodeficiency Virus
HRQOL	Health Related Quality of Life
НРТ	Heat Pain Threshold
HV	Healthy Volunteer
IASP	International Association for the Study of Pain
IENFD	Intra-Epidermal Nerve Fibre Density
ICH	International Conference on Harmonisation of Technical Requirements for Registration of Pharmaceuticals for Human Use
ICL	Imperial College London
IFNγ	Interferon-gamma
IL	Interleukin
ILEP	International Federation of Anti-Leprosy Associations
INFIR	ILEP Nerve Function Impairment and Reaction
IQR	Inter-quartile range
kPa	Kilopascal
LANSS	Leeds Assessment of Neuropathic Symptoms and Signs
LL	Lepromatous leprosy
LSHTM	London School of Hygiene and Tropical Medicine

Abbreviations

MB	Multibacillary
MDT	Multi-drug therapy
MDT	Mechanical Detection Threshold
MF	Monofilament
MHC	Major histocompatibility complex
mN	Millinewton
MPS	Mechanical Pain Sensitivity
MPT	Mechanical Pain Threshold
MRC	Medical Research Council
MRI	Magnetic Resonance Imaging
NCS	Nerve Conduction Study
NeuPSIG	Neuropathic Pain Special Interesting Group
NFI	Nerve function impairment
NP	Neuropathic Pain
NRS	Numeric Rating Scale
OR	Odds Ratio
РВ	Paucibacillary
PCR	Polymerase chain reaction
PD-Q	PainDETECT Questionnaire
PHN	Post Herpetic Neuralgia
PHS	Paradoxical Heat Sensation
PI	Principal investigator
PN	Peripheral Neuropathy
PNL	Pure neuritic leprosy
РРТ	Pressure Pain Threshold
PYAR	Person-years at risk
1	

QOL	Quality of Life		
QST	Quantitative Sensory Test		
RCTs	Randomised Controlled Trials		
RFT	Release from treatment		
R-J	Ridley Jopling		
RR	Reversal Reaction		
SD	Standard deviation		
SF-36	Short Form-36 Health Questionnaire		
SW	Semmes-Weinstein		
SWM	Semmes-Weinstein monofilaments		
T1R	Type 1 reaction		
ТВ	Tuberculosis		
TCAs	Tricyclic Antidepressants		
TENLEP	Treatment of Early Neuropathy in LEProsy		
TNF-α	Tumour necrosis factor alpha		
TRIPOD	Trial in Prevention of Disability study		
TSA	Thermal Sensory Analyzer		
TSL	Thermal Sensory Limen		
ТТ	Tuberculoid		
VAS	Visual Analogue Scale		
VDT	Vibration Detection Threshold		
VMT	Voluntary muscle test		
WDT	Warm Detection Threshold		
WHO	World Health Organization		
WUR	Wind Up Ratio		

Chapter 1 Introduction

1.1 PhD approach

This PhD work was a collaborative study between the leprosy group at the London School of Hygiene and Tropical Medicine (LSHTM) and the pain group at Imperial College London (ICL). Our international collaborators are from the London Pain Consortium (LPC) and the German Research Network on Neuropathic Pain (DFNS), and our local collaborators from India are: the Foundation for Medical Research (FMR) and Bombay Leprosy Project (BLP).

I came to LSHTM to study for an MSc in Tropical Medicine and International Health (TMIH) in 2007. I was supervised by Professor Lockwood and Professor Rice for my dissertation, which was on leprosy and neuropathic pain (NP). Since that time I have been interested in leprosy and research.

After I completed my MSc, and throughout my work as researcher at ALERT (The Leprosy Centre in Ethiopia), and at the Neglected Tropical Diseases at Federal Ministry of Health, Sudan, I noticed that although neglected tropical diseases are essentially preventable, little is done to control important endemic diseases such as Leishmaniasis, HIV/AIDS and Leprosy. The morbidity and mortality from infectious diseases are alarming. Since that time I have felt one of my responsibilities is to work for improvements in this situation.

Leprosy is still an endemic disease in developing countries. Management of leprosy (diagnosis and treatment) is still challenging. Antimicrobial therapy kills *M. leprae*, but immune mediated peripheral nerve damage can continue long after effective drug treatment. Consequently, as a result of persistent damage, patients who have been successfully treated with multi-drug therapy (MDT), suffer from NP and associated co-morbidity long after the infection has been cured. NP is now being recognized as an important long-term complication in a proportion of people previously treated for leprosy. Considering the high prevalence and morbidity of NP in treated leprosy patients, there is a pressing need for clinical trial evidence to assess the efficacy of pain therapies in the management of leprosy-associated NP.

This PhD work was conducted in Mumbai, India. The rationale for the study emerged in recognition of the growing challenges of NP, as well as a gap in the knowledge regarding identification and treatment of leprosy patients with NP, particularly in resource-limited settings. Preliminary discussion with leprosy and pain experts revealed that no single drug for the treatment approach stood out as a potential intervention study on NP in leprosy. Instead, key experts highlighted the need to accurately characterise the somatosensory profiles of leprosy patients with NP, in order to develop appropriate tools for the investigation and design of suitable treatment interventions. Preliminary work was then carried out to define the scope and design of the main PhD study of profiling leprosy NP patients. This information helped formulate the objective of the PhD work as conducting a case-control study using highly specialised tools to investigate the clinical characteristics of leprosy patients with NP and to inform development tools in the furure. As part of this PhD study, two validation studies were conducted in London and Mumbai. Validating the Mumbai site for study protocols in Hindi and Marathi, and establishing leprosy profiles in resource-limited setting were important preliminary steps to develop an optimal trial design to use for future studies.

1.2 PhD thesis timeline

I started my PhD work in October 2010. Recruitment for the validation study in London began in April 2011 and was completed in July 2011. During this time I trained and qualified in the use of the German Research Network on Neuropathic Pain DFNS-QST protocol (Appendix 40). I successfully upgraded from MPhil to PhD in November 2011. Recruitment for the local healthy volunteers study and the PhD main study took place between October 2012 and June 2013. Data recording, management, and writing up began in July 2013 and the thesis was submitted for examination in August 2014. The detailed timeline of my PhD thesis is shown in (Appendix 54).

1.3 Overview of my role in this PhD research

My roles in this study included writing the grant proposal, and designing the studies. I was responsible for the design and writing of study forms including consent forms and information sheets for both healthy volunteers and patients. I developed the study protocols with advice from Professor Lockwood, Professor Rice, Dr Nichols, Professor Maier, Professor Treede R-D, Dr Bennett, Dr Pfau, Dr Pai, Dr Shetty, Dr Walker and Dr Tudor. I was responsible for obtaining various ethical approvals. In addition, all

study work including patient recruitment, screening and enrolment procedures, clinical examination, administration of testing protocols, data recording and management presented in this thesis were performed by me, unless otherwise indicated. In particular, I conducted interviews in the local language through an interpreter. The study on the use of Semmes-Weinstein monofilaments to simplify the DFNS-QST protocol was conducted in collaboration with the DFNS group; I conducted the study and Dr Pfau completed the analysis and wrote up the findings. For the PhD main study, I conducted the data analysis with advice from Dr Nichols and Dr Tudor.

1.4 Structure of the PhD thesis

This thesis is organised into ten chapters:

Chapter one presents introductory and background information regarding NP in leprosy. It describes how the thesis is set-up, and introduces the reader to the burden of NP in leprosy patients. It then outlines the reasons for carrying out this research as well as the objectives of this study. Chapter two, the literature review, presents the current literature on leprosy, its complications and management; and looking particular at the central role of nerve damage in leprosy. Chapter three, the identification of neuropathic pain, gives a general introduction on the NP theme and, in particular, NP in leprosy. In chapter four the instruments for assessing sensory abnormalities, pain disorders and techniques for assessing the psychological impact of NP are assessed. Chapter five describes the clinical difficulties in identifying NP in diseases such as leprosy, with some current thoughts on NP pain assessment and the need to develop tools for the assessment of NP in leprosy. The background information from the above mentioned chapters aims to place the rationale of the PhD work in the context of current knowledge and identify a gap in existing research, thus providing a framework for this study.

Chapter six describes the study designed methods and chapter seven describes the validation studies. These include the validation of the investigator, the centre and study protocol. The aim of the first validation study was to validate the investigator in the London centre and the aim of the second was to validate the DFNS-QST protocol in the local Indian population and the site centre. The third validation study explored the utility of Brazilian Semmes-Weinstein monofilaments and aimed to simplify the DFNS-QST protocol. For these studies only healthy controls were recruited. This information

provided the basis of validating the investigator and the centre site in Mumbai for the use of DFNS-QST protocol.

Chapters eight and nine describe and discuss various aspects of the results. These include the baseline findings on the leprosy patients; the difference between healthy controls and the patient group; the assessment using pain questionnaires, clinical examination, quantitative sensory testing and psychological assessment techniques; and the role of somatosensory profiling in the assessment of leprosy-related NP.

Finally, based on the findings from this PhD work, the main summary of the findings and recommendations for future studies are presented in the last chapter, chapter ten.

1.5 Rationale of the PhD thesis

NP is a severe form of chronic pain caused by a lesion or disease of the somatosensory system and is associated with many diseases (Jensen et al., 2011). Leprosy, the most common cause of treatable neuropathy worldwide (Scollard et al., 2006), is now being recognised as one such disease. The Lancet Neurology highlighted the importance of leprosy as a neurological disease in an editorial in 2009 (Lancet, 2009), and we have recently shown in India and Ethiopia that 17-20% of treated leprosy patients cured of their infection but left with peripheral nerve damage, have significant NP, which is associated with reduced quality of life (Lasry-Levy et al., 2011, Haroun et al., 2012). Previous work has shown that patients with NP have significant levels of depression (Jensen et al., 2007, Doth et al., 2010). This makes NP one of the most prevalent disabilities among leprosy patients and thus a significant health problem worldwide. Thus, the accurate identification of NP in patients with chronic pain is crucial for targeting appropriate treatment as NP conditions require a different therapeutic approach from other pain types.

The German Research Network on Neuropathic Pain (DFNS) has developed a standardised QST battery that consists of 7 tests measuring 13 parameters (Rolke et al., 2006a, Rolke et al., 2006b), which may identify changes in sensory parameters related to chronic NP. This protocol has been used successfully to establish the somatosensory profile of patients with HIV-associated sensory neuropathy (Phillips et al., 2014). It has been also used in patients with different NP conditions such as post-herpetic neuralgia and diabetic mellitus, where a subgroup of patients with different somatosensory

profiles have been identified (Maier et al., 2010). Although NP occurs in approximately one-fifth of leprosy affected patients, little attention has been paid to assessing patients and treatment regimens have not been defined. No study to date has established the complete QST somatosensory profile of leprosy patients with NP. To better characterise this somatosensory profile, comparison was made to local Indian healthy control subjects, DFNS reference data, and to patients with and without neuropathy and pain. The ability to accurately identify sensory, psychological, and metabolic dysfunction profiles in leprosy patients with NP is essential for better understanding of the pathophysiological mechanisms as well as informing mechanism-based prescribing and thus has the ability to dramatically improve clinical trial design and NP treatment defined by drug response.

1.6 Aims of the PhD thesis

The overall aim of the study was to characterise the somatosensory profiles of leprosy patients with NP.

1.7 Objectives of the PhD thesis

- 1. To measure the somatosensory responses of leprosy patients including thermal detection and pain thresholds, paradoxical heat sensations, mechanical detection thresholds to von Frey filaments, vibration detection threshold, mechanical pain thresholds to pinprick stimuli and blunt pressure, stimulus/response-functions for pinprick and dynamic mechanical allodynia, and pain summation (wind-up ratio) and to compare these measures between leprosy patients with painful neuropathy; non-painful neuropathy, leprosy patients without pain and age and sex matched healthy controls.
- To elucidate the impact of leprosy NP on quality of life and psychological wellbeing in patients with painful neuropathy compared with patients with nonpainful neuropathy.
- 3. To stratify leprosy patients by symptoms, sensory profile and psychological state

1.8 Hypotheses of the PhD thesis

NP in leprosy patients could be caused by immune neuronal interaction which can be due to destruction or impaired functions of cells as a result of chronic inflammatory processes, or it could be caused by sensitisation and spontaneous activity in sensory neurons which can be due to the effect of *M. leprae*. As a functional compensation pain thresholds are lower, leading to increased pain sensation even though the stimulus is minor. I hypothesise that:

- The thermal and mechanical detection threshold, pain threshold (CPT and WPT), pressure pain threshold (PPT), and vibration detection threshold (VDT) are lower in leprosy patients with painful neuropathy than patients with non-painful neuropathy
- There is significant impact of neuropathic pain on quality of life and psychological well-being in leprosy patients with painful neuropathy compared to patients with non-painful neuropathy.

Chapter 2 Review of Literature

2.1 Literature review of leprosy

2.1.1 Definition of leprosy

Leprosy is a chronic granulomatous infectious disease, affecting the skin and peripheral nerves (Britton and Lockwood, 2004). It is one of the most common causes of peripheral neuropathies worldwide. The disease is caused by *Mycobacterium leprae* (*M. leprae*) (Job, 1989), which is a unique intracellular organism discovered by Armauer Hansen in 1873. The organism is the only bacterium that invades and multiplies inside Schwann cells, which this is the hallmark of the disease (Job, 1989). *M. leprae* cannot be cultured in routine laboratory media, but will multiply slowly in certain animal species, such as the nine-banded armadillo, which is a natural reservoir of the organism (Truman et al., 2011).

2.1.2 Leprosy transmission

Leprosy is thought to be spread through aerosol transmission of nasal secretions of untreated lepromatous leprosy patients, but the precise mechanism, the route of entry, and the role of skin contact are still debated (Pfaltzgraff R. E and G, 1994). It is widely believed that *M. leprae* probably enters the body via the nasal mucosa with subsequent haematogenous spread to the skin and peripheral nerves (Scollard et al., 1999).

2.1.3 Epidemiology of leprosy

2.1.3.1 Global Leprosy Prevalence

Globally the prevalence of leprosy has declined over the last 25 years (WHO, 2013b). At the beginning of 2013, the WHO reported a global prevalence of leprosy of less than one per 10 000 population, with 189 018 registered cases compared to 1.2 million cases in 1995 (WHO, 2013b) (Figure 2.1). The global case-detection rate remains high, with about 230 000 new cases being detected worldwide and reported to the WHO during the year 2012. The global rate of new cases with grade two disabilities per 100 000 population was 0.25, with 14 409 registered cases during the year 2012 (WHO, 2013b). However, due to recent major operational changes relating to the diagnosis and registration of cases in many countries, these leprosy statistics are probably underestimates (Fine, 2008).



Figure 2.1. Global leprosy prevalence 2012

2.1.3.2 Regional Leprosy Prevalence

Leprosy patients are found mainly in the tropical and warm temperate regions of the world, but cases have been also reported elsewhere, as patients may present with the disease long after leaving an endemic region. Currently 95% of new leprosy cases are concentrated in three geographical regions, namely, South Asia (India: 91 743, Nepal: 3 118 and Bangladesh: 3 848); Latin America (mainly Brazil: 34 894); and sub-Saharan Africa (Angola: 1 076, the Democratic Republic of Congo: 5 049, Mozambique: 1 207, and the United Republic of Tanzania: 2 349). South Asia, Latin America, and sub-Saharan Africa still have a prevalence of more than 1 case per 10 000, with 60% of all leprosy cases worldwide being concentrated in Brazil and India (WHO, 2013b).

Leprosy in India

India is the largest country in South Asia with a population of more than one billion (WHO, 2013a). It is the biggest contributor to the global burden of leprosy with 134 752 new cases were detected during the year 2012 (WHO, 2013b).

The prevalence of leprosy has declined over the last decade in India according to government records. The total number of leprosy cases in India fell strikingly from 265 781 in 2004 to 91 743 in 2013 (WHO, 2013b). This changing picture is attributed to introduction of multi-drug therapy (MDT) and the duration of treatment being reduced from 24 months to 12 months. Another contributing factor to the decreased prevalence observed in India is operational activities, such as level of case finding activity and integration of leprosy services into primary health care services.

In recent years, heavy migration into cities such as Mumbai from under developed or poorly surrounded region such as Uttar Pradesh and Bihar may have contributed to an increase in leprosy cases in urban areas. In the Bombay Leprosy Project clinic, 80% of the newly diagnosed leprosy patients from January to November 2013 were reported from urban slums and outside Mumbai.



Figure 2.2. Prevalence and incidence of leprosy in India

Data for leprosy prevalence and number of new cases in India are shown in Figure 2.2. These data clarify the importance of using the number of new cases as a marker of transmission of leprosy. High numbers of new cases continue to be reported. For instance, 91 743 new cases were reported in 2012 compared to 83 187 cases in 2011. A cross-sectional study from Maharashtra, showed rates of three-to-nine cases per 10 000 population, of which 30% of these newly diagnosed patients were children. Furthermore, the WHO global leprosy update reported that 9.9% of new leprosy cases in India in 2012 were among children. Multibacillary (MB) cases were the most frequent type of leprosy among the total number of new leprosy cases detected during the reporting year in India (Kumar, 2010). In 2012, 30% of the newly diagnosed MB cases had neuropathy and nerve damage. These data indicate ongoing transmission of leprosy in India.

2.1.4 Leprosy Classification

The clinical features of leprosy are determined by the host's immune response. Therefore, there are two main systems used to classify leprosy patients. In 1966, Doctors Ridley and Jopling proposed a system of leprosy classification known as Ridley-Jopling classification. They described a spectrum of leprosy from tuberculoid (TT) to lepromatous leprosy (LL) based on their clinical, bacteriological, immunological and histopathological features (Figure 2.3). Between these two poles there are three borderlines: borderline tuberculoid (BT), mid-borderline (BB), and borderline lepromatous (BL). Borderline patients can remain in positions or more towards either end of spectrum. This classification is mainly used in referral centres. It helps to predict patients at risk of reaction and their prognosis, in addition to their appropriate treatment (Ridley and Jopling, 1966). The other classification scheme was recommended by WHO, which suggests that for the purpose of treatment it is adequate to classify patients on the basis of clinical features. Patients are classified as paucibacillary (PB) and multibacillary (MB) by the number of skin lesions they have: PB (one-to-five skin patches); and MB (more than five skin lesions). This operational classification was introduced to simplify disease recognition and to ensure that patients were appropriately treated with MDT.



Figure 2.3. R-J scale and the relation with bacterial and immune response

The five forms of leprosy based on the Ridley-Jopling Scale are tuberculoid (TT), borderline tuberculoid (BT), midborderline (BB), borderline lepromatous (BL), and lepromatous (LL). CMI = cell-mediated immunity; AFB = acid-fast bacilli. Picture modified from (Walker and Lockwood, 2006)

2.1.5 Clinical Features of Leprosy

Patients with leprosy can present with prodromal symptoms and skin and nerve-related symptoms or complications of the disease. The prodromal symptoms, which is rare now and may occur only in LL such as nasal stuffiness, precede the typical skin and nerve features. The main presenting features are related to skin, nerve or reactions (Britton and Lockwood, 2004). A patient may present with a range of skin lesions such as macular hypo-pigmintation. Patients with affected nerves may present with new evidence of neuropathy such as, weakness and loss of function, or secondary to neuropathy e.g. ulcer. Patients with leprosy reaction can present with skin changes or neuritis. These features are diagnostic. The common presenting complications of leprosy are reactions. Patients may present with skin lesions that are erythematous, tender, and swollen due to type 1 leprosy reactions. Others may present with systemic manifestations, such as fever and malaise due to erythema nodusoum leprosum. Five-to-ten percent of leprosy patients may present with acute neuritic features, spontaneous nerve pain, new sensory or motor impairment of recent onset, or mixed-sign neuritis (Mahajan et al., 1996, Van Brakel et al., 1992).

Cardinal signs

There are three cardinal signs (WHO, 2012):

- 1. Definite loss of sensation in a pale (hypo-pigmented) or reddish skin lesion
- 2. Thickened or enlarged peripheral nerve, with loss of sensation and/or weakness of the muscles supplied by that nerve
- 3. Presence of acid fast-bacilli in a slit-skin smear.

The clinical diagnosis is confirmed when a patient has at least one of the cardinal signs (Britton and Lockwood, 2004).

2.1.6 Diagnosis of Leprosy

The diagnosis of leprosy is important for the individual and the community. For the individual with leprosy, accurate diagnosis is essential for providing appropriate management and to reduce stigma associated with the disease. Early identification of suspected cases, will reduce the incidence of impairments and their effects. Also, it reduces the spread of leprosy to other individuals in the community.

The diagnosis of leprosy is made clinically based on clinical history and a full clinical examination of the skin and peripheral nerves (WHO, 2012). Bacteriological and histological investigations also aid in the diagnosis.

In the presence of clinical symptoms, the first step towards diagnosing leprosy is to think of the possibility of leprosy among those who are at risk of exposure to infection which includes residence in an endemic country and being a household contact.

2.1.6.1 History and clinical examination

For every potential leprosy patient seen, a detailed clinical history must include: any changes in skin colour and texture; area of sensory loss; weakness of the hands, feet, and eyes; and any signs of reactions. The clinical history is followed by examination of the skin and peripheral nerves. When looking for signs of leprosy the procedure should be carried out in good light and the patients should be examined while maintaining and respecting privacy. *M. leprae* multiplies at relatively low temperature; therefore, areas such as the ears, face, lateral aspects of the limbs, the back, and the buttocks are to be examined. These areas of skin have relatively low temperatures compared to other less affected regions such as the axillae, groin, perineum, and hairy scalp. The body map is used to record the results of the examinations of skin lesions, nerve thickening, and any tissue damage. For each patient, the following information is required; number and type of skin lesions; loss of sensory and motor functions; bacterial index if skin smear is taken; eye assessment; and treatment regimen.

2.1.6.2 Bacteriological examination

M. leprae can be demonstrated in slit skin smear and/or tissue biopsies. In suspected case, slit skin smears are taken to look for acid-fast bacilli. This test is one of the essential component of definitive leprosy diagnosis by demonstrating acid-fast bacilli on microscopy. Patients with pure neuritic leprosy, in which no visible skin lesions are found, may require a nerve biopsy for accurate diagnosis of leprosy.

Histopathological evaluation is essential for accurate classification of skin lesions across the spectrum. The diagnosis is made by the presence of granulomata and lymphocytic infiltration of dermal nerves in anaesthetic skin lesions. These criteria help to confirm and exclude the diagnosis of leprosy.

2.1.6.3 Diagnostic immunology in leprosy

There is currently no established role for an immunodiagnostic test (serology, skin test, or polymerase chain reaction) of leprosy. There is no good serological test with adequate sensitivity and specificity for leprosy. For instance, the serological tests for phenolic glycolipid have a good but insufficient specificity for the diagnosis of leprosy patients and their sensitivity is generally high for the MB patients (>90%), but low for the PB patients (50-60%). It has also been detected positive (15-20%) in household contacts (Chanteau et al., 1992). Polymerase chain reaction methods for detection of *M. leprae* DNA have been developed and are potentially highly specific and sensitive, detecting *M. leprae* DNA in >95% of MB and 55% of PB cases, but they cannot be used routinely due to reliance on skilled technicians and expensive equipment (Oskam et al., 2003, Parkash, 2011).

The diagnosis of leprosy is typically clinical but sometimes slit skin smears are needed to confirm the *M. leprae*. In each patient, the definite diagnosis of leprosy includes: the spectrum of disease, type of reactions, and nerve function impairments.

2.1.7 Treatment of Leprosy

A comprehensive approach to treating leprosy patients may be used, including chemotherapy, treating reactions and complications of nerve damage, prevention of neuropathic damage, education, psychological support, and stigma reduction.

Curing the active mycobacterium infection through chemotherapy is essential in the treatment of leprosy patients. In 1982, WHO introduced and recommends MDT regimens for all leprosy patients. The regimens are combinations of rifampicine, clofazimine and dapsone. There are two types of WHO-MDT regimens used: the PB-MDT (given for six months: 600mg rifampicin once a month, 100mg dapsone daily), and MB-MDT (given for 12 months: 600mg rifampicin once a month, 100mg dapsonedaily, and 300mg clofazimine once a month and 50mg daily) (Table 2.1). A regimen for children is also available in reduced doses of the drug format. Within this combination, rifampicin has the highest bactericidal activity compared to other two drugs. There are several second line of treatments to MDT. These antibiotics such as minocycline, ofloxacin, clarithromycin and moxifloxacin required a shorter duration of therapy (Britton and Lockwood, 2004). A single dose of monthly for six months MDT

known as ROM (rifampicin, ofloxacin and minocycline) in combination is now available for PB disease (Setia et al., 2011).

The WHO-MDT regimen is highly effective and has been used successfully in treating leprosy patients with a high cure rate, few side effects, and low relapse rates; globally, more than 14.5 million patients have been treated with good clinical response (Lockwood, 2002). Moreover, it is safe in pregnancy and in breastfeeding mothers. WHO technical advisory group in leprosy chemotherapy noted low relapse rates (0.79% for MB patients compared to 1.09% for PB patients). These relapse rates were obtained from 20 141 MB patients and 51 553 PB patients, where 67 and 306 patients respectively were reported to have relapsed over a nine year period of follow up in 1994 (WHO, 1994). Relapse in MB patients, which is defined as the multiplication of *M. leprae*, is associated with clinical deterioration in skin and nerve function impairments and marked increase in bacteriological index. In contrast, relapse in PB patients is difficult to recognise as it is hard to distinguish from type 1 leprosy reaction. The committee also noted no evidence of *M. leprae* having resistance to the WHO-MDT regimen (WHO, 2012).

Additional benefits of WHO-MDT regimens include the prevention of drug resistance and better patient compliance due to a fixed duration of treatment. Another advantage is that supervised drug administration and health education provided by field workers review patients in each visit where leprosy complications such as reactions and nerve function impairments can be early detected and treated.

There are some reports on the adverse effects of WHO-MDT regimens. For instance, clofazamine is documented to cause troublesome skin pigmentations in many leprosy patients in particular among females (Maia et al., 2013). Other adverse effects associated with individual MDT drugs are listed below. After giving WHO-MDT treatment few patients develop an immune upgrading response and get reactions, and some of them may develop pain (Haroun et al., 2012). This indicates that neuritis may be worsened by WHO-MDT treatment and then cause pain. However, there is no association of neuropathic pain with starting WHO-MDT treatment. Although minocycline is effective in preventing neuropathic pain in animal models, it has not yet been demonstrated to be effective in clinical trials (Rojewska et al., 2014, Martinez et al., 2013). It has not been shown to have a role in leprosy neuropathic pain. Minocycline is only used as second line drug in the treatment of leprosy.

2.1.7.1 Rifampicin

Rifampicin is a potent bactericide for *M. leprae.* Previous studies in untreated MB patients have shown that rifampicin administered in single doses of 600mg killed more than 99% of viable *M. leprae* within three or four days (Levy et al., 1976). Its mode of action is via inhibition of DNA-dependent RNA polymerase, and it needs to be combined with other anti-leprosy drugs to minimize the risk of developing drug resistance. In untreated MB patients, a single monthly dose of rifampicin (1200mg) plus daily dapsone was as effective as daily rifampicin (450mg) plus dapsone (Yawalkar et al., 1982). A single monthly dose has been the current dose used in WHO-MDT regimen with few serious side effects. This drug dose produce red-brown discoloration of bodily secretions such as urine and faeces, so patients should be warned.

2.1.7.2 Clofazimine

Clofazimine is a drug with dual activity in leprosy: anti-bacterial and anti-inflammatory. It has weak bactericidal action for *M. leprae*, the mechanism of which is unknown. After a few weeks of commencing 300mg of clofazimine once a month and 50mg daily, active skin lesions start to improve. The most common adverse effects of clofazimine are in the skin (lesions and normal skin), and include troublesome pigmentation and dryness. These pigmentations occur in almost all patients in the first few days to weeks as the drug becomes clinically effective, but are not severe enough to warrant discontinuation, since most symptoms resolve spontaneously (Maia et al., 2013). Discoloration may also occur in hair, the eyes, and in body excretions, such as urine and faeces. Other reported side effects of clofazimine are gastrointestinal-related, such as nausea, vomiting, and diarrhoea.

2.1.7.3 Dapsone

Dapsone is both bacteriostatic and weakly bactericidal against *M. leprae*. Its mode of action is predominantly via inhibition of the synthesis of dihydrofolic acid. Dapsone can cause a few side effects, including haemolytic anaemia and skin reactions. These effects are rare, occurring within the first few months of treatment.

Type of	Drug treatment		Duration
leprosy			of
	Monthly, supervised	Daily, self-administered	treatment
РВ	Rifampicin 600mg; Dapsone 100mg	Dapsone 100mg	6 months
MB	Rifampicin 600mg; Clofazimine	Clofazimine 50mg; Dapsone	12 months
	300mg; Dapsone 100mg	100mg	

Table 2.1. WHO-recommended MDT doses and regimes for adult with leprosy

2.1.8 Immunological complications of leprosy

The role of immune system as causing leprosy complications is well documented (Walker and Lockwood, 2006). *M. leprae* invades and multiplies in Schwann cells and cutaneous macrophages (Britton and Lockwood, 2004). Such neuronal attack not only resulted in impairments of the sensory functions, but is also associated with a robust inflammatory immune response. As the damaged axons undergo Wallerian degeneration, non-myelinated Schwan cells proliferate and release chemokines, which recruit immune cells such as macrophages. Other resident cells such as mast cells and neutrophils provide the majority of the early immune cell attack, i.e. innate immune response. This process is followed by the accumulation of inflammatory mediators, which enhance the recruitment of immune cells such as CD4 and CD8, i.e. adaptive immune response. As a result, the transduction and transmission of signals are changed and cause ectopic activity and pain in affected nerve. The percentage of leprosy patients with nerve inflammation and pain is around 9% and 5% of patients with MB and PB respectively (van Brakel et al., 2005b, Richardus et al., 2004).

In neuropathic pain inflammatory models, activation of the Schwann cells, leukocytes, Toll-like receptors, inflammatory mediators and cytokines contributes to generation of pain and the severity of this responses is related to the degree of neuropathic pain (Gaudet et al., 2011, Lindenlaub and Sommer, 2003, Calvo et al., 2012). These mediators are important for the recruitment of immune cells such as neutrophils, macrophages, dendritic cells, and B and T lymphocytes to the site of nerve injury. The recruitment of inflammatory cells and cytokine expression such as TNF and IL-6 is also increased in nerve biopsies with inflammation in leprosy patients with neuropathic pain (Haanpaa et al., 2004). This suggests that the immune cells is the main player in the generation of leprosy nerve damage and thus neuropathic pain. The nerve, M. leprae and host immune response interactions are discussed in detail under the central role of nerve damage in leprosy.

2.1.9 Leprosy Reactions

Leprosy reactions are an immunological complication of the disease, which are associated with inflammation and acute peripheral nerve damage. Forty percent of MB leprosy patients experience significant appearance of symptoms and signs of acute inflammation in skin lesions and nerves at some stage in their disease course (Pocaterra et al., 2006, Walker et al., 2008). These reactions could occur before, during, and after successful completion of leprosy MDT and are serious because they cause an acute inflammatory immune response. Such acute onset of inflammation may rapidly and extensively damage a peripheral nerve already affected by the disease.

The two main types of reactions are Type 1 (reversal reaction) and Type 2 reactions, also known as Erythema nodosum leprosum (ENL). Neuritis, which is known as nerve inflammation without a Type 1 or 2 reaction, is the third form of leprosy reaction.

2.1.9.1 Type 1 Reactions

Background

Type 1 (reversal) reactions are acute inflammatory episodes, which may occur as a result of increased cell mediated immunity towards *M. leprae* antigen (Lockwood et al., 2002). Up to 30% of individuals with leprosy will be affected, but the prevalence shows large variations in a retrospective study with 386 leprosy patients in Nepal, one third of patients had a type 1 reactions at recruitment (Van Brakel et al., 1994a). In similar study in Ethiopia, type 1 reactions was reported in 16.5% of 594 patients followed for 10 years (Saunderson et al., 2000b). Patients with the borderline leprosy (BT, BB, and BL), whose immunological status is unstable, are at greater risk (Walker et al., 2008, Ranque et al., 2007). Another group of patients who are also at high risk for type 1 reactions is women after child birth (Lockwood & Sinha 1999). The onset of T1R is in the first six months of anti-leprosy treatment, but may occur rarely many years after completion of MDT (Croft 2000).
Immune responses occurring in type 1 reactions

Type 1 reactions are the result of delayed type hypersensitivity reactions to *M. leprae* antigens (Lockwood et al., 2002). They are characterised by T cell activation resulting in inflammation in the affected skin and nerve (Lockwood et al., 2002, Schenk et al., 2012). The increase in cell mediated immunity could lead to immunological upgrading of the patients towards the tuberculoid pole of the clinical spectrum of leprosy. In the affected sites, the CD4+ T cells increase in number and secrete Th1- type pro-inflammatory cytokines, such as IL-1, IL-2, IL-12, and IFN- γ 1, which may shift the immune response towards Th1 immunity. Th1-type cytokines could be responsible for the local tissue damage occurring in T1Rs, including: the swelling and painful inflammation in skin lesions and nerves. In addition, the increase in TNF- α is responsible for these local and systemic inflammatory symptoms that may occur in type 1 reactions (Khanolkar-Young et al., 1995, Little et al., 2001).

Clinical features of type 1 reactions

Patients with type 1 reactions presented with features localised to the skin lesions and peripheral nerves. Skin lesions develop swelling, redness and tenderness, with some emergence of new lesions. Erythema is often followed by desquamation and ulceration. Affected nerves will become rapidly swollen, extremely painful and tender. This is commonly followed by paraesthesia or pain in nerves distribution, accompanied very often by loss of sensory function. Loss of motor nerve functions also develops rapidly. If not treated, patients with type 1 reactions may end up with irreversible nerve damage.

Treatment of type 1 reactions

Type 1 reactions are usually treated with oral corticosteroids (40 mg/day of prednisolone reducing to 5 mg/day every month over six months) with monitoring of skin lesions and nerve function. However, a recent Cochrane systematic review, which looked at the frequency and features of nerve function impairment (NFI), noted that oral corticosteroids are accepted as treatment, but there is no consensus about the dose or duration of treatment, hence further randomised controlled trials (RCTs) are needed to identify the best treatment regimens (Walker and Lockwood, 2008).

2.1.9.2 Type 2 Reactions

Background

Type 2 reactions or (ENL) occurs as a result of antigen-antibody reactions with the formation of immune complexes that leads to systematic inflammation (Pfaltzgraff R. E and G, 1994). Type 2 reaction occurs in up to 50% of LL patients, and in 5-15% of patients with BL disease. The main risk factor for ENL is a high bacteriological index (Voorend and Post, 2013). The prevalence of ENL has decreased after the introduction of MDT, but there is still wide geographical variation. In an Indian retrospective study with 481 leprosy patients the prevalence of ENL was 49.4% in those with LL and 9% in BL cases (Pocaterra et al., 2006). In contrast, the prevalence in Ethiopia is low, where the AMFES study of MB patients found that 5.3% of individuals experienced ENL (Saunderson et al., 2000a). The onset of ENL often starts during the first year of MDT and can relapse intermittently over several years (Balagon et al., 2010).

Immune response occurring in type 2 reactions

Patients with type 2 reactions have high quantities of antibodies and *M. leprae* antigens. The relative increase in the antigen concentrations and precipitating antibodies could lead to formation of immune complexes, which are then precipitated out of the blood and can deposit in the tissues. This deposition can lead to vasculitis with immunoglobulin and complement activations in the vessel wall. During the ENL episodes, the immune complexes may lead to activation of neutrophils seen in vasculitis. Further, the episodes are associated with antigen-specific T cell activation. The mononuclear cell increase in number and release cytokines such as TNF- α (Oliveira et al., 1999). The over-production of TNF- α could be responsible for the tissue damage and systemic toxicity occurring during ENL reactions.

Clinical features of type 2 reactions

Clinical features of ENL reactions are usually systematic. Patients may present with painful, tender, erythematous skin nodules appearing in crops. Subsequently, they develop a generalized illness with high fever; malaise; and oedema of the hands, feet, and face. Other systemic involvement includes iritis, conjunctivitis, arthritis, arthralgia, dactylitis, lymphadenopathy, orchaitis, and renal disease. Peripheral nerves are also affected in ENL, as the recurrent episodes can repeatedly affect them, and may result in damage.

Treatment of type 2 reactions

The aim of type 2 reaction management is to control pain and inflammation, acute neuritis, and eye damage. For mild cases, aspirin may be used, but many cases require treatment with steroids and an increased dose of clofazimine (up to 300mg) (Britton and Lockwood, 2004) or thalidomide. The anti-inflammatory effect of clofazimine has reduced the incidence of ENL reactions (Lockwood, 1996). Thalidomide is the treatment of choice for severe ENL but its availability and teratogenicity limits its use (Walker et al., 2007). It is given in a dose of 400 mg daily until the reaction is controlled, and then reduced gradually to 50mg daily. Thalidomide treatment has been shown in other conditions to cause peripheral neuropathy, but not seen clinically in leprosy.

2.1.9.3 Neuritis

Neuritis is an important aspect of leprosy reactions, because it is difficult to classify, difficult to treat and is associated with disability. Patients with leprosy may experience neuritis at different times during the disease. Two types of patients can be distinguished: those with neuritis associated with both type of reactions (reversal and ENL) and those with pure neuritis pain. The latter may be defined as inflammation of the peripheral nerve trunk without features of the above mentioned reactions (Mahajan et al., 1996). 5-10% of leprosy patients may present with pure neuritis. Those patients may present with spontaneous nerve pain, new sensory or motor impairment of recent onset, or mixed signs neuritis (Mahajan et al., 1996, Van Brakel et al., 1992). The absence of skin lesions and negative skin smear in those patients contribute to the delay of diagnosis (Mishra et al., 1995). Acute loss of function is a hallmark of neuritis in leprosy, which, if not treated rapidly and adequately becomes permanent. Neuritis may occur in the absence of pain and may go unnoticed by the patient 'silent neuropathy'. Patient with neuritis is treated with standard dose of prednisolone starting at 40-60 mg daily, decreasing by 5 mg every 2-4 weeks after evidence of nerve functions recovery (Britton, 1998, Kamath et al., 2014). Neuritis is reviewed under the nerve in leprosy too.

2.2 The central role of nerve damage in leprosy

2.2.1 Introduction and definitions

Nerve damage is a central aspect of leprosy, affecting autonomic, sensory, and motor functions. It is the major cause of morbidity in leprosy. A landmark pioneering publication in early 1950s by Fite stated that there is no leprosy without nerve damage (Fite, 1951). Since then the disease is best described as a neurological condition rather than a simple skin problem. *M. leprae* has a predilection for nerve tissue and the associated perineural inflammation is a characteristic and hallmark of early leprosy. Further complications arise from the host's immune-mediated events that occur during disease progression. Pain due to peripheral nerve damage and/or treated skin lesions is now being recognised as a late complication of leprosy among treated patients. The consequences of nerve damage, such as hands and feet deformities are also responsible for the associated social stigma and disability.

For the purpose of the current study, the term "nerve" refers to the small dermal nerves and peripheral nerve trunk. The general term "neuropathy" is a clinical term used here to mean any detectable abnormality in a particular peripheral nerve. It includes: peripheral nerve damage; nerve function impairment (NFI), such as a motor or sensory deficit; and pain and tenderness.

2.2.1.1 Epidemiology

The epidemiology of nerve damage, in particular, its incidence and natural history of neuropathy in patients with leprosy is well documented. There have been major studies in large groups of patients in geographically diverse settings that have been very useful. Longitudinal data from the Bangladesh Acute Nerve Damage (BAND) cohort support the contention that most patients with leprosy have some demonstrable nerve damage (Richardus et al., 2004). The authors found that the level of nerve damage in 2664 newly diagnosed leprosy patients was high, with up to 60% of MB patients having clinically apparent nerve damage at the time of diagnosis. Presentation of nerve damage at diagnosis is also described in the ILEP Nerve Function Impairment and Reaction ('INFIR') study. Van Brakel et al. studied a cohort of 303 newly diagnosed MB leprosy patients for two years in India and found that 38% of patients had nerve damage at the time of intake into the study (van Brakel et al., 2005b). This high level of nerve damage at diagnosis reflects the delay of leprosy diagnosis, which often takes years and allows

the development of neuropathy (Van Veen et al., 2006). Damage during treatment is noted by Schreuder et al., who investigated 640 newly diagnosed leprosy patients from Thailand. The authors found that the presence of nerve impairment at time of leprosy diagnosis is a key risk factor for new nerve functional impairment (Schreuder, 1998). The study showed that the incidence of NFI among patients without any abnormalities at first examination while on treatment was 1.7 and 12 per 100 person-years at risk (PYAR) for the PB and MB patients, respectively. In addition, 2% of the PB and 11% of the MB patients who already had impairments at first examination developed new NFI while on treatment. Damage after treatment is illustrated by Saunderson et al. who examined 594 new leprosy cases in study known as ALERT MDT Field Evaluation Study (AMFES) in Ethiopia and found that 12% developed new NFI after starting MDT (Saunderson et al., 2000c). Nerve damage in leprosy is common; 60% of MB patients have nerve damage at the time of diagnosis, 30% of patients may develop further nerve damage during MDT treatment and 10% may develop new nerve damage after drug treatment.

2.2.1.2 Risk factors

The major risk factors for nerve damage, especially clinical rather than genetic or immunological, can be grouped into personal factors, visible signs, advanced stage of nerve involvement, and others. The personal factors are: age; sex; pregnancy; and lactation. The criteria for advanced stage of nerve involvement are the WHO disability grades and other factors are the clinical and physical state of the patient. In the INFIR study (van Brakel et al., 2005b), which was designed to assess measures that may predict NFI in a cohort of 303 previously untreated newly diagnosed MB patients, found that age, pre-existing sensory loss, count of enlarged nerves, and WHO grade two were the main predictors of NFI. A further independent risk factor for neuropathy in leprosy is the presence of skin lesions overlying nerve trunks (van Brakel et al., 2005b). Croft and colleagues (Croft et al., 2000) described a simple method, based on disease classification and the presence of loss of nerve function at registration, to identify patients at risk of developing NFI in Bangladesh. Patients who had PB leprosy with no clinical evidence of NFI had a low risk (about 1%) of developing further NFI with treatment. Patients with PB leprosy and clinically detectable nerve function loss who received standard treatment over two two-year observation periods had a risk of around 15% for developing further impairment. The highest risk (around 65%) of further NFI was seen in patient with MB leprosy and clinical NFI. Additional risk factors identified in the INFIR cohort study were deterioration in: ulnar above-elbow motor nerve conduction amplitude or latency; ulnar, median, radial or sural sensory nerve conduction amplitude or latency; and posterior tibial or sural cold or warm sensation (Smith et al., 2009).

2.2.1.3 Previous studies on leprosy neuropathy

While data from previous studies on leprosy neuropathy have contributed to the understanding of many aspects of neuropathy, research relying on leprosy neuropathy has important limitations:

- Limited opportunities to study persons early in process of neuropathy. Patients at diagnosis of leprosy may have already had neuropathy for some time.
- Limited generalisability to all leprosy patients with neuropathy, because patients in leprosy clinics often have severe or complicated cases of neuropathy; only few studies are based on patients in the field. Thus, by studying clinic-based patients, one cannot postulate the variability in and natural course of neuropathy.
- Limited opportunities to make a definitive diagnosis of neuropathy; various methods of nerve function assessment are used. The same case definition of leprosy neuropathy is not being applied by the different protocols. Thus, by using different protocols, one cannot identify the case definition of leprosy neuropathy.
- Limited generalisability to all leprosy patients with neuropathy, because the treatment may vary (steroids, thalidomide and MDT).

These limitations within leprosy neuropathy research suggest that the development of new tools may yield new insight into the course of neuropathy and its associated pain dysfunction.

2.2.2 The nerve in Leprosy

2.2.2.1 The nerve, M. leprae and host immune response interactions

The nerve damage in leprosy is characterised by the unique interaction of the *M. leprae*, the Schwann cell, and the host immune response; the mechanism underlying this process is still very poorly understood (Scollard, 2008).

Following entry into the human, M. leprae grows preferentially in the coolest and most superficial nerve branches, affecting the Schwann cells of the unmyelinated small sensory fibres. It has been suggested that in the skin, M. leprae first binds to exposed Schwann cells and then moves proximally within the nerve (Khanolkar, 1964). However, a more recent study from animal models indicated that M. leprae infects nerves via lymphatics and blood vessels. The authors postulated that M. leprae first aggregates in epineurial lymphatics and blood vessels and then enters the endoneurial compartment through its blood supply (Scollard, 2000, Scollard et al., 1999). At this stage, M. leprae targets the Schwann cell where several potential mechanisms of binding have been suggested (Ng et al., 2000, Rambukkana et al., 1997, Suneetha et al., 2001). The phenolic glycolipid-1 (PGL-1) of M. leprae binds specifically to the native laminin-2 in the basal lamina of Schwann cell axon units (Ng et al., 2000). Rambukkana has demonstrated that *M. leprae* specifically binds to α -dystroglycan in the presence of the G domain of the α 2 chain on laminin-2 (Rambukkana et al., 1997). Other studies have also demonstrated the ability of myelin P0 to bind M. leprae (Suneetha et al., 2001). Once invaded, the Schwann cell provides a suitable environment for M. leprae. The blood-nerve barrier in the peripheral nervous system protects M. leprae from host immune attack (Job, 1989). At this stage, M. leprae appears to persist and grows slowly within Schwann cells.

There are arguments about the M. leprae Schwann cell interactions. Proponents of interactions between M. leprae and Schwann cells claim that the benign characteristic of M. leprae may contribute to this favourable environment. For instance, M. leprae is nontoxic and therefore, it does not harm the Schwann cell. They also maintain that the presence of M. leprae in the nerve may enhance Schwann cell survival in order to maintain the affected cell in an active stage (Lahiri et al., 2010, Tapinos et al., 2006, Hagge et al., 2002, Rambukkana et al., 2002). However, such an argument discounts the fact that M. leprae constantly interacts with host cells in the nerves and skin. It should be evident that the argument against the M. leprae Schwann cell interaction is not valid. On the contrary, studies have shown that human Schwann cells express toll-like receptor 2 (TLR2), and that the activation of TLR2 by M. leprae can lead to apoptosis (Oliveira et al., 2003). Furthermore, it has been shown that even in the absence of immune cells, M. leprae induced ErbB2 activation mediates demyelination (Rambukkana et al., 2002, Rambukkana, 2004). More recently, Masaki et al. (Masaki et al., 2013), investigated the interactions between M. leprae and Schwann cells and suggested a novel model to explain the spread of the infection. This study suggested that M. leprae bacillus not only interact with Schwann cells, but also change its fate to become progenitor/stem cells with mesenchymal characteristics that promote bacterial dissemination (Masaki et al., 2013). However, these models have not been supported by human studies or even armadillo studies. In summary, the interaction between Schwann cells, *M. leprae*, and host immune responses yields nerve damage in leprosy.

2.2.2.2 Nerve damage mechanisms across the leprosy spectrum

The process of nerve damage in leprosy occurs gradually through the course of the disease and its presentation differs at different stages of the disease. Following entry into the human, *M. leprae* affects the Schwann cells of the unmyelinated small sensory fibres. In addition, the immune response to *M. leprae* may aggravate nerve damage, either by inflammatory and immune-mediated processes, or oedema induced mechanical processes (Haanpaa et al., 2004).

Several factors contribute to the development of nerve damage in leprosy. The direct presence of *M. leprae* within Schwann cells results in immunological stimulation. This interaction is mediated by toll-like receptors (TLRs), which kills Schwann cells and causes nerve damage (Krutzik et al., 2003, Oliveira et al., 2003). The presence of *M. leprae* enhances HLA-II expression on Schwann cells; mycobacterial peptides are presented to HLA-II-restricted CD4+ T cells, which can attack and damage Schwann cells. Additionally, activated natural killer cells may participate in these Schwann cell damaging processes. Another possible contributing factor for nerve damage is the ability of T cells to recognise the presence of mycobacterial antigens within the nerve, which contributes to chronic ongoing neural inflammation. Swelling within the perineurium leads to ischaemia, fibrosis, axonal death, and nerve damage. Furthermore, the presence of pro-inflammatory cytokines such as TNF- α in the affected nerve may also contribute to nerve damage. The direct effect of *M. leprae* on the nerve's protein also promote nerve damage (Save et al., 2004).

The unique pathology of nerve damage in leprosy is probably determined by the host's immune response, which differs across the leprosy spectrum. Additionally, the different clinical outcomes within leprosy sub-types may also contribute to the process of the nerve damage.

The two main processes of nerve damage are wallerian degeneration and demyelination. In tuberculoid disease (TT), the predominant process is wallerian degeneration. Patients in this group have vigorous cell-mediated immune responses to *M. leprae* that lead to granuloma formation with gross destruction of the nerve fibres. The epithelioid cell response to bacilli within the nerve is a characteristic of this type of leprosy. It forms cuffs of epithelioid cells that enclose the nerve as a whole (Fite, 1951). Histologically, the bacilli are found in fair numbers within the nerve itself, but rarely in the granuloma. Clinically, the nerve damage occurs early, and may be severe, but is not widespread in the body (Britton and Lockwood, 2004).

In contrast, the predominant process in the other group, lepromatous leprosy (LL), is demyelination. Patients with a lepromatous-type disease show very poor cell-mediated immune responses, but vigorous humoral responses. The bacilli are seen in every part of the nerve fibre, intensifying around Schwann cells. There is therefore demyelination and damage of the nerve fibres. The nerve fibres are less severely damaged compared to tuberculoid leprosy (Britton and Lockwood, 2004).

In the borderline (BB, BT and BL) groups, the nerve damage is obvious. It is explained by the instabilities between the cell-mediated and humoral immune responses. The nerve damage process may be cell-mediated and/or humoral to varying degrees. Patients in this group may have a mix of direct effects of the *M. leprae* and inflammatory immune-mediated pathology (Haanpaa et al., 2004).

2.2.3 Clinical presentations

The clinical presentation of leprosy patients with nerve damage depends on two features: damage of small dermal nerves and/or the peripheral nerve trunks. Although symptoms related to cutaneous loss of sensation occurs early in leprosy, the majority of patients present with symptoms of skin lesions or reactions (enlarged, painful and tender nerve). Besides this, a considerable proportion of patients present with a weakness in their hands, feet or face caused by motor involvement. The high number of patients presenting with clinically apparent nerve damage reflects the delay between the appearance of early leprosy skin lesions and diagnosis.

Loss of sensation (cutaneous anaesthesia)

The loss of sensation and autonomic nerve fibre functions in affected skin lesions is an early manifestation of the disease. Patients may have impaired thermal perception (to temperature), nociception and touch sensation. The skin may be dry due to autonomic

fibre damage. Damage to the peripheral nerve trunks leads to regional sensory loss. These sensory abnormalities may be confined to the innervated territory of the affected peripheral nerves or it may show a glove and stocking distribution when it is severe and widespread.

Nerve thickness, tenderness and pain:

Patients with affected peripheral nerve trunks have thickened peripheral nerves, with or without tenderness and standard regional patterns of sensory and motor loss corresponding to the nerve affected. A thickened or enlarged peripheral nerve, with loss of sensation and/or weakness of the muscles supplied by that nerve is one of the cardinal signs for a leprosy diagnosis (WHO, 2012). The nerve can be enlarged palpably in 40-55% of patients and it may be noticed by the patient, especially if it is painful or tender. In leprosy, certain nerves at certain sites are often more involved than others. This difference may be explained by the characteristics of *M. leprae*, which multiplies at relatively low temperatures and has an affinity for Schwan cells. The nerves most likely to be involved are therefore the peripheral nerve trunks in the upper and lower extremity, including: the ulnar nerve at the ulnar grove; the median nerve at the wrist; the cutaneous branch of the radial nerve at the lateral border of the radius; and the lateral popliteal nerve around the neck of the fibula. Figure 2.4 shows the most commonly enlarged nerves and the sites of enlargement in leprosy-affected patients.

Studies have shown that the ulnar nerve is the nerve most commonly affected by leprosy. Haroun et al. (2012) found that in a cohort of 80 leprosy patients who had completed MDT within 18 months, the ulnar nerve was the nerve most frequently affected (78%) (Haroun et al., 2012). Nerve enlargement can be identified by clinical examinations although this is not always reproducible. However, the degree of enlargement has little bearing on the severity of neuropathy. Nerve pain and tenderness is known as neuritis, which may be defined as inflammation of the peripheral nerve trunk without features of reactions (Mahajan et al., 1996). Patients with neuritis may present with spontaneous nerve pain or recent onset of new sensory or motor impairment (Mahajan et al., 1996). The nerve pain/tenderness may be so severe that even gentle palpation produces an electric shock-like sensation. A detailed description of the peripheral sensory fibres and function is discussed in Chapter 4.

Weakness:

Depending on the sites of the enlargement of the peripheral nerve trunk, the number of affected nerves and disease sub-type, the standard regional pattern of motor loss may include: claw hand; foot drop; and/or facial muscle weakness. The ulnar innervated muscles of the hand and forearm, the median innervated muscles in the hand, the tibial innervated muscles in the foot, the peroneal innervated muscles in the foot and the leg, and the facial innervated muscles are most commonly affected by the disease.

2.2.3.1 Clinical presentation of nerve damage across the leprosy spectrum

As described above, the involvement of peripheral nerves in patients with leprosy is clinically different across the disease spectrum as follows:

Neuropathy in tuberculoid leprosy:

The clinical neuropathy of tuberculoid leprosy is characterised by asymmetrical enlargement of a single nerve. Damage to the small dermal sensory and autonomic nerves produces hypoesthesia and anhidrosis (Bryceson and Pfaltzgraff, 1990). Marked nerve damage can occur in nerves underlying or surrounding skin lesions, which may result in wrist drop, clawing of the hand, and/or foot drop.

Neuropathy in lepromatous leprosy:

Nerve damage in lepromatous leprosy is slow and progressive. It characterised by bilateral symmetrical distal polyneuropathy. The damage to the small dermal sensory and autonomic nerves produces glove and stocking sensory loss. Damage to the nerve trunk produces bilateral and symmetrical thickening and tenderness, which may result in distal weakness of the intrinsic muscles of hands and feet.

Neuropathy in borderline leprosy:

In contrast to that seen in tuberculoid and lepromatous leprosy, nerve damage in borderline leprosy has a relatively rapid onset and acute progress. It is characterised by irregular and asymmetrical neuropathy. Damage to the small dermal sensory and autonomic nerves produces hypoesthesia and anhidrosis (Bryceson and Pfaltzgraff, 1990). Multiple affected nerves produces deformities in the hands and feet.



Figure 2.4. Commonly affected peripheral nerve in leprosy

Picture modified from Khambati and colleagues (Khambati et al., 2009)

2.2.4 Nerve function assessment

It is important to carry out a full/routine clinical assessment of nerve function in leprosy, as early detection of neuropathy will reduce the incidence of impairment and its effects. A comprehensive overview of the detailed methods for the assessment and detection of neuropathy will be explored further in Chapter 4, including the instruments for assessing of sensory abnormalities and pain-related phenomena.

2.2.4.1 Assessment of neuropathy:

History and clinical examination

The clinical history (described in section 2.1.6.1) is followed by palpation of the involved nerves to assess for nerve size, and tenderness.

Nerve assessment includes motor, sensory and autonomic functions. Table 2.2 shows different modalities of nerve fibre functions. A detailed description of the peripheral sensory fibres and function is discussed in Chapter 4. $A\alpha/\beta$ fibres are large in diameter, myelinated and have fast conduction velocity (Barrett et al., 2010). They are normally activated by non-noxious mechanical stimuli such as touch. In contrast, A δ fibres are medium in diameter, myelinated and have intermediate conduction velocity (Barrett et al., 2010).

al., 2010). They are normally activated by noxious stimuli and transmit sharp pain. C fibres are small in diameter, unmyelinated and have slow conduction velocity (Barrett et al., 2010). They are normally activated by noxious stimuli responsible for secondary pain, like burning and aching pain. These modalities can be assessed using electrophysiological methods such as nerve conduction study, cold and warm temperature sensation assessment or monofilaments.

Table 2.2. Physiological function of the motor, sensory and autonomic nerve fibres

Motor	Sensory			Autonomic
	Myelinated	Thinly myelinated	Un-myelinated	
Muscle	Touch, vibration,	Cold perception,	Warm	Sweating
control	position perception	pain	perception, pain	

The choice of the test for the nerve function assessment depends on the availability of the tool and availability of qualified trained staff. For leprosy, nerve functions have been assessed using different methods such as cotton-wool, monofilaments, voluntary muscle testing, nerve conduction study and biopsies.

Assessment of autonomic nerve fibre function: The assessment of autonomic components of the nerve includes testing for sweating of the hands and feet.

Assessment of sensory function: Ideally testing should be for light touch, light pressure, pain and temperature, but it is rarely done in the field. The performance of the test depends on the availability of the tools and qualified trained staff. Light touch is the most frequent sensory modality assessed using different methods such as cotton wool, a ball point pen, and Semmes-Weinstein Monofilaments (MFs) in resource-limited settings (Brandsma et al., 2014).

Assessment of motor function: The assessment of motor components of the nerve is done by voluntary muscle testing in the face, hands and feet. Voluntary muscle testing assesses innervated facial muscles, ulnar and median nerves of the hand and common peroneal nerves of the feet using a modified Medical Research Council (MRC) scale (Brandsma, 2000). MRC is a modified grading system to assess muscle weakness due to peripheral nerve involvement. It asses voluntary movement against resistance.

Voluntary muscle assessment is suitable for measuring NFI in a resource- limited setting. In a cohort study of 303 untreated MB patients from India, assessing the sensitivity and specificity of MFs and VMT using nerve conduction study as the gold standard for detecting NFI (Van Brakel et al., 2005a). The authors found that the concordance between VMT results and motor nerve conduction was good for the ulnar nerve, but very few median and peroneal nerves with abnormal conduction had an abnormal VMT. The authors also noted that both methods have good inter-tester reliability and reproducibility and good specificity of more than 80% for most nerves. A similar study on different tests of nerve function by Khambati and colleagues supports the conclusion that MFs and VMT have good specificity, but moderate-to-low sensitivity of less than 40% in detecting nerve involvement using NCS as a gold standard (Khambati et al., 2009).

2.2.4.2 Disability grading

The physical impairment associated with nerve damage in leprosy occurs mainly in the hands, feet and eyes. It can be assessed using WHO disability criteria, which defines grade zero as no loss of sensation or visible deformity, grade one as a loss of sensation without visible deformity, and grade two as the presence of visible deformity (WHO, 1988).

Physical disability from nerve damage in leprosy can occur before, during and after MDT. A recent survey of 1358 leprosy patients who had been released from MDT up to 5 years earlier in Indonesia found that 77% of the patients had a physical impairment (van Brakel et al., 2012). In the INFIR Cohort Study in India, 40.9% of the newly diagnosed Indian patients had a WHO disability grade of one and 9.6% had a disability grade of two at enrolment (van Brakel et al. 2005). The BANDS cohort had a prevalence of disability grades one and two of 9.6% and 6.0%, respectively (PB and MB patients) at enrolment. Among MB patients, the prevalence of disability grades one and two was 28.5% 18.2% respectively (Croft et al. 1999).

WHO's current 'Enhanced global strategy to further reduce the disease burden due to leprosy' aims to reduce the rate of new cases diagnosed with disability grade two (WHO, 2009a). The reduction of grade two disability is one of the new indices for successful leprosy burden reduction as well as a marker for early detection of nerve damage. In 2011, a WHO report showed the proportion of grade two disability in newly diagnosed leprosy cases varied between India at 3% and China at 27% (WHO 2012a).

2.2.5 The differential diagnosis of neurological lesions of leprosy

The neurological disorders that share the similar features to leprosy are many, but most do not have sensory loss. In a patient from an endemic area and with enlarged nerves, Amyloidosis, Neurofibromatosis, Sarcoidosis, and Vasculitis may need exclusion. For peripheral neuropathy with generalized neuropathy rather than attributed to a specific nerve innervation, conditions such as Guillain-Barré syndrome, Vitamin B12 deficiency, HIV/AIDS, Alcoholism, Hypothyroidism, Hereditary disorders, and heavy metal poisoning will require consideration and appropriate investigation. If motor involvement is prominent, the possibility of trauma, Dupuytren's contraction, Cervical rib, and Scleroderma will arise. Leprosy ulcer has to be distinguish from tropical ulcer and other trophic ulcers. Tropical ulcer is the commonest cause of ulceration in the poorer leprosy-endemic countries. Ulceration in the presence of sensory loss can be seen also in Diabetic Mellitus, the most common cause of sensory neuropathy.

2.2.6 Nerve damage consequences in leprosy

Nerve damage is incapacitating for many people in countries where leprosy is endemic. Patients may develop dryness of the skin, impairment of various sensory modalities, and motor impairment causing muscle weakness. But with time and without appropriate care, these impairments may lead to skin cracks, wounds, clawing of digits, contractures and shortening of limbs, and blindness (Bryceson and Pfaltzgraff, 1990). Impairments may interfere with quality of life and activities of daily living. Therefore, patients continue to be stigmatised and suffer from associated co-morbidities.

Paradoxically, although leprosy is classically associated with sensory loss affecting the hands and feet, pain due to peripheral nerve damage and/or skin lesion may become a problem for some patients. Those patients who do experience pain may develop an additional quality of life burden over that of leprosy itself or leprosy with painless nerve damage. It is not known why some leprosy patients develop painful neuropathy while others are pain-free throughout the course of the disease.

Chapter 3 Identification of neuropathic pain

3.1 Literature review of pain

3.1.1 Definition of pain

Pain is defined by the International Association for the Study of Pain (IASP) as "an unpleasant sensory and emotional experience associated with actual or potential tissue damage, or described in terms of such damage" (Merskey and Bogduk, 1994). The unpleasant is perceived in the brain in the response of afferent activity in nociceptive sensory neurons. However, pain is more than a sensation, or the physical awareness of pain; it also involves affective aspects. This gives further information on the pain's nature, intensity and location. Other factors such as social, psychological and behavioural aspects of pain are also not considered in this definition. These factors need to be taken into consideration in any interaction with the patient (McMahon SB et al., 2013). The inclusion of the various cultural and behavioural aspects to pain would add further definition to the overall concept of pain.

3.1.2 Classification of pain

Pain can be classified into "nociceptive" or "neuropathic" types, Table 3.1.

3.1.2.1 Nociceptive pain

Nociceptive pain occurs as a result of tissue damage in response to a noxious stimulus, such as impending tissue injury, ongoing tissue destruction, or inflammation, in the presence of intact sensory nerve system e.g. osteoarthritis.

3.1.2.2 Neuropathic Pain

Neuropathic Pain is defined by the IASP as "pain caused by a lesion or disease of the somatosensory system" (Jensen et al., 2011).

	Nociceptive Pain	Neuropathic Pin	
Definition	Pain caused by physiological	Pain arising as a direct consequence	
	activation of pain receptors	of a lesion or disease affecting the	
		somatosensory system	
Symptoms	Local +/- referred pain	At the neuro-anatomical lesion	
Quality of	Easy to describe by patients	Difficult to describe by patients	
symptoms	(Good clinical descriptors)	(Poor verbal descriptors)	
Treatment	Good response	Poor to moderate response	

Table 3.1. Nociceptive pain vs. Neuropathic pain

3.2 Literature review of Neuropathic pain

3.2.1 The concept of neuropathic pain

Neuropathy is defined as the disturbance of function or pathological change in a nerve (Merskey and Bogduk, 1994); therefore, NP is pain arises from this process. NP can occur as a result of a lesion or disease of the somatosensory system at the brain and spinal cord (central pain); or posterior roots and peripheral nerves (peripheral neuropathic pain). The definition of NP has been changed over the time due to a change in the use of terms referring to pain caused by sensory abnormalities, and uncertainties about the pathophysiology of chronic pain states. Originally NP term referred to pain due to peripheral neuropathies, but this brought out some difficulties such as the ability to distinguish neuropathic dysfunction from psychological interference. For instance, in patients with leprosy, chronic ulcer, arthralgia, skeletal deformities and depression may all significantly contribute to symptomatology of pain. Later in 1994, this definition was broadened to include the term "dysfunction", where the IASP defined NP, as "a pain initiated or caused by a primary lesion or dysfunction in the peripheral or central nervous system" (IASP, 1994). Recently, this has been also revised by a group of pain experts on the basis of how it could be distinguished from nociceptive pain (Treede et al., 2008). In the new definition the word "dysfunction" has been removed and the phrase "a lesion or disease affecting the nervous system" has been specified to be "a lesion or disease of the somatosensory system".

3.2.2 Epidemiology

The prevalence of NP among the general population is variable and influenced by the rates of aetiology of the underlying disease and the method of assessment (Smith and Torrance, 2012). Torrance et al. (2006), who surveyed 3 002 people from the general population in the UK through a postal survey using the Leeds Assessment of Neuropathic Symptoms and Signs score (S-LANSS), noted that 8.2% had pain of predominately neuropathic origin (Torrance et al., 2006). Bouhassira et al. (2008), who investigated the epidemiology of NP among 23 712 participants using "Douleur Neuropathique en 4 questions" (DN4) sent to 30 155 randomly selected adults in the "Acess Sante" data base, found that 6.9% of them had chronic pain with neuropathic characteristics (Bouhassira et al., 2008). NP affects as much as 8% of general population and in 5% of these cases, it may be severe (Torrance et al., 2006, Bouhassira et al., 2008). These prevalences are related to wealthy western societies and may not necessarily reflect the occurrence of NP in developing countries. Table 3.2 shows the prevalence of peripheral NP in different conditions estimated from single studies.

Condition	Neuropathic pain prevalence
Diabetic Mellitus	Prevalence 8% (Wu et al., 2007) and 20% (Davies et al., 2006)
Post herpetic neuralgia (PHN)	Prevalence ranging from 2.6 to 7.2% (Choo et al., 1997, Helgason et al., 2000, Scott et al., 2003)
HIV sensory neuropathy	42% (Smyth et al., 2007)

Table 3.2. The prevalence of NP in different conditions

3.2.3 Neuropathic pain classification and aetiology

Neuropathic pain may result from disorders of both the peripheral and central nervous system. This classification is based on the anatomical site of the lesion, even though traditionally NP has been classified according to the aetiological diagnosis. The underling nerve damage may occur due to metabolic, ischemic, hereditary, traumatic, toxic, infectious or immune-related cause. Of those, the major causes of NP include: peripheral nerve trauma, spinal cord injuries, diabetes mellitus, multiple sclerosis, herpes zoster, and HIV infection (Merskey and Bogduk, 1994, Attal et al., 2008).

3.2.4 Development of neuropathic pain

The development of NP arises from pathological responses in somatosensory systems in response to nerve damage. NP and associated sensory aberrances probably represent disordered repair process in response to damage. It is important to note that NP is not an inevitable consequence of nerve damage; the minority of patients develop NP after somatosensory insult. Pathophysiological underlying mechanisms, either in the peripheral or the central nervous system, may lead to symptoms such as spontaneous pain or stimulus-evoked pain, as well as other associated sensory disturbances such as sensory loss (anaesthesia dolorosa), hyperalgesia, allodynia and hyperpathia. These collections of symptoms and signs are known as NP syndrome. The exact constellation of NP characteristic, and associated sensory aberrations, which are manifested in an individual patient are variable and complex and often only fully revealed by the use of sophisticated investigation techniques (Jensen and Baron, 2003, Baron et al., 2009). These patterns do not necessarily reflect the condition from which the nerve damage arose, but importantly may well reflect pain generating mechanisms and therefore give a clue to mechanisms and thus likely drug responses on an individual patient level. Thus, it is vital to accurately phenotype the sensory abnormalities in each patients.

3.2.5 Neuropathic pain underling mechanisms

Animal research and the use of human investigative techniques such as microneurography, functional brain imaging, quantitative sensory testing, skin punch biopsies and experimental human pain models have furthered human pain research. The rationale for the mechanism-based approached is to stratify patients according to mechanism and therefore have mechanism-based prescribing of appropriate drugs at the individual patient level. However, the mechanisms of NP have not yet been fully established and research has yielded an array of potential pain generating mechanisms any of which may be operating in an individual patient. This further emphasises the necessity of properly phenotyping patients as discussed in the preceding section. The broad domains of possible underlying patho-physiological pain related phenomena which might contribute to NP development (Baron, 2006, Bridges et al., 2001, Baron, 2009, Baron et al., 2010a, Costigan et al., 2009), are described in Table 3.3. However none of them are unifying hypotheses for the NP mechanism (Bridges et al., 2001).

Neuropathic pain underling mechanisms *		
Peripheral mechanisms	Central mechanisms	
Sensitisation and spontaneous activity in sensory neurons	Sensitisation in the spinal cord	
Inflammation induced ectopic activities	Central sensitisation (changes in the brain)	
Loss of trophic support for neurons	Loss of inhibitory controls and increased facilitation of pain signalling	
Alteration in ion channel		
Immune-neuronal interaction	Neuronal immune and glial cell interactions	

 Table 3.3. Possible neuropathic pain mechanisms

* Cited in Ralf Baron (Baron, 2006), Wallace VCJ and Rice ASC (Wallace and Rice, 2008), Costigan and colleagues (Costigan et al., 2009), and Scholz J, Woolf CJ (Scholz and Woolf, 2007)

3.2.6 Clinical features of neuropathic pain

3.2.6.1 Symptoms

Patients with NP commonly present features of spontaneous pain and/or stimulus evoked pain. The onset of these symptoms usually occurs soon after nerve damage. The overall intensity of pain rated symptoms in patients with NP is higher compared to nociceptive pain, possibly due to paroxysmal episodes (Wilkie et al., 2001). A brief description of these different types of pain is given below.

Spontaneous pain

Spontaneous pain, which is described as pain arise without stimulus, can be continuous or paroxysmal (McMahon SB et al., 2013). Continuous pain is described in terms of dysaesthesia: burning, pricking, tingling, cutting, and stabbing; or deep pain which described as aching, cramping, throbbing, and crushing. A recent definition of spontaneous pain as pain due to neuropathic spontaneous discharge in somatosensory neurons that is caused by changes that are intrinsic to the neuron was introduced (Bennett 2012). This definition was introduced by Bennett due to uncertainty about the concept of spontaneous pain making no sense when the pain is the result of an ongoing inflammatory reaction. Although spontaneous pain can be hardly distinguished from

ongoing pain, which is defined as pain due to inflammatory ongoing discharge caused by the ongoing presence of the products of inflammation, it might lead to ongoing pain. Spontaneous pain commonly occurs in neuropathic pain patients. For example, 100% of 1 236 patients with NP reported spontaneous pain, usually shooting, burning, or electric-shock like sensation (Maier et al., 2010).

Stimulus-evoked pain

Stimulus-evoked pain, which is described as abnormal responses to stimuli, is often referred to hyperalgesic and allodynic. It is characterised by hypersensitivity of the nervous system.

Hyperalgesia (increased sensation of pain in response to normally painful stimuli), allodynia (pain in response to normally non-painful stimuli) and hyperpathia (explosive pain response when stimulus intensity exceeds sensory threshold) are features of sensory gain (Baumgartner et al., 2002) (Cavenagh et al., 2006).

3.2.6.2 Signs

The demonstration of nerve damage dysfunction is essential evidence in the diagnosis of neuropathic pain. Damage can be due to motor, sensory or autonomic dysfunction attributable to a lesion of the afferent transmission system. A loss of sensation/ "negative" signs may result due to complete or partial loss of input to the nervous system and a corresponding sensory loss. A reduction of afferent input caused by a nerve lesion may contribute to regeneration and loss of inhibitory output with development of hypersensitivity resulting in gain of sensation/ "positive" signs (Jensen and Baron, 2003). These loss and gain signs can be demonstrated by clinical examinations and laboratory testing such as quantitative sensory testing.

Negative signs:

Negative signs refer to loss of sensory and/or motor function due to nerve damage. Loss of sensory function is usually to thermal and noxious stimuli, indicating damage to small-diameter afferent fibres or to the spino-thalamic tract. Sensory loss manifestations can result in loss of thermal sensations, light touch, pinprick, and vibration.

Loss of motor functions result in muscle wasting with motor weakness.

Positive signs:

Positive signs in patients with NP usually result from stimulus evoked pain. It can be exaggerated responses to stimuli with either reduced pain threshold (allodynia), normal pain threshold (hyperalgesia) or increased pain threshold (hyperpathia). Allodynia is defined as pain due to a stimulus that does not normally provoke pain (Merskey and Bogduk, 1994). Different types of allodynia are associated with different stimuli: mechanical, thermal and dynamic. Hyperalgesia is defined as increased pain from a stimulus that normally provokes pain (Merskey and Bogduk, 1994). Hyperpathia is a painful syndrome characterized by an abnormally painful reaction to a stimulus, especially a repetitive stimulus, as well as an increased threshold (Merskey and Bogduk, 1994).

3.2.7 Diagnosis and assessment of neuropathic pain

Clinical evaluation is the common method for diagnosing neuropathic pain. It involves a series of systematic steps which include past and present history, a detailed description of pain distribution, quality and intensity of pain and a neurological examination with the emphasis on sensory testing (Jensen et al., 2001, Hansson, 2002). There are two definitive steps in the process: firstly to confirm that some form of damage to the somatosensory system has occurred, without this the diagnosis is impossible. Secondly, to determine that any pain is indeed neuropathic, this because the occurrence of nerve damage does not necessarily follow that any pain is neuropathic in origin, therefore screening tools such as DN4 may help in identifying symptoms and signs suggestive of NP. The sensory findings should be neuroanatomically logical and compatible with a definitive lesion site (Treede et al., 2008).

Patients may be classified as having NP according to the grading system for NP diagnosis, Table 3.4. The grading system considers pain to be definitive NP if all criteria are present, probable NP if criteria "1" and "2" are present, plus either "3" or "4," and possible NP if criteria "1" and "2" are present, without confirmatory evidence from "3" or "4" (Treede et al., 2008).

Cri	Criteria to be evaluated for each patient		
	1		
	1		
1.	Pain with a distinct neuroanatomically plausible distribution		
2.	A history suggestive of a relevant lesion or disease affecting the PNS or CNS		
2			
) 3.	Demonstration of the distinct neuroanatomically plausible distribution by at least one		
	confirmatory test		
4			
4.	Demonstration of the relevant lesion or disease by at least one confirmatory test		

 Table 3.4. Grading system for neuropathic pain

Recently, a comprehensive guideline on neuropathic pain assessment has been developed by the assessment committee of the Neuropathic Pain Special Interest Group (NeuPSIG) of the International Association for the Study of Pain (IASP) (Haanpaa et al., 2011a). The guideline identifies five areas for the assessment as follows: investigation of underlying disease, evidence of somatosensory system dysfunction, confirmation pain is neuropathic pain, severity of pain, and the impact of the pain.

The diagnosis and assessment of neuropathic pain is discussed in detail in Chapter 4.

3.2.8 Psychological co-morbidity, quality of life and neuropathic pain

Neuropathic pain is associated with psychological problems which impact upon quality of life. It interferes with physical and psychological functioning and causes disability (Fishbain et al., 1997). The International Classification of Functioning, Disability and Health describes functioning as "the complex interplay of body functions, body structures, activities and participation, environmental and personal factors" (WHO, 2009b). Disability is defined as "a physical or mental condition that limits a person's movements, sense or activities" (Oxford, 2009).

There is good evidence that depression is a consequence of chronic pain in other conditions (Fishbain et al., 1997). Williams et al. found a high prevalence of depression (33%) in 483 patients with neurological problems followed for 12 months; pain was more likely to persist in those with depression and depression was more likely to persist in those with depression and depression was more likely to persist in those with coexistent pain (Williams et al., 2004). This was supported by independent effect of antidepressant drugs on pain and depression. A study carried out by Meyer-

Rosberg et al. in 2001 assessed 126 patients suffering from NP due to a peripheral nerve lesion, recruited from two multidisciplinary pain clinics in Sweden (Meyer-Rosberg et al., 2001). The study identified depression, anxiety, altered sleep patterns, social isolation and reduced employment status as important co-morbidities (Meyer-Rosberg et al., 2001). Figure 3.1 shows co-morbidity associated with NP.



Figure 3.1. Co-morbidity associated with neuropathic pain

Patients with moderate to severe discomfort due to neuropathic pain symptoms (n=126), taken from study by Meyer-Rosberg (Meyer-Rosberg et al., 2001)

Daniel et al., 2008, who compared NP condition in 46 patients with post-herpetic neuralgia with a persistent pain of nociceptive origin in 55 patients with low back pain from pain management clinics in Scotland and England to determine the differences in physical and psychological function; factors that increase difficulties; responses to pain; beliefs about pain and problems experienced (Daniel et al., 2008). The authors noted that the differences between the two groups were not on the major variables of pain, mood, cognition and physical function. The main differences were in factors that increase pain, people's responses to pain, their beliefs about diagnosis and the cause of pain and the problems they reported as a result of experiencing pain.

3.2.9 Treatment of neuropathic pain

The management of patients with chronic NP is multifactorial. It may involve primary therapy for underlying neuropathy and disease, drug therapy for alleviation of pain and treatment of co-morbidity associated with pain.

3.2.9.1 Primary therapy for underlying neuropathy and disease

Early detection and treatment of the underlying cause may reduce the risk of nerve function impairment; controlling hyperglycaemia may attenuate diabetic neuropathy. However, permanent nerve damage that leads to chronic pain does not respond consistently to such treatment and other therapeutic agents may be needed.

3.2.9.2 Drug therapy for alleviation of chronic neuropathic pain

Both oral and topical drug therapies are used to treat NP. The main classes of oral therapy are tricyclic antidepressants, anticonvulsants, tramadol, and opioids. Topical therapies include capsaicin and local anaesthetic.

According to the European Federation of Neurological Societies (EFNS) guidelines on pharmacological treatment of NP, the first two classes of medication (tricyclic antidepressants and anticonvulsant) are recommended (Finnerup et al., 2010, Dworkin et al., 2007).

Other therapies such as opioids, topical local anaesthetics, topical capsaicin (0.075% and 8%) may be used, but the first two drugs still represent the main options for treating the condition and both of them are appropriate for use in developing countries (Finnerup et al., 2010).

Antidepressants

Tricyclic antidepressants have analgesic properties in addition to their antidepressant effect. This effectiveness has been shown in different clinical trials (Finnerup et al., 2005). Therefore, they are commonly used for pain relief in patients with NP (McCleane, 2003, Sindrup et al., 2005). The analgesic effects of antidepressants are not fully understood and their mode of action could be predominantly via inhibition of reuptake of serotonin and/or norepinephrine; blocking of sodium channels; or anticholinergic. The most common antidepressants used are imipramine, clomipramine, amitriptyline, desipramine, nortriptyline, venlafaxine, and duloxetine. A combination of drugs is often used. For instance, duloxetine and venlafaxine which are selective serotonin norepinephrine reuptake inhibitors were effective in patients with painful diabetic neuropathy (Baron, 2011).

Anticonvulsants

Anticonvulsants drugs such as oxcarbazepine, topiramate, lamotrigine, gabapentin, and pregabalin have an analgesic effect in chronic NP (McCleane, 2003). Anticonvulsants with Na+ channel actions such as carbamazepine are effective in trigeminal neuralgia (Baron et al., 2010a). Those acting on alpha2-delta-1 subunit of neuronal Ca++ channels, such as gabapentin and pregabalin, show efficacy in postherpetic neuralgia, diabetic painful neuropathy, central pain states and other NP conditions (Finnerup et al., 2005, Field et al., 2006).

Opioids

Opioid analgesics, such as morphine, codeine, methadone, and oxycodone, are used for the treatment of NP. Their mode of action could be predominantly via agonists at presynaptic and postsynaptic opioids receptors. Tramadol, which acts via inhibition of norepinephrine and serotonin reuptake, also has a role in the treatment of NP. Opioids have an analgesic efficacy the same as antidepressants; however, their use is limited, due to long term side effects and a lack of data supporting long term efficacy (Raja et al., 2002).

Capsaicin

Capsaicin is the active component of chilli pepper, produces burning and heat sensation in contact to skin. Interestingly, topical application of capsaicin was found to be useful in relieving chronic pain caused by nerve damage. Its mode of action was thought to occur at C-nociceptive fibres resulted in depletion of substance P and the nociceptor fibres (Attal, 2000). It is available in two forms: capsaicin cream (0.025-0.075%), and patch (8%). The former has had varying results in patients with painful diabetic neuropathy, and postherpetic neuralgia, whereas the topical high-dose capsaicin patch was found to be effective in patients with postherpetic neuralgia and HIV neuropathy (Backonja et al., 2008, Simpson et al., 2008). Use of capsaicin is limited due to its shortlived effect and burning sensation. It is predominately used as adjuvant therapy.

3.2.9.3 Treatment of co-morbidity

Depending on pain intensity and psychological assessment, patients with chronic NP may need further clinical, psychological and psychiatric treatment.

3.3 Pain in leprosy

3.3.1 Types of pain in leprosy

Patients with leprosy may experience skin and nerve pain at different times during the disease. Two types of patients can be distinguished: those with pain associated with tissue inflammation occurring during episodes of immune mediated reactions and those with leprosy affecting the somatosensory system.

3.3.1.1 Nociceptive pain

Leprosy nociceptive pain may occur due to activation of the nociceptive system either by tissue injury; ongoing tissue destruction such as pain around infected ulcer or swollen joint; or inflammation. Inflammatory pain is caused primarily by immune-mediated responses and can continue after completion of leprosy MDT.

3.3.1.2 Neuropathic pain

Leprosy neuropathic pain occurs due to damage to the somatosensory system. It may occurs in the skin lesions and/or nerves, even years after successful completion of the MDT.

3.3.1.3 Mixed pain

Leprosy mixed pain can occurs as a combination of both neuropathic and nociceptive pain; i.e., both components were present in different parts of the body.

3.3.2 Literature review of neuropathic pain in leprosy

3.3.2.1 Epidemiology

Recent studies show that NP is not uncommon among leprosy patients (Haanpaa et al., 2004). Table 3.5 shows NP studies that have been done in leprosy endemic settings over the last decade (Hietaharju et al., 2000, Haanpaa et al., 2004, Croft, 2004, Stump et al., 2004, Malaviya, 2005, Lund et al., 2007, Saunderson et al., 2008). The prevalence of NP range from 17.5% to 56.1%, however, these studies have been cross-sectional and hospital-based, which might over-estimate the prevalence of pain. In a cross-sectional study from Ethiopia with 96 patients who had been treated for leprosy more than 10 years prior to assessment, NP was found in 28 (29%) (Saunderson et al., 2008). Another

cross-sectional study on 358 leprosy patients from a referral centre in Brazil in 2004, reported 201 (56.1 %) of the patients had past or current moderate to severe NP that interfered with daily life activities (Stump et al., 2004).

Authors, year	Country	Type of	Number	outcome
		study	enrolled	
Hietaharju	Bangladesh	Cross-	16	Majority of patients had a
(Hietaharju et al.,		sectional		sensory changes suggestive
2000)				of NP
Stump (Stump et al.,	Brazil	Cross-	358	56.1% of patients with
2004)		sectional		treated leprosy had NP that
				interfered with daily life
				activities
Lund (Lund et al.,	India	Cross-	17	81.3% had neuropathy. All
2007)		sectional		nerve biopsies showed
				intraneural inflammation
Saunderson	Ethiopia	Cross-	96	29% of patients treated for
(Saunderson et al.,		sectional		MB leprosy more than ten
2008)				years previously had NP
Lasry-Levy (Lasry-	India	Cross-	101	21% of patients had NP
Levy et al., 2011)		sectional		
Haroun (Haroun et	Ethiopia	Cross-	80	17.5% of patients had NP
al., 2012)		sectional		
Reis (Reis et al.,	Brazil	Cross-	33	66.3% of patients had NP
2013)		sectional		
Ramos (Ramos et al.,	Ethiopia	Cross-	74	70.3% of patients had NP
2014)		sectional		

Table 3.5. Overview of studies carried out on neuropathic pain in leprosy

3.3.2.2 Psychological co-morbidity, quality of life and leprosy neuropathic pain

Patients with leprosy have a higher prevalence of psychiatric problems, probably because of the chronicity of the disease, disability caused by neuropathy and stigma. Although several studies have confirmed that depression was the most common psychiatric problem, affecting between 10% and 52.4% of patients, there is little published literature exploring the interactions between pain, psychological status and quality of life among leprosy patients (Fishbain et al., 1997, Williams et al., 2004, Meyer-Rosberg et al., 2001). A study from India in 2009, evaluated the association of chronic NP with psychological morbidity in 101 leprosy patients, the DN4 and GHQ-12 were used to identify NP and psychological morbidity, showed that 21.8% of the leprosy patients in the study had NP and that psychological morbidity was detected in 41% of patients with NP (Lasry-Levy et al., 2011). Another study, which included 80 patients from a leprosy centre in Ethiopia, noted that 68 (85%) of the patients had depression (Leekassa et al., 2004).

3.3.2.3 Mechanisms of NP in leprosy patients

In leprosy, there is increasing evidence that the development of neuropathic pain is probably immunologically mediated (Lund et al., 2007). Reactions, neuritis and inflammation, which are common among leprosy patients who develop NP, are a risk factor. The immune response in the peripheral nerves may recur and if they repeatedly affect the peripheral nerves, chronic-post inflammatory pain may result. Lockwood et al. (2002) have shown that *M. leprae* protein and lipid antigens are present in skin and nerves at the time of acute reversal reactions (Lockwood et al., 2002). A recent study, which examined 27 ulnar nerves of leprosy patients using nerve conduction studies in Brazil, showed the association between NP patients with reactions (Garbino et al., 2011). In addition, small fibre neuropathy (SFN), which is the most common neurological complication of leprosy, may have a significant contribution to the occurrence of NP in leprosy (Hietaharju et al., 2000). Table 3.6 shows the most typical symptoms of NP and possible related mechanisms that may occur due to nerve damage in leprosy.

Symptoms	Possible related mechanism		
Dysaesthesia	Ectopic discharges in Aβ fibres		
Spontaneous paroxysmal	Ectopic discharges in C fibres, Inflammation induced ectopic		
shooting or lancinating	activities or Immune-neuronal interaction		
pain			
Continuous spontaneous	Peripheral nociceptor sensitisation, loss of inhibitory control or		
evoked pain	ectopic discharges in C fibres		

Table 3.6. NP symptoms and related mechanisms in leprosy patients*

* Table adapted from Woolf and Mannion. Lancet 1999; 353: 1959-1964

3.3.2.4 Clinical presentation of leprosy patients with pain

Patients with leprosy NP commonly present with continuous burning pain, dysesthesia, paraesthesia and/or paroxysmal pain attacks (Hietaharju et al., 2000). Dysaesthesia and paraesthesia may occur as a result of increased expression of Na+ channels in injured nerve fibre which may leads to ectopic discharges in A β fibres (Woolf and Mannion, 1999). Ectopic discharges in C fibres may contribute to the occurrence of paroxysmal shooting or lancinating pain. Continuous burning pain may occur due to peripheral nociceptor sensitisation, loss of inhibitory control or ectopic discharges in C fibres in affected nerves (Woolf and Mannion, 1999).

A frequent finding in leprosy patients with NP is loss of pin-prick and temperature sensation (cold and warm). Other stimulus-dependent signs such as allodynia and hyperalgesia may not usually occur in leprosy patients (Hietaharju et al., 2000), but may occur in leprosy patients with chronic neuritis (Hietaharju et al., 2000). Allodynia, which is defined as pain due to a stimulus that does not normally provoke pain, and hyperalgesia, which is defined as an increased response to a stimulus that is normally painful, are typical in states with well preserved and irritated peripheral sensory fibre which are overactive and cause central sensitisation (Bridges et al., 2001, Baumgartner et al., 2002), but in leprosy there is silent inflammation and slow but complete destruction of the nerve (Job, 1989). Inflammatory pain is usually considered to be of nociceptive character, because it partly results from hyperexcitability of intact nociceptive dorsal root ganglion neurons innervating inflamed tissue. However, chronic inflammatory pain is often characterized by positive signs such as hyperalgesia and allodynia, suggesting

possible neuropathic component. Recent studies have shown that inflammation-induced nociceptor hyperexcitability is sustained by C-nociceptors, which may contribute to inflammatory hyperalgesia. This may also explain why in some inflammatory conditions both nociceptive and NP may overlap.

Hietaharju et al. reported detailed characteristics of chronic NP in 16 patients with treated multibacillary leprosy in Bangladesh (Hietaharju et al., 2000). 10 patients (62.5%) the pain had a glove and stocking like distribution, which suggests a distal symmetrical polyneuropathy, rather than mononeuropathy, in these cases. The quality of pain was burning in 9 (56.3%), and the occurrence of pain was continuous in 8 (50%) patients. A study from Brazil with 358 leprosy patients, showed that a glove (22.4%), and stocking (24.9%) like distribution of pain was common (Stump et al., 2004). In an Indian study with 17 leprosy patients with chronic NP who had completed MDT, the pain was burning in all patients (Lund et al., 2007). In leprosy, the presentation of painful neuropathy is heterogeneous, some patients have symptoms of polyneuropathy others may have mononeuropathy. Both neuropathies potentially may present in the same patients.

3.3.2.5 Diagnosis and assessment

Clinical evaluation is the common method for diagnosing NP in leprosy patients. It involves a series of systematic steps which include past and present history, a detailed description of pain distribution, quality and intensity of pain and a neurological examination with the emphasis on sensory testing (Jensen et al., 2001, Hansson, 2002). The diagnosis and assessment of NP in leprosy is discussed in detail in Chapter 5.

3.3.2.6 Treatment of neuropathic pain in leprosy

Amitriptyline for the treatment of neuropathic pain in leprosy

Overview of amitriptyline

Amitriptyline HCl is a tricyclic antidepressant drug, which possesses marked neurotrophic activity (Association, 2011). It was approved for depression by the Food and Drug Administration (FDA) in May 1983. Since then the drug has been widely and successfully used as a treatment for several conditions such as depression, nocturnal enuresis, migraine prophylaxis, and NP (Jang et al., 2009).

Amitriptyline has a well-documented efficacy on different types of NP. The drug dose is started at 10-25mg in the evening, and the dose is increased to adequate level of pain relief, with a maximum tolerated dose up to 150 mg per day. In addition, it is a known antidepressant drug and its effect on mood can be attained at lower doses. The amitriptyline dose for depression is started initially at 75mg daily in divided doses or as a single dose at bedtime escalated gradually as necessary to 200mg. The improvement in NP outcomes is independent of the effect of amitriptyline on mood (Max et al., 1987).

The anti-cholinergic adverse effects of amitriptyline include dry mouth, constipation, nausea, difficulty with micturition, sweating and cardiovascular effects (Association, 2011). These side effects are common in all tricyclic antidepressants drugs and can be reduced by starting with low dosages administered at bedtime.

Amitriptyline studies for the treatment of neuropathic pain

Previous studies on amitriptyline for the treatment of chronic NP have shown that a dose of 50-150mg is beneficial when used for treatment of painful neuropathy in a chronic disease (Max, 1987, Graff-Radford et al., 2000). These conditions include diabetic mellitus and post herpetic neuralgia. However, two studies found this drug to have no effect on HIV neuropathy compared with a placebo (Kieburtz et al., 1998, Shlay et al., 1998). Also, a similar study failed to find any effect of amitriptyline on chemotherapy induced neuropathic symptoms (Kautio et al., 2008). Appendix 53 shows various amitriptyline studies for the treatment of NP.

The results of the above studies were encouraging as it has been shown that amitriptyline was efficacious, and superior to placebo treatment, in treating NP caused by diabetes and post herpes infection. The median effective dose of amitriptyline in these studies was 75mg. Amitriptyline was not more effective than placebo in relieving pain in malignancy or chemotherapy induced neuropathy probably explained by low dose of 50 mg. In HIV neuropathy the results of amitriptyline have shown no significant pain relief. It is unknown whether this is explained by the underlying mechanism, because it also showed no effect in animal models (Phillips et al., 2010).

Although there are reports on the use of amitriptyline for the treatment of chronic pain in leprosy and clinicians often prescribe this drug and other antidepressants in treating NP, there are no data from controlled studies in well-defined groups of leprosy patients that demonstrate efficacy and guide their use. A single unpublished abstract work evaluating the treatment of 49 patients with leprosy and paraesthesia in a randomised non-blinded comparative 8-week trial at an Indian hospital was conducted in 2006. The author found that both amitriptyline and gabapentin produce modest improvements of the condition (Bhat and Khanna, 2006).

3.3.3 Summary

- Neuropathic pain is being increasingly recognised among leprosy patients
- No standard methods for assessing chronic neuropathic pain in leprosy
- Patients with leprosy may have a higher prevalence of psychiatric problems, probably because of the chronicity of the disease, disability and stigma caused by neuropathy
- High prevalence and morbidity of neuropathic pain in treated leprosy patients warrant clinical trials to assess the efficacy of pain therapies for leprosy-associated neuropathic pain.

Chapter 4 Instruments for assessing sensory abnormalities, pain disorders and techniques for assessing the psychological impact of neuropathic pain

4.1 Introduction

This chapter provides an overview of the somatosensory system and the main instruments for assessing sensory abnormalities used in the study, namely; instruments for symptoms: pain questionnaires (DN4 and PainDETECT), and quality of life and psychological factors (BPI and GHQ-12 questionnaires); and instruments for clinical signs: quantitative sensory testing (QST).

4.2 Overview of the structure and function of the somatosensory system

4.2.1 Introduction

The nervous system is divided into the central nervous system (CNS) and the peripheral nervous system (PNS) (Figure 4.1). It integrates sensory information and controls motor and cognitive function. This section will focus on the somatosensory components, which are responsible for the sensations of light touch, vibration, temperature and pain.

Nerve fibres in the skin are initiated in the spinal cord and traverse through the dorsal root ganglia which host the cell bodies, into peripheral nerves (Figure 4.1). After entering the skin, the nerve fibres pass in nerve bundles to the superficial dermis. These are small sensory fibres that provide protective sensibility. It includes small myelinated fibres (A δ) and unmyleinated axons. The latter, arranged in Remak bundles, is defined as a non-myelin forming Schwann cell and the unmyelinated fibres penetrate into the dermal-epidermal junction the smaller unmyelinated fibres penetrate into the epidermis, where individual epidermal nerve fibres emerge from the bundles and shed their collagen collar and Schwann cell sheath as they pierce the dermal-epidermal basement membrane. They penetrate through the epidermis to the stratum corneum, usually vertically, establishing free nerve endings (Griffin et al., 2001). In contrast the myelinated fibres tend to penetrate only into the dermis. Changes to these intra-epidermal nerve fibre densities (IENFD) are valuable in quantifying small fibre neuropathy.



Figure 4.1. The somatosensory system

Picture modified from Backonja and colleagues (Backonja et al., 2013)

4.2.2 Types and functions of the peripheral nerve fibre

Two different methods are used to classify types of peripheral nerve fibres: letter and numerical classification (McMahon SB et al., 2013). The ABC classification is based on function, size and myelination of the fibres. Using size classification, the largest diameter fibres are classified as A. This group is further sub-divided into four groups; α , β , δ and y. The number classification is based on conduction velocity I-IV in descending order of velocity. Table 4.1 shows the different types of peripheral nerve fibre and their classification. The table also shows the different modalities of somatosensory nerve fibre functions. The primary afferent fibres A δ , A β and C, which transmit the initial stimulus from the periphery, are located in the skin. A δ and C fibres are the main painmediating nerve fibre systems. A α/β fibres are large diameter, myelinated and have fast conduction velocity (Barrett et al., 2010). They are normally activated by non-noxious mechanical stimuli such as touch, vibration and pressure. Following injury they have been shown to respond to mechanical stimuli and contribute to mechanical allodynia (Treede and Cole, 1993). Ab fibres are medium diameter, myelinated fibres of intermediate conduction velocity (Barrett et al., 2010). They are normally activated by noxious stimuli and transmit the rapid phase of pain, which is sharp in nature. In contrast, C fibres are of small diameter, unmyelinated and have slow conduction velocity (Barrett et al., 2010). They are normally activated by noxious stimuli responsible for secondary pain; burning, dull and aching pain (Craig, 2003). Different receptors related to C fibres have been described, including thermoreceptors which respond to warming and cooling, chemo specific nociceptors and low threshold mechanoreceptors which respond to pressure (Meyer et al., 2006). In addition to thermorreceptors, some C fibres respond to mechanical, heat, irritant chemical stimuli and itch also have been described (Meyer et al., 2006, Lynn et al., 1996). Some groups of A δ and C fibres, known as 'silent nociceptors' may be insensitive to chemical or mechanical stimuli, but some sensitised following inflammation and then can be activated by mechanical stimuli. This may be the underlying mechanism for hyperalgesia (Xu et al., 2000). These fibres are thought to be important in inflammatory pain conditions and central sensitisation (Weidner et al., 1999). Damage to these fibres may also leads an ongoing pain and if this associated with sensory abnormalities the pain is defined as neuropathic pain (Jensen et al., 2011).

4.2.3 Somatosensory receptors

All the peripheral terminal branches of a primary axon form only one type of somatosensory receptor. Based on function, these sensory receptors can be divided into three groups: mechanoreceptors, thermoreceptors and nociceptors. The former contains ion channels that respond to stretching or changing in tension of the surrounding membrane. They mediate the sensations of light touch, pressure, vibration, flutter, limb position and movement. The second group has receptors for warm or cold stimuli. The third one is selective for different types of noxious stimuli such as thermal, mechanical, and chemical. These stimuli are those that can cause tissue damage.

Signals from these receptors are transmitted to the central nervous system (Table 4.1). Each sensory neurone in the peripheral has a cell body in the dorsal root ganglia of the spinal cord. These bipolar neurones have a long peripheral axon branch and a central axonal projection.

Following damage to somatosensory system, there may be a partial or complete loss of sensory functions, and the development of symptoms such as pain. This indicates that pain is a protective response preventing further damage to an affected area. For instance, injury to the sole of the foot leading to pain would cause the bearer to avoid putting further weight on the affected area until it was healed. Damage to this system
may also leads an ongoing pain and if this associated with sensory abnormalities the pain is defined as neuropathic pain (Jensen et al., 2011). This type of pain does not require any receptor stimulation and can be severe, chronic and intractable.

Туре	Diameter	Conduction	I-IV	Modality	Method of
of	(micro-	speed (m/s)	Class		Assessment
fibre	metres)				
Α-α	8 – 13	44 -78	n/a	Efferent,	n/a
				motorneurone to	
				muscle	
Α-δ	3 - 8	18 – 48	n/a	Efferent,	n/a
				motorneurone to	
				muscle spindle	
A	12 20	75 120	T	Afforent limb	
A-a	12 - 20	75-120	1	Allelent, mild,	-
				position and	
Α-β	6 – 12	30 - 75	II	Afferent, touch,	MFs, JPS, brush,
•				pressure, vibration	allodynia
				proprioception	
Α-δ	1-6	5-30	III	Afferent, fast pain,	
				cold, crude touch	
	4 0	2 15			
В	1 - 3	3-15	n/a	Autonomic pre-	-
				ganglionic	
				neurones	
C	<1.5	0.5 – 2	IV	Afferent, slow	Pinprick, hyperalgesia
				pain. warm	warm sensation, heat
				[and cold pain
					torn parts

Table 4.1. Peripheral nerve fibre types and classification

4.3 Instruments for assessing sensory and pain-related phenomena

A number of assessments tools are available for the identification of symptoms and clinical signs of pain-related sensory abnormalities. Symptoms including pain and psychological factors are assessed with a validated questionnaire. This is a list of purposely designed questions that captures the quality and intensity of pain-related sensory symptoms perceived by patients, i.e. patient-reported outcomes. Clinical examination, supplemented by additional diagnostic methods, is commonly used to ascertain the clinical signs of the various sensory perturbations associated with nerve damage. These include quantitative sensory tests (QST), neurophysiological methods, microneurography and skin biopsy (Haanpaa et al., 2011b). QST is a psychophysical method that provides a comprehensive measure of the somatosensory functions. The perception thresholds to various thermal and mechanical sensory stimuli are accurately measured using the QST battery, but the outcome relies on the patient's subjective response. Neurophysiological methods, which include nerve conduction studies, somatosensory evoked potentials and laser-evoked potentials, are objective tests that assess function of large and small afferent fibres (Cruccu and Truini, 2009). Skin biopsy enables quantification of the number of intra-epidermal nerve fibres, which provides a measure of small fibre density (Griffin et al., 2001). Composite tools of symptoms screening questionnaires and diagnostic procedures are often used.

Neuropathic pain, which is defined as pain caused by a lesion or a disease of the somatosensory system, may be manifest in a variety of ways, depending on the location, severity and the underlying cause (Jensen et al., 2011).

The symptoms manifest in association with heterogeneous sensory disturbance range from sensory gain to sensory loss. For example, many patients with NP due to HIV neuropathy have loss of mechanical and vibration detection thresholds. Other "sensory gain" symptoms such as allodynia and hyperalgesia may have a variety of causes such as complex regional pain syndrome (CRPS), Trigeminal Neuralgia (TN) and Postherptic Neuralgia (PHN)(Maier et al., 2010). Analysis of somatosensory profile graphs may help in identifying the presence of sensory gain or sensory loss. However, this profiling and grouping of patients with sensory disturbance cannot be determined by clinical examination alone.

The ability to stratify patients by symptom, clinical signs and psychological state has greatly influenced the identification of NP. A mechanism based approach to NP treatment, guided by symptoms and sensory profiles, has been significantly enhanced by the use of different methods of assessing the pain-related sensory abnormalities. In the following sections, an overview is given of the standard tools that are currently available in existing NP clinical practice to determine whether these are valid for the purpose.

4.3.1 Pain scales and questionnaires for symptoms

Clinical investigators have long recognised the variety of pain experienced by patients, such as shooting, fearful descriptors of neurologic pain to the burning pain after peripheral nerve damage.

These symptoms, along with clinical examination, provide the key to diagnosis and may even suggest the course of treatment. In assessing such patients, it is important that the measure is: valid, reliable, reproducible and useful. I will now describe some instruments that have been validated and are reliable in assessing pain symptom.

4.3.1.1 Pain scores and scales

Pain intensity can be measured on Likert scales, Visual analogue scales (VAS), Verbal rating scales (VRS), or a combination of verbal and numerical rating (Gracely Pain Scale) (Haanpaa et al., 2011a). Verbal rating scales typically consist of a series of verbal pain descriptors ordered from least to most intense (no pain, mild, moderate, severe). The patient reads the list and chooses the one word that best describes the intensity of pain at that moment. A score of zero is assigned to the descriptor with the lowest rank; a score of 1 is assigned to the next lowest rank (McMahon SB et al., 2013).

The Likert numerical rating scale is the most frequently used scale for pain intensity (Farrar et al., 2001). It is recommended by the IASP for assessing pain intensity (Haanpaa et al., 2011a). Likert is an 11- point numerical scale ranging from "0" (no pain) to "10" (worst possible pain) (Farrar et al., 2001). The scale is validated and may be easier to use than the VAS (Dworkin et al., 2005). It is commonly used to assess treatment effect in chronic pain (Dworkin et al., 2005). Using this scale, patients are asked to describe the average intensity of pain by choosing the appropriate number between 0 and 10 in response to the question "tell me what number best represents the greatest pain you have had in the last week".

The other options for recording pain - the NRS and the VAS - are represented by a 10cm line, with one end representing no pain and the other representing worst pain; the patient is asked to mark a point on this line that represents their pain level, and this line is then measured to arrive at a numerical measurement (McMahon SB et al., 2013).

4.3.1.2 Pain questionnaires

To identify patients with sensory abnormalities and possible NP, no single symptom is found, but a combination of symptoms and signs. Several available screening tools are used, such as the PainDETECT Questionnaire (PD-Q), the Leeds assessment of neuropathic symptoms and signs (LANSS), the neuropathic pain questionnaire (NPQ), the McGill Pain Questionnaire (MPQ) and the Douleur Neuropathique en 4 Questions (DN4). These questionnaires are particularly recommended for non-specialists (Haanpaa et al., 2011a).

Pain Detect Questionnaire (PD-Q)

The PD-Q questionnaire is one of the most widely used questionnaires for recording somatosensory systems of NP. It was designed by Thomas R. Tolle in 2006 based on the German Research Network on Neuropathic Pain data base, and takes into account a large number of descriptors of pain (Freynhagen et al., 2006a). The PD-Q comprises nine questions regarding the severity, course, quality and nature of the patient's pain and the specific NP symptoms. Please see the Methods section for an illustration of the PD-Q score described here.

Leeds assessment of neuropathic symptoms and signs (LANSS)

The Leeds assessment of neuropathic symptoms and signs questionnaire was designed as a scale measure for identifying patients whose pain is dominated by neuropathic mechanisms. It was developed and validated in two different populations of chronic pain patients by Michael Bennett in 1999 (Bennett, 2001). The scale consists of seven items measuring five symptoms and two aspects of sensory dysfunction. Each item requires yes or no responses and the scores is compared with the cut-off values; a score of 12 or more suggests pain of predominately neuropathic origin (Bennett et al., 2005).

Neuropathic Symptom Inventory (NPSI)

Neuropathic Symptom Inventory (NPSI) is a validated questionnaire designed to evaluate the different symptoms of NP (Bouhassira et al., 2004, Crawford et al., 2008). It was validated in 176 consecutive patients with NP in France and Belgium (Bouhassira et al., 2004). It includes 10 items (plus two temporal items), quantified on a (0–10) numerical scale, that allow discrimination and quantification of five distinct clinical

relevant dimensions of NP syndromes and that are sensitive to treatment (Bouhassira et al., 2004, Crawford et al., 2008).

McGill Pain Questionnaire (MPQ)

The McGill pain questionnaire consists of three major classes of word descriptors (sensory, affective and evaluative) that are used by patients to specify subjective pain experience. It was designed by Roland Melzak in 1975, but later he recognised that the instrument was too long for use in clinical trials and introduced a short-form McGill pain questionnaire (SF-MPQ) (Melzack, 1975, Melzack, 1987). The SF-MPQ consists of 15 descriptors (11 sensory and 4 affective), along with a visual analogue scale for pain intensity. The sensory and affective descriptors are rated on an intensity scale as 0 = none, 1 = mild, 2 = moderate or 3 = severe. A maximum score of 55 indicate severe symptoms (Melzack, 1987).

Douleur Neuropathique en 4 Questionnaire (DN4)

The DN4 questionnaire is a widely used questionnaire in NP clinical practice. It was developed in France in 160 patients with either NP or nociceptive pain. It has been translated into several languages (Bouhassira et al., 2005). Please see the use of DN4 and the Methods section for an illustration of the score described here.

4.3.2 Clinical signs assessment

The assessment of clinical signs in patients with sensory abnormalities and pain is essential, because it helps the diagnosis and classification of patients. A range of methods from simple bedside examination to more sophisticated neurological techniques such as nerve conduction study and QST, are used to assess sensory abnormalities (Haanpaa et al., 2011a).

There are two definitive steps the clinician can use when examining patients for the diagnosis of the pain-related sensory abnormalities. Firstly to confirm that some form of damage to the somatosensory system has occurred, without this the diagnosis is impossible. The sensory findings should be neuroanatomically logical and compatible with a definitive lesion site (Treede et al., 2008). Secondly, to determine that any pain is indeed neuropathic, this because the occurrence of nerve damage does not necessarily follow that any pain is neuropathic in origin.

For the purposes of this study, an ideal test for an individual patient should:

- i. provide specific information about pathophysiological nature of the mechanisms of disease when disease processes are taking place
- ii. have sensitivity and specificity and positive predictive accuracy
- iii. be interpretable within the context of clinical practice
- iv. provide information about where the pathophysiology of pain takes place, i.e., the location of a lesion for neuropathic pain.

4.3.2.1 Clinical examination

Clinical examination is of paramount importance in assessing neurological disorder. It involves sensory, motor and autonomic signs (Hansson et al., 2001). The examination involves a series of systematic steps, including past and present history, a detailed description of pain distribution, quality and intensity of pain and a neurological examination with the emphasis on sensory testing (Jensen et al., 2001, Hansson, 2002).

Standard bedside neurological sensory testing

Neurological examination involves assessment of muscle tone and power, tendon reflexes and sensory examination. Sensory examination is critical in providing evidence necessary for the diagnosis of NP (Haanpaa et al., 2011b). Sensory abnormalities are mapped out using cotton wool for tactile sensation (A β fibres), pin-prick sensation with a small pin or cocktail-stick (A δ fibres), gross temperature sensation with warm or cool objects (warm – C-fibres, cool – A δ fibres) and vibration sense with a tuning fork (128Hz) (A β fibres) (Cruccu et al., 2004). This examination is performed to determine the presence of suspected sensory loss or gain of function that are associated with NP and at the same time to document the presence of allodynia, hyperalgesia and hyperpathia as hallmarks of NP.

However, clinical examination of a patient with pain which aims to assess the somatosensory system and may provide supporting evidence for altered function of the nervous system, can never prove any pain to be of neuropathic origin. In patients with a possible NP condition, the sensory dysfunction is coordinated by somatosensory abnormalities (Treede et al., 2008), and cannot be determined by clinical examination alone. For instance, if afferent fibres from the skin are affected, sensory abnormalities can be detected using simple bedside tools for supra threshold stimulation (Haanpaa et

al., 2011a). The light touch of cotton wool and the vibration stimulus of the tuning fork, which activate large A-beta fibres, as well as the dorsal columns and their thalamocortical extension, are used to assess the sensitivity to touch and vibration, respectively. Other test stimuli, such as cold and warm metallic rollers and pins could be used to assess thermal and pain sensation and hence activate different types of A-delta and Cfibres and spino-thalamo-cortical system. But, despite its clinical significance and identification of the neuroanatomical distribution of symptoms, pain in an area with sensory dysfunction is not to be associated with NP since other types of pain may be expressed in such an area (Hansson, P. and Lindblom 1993). A study by Freynhagen et al, reported that 5 out of 12 patients classified with painful radiculopathy had normal sensory function on bedside examination (Freynhagen et al., 2008). The diagnostic prerequisite in NP conditions, i.e. sensory abnormalities in the distribution of the affected nervous structure, is not identified by bedside examination. Given that sensory abnormalities are not confined only to NP states, the outcome of sensibility examinations, especially by clinicians lacking experience in detailed sensory examination, could be a source of confusion and possible diagnostic errors. Another drawback of these tests is that they are often not sensitive enough to show longitudinal change. Therefore, the characteristics of the pain need further assessment to allow for its classification. Quantitative Sensory Testing of perception thresholds in the above mentioned somatosensory channels could be used to complement the assessment of somatosensory abnormalities at bedside examination (Leffler and Hansson, 2008a).

4.3.2.2 Conventional electrophysiological studies

Nerve conduction studies

Nerve conduction study is considered as an extension of the patient history and clinical examination. It comprises an electophysiological test that assesses the motor and sensory function of the nerve. These are obtained by stimulating the nerve and a response is recorded directly from the nerve or from a muscle innervated by the tested nerve. In this way, NCS are able to detect and differentiate general from focal nerve abnormalities, type of neuropathy (demyelinating, axonal and conduction block), type of involved axons (motor, sensory and autonomic) and localisation (distal, proximal and entrapment site).

While NCS is a relatively non-invasive method, its main limitation is that it cannot be used to assess small fibre neuropathy or the function of nociceptive pathways (Cruccu et al., 2004). In such conditions, special methods are used such as testing thermal thresholds and quantitative mechanical testing. Quantitative sensory testing, in general, quantifies the functional status of the peripheral nervous system by nerve fibres of various sizes and by central pathways. The QST is most useful in the diagnosis of small fibre neuropathy.

4.3.2.3 Quantitative measurements

"Quantitative measurement" describes tests where the intensity and characteristics of the test stimulus are well controlled and reproducible, and the detection threshold is determined in parametric units that can be compared to established normal values. Stimuli are usually delivered in accordance with specific testing algorithms, and the subject's response is predefined according to standardised instructions. A number of instruments and validated tools for quantitative assessment of neuropathy have been established, ranging from simple instruments such as monofilaments to more sophisticated computer-aided systems such as quantitative sensory testing (Valk et al., 1997, Rolke et al., 2006a).

Semmes-Weinstein Monofilaments (MFs)

Monofilaments, which were originally made from horsehair, have been developed over time from simple, natural materials to synthetic devices. In the late 19th century, the horsehair was used as a method to quantify pain induced by punctate stimulation (Weinstein, 1968). Von Frey used various thicknesses of horsehair to determine the thresholds of touch recognition. Later this technique was refined and amended by others, such as Semmes and Weinstein in the 1960s. They developed a standard set of nylon monofilaments that exert predefined forces onto the skin (Semmes et al., 1960). Now, however, a more field friendly method has been introduced; Von Frey hairs made from optical glass. A testing kit comprises a standard set of glass filaments which are widely used in clinical practice by neurologists for assessing sensory abnormalities.

Semmes-Weinstein Monofilaments are a standard set of six coloured monofilaments ranging from 5mg, 200mg, 2g, 4g, 10g and 300g (Bell-Krotoski, 1990). When used for the hand specifically the stimulus was found to have a cut off 200mg, but for the foot the cut off was found to be 2g. Semmes-Weinstein Monofilaments were developed to

detect sensory loss in leprosy programmes but have been widely used by other programmes as a diagnostic technique for routine clinical and research purposes (Jamison, 1969). Their use has led to a significant improvement in screening diabetic patients; for example inability to feel a 10gm monofilament is recognised as a risk factor for ulcer (Birke and Sims, 1986).

Both MFs and VMT using a modified MRC scale are suitable and reliable tools for measuring nerve function impairment in a resource-limited setting (Brandsma et al., 2014). In a cohort study of 357 untreated multibacillary patients from India between 2001 and 2005, which assessed the sensitivity and specificity of MFs and VMT using nerve conduction study as gold standard for detecting nerve function impairment, the authors noted that both methods have good inter-tester reliability and reducibility and good specificity of more than 80%. The sensitivity of both tests in detecting nerve involvement was low, less than 40% (Khambati et al., 2009).

Quantitative Sensory Testing (QST)

Quantitative sensory testing is defined by the Peripheral Neuropathy Association as the technique(s) used to measure the intensity of stimuli needed to produce specific sensory perceptions (PNA., 1993). For more details on QST please, see section 4.6.

4.3.3 Skin biopsy

Intra Epidermal Nerve Fibre Density (IENFD)

Intra-epidermal nerve fibre density is a technique for measuring the endings of small peripheral nerve fibres in the epidermis (Lauria et al., 2005). It has been used for identifying the presence of nerve damage (Lauria et al., 2005). Both myelinated ($A\beta$ and $A\delta$) and unmyelinated (C) nerve fibres can be assessed (Devigili et al., 2008). These nerve fibres along with sweat glands, blood vessel, epidermis cells and superficial dermis are investigated using skin biopsy. Skin biopsy is a safe and reliable technique used to investigate IENF (Lauria et al., 2005). Nerve fibres are immunostained by antibodies against PGP 9.5 using either immunohistochemistry or immunofluorescence, and fixed by 2% paraformaldehyde-lysine-periodate (2% PLP) or Zamboni's solution. Fibres crossing the dermal epidermal junction are counted and quantified to confirm the clinical diagnosis of neuropathy (Lauria et al., 2005). The density is calculated in at least three sections as the number of IENF per length of the section (IENF/mm) (Lauria et al., 2005).

al., 2005). Devigili et al. who screened 486 Italian patients and collected 124 patients with neuropathy found 67 patients had small fibre neuropathy using skin biopsy (Devigili et al., 2008). The author noted that quantification of IENF density in skin biopsy for diagnosis of neuropathy showed a diagnostic efficiency of 88.4% (Devigili et al., 2008).

European Federation of Neurological Societies (EFNS) has recommended the use of skin biopsy with linear quantification of IENF density as a reliable and efficient technique to confirm the clinical diagnosis of small fibre neuropathy (Lauria et al., 2010).

Previous studies on IENF, pain and QST parameters

As peripheral neuropathic pain abnormal sensations may be related to dysfunction of A β , A δ or C- fibres, IENFD may be correlated with pain. Studies have identified IENFD, pain and QST parameters correlation in post herpetic neuralgia, Diabetics and HIV pain neuropathy (Zhou et al., 2007, Sorensen et al., 2006, Devigili et al., 2008, Loseth et al., 2008, Vlckova-Moravcova et al., 2008a).

In patients with post herpetic neuralgia, the number of intra-epidermal fibres is lower in the biopsies taken from pain area compared to the control site. (Oaklander, 2001) Sorensen et al. 2006, who studied the correlation between IENF density and pain in 38 patients with diabetes using skin punch biopsy, noted that IENF density was significantly lower in the biopsies taken from those with pain compared with those without pain (Sorensen et al., 2006). A similar conclusion, greater fibre loss correlated with more severe pain, was obtained also by Zhou et al. 2007, who studied the relationship between IENF density and pain in 101 patients with HIV neuropathy (Zhou et al., 2007).

Other studies have shown the correlation between IENFD and QST parameters. Studies carried by Devigili et al. 2008, Loseth et al. 2008 and Moravcova et al. 2008, who assessed patients with sensory neuropathy to demonstrate the relationship between the IENF density and QST, have shown an inverse correlation between IENFD and thermal threshold, a significant correlation between IENFD and QST parameters and IENFD correlated with warm detection threshold on QST, respectively (Devigili et al., 2008, Loseth et al., 2008, Vlckova-Moravcova et al., 2008a).

4.4 Techniques for assessing psychological impact of neuropathic pain

4.4.1 Introduction:

Measurement of subjective experience, such as pain, inevitably relies heavily on selfreport, which is valuable but its impact is difficult to determine. The complexity of pain phenomena in an area of sensory loss, also known as NP, is a factor that hinders its management and control. For instance, pain intensity and interference with function can have a marked impact upon daily routine, affective and motivational states, social relationships, sleep and economic factors. This indicates the importance of evaluating the relationship between the intensity of pain, disability and depression with the quality of life of individuals with chronic pain. Instruments, such as quality of life questionnaires are used to deepen knowledge of the perceived pain and thus enable an evaluation of effectiveness of the treatment used. I will now describe the instruments that have been validated and are reliable in assessing psychological impact of NP.

4.4.2 Quality of life and psychological state questionnaires literature review

Health is defined by the World Health Organization (WHO) as a state of complete physical, mental and social well-being and not merely the absence of diseases and infirmity (WHO, 1946).

4.4.2.1 General Health Questionnaires (GHQs)

The GHQs is the most widely used standardised self-completion measure of psychological well-being globally. It was developed by Godberg in the 1970s for use in primary practice settings (Goldberg, 1972). Its main focus is to assess psychological components of ill health, in particular screening for common psychiatric disorders such as anxiety and depression.

Content: the initial version of the GHQ contains 60-items which measure common mental health problems/domains of depression, anxiety, somatic symptoms and social withdrawal dimensions. Shorter versions of 30, 28, 20 and 12-items have also been developed. The 12-items version, is in fact, as efficient as the 30-items version as a case detector (Bowling, 2005). The questions assess psychological well-being state over the past few weeks, including:

Have you recently:

- Been able to concentrate on whatever you are doing
- Lost much sleep over worry
- Felt that you are playing a useful part in things
- Felt capable of making decisions about things
- Felt constantly under strain
- Felt you couldn't overcome your difficulties
- Been able to enjoy your normal day to day activities
- Been able to face up to your problems
- Been feeling unhappy and depressed
- Been losing confidence in yourself
- Been thinking of yourself as a worthless person
- Been feeling reasonably happy, all things considered

Scoring: Detailed instructions on the rating, coding and scoring procedures are described in Chapter 6.

The GHQ-12 is one of the most commonly used screening tool for mental health assessment in community settings (Furukawa and Goldberg, 1999), and has been validated in other mental disorders around the world (Goldberg and Williams, 1988, Goldberg et al., 1997). Particularly relevant to this study is its validation both in India and leprosy (Gautam et al., 1987, Verma and Gautam, 1994, Senturk et al., 2007, Jindal et al., 2013, Bandyopadhyay et al., 1988, Sriram et al., 1989). Further, The GHQ-12 has been demonstrated to be a valid screening in patients with leprosy NP (Lasry-Levy et al., 2011, Haroun et al., 2012)

4.4.2.2 Brief Pain Inventory (BPI)

The short form of the BPI is constructed of nine self-report items and measures two main domains: the intensity of pain (sensory dimension) and interference of pain in the patient's life (reactive dimension). It is probably the most widely used measurement scale for clinical pain (Cleeland and Ryan, 1994), and its validity comes from several studies of cancer pain and pain of other diseases. It also demonstrates good test-retest item correlations over short time intervals (Daut et al., 1983). Patients rate their pain on a 11 – point numerical scale for the average, worst and current pain in the preceding 24 hours. The second part of the questionnaire ask patients to indicate the extent to which

pain interferes, on a scale of 0 ('pain has not interfered') to 10 ('pain completely interfered'), with the daily activities addressed by the subscales (e.g. general activity, mood, mobility, normal work, relations with others, sleep, enjoyment of life, self-care, recreational activities and social activities).

4.5 The use of DN4 and PD-Q questionnaires for assessing NP

This section describes the items and structure of the main questionnaires, namely; DN4 and PD-Q, used for the screening and measurement of NP in this study, as well as their potential value and limitations.

4.5.1 Overview:

DN4 consists of 10 items: seven interview items and three clinical signs. The interview items are related to the quality of pain (burning, painful cold, electric shocks) and its association to abnormal sensation (tingling, pins and needles, numbness, itching). The clinical signs are related to sensory examination in the most affected area (touch hypoesthesia, pinprick hypoesthesia, tactile allodynia) (Bouhassira et al., 2005). Items are grouped in four sections; each one requires yes or no responses to questions on the quality of pain or clinical signs. The DN4 rating and scoring is simple; a score of 1 is given to each positive item and a score of 0 to each negative item. The total score is calculated as the sum of the 10 items and the cut-off value for the diagnosis neuropathic pain is a total score of 4 out of 10 (Bouhassira et al., 2005).

DN4 is often compared to other screening tools for neuropathic pain (LANSS, NPQ, PD-Q); however some differences should be recognised. In particular, these tools require no clinical examination. In contrast, the full versions of DN4 and LANSS are clinician-administered questionnaires. In these screening tools the clinical signs are tested by the examiner. Short versions of DN4 (DN4-Interview) and LANSS (S-LANSS), which omit the items related to sensory examination, have been developed for use as self-administered questionnaires (Bouhassira et al., 2008, Bennett et al., 2005). Another relevant difference from screening tools is the method of validation. The DN4 validation study included patients with either peripheral or central NP, whereas other studies included only patients with peripheral NP. In addition, the DN4 validation study included only patients with pure NP, while other studies included patients with mixed pain (PD-Q) or complex regional syndrome type 1 (LANSS). The number of items,

their phrasing and the scoring methods also differ between DN4 and the other screening tools.

Despite the methodological differences between DN4 and other questionnaires mentioned above, it appears that most of the DN4 items are also present in the final versions of these questionnaires. For instance, the DN4 pain descriptors "hot or burning," "shooting or electric shock," "numbness," "tingling," pins and needles" and items related to abnormal clinical signs "allodynia" are used in all the other questionnaires. This may suggest that DN4 has the main common symptoms of NP conditions.

4.5.2 Limitations of the DN4 and PD-Q questionnaires

As will be discussed in Chapter 5, DN4 has become one of the most common screening questionnaires for NP; in particular for the identification of possible NP among leprosy patients. As discussed earlier, no single symptom is diagnostic of NP, but combinations of certain symptoms, pain descriptors and clinical findings increase the possibility of a NP condition. Both DN4 and PD-Q attempt to provide an accurate selection of patients with symptoms and signs suggestive of NP. However, as in other questionnaires, the following issues have been identified as drawbacks:

First, DN4 has limited diagnostic value in patients with widespread pain. It has been validated in patients with pain at a single body location. Their ability to distinguish between pain and pain-free neuropathy is reliable only when applied to a limited painful area. A study carried by Attal and colleagues, which investigated the neuropathic components of chronic low back pain in 132 patients with and without lower limb pain using the DN4 questionnaire, showed that DN4 can be used to assess patients with up to three pain locations (Attal et al., 2011b). Hence, DN4 is less practical in patients with multiple pain locations, i.e. more than three. Secondly, DN4 is increasingly used for NP epidemiological studies in different settings, but validation studies for this purpose are necessary. Furthermore, NP screening questionnaires fail to identify 10 to 20% of patients with clinician diagnosed NP (Haanpaa et al., 2011a). This implies that screening questionnaires cannot replace clinical judgment, but may play role in guidance for further diagnostic evaluation and pain management. Another limitation of the DN4 questionnaire is that it provides no information about the relationship between symptoms and lesions or disease mechanism. This is illustrated by Rasmussen et.al, in a

study that compared verbal pain description and detailed sensory testing using the short form of McGill pain questionnaire (Rasmussen et al., 2004). The authors examined 214 patients with suspected chronic NP of moderate to severe intensity. They proposed clinical criteria for NP based on pain aetiology and presence of pain sensory loss, and labelled patients as having "unlikely", "possible" and "definite" NP. The authors found no differences in verbal descriptions across the groups. The identification of NP conditions may require a further clinical examination, together with imaging, laboratory, or electrophysiological tests in some cases (Baron et al., 2010a, Haanpaa et al., 2011a, Haanpaa et al., 2009). Finally, PD-Q has limited applicability for assessment of the effects of treatment.

4.6 The use of QST methods for assessing sensory abnormalities and NP

4.6.1 Introduction

This section explores the background information behind quantitative sensory testing measurement of sensory function, the basic principles of QST, the type of information obtained and their potential clinical utility as an aid to the diagnosis of NP. There are several protocols for QST and this will be discussed further in section 5.2.2 under available protocols for QST. A comprehensive overview of the detailed parameters and methods of DFNS-QST protocol will be provided in Chapter 6.

4.6.2 History and background

Quantitative sensory testing (QST) in medicine has roots in quantification and noninvasive testing and the first descriptions of the potential of QST as a standard evaluation procedures took place in the 1970s. A landmark pioneering publication by Fruhstorfer (Fruhstorfer et al., 1976) described the use of QST for thermal thresholds that may detect preclinical diabetic neuropathy. Fruhstorfer's group went on to develop a quantitative technique for the examination of thermal sensibility and, in parallel with other researchers across the world, developed a method for quantitative thermal thresholds in the late 1970s. At this early stage, the main advantage of the quantification technique was that its ease of use enabled it to be employed routinely and repeatedly in patients with symptoms of, or the potential, for neurologic damage or disease. Since then, there has been increasing interest in using QST to give insights into the underlying pathophysiological mechanisms of pain. The next break through was the development of various quantitative methods for assessing sensory abnormalities, such as Von Frey hairs for touch, which enable the clinician to assess other modalities of sensory abnormalities, rather than just using a selected thermal test. A major improvement in quantitative sensory testing quality arrived with the development of electronic devices in the late 1980s and early 1990s. Table 4.2 shows the devices cleared by the Food and Drug Administration (FDA) in the 1980s and 2000s. Despite the advantages of being non-invasive, non-interventional techniques to complement standard neurological bedside examination, in order to help detect and quantify positive and negative sensory phenomena, there was no standardised QST testing procedure. For instance, there was no consensus regarding which specific QST device or algorithm should be used preferentially. This showed the growing need to develop standardised QST protocols. In recent years, the German Research Network on Neuropathic Pain-DFNS has developed a standardised and comprehensive QST protocol (Rolke et al., 2006a).

The emergence of DFNS-QST as a new and comprehensive protocol for quantifying somatosensory changes in human skin and even pain-related phenomena has made it the technique of choice for assessing diseases of the central and peripheral nervous system. The DFNS-QST battery consists of seven tests measuring 13 parameters, including various types of mechanical and thermal detection and pain thresholds for the hand, foot and face. The QST protocol was implemented in 180 healthy volunteers, thus providing a complete profile of sensory function (gender, age and location matched) and normative data that can be used as reference values for statistical analysis in studies on patients with NP (Rolke et al., 2006a). By affording such accurate measurement of sensory loss and gain, as well as psychophysical responses, DFNS-QST offers a high degree of detailed and precise information in the clinical diagnosis domain.

Year	Product	Comments
1987	Thermal Threshold Tester (ITT)	Teca, Inc.
1992	CASE IV Computer Aided Sensory Evaluator	Vibration & thermal threshold testing
1993	Thermal Sensory Analyzer (TSA)	Medoc Corporation
1994	Nk Pressure-Specified Sensory Device	NK Biotechnical Corporation

Table 4.2. Devices cleared for marketing by the FDA (1980-2003)

1994	Neurometer	Current Perception Threshold
1994	Pressure-Specified Sensory Device	Sensory Management Services LLC
1997	Medi-Dx 7000	Neuro Diagnostic Associates
2003	Vibration Perception Threshold (VPT) meter	Xilas Medical

4.6.3 Literature review of QST

In this section, I describe the basic principles and information behind QST. I review the different methods of stimulation that are currently available and provide an overview of the methods used by the DFNS. For the purpose of the current study, the DFNS-QST protocol will be the main protocol for QST.

4.6.4 Overview of the QST principle

QST is a non-invasive, sophisticated clinical examination of the sensory nervous system, whereby the perception thresholds to various thermal and mechanical sensory stimuli are accurately measured using the QST battery. The thermal testing modality assesses small myelinated (A- δ fibre) and unmyelinated (C-fibre) sensory nerve function, whereas the mechanical testing (light touch and vibration) evaluates the large myelinated A- α and A- β sensory fibres (Vinik et al., 1995). The pain modality, heat-pain and cold-pain threshold tests have been found to document increased pain sensitivity (hyperalgesia, allodynia and hyperpathia) (Verdugo and Ochoa, 1992). Others, such as mechanical pain and pressure pain have been used to document hyperalgesia (dynamic and static) and pain sensitivity (cutaneous and deep), respectively (Ochoa and Yarnitsky, 1993, Treede et al., 2002). Therefore, QST and in particular DFNS-QST, can study large myelinated, small myelinated and unmyelinated fibres in addition to documenting sensory gain and loss (hyperalgesia and hypoesthesia).

4.6.5 QST algorithms

A number of algorithms of testing and finding thresholds are used to quantify the sensory thresholds and pain-related phenomena in clinical practice. The test should yeilded accurate and reproducible results within a reasonable amount of time. Tests for pain sensation have the additional requirement of minimizing the number of stimuli that are unpleasant to the patient. In QST, the method of limits and the method of levels are the most common algorithms used. This makes QST reasonably reproducible over the course of several days to a week (Heldestad et al., 2010). However, describing an algorithm in these terms does not in itself ensure that a particular standard or adequate algorithm is being used. A number of issues should be taken into consideration for any QST algorithm.

Factors influencing QST algorithm:

- Type of instrument
- Room temperature and humidity
- Site of stimulus
- Patient related factors: age, gender, cooperation and motivation
- Availability of standardised protocol.

The method of limits and the method of levels – the two common algorithms for pain threshold - are described in more detail. Others, such as tolerance, magnitude estimation of supra-threshold pain intensity and summation are excluded from the present review.

Method of limits: is one of the most commonly used algorithms for quantitative sensory testing (Figure 4.2). The threshold is determined with ramped stimuli that are stopped immediately when the subject presses a button. Two types of ramp stimuli are used: the ascending ramp and the descending ramp. In the former, the intensity of the stimulus is gradually increased until the subject perceives the stimulus as painful; the so-called appearance threshold. Another, less common, approach is the determination of a disappearance threshold, by decreasing the stimulus intensity until it is no longer detected. Thermal and vibratory thresholds are frequently assessed with this method (Verdugo and Ochoa, 1992, Chong and Cros, 2004). The advantage of this algorithm is that pain threshold can be determined very quickly (Dotson, 1997). Thus the chance that fatigue, loss of motivation, inattention and malingering will occur is minimal.

The feedback mechanism, however, is the main disadvantage of the method of limits. The subject's reaction time is the period between processing the information and the subject's indication of a response. For example, a subject needs to perceive the stimulus, process the information and generate an action to indicate a response. This may lead to an overestimation of the pain threshold. The extent of overestimation of the threshold depends on the rate of stimulus change, i.e., the slope of the ramp (Dyck et al., 1990, Shy et al., 2003, Chong and Cros, 2004, Hansson et al., 2007).

Method of levels: is an algorithm method that obtains results by applying a series of predefined stimuli to the skin in ascending or descending order (Figure 4.2). The stimulus has a defined intensity and duration and the subject has to choose whether or not the stimulus is felt after each trial. Hence, it is also referred to as "forced choice" algorithm (Shy et al., 2003).

While the method of levels does not depend on reaction time, its main limitation is that it cannot be used frequently. The method of levels is generally more time-consuming and may subsequently lead to boredom and inattention (Hansson et al., 2007).



Figure 4.2. Summary of the methods used for QST

4.6.6 **QST** instruments

Detailed descriptions of the thermal and mechanical instruments are described in Chapter 6.

4.6.7 QST and standard bedside neurologic sensory testing compared

While the focus of this study is the profiling and stratification of leprosy patients, it is useful to make a comparison, in this preliminary chapter, of the known advantages and disadvantages of the validated techniques. I have summarised the practicalities of the QST compared to other methods in Table 4.3 and Table 4.4.

	Quantit	ative Sensory Testing	Standard bedside sensory testing	
	0 3	Stimuli delivered in	0	Highly reproducible (Bouhassira
	6	accordance with specific		et al., 2005)
	t	testing algorithms	0	Generates and ranks other types
	0	Greater precision		of pain as matter of differential
	0	Potential possibility of		diagnosis
		diagnosis of sensory	0	Answers the question where on
	1	neuropathy (Shy et al.,		the somatosensory system is the
Advantages	2	2003, Chong and Cros,		pathology that generates
	2	2004)		neuropathic pain
	0	Possible to follow up	0	Can distinguish between pain
				and pain-free condition
				(Rasmussen et al., 2004)
			0	More sensitive than QST
	0 1	Time-consuming	0	Stimuli are not calibrated
Disadvantages	0]	Lack standardization	0	Not able to prove any pain to be
	0 5	Subjective		neuropathic origin

Table 4.3. QST and standard bedside neurologic sensory testing

	Quantitative Sensory Testing	Conventional electrophysiological		
	(QST) (Cruccu et al., 2004)	techniques (NCS)		
	o Psycho-physical	o Does not require responses		
		from the subject		
		• No active cooperation		
		required		
		• Useful to localise the nerve		
		lesion, its severity and to		
		suggest prognosis		
Advantages		• Training required for		
		investigators but not for		
		subjects		
		• Published normative data		
		and available data from most		
		electro-physiological		
		laboratories		
		0 Objective		
	• Requires a response from	• Tests only large fibres		
	the subject, so is subjective	, , ,		
	and not objective			
	o Standardisation			
	o Time-consuming			
Disadvantages	• High cost			
	• Lacks reproducibility (Bird			
	et al., 2006)			

Table 4.4. QST	and conventional	electrophysiol	ogical techniques
----------------	------------------	----------------	-------------------

*cited in (Backonja et al., 2013)

4.7 Previous studies on painful and painless neuropathy using QST

Several studies have considered neuropathy and pain using QST, (Table 4.5). The main findings of these studies were encouraging as it appeared that QST parameters may

show distinct changes in patients with painful neuropathy compared to patients with pain-free neuropathy.

Vrethem et al. measured the different responses to the panel of stimuli used in the QST in 55 diabetic patients, with neuropathy, in Sweden. The study showed that touch was more affected in patients with painful neuropathy compared with patients with painless neuropathy; otherwise, there were no differences between the patient groups (Vrethem et al., 2002). In HIV-related neuropathy, Martin et al. 2003 examined 36 HIV infected patients with painful (20 patients) and non-painful (16 patients) sensory neuropathy assessed by clinical, quantitative thermal testing and nerve conduction examination. Control reference data were obtained from 49 healthy participants with a corresponding age and sex match. The authors showed that patients with painful neuropathy had a significantly lower cold pain threshold than healthy controls which demonstrates impairment of C-fibres function (Martin et al., 2003). A similar conclusion, low cold detection threshold associated with HIV neuropathy compared to healthy controls, was also obtained by Simpson et al. 2002, who studied a cohort of 270 HIV patients from the United States (Simpson et al., 2002). In contrast, Ulf et al. 2002, who assessed the patterns of sensory changes caused by different conditions using the QST parameters in 30 patients with dysaesthesia and 15 controls in Germany, noted no association of any parameter obtained by QST with a particular disease (Baumgartner et al., 2002). In a QST profile study of 66 HIV infected participants, Phillips and colleagues (Phillips et al., 2014), reported no differences in regard to thermal and mechanical perception thresholds between painful and non-painful sides of denervated skin.

Despite the differences in assessing sensory changes caused by different conditions in these studies, QST may help to identify the sensory modalities mediated by different nerve fibres. In diseases such as PHN and traumatic lesions QST parameters were found to differentiate between patients with and without pain, whereas in DM and HIV neuropathy the finding were not generalised. Overall, the findings emphasise the importance of sensory profiles of patients who presented with neuropathy and pain, which might help in grouping patients according to the changes of the sensory patterns identified by QST.

Authors,	Purpose of	Study		Interventio	Main findings
year,	the study	population,	Controls	n / test	
country			Controls		
Vrethem et al. (Vrethem et al., 2002) Sweden	To study pain characteristic s in diabetic patients	55 patients: painful neuropathy 37, painless neuropathy 18	Healthy controls: 14 and non- DM neuropathy: 18	Clinical, neurophysio logy and QST	there were no differences between the patient groups, except for touch (p = 0.02)
Martin et al. (Martin et al., 2003) Sweden	To investigate sensory profile in HIV patients	36 AIDS patients: 20 painful, 16 non painful	Healthy controls: 49 participants	Clinical, QST and NCS	Warm Hypothaethesia (p =0.06) and CPT (p =0.03) gain greater in painful group
Simpson et al. (Simpson et al., 2002) United States	To investigate HIV associated neuropathy	236 patients with HIV neuropathy nested from a cohort of 270	No controls	QST	CDT associated with pain in HIV neuropathy
Fitzek et al. (Fitzek et al., 2001) Germany	To identify clinical predictors in patients with pain	12 patients with dorso lateral medullary infarction	No controls	Thermal testing and MRI	Facial pain predicted by loss of pinprick heat/cold pain, TSL
Finnerup et al. (Finnerup et al., 2003) Denmark	To investigate NP mechanism	40 patients with SCI: painful 20, pain free 20	Healthy controls: 20 participants	Clinical, QST and somatosens ory evoke potentials	At lesion level hyperalgesia differs (p =0.03)
Ulf et al. (Baumgart ner et al., 2002) Germany	To assess underlying mechanism of NP	Patients with dysaesthesia: 30	Healthy controls: 15 participants	MDT, S/R (stimulus/re sponse) functions	No association of QST parameters in a particular disease
Phillip (Phillips et al., 2014), UK	To assess sensory profile in HIV patients	HIV-SN: 38 HIV-no SN 28 patients	Healthy controls: 66 participants	Clinical, QST, NCS, IENFD	No single QST parameters were different between the groups

Table 4.5. QST studies on painful neuropathy and painless neuropathy

4.8 Chapter summary

This chapter summarised neuropathic pain and the methods, particularly profiling methods, used to aid in its diagnosis and impact. Only two of the profiling methods discussed have the capacity of adding somatosensory profiling measurements: QST and skin biopsy. While QST techniques are well established in term of quantification of sensory profile, they are often limited by its applicability in limited resource setting. QST techniques have the advantage of being able to test the entire sensory axis from receptors to brain. However, as will be outlined in Chapter 5, the use of QST techniques to quantify the peripheral sensory system has not yet been fully investigated in leprosy.

Chapter 5 The clinical problem and aims of the thesis

5.1 The clinical difficulties in NP in diseases such as leprosy

5.1.1 Problems of defining neuropathic pain in leprosy

5.1.1.1 Case study: a problem of identifying a case of NP in leprosy

Case history 1:

A 49 year old female from Worli village, Mumbai sub-urban district. She is a service worker and the family's only earning member. She has received full treatment for multibacillary (MB) leprosy, completed in September 2002. She has no other medical comorbidities and notably has no clinical evidence of hypertension or diabetes mellitus on laboratory investigations.

Presenting complaint

Severe pain, tingling, numbress and burning sensations in both hands and feet bilaterally for the last 5 years

Origin, duration and progress of leprosy

In 2001, when the patient was 37 years of age and in the third trimester of her first (and last) pregnancy, she presented with painful nodules on her leg arms and back with associated fever and joint pain. She was admitted to hospital, diagnosed as having MB leprosy with erythema nodosum leprosum reaction (ENL), and commenced one year multi-drug treatment (MDT). She had a normal vaginal delivery, and gave birth to healthy baby with no medical problems.

Post-delivery she had repeated episodes of ENL reactions. During the first episode, in 2001, she received a six months course of prednisone. The second episode in June 2003 was managed with a further six months of prednisone. After the third recurrent episode of ENL reaction in 2004, she was diagnosed as chronic ENL and commenced and completed another 6 months treatment with prednisone.

After anti reaction treatment, she showed steady improvement. Her pain and skin lesions subsided, and she felt better. However, seven years later in 2008; she developed

burning sensations, tingling and numbness in both hands and feet. By late 2011, she was admitted to hospital with bilateral weakness, sensory loss and severe burning sensations in her hands and feet. Two weeks later, she complained of insomnia due to her pain. The pain in her hands and feet was described as sharp, stabbing, burning and "electric shock" like and the patient described it as "putting her hands and feet into a fire". No formal assessment of her pain symptoms was made by the attending physician. Neurological examination of her limbs was normal. She completed steroid treatment after consultation with a dermatologist to relieve pain, but her description of the pain remained as before. Treatment with MDT for the second time had been recommended by her doctor in a private clinic along with analgesics, but this did not improve her pain.

Radiological imaging (MRI of cervical spines) on 26/11/2012 was normal and included the basal ganglia. Nerve conduction studies (NCS) revealed predominately a severe sensory peripheral neuropathy in all four limbs, so a neuropathic mechanism was suspected as the dominant cause of her pain. The second course of MB-MDT was stopped after three months and her steroid dose was reduced. She commenced analgesic treatment with Amitriptyline 10 mg initially, increased to 75 mg which was effective in relieving her pain and then shifted to combination of Gabapentine 30 mg TDS and Duloxetine 20 mg BD. This produced substantially better pain relief without any disability for several weeks.

Assessment

Initially, her pain was considered to be nociceptive in type, i.e., pain caused by normal activation of peripheral nociceptors following tissue damage. It was managed by combination of MDT and increasing oral prednisolone up to 60 mg per 24 hours (median daily dose between 40 mg and 60 mg tapering over 6 months). Despite this she described only 50 % pain relief and showed symptoms and signs suggestive of psychological disturbance (GHQ score = 7). Over the last five years she had missed her work several times because of her pain.

Throughout her illness, this patient always described the pain as tingling, numbress, burning and "electric shock" like. Several pathological processes were suspected as aetiological factors such as, an immune mediated inflammatory process, leprosy relapse or reinfection. However, her clinicians were unable to elucidate the neuropathic component of her pain until a late stage and therefore the period of ineffective pain management could lead to psychological disability.

Case history 2:

An 18-year-old boy from Dharavi, Mumbai' largest slum (Figure 5.1). He is a right handed student. He was newly diagnosed with PB leprosy on 30.03.2013 and he is on pain medication for the last week. He has no other medical comorbidities and notably has no clinical evidence of hypertension or diabetes mellitus on laboratory investigations.

Presenting complaint

Severe pain, burning, and tingling sensations in the right medial aspect of the forearm, and associated with right hand weakness for the last three days.

Origin, duration and progress of the condition

The condition started in early March 2013 (four weeks prior to the current visit), when the patient developed an acute onset of pain sensations in the medial aspect of the right hand. The pain was described as "insects crawling" in his hand and associated with burning and shooting pain along the medial side of his right forearm. The patient sought medical advice at a private clinic and received combiflam pain-killers (a combination of Paracetamol and Ibuprofen), one tablet four times a day. After pain killer treatment, he showed improvement. His pain subsided and he felt better. However, one week later; he developed sever pain around his right ulnar nerve associated with reduced function of the right hand. By late March 2013, he was referred to the BLP clinic by a leprosy affected member of the Dharavi slum community. At the BLP clinic he was diagnosed as having PB leprosy with neuritis, based on clinical examination: the presence of single skin lesions located on the face, and painful right ulnar nerve with reduced function. Further investigation such as skin smear was requested. The treatment plans was a high dose of prednisolone (60 mg), PB-MDT treatment for the next six months, and hand physiotherapy.

Assessment

This patients with leprosy presented with symptoms of acute pain while seeking treatment. Initially, the nature of the pain was nociceptive inflammatory pain, which is

usually amenable to treatment such as steroids, other anti-inflammatory medications, or immobilisation, but the nerve function impairment may continue even after starting MDT, which may serve as a source for the development of NP (Haanpaa et al., 2004, Haroun et al., 2012). This patients had right ulnar neuritis with sensory loss in the right ulnar nerve territories, hence the possibility of having acute NP cannot be ruled out.



Figure 5.1. Dharavi, Mumbai' largest slum

Source: this picture was taken in March 2013 in Mumbai

5.1.1.2 Summary of the case studies

From the above two different scenarios, leprosy patients with neuropathy and pain could benefit from a much clearer distinction between mono-neuropathies-the usual presentation of NP in leprosy-and distal symmetrical poly-neuropathies which are much more unusual.

5.1.2 Challenges in defining NP in leprosy patients

5.1.2.1 Introduction

NP, recognised to occur in approximately one-fifth of treated leprosy patients, is a major issue for patients and the health system care. Previous work has shown that patients with NP have significant levels of depression (Lasry-Levy et al., 2011, Haroun et al., 2012). For individuals, life with neuropathic pain can be disabling even after their disease has been "cured" adding a burden to the patient's suffering and health service cost. As diagnosis is challenging for clinicians, misdiagnosis often leads to further health and social consequences. Because of the resource poor setting we need better diagnostic

tests purely for identification as this the main barrier to effective treatment. This is caused by a number of obstacles which still exist in clinical practice.

The first difficulty is the variety of symptoms and signs of pain-related sensory abnormalities. The exact constellation of neuropathic pain characteristics, and associated sensory aberrations, which are manifested in an individual patient are variable and complex and often only fully revealed by the use of sophisticated investigation techniques (Jensen and Baron, 2003, Baron et al., 2009). This can result in common clinical features such as spontaneous pain or stimulus-evoked pain, as well as other associated sensory disturbances such as sensory loss (anaesthesia dolorosa).

It has been argued that rather than categorising neuropathic patients according to their aetiological diagnosis, the stratification of patients on the basis of individual symptom sensory profiles should be done to understand better the underlying processes (Baron et al., 2012). In a study from Germany, Baron and colleagues (Baron et al., 2009) examined symptom profiles in more than 2000 patients with diabetic neuropathy and post herpetic neuralgia using standard clinical examination and the PD symptom questionnaire. The authors found that patients with NP could be sub-grouped based on specific symptom profiles. The authors identified 5 subgroups where the symptom profiles were found to be different on the basis of the prominent features. For instance, subgroup 1 report spontaneous burning pain. Whereas, subgroup 2 demonstrates only severe pain attacks. In subgroup 3 the values of the sensory profile are mainly concentrated around the zero-line for all parameters. In contrast, subgroup 4 demonstrates considerable evoked pain symptoms and less burning sensations and paresthesia. Others such as subgroup 5 demonstrate considerable spontaneous symptoms without cutaneous allodynia or hyperalgesia. This indicates that the information obtained from pain questionnaires and clinical examination can be used to distinguish symptom profiles in patients with NP across different conditions. A similar grouping of NP patients based on sensory symptom profiles and co-morbidity, obtained in study of more than 2000 patients with painful radiculopathy (Mahn et al., 2011). These patterns do not necessarily reflect the condition from which the nerve damage arose, but importantly may well reflect pain generating mechanisms and therefore give a clue to mechanisms and thus likely drug responses on an individual patient level. Thus it is vital to accurately phenotype the sensory abnormalities in each patient with leprosy and NP.

Another major difficulty is the distinction between the different types of pain which are associated with leprosy. These are difficult to distinguish clinically, yet require different management strategies. Patients with leprosy may experience skin and nerve pain at different times during the disease. Skin lesions may affect sensory processing, but the pain mechanism differs from patients with nerve damage (i.e. not peripheral mechanism). Two main groups of leprosy patients experiencing pain can be distinguished: those with pain associated with reactions, and those with NP. However the two categories overlap (Haroun et al., 2012).

In addition to the difficulties of assessing the heterogeneous features following nerve damage and the distinction between the different types of pain, relationships between underlying pain and sensory deficit cannot be accurately determined. The process is based primarily on patients' descriptions supported by examination and investigation. Whilst any leprosy patients with neuropathy and pain is highly categorised as NP, the evidence of associations are required. This may not be a proven causation; neuropathy can be identified objectively, but it cannot be assumed that a causal relationship exists with the patient's pain. This scenario is further complicated, as illustrated in the preceding section, when mixed types of pain exist in the presence of a progressive immunological and pathological process or in the occurrence of pain in subclinical neuropathy. There are several advantages of accurately phenotyped abnormalities, if uncertainties regarding the relationships between different sources of underlying pain are to be avoided.

Finally, assessing an experience of pain itself is difficult. According to the IASP, irrespective of the underlying mechanism, pain is always subjective (IASP, 2009). Each individual feels and reports their pain experience differently, and the sensation of pain itself cannot be objectively measured. Even though certain behaviours have been identified as associated with pain, these are only corroborative. Also, although numerous psychophysical methods exist for the measurement of pain, these are entirely subjective (Chong and Cros, 2004).

The variability of nerve damage in leprosy, the existence of different types of pain, the uncertainty in the relationships between underlying pain mechanisms and the subjective experience of pain makes the identification of pain-related sensory abnormalities a continuing challenge. Therefore, a new classification of leprosy patients with NP could take into account subgroups of patients with different sensory profiles. This provides information about the pathophysiological process that helps improve understanding of the various NP mechanisms operating in leprosy.

5.1.2.2 How to define neuropathic pain in leprosy

For this current study, a two-step case definition was used to define NP in leprosy:

1) Is there evidence of nerve damage?

In accordance with the above definition the demonstration of nerve damage is an essential pre-requisite to diagnosing the presence of neuropathic pain.

2) Is the pain neuropathic?

Although demonstration of nerve damage is an essential first step in this diagnostic triage, the mere presence of nerve damage does not necessarily indicate that any pain is neuropathic in origin. Therefore, an essential second step is to classify any pain as being likely neuropathic origin. For this two criteria are required:

- i. Is the pain distributed in a "neuroanatomically plausible" location (e.g. a single peripheral nerve innervation)? A body chart where the pain location is drawn is used.
- ii. Is the symptom profile/ pain descriptors characteristic of neuropathic pain? For example by a score of $\geq 4/10$ using the DN4 questionnaire.

There are two definitive steps in the process: firstly to confirm that some form of damage to the somatosensory system has occurred, should be neuroanatomically logical and compatible with a definitive lesion site, without this the diagnosis is impossible (Treede et al., 2008). In leprosy, somatosensory system damage is a recognised complication. Body charts, which are routinely used in leprosy clinics to map skin lesions, can be used to identify distribution of pain by drawing pain location; in addition to patient's pain description (i.e. is the pain for example burning, stabbing or aching). The demonstration of pain distribution. Also, pain patterns such as glove and stocking distribution can be mapped on the body template. This may help in understanding the relationship between the location and quality of pain. Likert scales, Visual analog scales (VAS), Verbal rating scales (VRS), or a combination of verbal and

numerical rating (Gracely Pain Scale) can be used to determine the severity of pain. BPI can be used for assessing fluctuation of pain over time (Haanpaa et al., 2011a).

Secondly, to determine that any pain is indeed neuropathic, because nerve damage does not necessarily mean that pain is neuropathic in origin. To identify patients with possible NP, several screening tools are available such as the DN4, LANSS, NPQ and PD-Q. These screening tools are recommended particularly for non-specialists to consider the diagnosis of NP (Haanpaa et al., 2011a). It helps to determine the prevalence of NP in epidemiological studies, and it may also help to distinguish between inflammatory pain that is due to ongoing acute leprosy reactions and NP that is due to the effects of leprosy on sensory fibres. Nevertheless, these screening tools cannot replace clinical judgment in diagnosing NP in leprosy patients.

5.1.2.3 How to apply a case definition of NP in leprosy

Variation of defining NP in diseases such as leprosy

Different types of leprosy-related pain may occur during the course of the disease; inflammatory and NP (Haroun et al., 2012). Pain associated with neuritis reaction, which is defined by the development of inflammation of a nerve sheath without abnormal findings in sensory testing, is clinically defined as nociceptive pain (Bove and Light, 1997). Importantly this term has been noted not to be used unless inflammation is thought to be present (IASP, 2012). However, if an inflammatory neuritis causes nerve damage then the pain is by definition neuropathic. Another subtype of NP is neuralgia, which is defined as pain arising in the distribution of a nerve or nerves (IASP, 1994). Although neuralgia is the preferred term used generically to describe chronic pain following herpes zoster reactivation, it is used to describe NP arising from a lesion of specific nerves. NP in leprosy may occur even years after completion of the MDT. It usually occurs in distribution that is anatomically appropriate to the affected nerve(s) and in skin lesions.

5.1.3 The impact of diagnosis on treatment

Although there are no data from controlled studies in well-defined groups of leprosy patients that demonstrate the efficacy of the NP drugs, the treatment of the condition is less satisfactory. There are three principle reasons for this, the first being the difficulty in identifying and defining NP cases in leprosy. The second is the failure to identify the presence of NP mechanisms. The third is the use of treatment that is based on one specific mechanism when each patient with NP is likely to have more underlying causes leading to pain. For instance, clinicians often prescribe a tricyclic antidepressant when treating chronic pain in leprosy based on reports on the use of these drugs with no clinical evidence. Thus for some patients, even optimal use of current treatments by experienced clinicians will not relieve their pain.

5.1.4 The need of profiling and stratification of leprosy patients group

Despite the problems outlined above regarding the identification, defining and treatment of neuropathic pain in leprosy, a majority of leprosy patients will benefit from the new profiling and stratification of neuropathic pain in order to:

- i. improve clinical trial design
- ii. identify neuropathic pain mechanisms
- iii. shape the development of new drugs
- iv. individualise treatment leading to improved pain control

Given that the neuropathic pain mechanism-based approach is guided by targeted treatment according to pain mechanisms, it seems unlikely that there is a way to measure the mechanisms routinely. Therefore, identification of neuropathic pain mechanisms from symptoms and sensory profiles stratification would appear to be a sensible approach to inferring mechanisms operating at the individual patient level. In fact, Baron described an ideal situation in which subgrouping of patients with different sensory profiles guides the clinician in matching a particular treatment to a particular patient with predictable responses especially in clinical trial setting (Baron et al., 2012).

The identification of sensory profiles can be best achieved with validated questionnaires such as the DN4 or BPI as regards symptoms, and with an extension of the clinical examination such as QST for sensory signs (Haanpaa et al., 2011a, Backonja et al., 2013). For the purpose of this study the different approaches related to mechanisms and patients subgrouping are explained as follows: the mechanism-based is approach adopted by NP expert to target treatment with mechanisms; Patient profile (phenotype) is grouping of patients according to their symptoms and signs. Any profiling approach should stratify patients by symptom, sensory and psychological state. This has the potential to improve clinical trial design and might be adopted into routine practice.

5.2 The role of instruments for assessing sensory abnormality and psychological impact of NP in leprosy

This section describes the role of different instruments for assessing pain-related sensory abnormalities, their availability and justification for use in leprosy.

5.2.1 The role of pain questionnaires

5.2.1.1 Background

Pain questionnaires are tools that can accurately identify patients with symptoms and signs suggestive of neuropathic pain (Haanpaa et al., 2011a).

5.2.1.2 Available instruments

There are several validated questionnaires designed to identify neuropathic pain characteristic. These instruments can be classified into two groups: diagnostic screening and symptom profiling questionnaires. The former include LANSS, NPQ, DN4, PD-Q, StEP and ID-pain questionnaires, which have sensitivities ranging from 66% to 94% and specificities in the range 69% to 97% (Üçeyler and Sommer, 2011). Whereas, the symptom profiling questionnaires include NPS, NPSI, PQAS and SF-MPQ. For leprosy work the linguistic and cultural (context) validation are required in local languages and the DN4, LANSS and the NPSI have been used (Saunderson et al., 2008, Lasry-Levy et al., 2011, Haroun et al., 2012). I will present here the screening tools used in leprosy neuropathic pain, others are explained in detail in Chapter 3.

Pain screening questionnaires and leprosy

In recent years, more attention has been given to screening tools in identifying neuropathic pain among leprosy patients. The preference is given to a tool validated in the language in which it will be applied. Stump (Stump et al., 2004) described the use of McGill pain questionnaire which identified 53 (15%) patients with pain in sample of 358 Brazilian leprosy patients, but there was no information regarding the validation. In contrast, DN4 has been used for studies in India and Ethiopia (Lasry-Levy et al., 2011, Haroun et al., 2012). The different screening tools are summarised in Table 5.1. I elected to use the DN4 questionnaire.

Authors, year	Country	Study population	Tools
Hietaharju (Hietaharju	Bangladesh	16 leprosy patients with chronic	Clinical
et al., 2000)		pain	assessment
Stump (Stump et al.,	Brazil	358 leprosy patients from a referral	McGill Pain
2004)		centre	Questionnaires
Lund (Lund et al.,	India	17 leprosy patients who had	Clinical
2007)		completed MDT	assessment
Saunderson	Ethiopia	96 leprosy patients who had	NPSI
(Saunderson et al.,		completed MDT more than 10	
2008)		years	
Lasry-Levy (Lasry-	India	101 leprosy patients who had	DN4 and
Levy et al., 2011)		completed MDT	LANSS
Haroun (Haroun et al.,	Ethiopia	80 leprosy patients who had	DN4 and
2012)		completed MDT within 18 months	LANSS
Chen (Chen et al.,	China	275 leprosy patients	NPSI
2012)			
Felipe Reis (Reis et al.,	Brazil	33 leprosy patients with pain	DN4
2013)			
Raicher (Raicher et al.,	Brazil	90 leprosy patients with pain	DN4 and NPSI
2013)			
Gosling (Gosling et al.,	Brazil	114 leprosy patients with pain	DN4 and
2013)			McGill Pain

Table 5.1. Overview of screening tools in studies carried out on leprosy NP

5.2.1.3 Justification for using DN4 in identifying NP among leprosy patients

In the current study, leprosy patients have been assessed for NP using highly specialised assessments tools. Of these validated tools, the DN4 was selected, which may help in identifying NP characteristics.

First, DN4 is a simple, short and validated questionnaire to identify patients with NP. It uses both interview questions and brief bedside examinations, so it helps in assessing symptoms as well as clinical signs of NP. Whereas, other questionnaires are more complex, lengthy and have limited clinical examinations.

Secondly, DN4 has a good diagnostic yield. Previous studies evaluating diagnostic characteristics of DN4 for neuropathic pain of different aetiology have found values of sensitivity and specificity for neuropathic pain from 82% to 95% and from 78% to 97%, respectively (Bouhassira et al., 2005, Perez et al., 2007, Unal-Cevik et al., 2010).

Furthermore, in leprosy, DN4 is commonly used to identify patients with NP. It has a higher sensitivity as screening tool for NP in leprosy. In the 18th International Leprosy Congress, Raicher (Raicher et al., 2013), who investigated the prevalence of NP among 90 leprosy patients with pain using DN4 in Brazil, reported sensitivity and specificity of 96% and 58%, respectively. Similarly, in the 14th World Congress on Pain, Stump (Stump et al., 2012) reported high sensitivity of DN4 as a screening tool for NP in leprosy (>90%) in a study with 358 patients. In an study performed by our group in 2009; 80 leprosy patients, who had completed MDT within 18 months in Ethiopia, were assessed using DN4 and LANSS (Haroun et al., 2012). In this study we asked patients to evaluate the screening questionnaires. The study found that the DN4 was easier to administer than other tools in assessing NP. This finding was proved by patients' choices. Although the sensitivity of DN4 was found to be excellent (100%), its specificity was far lower than in the validation studies (45%). This could be because of the high numbers of patients with inflammatory pain that were recruited. This should not have affected the results concerning DN4 diagnostic accuracy; rather, it indicates that patients studied were similar to those usually encountered in regular clinical practice. Another study performed by our group in India, has come to a similar conclusion; DN4 is easier to apply in identifying NP among leprosy patients (Lasry-Levy et al., 2011).
5.2.2 The role of Quantitative Sensory Testing

5.2.2.1 Background

Quantitative Sensory Testing (QST) is a diagnostic method for accurately assessing somatosensory changes in human skin caused by nerve damage (Maier et al., 2010). It has been also recommended by NeuPSIG as a useful diagnostic instrument in the assessment of NP (Haanpaa et al., 2011a). It is a non-invasive procedure which is a sophisticated clinical examination of the sensory nervous system, whereby the perception thresholds to various mechanical and thermal sensory stimuli are accurately measured. Although it's utility in routine clinical NP assessment may have limitations such as difficulty in standardising and being time consuming (Cruccu et al., 2004), there is increasing interest in using QST to give insights into the underlying pathophysiological mechanisms of chronic pain.

5.2.2.2 Available instruments

DFNS-QST protocol

The German Research Network on Neuropathic pain (DFNS) has developed the DFNS-QST protocol in 2006 (Rolke et al., 2006a). It is a comprehensive and validated tests including all somatosensory modalities mediated by different nerve fibres (A β , A δ and C), that measures: cold and warm detection thresholds, number of paradoxical heat sensations during the thermal sensory limen procedure, cold and pain thresholds, mechanical detection threshold and mechanical pain sensitivity, dynamic mechanical allodynia, temporal pain summation and pressure pain threshold (Rolke et al., 2006b). This offers a high degree of detailed and precise information in the clinical diagnosis of leprosy related NP. The protocol is a well-established instrument for the assessment of NP and data has been collected internationally in over 3,000 neuropathic pain patients. In a study of 43 German patients who had neuropathy and dysesthesia in 2008, the QST parameters showed a high specificity (80%) but low sensitivity (37%) in the prediction of a reduced intraepidermal nerve fibre density as correlate for neuropathy (Scherens et al., 2009).

Quantitative sensory testing and leprosy

Although QST is widely used as an assessment tool for small fibre function and sensory profiles in neuropathies associated with pain (Maier et al., 2010, Rolke et al., 2006b), it

has not been used much leprosy patients and certain individual components have been measured (e.g. thermal and vibration thresholds) as opposed to the full battery of tests required to give the complete sensory assessment, Table 5.2.

Villarrole et al. in 2007 measured the different responses to the panel of stimuli used in a thermal testing analyser and monofilaments in 108 leprosy patients with skin lesions. They found that all patients had impaired warm and cold perception (Villarroel et al., 2007b). This study found that the cut-off points for warm and cold perception threshold determined from thermal sensory analysis were 35.1°C and 28.95°C, respectively (Villarroel et al., 2007a). Facer et al. 1998, who measured the responses to the thermal sensory analyser applied in the skin lesions in 28 leprosy patients, has also concluded that thermal threshold for cold and warm were significantly different (Facer et al., 1998). However, these studies have assessed the skin lesions in leprosy patients with no evidence of nerve function involvement and measured only thermal testing. A similar result, showing that warm and cold detection threshold were commonly affected, was obtained by van Brakel et al. 2008, who measured the different responses to the panel of QST thermal stimuli in a cohort of 303 Indian leprosy patients (van Brakel et al., 2008b).

Lund et al. 2007 measured the different responses to the panel of stimuli used in the QST in 17 leprosy patients with chronic pain in India and found 65% of the patients had sensory loss for all tested modalities (Lund et al., 2007). The study found that patients with chronic pain had lower IENF density and QST parameters (Lund et al., 2007), however this study has contained relatively small numbers of subjects.

Author, year, country	Main purpose of the study	Study population, N	Intervention / test	Main findings
Villarroel et al. (Villarroel et al., 2007b) Brazil	To assess leprosy skin lesion sensory impairment	Leprosy patients: 108	Thermal Sensory Analyser and Monofilaments	All patients had impaired warm and cold perception
Villarroel et al. (Villarroel et al., 2007a) Brazil	To determine the frequency of thermal sensation	Leprosy patients: 108	Thermal Sensory Analyser (TSA- 2001)	Warm perception threshold: 35.1 o C and CPT: 28.95 o C
Abbot et al. (Abbot et al., 1996) Iran	To assess un- myelinated fibre impairment	Leprosy patients: 39	Laser Doppler flowmetry, electo- pysiology and QST	Significant relationship between fibres and sensory impairment
Facer et al. (Facer et al., 2000) India	To explain early loss of cutaneous pain sensation	Leprosy patients: 28	Thermal test, light touch, Laser Dopler and skin biopsy	Sensory loss in affected skin
Brakel et al. (van Brakel et al., 2008a) India	To compare diagnostic test for neuropathy	Leprosy patients: 303 INFIR study	NCS, Quantitative thermal sensory test, MFs and VMT	WDT more frequently affected (29%), CDT (13%)
Lund et al. (Lund et al., 2007) India	To demonstrate the possible factors for NP	Leprosy pts completed treatment 17	QST and skin biopsy	65% had total sensory loss for all modalities

Table 5.2. Quantitative Sensory Testing studies in leprosy neuropathy

5.2.2.3 Justification for using DFNS-QST protocol in leprosy

This is the first study to document pain sensation using DFNS-QST in leprosy patients. These reasons for using the DFNS protocol include the standardisation, the highly precise detection of the sensory loss and gain, the validity of the DFNS-QST protocol, and its ability to assess the individual's sensory profile. This protocol is commonly used in other conditions such as peripheral nerve injury, postherpatic neuralgia and trigeminal neuralgia related neuropathic pain (Maier et al., 2010). The assessment of the exact sensory phenotype by QST is also a crucial part of this research project and this comprehensive technique is not yet used in leprosy.

The DFNS-QST protocol is a reliable and well-validated protocol. In recent years, several studies indicate a high diagnostic value of its results in both healthy subjects and in patients with NP. Maier (Maier et al., 2010) explored the spectrum of sensory abnormalities in 1236 patients with neuropathic pain due to different underlying diseases. In this large cohort of patients, DFNS-QST was found to be reliable; 92% of all patients with proved neuropathy had at least one sensory abnormality compared with the contralateral unaffected body area or with the reference data obtained from healthy controls. This indicates a good validity of the QST-DFNS protocol as a tool for quantifying somatosensory changes and even pain-related phenomena.

The DFNS-QST protocol is sufficiently sensitive to document the results related to the loss of function, i.e. sensory deficit and to abnormal pain-related phenomena, which clinically present as various manifestations of pain. This may help to assess the patient's sensory profile and subgrouping of the patients, and thus can be valuable to evaluate the underlying NP mechanisms in leprosy.

This protocol has been recommended by the IASP for the assessment of patients with NP, and leprosy patients will benefit from such tool (Haanpaa et al., 2011a, Backonja et al., 2013).

5.2.3 The role of skin biopsy and IENFD

5.2.3.1 Background

Skin biopsy provides insight into disease pathophysiology, which may lead to improve diagnosis and treatment of chronic pain condition. Its diagnostic yield has already been established in many peripheral neuropathies and especially useful when small fibre neuropathy is being considered (Lauria et al., 2009). For instance, skin biopsy is used to aid in the diagnosis of Parkinson's disease specifically to evaluate the potential role of alpha-synuclein as a biomarker for this disease (Nolano et al., 2008). Moreover, skin biopsies have demonstrated involvement of epidermal small sensory fibres in patients with Amyotrophic Lateral Sclerosis (ALS) (Weis et al., 2011). Furthermore, skin biopsies have demonstrated involvement of cutaneous innervation in patients with Spinobulbar Muscular Atrophy (Kennedy's disease) (Manganelli et al., 2007). These utilities help to evaluate and better understand of somatosensory dysfunction.

The principal role of skin biopsy in the diagnosis of NP is to determine intra-epidermal nerve fibre density (Lauria et al., 2010). Unmylinated C-fibre, which is the only fibre that penetrates into the epidermis, is frequently involved in patients with NP. This can be assessed by quantifying IENF density in the affected area (Lauria et al., 2010).

Several studies in clinical settings have examined the correlation between IENF density and pain (Devigili et al., 2008, Sorensen et al., 2006, Vlckova-Moravcova et al., 2008b, Quattrini et al., 2007, Polydefkis et al., 2002, Zhou et al., 2007). Sorensen (Sorensen et al., 2006) investigated 25 diabetic patients with NP and 13 patients without pain using skin biopsy obtained from distal leg. The authors found IENF density was lower in patients with NP compared to those without. In HIV-related sensory neuropathy, 101 patients underwent standardised NP assessment, IENF density was found inversely correlated with pain severity assessed with both VAS and the Gracely Pain Score (Zhou et al., 2007). Devigili and colleagues (Devigili et al., 2008) investigated 67 patients with pure small fibre neuropathy (diagnosed by the presence of at least two abnormal results on clinical examinations, QST, and skin biopsy) selected from a cohort of 124 patients with sensory neuropathy. The authors noted that Lower IENF density may be associated with the presence of NP, but it does not correlate with the intensity of pain.

5.2.3.2 Available instruments

Two main methods; punch biopsy and blister techniques are often used to obtain skin biopsy samples for assessing small fibre neuropathy. The former one is the most commonly performed using 3-mm disposable punch. It is standardised procedure that provides information on epidermal nerve fibres, sweat gland, hair follicles, and arterovenous anastomosis. The technique is validated, safe and minimally invasive (Lauria et al., 2010). Whereas, the latter is less invasive and has not been systematically used to

investigate patients with small fibre neuropathy. In addition, it does not provide information on dermal and sweat gland. In this study I use the 3mm disposable punch biopsy technique.

5.2.3.3 Intra-epidermal nerve fibre density (IENFD) and leprosy

In leprosy, skin biopsy is often used to define disease classification, but there have been only two studies on IENF density measurement and its correlation. The first study is performed by Facer in India, investigated 28 leprosy patients and found an inverse correlation between nerve fibres in the sub epidermis and thermal threshold (Facer et al., 1998). The second one is performed by Lund examined 17 leprosy patients with chronic pain (Lund et al., 2007). The authors found IENF density was significantly lower compared to the control skin biopsies.

Studies have revealed a significant correlation between IENFD, pain and QST parameters in post herpetic neuralgia, diabetics and HIV neuropathy (Zhou et al., 2007, Sorensen et al., 2006, Devigili et al., 2008, Loseth et al., 2008, Vlckova-Moravcova et al., 2008a). It is uncertain whether intra-epidermal nerve fibre density is correlated with pain and quantitative sensory testing parameters in leprosy, although studies confirm leprosy-related peripheral sensory neuropathy is a small fibre neuropathy. Previous study in leprosy has also suggest that a significant correlation between IENFD and QST parameters in patients with chronic pain (Lund et al., 2007), however this study has contained relatively small numbers of subjects.

5.2.3.4 Justification for using skin biopsy and IENF density in leprosy

In this current study, leprosy patients and controls have been assessed for NP using standardised clinical assessments and a skin biopsy taken from a pain affected area. A crucial part of the pain evaluation is the determination of the density of intra-epidermal nerve fibres in the affected areas.

The 3-mm punch biopsy with linear quantification of IENF density is a reliable and efficient technique to confirm the clinical diagnosis of small fibre neuropathy. The technique is ethically approved and widely used in other conditions. No side effects have been reported in published studies. Recently, it has been recommended by the European Federation of Neurological Societies/Peripheral Nerve Societies (EFNS/PNS).

5.2.4 The role of Psychological co-morbidity and HRQoL instruments

5.2.4.1 Background

NP is often associated with psychological conditions such as depression and anxiety, which may affects daily activities and overall quality of life (Meyer-Rosberg et al., 2001). This section describes the role of psychological co-morbidity and HRQoL questionnaires in the assessment of the impact of leprosy NP.

According to the World Health Organization health is defined as "a state of complete physical, mental, and social well-being, and not merely the absence of disease or infirmity" (WHO, 1948). Quality of life questionnaires are important measures of the general aspects of an individual's life, whereas HRQoL questionnaires more specifically measure the impact of disease on the patient's physical, psychological and social functioning.

NP is often associated with a reduced quality of life (Meyer-Rosberg et al., 2001, Haanpaa et al., 2011a). A systematic review of the association between NP and health related quality of life (Jensen et al., 2007) revealed strong evidence that the presence and severity of NP are associated with greater impairments in a number of important HRQoL domains. For example, pain intensity and pain interference with function can have a marked impact on daily functioning activities, affective and motivational states, social relationships, sleep hygiene and economic factors. This impact varies as a function of the HLQoL domain being considered and that different measures of HRQoL are differentially sensitive to the effect of NP. The principle role of HRQoL questionnaires in patients with NP is to provide information about the impact on quality of life, particularly when associated with chronic severe pain and suffering (Guyatt et al., 1993, Nelson and Berwick, 1989).

5.2.4.2 Available instruments

Depending on the outcome measures, the HRQoL instruments can be grouped into generic, condition-specific and preference-based measures (Vetter, 2007). The generic HRQoL instruments, such as SF-36 and WHOQOL questionnaires, are more general and comprehensive. These are often used for evaluating the impact of pain on the common elements of health, well-being and functionality. Whereas, the condition-specific instruments are more suitable for detecting changes due to disease progression

or remission. They are also used to detect treatment response (Patrick and Deyo, 1989). For instance, NePiQoL and Neuroqol are specifically designed for neuropathy and pain. Another important condition-specific instrument is the Brief Pain Inventory (BPI), which assesses the impact of NP on the patient (Coplan et al., 2004). BPI has been validated in patients with non-cancer pain (Keller et al., 2004). On the other hand, the preference-based instruments such as SF-6D and EQ-5D are designed to incorporate patients' opinions of the utility value of a particular health state rather than simply describe the condition. They are suitable for cost-effectiveness analysis and can be used for a comparison across diseases (Vetter, 2007).

5.2.4.3 Justification for using GHQ-12 and BPI questionnaires in leprosy

For this current study, I have looked at the HRQoL questionnaires and found that it is more general and not relevant to our patient group, category of measurement, primary purpose or setting. For instance, the WHOQOL social relationship questions are not relevant to our ultimate objective. Moreover, while WHOQOL includes overall pain as one domain, a condition-specific measure can evaluate in depth the impact of NP on quality of life. For example, the WHOQOL's specific pain focused questions, such as "Do you worry about your pain or discomfort?" may not detect a clinically significant change in leprosy related sensory abnormalities. Furthermore, even if pain is assessed as a separate dimension on WHOQOL, the effects of pain severity on health-related quality of life is not considered. Hence, I decided to use condition-specific tools for NP such as BPI, which are designed to assess specific diagnostic groups particularly with the aim of determination of the impact of NP.

Another important reason is the validity and availability of BPI. Although, no recommendations exist on the use of specific HRQoL questionnaires for the assessment of quality of life (Haanpaa et al., 2011a), BPI is preferred to be used in cases of severe neurological conditions or in short-lived NP conditions (Coplan et al., 2004, Zelman et al., 2005). In painful diabetes neuropathy and herpes zoster studies, the usefulness of BPI measures of functionality and quality of live have been demonstrated. In an early study performed by our group in 2009; 80 leprosy patients, who had completed MDT within the previous 18 months in Ethiopia, were assessed using BPI (Haroun et al., 2012). In this study the intensity of patient's pain on health related quality of life, such as physical functioning, sleep and mood were assessed. The short version of BPI and validated tools are freely available.

5.3 Aims and objectives of the thesis

5.3.1 Aims of study

The overall aim of the study is to characterise the somatosensory phenotype of leprosy patients with chronic neuropathic pain.

5.3.2 Statement of the hypothesis

I hypothesised that;

- The thermal and mechanical detection threshold, pain threshold (CPT and WPT), pressure pain threshold (PPT), and vibration detection threshold (VDT) are lower in leprosy patients with painful neuropathy than patients with non-painful neuropathy
- There is significant impact of neuropathic pain on quality of life and psychological well-being in leprosy patients with painful neuropathy compared to patients with non-painful neuropathy.

Specific objectives

- i. To measure the somatosensory responses of leprosy patients including thermal detection and pain thresholds, paradoxical heat sensations, mechanical detection thresholds to von Frey filaments, vibration detection threshold, mechanical pain thresholds to pinprick stimuli and blunt pressure, stimulus/response-functions for pinprick and dynamic mechanical allodynia, and pain summation (wind-up ratio) and to compare these measures between leprosy patients with painful neuropathy; non-painful neuropathy, leprosy patients without pain and age and sex matched healthy controls.
- To elucidate the impact of leprosy NP on quality of life and psychological wellbeing in patients with painful neuropathy compared with patients with nonpainful neuropathy.
- iii. To stratify leprosy patients by symptoms, sensory profile and psychological state

Chapter 6 Materials and Methods

6.1 Overview

This chapter presents the search strategy and definitions of terms used in the study, and an overview of the study design including the description of a case-control study as a research design and its applicability to this study. The methods of clinical and laboratory assessments, including the DFNS-QST protocol are then defined. Finally, an account of the data recording and management approaches are provided, including strategies of analysis that were used in this study.

6.2 The search strategy and search criteria

Literature published up to August 2014 related to leprosy and NP was searched. The literature review of leprosy and neuropathic pain was performed using search terms listed in Table 6.1. The search was limited to articles published in English. Various combinations of the terms were employed (leprosy and neuropathic pain). The main research resources used were search engines and bibliographic data-bases. These included: PubMed (chosen as it provides a wide coverage of health topics), EMBASE (chosen as it provides access to articles with a focus on general medicine), MEDLINE (chosen to complement the EMBASE search), and the Cochrane library (chosen to ensure capture of articles with clinical interventions). Reference lists in the articles found were also searched for relevant articles and the function "related articles" in PubMed was used. WHO documents on leprosy were checked on the WHO website. DFNS publications on QST were also checked on the DFNS website. Additional references were gathered from conference lists and Google internet searches. PhD theses available on EThoS (UK theses) and through the LSHTM, and ICL libraries were also checked for relevant information.

Leprosy	Reactions	Neuropathy	Pain	QST
Hansen's disease	type 1 reaction	subclinical	Pain	DFNS-QST
	"T1R"	neuropathy		
	reversal reaction	Nerve damage	Neuropathic	Quantitative
			pain	sensory testing
	erythema	peripheral nerve	non-neuropathic	DFNS
	nodosum	damage "PN"	pain	
	leprosum			
	"ENL"	nerve function	nociceptive pain	
		impairment		
		"NFI"		

Table 6.1 Search terms used in the study

6.3 Definitions of terms used in this thesis

6.3.1 Definitions

Motor test: test of the function of motor fibres of the nerves, normally performed by checking muscle power using voluntary muscle testing (Appendix 24).

Voluntary muscle test (VMT): test of the function of (voluntary) muscles innervated by a particular nerve trunk, normally performed by checking the ability of the patient to put a limb into a given position and to hold that position against resistance applied by the tester. For standardisation where possible a comparison is made to the unaffected side.

Motor impairment: motor neuropathy resulting in obvious weakness of the muscles innervated by a given nerve.

Sensory test: test of the function of sensory fibres of the nerves, normally performed by checking light touch sensation using application of monofilaments (Appendix 25).

Semmes-Weinstein monofilament test (MFs): a graded test of touch sensibility based on indenting the skin surface with a series of increasing thickness of standard nylon filaments. For each thickness it is recorded whether or not the patient feels the touch. Three or four sites per nerve may be tested, the severity of the sensation impairment being measured by the thickness of filament just felt by the patient. A normal level of 0.2g for the hand and 2g for the foot (excluding the heel) is appropriate.

Sensory impairment: neuropathy of the sensory fibres resulting in obvious reduction in the sensory ability of the patient using Semmes-Weinstein monofilaments.

Clinical neuropathy: sensory, motor or autonomic neuropathy evidenced by clinically detectable obvious reduction in function in sensory and/or motor using Semmes-Weinstein monofilaments and/or VMT.

6.3.2 Case definitions

1. Leprosy neuropathy

A leprosy neuropathy case was defined as a patient with a clinical evidence of sensory and/or motor impairment in an area innervated by one or more nerve using Semmes-Weinstein Monofilaments (MFs) and MRC scale.

Sensory impairment was defined by a decrease in sensation as measured by Semmes Weinstein monofilament testing. In the hands, this was defined as not being able to perceive the 0.2gm monofilament at 2 points out of 3 in each nerve of the hand. In the feet, this was defined as not being able to perceive the 2gm monofilament at 3 out of 4 sites of the foot.

Motor impairment was defined by a decrease in voluntary muscle testing (VMT) score, by 1 point or more from the normal score of 5, using the modified MRC scale.

2. Sub-clinical neuropathy

Patients with no clinical evidence of neuropathy based on Semmes-Weinstein Monofilaments (MFs) and/ or MRC scale, but who showed abnormal NCS or thermal testing were allotted to "Subclinical neuropathy".

3. No clinical evidence of neuropathy

For the purpose of this study, no clinical evidence of neuropathy was clinically defined by the normal sensory and motor impairment result using Semmes-Weinstein Monofilaments (MFs) and MRC scale, respectively. Sensory neuropathy detectable by Semmes-Weinstein monofilaments, but not meeting the criteria of sensory impairment is clinically defined as no clinical evidence of neuropathy.

4. Neuropathic pain

A case of NP was clinically defined by the presence of negative or positive neurological signs concordant with the distribution of pain at the affected peripheral nerves based on a score of $\geq 4/10$ using the DN4 questionnaire and clinical neurological examination. Duration of this pain for three months or more is defined as chronic NP.

5. Non-neuropathic pain (nociceptive pain)

For the purpose of the curent study, a case of non-neuropathic pain was clinically defined by the occurrence of sharp, dull, or aching pain score of < 4/10 using the DN4 questionnaire (Costigan et al., 2009). Pain associated with neuritis reaction, which defined by the development of inflammation of nerve sheath without abnormal findings in sensory testing, is clinically defined as nociceptive pain (being originated from nervi nervorum) (Bove and Light, 1997, Sauer et al., 1999). Also, pain associated with type 1 and type 2 reactions is clinically defined as non-neuropathic pain.

6. Type 1 reaction

A type 1 reaction was diagnosed when the patients had erythema and oedema of skin lesions. There may be accompanying neuritis and oedema of the hands, feet and face. The skin signs were obligatory; the nerve and general signs optional (Van Brakel et al., 2005a).

7. Type 2 reaction

A type 2 reaction was diagnosed when the patients had crops of tender subcutaneous skin lesions. There may be accompanying neuritis, iritis, arthritis, orchitis, dactylitis, lymphadenopathy, oedema and fever. The skin signs were obligatory; the nerve and general signs optional (Van Brakel et al., 2005a).

8. Neuritis

A case of neuritis is clinically defined if the patient has any of the following: Spontaneous nerve pain, paraesthesia or tenderness; new sensory, motor or autonomic impairment of recent onset; or mixed signed.

6.3.3 Pain terminology

The pain terminology, which was based on the updated IASP taxonomy (IASP, 1994), is given in (Appendix 50).

6.4 Study design

6.4.1 An overview of the study design and research procedures

A range of study designs could be used for this type of analysis. In cross-sectional studies, which is a "snapshot" of the population at a single point in time, the exposure and disease status are assessed simultaneously in each individual. It is good for measuring the scale of a problem e.g. prevalence of disease, but it cannot assess the sequence of events, so cannot show that exposure came before the outcome. In case control studies, people who have the disease in question (cases) and those who do not (controls) are compared with respect to the past exposure of potential causative factors. Sometimes case control studies referred to as retrospective studies, because they look backwards from the disease to potential causes. In cohort studies, a group of people with a particular risk factor is followed to determine whether they develop the disease of interest.

I adopted a case control study to assess the somatosensory parameters related to NP in leprosy patients using the QST in 90 leprosy patients. A defining feature of a casecontrol study is that the starting point is identification of people with the outcome in question. In relation to this study, I chose to identify NP cases among leprosy patients. The case definition of NP was clinically defined, and I included all prevalent cases of NP (i.e. all NP cases within leprosy population in the BLP catchment area at a specific point in time). Our controls were drawn from the same population and were classified into different groups by their neuropathy status and the presence of pain (Table 6.2). Three control groups were randomly selected from the leprosy population presenting at the BLP and FMR clinics. One group of controls was selected from patients with no pain and no clinical evidence of neuropathy. The other two groups were patients with pain-free neuropathy and patients with pain and no clinical evidence of neuropathy. In addition, a group of healthy volunteers was recruited from the local Indian population. Although, this group was not part of the case control series, their data were used for comparison with patients along with the DFNS database reference. The reason I selected more than one group was due to the heterogeneity of leprosy neuropathy and pain. The aim of the study was to investigate the clinical aspect of NP in leprosy. If the study had used only healthy controls, the comparison would have been between patients who had developed NP and healthy, which might may have introduced inaccuracies in the profiling if the patients with leprosy who had pain reported similar profiles to healthy controls. In addition, we felt that profiling leprosy without pain might detect new neurological abnormalities that could be tested by comparing cases with the control group who had neuropathy and pain. Individual matching methods were applied in this study. For each individual case, a control was selected who was similar to that case with respect to the age and gender. Cases and controls were individually matched by age (within approximately 10 years) and sex. For example, if our case was a 34-year-old female, I selected a control who was a female aged 30 - 39.

		Neuropathy	
		Yes	No
Pain	Yes	? NP	? Non-NP
	No	Pain free neuropathy	No evidence of neuropathy

Table 6.2. Design of study groups selection

In relation to this study, the selection of controls was appropriate. The use of these methods helped the study as follows: firstly, the population-based controls minimised the systematic selection bias. Secondly, having three controls per case improved the statistical precision of the profiling estimate. However, increasing the number of controls was logistically difficult because of additional resources and time required to interview a greater number of controls. Finally, the advantages of the age and gendermatched control techniques used for each patient in this study reduced potiential sources of confounding and increased the efficiency of the study. Matching our controls to cases by age ensured that the age distribution of cases and controls was the same. This meant that the effect of the matched age was removed from analysis (i.e. I controlled for any confounding effect of age). Another advantage of matching was examining fewer patients per group, because the cases and controls were more similar to each other than they would be if they were not matched. I tried to match the recruitment across the leprosy spectrum (TT, BT, BB, BL and LL), but it became difficult to find appropriately matched controls for each case, because the R.J classification was poorly documented. However, the matching method is prone to bias by making cases and controls more similar than they would otherwise be. Therefore, it is not possible to examine the effect of an exposure that has been matched.

In this study, I chose to characterise the somatosensory profile of leprosy patients with NP, and compared the profile among those with leprosy neuropathy and those without. However, the study did not consider the association between the leprosy neuropathy and NP, or the effect of leprosy or exposure to other variables, for these would fall outside the timeframe and objectives of the study. The study would only look at profiling characteristic of leprosy patients with NP at one point in time.

The benefit of a case-control study design is that it can be carried out rapidly and relatively cheaply compared to cohort studies. It is also useful for studying rare diseases such as leprosy that may take a long time to manifest. However, case-control studies are prone to selection bias, particularly in the selection of controls. The design is also prone to information bias, because exposure status is determined after the outcome has occurred. In addition, case-control studies may not provide information about the sequence of events leading up to an exposure or outcome. These studies are also not suitable for estimating disease incidence or prevalence. Therefore, in this this study, I was unable to estimate the occurrence of NP among leprosy patients. I was also unable to establish the causality of NP, and whether leprosy is a consequence or a cause. Information on cause-and-effect relationships can be collected by applying a longitudinal study design.

A cohort study, similar to a case-control study, is observational. Two types of cohort study are known: descriptive and analytic. In descriptive cohort studies, a group of participants who have experienced an exposure of interest are selected and followed over a period of time to determine the incidence of one or more outcomes. In analytical cohort studies, the association between an exposure and an outcome are tested. Study participants are classified as exposed or unexposed to the risk factor of interest.

Cohort studies are particularly useful for rare exposures and in situations where more than one outcome is of interest. In cohort studies, the exposure is measured at the start of the study before the outcome occurs, and so measurement of the exposure is not biased by the presence or absence of the outcome. These studies can provide data on the time course of the development of the outcome(s), including late effects. Rare exposures can be investigated using appropriately selected populations.

Our study requires the description of the distribution of leprosy NP related to time, place, and person. Since the onset of leprosy is usually gradual, if we want to describe the onset and characteristics of NP, a cohort study would be the best study design. Leprosy is a rare disease, so such a study would require a large study population. In nested case-control studies, both cases and controls are derived from the same population, with controls being representative of a sample of non-cases. The benefit of such study over a case-control study one is that the risk of selection bias is reduced. It would thus seem that a cohort study would be the design of choice as it provides a wider scope, but due to its nature, some disadvantages are unavoidable. Prospective cohort studies are slow and potentially expensive if there is a long period between the exposure and the outcome. They are inefficient for rare diseases. Retrospective cohort studies depend on pre-existing records of exposure being available and being reliable. The exposure status may change during the study in which case exposure status may need to be determined again at intervals throughout the study. Furthermore, differential loss to follow-up may introduce bias, which is a particular problem when follow-up is of a long duration. In long-term cohort studies, it may be hard to ensure that diagnostic criteria remain consistent throughout the study, particularly if outcomes are ascertained from routine data sources. In nested case-control studies, a cohort study needs to be done first. The rarity of disease and long latency are not contra-indications for cohort studies, although their disadvantages need to be balanced against the superior quality of evidence cohort studies deliver compared with other study designs, such as case-control studies.

For this study, a single time case-control study was feasible in the time frame available. This design is quick by nature and it also solves some of the problems associated with the cohort study designs, and most importantly, it can provide information of interest as discussed above.

6.4.2 Study setting

6.4.2.1 Study sites

India

Each year India registers about 130 000 new leprosy patents so doing this study in India is important and will increase the impact of the study (Figure 6.1). For the current study, we collaborated with two Indian centres, The Foundation for Medical Research Indi (FMR) and the Bombay Leprosy Project (BLP), where our leprosy group have successfully recruited to other leprosy studies, 101 patients were recruited for a study on NP. In addition, an ongoing study (TENLEP) associated with Professor Lockwood is also based on these two sites. FMR was chosen for the laboratory work (QST testing), while BLP was acted as sources of patients into our study.



Figure 6.1. Study site - Mumbai

Foundation for Medical Research (FMR)

The Foundation for Medical Research was established in 1975 as a Public Trust and is recognised as a Scientific and Industrial Research by the Government of India. It is a continuation of the founder H. Antia interest in leprosy. Dr Antia is a pioneer plastic and reconstructive surgeon for leprosy and this continuation of his initial interest of leprosy is maintained and extended to involve research on drug-resistant TB and medical plants. The foundation has a well-established laboratory facilities (Figure 2.1).

The leprosy research at the Foundation for Medical Research focuses on neuropathy and leprosy treatment. Dr Shetty, who leading the leprosy group, has a pioneering and sustained research work on mechanisms of nerve damage which has implications for treatment and regeneration. The leprosy group has also been actively involved in assessment of drug regimens for treatment of leprosy and prevention of nerve damage. FMR is also a recognised centre for epidemiological study and its estimate of the current load of leprosy in rural and urban areas challenged the claim of elimination and provided the basis for further course of action by the national control programme.



Figure 6.2. Foundation for Medical Research – Mumbai

Bombay Leprosy Project (BLP)

The Bombay Leprosy Project, founded in 1976, is the largest referral centre for people affected with leprosy and other skin disease in Mumbai. It was established by Dr Ganapati with the objective of improving the quality of life of leprosy patients. The project covers 23 health posts in Mumbai covering a population of approximately 2

million. All consultations at BLP outpatient services and rehabilitation activities are provided free of charge for people affected with leprosy. The services are divided into satellite leprosy daily outpatient's clinic, general clinics for people affected with leprosy and their families, inpatient services in collaboration with Somayia Hospital and clinics for people with other skin diseases. The BLP in particular serves a wide population and receives referrals from all over the city as well as the Maharasha region and the whole country. Also, it has an intensive data on slum record in the city.

An average of 2 to 3 new leprosy cases were diagnosed weekly at BLP over the period from January to June 2013. In addition, an average of 30 to 35 patients were seen daily at the referral centre during the same period and these were mainly leprosy cases with complications, many of whom were referred from regions.

Over the last three decades of leprosy work BLP has reached 1.95 million people of which 60% are from slums of Dharavi and other similar slums in G and H wards of Mumbai (BLP annual report, 2013). 30,000 patients have been cured with MDT. Disabilities have been prevented in 2500 patients. 300 leprosy patients and general handicapped persons have been rehabilitated. In addition to the activity related to diagnosis and treatment of patients, the BLP has been carrying out operational and technical research in the field of leprosy and has published over 300 scientific papers in India and International journals. The efforts of BLP have been highly recognised for its excellence in leprosy research.



Figure 6.3. Bombay Leprosy Project – Mumbai

6.4.2.2 Study team

The study teams from the BLP and FMR clinics received regular training and updates throughout the study. The clinical psychologist had to undergo piloting study questionnaire exercises, and the research assistant had onsite training on diagnosing leprosy and NP. This step aimed to familiarise teams with the study materials and to ensure that they understood the procedures. This was done at the beginning of the study. A joint meeting with the director of the BLP (Dr Pai) and Dr Shetty, senior researcher at FMR (Figure 6.3), was held on a regular basis (every two weeks for the first three months, then monthly till the end of the study). This helped to recruit enough patients and to harmonise the work between the two centres.

6.4.2.3 Study population

Participants and recruitment

The following group of participants were recruited for this thesis:

- i. Leprosy patients without pain and no clinical evidence of neuropathy
- ii. Leprosy patients without pain and sub-clinical neuropathy
- iii. Leprosy patients without pain and established clinical evidence of neuropathy
- iv. Leprosy patients with established pain and clinical evidence of neuropathy
- v. Healthy control participants, aged matched to the patients group in India
- vi. Healthy control participants for QST investigator validation in UK
- vii. Healthy control participants for monofilament comparison study in Germany.

Recruitment

Patient recruitment

Patient cohort were recruited from leprosy affected people attending two main centres: the BLP and FMR clinics in Mumbai during the period October 12th 2012 to June 30th 2013 (Appendix 11 and Appendix 12). I attempted to recruit patients across the leprosy spectrum (TT, BT, BB, BL and LL) in each cohort where applicable, but as most of participants were treated patients at the time of the recruitment it was very difficult to retrospectively identify the type of leprosy. All patients had to fulfil the diagnostic criteria for leprosy, which includes hypo-pigmented lesions with definite loss of sensation, thickened peripheral nerves, and acid fast-bacilli on skin smears (Britton and Lockwood, 2004). All potential participants underwent an initial screening (Appendix 23). Prior to inclusion in the study, patients underwent a comprehensive assessment in order to further determine if they met the eligibility criteria for recruitment. A study team member interviewed each participant using a pre-tested questionnaire. The assessment of each patient took one hour on average. The clinical assessment was comprised of patient's history and pain drawings including the location, description, and intensity of pain. In addition, documentation of clinical evidence of neuropathy were collected and related neurological bedside examinations of sensory and motor function were conducted. Information from medical records and data relevant to peripheral neuropathy and results from any other medical investigations that were available were reviewed. All participants were asked to provide a urine sample for sugar testing and a blood sample for complete blood counts, blood glucose, thyroid function, vitamin B_{12} level, syphilis, HIV and pregnancy (for women).

The patient's records, including the findings of the clinical examination and the available investigations, were reviewed by the investigator. Individuals who were anaemic, diabetic, had hyperthyroidism or hypothyroidism, were B_{12} deficient, had positive serological test for syphilis and/or HIV or women who were pregnant were not eligible to enter the study. These patients were able to access routine services and were also referred to the nearest facility for special service according to their condition. Those who had satisfied the eligibility criteria were invited to take part in the study.

In particular, patients with symptoms and signs suggestive of NP were recruited through a two stage process (Figure 6.4). All patients were screened for neuropathy using MFs and the MRC scale after they signed the informed written consent sheet. Those who had neuropathy were further screened for pain. Of them, patients with pain were assessed for NP using the DN4 questionnaire. Patients who proved to have the two criteria for the case definition were considered as leprosy NP.



Figure 6.4. Assessment of neuropathy and NP case-definition

PN: Peripheral Neuropathy

Subclinical Neuropathy: Patients with no clinical evidence of neuropathy based on Semmes-Weinstein Monofilaments (MFs) and/ or MRC scale, but who showed abnormal NCS or thermal testing were allotted to "Subclinical neuropathy".

All participants had the following assessments:

- Demographic description and medical and drug history
- Completion of pain questionnaires
- Clinical assessment
- Quantitative sensory testing

Healthy volunteer recruitment

Healthy volunteer participants were recruited from the general population by personal invitation and word of mouth. For the study in India, participants were selected from the patient's relatives attending the leprosy clinic based at BLP and staffs from BLP and FMR. Potential participants were approached and invited to participate in the study (Appendix 5 and Appendix 6). Those who volunteered in response to the invitation were given brief feedback asking them to contact the research team should they wish to discuss the study or be sent further information. Potential participants were given a participant information sheet and then asked to sign a consent form. Eligible participants received a questionnaire about their general health status, the "Participant's State of Health Questionnaire" used in the DFNS guidelines. All participants were asked to provide a urine sample for sugar testing and a blood sample for complete blood counts, blood glucose, thyroid function, vitamin B₁₂ level, syphilis, HIV and pregnancy

(for women). Then they underwent a comprehensive clinical examination using DFNS QST measurement parameters.

More details on the recruitment of healthy volunteers in Germany and UK are presented in Chapter 7.

Participants

Patients with leprosy

Eligibility

The study participant had to be a confirmed leprosy case irrespective of whether they were receiving, or had received multi-drug therapy or have symptoms of a peripheral neuropathy. Study specific entry criteria are described below:

Inclusion Criteria

- Able to give informed consent
- Age 18 to 65 years
- Able to comply with the protocol

Exclusion criteria

- A history of concomitant severe infection such as TB or any other serious underlying disease (cardiac, renal, or hepatic) that potentially might affect the evaluation of the patient's pain response.
- A history of other conditions associated with peripheral neuropathy such as Diabetes Mellitus (DM), HIV/AIDS, or nutritional deficiency (Thiamine, B₁₂ deficiency)
- Other neurological or psychiatric disease
- A history of regular, excessive intake of alcohol (alcoholism)
- Evidence of Thalidomide treatment
- Evidence of pregnancy or lactating mother
- Insufficient level of communication (i.e., lack of fluency in any of the three languages of the study: English, Hindi, or Marathi).

Healthy controls

Inclusion Criteria

- Signed the written consent form themselves
- Age 18 to 65 years

Exclusion Criteria

- A history of pain and/or paraesthesia and/or nerve lesion in the extremities
- Evidence of pain treatment scheduled for the time of the study and intake of medication during the time of the study (antidepressant, analgesics, or hypnotics)
- A history of surgery that potentially might affect the sensations in the tested site
- Evidence of medical disease
- Other neurological or psychiatric disease
- Insufficient level of communication as described above

6.4.3 Sample sizes

A sample size of 15 participants in each patient group was estimated to be sufficient to adequately assess the sensory changes in leprosy patients with NP. The study was powered against sensory modalities data required to detect the prevalence of sensory changes associated with the risk of developing HIV neuropathy from the pain in the HIV-related neuropathy study (Phillips et al., 2014). The HIV study is one of a few studies using quantitative sensory testing to assess sensory parameters in NP caused by infectious diseases. I had to choose one of the 13 parameters of the DFNS profile on which to conduct our power calculation; since loss of mechanical sensation is one of the primary manifestations of leprosy we determined that this was the most useful parameter. In the previous HIV study, the mechanical detection threshold (MDT) showed a significant difference between the z-scores for QST parameters in multi-ethnic population (MDT mean Z-score difference was 1.4 with a standard deviation of 1.0). The sample size was calculated based on this result, using a standard deviation of 1.0, power of β =90% and significance of α =0.05%.

6.4.4 Ethics and consent practice

The study protocols and recruitment procedures were approved by the following ethics committees in London and Mumbai:

- 1. London School of Hygiene and Tropical Medicine Research Ethics Committee, ethical approval reference number: 6181 (Appendix 1).
- Imperial College Research Ethics Committee (ICREC), ethical approval reference number: ICREC_11_2_3 (Appendix 2).
- Foundation for Medical Research Ethics Committee, IEC No _ FMR/ IEC/ LEP/ 04/ 2012 (Appendix 3).

I undertook two courses in good clinical practice (GCP), one at the LSHTM prior to starting the study, followed by a refresher course (online) organised by Imperial College London. All studies were conducted in adherence to the recommendations for physicians involved in research on human subjects adopted by the 18th World Medical Assembly, Helsinki 1964 and later revisions.

Subjects participated in the study after giving written informed consent. The process was conducted by a native Hindi/Marathi speaker after he had fully explained the study and answered any questions. The study information leaflets (Appendix 13, Appendix 14, and Appendix 15) and consent forms (Appendix 20, Appendix 21, and Appendix 22) were available in Hindi, Marathi and in English. Participants were informed that they had the right to refuse to participate without given reasons. They were also assured that they could withdraw at any time from the protocol without giving reasons and without prejudicing further treatment. The consent forms were signed by all participants prior to inclusion in each study (if they were unable to sign, a thumb print was used instead and witnessed by the person obtaining the consent).

6.5 Clinical and laboratory assessment

6.5.1 Patient interviews and examinations

6.5.1.1 Clinical history

The patient's demographic information was collected for age, sex, time since leprosy symptoms first developed, the clinical R-J classification of their disease, treatment with MDT, previous reactions, and past medical history (Appendix 23). A detailed history of their skin and nerve symptoms was taken. The number and morphology of skin lesions, the presence of peripheral oedema, nerve tenderness, and paraesthesia or nerve pain was recorded. The location of pain was recorded by using a template body map (Appendix 26); the patient marked the distribution of any type of pain he had. Patients were asked about their type of pain, its duration, and the treatment used for it.

6.5.1.2 Measures

Translation

Using standard translation and back translation methods, the English version of all questionnaires including the DFNS-QST protocol were translated into local languages (Hindi/Marathi) by an independent translation centre in Mumbai (Appendix 52). The translated questionnaires were then reviewed by a panel of experts, using the repeated 'forward-backward' procedure. The experts were from BLP, FMR, and the Neurology department at JJ Hospital in Mumbai. They are fluent in both English and Hindi/Marathi as well as the terminology used in the leprosy and pain field. The translation procedure review was conducted several times until an agreement was reached for the final version. The translation procedure is further detailed in Chapter 7.

Patient groups

Patients who were positive on the screening test for pain then completed the DN4, PD-Q, and BPI questionnaires.

Pain intensity

A supervised assessment of patient pain intensity was adopted due to variable barriers of using a pain diary. The patient's weekly average pain intensity was determined using an intensity numeric rating scale which consists of an 11- point numerical scale ranging from "0" (no pain) to "10" (worst possible pain) (Farrar et al., 2001). Patients were requested to describe the weekly average intensity of pain by choosing the appropriate number between 0 and 10 in response to the question, "tell me what number best represents the greatest pain you have had in the last week" (Appendix 23).

The intensity of the various pain components was documented on verbal rating scale (mild, moderate, and severe) as part of PRF.

The worst and average pain intensity over the week preceding the testing session and pain intensity at the time of questionnaire completion were documented on a numeric rating scale as part of PD-Q (0 = no pain, 10 = maximum pain).

Screening questionnaire for NP

Two NP screening tools were used in this thesis to identify the likely presence of suggestive symptoms and signs of NP: the DN4 and the PD-Q. The former questionnaire was used for the case definition in this study. The local version of the DN4 questionnaire (Appendix 29 and Appendix 30) was obtained from our leprosy group study in Mumbai (Lasry-Levy et al., 2011). The PD-Q that was validated in Hindi and Marathi was obtained with permission from Pfizer Medical information, India (Appendix 32 and Appendix 33).

DN4 Questionnaire

The DN4 was applied in an interview format and contained seven sensory descriptor items and three clinical examination items. The latter test for allodynia using cotton gauze and altered sensation (hypoaesthesia) to touch or a pinprick in the painful area. The Hindi/Marathi version of DN4 questionnaire was applied by the local team member in the presence of the principle investigator. The first seven items were answered and the clinical assessment was done by the principle investigator. The scoring of the items ranged from 0 to 1. A score of 1 is given to each positive item and a score of 0 to each negative item. The total score was calculated as the sum of all 10 items, and the cut-off value for the diagnosis of NP was a total score of 4/10 (Bouhassira et al., 2005).

Pain detect questionnaire (PD-Q)

The PD-Q is comprised of nine questions regarding the severity, course, quality and nature of the patient's pain and the specific NP symptoms (Appendix 32 and Appendix 33). The scoring of the sensory descriptors ranged from "0" (indicating that the person does not experience the relevant sensation) to "5" (indicating that the person feels the sensation very strongly). Based on PD-Q results, patients were grouped as follows: unlikely NP (a score of 0-12, which indicates a negative result and a NP component is unlikely); probably NP (a score of 13-18, which indicates an unclear or ambiguous result

that does not preclude a NP component); and definitely NP (a score of \geq 19, which indicates a positive result and NP is likely) (Freynhagen et al., 2006b).

Brief pain inventory (BPI)

Patients were asked to respond to the Hindi/Marathi version of the BPI (Appendix 35 and Appendix 36), rating their current pain intensity and also pain in the last 24 hours at its worst, least, and average by using a numeric 11-point scale ranging from 0 "no pain" to 10 "pain as bad as you can imagine".

Pain-related interference of activities of daily living was assessed using the pain interference scale of the BPI (Cleeland and Ryan, 1994). The scale assesses pain interference within seven domains: general activity, walking, work, relationships, mood, life enjoyment, and sleep. Patients were asked to rate the extent to which their pain interfered with these seven quality of life domains on an 11-point scale ranging from 0 "does not interfere" to 10 "completely interferes".

All patients

Psychological co-morbidity measures

Psychological factors were assessed with the GHQ-12 (Appendix 37), which is an established instrument used to screen for the presence of mental distress (Goldberg, 1972). In particular, the GHQ-12 asks questions about anxiety and depressive symptoms over the past four weeks (Goldberg and Williams, 1988). The GHQ-12 version, which was validated in Hindi (Gautam et al., 1987), was obtained with permission from Professor Shiv Gautam, former president of the Indian Psychiatric Society (Appendix 38 and Appendix 39). Before asking the GHQ-12 questions, patients were asked to refer to how they had felt during the past four weeks. If they had any unhappy feelings, they were asked what causes contributed to them and how much was due to their pain, leprosy itself, or general life. These responses were all recorded in the PRF. The GHQ-12 questions were administered by a native Hindi/Marathi speaker. The interview lasted 5 - 10 minutes. The patients' responses were scored on a four point scale "not at all", "same as usual", "more than usual" and "much more than usual" giving a score from 0 "no problem" to 12 "severe problem". Higher scores indicated greater psychological distress. For the coding and interpretation of the answers see data analysis section below.

6.5.1.3 Structured clinical examination

A comprehensive structured clinical examination was developed to detect clinical signs of peripheral neuropathy. The clinical examination was performed on each patient and included assessment of neurological examination with special attention to loss of sensation over skin lesions and nerves and disability (Appendix 23).

Neurological clinical assessment

The neurological examination was performed on each patient and included assessment of light touch using cotton gauze and a pinprick sensation using a disposable safety needle. Thermal and vibration perception were assessed as part of QST testing. Joint position sense of the index finger and big toe were assessed and graded as normal or absent. Deep-tendon reflexes (knee, ankle) were also assessed and graded as normal, decreased (if present with reinforcement), or absent. Muscle wasting and motor power were assessed using a modified MRC scale (described below). An abnormal result was taken as two or more symmetrical signs in the hands or feet consistent with peripheral neuropathy.

Skin assessment

The location and appearance of skin lesions and whether they were overlying the course of a peripheral nerve trunk or pain site was recorded on a body map (Appendix 26).

Nerve assessment

Nerve function impairment present for more than six months was recorded. The nerve involved and the functional modality affected (sensory or motor) was also documented.

Nerve thickening and tenderness

The main peripheral nerves, namely the greater auricular, ulnar, median, radial cutaneous, lateral popliteal, and posterior tibial nerve were assessed for enlargement and tenderness. A palpable nerve was assessed clinically by pressing the nerve against bone. The results of nerve palpation, along with the presence of a skin lesion and/or a positive skin smear were used to confirm the diagnosis of leprosy, but it was not considered in the diagnostic criteria for neuropathy in this study.

Nerve function assessments

Sensory nerve function was assessed using a standard set of coloured Semmes– Weinstein monofilaments. The monofilaments were designed to apply a graded test of touch sensibility based on indenting the skin surface with a series of increasing thickness of standard nylon filaments (0.2g, 2g, 4g, 10g and 300g). For each thickness it was recorded whether or not the patient felt the touch. Three sites per nerve in the upper limb and four in the lower limb were tested. The test sites in the upper and lower limbs are shown in Appendix 25. The severity of the sensation impairment was measured by the thickness of the filament felt by the patient. A result of monofilament of 0.2gm for the hand and 2gm for the foot was taken as a normal reference (van Brakel et al., 2005b). A patient was diagnosed as having sensory impairment when the monofilament threshold was increased from the normal reference. In the hands, this was defined as not being able to perceive the 0.2gm monofilament at two points out of three in each nerve of the hand. In the feet, this was defined as not being able to perceive the 2gm monofilament at three out of four sites of the foot.

Motor nerve functions were assessed by voluntary muscle testing (VMT) using the 0-5 modified Medical Research Council (MRC) scale (Appendix 24). An abnormal result in the hands or feet was taken as a decrease in VMT score by one or more points from the normal score of five using the modified MRC scale.

Disability assessment

Leprosy-related disability was assessed using the WHO disability criteria, which defines grade 0 as no loss of sensation or visible deformity, grade 1 as loss of sensation without visible deformity, and grade 2 as presence of visible deformity (WHO, 2006).

The patients' record forms, including the clinical examination, pain symptomatology, quality of life, and psychological co-morbidity measures were administered before the DFNS-QST testing was performed. Of these questionnaires, the sensory testing assessment of the DN4 questionnaire was performed at the area of maximum pain of the ulnar nerve territory (Dermatome C8) just prior to administration of the DFNS-QST testing protocol.

6.6 Quantitative Sensory Testing

6.6.1 Testing sites

Quantitative sensory testing (QST) was performed for all subjects in the ulnar side of the dorsum of the hand bilaterally (supply area of the ulnar nerve dermatome C8) or S1 dermatomes of the foot bilaterally. These sites were defined and documented on each patient's body map during the screening assessment based on the maximum pain in the most affected area as determined by the patients. In patients with bilateral pain, the most painful area was chosen as the test site, and the contralateral mirror site as a control. However, during the course of patient recruitment it became apparent that many patients experienced their maximum pain along the ulnar nerve territories proximal to the dermatome C8 or in other body regions. The pressure pain threshold (PPT) for dermatome C8 of the hand was assessed over the hypothenar eminence muscle. The PPT for the S1 of the foot was assessed over the medial aspect of the plantar. Vibration detection threshold (VDT) of the hand and foot was recorded over the bony prominence of the ulnar (ulnar styloid process) and medial malleolus, respectively (Rolke et al., 2006a, Rolke et al., 2006b). The dorsum of the hand (dermatome C8) was the main site for the QST measurements in this study (Figure 6.5). This site was selected because the ulnar nerve is the most common nerve affected in leprosy, and also to obtain a consistent QST measurement. The lower limb site was selected for patients with foot pain. The reason for recruiting four patients with foot pain was to assess the IENFD.



Figure 6.5. QST Testing sites

6.6.2 Testing protocol

QST measures were taken according to the standardised DFNS-QST protocol of the German Network on Neuropathic Pain, using the same equipment and standardised instructions. Applications of the procedure can be seen in the published guidelines by Rolke (Rolke et al., 2006a, Rolke et al., 2006b), and an updated version of the protocol in Hindi and Marathi can be seen in Appendix 41 and Appendix 42. The test was conducted at the FMR laboratory which was quiet, spacious and had a constant room temperature. All measures were performed on each subject by the same investigator, who was trained and qualified in using the DFNS-QST assessments and instructing subjects. Standardised verbal instructions were given to all subject by a native Hindi/Marathi speaker. The subject's positioning remained unchanged for each body site to be tested. All tests were first performed over a demonstration site that was not later tested during the session. This step aimed to familiarise subjects with the tests and to ensure that they understood the procedures. For all patients, testing was performed in a standardised order (i.e. control side was tested prior to affected side) (Rolke et al., 2006a, Rolke et al., 2006b), and for those who were bilaterally affected, the maximum pain site was selected as the test site. For healthy subjects, the control and test sites were determined randomly from the list. The time needed to complete testing of the full protocol per test site was approximately 30 minutes, with the total examination time taking 2-2.5 hours. All QST measures were recorded on a specific data collection sheet provided by the DFNS.

The DFNS-QST protocol comprises a set of psychophysical tests that assess the functional status of specific somatic sensory modalities in the following standardised order:

- Thermal detection and pain thresholds
 - Detection threshold for cold (CDT)
 - Detection threshold for warm (WDT)
 - Difference threshold for cold and heat (TSL)
 - Cold pain threshold (CPT)
 - Heat pain threshold (HPT)
 - Paradoxical heat sensations (PHS)
- Mechanical detection threshold (MDT)

- Mechanical pain threshold (MPT)
 - o Detection threshold for pressure
 - o Pressure pain threshold
- Stimulus-response functions
 - Mechanical pain sensitivity (MPS)
 - Dynamic Mechanical Allodynia (DMA)
- Wind-up ratio (WUR)
- Vibration detection threshold (VDT)
- Pressure pain threshold (PPT)

A description of the QST procedures and the needed equipment for each test follows:

Determination of thermal detection and thermal pain thresholds

Methods and background: The thermal QST tests were the first modality assessed and consisted of six different parameters. First, thermal detection thresholds for the perception of cold (CDT) and warm (WDT) were measured using the thermal machine, followed by thermal sensory limen (TSL: the different threshold for alternating cool and warm stimuli), to PHS. Then, the thermal testing battery was conducted by measuring cold and heat pain thresholds (CPT; HPT). The method of limits was used (Chong and Cros, 2004). For each threshold test, three repetitions were performed at each site. All thresholds were determined by continuous ramping of temperature by $1^{\circ}C/s$, which were immediately stopped when the subject pressed a button. The subject was instructed to press the stop-button as soon as the slightest change of temperature (for detection threshold) or the first painful sensation (for pain threshold) was perceived (Appendix 40). The base line temperature to which the thermode returned before each test was $32^{\circ}C$, and the cut-off temperatures were $10^{\circ}C$ and $50^{\circ}C$. The contact surface area for the thermode was 2.5 cm x 5 cm. An average threshold was calculated from three measurements in each area.

Equipment: The MSA Thermal Sensory Analyzer (Somedic AB, Farsta, Sweden) was used to determined thermal detection and pain thresholds (Figure 6.6). It is widely used in routine clinical diagnosis to functionally diagnose pain and temperature-conducting nerve fibres (C and A-delta fibres).



Figure 6.6. MSA Thermal Sensory Analyzer - Somedic

Mechanical detection threshold (MDT)

Methods and background: The MDT was determined with a standardised set of modified von Frey hairs that exert forces between 0.25 and 512mN. The threshold was defined as the smallest force necessary for patient perception in response to one von Frey filament bending on the skin. This was determined by performing a modified "method of limits" manner using series of alternative ascending and descending stimulus intensity in five series. Subjects were asked to indicate when they felt the slightest touch of the filament (Appendix 40). The von Frey hairs eliciting 16mN force were applied first, followed by hairs of consecutively lower intensity until the patient could not detect the stimulus being applied. This respective force represents the first threshold value. The order in which the stimuli were applied was then reversed and stimuli of consecutively greater intensity were applied until sensation was detected (this intensity became the second value). Again, hairs with decreasing intensity were applied until in total five upper and lower values of detections were fulfilled from which the mechanical detection threshold was determined. If the first von Frey filament with an intensity of 16mN was not detected, the next highest intensity filament to be detected was used as a starting intensity, and the above procedure were applied. This procedure were repeated five times. The final threshold was the geometric mean of the five series of ascending and descending stimulus intensities (i.e. 10 determinations) (Rolke et al., 2006b).

Equipment: A set of standardised von Frey hairs (0.25 mN, 0.5 mN, 1 mN, 2 mN, 4 mN, 8 mN, 16 mN, 32 mN, 64 mN, 128 mN and 256 mN) (Optihair2-Set, Marstock Nervtest, Schreisheim, Germany). The contact area of the hairs with the skin is of uniform size (about 1 mm2) and texture (rounded contact area to avoid sharp edges, which could facilitate nociceptor stimulation).

Mechanical pain threshold (MPT)

Methods and background: The MPT was measured using a standardised set of pinprick punctuate probes that exert fixed stimulus intensities between 8 mN and 512 mN. The patient's skin response was sensed to the needle probe itself and not to the guided tube. The introduction and removal of the pinprick probes were carried out in smooth movements, allowing the probe to be in contact with the skin for two seconds. The threshold was defined as a geometric average of the MPT in the tested skin area, which was determined by performing a modified "method of limits" using a series of alternative ascending and descending pinprick stimulus intensities in five series. A pinprick probe with 8 mN was used as starting force. The tip of the needle was gently placed perpendicular to the skin's surface, then a weight was applied. Subjects were asked to indicate if the sensation was felt "sharp" or "blunt". Beginning with an applied force of 8 mN stimuli, intensity increased until the sensation induced was described as "sharp". The corresponding force used represented the first threshold value. To follow, the order in which the stimuli were applied was reversed by applying stimuli of consecutive lower intensities until the quality of the sensation was detected as not sharp (i.e. blunt). This force became the second value. The procedures were continued in this fashion until, in total, five first and second values of detection were completed from which the MPT was determined. The final threshold was the geometric mean of five series of ascending and descending stimulus intensities (Appendix 40).

Equipment: A set of seven metal probes with standardised stimulus intensities (8 mN, 16 mN, 32 mN, 64 mN, 128 mN, 256 mN and 512 mN) (MRC Systems GmbH, Germany) with uniform skin contact area (0.2 mm diameter). Penetration of the skin with these stimuli is not possible.

Stimulus-response functions: mechanical pain sensitivity (MPS) for pinprick stimuli and dynamic mechanical allodynia (DMA) for light touch
Methods and background: The MPS was assessed using the same weighted pinprick probe stimuli of different stimulus intensities as for MPT. These seven pinprick stimuli were applied in a balanced order; each stimulus was applied five times. Subjects were asked to give a pain rating for each stimulus on an NRS (range 0-100) where 0 means "no pain" but is not synonymous with "not felt/ detected" (Appendix 40). An undetected stimulus should be noted as Ø. The most intense pain imaginable is represented by 100. Pain in response to light touch (i.e. DMA) was tested using innocuous stimuli (Q-tip, cotton wisp, and soft brush) applied in between the pinprick stimuli in balance and standardised order, and subjects were asked to give a rating on the same scale as for pinprick stimuli. The tests for MPS and DMA were applied given in runs of 10 (five runs per test sites) and each run consisted of a different randomised sequence of seven pinprick stimuli and three tactile stimulus. MPS was calculated as the geometric mean of all numerical rating for pinprick stimuli and DMA as the geometric mean of all numerical rating across all three different types of light touch stimulators. Thus, a stimulus-response function aimed to explore whether a hyper- or hypoalgesia or mechanical allodynia was present.

Equipment: A set of seven metal probes with standardised stimulus intensities (8 mN, 16 mN, 32 mN, 64 mN, 128 mN, 256 mN and 512 mN) (MRC Systems GmbH, Germany) with uniform skin contact area (0.2 mm diameter). In addition, a set of three light intensity stimuli: a cotton wisp (3 mN), a cotton wool tip (Q-tip) fixed to an elastic strip (100 mN), and a paint brush with an applied force of between 200 mN and 400 mN was used.

Wind-up phenomenon (WUR)

Methods and background: WUR is the perceptual correlate of temporal pain summation for repetitive pinprick stimuli. In this test the perceived magnitude of a single pinprick stimulus (256 mN) was compared with that of a series of 10 pinprick stimuli of the same force repeated at a 1/s rate within an area of 1 cm². The time interval was standardised using a metronome timer (Korg MA-30, Japan). In general, immediately following the single stimulus and series of stimuli, an evaluation of the sensation was provided according to a verbal numerical scale (0–100: 0, "no pain"; 100, "most intense pain imaginable"). First, a single stimulus was applied and the subject was asked to give a pain rating for this stimulus. Then the repeated stimuli were applied and the subject was requested to give a pain rating representing the pain over that whole series of the 10 stimuli. This procedure of single pinprick stimuli was alternated with the series of 10 stimuli until both were performed five times at five different skin sites within an area of 1 cm² of the same testing site. A geometric average of the "wind-up" was calculated from the five ratios (Appendix 40).

Equipment: A single pinprick stimulus with standardised intensity (256 mN) of the pinprick set applied in a flat contact area of 0.25 mm diameter.

Vibration detection threshold (VDT)

Methods and background: A standardized tuning fork (64 Hz) was used for the measurement of vibration detection threshold. The tuning fork had two arms, both of them with calibrated weights at their ends. A shape of an elongated triangle and a ninepoint arbitrary scale from 0–8 beside the triangle were imprinted on each weight. Once the turning fork start to vibrate, the triangle on each arm appears as two virtual, intersecting triangles. The triangles moves exceptionally up the scale, as the arms' vibration decreased. Once the subject indicated that the vibration was no longer felt, the nearest value (to the closest half-point) to the point of intersection of the triangles was then recorded as the vibration threshold. The threshold was determined by performing three series of descending stimulus intensities decided from the "wandering" tip of a triangle moved by means of the vibration and indicated on the tuning fork that was placed over a bony prominence of the ulnar (ulnar styloid process) for upper extremities or medial malleolus for lower extremities (Rolke et al., 2006b). It was measured three times as the amplitude at which the vibration was no longer detected, which was indicated verbally (Appendix 40).

Equipment: A Rydel-Seiffer tuning fork (64 Hz, 8/8 scale) (Martin, Tuttlingen, Germany) as used routinely in the clinic (Figure 6.7).



Figure 6.7. Tuning fork

Pressure pain threshold (PPT)

Methods and background: The deep PPT was measured above the hypothenar eminence muscle of the test and control sites. Pressure was increased continually using a pressure stimulus (with an application rate of approximately, 0.5 kg/s corresponding to 50 kPa/s) until pain was indicated verbally by the subject. The subjects were asked to say "now" as soon as the sensation changed from pressure alone to pressure and pain. An average of three measurements per site were taken and the mean value of these was used for analysis (Appendix 40).

Equipment: A blunt mechanical stimulus (contact area of the probe is 1 cm², applied force up to 20 Kg/2000 kPa/200 N), with an in-built pressure gauge (Pressure algesiometer Wagner Instrument – Somedic AB, Farsta, Sweden).

6.7 Data recording and management

6.7.1 Overview

All data were recorded during the interview assessment on standardised patient record forms (PRF) (Appendix 23). The study forms were kept in a separate set of case notes from the usual clinical records. All study records were kept in a locked area accessed only by two nominated study team members.

The EpiData 3.1 software, which has an adequate electronic data capture module especially for data validation, was chosen as a database for the study. The design and

development of the database was done by me. The EQUISTA database system (described below), which was developed and provided by the DFNS, was used for the QST data.

6.7.2 Data preparation

Various steps were taken to prepare the data for statistical analysis. The completed questionnaires from healthy participants and patients were checked for missed and/or unclear answers. Data from the PRF and questionnaires (GHQ-12, BPI) were coded to represent the category they belonged to, for example, female subjects were assigned the code 0, and males were assigned 1. The coding sheets were designed before starting data collection. The coding system was standardised to maintain consistency through all data sets. The data were then verified and entered from the PRF into the data base by myself. As there were validation checks in the EpiInfo, the single entry was considered viable to ensure that the recorded data on the PRF was transcribed into the database. In addition, double entry was done for five patients who were randomly selected for validating the quality of data entry process.

6.8 Statistical analysis

6.8.1 Strategy of analysis

Data analysis was conducted using Stata/IC version 13.1 (Stata Corp LP, College Station, Texas, USA). Further, QST data analysis was conducted using GraphPad Prism version 6.0 for Windows (GraphPad Software, San Diego California, USA). Patients with NP or who had a GHQ-12 of three or above (i.e. symptoms and signs suggestive of depression) were considered as cases and the outcome was coded as 0 or 1 for the purposes of logistic regression. The controls were selected from group 1, 2, and 3, and from healthy volunteers.

The statistical analysis was conducted in two steps: descriptive and analytical. The descriptive analysis was carried out for all patients and in groups; the participants were described in terms of demographic data and their responses to study questionnaires. The second step was the test for the association between exposures and outcome variables. Data is presented throughout as mean \pm standard error mean unless stated otherwise. The QST measures were plotted as dot plots with overlaid box plots for mean/median, variance (box), and range (bars).

The statistical tests used were: paired and unpaired Student's t-tests; the Mann-Whitney U test (also called Wilcoxon rank-sum test); the Kruskal-Wallis test; Pearson's Chisquared test; and Spearman's correlation coefficient. A statistical advisor supervised the statistical analysis of the data and verified the appropriateness of the tests. The flow chart of the appropriate statistical tests is shown in Figure 6.8.



Figure 6.8. Flow charts indicating appropiate statistical test

The flow chart shows the appropriate techniques in different circumstances. Modified from basic statistics road map by Petrie (Petrie and Sabin, 2009)

The normality of the distribution of the continuous data was assessed by generating normal histograms for the variables of interest. Two steps were performed to check for the skewed or non-normal results: first the bell-shape was checked based on the eyeball test, then a confirmatory test was performed using the Shapiro-Wilk test. The null-hypothesis for this test is that the variable has a normal distribution - a non-significant result indicates normality. In case the assumption of normality could not be rejected then a parametric test was applied. If the assumption of normality was not reasonable and had to be rejected then the non-parametric Mann-Whitney U test was applied. The unpaired Student's t-tests were performed to assess for significant differences between

two independent samples testing one variable. One-way ANOVAs were used to assess for significant differences between two or more independent samples with one variable tested. The two-way ANOVAs were calculated to assess for significant differences between two or more independent samples with two variables tested. Non-parametric tests were used to assess statistical significance. The Mann-Whitney U test was used to assess non-parametric result. The Mann-Whitney U test was used for all statistical tests of continuous variables and Fisher's exact test was used to compare dichotomous variables. A significance level of at least p-value ≤ 0.05 was used to state whether a result was significant or not, unless stated otherwise. The number of data values that were included in the calculation was represented by "n".

6.8.1.1 Descriptive analysis

In this study the distributions of each of the variables were created to see the characteristics of the study population and the validity of the data. For quantitative data, the frequency of distribution, cumulative relative frequency, and percentage were calculated. For each grouping variable, such as age, the minimum and maximum values were obtained and the row data were checked for the accuracy of the limits. The variable age was divided into five age-group (18-19/20-29/30-39/40-49/50-59). However, the scarcity of events led to a new grouping for analysis purposes. Therefore, during multivariate analysis, age was categorized into two groups ≤ 30 years and >30years. This cut off (30 years) was used as the median age skewed to the right on the age histogram. Similarly, the period of symptoms prior to diagnosis was grouped into the delay in presentation and divided into <six months, from six months to one year, and >one year. Other variables that were grouped were: thickened nerves, tender nerves, sensory impairment, motor impairment, and depression. These were grouped into present or absent. Mean, median, and mode were calculated as measures for data location. Range, percentile, and standard deviation (SD) were calculated as measures of data spread. The mean and its confidence interval (CI) were reported for continuous normally distributed variables, whereas the median and inter-quartile range (the difference between the 25th and 75th quartiles) were reported for non-normally distributed variables.

Data were also displayed in graphs. Bar and/or pie charts were used for graphing frequency distribution of categorical data, while for the quantitative continuous data the histogram was used. Bar charts were also used to represent qualitative data.

Brief pain inventory (BPI)

The descriptive statistics including mean, SD, median, and interquartile range were calculate for each patient. To further explore BPI, the worst pain score was taken as a reference point, and all patients with a baseline pain score of 0–3 were labelled as having mild pain, all patients with a baseline pain score of 4–7 were labelled as having moderate pain, and all patients with a baseline pain score of 8–10 were labelled as having severe pain. Pearson correlations were then performed to assess associations between all of the BPI pain scores in relations to the GHQ-12 scores within each group.

General health questionnaire (GHQ-12)

The scoring procedure and calculation methods were applied according to recommendations by the author of the GHQ (Goldberg and Williams, 1988). The binary scoring using the 0 0 1 1 method for questions on the GHQ, with 0 signifying absence of illness was used. The threshold value of three was made to maximise sensitivity. Previous studies on leprosy mental disorder used a threshold values range from two-to-four (Bhatia M.S et al., 2006, Senturk et al., 2007, Haroun et al., 2012, Jindal et al., 2013). The cut-off point of three was considered appropriate, as the utilisation of the GHQ-12 in my study was to identify the probable cases and not definite ones. In addition, the mean GHQ-12 score of three for the previous study on psychological disorders among treated leprosy patients with NP in Mumbai (Lasry-Levy et al., 2011), also provided a guide to the best cut-off points as suggested by Goldberg and colleagues (Goldberg and Williams, 1988).

Descriptive statistics including mean, standard deviation, median, and interquartile range were calculated for GHQ-12 items. I first graphically compared the distributions of the male and female scores for leprosy patients. I then compared the scores between patients with no neuropathy, neuropathy, and NP.

6.8.1.2 Analysis

Logistic regression was used for the analysis to produce odds ratios and 95% confidence intervals for the associations and p-values for any differences in proportions seen using the Pearson's Chi Squared Test, and Mantel Haenzel Odds Test. P-values ≤ 0.05 indicated significance. Analysis was done at two levels: first univariate analyses followed by multivariate analyses.

Univariate analysis

Initially univariate analysis was applied to assess any possible univariate association between all the potential variables and outcomes. It was used to look at the strength of associations of the NP (Appendix 46), depression outcomes (Appendix 48), and the differences between two groups.

Multivariate analysis

To investigate the association between these variables more closely, a logistic regression model was developed adjusting each variable for the other one. First multiple logistic regression was carried out to look at groups of variables together, followed by stepwise logistic regression, which identified an overall simplified model including only the important variables. Using this model, variables were added using step-wise methods. They were added in the order of effect estimated from the univariate analysis (Appendix 47 and Appendix 49).

6.8.2 QST data analysis

QST data were first entered into an Excel-based spreadsheet (Excel 2007, Microsoft USA) data analysis system provided by the DFNS (EQUISTA, Casquar GmbH Germany). EQUISTA was designed for data entry of patient demographic and QST raw data. The programme automatically performed z-score transformations of raw QST data values by comparing them against normative reference data published by the DFNS (Magerl et al., 2010); generated thresholds, average ratings, and numbers of observed paradoxical heat sensations. The obtained data were entered into Prism 6 (GraphPad Prism 6.02) to generate specific QST graphs.

QST data transformation

The mathematical transformation of QST data to z-scores has been described by Rolke and colleagues (Rolke et al., 2006b). QST data were log-transformed (log10 units) prior to statistical analysis except HPT and VDT which were normally distributed as raw data (Rolke et al., 2006a, Rolke et al., 2006b). To compare a patient's QST data profile with control data independent of the different units of measurement across QST parameters, the patient's data were z-transformed for each single parameter by using the following expression:

Z-score = (X [single patient] - Mean [healthy controls])/SD [healthy controls]

Z-score values were calculated based on the included healthy control group data. This approach allowed site-specific normalisation of QST data, where each individual parameter was related to its age, gender, and anatomical test site-specific reference range. This procedure resulted in a QST profile where all parameters were displayed as the number of standard deviations above or below the healthy controls. The algebraic sign of z-score values was adjusted for clarity of presentation for each parameter so that it reflects the patient's sensitivity for this parameter. Z-scores above 0 indicate a gain of function, where the patient is more sensitive to the tested stimuli compared with healthy controls (hyperalgesis, allodynia, hyperpathia), while z-scores below 0 indicate a loss of function, referring to lower sensitivity (hypoaesthesia, hypoalgesia) of the patient. Thus, elevations of threshold for any of the 13 parameters measured resulted in negative z-scores.

After this z-transformation it was straightforward to compare a single patient with the group mean of healthy controls, since the reference range of a standard normal distribution is defined as follows:

95% reference range = Mean [healthy controls] \pm 1.96 SD [healthy controls]

The QST scores for individual patients and groups were summarised and presented graphically. Significance was accepted at p-values <0.05 for all analyses. The correlation between leprosy pain symptomatology and physiological and psychological morbidity were assessed by fitting an analysis of rank correlation using Spearman's test.

The DFNS coding system published by Maier and colleagues (Maier et al., 2010) was used to examine combinations of somatosensory function in leprosy patients with neuropathy and pain. A value of 0 was designated for a QST parameter that was found within the normal DFNS reference range; the presence of loss of thermal modalities (i.e. a loss of WDT or CDT) was designated as L1, and the presence of loss of mechanical modalities (i.e. the loss of MDT or VDT) was designated L2. Gain of sensory function to thermal modalities was designated G1 and gain of sensory function to mechanical modalities as G2. When both thermal and mechanical abnormalities were present they were designated as L3 and G3 respectively.

Chapter 7 Validation studies

This chapter describes the pre-implementation work which was conducted prior to the introduction of the main QST profiling study. This stage aimed to validate the investigator, local centre, and population in India in using the DFNS-QST protocol. Further, the training courses on clinical assessment of NP are also discussed. Finally, the chapter presents reflective approaches to simplify the DFNS-QST protocol.

7.1 Background

Normally for the validation of a new investigator or centre for a certification protocol of QST laboratories, the DFNS requires 18 healthy controls in order to validate compatibility (Geber et al., 2009). This is for quality internal assay control purposes to ensure that the values obtained for this population are within the normative ranges held in the large German Research Network on Neuropathic Pain database (DFNS) and giving the guarantee to deliver highest possible objectivity within the range of psychophysical testing.

Two validation studies were conducted among healthy volunteers in London (18 participants) and Mumbai (52 participants). The first study was done in London with 18 healthy volunteer participants. However, in term of validating the site in India, I decided to increase this number to 30 in case there were environmental site differences and ethnic differences in the population studied, since the DFNS normative data were almost entirely collected in European centres with predominately Caucasian population. Should the normative data from this site have differed from those held in DFNS database, we would have been able to use this local population of 30 healthy controls for data analysis. An additional reason for selecting more local Indian healthy volunteers was the targeted number for skin biopsy. As obtaining a biopsy from this population was challenging, the recruitment was continued till I reached 30 participants with both QST and biopsy. The validation studies were important preliminary work in order to conduct a comprehensive somatosensory profile of leprosy patients with NP in a resource-limited setting.

7.2 Overview of my preparation in this PhD research

During my PhD research in Professor Lockwood and Professor Rice's laboratories, I sought to understand the clinical aspects of NP in leprosy. Throughout this time I was continually developing my experience by attending lectures, seminars, meetings, international conferences and training days, all of which employed various teaching and learning techniques. In particular, the training in clinical assessment of NP that I had was as follows:

- Professor Lockwood's clinic (Leprosy clinic at HTD)
- Professor Rice's clinic (Pain clinic at CWH)
- Professor Hadi Manji's clinic (Neurology clinic Queens Square)
- Neuropathic pain training course in Finland (Professor Aki Hietaharju)
- DFNS-QST Training course in Germany (Professors Maier and R-D Treede)
- Dr Bennett's lab (Skin biopsy techniques at KCL)

I developed a broad knowledge of leprosy and NP, and gained the technical skills necessary to clinically assess patients with NP. The knowledge and technical skills obtained have been critical during my PhD research.

7.3 Healthy volunteers study to validate investigator

7.3.1 Introduction

In order to obtain accurate data, high quality training and validation of QST investigators is essential. This training occurs in two phases: firstly the investigator visits a training centre in Germany to be trained in the technique (Geber et al., 2009). The QST-training includes information on the equipment, the investigation technique, recommended procedures for quality control and analysis and interpretation of results. Secondly, the investigator is validated by performing QST on a number of healthy volunteers; these data are then submitted to the central DFNS database in order to check their validity against a large normal dataset. I qualified in the use of DFNS-QST protocol after attending QST-training in Bochum, Germany, and completed the healthy volunteers study in London.

7.3.2 Participants and methods

I examined the QST sensory parameters in 18 healthy participants aged between 18 to 55 years (mean ages 32.2 years, SD 9.8). There was more female (n=11; 61%) than male (n=7; 39%) included in the study. Participants were recruited from the general population and/or staffs and students from the LSHTM and ICL by personal invitation and word of mouth (Appendix 4). Participants were screened for relevant medical history and were specifically questioned about migraine headaches and low back pain. Participants suffering from any acute or chronic pain condition were excluded. All participants were without any pain medication for at least 24 hours before the investigation (Appendix 43). The study was approved by the Imperial College Research Ethics Committee (ICREC_11_2_3) (Appendix 2). All subjects participated after giving written informed consent.

I assessed the sensory parameters in the dorsum of the hands: the ulnar side of the dorsum of the hand (supply area of the ulnar nerve territory, dermatome C8) and the radial side (supply area of the radial nerve, territory C6), using DFNS-QST protocol. Testing took 30 minutes per site. The study was conducted at Chelsea and Westminster Hospital. The detailed testing procedures of the DFNS-QST protocol are described in Chapter 6, in summary the following parameters (Figure 7.1) were assessed:

Thermal detection and pain thresholds and the number of paradoxical heat sensations: The thermal tests were performed using a MSA (SOMEDIC, Sweden, available in three centres). Cold and warm detection thresholds were measured first (CDT, WDT). In addition, subjects were asked about PHS during the thermal sensory limen (TSL) procedure of alternating warm and cold stimuli. Then cold pain and heat pain thresholds were determined (CPT, HPT). The mean threshold temperature of three consecutive measurements was calculated.

Mechanical detection threshold: MDT was measured with a standardised set of von Frey hairs that exerts forces upon bending between 0.25 mN and 512 mN. Using the methods of limits, five thresholds determination were made, each with a series of ascending and descending intensities. The final threshold was the geometric mean of these five series.

Mechanical pain threshold: MPT was measured using pinprick stimuli as a set of seven pinprick mechanical stimulators with fixed stimulus intensities that exerted forces of 8 mN, 16 mN, 32 mN, 64 mN, 128 mN, 256 mN, and 512 mN. The stimulators were applied at a rate of 2 second on, 2 second off in an ascending order until the first percept of sharpness was reached. The final threshold was the geometric mean of five series of ascending and descending stimuli.

Mechanical pain sensitivity: MPS was assessed using the same set of seven weighted pinprick stimuli to obtain a stimulus–response function for pinprick-evoked pain. Subjects were asked to give a pain rating for each stimulus on a '0–100' numerical rating scale ('0' indicating "no pain", and '100' indicating "most intense pain imaginable").

Dynamic mechanical allodynia: ALL was assessed as part of the test above, using a set of three light tactile stimulators as moving innocuous stimuli: Cotton wisp exerting a force of 3 mN, a cotton wool tip fixed to an elastic strip exerting a force of 100 mN, and a standardized brush exerting a force of 200–400 mN. The tactile stimuli were applied with a single stroke of approximately 2 cm in length over the skin.

Wind-up ratio: Participants were asked to give a pain rating intensity of a single pinprick stimulus (256 mN pinprick) this was compared with the estimated mean over the whole series of 10 repetitive pinprick stimuli of the same physical intensity (1/s applied within an area of 1 cm²) using a '0–100' numerical rating scale. The whole procedure was repeated five times. WUR was calculated as the ratio: mean rating of the five series divided by the mean rating of the five single stimuli.

Vibration detection threshold: VDT was performed with tuning fork (64 Hz, 8/8 scale) that was placed over ulna styloid process and a head of the radius, and left there until the subject could not feel vibration any more. VDT was determined as a disappearance threshold with three stimulus repetitions.

Pressure pain threshold: PPT was performed over the thenar and hypothenar muscle with a pressure gage device with a probe area of 1 cm² that exerts forces up to 20 kg/cm² corresponding to 2000 kPa. The pressure pain threshold was determined with three series of ascending stimulus intensities, each applied as a slowly increasing ramp of 50 kPa/s (0.5 kg/cm² s).



Figure 7.1. Complete set of QST devices Picture taken from Professor Rice's lab, 2012

7.3.3 Data evaluation and results

Data entry and transformation was carried out using the Equista database that developed by the DFNSp. The following analyses were carried out:

- i. Comparison of the results to the 95% confidence interval of the DFNS reference data base.
- ii. Number (percentage) of abnormal values
- iii. Calculation of Z-values: data are presented as mean \pm SD and based on the DFNS reference database, I calculated the z-score for each participant:

Z-score = (value [single participant] - Mean [controls])/SD [controls]

iv. Statistical analysis of the z-values of right vs. left ulnar and right vs. left radial using t test for the differences between the parameters within each group.

The result was compared to the 95% confidence interval of the DFNS reference data base, DFNS. Of the total 18 subjects only one had 3 parameters; namely (CPT, HPT and VDT) for the test side, Figure 7.2, and (WDT, CPT and HPT) for the control side, Figure 7.3; which fell outside of the DFNS reference database. The percentage of abnormal values in this group is less than 5% and it is therefore acceptable for DFNS investigator validation.



Figure 7.2. Abnormal parameters in test area compared to DFNS reference



Figure 7.3. Abnormal parameters in control area compared to DFNS reference

Of the 13 parameters obtained for 2 body regions, dynamic mechanical allodynia did not occur among healthy subjects. No significant right-left differences for the ulnar nerve side were observed (p-values for each parameters where greater than 0.05 and range from 0.1 to 0.9) except for MPS (p-value < 0.05) (Figure 7.4 and Figure 7.5).



Figure 7.4. Z-score of Rt vs. Lt ulnar of the 18 subjects compared to DFNS

Graph shows the Ulnar nerve mean values and 95% confidence intervals of all QST parameters in 50 healthy volunteers. The upper confidence limits of all QST parameters were within the limits of the reference data. Rt: Right, Lt: Left



Figure 7.5. Z-score of Rt vs. Lt radial of the 18 subjects compared to DFNS

Graph shows the Radial nerve mean values and 95% confidence intervals of all QST parameters in 50 healthy volunteers. The upper confidence limits of all QST parameters were within the limits of the reference data. Rt: Right, Lt: Left

7.3.4 Discussion

Despite using crude thermal sensory testing for more than a century in the assessment of neuropathy in leprosy patients, its application in clinical and research practice has been limited (Backonja and Lauria, 2010). The DFNS-QST test is a standardised protocol, including standardised instructions for the investigator and subject. It is a noninvasive method used comprehensively to assess the function of both un-myelinated and thinly myelinated small fibres, as well as of the large fibres. Therefore, it has ability to characterise somatosensory functions across the full spectrum of nerve fibres (Krumova, 2010), (Table 7.1). In addition, because it tests the entire system from transduction to perception, the loss of function which manifested clinically as sensory deficit, and the abnormal facilitatory phenomena, which clinically present as various manifestations of pain, can be easily assessed (Walk et al., 2009). It also characterises the function of nociceptive system, which is not possible with standard methods of clinical electro-neurophysiology (Cruccu et al., 2004). These factors make it an appropriate tool for assessing sensory parameters related to pain in leprosy patients.

Axon type	QST parameters
Aβ-fibre	Mechanical detection threshold for von Frey hairs and vibration tests
Aδ-fibre	Cold detection threshold and the mechanical pain threshold for pinprick stimuli
C-fibre	Warm detection threshold and heat pain threshold

Table 7.1. Sensory functions represented by different QST parameters

Furthermore, recent guidelines on NP assessment recommend the importance of exploring QST for prospective therapeutic outcome prediction (Haanpaa et al., 2010). In a study by Dyveke and colleagues, the differentiation between the DFNS somatosensory profile of sensory loss and gain functions was found useful to predict the responses to treatment. This was a randomised, double-blind, placebo-controlled, and phenotype-stratified clinical trial in which 97 patients with peripheral NP due to polyneuropathy were treated with oxcarbazepine or placebo in two 6-week periods. The authors reported that oxcarbazepine is more efficacious for relief of NP in patients with a sensory gain profile than with a sensory loss profile (Demant et al., 2014).

7.4 Healthy volunteers study to validate local centre and population in India

7.4.1 Introduction

The aim of this study was to validate the QST parameters in the Indian population and the site centre. This was for the purposes of internal assay quality control to ensure that the values obtained for this population were within the normative ranges held in the large DFNS database, which is based on data from European centres and mainly Caucasian subjects.

7.4.2 Participants and methods

7.4.2.1 Study design

This study was initiated as a prospective study to investigate the DFNS-QST protocol in an Indian population. It was done at the FMR clinic, from October 2012 to June 2013.

The study was conducted according to the principles of research in humans specified in the Declaration of Helsinki. Ethical approval was obtained from the local Ethics committee of the Foundation for Medical Research, Mumbai, India (IEC No $_$ FMR/ IEC/ LEP/ 04/ 2012) (Appendix 3).

7.4.2.2 Participants

Fifty-two healthy Indian participants, 28 women and 24 men, mean age 30.75 years, range from 18 to 56 years, were recruited in the study (see section 6.4.2.3). All subjects participated voluntarily after giving written informed consent. Participants were excluded if they had been diagnosed with, or suspected to potentially have, any neurological disease including different forms of neuropathy, cutaneous lesions in the tested site, systematic disease, chronic pain, or were taking medication during the time of study (see Methods section 6.4.2.3). Of the 58 volunteers screened for the study, six subjects were excluded. Of these, four had a low Vitamin B_{12} level, one was anaemic and one was pregnant. Travel expenses were reimbursed for all volunteers including those who were excluded.

7.4.2.3 DFNS-QST protocol translation

A rigorous method for translating and checking DFNS-QST protocol was adopted. The forward-translation and back-translation methods were used. This process was only of

rigorous translation, but the protocol was not validated (Maki et al., 2014). The aim of the translation was to achieve an Indian version of the English DFNS-QST questionnaire that was cross-cultural and conceptually equivalent to the local Indian population (Maki et al., 2014, Maneesriwongul and Dixon, 2004). The validation process, which is the process of investigating the reliability, conceptual equivalence, and content validity of translations of questionnaires measures, is beyond the scope of this thesis to explain it in detail. The guidelines provided by WHO were considered during the translation process (WHO, 2014).

Forward translation was conducted first by an independent translation centre in Mumbai (Appendix 52). The local language versions were then reviewed by a member of our research team, Ms Maitreyi Nigwekar, who is a bilingual clinical psychologist. She is familiar with the psychophysical terminology of the DFNS-QST instrument and instructions, as well as being equipped with the interviewing skills required for the task. Also, she is fluent in written and spoken English and is a native speaker of Marathi Indian.

Using the same approach as that outlined in the forward translation, the protocol was then translated back into English by an independent translator (Appendix 52). Final consensus versions were conducted via an expert panel. The panel included the original translator, a clinical psychologist and Dr S. Khadilkar, a neurologist from the Neurology department at JJ Hospital. The panel also included Dr Pai from BLP, Dr Vanaja Shetty, Ms. Capadia and Dr Pandya from FMR. They are fluent in both English and Hindi/Marathi, and are familiar with the terminology used in the leprosy and pain fields. The local language versions were reviewed by the panel and any inadequate expressions/concepts in the translation were identified and discussed. In addition, any discrepancies between the forward translation and the existing or other comparable previous versions of the questions were discussed. Discrepancies were discussed until a satisfactory version was agreed. The translation procedure review was conducted several times until an agreement was reached for the final Hindi/Marathi version.

Following completion of the translation work, the final versions of the translated protocol were sent to the DFNS for approval. Pre-testing of the protocol on the target population was done before the final version was used for the study. The pilot was tested on a sample of five participants: two volunteers from FMR staff and three patients from the BLP clinic. The goal in this step was to test the DFNS-QST instructions and the time taken to complete an interview. The main methodological change and modification as a consequence of the piloting was to test only one site. The decision to restrict the site of the test to the hand (dermatome C8 bilaterally) instead of the hand and feet was made in order to minimise the duration of the interview.

7.4.2.4 Recording techniques and testing protocol

The DFNS-QST protocol was tested bilaterally on a defined area on the dorsum of the hand (sensory region of the ulnar nerve, dermatome C8) or dorsum of the foot (dermatome S1). Participants were tested under identical conditions in a comfortable position. Testing was conducted in a quiet room with the ambient temperature controlled at 21°C to 23°C. All participants underwent a practice session in which the different parameters were applied to the forearm until they were familiarised with the measurement procedure and the equipment. Participants were unable to watch the computer screen at any times during the test procedures. They were also not given auditory or visual clues to indicate the start of individual stimuli. Testing of each site took \sim 30 minutes and a complete test at all two sites took \sim 60 minutes. All tests were carried out by the same investigator who was trained in the use of QST according to the DFNS protocol. All participants who underwent the protocol began with the determination of thermal thresholds followed by mechanical thresholds. The detailed testing procedures of the different parameters of the standardised QST battery according to DFNS protocol is described in chapter 6. These different parameters can be listed chronologically as follows:

- Thermal detection for cold and warm detection threshold (CDT, WDT);
- Number of paradoxical heat sensations (PHS) during thermal sensory limen procedure (TSL);
- Thermal pain threshold for cold and heat pain threshold (CPT, HPT);
- Mechanical detection and pain threshold (MDT, MPT);
- Mechanical pain sensitivity for pinprick stimuli (MPS);
- Dynamic mechanical allodynia (DMA) for stroking light touch;
- Wind-up ratio (WUR);
- Vibration detection threshold (VDT);
- Pressure pain threshold (PPT).

7.4.3 Data evaluation and results

Data entry and transformation was conducted according to the DFNS guidelines. Data were first entered into an Microsoft Excel-spread sheet (Equista database) provided by DFNS, which automatically generated thresholds and average ratings and number of observed paradoxical heat sensations. The obtained data were entered into a STATA data file and GraphPrism to generate analysis and graphs, respectively. The following analyses were carried out:

- Comparison of the results to the 95% confidence interval of the DFNS reference data.
- Number (percentage) of abnormal values
- Calculation of Z-values: data are presented as mean ± SD and based on the DFNS reference database, I calculated the z-score for each participant:

Z-score = (value [single participant] - Mean [controls])/SD [controls]

• Statistical analysis of the Z-values of right ulnar vs. left ulnar and using t test for the differences between the parameters within each group.

Fifty-two volunteers participated in this study. All subjects completed all QST measurements and 30 subjects completed both QST and skin biopsy interventions; there were no drop-outs. None of the subjects reported any adverse effects. Forty-eight subjects completed all QST measures in their hands, two subjects on feet, and two subjects on both hands and feet. The data obtained from all QST measurements for the two test sites are presented as mean and SDs in Table 7.2, Figure 7.6, and Figure 7.7. QST in two different test sites yielded values that were within the published DFNS reference values; the number of values outside the 95% confidence interval (5.27%) was actually slightly higher than its expected published value (5%) (Rolke et al., 2006a, Maier et al., 2010). The result was validated by the German Neuropath Pain Group (DFNS).

7.4.4 Discussion

This is the first study to use the DFNS-QST protocol in a resource-limited setting outside of Europe and validates the Mumbai Centre for the use of the QST. Normative QST data are generated by evaluating somatosensory function in healthy volunteers, a process in which one body area is assessed using the QST measures according to the DFNS protocol.

The findings in our study revealed that one or more somatosensory abnormalities were present in healthy controls. The percentage of abnormal findings expected in healthy controls is 5%, with the exception of DMA, which is not present in healthy controls. In accordance with this exception, our data showed abnormal sensory function for 5.27%. Although our healthy volunteer data is in line with data published by the DFNS and others, the main abnormality occurred mostly in one parameter. For instance, of those volunteers who had abnormal findings at the test site "dorsum of the hand", the wind-up ratio represented one-third of these abnormalities. This could be explained by the complexity of the test and the interpretation by subjects.

The normative QST data generated from the dermatome C8 of a non-Caucasian population may help to increase the generalizability of the DFNS reference database. In addition, validating the Mumbai site for QST protocol in Hindi and Marathi may help to establish somatosensory profiles in other diseases such as diabetics and HIV/AIDS in India.

QST parameter *	Test sites				
		Hand (n=50)		Foot (n=4)	
		Mean \pm SD	Number of subjects	Mean ± SD	Number of subjects
			outside the 95% CI		outside the 95% CI
I hermal thresholds					
Cold detection threshold (CDT)	Δ °C from baseline 32°C	-0.29 ± 0.87	2	-0.23 ± 0.25	0
Warm detection threshold (WDT)	Δ °C from baseline 32°C	-0.45 ± 0.87	3	-0.77 ± 0.57	0
Thermal sensory limen (TSL)	Δ°C	-0.78 ± 0.59	1	-1.04 ± 0.48	0
Paradoxical heat sensations (PHS)	x/3	0.00 ± 0.00	0	0.00 ± 0.00	0
Cold pain threshold (CPT)	°C	1.20 ± 0.49	3	1.01 ± 0.79	0
Heat pain threshold (HPT)	°C	0.40 ± 0.63	1	0.36 ± 0.29	0
Mechanical thresholds					
Mechanical detection threshold (MDT)	mN	0.97 ± 0.67	1	0.39 ± 0.38	0
Mechanical pain threshold (MPT)	mN	0.27 ± 0.41	0	0.42 ± 0.15	0
Mechanical pain sensitivity (MPS)	NRS	0.14 ± 0.66	4	0.43 ± 0.18	0
Dynamic mechanical allodynia (DMA)	NRS	0.00 ± 0.00	0	0.00 ± 0.00	0
Wind-up ratio (WUR)		1.15 ± 1.44	0	0.23 ± 1.00	0
Vibration detection threshold (VDT)	x/8	0.41 ± 0.42	0	0.41 ± 0.42	0
Pressure pain threshold (PPT)	kPa	-1.22 ± 0.86	4	-1.22 ± 0.86	0
	·		5.27%		0%

Table 7.2. QST results performed in hands (n=50) and feet (n=4)

* Data are presented as log data (mean and SDs). In addition, number of values outside the 95% confidence interval of the published DFNS reference values for each parameter are given (in %) which are calculated as follows: 11 tests x 50 subjects = 550 divided by the number of values outside the 95% confidence interval.





7.5 Comparability of mechanical detection and pain thresholds in QST using different devices

7.5.1 Introduction

The aim of this study was to investigate comparability between different devices namely: the Semmes-Weinstein monofilaments, von Frey filaments and "electronic von Frey" using the same psychophysical methods for mechanical detection threshold (MDT), mechanical pain threshold (MPT), and mechanical pain sensitivity (MPS) within the DFNS criteria. The study was designed jointly with Prof Rolf-Detlef Treede from the DFNS group. I conducted this study in Prof Treede's laboratory in Germany and Dr Doreen Pfau from the DFNS group completed the analysis and wrote up the findings.

7.5.2 Materials and Methods

7.5.2.1 Subjects and test areas

Thirteen healthy control subjects (7 female, 6 male; mean age 29 years) were investigated between March 1st and 25th 2011, after giving written informed consent for the QST procedures. Participants were invited and recruited from Mannheim Medical Centre, Mannheim. Ethical approval was obtained from the Ethics committee of University of Mannheim, Mannheim, Germany. Procedures were in accordance with the Declaration of Helsinki.

Tests were performed in the innervation territories of the right and left ulnar nerve on the hand dorsum and on the palmar fingertip (small finger). In the face, test area was within the innervation territory of V2 (maxillary nerve) bilaterally.

7.5.2.2 Equipment

Von Frey filaments

The set of von Frey filaments commonly used within the DFNS protocol for testing the mechanical detection threshold (MDT) (Marstock nervetest; Marburg, Germany) consists of glass fibre filaments with different length and strength. Those factors determine the intensity of the force applied to the skin. The tip of the filaments is rounded tip in order to avoid sharp edges and subsequently in order to avoid a possible nociceptor activation due to sharp edges (Greenspan and McGillis, 1991). Nominal

bending forces are logarithmically increasing forces of 0.25 mN, 0.5 mN, 1 mN, 2 mN, 4 mN, 8 mN, 16 mN, 32 mN, 64 mN, 128 mN, 256 mN and 512 mN.

Semmes Weinstein monofilaments

The Semmes Weinstein monofilament (Brazilian filaments; SORRI-Bauru; Brazil) consists of a plastic handle supporting a nylon filament. Applying forces of 0.05 g, 1 g, 2 g, 20 g, 40 g, and 300g (corresponding to 0.5 mN, 9.8 mN, 19.6 mN, 196.1 mN, 392.3 mN and 2942.0 mN) on the skin using different filaments with the same length but different strength. In contrast to the DFNS von Frey filaments, tips presented an edge leading to a sharp sensation in some subjects at higher stimulus intensities. For this reason, this set of filaments was also used to test the mechanical pain threshold (MPT). They are designed to be cheap robust and suitable for use in low/medium resources settings.

Devices for Mechanical pain threshold and Mechanical Pain Sensitivity

The set of weighted pinprick stimulators used within the DFNS protocol (The Pinprick, MRC systems, Heidelberg, Germany) consists of different steel tubes with a standardized diameter of about 1 cm. Inside the tubes, weights rest on a rim at the end of the tubes when held perpendicularly without touching the skin and end in blunt steel rod with a standardized tip diameter of 0.25mm. As soon as the tips touch the skin and the tubes are moved slightly towards the skin, forces delivered by the weights are completely carried on the skin by the small steel rods, resulting in healthy subjects in "blunt" sensation for lower and "sharp" sensations for higher forces. This set of stimulators covers intensities representing logarithmically increasing forces of 8 mN, 16 mN, 32 mN, 64 mN, 128 mN, 256 mN and 512 mN.

Electronic von Frey filament

The electronic von Frey filament (Somedic, Sweden) represents a device eligible for the testing of a mechanical pain threshold (MPT). The tip diameter is 0.25 mm and thus comparable with the set of DFNS pinprick stimulators. The replaceable and sterilisable tip tapers in a cylindric form to avoid skin penetration and is connected with a handpiece and a force sensor and was connected via a force transducer and a computer with the corresponding software. A visual feedback control of the applied ramp rate is possible due to flashing lights on the back of the handpiece. Subjects could indicate the

first sharp sensation (mechanical pain threshold; MPT) by a stop button or the pain intensity (mechanical pain sensitivity; MPS) by an electronic visual analogue scale (VAS).

Optical feedback control indicated by flashing lights on the holder was limited to a lowest ramp rate of 10g/s, corresponding to about 100 mN/s. As the lowest mechanical pain threshold within the normal range according to the DFNS protocol depending on age and gender of tested subjects (Magerl et al., 2010) is expected at about 40mN, threshold would be missed by applying a ramp rate of 100 mN/s. For this reason we added a testing trial with a ramp rate of 1g/s corresponding to ≈ 10 mN/s. Instead of the optical feedback control directly on the handpiece, the control of ramp rates was possible by an optical feedback given by functions for pressure vs. time on the computer screen. Applied stimulus ramps were thus 10g/s and 1g/s, respectively, to test a possible influence of subject's reaction time or the influence of steeper increasing pressure ramps delivered by sharp stimuli per se (List et al., 1991).

Cut-off value for testing on the hand was 74 g, corresponding to 724mN as presenting the calculated cut-off value within the DFNS procedure using "The Pinprick" to avoid tissue damage. Those values are calculated when the 512mN pinprick stimulator is perceived as non-pricking, and a virtual value of 1024 is then defined as pricking stimulus intensity, resulting in a geometrical mean value of 724 mN for the mechanical pain threshold. Similarly, the cut-off value for testing in the face was 37g, corresponding to 362 mN. As the 512 mN Pinprick stimulus is recommended not to be used in the face to avoid any skin damage, 512 mN is defined as virtually pricking value if the 256 mN device is perceived as blunt sensation, resulting in a geometrical mean value of 362mN.

7.5.2.3 Procedures

Mechanical detection threshold (MDT)

Mechanical detection threshold was performed according to the DFNS protocol using a method of limits (Rolke et al., 2006b),

- with the DFNS von Frey filament, starting with a probable suprathreshold filament of 16mN
- with the Semmes Weinstein monofilaments, starting with a probable suprashreshold stimulus intensity of 2g, corresponding to 20mN

Five subthreshold and five suprasthreshold values were defined. Geometric mean values of sub- and suprashreshold stimuli then represent the mean tactile detection threshold.

Mechanical pain threshold (MPT)

Mechanical pain threshold was performed with different procedures as described below:

- 1. according to the DFNS protocol using the method of limits and asking the subject of differentiate between a sharp or blunt sensation evoked by
 - i. a set of seven weighted pinprick stimuli (,The Pinprick'), starting with a probable sub-threshold stimulus intensity of 8mN
 - ii. Semmes Weinstein monofilaments with a probable sub-threshold stimulus intensity of 2 g, corresponding to ≈20mN; as filaments are more flexible, any pain sensation was expected for higher intensities compared to the blunt needle stimulator (The pinprick).

Then five sub-threshold and five supra-threshold values were defined. Geometric mean values of sub- and supra-threshold stimuli then represent the mean mechanical pain threshold.

2. using a three-time threshold determination of a continuously increasing ramp rate at both, 1g/s or 10g/s using the electronic von Frey Filament with a stop button. Subjects were asked to press the stop button as soon as they perceive any sharp sensation representing the activation threshold of nociceptors (cite).

Mechanical pain sensitivity

Using the weighted pinprick stimulators and the Semmes-Weinstein filaments, the procedure for testing the mechanical pain sensitivity (MPS) used within the QST protocol according to DFNS protocol was used. Mechanical pain sensitivity was tested with the same standardized punctate probes as used for testing of the mechanical pain threshold. Dynamic mechanical allodynia (DMA) was tested using standardized light touch stimuli: (1) a cotton wisp applying a force of about 3 mN, (2) a Q-tip, fixed in a flexible plastic mount, exerting a force of about 100 mN when slightly bent, and (3) a standardized brush applying forces of 200–400 mN (Senselab Brush-05, Somedic, Sweden). To evaluate the pain intensity of subjects, a numerical rating scale from 0 (no pain) to 100 (most intense pain imaginable) was used for all test stimuli. Subjects were

free to use integers as well as fractions ad libitum. They were instructed to distinguish pain from the perception of touch or pressure by the presence of a sharp or slightly pricking or burning sensation.

Using the electronic von Frey filament subjects rated the intensity of any pain-related sensation on an electronic VAS which was connected via the force transducer (SenseBox) with the computer, recording ratings and applied forces. Ramp rates were the same as for the testing of the mechanical pain threshold via the evF as are 1g/s and 10g/s. For data analysis, corresponding ratings to the forces within the DFNS protocol applied (8, 16, 32, 64, 128, 256 and over extra-trigeminal areas additionally 512 mN) were extracted. A small constant of 0.1 was added to each rating to avoid a loss of zero values during data analysis (Magerl et al., 2010).

Order of test procedures

Within the test protocol, test procedures were performed in the following order:

- i. Mechanical detection threshold (MDT)
- ii. Mechanical pain threshold (MPT)
- iii. Mechanical pain sensitivity (MPS)

Order was balanced over test areas (face, fingertip, hand dorsum) and over the application order of used test devices.

7.5.3 Results

7.5.3.1 Mechanical detection threshold

Mean values for MDT tested by von Frey vs. Brazilian monofilaments did not differ over all test areas (Figure 7.8), (mean values over face: 0.46mN vs. 0.59mN (log-mean \pm SEM: -0.338 \pm 0.098 vs. -0.229 \pm 0.067); fingertip: 0.57mN vs. 0.62mN (-0.241 \pm 0.103 vs. -0.208 \pm 0.053); hand dorsum: 1.88mN vs. 1.83mN (0.273 \pm 0.121 vs. 0.263 \pm 0.129) and were highly correlated (r=0.71-0.81).



Figure 7.8. MDT results from the face, fingertip and hand

von Frey (dark bar column) and Brazilian monofilaments (white bar column)

7.5.3.2 Mechanical pain threshold

Mean values for MPT differed significantly for pinprick stimuli vs. Brazilian monofilaments by 24.9mN vs. 12.7mN over face (log-mean \pm SEM: 1.396 \pm 0.041 vs. 1.103 \pm 0.094; p-value <0.01) but not for the fingertip: 55.1mN vs. 78.1mN (1.741 \pm 0.074 vs. 1.893 \pm 0.206) and the hand dorsum: 55.4mN vs. 69.8mN (1.744 \pm 0.064 vs. 1.844 \pm 0.182). Compared to values tested by pinprick stimuli, mean values tested by the evF differed significantly for both ramp rates in all test areas; in the face with 90.3mN (1.956 \pm 0.054; 1g/s) and 176.0mN (2.245 \pm 0.055; 10g/s), on the fingertip with 121.8 mN (2.085 \pm 0.060; 1g/s) and 223.4mN (2.349 \pm 0.034; 10g/s) and on the hand dorsum with 144.6mN (2.160 \pm 0.061; 1g/s) and 239.6mN (2.379 \pm 0.061; 10g/s) (Figure 7.9).





Comparison of the set of weighted Pinprick stimuli (black bars) with the Brazilian monofilaments (white bars) and an electronic von Frey filament with the ramp rates of 1 g/s and 10g/s (light/dark grey bars). n.s. = non-significant, *p-value <0.05; **p-value <0.01; ***p-value <0.001; paired t-test.

7.5.3.3 Mechanical pain sensitivity

Mean values for complete MPS-DFNS procedure were 1.09/100 in the face (0.039 ± 0.082) , 0.92 on the fingertip (-0.037 \pm 0.117), and 0.84 on the hand dorsum (-0.077 \pm 0.083). The calculation of MPS with the evF was not possible as subjects did not rate pain below stimulus intensities of 128mN, and cut-off value for the face was set at 256mN. For this reason, we used a stimulus response function with only 2 stimulus intensities for comparison. Mean pain rating for 'short' MPS in the face was 7.75 (0.889 \pm 0.103) for Pinprick stimuli vs. evF 1g/s: 0.56 (-0,254 \pm 0.138) and evF 10g/s: 0.25 (-0.601 \pm 0.066). On the fingertip, rating was 3.56 (0.551 \pm 0.178) for the Pinprick vs. evF 1g/s: 0.71 (-0.149 \pm 0.153) and evF 10g/s: 0.24 (-0.621 \pm 0.089). On the hand, rating was 3.59 (0.555 \pm 0.134) vs. evF 1g/s: 0.6 (-0.210 \pm 0.148) and evF 10g/s: 0.17 (-0.758 \pm 0.041).

7.5.4 Discussion

The DFNS-QST protocol is a standardised method. It has been used globally to explore somatosensory profiles in patients with neuropathic pain. The high standardization of this QST protocol requires specific devices to be used within the protocol. This may represent one of the most important limitations of further expansions of the use of the protocol due to high costs and restricted practicability.

In this study the Semmes-Weinstein monofilaments yield similar results for the testing of MDT and MPT according to the DFNS protocol in healthy subjects. Due to larger increments of applied intensities compared to the DFNS von Frey filaments, neuropathies may be underestimated in more sensitive test areas (face, fingertip). Additional variance may arise from non-standardized tip surfaces of the Semmes-Weinstein monofilaments. The suitability of the use on the hand should be confirmed by testing of subjects with sensory loss and gain, i.e. under pathological condition. Semmes-Weinstein monofilaments are more robust than the von Frey filaments used within the DFNS and cost effective. Further, they are not usually affected with dust or required calibration. For these reasons their use would be favourable in resource-limited settings, and may also be used for the testing of MDT and replace a set of von Frey filaments stimuli used within the DFNS.

MPT and MPS tested with the electronic von Frey filament are not comparable with the responses measured by pinprick stimuli within the DFNS-QST protocol which may be caused by a different mode of application (static stimuli vs. dynamic stimuli). For the dynamic application, reaction time may play an important role, which is demonstrated by different MPT and MPS values for different ramp rates. This indicates that investigating comparability between inter-observer testing and test-retest in same subjects are needed.

7.5.5 Conclusions

The use of the Brazilian monofilaments yield similar results as the DFNS instruments for both, MDT and MPT on the hand dorsum but has a limited range of use in more sensitive areas due to a limited grading of intensities at lower forces. The electronic von Frey filament was frequently insufficient to reach the threshold for sharp sensation as used for the DFNS protocol and is not suitable for the use within that protocol.

Chapter 8: Results

Chapter 8 Results

8.1 Introduction

This chapter presents the results of the data analysis as follows: first, descriptive results are shown for all leprosy patients, and study groups; followed by results for pain description questionnaires; clinical examinations; and Quantitative Sensory Testing. The last result section describes the impact of neuropathic pain on quality of life.

8.2 Descriptive analysis

8.2.1 Baseline characteristics of study population

8.2.1.1 Participants

Ninety-nine leprosy patients with and without pain; and 52 healthy volunteers were enrolled into Leprosy Pain India study (LePaIn) between 10^{th} October 2012 and 30^{th} April 2013. Patients were allocated into four groups based on clinical evidence of neuropathy and presence of pain (Figure 8.1). Thirty-six patients had pain at interview. Of these, twenty-four participants were clinically identified as having neuropathic pain in the affected ulnar nerve based on a score of $\geq 4/10$ of the DN4 questionnaire and clinical neurological examination.



Figure 8.1. Flow diagram for LePaIn study

G1: Group 1; G2: Group 2; G3: Group 3; G4: Group 4; UL: Upper limbs; LL: Lower limbs

Demographic and clinical characteristics

Age and sex distribution

The mean age of included patients was 32.8 years (range 18-60 years). More than two third of patients were in the age groups 20-29 and 30-39 years. Approximately one fifth of the patients were female. The proportion of male and female patients was not evenly distributed across age groups. There were relatively few patients aged below 20s and above 50s age group (Table 8.1).

Age group	Male	Female	Total
(Years)	N = 69 (77%)	N = 21 (23%)	N = 90 (100%)
<20	5 (7.3%)	1 (4.8%)	6 (6.7%)
20-29	31 (44.9%)	6 (28.6%)	37 (41.1%)
30-39	21 (30.4%)	4 (19.1%)	25 (27.8%)
40-49	6 (8.7%)	6 (28.6%)	12 (13.3%)
50-60	6 (8.7%)	4 (19.1%)	10 (11.1%)

Table 8.1. Age distribution of the sample by sex

Geographical distribution

Ninety patients were seen during the period under study, of which 74 patients (82.2%) were recruited from the BLP clinic in Mumbai. Patients came from a wide range of leprosy endemic areas in India. Approximately 50% had migrated from outside BLP catch-up area of services; the remainder were within the state of Maharashtra. The largest single group of patients (42 patients (46.7%)) came from Maharashtra state, with 33.3% from Mumbai; followed by Uttar Pradesh (31 patients (34.4%); and Bihar (10 patients (11.1%)). One patient was from Nepal, and had acquired his leprosy in India, where he had been living since 1990. The majority of the study population (64 patients) had primary education (50 patients) or no formal education (14 patients). Figure 8.2 shows the geographic origin of 90 patients presenting with leprosy to the BLP and FMR.


Figure 8.2. Geographical origin of 90 patients presenting with leprosy

This map is developed by the WHO office in Delhi for the purpose of the current study; with kind permission from GIS / SEARO.

Patients' clinical characteristics

The demographic and clinical details of the study population are shown in Table 8.2. In 65 (72.2%) patients, the diagnosis of leprosy was made after six months from the patients' first symptoms. There was a long lag time between the appearance of first symptoms and disease diagnosis, mean 2.1 years (range 0.3 - 22 years). In 14 (7BL, 7LL) patients, the latent time to diagnosis was 2.1 years (range 0.4 - 10 years). This subgroup were potentially infectious to others before diagnosis and treatment. The main reason for delay was misdiagnosis or an unusual presentation mimicking other common conditions, such as skin diseases. In 33 patients a misdiagnosis was made, all were dermatological related conditions such as psoriasis. Two patients had an unusual presentation described as neurological condition with no skin involvement. Patient factors can cause delay: 31 patient were unaware or neglected their symptoms.

Patients reported a variety of symptoms at the start of their problems, although none of them thought of the possibility of leprosy. In the vast majority (62 patients), these were the descriptions of typical leprosy skin lesions. 29 patients mentioned symptoms related to nerve damage: eighteen had anaesthesia; two had loss of warm sensation; four had burning and tingling sensations; and five had loss of muscle strength, at their first consultation at the BLP clinic. Patients with leprosy frequently presented with complications such as reactions. Seven patients were in reaction at the time of presentation: three of these were Type I (reversal) reactions requiring treatment with oral prednisolone. Four had ENL. In these patients the signs and symptoms of the reaction were the stated reason to seek medical assistance. Over three quarters of the patients (69 (76.6%)) presented their problem to a private doctor. Only 11 (12.2%) of the patients presented directly to leprosy hospital or clinic. Traditional healers, alternative medical practitioners and pharmacists were first consulted in 5.7%, 3.3% and 2.2% of the patients, respectively.

Of the total sample, 18 (20.0%) patients were newly diagnosed and not started MDT at the time of interview; 26 (28.9%) patients had been diagnosed and received MDT within the first six month; 11 (12.2%) between six month and one year; and 35 (38.9%) more than one year after developing their first symptoms of leprosy (Table 8.2).

Patients were classified according to WHO field classification as follows: 57 (63.3%) patients had MB, and 33 (36.7%) had PB. Only 68 (75.6%) patients had been classified clinically using the RJ clinical classification, the remaining 22 (24.4%) was not classified. Of the 68 patients, four patients (5.9%) had TT leprosy, thirty (44.1%) had BT, two (2.9%) had BB, thirteen (19.1%) had BL, nine (13.2%) had LL, six (8.8%) had pure neuritic leprosy, and three (4.4%) had indeterminate leprosy. Of the 90 patients, 18 (20.0%) patients had not yet received leprosy MDT, 26 (28.9%) patients were already taking MDT, and 46 (51.1%) had been released from leprosy treatment (RFT). Of these RFT, the mean time since completed treatment was 3.5 years (range 0.04 – 30.9 years). Twenty five (27.9) patients had received at least two courses of leprosy treatment.

Table 8.3 documents the severity of nerve involvement at diagnosis, 36 (40.0%) patients had evidence of Grade 1 disability from nerve damage involving their hands or feet, with 10% having Grade 2 disability of hands or feet. Two (2.2%) patients had ocular disability due to leprosy, one of which had a Grade 2 disability.

Variable		Frequency	Percentage	Variable		Frequency	Percentage
Centre	BLP	74	82.2%	Dominant hand	Right	86	95.6%
	FMR	16	17.8%		Left	4	4.4%
	Hindi	56	62.2%		Illiterate	14	15.6%
Language	Marathi	32	35.6%		Primary	50	55.6%
	English	2	2.2%	Education	Secondary	11	12.2%
	Hindu	63	70.0%		High school	3	3.3%
Religion	Muslim	25	27.8%		Higher secondary	4	4.4%
0	Christian	1	1.1%		College and above	8	8.9%
	Others	1	1.1%	Alcohol status	No	64	71.1%
	Current smoker	9	10.0%		Yes	26	28.9%
Smoking	Former smoker	8	8.9%	E (Daily	3	3.3%
status	Occasionally	9	10.0%	alcohol use	Weekly	6	6.6%
	Never smoked	64	71.1%		Occasionally	17	18.9%

Table 8.2. Demographic and characteristics of the study population (n=90)

Variable		Frequency	Percentage	Variable		Frequency	Percentage
	None	3	3.3%		TT	4	4.4%
	Housewife	14	15.6%	RJ classification	BT	31	34.4%
Occupation	Labour	41	45.6%		BB	2	2.2%
Cecupation	Farmer / skilled labour	9	10.0%		BL	13	14.4%
	Office worker / Business	3	3.3%		LL	9	10.0%
	Student	7	7.8		PN	6	6.7%
Delay in	No	25	27.8%		Indeterminate	3	3.3%
diagnosis	Yes	65	72.2%		Not known	22	24.4%
	<6month	26	36.1%	WHO	PB	33	34.4%
Delay in diagnosis	6months-12month	11	15.3%	classification	MB	57	65.6%
	>12months	35	48.6%		Newly diagnosed	18	20.0%
Main reasons for delay	Misdiagnosis	33	50.7%	Duration of	<6months	26	28.9%
	Patients factors	31	47.7	disease	6months up to 1yr	11	12.2%
	Admin factors	1	1.54		Longer than 1year	35	38.9%

	Grade 0	Grade 1	Grade 2
Hands and feet	54 (60.0%)	36 (40.0%)	9 (10%)
Eyes	89 (98.9%)	-	1 (1.1%)

Table 8.3. Disability grades present in patients at diagnosis (n=90)

8.2.1.2 Study Groups

Number of participants by group

Patients were divided into the following groups: leprosy patients with no clinical evidence of neuropathy and no pain (Group 1); leprosy patients with subclinical neuropathy and no pain (Group 2); leprosy patients with clinical evidence of neuropathy and no pain (Group 3); leprosy patients with clinical evidence of neuropathy and pain (Group 4). Healthy volunteers group (H.V) (Table 8.4). Group 4 is further divided to neuropathic pain subgroup (NP) and non-neuropathic pain subgroup (Non NP).

Patients with No pain (Group 1, 2 and 3)

Fifty four patients were recruited in three groups of patients with no pain. Of these, 29 patients had a diagnosis of leprosy and no clinical evidence of neuropathy: one group of 14 newly diagnosed leprosy patients had no evidence of neuropathy prior to the study (G1); and the other group consisted of 15 patients with leprosy and had subclinical neuropathy (G2). Another group consisted of 25 patients had leprosy and had clinical evidence of neuropathy (G3).

Patients with pain (Group 4)

In total, 36 patients were recruited with pain. There was poor recruitment in the leprosy patients with pain and no clinical evidence of neuropathy group mainly because patients having completed leprosy treatment with no neuropathic complication may not come to the leprosy clinic for their pain condition. Out of 36 patients with pain, 32 (88.9%) reported pain in their hand and 4 (11.1%) had lower limb pain.

Demographic and clinical characteristics of participants by group

The five groups were not significantly different with respect to age, weight, height, or metabolic factors: Thyroid hormone profiles, glucose, BMI and vitamin B_{12} level. Patient demographics and characteristics by groups are shown in Table 8.5.

Group	Number	Gender	Age (mean,
			range)
Healthy volunteers	52	24 male, 28 female	30.75 (18-55)
No evidence of neuropathy and no pain	14	13 male, 1 female	31.74 (20-51)
Subclinical neuropathy and no pain	15	10 male, 5 female	30.96 (18-56)
Clinical evidence of neuropathy / no pain	25	18 male, 7 female	30.39 (18-53)
Clinical evidence of neuropathy and pain	36	24 male, 12 female	35 (18-60)

Table 8.4. Number of participants and study groups (n=142)

Variable©	Groups	Groups					
	Healthy	Patients					
		Group 1	Group 2	Group 3	Group 4		
Participant's §	52 (63.6)	14 (9.9)	15 (10.6)	25 (17.6)	36 (25.4)		
Male	24 (25.8)	13 (12.9)	10 (10.8)	18 (19.4)	28 (30.1)		
Female	28 (57.1)	1 (2.0)	5 (10.2)	7 (14.3)	8 (16.3)		
Age (years) *	30.8±10.1	31.1±9.6	30.9±10.5	30.4±9.2	35.8±13.1		
Male	30.8±11.2	31.6±9.9	28.2±10.4	30.8±9.6	32.4±12.3		
Female	30.7±9.3	25.1	36.6±9.1	29.3±8.7	47.9±7.9		
Weight *	59.9±13.6	60.1±9.8	54.6±11.3	58.4±12.8	62.3±10.1		
Male	65.1±14.0	61.9±7.5	56.6±9.7	61.7±13.5	62.7±10.9		
Female	55.0±11.4	37.0	50.6±14.3	49.9±5.4	61.3±7.3		
Height †	159.4±8.4	163.4±9.5	157.7±11.0	159.2±7.8	161.0±8.4		
Male	163.0±8.1	164.8 ±8.2	163.9±6.7	162.7±5.5	164.4±6.7		
Female	156.1±7.4	145.0	145.2±5.9	150.3±5.3	150.5±5.2		
BMI (kg/m2)	23.6±4.9	22.5±2.8	22.0±4.2	22.9±4.0	24.1±3.8		
Male	24.5±5.2	22.8±2.5	21.0±3.0	23.2±4.5	23.1±3.3		
Female	22.7±4.6	17.6	23.8±5.9	22.1±2.2	27.2±4.0		
HbA1C *	5.7 (0.5)	4.9 (1.3)	4.9 (1.0)	4.8 (1.1)	6.9 (8.9)		
Male	5.8 (0.5)	4.9 (1.3)	5.0 (1.0)	4.7 (1.1)	7.5 (10.4)		
Female	5.4 (0.4)	1.0 (5.0)	4.6 (1.2)	4.9 (1.2)	5.3 (1.5)		

Table 8.5. Demographic and clinical characteristics of participants by group (n=142)

T3 *		133 (27)	94 (52)	125 (17)	112 (22)	94 (20)
	Male	145 (29)	93 (59)	122 (24)	112 (22)	98 (19)
	Female	115 (8)	100	130	121 (34)	85 (26)
T4 *	1	8.6 (2.6)	6.6 (2.1)	8.4 (0.6)	9.0 (2.1)	17.8 (53.3)
	Male	10.0 (2.4)	7.0 (2.2)	8.0	7.5 (1.3)	6.5 (1.5)
	Female	15 (24.9)	5.0	9.0	11.0	50 (103)
TSH *	k	1.9 (0.9)	1.8 (0.4)	1.3 (0.6)	1.9 (1.3)	2.0 (2.0)
	Male	1.9 (1.0)	1.8 (0.5)	1.0 (0.0)	2.5 (1.3)	1.8 (1.5)
	Female	2.0 (0.5)	2.0	2.0	1.0 (1.0)	3.0 (3.1)
Vitam	in B ₁₂ *	387 (168)	267 (173)	303 (106)	348 (167)	523 (410)
	Male	360 (118)	234 (180)	333 (131)	281 (114)	436 (207)
	Female	427 (248)	400	243.0	436 (207)	927 (595)

©Data are means (SD) (mean \pm standard deviation) or numbers (%); HbA1C – Glycated Haemoglobin results are displayed as percentage, reference range (4 to 6%); T3 total are displayed as ng/dl, reference range (70 to 204ng/dl); T4 total are displayed as µg/dl, reference range (4.87 to 11.72 µg/dl); TSH are displayed as µIU/ml, reference range (0.45 to 4.5 µIU/ml); Vitamin B₁₂ results are displayed as pg/ml, reference range (187 to 883). *Continuous data if normally distributed were analysed with one way anova test (ANOVA). Mean values and SDs shown; † Continuous data not normally distributed were analysed using Kruskal Wallis test. Mean values and SDs shown. § Categorical data were analysed using Chi squared test of association. Values and percentages shown. BMI Body Mass Index.

8.3 Clinical symptoms findings

8.3.1 Pain and sensory symptoms in leprosy patients with pain

8.3.1.1 Patient details

Twenty-six patients (72%) had symptoms and signs suggestive of NP identified by the DN4 questionnaire and 10 patients (28%) with predominately musculoskeletal/ nociceptive pain (non-NP) conditions.

8.3.1.2 Patient characteristics for pain group

General characteristics

The sample of 36 leprosy patients with pain represented a range of neuropathic and non-NPs. There were no significant differences between the two pain subgroups with respect to age, sex, average pain intensity and frequency in the last four weeks and pain intensity during interview. Table 8.6 shows the characteristics of patients with pain classified according to the case definition as having Non-NP or NP in both upper and lower limbs.

Site of pain

Although the most frequent sites of pain for all patients were upper extremities (88.9%), which reflects the study methodology, pain frequently occurred in skin lesions, peripheral nerves or generalized pain (Table 8.7). Of the 36 patients, 13 had skin lesion pain: in 5 patients pain was located in the active untreated skin lesions, and 8 in the treated skin lesions. A wide spectrum of pain site presentation was reported in peripheral nerves: 9 patients had hand and feet pain; 17 patients reported pain at the site of the nerve; and 18 patients had pain in the area of sensory loss over nerve distribution, the nerve most often affected by pain was the ulnar nerve in the upper limbs followed by the lateral common peroneal nerve in the lower limbs. 14 patients had joint pain, and 2 patients had foot neuropathic ulcer pain.

Variable	Non-NP	NP pain	P-
	pain group	group	value
	(n=10)	(n=26)	
Age (years) ^a	31.9 ± 11.8	37.3 ± 13.8	0.28 *
Sex (female/male)	3/7	5/21	0.49 §
Pain duration (weeks) b	4.5 (1; 10)	36 (4; 72)	<0.001†
NRS at interview (NRS 0-10) b	5.5 (5; 8)	6.5 (5; 9)	0.87 *
Maximum pain intensity during last 4 weeks b	9.5 (7; 10)	7.5 (6; 9)	0.10 †
Average pain intensity during last 4 weeks b	5 (4; 6)	5 (4; 7)	0.17 *
Current pain treatment ^c	8 (80.00%)	21 (80.77%)	0.96 §
Number on antidepressant/anticonvulsants ^c	0	5 (19.23%)	0.13 §
Number on analgesics (NSAIDs) ^c	4 (40.00%)	14 (53.84%)	0.71 §
Number on analgesics (steroid) ^c	4 (40.00%)	6 (23.08%)	0.41 §
Pain relief by medication ^c	7 (70.00%)	13 (50.00%)	0.28 §
NRS pain relief (NRS 0-10) ^b	1 (0; 7)	1 (0; 5)	0.09 †

Table 8.6. Characteristics of patients (n=36) with pain

^a mean ± standard deviation; ^b median (25th percentile; 75th percentile); ^c number (%); *Continuous data if normally distributed were analysed with Student t test. Mean values and SDs shown; [†] Continuous data not normally distributed were analysed using Mann-Whitney Rank Sum test. Median and quintile (25th percentile; 75th percentile) shown in brackets. § Categorical data were analysed using Chi squared test of association. Values and percentages shown.

Site		Frequency (%)
Skin lesions	Active untreated skin lesions	5 (13.9%)
	Treated skin lesions	8 (22.2%)
Peripheral nerves	Hands and feet	9 (25.0%)
	Nerve pain	17 (47.2%)
	Area of sensory loss over nerve distribution	18 (50.0%)
Others	Joint pain	14 (38.9%)
	Ulcer pain	2 (5.5%)

Table 8.7. Pain localisation in 36 patients with leprosy

Duration and severity of pain

Patients with NP had longer pain duration than those with non-NP and less likely to have current pain due to reactions, Table 8.6. Pain had been present for one month or less in non-NP patients; whereas neuropathic patients reported pain for longer than nine months. The timing of the pain was reported as all the time by 27 (75.0%) patients, as less than two hours every day by 6 (16.7%) patients, as occasionally by 3 (8.3%) patients, and no patients reported as at least once a week (Figure 8.3).



Figure 8.3. Pain frequency in 36 patients with leprosy

Patients with NP rated their present pain as more intense than those with non-NP (Table 8.6 and Figure 8.4). This finding was not significant and influenced by taking into account gender, or interview setting. Patients with NP were more likely to be taking analgesic treatments during the 24 hours prior to the interview than those with non-NP. A larger proportion of patients with NP subgroup were on pain medication compared to the subgroup with non-NP. Pain intensity at interview (interview NRS) was significantly correlated to pain intensity and frequency over the previous week (Spearman's correlation, P value<0.001). Patients more likely to have non-NP had higher maximal pain scores during the preceding weeks.



Figure 8.4. Pain grade in 36 patients with leprosy

Reaction and pain groups

Of the total sample, 26 patients had reactions at the time of interview; and 45 patients had never experienced reaction. Of these 26 patients, 9 (34.6%) had a T1Rs, 5 (19.2%) had ENL and 12 (46.1%) had neuritis (Table 8.8). 40 (56.3%) patients had a previous history of reaction either at diagnosis, during or after MDT treatment. Of these, neuritis was commonest (64.4%).

Non-NP (nociceptive/inflammatory pain) was found in 10 (11.1%) patients of the total sample. Of these 10 patients, 7 (70.0%) had a current reaction, 2 (20%) had ulcer in their hands or feet, and the remaining 1 (10.0%) had neither reaction nor ulcer.

Leprosy	No pain	Pain		Total	
reaction		Non-neuropathic	Neuropathic	-	
T1R n (%)	2 (22.2%)	5 (55.6%)	2 (22.2%)	9 (100%)	
ENL n (%)	1 (20.0%)	2 (40.0%)	2 (40.0%)	5 (100%)	
Neuritis n (%)	2 (16.7%)	2 (16.7%)	8 (66.7%)	12 (100%)	
Total	5 (19.2%)	8 (34.6%)	12 (46.2%)	26 (100%)	

Table 8.8. Association of pain with reactions at the time of examination (n=26)

Disability assessment

Disability assessment results reflected high disability for patients with NP and mild disability for patients with non-NP, see clinical examination findings below.

8.3.1.3 Pain description questionnaires

Pain symptoms using DN4 and PD-Q

The Hind/Mahrati version of the DN4 was administered to 36 patients with leprosy and pain presenting at the clinic. Of the 36 patients who had pain, the description of numbness occurred mainly in NP subgroup. Itching, aching and dull type pain was most frequently reported in the non-NP subgroup (44.4% - 61.1%). Burning; tingling; sharp; pins and needles sensation; and the descriptor electric-shock like were reported in more than two thirds of NP patients compared to less than one fifth of non-neuropathic. Spontaneous pain was reported in 28 patients during the clinical examination and on PD-Q with increased frequency and increased likelihood of neuropathic pain (nonneuropathic, n=2; neuropathic pain, n=26). Figure 8.5 shows the frequency of reported pain descriptors from patients classified as having neuropathic pain, or non-neuropathic pain using the DN4 questionnaire.



Figure 8.5. Frequency of pain descriptors

Frequency of pain symptoms (seven symptoms from DN4 questionnaires; others: constant and aching or dull are from patients history) reported by 36 patients with pain, classified as non-neuropathic pain and neuropathic pain.

DN4 questionnaire analysis

Of the 32 patients with pain concordant with the distribution of sensory loss at the affected ulnar nerve in the upper limb, the DN4 identified 24 patients with symptoms and signs suggestive of NP (of the 32 patients with upper limb pain, 24 patients had a score of 4 or more indicating the presence of NP), (median score 6, interquartile range (IQR) 2.0; mean score 5.6, standard deviation (SD) \pm 1.4) and 8 patients without NP (median score 2.5, interquartile range (IQR) 1.0; mean score 2.5, standard deviation (SD) \pm 0.5). Figure 8.6 shows the DN4 scores in patients with pain.



Figure 8.6. DN4 scores in 32 leprosy patients with upper limb pain

Frequency of the DN4 scores from 32 patients with upper limb pain, classified as non-neuropathic pain (8 patients) and neuropathic pain (24 patients).

Pain symptoms data obtained from patients, who were classified as having non-NP and NP using the DN4 and clinical examination of upper limbs, were analysed. A nonparametric analysis (Mann-Whitney two-sample statistic) between these two subgroups was performed. The result showed that questions 1 (burning sensations); question 5 (pins and needles); question 6 (numbness); question 8 (hypoesthesia to touch); and question 9 (hypoesthesia to pin-prick) were associated symptoms between non-NP and NP subgroup (p-value ≤ 0.05 ; more present in NP patients). When the DN4 score was considered in total points, there was also a difference between the two pain subgroups (p-value ≤ 0.001). When the analysis was repeated using binary coding for the DN4 questionnaire responses in the upper limbs instead of categorical coding, the findings were unchanged. Subsequent analyses were performed using binary codes for pain symptoms. Significantly, different symptoms were found for five pain symptoms between the two pain groups (Table 8.9).

Q1-10: Clinical relevant complaint	nt Frequency of patient's descriptor present		
S			
	Non-NP (n=10) (%)	NP (n=22) (%)	P- value
Q1, burning sensation	3 (30.0%)	16 (72.7%)	0.02*
Q2, painful cold	0	5 (22.7%)	0.10
Q3, electric shocks	6 (60.0%)	17 (77.3%)	0.31
Q4, tingling sensation	8 (80.0%)	20 (90.9%)	0.34
Q5, pins and needles	2 (20.0%)	14 (63.6%)	0.02*
	1 (10,00/)	10 (0(40/)	0.001*
Q6, numbress	1 (10.0%)	19 (86.4%)	0.001*
07 itching	4 (40,0%)	5 (22 7%)	0.31
	+ (+0.070)	5 (22.770)	0.51
O8. hypoesthesia to touch	1 (10.0%)	17 (77 3%)	0.001*
			0.001
Q9, hypoesthesia to pin-prick	2 (20.0%)	15 (68.2%)	0.01*
Q10, painful brush	3 (30%)	5 (22.7%)	0.66

Table 8.9. A comparison of symptoms detected by DN4 in pain groups (n=32)

↑ DN4 questions significant levels Cochran's nonparametric analysis (McNemar chisquared test) for the upper limb pain (n=32). § Q1-10: Question 1 to 10 of DN4 questionnaire. *P-value ≤0.05

PD-Q questionnaire analysis

Of the 32 patients with pain concordant with the distribution of sensory loss at the affected ulnar nerve in the upper limb (Figure 8.7), the PD-Q identified 13 patients with a likely neuropathic pain component (median score 6, IQR 2.0; mean score 5.6, SD \pm 1.4) and 19 patients with an unlikely NP component (median score 6, IQR 2.0; mean score 5.6, SD \pm 1.4). The PD-Q's final results classify patients into no NP, unclear and NP. In 28.1% of patients (9/32), results were unclear (median score 6, IQR 2.0; mean score 5.6, SD \pm 1.4); however a NP component might be present.



Figure 8.7. PD-Q scores in 32 leprosy patients with upper limb pain

Frequency of the PD-Q scores from 32 patients with upper limb pain, classified as non-neuropathic pain (8 patients) and neuropathic pain (24 patients).

A non-parametric analysis (Mann-Whitney two-sample statistic) between reported pain descriptors from patients classified as having non-NP and NP in upper limb using the PD-Q questionnaire was performed. The result showed that responses to questions 1 (burning sensations); question 2 (pins and needles); and question 6 (numbness) were significantly different between these two subgroups (p-value ≤ 0.05 ; more present in NP patients). When overall PD-Q questionnaire responses were considered in total points, there were also a difference between the two pain subgroups (p-value ≤ 0.001). When the analysis was repeated using binary coding for the PD-Q questionnaire responses in the upper limb instead of categorical coding, the findings were unchanged (Table 8.10).

Q1-7: Clinical relevant complaint §	Frequency of patients descriptor present				
	Non ND $(n=10)$ $(0/2)$	NIP $(n-22)$ (%)	D value		
	1001-10F(11-10)(70)	1NP(11-22)(70)	r-value		
Q1, burning sensation	3 (30.0%)	17 (77.3%)	0.02*		
Q2, prickling	7 (70.0%)	20 (90.9%)	0.01*		
Q3, allodynia	2 (20.0%)	4 (18.2%)	0.90		
Q4, electric shocks	6 (60.0%)	17 (77.3%)	0.31		
Q5, thermal	0 (00.0%)	5 (22.7%)	0.10		
Q6, numbness	1 (10.0%)	19 (86.4%)	0.001*		
Q7, pressure	6 (60.0%)	9 (40.9%)	0.31		

Table 8.10. A comparison of symptoms detected by PD-Q in pain groups (n=32)

 A comparison of pain symptoms detected by PD-Q questionnaire for the two pain subgroups, using chi-squared test. § Q1-7: Question 1 to 7 of PD-Q questionnaire. *P-value ≤0.05

The responses to DN4 questionnaire had agreement with PD-Q responses when identifying neuropathic pain in 21 of the 32 patients with upper limb pain (NP: n = 13; Non-NP: n = 8), yielding a 65.6% agreement between questionnaire final results (Table 8.11). In all discordant (11 patients), a neuropathic pain component was detected by DN4 questionnaire, but not with the PD-Q.

By using the PD-Q final classification as having no neuropathic pain, unclear or neuropathic pain; the responses to DN4 questionnaire had agreement with PD-Q responses in 13 patients (Non-NP: n = 8, NP: n = 5), which yielded a 40.6% agreement.

		DN4		Total
		NP	No NP	
PD-Q	NP	13	0	13
	No NP	11	8	19
	Total	24	8	32
		1		
	NP	5	0	5
PD-Q	Unclear	8	0	8
	No NP	11	8	19
	Total	24	8	32

Table 8.11. A comparison of NP detected by DN4 and PD-Q (n=32) [†]

[†]A comparison of NP in 32 patients with hands pain detected by DN4 and PD-Q questionnaires. Two classifications for PD-Q were used: non-NP and NP; and non-NP, unclear and NP

Patient's responses to DN4 and PD-Q

In the 36 patients with pain present at the time of interview, DN4 had been evaluated as very easy by 2 (6.5%) patients, easy 21 (67.7%), and fair 8 (25.8%). Similarly, PD-Q was evaluated as very easy by 3 (9.7%) patients, easy 16 (51.6%), and fair 12 (38.7%) (Figure 8.8). No statistical difference between the two groups.



Figure 8.8. Patients' responses for the DN4 and PD-Q questionnaires (n=36)

8.4 Clinical examination findings

Clinical examination included the following: assessment of skin; evaluation of enlargement and tenderness of main peripheral nerves affected by leprosy (greater auricular (GA), median, ulnar, radial cutaneous, ulnar branch, lateral popliteal, posterior tibial, and sural nerves); nerve function assessment; and disability assessment.

8.4.1 Skin assessment

Seventy patients (77.78%) had skin lesions at the time of interview, Table 8.12. Of these 70 patients, 38 (54.29%) had old skin lesions: fully treated in 32 patients, and partially treated in 6.

Skin lesion	Number (%)
No skin lesion	20 (22.22%)
Single skin lesion (SSL)	12 (13.33%)
2-5 skin lesions	28 (31.11%)
6-10 skin lesions	16 (17.78%
More than 10	14 (15.56%)

Table 8.12. Distribution of skin lesions in leprosy patients (n=90)

8.4.2 Evaluation of enlargement and tenderness of main peripheral nerves

For the 90 studied patients, 1620 peripheral nerves were assessed for enlargement and tenderness in both right and left body side. Nerve enlargement was common (90% of the patients had one or more enlarged nerves), and 21% had tenderness of nerves on palpation. The ulnar nerves were found to be both the most frequently enlarged and tender nerves, followed by the posterior tibial, lateral popliteal, and superficial peroneal nerves. The prevalence of palpable nerve enlargement and tenderness on palpation in each nerve is shown in Table 8.13.

8.4.3 Nerve functions assessment

Nerve function assessments: sensory and muscle strength examination were completed for the 90 patients. Nerve function impairment of less than 6 months duration (new NFI) was reported for 10 patients (11.11%). A further 36% of patients reported that their impairment had been present for longer than 6 months (old NFI). In both old and new NFI, sensory loss was more frequent than motor loss or mixed loss. Sensory impairment was found in 62 (78%) patients, and motor impairment in 47 patients (59%); the ulnar nerve was the most commonly affected nerve.

The frequency of abnormal tendon reflexes or absent joint position sense in each nerve in this cohort was very few. The percentage of absent reflexes in each nerve as follow: right biceps (1.11%), right triceps (1.11%), left supinator (1.11%), right knee jerk (1.11%), and left ankle (2.22%). Two patients (2.22%) had absent JPS in their right index finger, and other two (2.22%) in their right big toe. Impaired JPS or reflexes were not associated with the presence of neuropathic pain or depression. 4 - 6

The frequency of ulnar neuropathy (sensory and motor impairment) detected by the different tests is shown in Table 8.14. Of 29 patients with no clinical evidence of neuropathy, 15 patients had subclinical neuropathy. Of these 15 patients, 14 patients (93.33%) had impairment of two or more QST parameters and classified as having subclinical neuropathy (Figure 8.13 – A and B, page – 214). Pooling results for right and left dermatomes C8 of the ulnar nerves, where a response was recorded, the QST parameters were abnormal in 112/176 (63.63%) of participants (Figure 8.10 – A and B, page – 208). The most frequently affected were CDTs and WDTs (thermal QST parameters). CDTs were affected as least twice as often as WDTs in the ulnar nerves in the subclinical group. This difference was less pronounced on the neuropathy and pain

groups. A discrepancy was found between the frequencies of ulnar nerve impairment detected by MF and QST thermal tests. This was particularly pronounced in the subclinical neuropathy group. For neuropathy and pain groups, WDTs and CDTs were significantly more often abnormal than MF and VMT (e.g. 31% versus 89% for the neuropathy and pain group, p-value<0.001). Interestingly, vibration sense in leprosy ulnar neuropathy were less frequently affected (15%) than other modalities. Overall, on the ulnar nerve tests, the sensation conducted by small fibres were the most frequently impaired. This confirms that small fibre neuropathy usually precedes large fibre damage.

Table 8.15 shows the concordance between ulnar neuropathy diagnosed with monofilament testing and results of quantitative sensory testing (thermal, vibration and combined QST parameters (two or more abnormal parameters)) in 88 patients. Combining results for right and left ulnar nerves, monofilaments and QST testing are both detected abnormality in 94 of the 176 nerves (ulnar nerve impaired: n= 30; no ulnar nerve impairment: n= 64); yielding 83.9% agreement. Concordance between monofilament results and QST parameters was best for CDTs and WDT. Unlike thermal tests, the VDT results were not concordant with the monofilament results. The highest positive concordance was seen between monofilament results and combined QST. Combining impairment of any of the QST parameters in one variable improved agreement with the monofilament test, over that of individual QST parameters, but negative concordance was substantially higher. Up to 50% of nerves with a normal MF result had one or more abnormalities in quantitative sensory testing. Similar results were observed for the motor assessments.

8.4.4 Disability assessment

Disability was assessed using the WHO disability criteria, which define grade 0 as no loss of sensation or visible deformity, grade 1 as loss of sensation without visible deformity, and grade 2 as presence of visible deformity. 55.56% of participants had physical impairment (28.9% grade 1 and 26.7% grade 2). Impairment was associated with the hand (46%), followed by 37% associated with the feet and 1% associated with eyes (grade 2).

	Greater A		Ulnar		Media	n	Radial	l	Uln. b	ranch	LPN		Superf	icial	Post T	'ibial	Sural	
	Rt.	Lt.	Rt.	Lt.	Rt.	Lt.	Rt.	Lt.	Rt.	Lt.	Rt.	Lt.	Rt.	Lt.	Rt.	Lt.	Rt.	Lt.
Enlarged	9%	12%	64%	73%	26%	22%	40%	37%	11%	10%	37%	56%	44%	51%	56%	48%	24%	24%
Tender*	0	0	17%	12%	4%	5%	8%	6%	0	0	6%	8%	15%	17%	14%	19%	23%	18%

Table 8.13. Prevalence of palpable nerve enlargement and tenderness (n=90)

*As the tender nerves were always enlarged, the denominator is the enlarged nerves

Test	No neuropathy		Subclinical neuropathy		Neuropathy		Neuropathy and pain		
	Ulnar nerve		Ulnar nerve (n=15)				Ulnar nerve (n=21)*		
					Ulnar nerve (n=	=24)			
	Right	Left	Right	Left	Right	Left	Right	Left	
MF	0	0	0	0	10 (41.67%)	6 (25.00%)	6 (31.58%)	6 (31.58%)	
VMT	0	0	0	0	14 (58.33%)	13 (54.17%)	13 (68.42%)	11 (57.89%)	
WDT	0	0	5 (33.33%)	5 (33.33%)	17 (70.83%)	18 (75.00%)	17 (89.47%)	15 (78.95%)	
CDT	0	0	10 (66.67%)	7 (46.67%)	15 (62.50%)	15 (62.50%)	15 (78.95%)	14 (73.68%)	
VDT	0	0	0	0	3 (12.5%)	1 (4.17%)	3 (15.79%)	3 (15.79%)	

*of the 24 patient's classified as neuropathy and pain over ulnar nerve territory, 3 patients had neuropathy and pain over skin lesion.

Test*	Right Ulnar nerve (n = 88)		Left Ulnar nerve (n = 88)			
	MF impaired (n=18)	MF not impaired (n=70)	MF impaired (n=12)	MF not impaired (n=76)		
WDT impaired	94% (17)**	50% (35)	83% (10)	43% (33)		
CDT impaired	89% (16)	40% (28)	92% (11)	41% (31)		
Thermal impaired combined	100% (18)	63% (44)	92% (11)	68% (52)		
VDT impaired	11% (2)	6% (4)	17% (2)	3% (2)		
≥2 impaired QST parameters	100% (18)	56% (39)	100% (12)	57% (43)		

Table 8.15. Comparability of impairment by MF test and QST (n=88)

*MF = monofilament test, WDT = warm detection threshold, CDT = cold detection threshold, VDT = vibration perception threshold **Column % (number of nerves)

8.5 Quantitative Sensory Testing (QST) findings in leprosy patients

DFNS-QST database was used to transform QST raw scores into z-scores, and healthy controls data was used for statistical comparisons. The healthy controls, consisting of 52 age-matched participants, were recruited from the local community. The normative data from local population showed similar distribution to those held in DFNS database.

The results of the QST data analysis are presented as follows: the first part is the descriptive results, and comparison of the QST findings between all leprosy patients and healthy controls. The second part contains the somatosensory profiling, and the number of abnormal QST parameters in leprosy patients with no pain; followed by results for patients with pain. The last part describes the QST findings and comparisons of the leprosy patients with neuropathic pain.

8.5.1 QST observations in healthy control participants

From the 52 healthy volunteers investigated in this study, 102 locations in the upper limb (dermatome C8) and 8 locations in the lower limb (dermatome S1) were assessed. The measurements were analysed by z-score profiling (Chapter 6), and data was used for statistical comparisons.

8.5.2 QST observations in patients

For the 90 study patients, 27 990 QST data measurements were obtained from the affected and contralateral side and analysed by z-score sensory profiling.

8.5.3 Distribution of QST measures

Figure 8.9 illustrates distributions of QST data from the 52 healthy control subjects and 88 leprosy patients tested in the ulnar nerve territory (dermatome C8). DFNS-QST database was used to transform QST raw scores into z-scores, which adjusts for test site, gender and age. The horizontal axis (x-axis) demonstrates the QST parameters after z-transformation, and the vertical axis (y-axis) indicates the percentage of cases (patients or healthy controls). As paradoxical heat sensations and dynamic dynamic mechanical allodynia normally do not occur in healthy participants, z-transformation could not be calculated. Thus, data are shown as percentage of participants showing PHS and DMA. For healthy control subjects, QST measures fell within the normal range of the DFNS-QST references (Chapter 7). QST measures for leprosy patients demonstrated similar distribution shapes compared to healthy control subjects, but with larger standard deviations indicating diverse sensory findings. For thermal and mechanical detection thresholds (non-nociceptive parameters: CDT, WDT, TSL, MDT, VDT), there were significant leftward shifts, suggesting the presence of hypoesthesia. For pain thresholds (nociceptive parameters: CPT, HPT, PPT, MPT, MPS, WUR), there were slight leftward shifts, suggesting the presence of hypoalgesia. The high prevalence of hypoethesia and hypoalgesia led to difficulties in performing WUR, which was the most frequently missing QST parameter in the cohort (21%). Overall, abnormal findings for loss of function (30.5%) across all QST parameters in patients with leprosy were more frequently observed than gain of function phenomenon (1.7%).

For thermal detection threshold (CDT, WDT, TSL), only sensory loss signs (thermal hypoethesia) were detectable in the affected area. For pain thresholds, negative sensory signs (thermal hypoalgesia) also dominated in 7.9% of the patients for CPT and in 31.8% for HPT. Sensory gain signs (thermal hyperalgesia) were absent in affected area for both CPT and HPT. Patient's measures of MDT demonstrated a broader distribution and showed a leftward shift compared to the healthy controls (Figure 8.9 - J). Thus indicating that the negative sensory sings (mechanical hypoesthesia) were frequent in the affected area. VDT, WUR and PPT patient's data exhibited similar distribution to the control's data.

For dynamic mechanical allodynia (DMA) and paroxysmal heat sensation (PHS), which are pathological phenomena and normally do not occur in healthy subjects, the occurrence in leprosy patients was rare (range between 2.3% for DMA and 13.6% for PHS). DMA was present in 2.3% of patients, but mostly of very mild intensity.

PHS in the affected area was reported in twelve patients, but it is not clear whether this phenomenon is part of sensory gain. In these twelve patients, PHS was reported once in three patients, twice in one patient, and three times in eight patients.



Figure 8.9. Distribution of the QST parameters after z-transformation

Distribution of the QST data using DFNS reference data. Affected ulnar nerve territory of all leprosy patients (n=88) (red circles/ solid line) in comparison with the controls (n=50) (green square/solid line). The y-axis indicates the percentage of cases (patients or controls). For PHS and DMA % are plotted versus original data: occurrences of PHS (0–3), log numerical ratings scale for DMA (0–100)

8.5.3.1 Frequencies of abnormal QST values

Of the 90 patients with leprosy, 86 patients had completed a full DFNS QST measures in the upper limbs, 4 patients in the lower limbs and 2 patients in both upper and lower limbs. The frequencies of abnormal QST values for each parameter was identified by absolute (outside 95% CI of DFNS reference data) and relative (side-to-side differences) sensory abnormalities (Figure 8.10). In these 88 patients tested in the ulnar nerve territories (dermatome C8), the abnormal sensory loss was highly prevalent, and significant rates were found primarily for non-nociceptive thresholds in approximately two third of the patients (60%) for thermal detection; and in about half of the patients for mechanical detection thresholds (46%), but rarely for vibration thresholds (8%). Almost no one had sensory gains for the non-nociceptive parameters.

For nociceptive parameters (pain), sensory loss (hypoalgesia) was frequent (range 7.9% - 42%), but sensory gain was rare (18%). Of these detected abnormalities, about one third of the patients with relative sensory loss were identified by side-to-side comparison (for different parameters between 2.3% – 15.9% additional patients), but again almost none with relative sensory gain for the nociceptive parameters. Remarkably, cold pain hypoalgesia was only detectable by side-to side comparison. Hypoalgesia was most frequently detected for pinprick, followed by heat, blunt pressure, and cold. Paroxysmal heat sensation was about as frequent as Wind up ratio. Dynamic mechanical allodynia was rare.



Figure 8.10. Frequencies of abnormal QST measures

Frequencies of abnormal QST measures (values outside the 95% CI of the reference data base) in the affected ulnar nerve site. Shaded areas of the bars illustrate percentage of abnormal findings according to absolute reference data, open areas of bars according to abnormal side-to-side difference. The y-axis shows percentage of patients (n = 88), with positive sensory signs plotted upwards and negative sensory signs plotted downwards. (A) QST parameters that display loss or gain of function (B) Parameters, which are absent in normal subjects and can only present as sensory gain as defined by DFNS.

8.5.3.2 Differences between patients with leprosy and healthy controls

There were significant differences between the patients with leprosy and healthy controls in all QST thermal measures (mean and standard deviation of the DFNS z-score); CDT, WDT, TSL, CPT and HPT (p-value <0.0001). Patients had increased cold and warm thresholds; and were less sensitive to heat, and cold pain stimuli. For QST mechanical measures, there were significant differences in MDT, MPT, and MPS (p-value <0.0001). No significant differences in VDT, WUR, or PPT were found between the two groups (p-value >0.5, p-value >0.6, and p-value >0.5, respectively) (Table 8.16, Figure 8.11 and Figure 8.12).

QST	Controls (n=	=50)	Leprosy p	Leprosy patients (n=88)			
Parameter					P value		
	Mean	SD	Mean	SD			
CDT (°C)	-0.12	1.8	-2.94	0.89	< 0.0001		
	0.10						
WDT (°C)	-0.43	0.74	-2.84	1.52	< 0.0001		
	0.75	0.57	2.92	1.20	<0.0001		
15L (°C)	-0.75	0.57	-2.02	1.39	<0.0001		
CPT (°C)	1 24	0.48	0.30	0.65	<0.0001		
	1.21	0.10	0.50	0.05	<0.0001		
HPT (°C)	0.55	0.53	-0.87	1.03	<0.0001		
MDT	1.08	0.66	-1.69	2.86	< 0.0001		
МРТ	0.27	0.38	-0.72	1.06	< 0.0001		
MPS	0.17	0.55	-0.43	1.21	< 0.0001		
WUR	0.62	1.02	0.59	1.44	0.45		
	0.40	0.20	0.47	0.00	0.40		
	0.48	0.28	0.17	0.98	0.49		
ррт	1.07	0.68	1.05	0.90	0.62		
	-1.07	0.00	-1.05	0.70	0.02		

 Table 8.16. A comparison of QST measures between patients and controls

 \uparrow Means and standard deviation (SD) of QST parameters of leprosy patients tested on the ulnar area (C8) compared to controls tested on the same site (n=50). QST data are shown as mean for untransformed data (HPT, VDT) and transformed mean for lognormally distributed data.



Figure 8.11. Thermal QST measures for patients (n=88) and controls (n=50)



Figure 8.12. Mechanical QST measures for patients (n=88) and controls (n=50)

8.5.4 Somatosensory profiles of leprosy patients

8.5.4.1 Sensory profile and number of abnormal findings

Z-score sensory profiles of all leprosy patients compared to control

The QST sensory profiles for ulnar nerve territory (maximum pain area on dermatome C8) and by group (HC, patients with no pain, and patients with pain) shown as z-scores are presented in Figure 8.13 and Table 8.17. For the current study, the QST parameters are grouped into thermal and mechanical parameters; and to allow for easy visual comparison, the z-scores sensory profile are also shown for all patients by each group independently.

8.5.4.2 Distribution of QST measures in patients without pain

The z-score DFNS-QST sensory profiles for all thermal and mechanical parameters in patients with leprosy and no pain in the ulnar nerve territories (dermatome C8) are illustrated in Figure 8.13 and Table 8.17. For patients with no clinical evidence of neuropathy (Group 1), the mean values of all QST parameters were within the 95% confidence interval of the DFNS and local Indian healthy controls references (Figure 8.13 - A). Patients with no clinical evidence of neuropathy based on MFs and/ or MRC scale, but who showed abnormal nerve conduction study (NCS) or thermal testing were classified as "Subclinical neuropathy".

Of the 29 patients with no clinical evidence of neuropathy, 14 patients were identified as having subclinical neuropathy using the thermal testing compared to 15 patients using NCS (data for NCS were not presented), indicating that QST measures can differentiate neuropathy from no neuropathy (Figure 8.13 - A and B).

Patients with subclinical neuropathy (Group 2) and clinical evidence of neuropathy (Group 3) had z-scores beyond the 95% confidence interval of the DFNS and local Indian healthy controls references. Their sensory profiles were characterised predominately by a loss of function, indicated by increased thermal and mechanical thresholds (Table 8.17). In these two groups, a thermal loss of function was demonstrated for non-nociceptive CDT, WDT, TSL; and nociceptive parameters CPT, and HPT in the ulnar side compared to healthy controls (Figure 8.13 - B and C). For mechanical QST parameters, a loss of function was demonstrated for the non-nociceptive parameters MPT; and for the nociceptive parameters MPT, and MPS.

Although MPT, MPS, VDT, WUR, and PPT in the subclinical group; and VDT, WUR, and PPT in the neuropathy group, were abnormal in the affected side (Table 8.17) they did not reach statistical significance compared to healthy controls data (p-values >0.05). In both groups, there was no evidence of sensory gain.

The thermal and mechanical frequencies of z-score values outside the 95% confidence interval of the DFNS and HC group which indicating a loss of function (<- 1.96), for patients with subclinical neuropathy, were as follows: CDT (53.3%), WDT (40.0%), TSL (56.6%), CPT (6.7%), HPT (13.3%), MDT (26.7%), MPT (10.0%), MPS (30.0%), and PPT (30.0%) (Table 8.22). For the clinical evidence of neuropathy group, the frequencies of the abnormal values were as follows: CDT (53.3%), WDT (40.0%), TSL (56.6%), CPT (6.7%), HPT (13.3%), MDT (26.7%), MPT (10.0%), MDT (40.0%), TSL (56.6%), CPT (6.7%), HPT (13.3%), MDT (26.7%), MPT (10.0%), MPS (30.0%), and PPT (30.0%) (Table 8.22). WUR was not consistently present in any of the affected and/ or contralateral ulnar side. No patients had DMA in any of the affected and/ or contralateral ulnar side. PHS in the affected area was frequently reported in nine patients. In these nine patients, PHS was reported once in five patients, twice in two patients and three times in two patients (Table 8.22).





B) Sub-clinical neuropathy







D) Clinical evidence of neuropathy and pain



Figure 8.13. Leprosy sensory profiling

The z-score sensory profiles are shown of (A) patients with no clinical evidence of neuropathy (n=15), (B) patients with sub-clinical neuropathy (n=14), (C) patients with clinical evidence of neuropathy (n=25), and (D) patients with pain (n=32)

QST	Parameter †	Groups						
		Healthy	Healthy Patient (n=88) control					
		(n=50)	Group 1	Group 2	Group 3	Group 4		
	CDT (°C)	-0.12 (0.89)	-0.54 (0.88)	-2.13 (1.01)*	-3.13 (2.06)*	-3.24 (1.99)*		
leters	WDT (°C)	-0.43 (0.74)	-0.51 (0.75)	-1.80 (1.20)*	-3.07 (1.59)*	-3.23 (1.41)*		
param	TSL (°C)	-0.75 (0.58)	-0.77 (0.67)	-2.08 (0.77)*	-2.78 (1.55)*	-3.27 (1.34)*		
al QST	CPT (°C)	1.24 (0.48)	0.81 (0.77)	0.58 (0.56)*	0.13 (0.69)*	0.30 (0.62)*		
Therm	HPT (°C)	0.55 (0.53)	0.13 (0.84)	-0.33 (1.11)*	-0.99 (0.97)*	-1.08 (0.93)*		
	MDT(mN)	1.08 (0.66)	0.80 (0.85)	0.10 (1.73)*	-2.44 (3.34)*	-2.83 (3.10)*		
rs	MPT (mN)	0.27 (0.38)	0.41 (0.60)	-0.11 (0.80)	-1.00 (1.07)*	-0.80 (1.05)*		
nical QST paramete	MPS (0-100)	0.17 (0.56)	0.34 (0.68)	0.33 (1.12)	-0.68 (1.16)*	-0.63 (1.14)*		
	VDT (x/8)	0.48 (0.28)	0.34 (0.59)	0.41 (0.47)	0.20 (0.89)	-0.24 (2.04)*		
	WUR (ratio)	0.62 (1.02)	0.73 (1.07)	0.78 (1.29)	0.84 (1.75)	0.25 (1.19)		
Mecha	PPT (kPa)	-1.07 (0.68)	-1.19 (0.68)	-1.31 (0.78)	-1.25 (0.90)	-0.71 (0.90)		

Table 8.17. Descriptive statistics for QST parameters †

[†] Means and standard deviation (SD) of QST parameters of healthy controls (HC), and leprosy groups (Group 1, 2, 3, and 4) in the maximum pain area (C8). QST data are shown as mean for untransformed data (HPT, VDT) and transformed mean for log-normally distributed data. *p-value <0.05.
WDT_ Healthy vs. Groups



В

Α

CDT_ Healthy vs. Groups

Controls Group 1 Group 2 Group 3

Figure 8.14. Thermal QST measures for leprosy patients with no pain (n=54)



Figure 8.15. Mechanical QST measures for leprosy patients with no pain (n=54)

8.5.4.3 Distribution of QST measures in patients with pain

The z-score DFNS-QST sensory profiles for thermal and mechanical parameters in patients with leprosy and pain (Group 4) are illustrated in Figure 8.13 - D (above), Figure 8.16 and Figure 8.22 (below); and Table 8.18. Patients with leprosy and pain were further sub-grouped according to the DN4 score and neuropathy which was classified clinically by using MFs and/ or MRC scale (but no NCS), as follows: no neuropathy and DN4 <4 (2 patients); no neuropathy and DN4 \geq 4 (2 patients); neuropathy and DN4 <4 (2 patients); and neuropathy and DN4 \geq 4 (22 patients) (Table 8.18). In all these subgroups patients had z-scores beyond the 95% confidence interval of the normal reference values for CDT, WDT, and TSL. The dominant sensory characteristics was a loss of function in the C8 area of the ulnar nerve territory. For mechanical QST parameters, a loss of function was demonstrated for the MDT and WUR in the no neuropathy subgroup (p-value <0.05) (Table 8.18); MDT, and MPS in patients with neuropathy and DN4 \geq 4.

In addition, few leprosy patients with any type of pain demonstrated signs of a gain of sensory function indicated by the presence of WUR and DMA. The former was present in eight (12.5%) patients. Two (3.1%) patient demonstrated DMA in the QST tested site (dermatome C8), but it was present nine (41.1%) patients along the ulnar nerve territory. PHS in the affected area was frequently reported in eleven (17.2%) patients. In these four patients, PHS was reported once in three patients, and three times in eight patients (Table 8.22).



Figure 8.16. Sensory profiling for pain patients (n=32) [†]

The z-score sensory profiles of the ulnar tested sites are shown of patients with no clinical evidence of neuropathy and DN4 <4 (Green), or DN4 \geq 4 (Blue); clinical evidence of neuropathy and DN4 <4 (Pink), or DN4 \geq 4 (Red); and controls (Black)

QST	Healthy	Patients with pain in upper limbs (n=32)						
parameter	controls							
	(n=50)	No neuropathy		Neuropathy				
		DN4<4	DN4≥ (n=2)	DN4<4	DN4≥			
		(n=2)		(n=6)	(n=22)			
CDT (°C)	-0.12 (0.89)	-2.26 (1.38)*	-2.22 (2.59)*	-2.84 (2.11)*	-3.13 (2.08)*			
WDT (°C)	-0.43 (0.74)	-2.97 (1.29)*	-2.62 (1.57)*	-2.31 (1.97)*	-3.15 (1.39)*			
TSL (°C)	-0.75 (0.58)	-2.61 (0.80)*	-2.29 (1.50)*	-2.36 (2.05)*	-3.24 (1.27)*			
CPT (°C)	1.24 (0.48)	0.43 (0.70)	0.94 (0.66)	0.28 (0.71)*	0.35 (0.64)*			
HPT (°C)	0.55 (0.53)	-1.27 (0.87)*	-0.73 (1.10)	-0.69 (1.12)*	-0.92 (1.03)*			
MDT(mN)	1.08 (0.66)	-0.80 (1.06)*	-1.34 (0.04)	-2.38 (3.47)*	-3.13 (2.08)*			
MPS	0.17 (0.56)	-0.13 (0.78)	-0.26 (0.79)	-0.82 (0.57)*	-0.57 (1.27)*			
MPT (mN)	0.27 (0.38)	-0.18 (0.47)	0.33 (0.22)	-0.45 (0.72)	-2.94 (3.17)*			
VDT (x/8)	0.48 (0.28)	0.33 (0.57)	0.64 (0.00)	-0.41 (1.64)	-0.28 (2.18)			
WUR	0.62 (1.02)	-1.22 (0.10)*	0.27 (1.28)	1.00 (1.05)	0.27 (1.15)			
PPT (kPa)	-1.07 (0.68)	-0.65 (0.11)	-0.98 (0.15)	-0.51 (0.74)	-0.76 (0.97)			

Table 8.18. QST parameters for leprosy patients with pain \uparrow

↑ Means and standard deviation (SD) of QST parameters of leprosy patients with pain in the ulnar area (C8). QST data are shown as mean for untransformed data (HPT, VDT) and transformed mean for log-normally distributed data. *p value <0.05

8.5.4.4 Distribution of QST measures in patients with and without pain

Z-scores of sensory profiles of the patients with pain and without pain in the upper limbs are illustrated for each group (Table 8.19 and Figure 8.17). Healthy control participants are represented by a z-score of "zero". In both subgroups all QST parameters fell outside the 95% confidence interval of our healthy controls data (i.e. zscore >-1.96 or <1.96 standard deviation). QST measures cannot be used to differentiate leprosy patients with and without pain.

In patients with or without pain, thermal and mechanical detection thresholds; pain thresholds were significantly reduced on the affected side compared to the healthy control subjects (CDT: p < 0.0001, WDT: p < 0.0001, TSL: p < 0.0001, CPT: p < 0.001, HPT: p < 0.001, and MDT: p < 0.0001) (Figure 8.17). Other QST parameters (VDT, WUR and PPT) were not statistically significant compared to healthy control subjects in affected side. Although patients with and without pain both had QST parameters beyond 95% confidence interval of the references, these measures cannot be used to differentiate leprosy patients with and without pain (Table 8.19). In this study, QST results revealed that the number of sensory abnormalities did not differ between patients with and without pain.

QST	Patients with	out Pain (n=54)	Patients w	ith pain (n=32)	P-value
parameter					
-	Mean	SD	Mean	SD	
CDT (°C)	-3.74	2.13	-3.29	2.10	0.35
WDT (°C)	-3.62	1.36	-3.31	1.44	0.59
TSL(°C)	-3.10	1.66	-3.23	1.44	0.91
CPT (°C)	-0.10	0.49	0.34	0.66	0.35
HPT (°C)	-1.22	0.94	-1.02	1.06	0.81
MDT(mN)	-3.27	3.21	-2.76	3.36	0.37
MPT (mN)	-1.43	1.06	-0.92	1.13	0.34
MPS (0-100)	-1.08	1.16	-0.81	1.01	0.74
VDT (x/8)	0.08	1.03	-0.19	1.38	0.74
WUR (ratio)	1.10	1.72	0.17	1.26	0.08
PPT (kPa)	-1.34	0.10	-0.74	0.95	0.27

Table 8.19. QST parameters in leprosy patients with and without pain \uparrow

[†] Means and standard deviation (SD) of QST parameters are shown of leprosy patients with and without pain in the ulnar area (C8). QST data are shown as mean for untransformed data (HPT, VDT) and transformed mean for log-normally distributed data





Figure 8.17. QST findings for leprosy patients (n=88)and controls (n=50)

8.5.4.5 QST findings in leprosy patients with neuropathic pain

Sensory phenotypes

Patients with leprosy NP demonstrated z-scores beyond the 95% confidence interval of the normal reference values in almost all QST parameters, except for VDT, PPT and WUR, in the maximum pain area over the ulnar nerve territory (dermatome C8) (Figure 8.18). Data analysis on the individual level for frequencies of abnormal values gave a similar findings (Figure 8.19 and Table 8.20). Two thirds of these findings were identified by direct comparison to reference data, which is defined as an absolute abnormality; and about one third by side-to-side difference i.e. relative abnormality (if a patient's values were abnormal in both tests, only abnormality with respect to absolute reference data was counted). Figure 8.19 shows the percentage of abnormal values in the leprosy patients presented with NP.

Of the twenty four patients with NP components in the ulnar nerve territory, three patients had NP in their skin lesions which were not tested as part of the QST protocol. In the remaining 21 patients, 84-94% had abnormal sensory loss and significant rates were found primarily for non-noxious detection for thermal detection transmitted by small nerve fibres (CDT (84.2%), WDT (89.5%), TSL (94.7%): p-value <0.0001); and in about one quarter of the patients for mechanical detection conducted by large nerve fibres (MDT (73.7%): p-value <0.0001). Remarkably, sensory loss was also frequent for pain parameters, in particular for pinprick (MPT (85.7%), MPS (78.9%): p-value <0.0001); and heat pain (HPT (52.6%): p-value <0.0001) (Table 8.18). Although patients had abnormal PPT, WUR and VDT in the affected side (VDT (26.3%), PPT (15.8%), and WUR (15.8%)) (Table 8.19 and Figure 8.17 - B), they were not significantly different from healthy controls data (PPT, WUR and VDT: p-values >0.05).

The sensory gain in patients with leprosy NP was not consistently present in the tested site (dermatome C8). The most prevalent findings of sensory gain was the presence of WUR in three (15.8%) patients, and in nine (47.4%) patients the thresholds was not detected as the patients did not feel the stimulus. Only one (5.3%) patient had pain to light touch (DMA) demonstrated in the QST tested site (dermatome C8). In addition, DMA was present in six (31.6%) patients along the ulnar nerve distribution, but not at the QST tested site. No abnormal hypersensitivity were found in all other pain

parameters. PHS in the affected area was frequently reported three times in 3 (15.8%) patients, but it is not clear whether this phenomenon is part of sensory gain.



Figure 8.18. Leprosy neuropathic pain sensory profiling

QST profiles QST profile of 24 patients with clinical evidence of neuropathy and DN4≥4 in ulnar nerve territory (dermatome C8) (red symbols), using a QST test protocol according to the German Research Network on Neuropathic Pain (DFNS). Data are presented as z-scores, using the following expression: Z-value = (Value patients – Mean control) / SD control. Data of healthy control patients are represented by a z-score of "0"; patients' data are presented as positive or negative z-values. The grey area represents the confidence interval of healthy control patients. Values are defined as pathological when deviating more than two SD from the respective control sample (age- and sex-matched). This profile shows signs of sensory loss detected by increased thermal detection thresholds (CDT, WDT) and increased mechanical detection thresholds (MDT). No signs of sensory gain are found by pinprick hyperalgesia (MPT, MPS). QST quantitative sensory testing; SD standard deviation; CDT cold detection threshold; WDT warm detection threshold; TSL thermal sensory limen; CPT cold pain threshold; HPT heat pain threshold; PPT pressure pain threshold; MPT mechanical pain threshold; MPS mechanical pain sensitivity; WUR windup ratio; MDT mechanical detection threshold; VDT vibration detection threshold; NRS numerical rating scale; DMA dynamic mechanical allodynia; PHS paradoxical heat sensation.

The abnormal values from the QST testing were categorised with respect to loss (L) and gain (G) of sensation from reference data taking age and sex into account (Magerl et al., 2010). Loss of sensation was further categorised to L0: no loss, L1: thermal hypoaesthesia, L2: mechanical hypoaesthesia, L3: thermal and mechanical hypoaesthesia; and gain of sensations was G0: no hyperalgesia, G1: thermal hyperalgesia, G2: mechanical hyperalgesia, G3 thermal and mechanical hyperalgesia (Table 8.20 and Table 8.22). This defines the abnormal sensitization of unmyelinated cutaneous nociceptors (irritable nociceptors) phenotype as L0G1, L0G2, L0G3, L2G1, L2G2, and L2G3; and the abnormal sensation of complete deafferentation of both large and small diameter fibres "non-irritable nociceptor" phenotype as any combination including L1 or L3, and L0G0 and L2G (Fields et al., 1998). The sensory phenotypes of patients with leprosy neuropathic pain was characterised predominately by a loss of function, indicated by increased thermal and mechanical threshold in the affected areas. The most prevalent combinations characterising most of the patients were sensory loss for both thermal and mechanical stimuli, combined with no sensory gain (Table 8.20). Only 4/21 patients had some form of abnormal mechanical pain (19%). Thus mechanical hyperalgesia was very rare and no thermal hyperalgesia. Mechanical hyperalgesia, if present, was usually accompanied by thermal and mechanical sensory loss.

Based on the QST findings patients were grouped as follows: patients with pain without hyperalgesia or allpdynia "non-irritable nociceptor phenotype". In this group, which including most of the patients, profound loss of small and large diameter fibre functions was documented. The second group had pain associated with small fibre deafferenation. In these patients, which represent the minority of the study cohort, pain and temperature sensation were profoundly impaired but allodynia was present "irritable nociceptor phenotype".



Figure 8.19. Percentages of abnormal QST values in NP patients

Percentages of abnormal QST values in 21 patients with NP in C8. Shaded areas of the bars illustrate percentage of abnormal findings according to absolute reference, open areas of bars according to abnormal side-to-side difference. (A) Parameters that display loss or gain of function (B) Parameters, which are absent in normal subjects and can only present as sensory gain as defined by DFNS.

Loss	Gain (pain)				
(detection)					
	No gain (G0)	Thermal (G1)	Mechanical (G2)	Both (G3)	All
No loss (L0)	- (0%)	- (0%)	- (0%)	- (0%)	- (0%)
Thermal (L1)	- (0%)	- (0%)	- (0%)	- (0%)	- (0%)
Mechanical (L2)	- (0%)	- (0%)	- (0%)	- (0%)	- (0%)
Both (L3)	15 (86%)	- (0%)	4 (19%)	- (0%)	21 (100%)
All	15 (86%)	- (0%)	4 (19%)	- (0%)	21 (100%)

Table 8.20. Frequency of abnormal values in NP (n=21) †

↑ Frequency of different combinations of abnormal values in 21 patients with NP in dermatome C8. L0: no loss, L1: thermal hypoaesthesia, L2: mechanical hypoaesthesia, L3: thermal and mechanical hypoaesthesia; G0: no hyperalgesia, G1: thermal hyperalgesia, G2: mechanical hyperalgesia, G3 thermal and mechanical hyperalgesia.

8.5.4.6 Comparison of sensory profiles between groups

Side-to-side comparison of QST sensory profiles (pain group)

Sensory profiles of the symptomatic and asymptomatic upper arms are illustrated for each pain subgroup (non-NP subgroup Figure 8.20 and NP subgroup Figure 8.21) in the maximum pain area. Healthy control subjects are represented by a z-score of "zero". In both subgroups QST parameters fell outside the 95% confidence interval of our healthy controls data. Compared to asymptomatic side, patients with painful neuropathy have lower thermal detection threshold and pain threshold (CPT, HPT). Remarkably, in these patients, the responses to QST stimuli can identify abnormalities in somatosensory system, but it cannot be used to differentiate leprosy patients with and without pain.

Patients with leprosy non-neuropathic pain

In patients with non-NP in upper limbs, all thermal detection thresholds; and pain parameters, in particular for heat, cold and pinprick pain thresholds were significantly reduced on the symptomatic side compared to the asymptomatic side in the tested area (p-value < 0.05) (Figure 8.20).

Patients with leprosy neuropathic pain

Of the 24 patients with NP in upper limbs, 4 patients had NP in the skin lesions, one of them located in the tested site (dermatome C8). So, 21 patients with maximum pain in the dermatome C8 were included in this comparison. Of these 21 patients, six had bilateral pain and the remaining 15 had unilateral pain. In these 21 patients, all thermal detection thresholds; and pain parameters, in particular for heat, cold and pinprick pain thresholds were significantly reduced on the symptomatic side compared to the asymptomatic side in the tested area (p-value <0.05) (Fig). Side-to-side comparison of all other QST parameters (MDT, WUR, VDT and PPT) were not significant.

Figure 8.21 -B demonstrates the percentages of the occurrence of PHS (0-3) and DMA (log numerical rating scale 0-100). The reports of both were infrequent. One patient with NP demonstrated DMA, but none in the asymptomatic side. PHS was reported by one patient once; and by four patients three times on the symptomatic side. Two patients reported PHS three times on the asymptomatic side.



Figure 8.20. Somatosensory profiles in Non-NP patients

Somatosensory profiles (A), and occurrence of DMA and PHS (B) of the symptomatic (empty circle symbol "red") and asymptomatic (empty square symbol "black") side in 8 patients with leprosy non-neuropathic pain in ulnar nerve territory. Error bars indicate the standard error of measurement. Healthy control subject are represented by a z-score of "zero"



Figure 8.21. Somatosensory profiles in NP patients

Somatosensory profiles (A), and occurrence of DMA and PHS (B) of the symptomatic (empty circle symbol "red") and asymptomatic (empty square symbol "black") side in 21 patients with leprosy NP in ulnar nerve territory. Error bars indicate the standard error of measurement. Healthy control subject are represented by a z-score of "zero"



A. Thermal measures for pain groups and controls

B. Mechanical measures for pain groups and controls



Figure 8.22. QST findings for patients with pain (n=32) and controls (n=50)

QST parameter	Maximum pain area over ulnar nerve territory (C8)								
	Neuropathic pain			Non-neuropathic pain					
	Asymptomatic	Symptomatic	P-value	Asymptomatic	Symptomatic	P-value			
CDT (°C)	-2.05 (1.89)	-3.70 (2.07)	0.018	-1.31 (0.98)	-3.23 (1.93)	0.088			
WDT (°C)	-2.29 (1.23)	-3.57 (1.28)	0.001	-0.88 (0.77)	-3.15 (1.66)	0.024			
TSL (°C)	-2.44 (1.24)	-3.58 (1.24)	0.009	-0.95 (0.80)	-3.02 (1.69)	0.039			
CPT (°C)	0.54 (0.63)	0.15 (0.54)	0.052	0.83 (0.67)	0.12 (0.60)	0.075			
НРТ (°С)	-0.38 (1.01)	-1.30 (0.85)	0.005	0.28 (0.84)	-1.31 (0.76)	0.005			
MDT (mN)	-2.04 (2.61)	-3.78 (3.21)	0.092	0.57 (0.34)	-2.35 (2.97)	0.079			
MPT (mN)	-0.17 (1.02)	-1.18 (1.13)	0.009	0.08 (0.59)	55 (0.61)	0.102			
MPS (NRS 0-10)	0.23 (1.34)	-0.96 (1.09)	0.006	-0.34 (0.85)	-0.74 (0.62)	0.350			
WUR (ratio)	0.28 (1.14)	0.15 (1.28)	0.780	1.44 (1.74)	-0.06 (0.97)	0.058			
VDT (x/8)	0.46 (0.43)	-0.70 (2.67)	0.117	-0.65 (1.47)	-0.01 (1.46)	0.470			
PPT (kPa)	-0.48	-0.83	0.331	-0.69 (0.30)	-0.50 (0.74)	0.612			

Table 8.21. Comparison between symptomatic and asymptomatic sides (n=21) [†]

[†] Comparison between symptomatic and asymptomatic sides of 21 leprosy patients with neuropathic pain in the upper limbs using the QST parameters. Data are shown as mean for untransformed data (HPT, VDT) and retransformed mean for log-normally distributed data

		Patients without pain				Patients with pain						
QST	НС		Group 1		Group 2	Group 2 Group3			Non-NP		NP	
Parameter	n=52		n=14		n=30 (two	sites)	n=48 (two	sites)	n=10 (test	site)	n=22 (test	site)
	Gain	Loss	Gain	Loss	Gain	Loss	Gain	Loss	Gain	Loss	Gain	Loss
CDT	0	0	0	0	0	16	0	33	0	5	0	16
WDT	0	0	0	0	0	12	0	38	0	6	0	17
TSL	0	0	0	0	0	17	0	34	0	6	0	18
СРТ	0	0	0	0	0	2	0	4	0	1	0	2
НРТ	0	0	0	0	0	4	0	17	0	3	0	10
MDT	0	0	0	1	0	8	0	23	0	2	0	15
MPT	0	0	0	0	0	3	0	22	0	2	0	12
MPS	0	0	0	0	0	9	0	19	0	1	0	14
WUR	0	0	4	0	8	0	13	0	0	1	0	3
VDT	0	0	0	0	0	0	0	3	0	1	0	5
РРТ	0	0	0	0	0	0	0	13	0	1	0	3
DMA	0	0	0	0	0	0	0	0	1	0	1	0
PHS	0		0		1		9		1		3	

Table 8.22. Distribution of abnormal findings in study population (n=142) \uparrow

↑ Number of individuals within each group with z-score values outside the 95% confidence interval of healthy control participants (\pm 1.96 SD). HC: Healthy control participants; Group 1: No evidence of neuropathy and no pain; Group 2: Subclinical neuropathy and no pain; Group 3: Clinical evidence of neuropathy and no pain; Non-NP: Inflammotory/ nociceptive pain; NP: Clinical evidence of neuropathy and pain (DN4≥4). Gain: Number of patients with positive individual z-score values, indicating an increased sensitivity compared to normative data (>+ 1.96 standard deviation). Loss: Number of patients with negative individual z-score values, indicating a decreased sensitivity compared to normative data (>+ 1.96 standard deviation). As no DMA occurred in healthy subjects, z-score values could not be calculated. Data are shown as absolute number of participants showing DMA.

8.6 Pain intensity and quality assessment

Patients with pain were asked to rate their current pain and estimate a level for their worst and average pain in the previous week using the NRS and BPI. All patients with pain (36) completed pain questionnaire. The individual scores in total and scores within each pain subgroup are shown in Figure 8.23, Table 8.23, and Table 8.24.

Figure 8.23 shows the distribution of pain scores (worst, average and current pain). The worst pain scores were not normally distributed, and the Shapiro-Wilk test was significant (W = 0.94, p-value = 0.03). Kurtosis was 1.81 and skewness was 0.74, which confirms that worst pain score is not a form of normal distribution, as in a normal distribution, both values should be zero (Kim, 2013).

The mean of current pain scores was slightly worse at 5.9 than average pain, and the median was six (interquartile range from 3 - 10). The mean of the worst pain intensity experienced by patients in the last week was high at 7.5 out of 10 (maximum score), and the median was eight (interquartile range from 4 - 1). The mean of the average pain intensity reported by patients in the previous week was relatively high at 5.3. Descriptive statistics for pain intensity measures are presented in Table 8.23.

Table 8.24 and Figure 8.24 display the scores for the BPI severity measure. Mean and median scores are displayed as well as how the scores were distributed between the neuropathic and non-NP groups. The two groups were compared according to pain intensity. The Mann-Whitney U test (for worst pain) and mean comparison test (for average and current pain) were conducted. None of the test results showed a significant difference (p-values: 0.09, 0.85, and 0.5 respectively).

The proportions of men and women who reported pain were quite similar (40% of men, 38% of women) (p-value =0.8). Women were slightly more likely than men to report a pain score of 4 or more, indicative of pain severity. The mean pain value for women was 7.4 and for men was 7.6 (p-value=0.06). No differences in mean score for men and women who reported pain interference problems. The mean interference value for men was 37 and for women was 36.7 (p-value=0.9).



Figure 8.23. Distribution of pain scores (n=36)

Pain measures	Mean	SD	95% CI	Median	Interquartile range
NRS current pain	5.97	2.47	5–7	6	3-10
NRS worst pain	7.50	2.35	7–8	8	4-10
NRS average pain	5.31	1.94	5-6	5	3-8

Table 8.23. Descriptive statistics for pain intensity measures (n=36)

Table 8.24. Pain intensity between pain groups (n=36)

Variable	Non-NP group		NP group		
	Mean (SD)	Median (IQ)	Mean (SD)	Median (IQ)	
NRS current pain	5.5 (2.42)	5.5 (5–7)	6.15 (2.51)	6 (5-8)	
NRS worst pain	8.5 (2.01)	9.5 (7–10)	7.12 (2.01)	8 (5-9)	
NRS average pain	2.49 (2.01)	5 (3–7)	5.35 (2.01)	5 (4-6)	



Figure 8.24. Pain intensity in NP and non-NP sub-groups (n=36)

8.7 Evaluation of psychological co-morbidity and HRQoL

8.7.1 Health related quality of life

8.7.1.1 Pain interference with daily life

The BPI is designated to evaluate pain interference over a variety of domains over the last 24 hours using a standard 0 - 10 Likert scale (0 = Does not interfere versus 10 = completely interferes). Patients were asked to rate how much their pain was interfering with aspects of daily life in the last 24 hours. Of the 36 patients with leprosy and pain, 35 (97%) patients completed the BPI related questions (Table 8.25).

Pain interference levels were moderate-to-severe, with the mean ranging from 4.1 to 6.0 across the selected daily life aspects. The collective patient group had a total interference score of more than four, which may be considered to be a high level of interference (Cleeland, 2009). Pain substantially interfered (≥ 4 on 0 – 10 scales) with normal work (mean 6.06, SD 3.11), sleep (mean 5.94, SD 3.30), and mood (mean 5.71, SD 3.22). Overall, patients reported variable levels of interference per domain; however, the highest levels of interference were observed in sleep, work, and mood. The lowest levels of interference were observed in sleep, work, and mood. The lowest levels of interference were observed in walking ability (Figure 8.25). The former three vital aspects of daily life have been found to be affected due to leprosy patients feeling pain.

Pain interference	Mean	SD*	95% CI**	Median	IQ range
General activity	5.23	2.81	46	5	4-7
Mood	5.71	3.22	7–8	5	5–7
Walking ability	4.11	3.12	3–5	5	0-7
Normal work	6.06	3.11	5–7	7	3–9
Relations	4.26	3.51	3–5	4	0-7
Sleep	5.94	3.30	5–7	7	48
Enjoyment of life	5.69	2.98	5–7	6	48

Table 8.25. Descriptive statistics for pain interference using BPI (n=35)

*SD = standard deviation **CI = confidence interval



Figure 8.25. BPI Pain interference

Pain interference in various activities of daily life in 35 leprosy patients with pain

8.7.1.2 Comparing pain interference scores between pain groups

The NP and non-NP sub-groups were compared with pain interference on BPI. As expected, the BPI interference mean total score was significantly higher in the NP group (mean 41.5, SD 11.9) compared to the non-NP sub-group (mean 25.7, SD 17.8) (p-value <0.004). The high scores present in the NP group indicate that a patient's pain has an impact on their day-to-day living and quality of life.

The scores for each interference domain were normally distributed (the Shapiro-Wilk test W: 0.96 - 0.980, P-values: 0.19 - 0.90). The Mann-Whitney U test as well as mean comparison tests were conducted for each domain between the two pain sub-groups, and both showed consistent results. All of the test results were significantly different (p-value <0.05) except for mood (p-value = 0.07), walking ability (p-value = 0.47), and normal work (p-value = 0.17) when analysed by type of pain. These results indicate that NP in patients with leprosy has a greater negative impact on sleep (p-value <0.001), relations with other people (p-value <0.03), enjoyment of life (p-value <0.04), and compromised ability to perform general activities (p-value <0.05) compared to patients with non-NP (Figure 8.26). A further sub-analysis for the patients with upper limb pain showed that normal work–Including both work outside of the home and housework–was significantly different between pain sub-groups (p-value = 0.04), (Table 8.26).

Compared to the mild and moderate pain groups, patients in the severe pain group had higher ratings on the BPI interference items (mood, relations with others, sleep, and enjoyment of life), and the mean interference scores increased with increasing pain intensity. However, results of analysis of variance showed no statistical difference (p-value >0.05).

Overall, NP frequency was found to be high among leprosy patients. In addition, the patients' worst pain scores on average were high and severe. Furthermore, NP affected patients' life activities. Thus, the impact of NP on quality of life seems to be prevalent among leprosy patients. Its psychological effect on leprosy patients will be discussed in the following section.





Mean scores in BPI interference in patients with non-neuropathic pain (n=10, white bar) vs. neuropathic pain (n=25, black bar). **P-value* < 0.05

BPI interference		Non-NP (n=10)	NP (n=25)	P- value
BPI general	No interference	3	0	0.02
activity	Mild	1	0	
	Moderate	5	15	
	severe	1	6	
BPI mood	No interference	3	0	< 0.01
	Mild	4	3	
	Moderate	0	9	
	severe	3	9	
BPI walking	No interference	3	5	0.80
ability	Mild	2	2	
	Moderate	4	11	
	severe	1	3	
BPI normal	No interference	1	0	0.04
work	Mild	3	2	
	Moderate	4	9	
	severe	2	10	
BPI relations	No interference	6	2	0.03
with other	Mild	1	5	
people	Moderate	2	8	
	severe	1	6	
BPI sleep	No interference	5	0	< 0.01
	Mild	0	2	
	Moderate	4	10	
	severe	1	9	
BPI enjoyment	No interference	2	0	0.1
of life	Mild	3	3	
	Moderate	3	11	
	severe	2	7	

Table 8.26. BPI interference in pain groups

8.7.2 Psychological well-being

8.7.2.1 Evaluation of the psychological well-being with the GHQ-12

Descriptive analysis

A total of 89 patients (98.9%) completed the GHQ-12 questions in the study. Sixty six point three percent of the respondents had a GHQ-12 score of three or more. The overall mean score for the GHQ-12 of the patients was 3.52 (SD 2.11) (Table 8.27) and the median score was four (inter-quartile range 2–5). The distribution of scores obtained from men and women are shown in Figure 8.27. The median value for men was three and for women was 4.5. No significant differences were observed.

Over 57% of patients had a feeling of "under strain" and suffered from stress. The other most common symptoms were "feeling unhappy and depressed" (51.7%), sleep problems almost every night (43.8%), could not overcome difficulties (38.2%), and lost confidence (37.1%). In addition, 29 patients (32.6%) felt worthless, 24 patients (27%) were not feeling happy, and 20 patients (22.5%) could not face problems. Other symptoms of mental disorders were much less common, and included not enjoying activities (15.7%), could not make a decision (11.2%), could not concentrate (8.99%), and a feeling of "not playing a useful part" (6.7%). Thus, more than two-thirds of the patients were considered to have possible psychological distress (anxiety and depression).

In patients with no clinical evidence of neuropathy, the proportion of patients scoring three or higher using the GHQ-12 was 35.7% (mean 2.29, SD 1.94), compared to 66.7% (mean 3.07, SD 1.83) in the sub-clinical neuropathy group, 58% (mean 3.08, SD 1.97) in the neuropathy group, and 83.3% (mean 4.5, SD 2.01) in the pain group (Table 8.27). Patients with neuropathy and pain had a poorer mental health compared to other groups (p-value <0.001) (Figure 8.28).

Among patients with pain, 30 individuals (83.33%) had a GHQ score of three or more, compared to 29 (54.72%) among non-pain patients. 24 patients (92.31%) with NP had a GHQ score of three or more, compared to six patients (60%) in the non-neuropathic group. Figure 8.29 shows the distribution of GHQ-12 scores of three or more in leprosy patients with and without pain.



Figure 8.27. Distribution of GHQ-12 scores (n=89)

Distribution of GHQ-12 scores in men (median 3, above thresholds: 64%) and women (median 4.5, above thresholds: 75%)

Table 8.27.	GHQ-12 scores †

GHQ-12 item	Groups				Total
					scores
	Group1	Group2	Group3	Group4	(n=89)
	(n=14)	(n=15)	(n=24)	(n=36)	
Could not concentrate	1.9 (0.5)	1.8 (0.7)	1.5 (0.7)	1.5 (0.7)	1.6 (0.7)
Lost sleep	2 (0.7)	2.5 (1.0)	2.3 (0.7)	2.6 (1.0)	2.4 (0.9)
Not playing a useful part	1.7 (0.5)	1.9 (0.5)	1.7 (0.7)	1.5 (0.6)	1.6 (0.6)
Could not make a decision	1.8 (0.6)	1.6 (0.6)	1.8 (0.7)	1.9 (0.7)	1.7 (0.7)
Felt under strain	1.9 (1.1)	2.1 (1.0)	2.5 (1.1)	2.7 (0.9)	2.4 (1.0)
Could not face difficulties	2.1 (0.7)	1.9 (0.9)	1.9 (1.0)	2.4 (0.9)	2.1 (0.9)
Not enjoying activities	1.9 (0.7)	2 (0.7)	1.8 (0.9)	1.5 (0.7)	1.7 (0.8)
Could not face problems	2.1 (0.7)	1.9 (0.6)	1.9 (0.9)	1.7 (0.7)	1.9 (0.8)
Unhappy and depressed	1.6 (0.9)	2.2 (1.2)	2.1 (1.1)	2.7 (1.0)	2.3 (1.1)
Lost confidence	1.9 (0.7)	2.1 (0.8)	1.8 (0.8)	2.4 (0.9)	2.1 (0.9)
Felt worthless	1.5 (0.8)	1.4 (0.8)	2 (0.9)	2.2 (0.9)	1.9 (0.9)
Not feeling happy	2.1 (0.8)	1.9 (0.8)	1.8 (0.8)	2.1 (1.0)	1.9 (0.9)
All GHQ-12 items	2.3 (1.9)	3.1 (1.8)	3.1 (1.9)	4.5 (2.0)	3.5 (2.1)

↑ A table displaying the scores for the GHQ-12 and how these scores were distributed between the groups (No neuropathy, sub-clinical neuropathy, neuropathy, and neuropathy and pain), as well as total scores. Mean values and SDs shown



Figure 8.28. Overall score of GHQ-12 in 89 leprosy patients

The relation between the presence of neuropathy, pain and overall score of depression in 89 leprosy patients.



Figure 8.29. GHQ-12 score of 3 or more in 89 leprosy patients

Analysis

Univariate analysis

Univariate analysis was carried out for the association between the variables and mental health (depression) as an outcome. The following factors were strongly associated with the outcome: patients' motor nerve impairment, delay in presentation, WHO-disability classification, disease duration, and the presence of pain particularly neuropathic pain (p-value <0.001). There was, however, some evidence of an association between mental health status and educational qualifications (p-value = 0.04), disease duration (p-value = 0.05), presence of skin lesions on examination (p-value = 0.06), number of skin lesions on examination (p-value = 0.06), number of skin lesions on examination (p-value = 0.07), sensory nerve impairment (p-value = 0.08), and the presence of a reaction on examination (p-value = 0.1). Other variables showed no evidence of an association (p-value >0.2). Appendix 48 shows the distribution of GHQ-12 (above a threshold) cases by patients' characteristics and the odds ratios for each variable.

GHQ scores and social, clinical features and pain variables

In the following section the association between psychiatric morbidity and social and clinical findings are described.

Association between socio-demographic variables and GHQ scores

The prevalence of significant psychiatric morbidity between those in different age groups and sexes were not statistically different (Appendix 48). This may be due to age and gender match of the study population. However, at all ages and for both sexes, those who have some formal educational qualifications had better mental health than those who had no such qualifications. The difference between these two categories was more pronounced in the 20–29-year old age group. For men there was a statistically significant difference between those with and without qualifications in relation to their mental health. While those with no qualifications reported a higher prevalence of psychiatric morbidity in the youngest age groups, the differences were small and there was no difference at ages 30–39, 40–49 and >50. There was no difference for women across the five age bands. The odds ratios for educational level were found to be marginally significant (p-value <0.04). Married participants had approximately the same prevalence of psychological morbidity as single or separated, widowed, or divorced

participants. Of currently married participants, 67.8% had a higher GHQ-12 score, while 63.3% of the non-married group had a higher GHQ-12 score.

Association between clinical findings and GHQ scores

Patients who had clinical evidence of motor impairment were significantly more likely to have scored above the GHQ-12 threshold score (crude OR 3.13, 95% CI 1.20–8.18). Of patients with motor impairment, 78.26% obtained higher GHQ-12 scores, compared to 53.49% of patients who did not have motor impairment. In comparison with motor impairment, neither sensory impairment nor neuropathy (both sensory and motor) showed a statistical association with a GHQ-12 score, p-values 0.08 and 0.07 respectively. There was, however, a clear relationship between ulnar neuropathy and GHQ-12 score (crude OR 4.46, 95% CI 1.55–12.82).

There was a clear relationship between the presence of disability and GHQ-12 score. Patients who reported having a disability at diagnosis or at the time of interview were more likely than those who did not to have a GHQ-12 score of three or more. For respondents who had a disability at diagnosis, the figures were 82.9% and 17.1% respectively, and for those who had a current disability they were 75.5% and 55.0% (Appendix 48). Of those who had a previous reaction, 90% had a higher GHQ-12 score compared to 80% of patients who did not experience a reaction.

There was also a clear relationship between the reaction and GHQ-12 score. Patients who reported having a previous reaction were more likely than those who did not to have a GHQ-12 score of over three (crude OR 2.25, 95% CI 1.13–8.21).

Association between pain and GHQ scores

Large differences were evident when comparing those respondents who reported having pain symptoms and those who did not. Of the former, 83.3% had a GHQ-12 score of three or more, while the equivalent figures for the latter were 54.7%. Among those who had pain, respondents with symptoms and signs suggestive of neuropathic pain were more than nine times more likely to have an above-threshold GHQ-12 score than those with non-neuropathic pain (crude OR for neuropathic pain 9.93, 95% CI 2.13–46.35; crude OR for non-neuropathic pain 1.24, 95% CI 0.31–4.91). Twenty-four out of 26 patients (92.3%) with a presentation suggestive of NP using the DN4 questionnaire had a higher GHQ-12 scores compared to 60% of those patients presenting with complaints

of non-neuropathic pain, and 54.72% of the no pain group. The greatest difference was for men at age 20–29 and 30–39 where those with neuropathic pain were more likely to have a high GHQ-12 score than those in the non-neuropathic pain group. At the oldest female age group, smaller numbers obscure the pattern. Overall, there were clear relationships between the presence of neuropathic pain and GHQ-12 scores. Those who were diagnosed with neuropathic pain had the worst mental health.

Multivariate analysis

Logistic regression

Adjusted odds ratios for experiencing mental health disturbances were estimated using logistic regression. Stepwise logistic regression modelling was done (Appendix 49). The main social and clinical findings were included in the model for GHQ-12. This modelling allowed us to assess the association for each variable, adjusting for others, and to examine which factors affected the odds of having a high score on the GHQ-12.

All variables and not only those significantly associated variables (above a certain pvalue in the univariate analysis), were considered for the multivariate regression model. For instance, gender and age group did not have a significant effect on the association, but were included in the multivariate regression model, as both are known universal confounders. Variables were retained in the final multivariate regression model when the model found significant evidence (p-value <0.05) with that variable included than without it, using a likelihood ratio test. In the model, the variable most significant in the univariate analysis was included first (forward regression). The next variable selected in the model for inclusion was then the one that best improved the model based on the LRT test until no additional variables improved the model significantly. In addition, all potential variables were included in the model first and then removed one-by-one starting with those least associated with the outcome (backward regression).

After adjusting for all other factors in the model, the odds of having significant psychiatric morbidity fell by more than one third (from 9.60 to 6.25). This result showed that the magnitude of the association decreased, but strong evidence of an association between psychiatric morbidity (anxiety and depression) and presence of neuropathic pain (p-value = 0.03) remained. There was strong evidence of reduced odds for the presence of neuropathic pain compared to patients with no neuropathic pain.

While there was a tendency for the presence of disability, particularly motor nerve impairments, to have higher psychiatric morbidity, there was no statistical significance (p-value 0.09). Neither socio-demographic factors nor disease duration were associated with mental health in either men or women.

The major determinant of mental health of the cohort was the presence or absence of chronic pain, "neuropathic pain", or presence of disability, "motor nerve impairment". The likelihood of psychiatric morbidity increased if neuropathic pain was present. The odds of psychiatric morbidity in those with neuropathic pain was 6.25 the odds of disorder in those without pain. At this stage in the study, we would suspect that those with chronic neuropathic pain were at a higher risk of developing worse mental health. This result would make sense, as patients with neuropathic pain are more likely to suffer from its impact on all quality of life modalities.

Chapter 9 Discussion

9.1 Introduction

The aim of this doctoral thesis was to investigate the clinical characteristics of patients with leprosy-associated NP and to establish the somatosensory profiles of such patients. Recent advancement of NP assessment has included two aspects. First, the identification and baseline profiling of patients with NP, including different dimensions of chronic pain problems, and second, the adaptation and development of a comprehensive clinical trial design for NP treatment. I hypothesised that:

- i. Patients with painful neuropathy have a different sensory profile compared to patients with non-painful neuropathy
- ii. Patients with NP endure a greater quality of life and psychological well-being burden than those with leprosy and neuropathy, but without NP.

These baseline comprehensive hypotheses were tested in a case-control study of a leprosy cohort consisting of patients with established pain and neuropathy, patients with pain-free and established clinical evidence of neuropathy, patients with no pain and no clinical evidence of neuropathy, and local healthy volunteers recruited in Mumbai, India.

9.2 Summary and discussion of the findings

Somatosensory profiles were compared to those from the DFNS-QST reference data and local Indian healthy control subjects. The pattern in all leprosy patients revealed a novel profile not previously seen in other NP conditions whereby cool and warm detection thresholds and also mechanical detection were lowered but vibration perception was preserved. This is somewhat different to profiles seen in other NP conditions. Patients with leprosy NP had a high rate of abnormal findings in almost all QST parameters when measured in the maximum pain area in the ulnar nerve territory. Their sensory profiles were categorised into two subgroups. The majority of patients have spontaneous pain with evidence of sensory loss, but no sign of sensory gain, these findings are consistent with peripheral neuronal damage. The second subgroup showed pain and temperature sensation to be profoundly impaired, but light mechanical stimuli produced pain (dynamic mechanical allodynia). Surprisingly, the obtained QST profiles in leprosy patients with pain were not significantly different from those patients without pain. Patients with NP had poor quality of life and psychological well-being compared to those with pain-free neuropathy. Hence, the characterisation of patients with respect to the NP is of therapeutic significance.

9.2.1 Sociodemographic, clinical and epidemiological characteristics of study participants

The sociodemographic profile found in this study is comparable with those in other studies conducted in different parts of India, which point out gender bias, illiteracy, and a move to a city as constant characteristics of leprosy populations (Thakkar and Patel, 2014, Van Brakel et al., 2005a). The age profile of the patients was younger than the general population. In India more than 65% of the population are below the age of 35 years (WHO, 2014). Approximately one third of the population sample were in their third decade. The relatively few patients in either the under 20s age group or the above 50s age group probably reflect the sampling method and the nature of the disease.

In leprosy, the proportion of male to female is identical (1:1) up to puberty, then changes to 2:1 which reflects the natural history of the disease (Guinto and Rodriguez, 1936). In our cohort, the proportion of male and female patients was not evenly distributed across age groups and the ratio of 3:1 was also slightly different from that in the general leprosy population. This could be explained by the sampling method of the study. In many developing countries leprosy clinics are accessed by more men than women. In addition, bias in favour of men is also found in India (Hausmann, 2013). The high illiteracy and low educational level in our cohort reflects the strong association in most leprosy populations with complex variables of poverty such as income, housing quality hygiene and education. The geographical distribution of our cohort sample was also typical of the general leprosy population in India. In this study, half of the patients recruited at the BLP had moved from rural areas to Mumbai. This could be explained by the strong association between leprosy and poverty (Murto et al., 2013), as leprosy may be characterised by rural incidence and urban prevalence, i.e. cases move to cities.

Regarding clinical characteristics, it is worth noting that the high prevalence of MB cases associated with the high percentage of disability grade 2 at diagnosis is indicative of late diagnosis and lack of early detection of cases. Grade 2 disability at diagnosis, which is defined as the presence of visible deformity, is an indicator of the late diagnosis and severity of the disease. In our cohort, 10% of patients had disability grade 2 at diagnosis.

Our finding is similar to results from the INFIR cohort study (9.6%). The proportion of grade two disability in newly diagnosed leprosy cases in India reported by WHO at 3% (WHO, 2013b) this could be explained by the fact that the WHO figure included PB cases who usually have lower rates of disability. The high percentage of grade 2 disability at diagnosis in our cohort suggests that the risk of leprosy transmission is still ongoing in the study area. Delayed presentation is a known risk factor for disability in leprosy (Schreuder, 1998, Meima et al., 1999). A study from Thailand (Schreuder, 1998) has shown a highly significant correlation between the proportion of new cases with disability and delay in diagnosis. An Ethiopian study found an odds ratio of 2.1 for grade 2 disability when registration was delayed by more than 2 years (Meima et al., 1999). In this study, we found that more than 70% of the patients had had symptoms for more than 6 months before starting their MDT treatment. Data on the main reasons for this delay were related to misdiagnosis and patient's unawareness of the disease; other studies have found similar level of delay, explained by stigma and difficulties accessing services (Lockwood and Reid, 2001, Nicholls et al., 2003). These findings indicate that there is still more work to be done to prevent and manage disability in leprosy. A recent publication on the use of the WHO disability grading system by Cross and colleagues (Cross, 2014), used Delphi methods to reach a consensus among fifteen experts on the prevention of disability due to leprosy. The authors defined the terms more precisely and provided guidelines for use in the clinic, which can be immediately applied. There is limited time for the current global strategy, which proposed the target of reducing the rate of new cases with grade 2 disabilities per 100 000 population by at least 35% by the end of 2015, compared to the baseline at the end of 2010 (WHO, 2009a). However, the new guidelines may help to improve disability assessment, monitoring and successful leprosy burden reduction.

Twenty six patients (28.9%) had leprosy reactions at the time of examination, and previous history of reaction was reported by 40 patients (56%). Nociceptive joint pain is likely to be reaction associated. Pain associated with neuritis reaction, which is defined by the development of inflammation of a nerve sheath without abnormal findings in sensory testing, is considered to be of inflammatory origin and clinically defined as nociceptive pain (originating from nervi nervorum) (Bove and Light, 1997). However, if an inflammatory neuritis causes nerve damage then the pain, is by definition, neuropathic. Inflammatory pain is usually considered to be of nociceptive character, because it partly results from hyperexcitability of intact nociceptive dorsal root ganglion

neurons innervating inflamed tissue. However, chronic inflammatory pain is often characterized by positive signs such as allodynia, suggesting a possible neuropathic component. Recent studies have shown that inflammation-induced nociceptor hyperexcitability is sustained by C-nociceptors, which may contribute to inflammatory hyperalgesia (Flynn et al., 2014). This may also explain why in some inflammatory conditions nociceptive and NP may overlap. In our study, patients with reactions could have been over-represented, but this group is still attending leprosy services and therefore accessible to our recruitment tactics. The high prevalence of reactions in our cohort supports the growing evidence that the development of NP is probably immunologically mediated (Lund et al., 2007). The immune response in the peripheral nerves may recur and if it repeatedly affects the peripheral nerves, chronic-post inflammatory pain may result. Lockwood and colleagues (Lockwood et al., 2002) have shown that *M. leprae* protein and lipid antigens are present in skin and nerves at the time of acute reversal reactions. These data show the importance of reactions in pain among leprosy patients; they can be significant risk factors for the development of chronic NP.

The epidemiological profile in this cohort had a prevalence of leprosy neuropathy of 68%, with 60% of those reporting pain. NP was found in 28.8% of the patients. This finding is similar to the large epidemiological studies: In the INFIR cohort study, the neuropathy is consistently reported at around 40% (van Brakel et al., 2005b). New peripheral nerve damage is present in about 65% of cases. In our previous study in Ethiopia, pain was experienced in 60% of the patients who had completed their MDT within 18 months (Haroun et al., 2012). The prevalence of NP in this study is lower than other NP due to infectious disease, such as HIV-SN, which stand at 40%. Previous epidemiological studies on leprosy neuropathy and pain showed that NP occurs in up to 30% of patients in the long term (Hietaharju et al., 2000, Stump et al., 2004, Saunderson et al., 2008, Lasry-Levy et al., 2011, Haroun et al., 2012, Ramos et al., 2014). This supports the validity of the NP criteria used, and it is likely that these findings would be applicable to larger leprosy populations and in different settings. However, this study did not find some of the well-established risk factors for leprosy neuropathy and pain, such as reaction (Appendix 47). This could be because the case control study was powered against the sensory changes associated with risk of developing HIV neuropathy from the pain in HIV related neuropathy study (Phillips et al., 2014), and was not designed to elucidate these risk factors in the same way as larger epidemiological studies.
The HIV study is one of few studies to use quantitative sensory testing to assess sensory parameters in NP caused by infectious diseases.

9.2.2 Diagnostic tools

9.2.2.1 Case definition

The ability of health personnel in remote leprosy clinics to identify NP in patients with chronic pain is unclear. In this study, leprosy patients with NP were defined in a stepwise manner. First monofilaments and VMT, which are widely used in leprosy clinics, were used to confirm that some damage to the somatosensory sensory system along ulnar nerve territory had occurred. A sensory deficit in the innervation territory of a lesioned nerve is a diagnostic criterion of NP (Treede et al., 2008, Haanpaa et al., 2011a). Secondly, as the presence of nerve damage per se does not necessarily indicate that pain is neuropathic in origin, patients were further classified based on a score of \geq 4/10 using the DN4 symptom descriptors questionnaire. Using this definition, would provide simplicity, validity, utility and affordability for use in most routine clinical practice in leprosy population. This indicates that our case definition may help to develop a new tool for the assessment of NP in leprosy that might be useful in resource-limited settings.

9.2.2.2 Pain symptoms

Screening tools are used to alert clinicians to the possibility of NP (Haanpaa et al., 2011a). In previous studies done in collaboration with the Bombay Leprosy Project in India, ALERT hospital in Ethiopia, and LSHTM we demonstrated that the DN4 questionnaire is valid in its application in the leprosy NP in different languages and settings (Lasry-Levy et al., 2011, Haroun et al., 2012). In this study, I used the DN4 questionnaire for the case definition of NP; a score of 4 or higher in patients with evidence of neuropathy. In addition, I used PD-Q to verify the DN4 result.

NP, based on the identification of common pain symptoms and sensory tests over the ulnar nerve territory (C8) obtained with the DN4 questionnaire, was identified in the pain group with mean scores of 5.6 out of 10, specifically with the presence of "numbness" in the NP and "aching or dull pain type" in non-NP patients. The presence of NP in patients with upper limb pain was identified in 24 patients out of 32. The number of DN4 sensory pain descriptor items described by those patients was higher

than that identified by examination of sensory function. Similar findings were also observed in a study by Bouhassira, who compared the clinical features of NP and non-NP in 160 patients in France (Bouhassira et al., 2005). In our study "tingling", "burning", and "numbness" were the sensory descriptors most widely used by NP patients. Hypoesthesia to touch and pin-prick were much more frequent in NP than in non-NP patients. These findings are consistent with previous studies on NP in Fibromyalgia (Petzke et al., 2003, Staud et al., 2003). The presence of "painful brush" sensation was also more frequent in NP compared to the non-NP group, although it was not statistically significant.

In this study clinical examination, and the DN4 questionnaire, were used for the case definition of leprosy NP. The PD-Q identified fewer patients (13 subjects) with NP components compared to DN4. Of 32 patients with upper limb pain, 24 patients were clinically classified by DN4 as definite NP, while PD-Q identified only 5 of these patients. There were inconsistent responses to the common questions between DN4 and PD-Q in 60% of cases. It is unclear if the lowered sensitivity of PD-Q in our cohort might be related to the validity of the questionnaire in this population and setting, as there are no previous studies documenting the clinical diagnostic accuracy and reliability of PD-Q in leprosy. Similar findings were also observed in patients with neck/upper limb pain by Tampin and colleagues (Tampin et al., 2013). In this study, the author investigated the application of LANSS and PD-Q in 152 patients with neck/upper limb pain. Both questionnaires failed to identify a large number of patients with clinically classified definite NP. These findings suggest that the application of PD-Q in patients with leprosy NP is limited.

Another potential observation regarding the differences between the DN4 and PD-Q is questionnaire design. For instance, the number and type of questions are different: the DN4 consists of 10 symptom and clinically-related questions; while PD-Q contains only seven symptom-related questions. The phrasing of the questions and the scoring method are also different. Whilst DN4 uses yes or no fixed scores for each question, sensory descriptor questions are score-weighted in PD-Q. The latter scoring may introduce bias, as responses could be vulnerable to subjective psychological factors, which potentially contribute to an overall higher score. These observations in questionnaire design and low sensitivity between the two instruments, support the use of DN4 as screening tools for leprosy NP. Patients were asked to evaluate the administration of NP screening tools; no difference was found between DN4 and PD-Q. In a previous study, we documented that DN4 was easier to administer than the LANSS in patients with leprosy and NP (Haroun et al., 2012).

9.2.2.3 Clinical examination

The nine nerves commonly affected in leprosy, namely: greater auricular, median, ulnar, radial cutaneous, ulnar branch, lateral popliteal, superficial peroneal, posterior tibial, and sural nerves were assessed in a cohort of 90 patients with and without neuropathy and pain. Of these nine nerves, I compared the sensory and motor findings from ulnar nerve examinations between groups. I also compared the results of the QST tests against monofilament testing.

In this study, abnormal tendon reflex or JPS was present in very few patients (only around 2%). This is typical for leprosy, and accords with previous observations. Jennekens and colleagues, who examined 28 male leprosy patients, found abnormal JPS of one or more digits in 33% of the participants (Jennekens and Jennekens-Schinkel, 1992). In the INFIR cohort study around 2% of the 303 subjects had abnormal reflexes or JPS (van Brakel et al., 2005b). Ramadan and colleagues found 'diminished' reflexes in 45% of their patients and 'diminished joint and vibration sensation' in 33%. However, the patient group in the latter study was older and had longer histories of leprosy. In our study, the presence of abnormal reflexes or JPS could be explained by neuropathy of long duration (van Brakel et al., 1994b).

In the subclinical neuropathy group, the QST findings showed ulnar impairment in up to two third of the patients. This indicates that QST testing is a more efficient method of clinical detection than those currently used in leprosy neuropathy. A similar conclusion, that monofilament and VMT are not very sensitive methods in detecting neuropathy, was also observed by McKnight, who further analysed the data from the INFIR cohort study (McKnight, 2010). However, the use of monofilament and VMT in resource-limited setting is reliable (Brandsma et al., 2014). In Chapter 7, I showed that the use of monofilament in detecting touch sensation in healthy volunteers is comparable to electronic von Frey, but may be different in pathological conditions. The low sensitivity of clinical evidence of neuropathy in detecting ulnar impairment compared to QST in the subclinical neuropathy group can be explained by the fact that thermal (warm and cold) sensation occurs early in leprosy, which cannot be detected by monofilament. Another explanation could be related to the scale used for the VMT grading range; 4 scores instead of 3 for abnormal limits (Van Brakel et al., 2007). The new grades of weakness are more difficult to assesses in comparison to no weakness (Brandsma, 2000).

In the current study, I found that all nerves classed as impaired on the monofilament test also had two or more QST based abnormal parameters. In the ulnar nerve the combined thermal tests were impaired in all right ulnar nerves, and in more than 90% of left ulnar nerve impaired according to the monofilament test. In addition, a substantial proportion (>60%) of the nerves had impaired thermal sensation, but normal touch sensation. These findings indicate that, if touch sensation is affected, one of the QST parameters will also be abnormal. Unlike the INFIR cohort study findings (Van Brakel et al., 2005a), our study showed that impaired thermal sensation may be detectable before touch sensation. Our findings could be explained by the fact that small, unmyelinated fibres are the first to be affected in leprosy (Shetty et al., 1988, Shetty et al., 1977).

In our cohort, the pattern assessing large afferent fibre using monofilament and vibration tests revealed a loss of touch sensation, but vibration perception was preserved. These findings were fully compatible with what might be expected from leprosy neuropathy, i.e., not all ulnar nerves with impaired touch had impaired vibration sense. This is similar to the findings for neuropathy assessment in the INFIR cohort study by Brakel and colleagues (Van Brakel et al., 2005a). The authors found high negative concordance between monofilament and vibration perception; if the monofilament test was normal, the vibration test was also normal in the great majority of nerves. This indicates that the vibration perception test is less sensitive for detecting sensory neuropathy in leprosy than monofilament testing.

9.2.2.4 Metabolic factors

In this cohort, the mean values of plasma level of vitamin B_{12} across the study group were high, unlike the healthy cohort where three volunteers with very low levels of B_{12} were identified. Vitamin B_{12} deficiency is an important factor in the development of peripheral neuropathies (Reynolds, 2014). The low vitamin B_{12} observed in our healthy volunteers could be explained by Indian dietary habits, as the majority of volunteers were vegetarian. Higher plasma levels in our patients, particularly the neuropathy and pain group, could be explained by the fact that 60% of them were on vitamin supplementation. Vitamins are givens to all patients as a routine treatment at the BLP clinic.

9.2.3 Quantitative sensory testing

This is the first study to use the DFNS-QST protocol in a resource-limited setting and validates the Mumbai centre for the use of the QST. The QST machine (MSA Thermotest, Somedic) used in this study was easy to operate. The protocol is highly relevant in leprosy, because both small and large fibres are damaged.

QST results are interpreted on the basis of a DFNS reference data set that is stratified for age and gender (Rolke et al., 2006a, Magerl et al., 2010). It contains a description of somatosensory profile and, subsequently, refers to possible underlying mechanisms.

9.2.3.1 Differences between healthy controls and patients

Consistent differences in quantitative sensory testing results were observed between our local Indian samples of healthy control participants and patients with leprosy. The most frequent differences was sensory loss indicated by increased thermal detection thresholds (CDT, WDT) and increased mechanical and vibration detection thresholds (MDT, VDT) in leprosy patients compared to healthy controls. This highlights the fact that nerve damage is a characteristic of leprosy, even if there is no clinical evidence of neuropathy using monofilaments and/or voluntary muscle testing.

QST measurements in the current study revealed that patients with leprosy had a widespread loss of function in the tested site (dermatome C8) in most DFNS-QST thermal and mechanical sensory parameters, but also that vibration sense tend to be preserved. About two thirds of participants showed a loss of function of at least one sensory modality. The same was found for leprosy patients in the INFIR study by Van Brakel et al; loss of temperature sensation, but retained vibration perception (Van Brakel et al., 2005a). Vibration "sense" is mediated by large afferent $A\alpha\beta$ fibres (Light and Perl, 1993). VDT and monofilaments tests both assess large afferent fibre function, it was therefore expected that vibration test would also be affected. One explanation would be that the preserved vibration sense reflects the type of nerve fibres that are damaged in leprosy (i.e. there is a preservation of the large myelinated sensory fibres that transmit

vibration sense in leprosy). Another explanation for the preserved vibration in this cohort could be because vibration is transmitted by bone not skin which may not reflect abnormality in the cutaneous nervous system. Further explanation again would be the preserved spinal dorsal columns in leprosy. The supporting explanation for this may be that since the spinal dorsal columns are preserved in leprosy, one needs less intense afferent input to perceive vibration and the remaining sensory input is sufficient for vibration and yet not of other thermal and mechanical parameters.

In concordance with our hypothesis, patients with leprosy had localised sensory abnormalities in the tested ulnar site (negative sensory signs: reduced thermal and mechanical modalities), indicating a loss of small and large sensory fibre function. The main sensory finding was loss of function of the C and A δ mediated sensory modalities of thermal non-nociceptive parameters (60%) and functions mediated by large fibres (A β) (46%), except for vibration sense (8%). This is similar to previous findings for leprosy patients in the INFIR study; loss of thermal sensation, but retained vibration sense (Van Brakel et al., 2005a, van Brakel et al., 2008a). In contrast, abnormal negative nociceptor QST parameters ranged from 7.9% (CPT) to 42% (MPS). The presence of these negative sensory findings is indicative of peripheral nerve damage (Hansson, 2002). Loss of function occurred in all sensory fibres tested (C, A δ and A β), which is consistent with previous findings in patients with peripheral nerve damage and in patients with HIV sensory neuropathy (Kleggetveit and Jorum, 2010, Phillips et al., 2014).

9.2.3.2 Somatosensory profiles in patients with and without pain

Patients with and without pain had a similar sensory abnormalities for nociceptive and non-nociceptive QST parameters. Abnormality of thermal, and mechanical detection thresholds; and pain thresholds to the nociceptive parameters were common and similar in both patients groups compared to healthy controls. Although the cumulative frequency of abnormalities in any of the thermal and mechanical measures were higher in patients with painful neuropathy, it was not statistically significant. The QST findings showed a pattern of profound sensory loss in both groups. A similar distribution of QST profile in patients with and without pain has previously been reported in other conditions. In a recent study of HIV painful neuropathy, the results resemble those of the present study. In a QST profile study of 66 HIV infected participants, Phillips and colleagues (Phillips et al., 2014), reported no differences in regard to thermal and

mechanical perception thresholds between painful and non-painful sides of denervated skin. In addition, the authors found that patients with painful neuropathy had profound loss to vibration sense. The implications of our findings that leprosy patients with pain do not have significantly different QST profile from leprosy patient without pain is discussed in detail in section 9.3.6, page 276.

In this study, the differences in QST parameters between symptomatic and asymptomatic sides in 21 leprosy patients with symptoms and signs suggestive of NP and 10 patients with non-NP, was investigated. The QST findings demonstrated a significant loss of function mediated by nociceptive and non-nociceptive sensory fibres in both sides of the NP groups; these findings are consistent with the characteristics of NP (Haanpaa et al., 2011a). However, in patients with non-NP, the QST findings demonstrated a significant loss of function on the symptomatic side compared to the asymptomatic side. These findings indicate that unlike non-NP, in patients with NP there was no significant side-to-side difference in any QST parameters in the maximum pain area. Our findings could be explained by the fact that in NP patients, loss of thermal and mechanical detection may occur bilaterally. Contralateral loss of thermal detection has been observed in patients with nerve damage (Leffler and Hansson, 2008b, Jaaskelainen et al., 2005). In patients with trigeminal neuropathy, contralateral loss of thermal detection was associated with the presence of ipsilateral NP. This was explained by peripheral nerve damage induced inhibition or disturbed excitatory connections within the central pathways mediating non-noxious thermal information from the contralateral side (Jaaskelainen et al., 2005, Davis et al., 2011).

Our results indicate that the QST measures allow us to differentiate between patients with and without neuropathy, but cannot be used to differentiate between leprosy patients with and without pain, which is similar to the study by Phillips and colleagues (Phillips et al., 2014).

9.2.3.3 Somatosensory profiles in patients with NP

The description of the somatosensory profiles in patients with leprosy NP is the central focus of this doctoral thesis. As mentioned in the introduction, baseline profiling is an important step towards better understanding of pathophysiological mechanisms as well informed mechanism based prescribing, which in turn influences clinical trial design and drug responder defining NP treatment (Reimer et al., 2014).

To our knowledge, this is the first study to comprehensivly investigate the clinical presentation of NP in leprosy. A detailed description of the leprosy NP phenotype is given. As expected, the majority (90%) of leprosy NP patients showed sensory abnormalities at their ulnar affected side. This is similar to the findings for NP patients by Maier and colleagues (Maier et al., 2010): In this study, 1236 patients with NP of different aetilogies were investigated and somatosensory profiles of all patients were assessed, the authors reported a high percentage (92%) of patients with at least one QST abnormality. The most prevelant profiles were thermosensory and mechanical hypoesthesiac (in up to 41%).

In our study, across different thermal and mechanical modalities, the predominant profile in leprosy NP was loss of sensory function; 90% of leprosy NP patients showed a loss of function of at least one sensory modality. In concordance with our hypothesis, sensory loss of functions was predominately found in non-nociceptive parameters, with higher incidence for thermal function mediated by small nerve fibres (CDT 84.2%, WDT 89.5%, TSL 94.7%) and mechanical function mediated by large nerve fibres (MDT 73.7%), whereas loss of vibration sense mediated by dorsal column tract occurred in 26.3%. Similarly, sensory loss of function in nociceptive parameters occurred in most patients (MPT 85.7%, MPS 78.9%, and HPT 52.6%), except for pressure pain (15.8%), which reflects the innervation of deep muscles rather than cutaneous sensory receptors. On the other hand, sensory gain of functions for both non-nociceptive and nociceptive parameters was rare. 5.3% of the patients had DMA and 15.8% had abnormal WUR. 15.8% of the patients had PHS.

In patients with pain and neuropathy group, I found that 26.5% of them had increased vibration detection threshold (i.e. loss), while our finding of generalized retained vibration sense in non-NP patients is consistent with previous studies (Van Brakel et al., 2005a). Our demonstration of increased vibration detection threshold in patients with NP has not been reported in leprosy, but corresponds with other NP conditions (Maier et al., 2010, Konopka et al., 2012).

In studies conducted by Hammond and colleagues in India, the vibration thresholds, which were assessed using a biosthesiometer technique, were found useful for predicting the risk of plantar ulcer (Hammond and Klenerman, 1988), risk of tarsal disintegration (Klenerman et al., 1990), and for diagnosing sensory impairment in skin lesions

(Klenerman and Hammond, 1988). The authors showed that vibration sense was affected in 90% of feet at risk of ulceration.

This important findings, as it reflects that a considerable percentage of patients (26.5%) had impaired vibration sense, wich could be a potiential predictor for the development of NP in leprosy. This observation is again indicative of peripheral nerve damage.

Previous studies of patients with NP have shown that loss of sensory function is associated with central or peripheral nerve damage which may be complicated by ongoing pain via increased ectopic activity (Liu et al., 2000, Orstavik et al., 2006, Zhao et al., 2006, Vaso et al., 2014). This is usually associated with high incidence of thermal and mechanical loss of function. The frequent sensory loss found in leprosy NP patients may indicate peripheral nerve damage.

Although PHS in the affected area was frequently reported (15.8%), its consideration in regards to sensory function is still controversial: some believe it to be a sensory gain phenomena and others a sensory loss.

Interestingly, in this group of leprosy patients with NP I found an appreciable percentage of patients (5.3% and 19%) who had pain due to light moving mechanical stimuli (allodynia) in the tested site C8 and along the ulnar nerve territory, respectively. This finding is consistent with peripheral nerve damage. Sandkuhler had similar findings assessed behavioural hypersensitivity to stimuli after inducing a mechanical peripheral damage in animal models of NP; high rates of positive sensory signs, such as, blunt pressure pain, dynamic mechanical allodynia, and pinprick hyperalgesia were also found in patients with peripheral nerve damage (Sandkuhler, 2009).

The pathophysiological mechanisms of NP associated with leprosy are not well established. They may involve the development of peripheral nerve damage or central mechanisms, or both. In this study, the QST findings have shown that the majority of patients have spontaneous pain with evidence of sensory loss, but no signs of sensory gain (hyperalgesia or allodynia). Such patients characteristically have lost both small and large diameter fibres. In this group, the pain is possibly due to increased spontaneous activity in deafferented central neurons and/or reorganization of central connections (Wallace and Rice, 2008). A few other leprosy patients have abnormal sensitization of unmyelinated cutaneous nociceptors (irritable nociceptors). In these patients, the QST

results have shown that pain and temperature sensation are profoundly impaired but light mechanical stimuli often produce pain (allodynia). In these patients, allodynia may be due to the formation of new connections between non-nociceptive large-diameter primary afferents and central pain transmission neurons (Fields et al., 1998).

In summary, the profile of NP associated with leprosy shows signs of sensory loss by increased thermal detection thresholds (CDT, WDT) and increased mechanical and vibration detection thresholds (MDT, VDT). No signs of sensory gain are found by pinprick hyperalgesia (MPT, MPS), except for DMA and WUR. This sensory phenotype of sensory loss in patients with leprosy is likely to reflect the underlying pathology peripheral nerve damage.

9.2.4 Pain intensity and quality assessment

Self-reported pain intensity in leprosy patients was assessed using the BPI. Despite different pain distribution in NP and non-NP, the two groups reported similar pain intensity. Similar findings were also reported in PHN and LBP (Daniel et al., 2008). These finding support the evidence that patients with NP suffer to a similar extent to those with nociceptive pain (Haythornthwaite and Benrud-Larson, 2000). The average pain intensity among patients with NP was 5.4. Previous pain studies on NP describe "worst pain (rated as five or above)" as significant pain, and report it as contributing disproportionately to more functional impairment. In this study, around 40% of patients with NP rated their worst pain as five or more, which is lower than in the Ethiopian and Brazilian study of leprosy patients with NP (Haroun et al., 2012, Stump et al., 2004).

The genders different in responses to pain severity or pain-related functioning in patients with leprosy are not known. In this study I found that no significant sex differences in measures of pain and functioning (depression symptoms, pain severity or interference). A similar finding, no differences between men and women in their responses to pain, were reported by Racine in study from UK, USA and Canada. The authors used a cross-sectional design with a cohort of 747 women and 48 men with fibromyalgia syndrome referred to The Fibromyalgia Day Program (Racine et al., 2014). These findings oppose the fact that women typically report more frequent and/or severe pain than men (Unruh, 1996). Our findings suggest that no important differences exist between men and women in the study. However, there might be a gender bias, as we know that more men than women develop leprosy. The inclusion of the differences

between men and women in how they view and cope with leprosy-related pain would add further knowledge to the overall concept of pain in leprosy.

9.2.5 Psychological co-morbidity and HRQoL

This study is the first to use validated structured instruments to evaluate the impact of pain on health-related quality of life and psychological state among leprosy patients with and without pain in India. The BPI interference subscale mean total score (5.29) was significantly higher in the patients with NP sub-group than the non-NP. This indicates that NP may cause substantial interference with the activities of daily living among leprosy patients. The domain of "Sleep" was identified as a major independent variable affected by pain in leprosy. The prevalence of psychological co-morbidity in this cohort was 66.3%. Our study has shown that anxiety and depression were highly prevalent in NP patients.

Overall, NP is associated with poor general health. In a French nationwide survey on the impact of NP on quality of life among 4,554 members of the general population, respondents who reported pain with neuropathic characteristics had a higher anxiety/depression score and higher degree of impairment in all dimensions relating to quality of life compared to those reporting pain without NP characteristics and those without pain (Attal et al., 2011a). Similarly, in a study from the UK population using SF-36 General Health Questionnaire, found that the quality of life was worse in the presence of NP than non-NP of the same severity (Smith et al., 2007). The same findings were also confirmed in a recent systematic review of health utilities on NP (Doth et al., 2010). The authors also found that the intensity of NP was more important in determining the extent of its health impact. Cognitive behavioural implications are important in the management of patients with chronic neuropathic pain (Daniel et al., 2008).

9.2.5.1 Health-related quality of life (HRQoL)

For the BPI interference scores, a measure of how the patients' pain interferes with every day activities, the greatest impact of pain on leprosy patients was in terms of "normal work", "sleep", and "mood". Between patients groups, there were significant differences of pain on the domain of "general activity", "relations with other people", "sleep" and "enjoyment of life". Patients in the NP group with upper limb pain had significantly poor scores for all pain interference domains of the BPI except "mood" and "walking ability" compared to non-NP.

BPI interference has been previously measured in patients with leprosy NP in Ethiopia (Haroun et al., 2012). In our previous study, Ethiopian patients who had completed MDT within the past 18 months were selected. Results were compared with non-NP, and found that NP interfered in the domains of "general activity", "normal work", and "enjoyment of life".

Patients with leprosy NP show interference from pain and reduced quality of life compared to patients without NP across most domains of the BPI. Findings were comparable to the impact of NP on quality of life in other infectious painful neuropathies, but NP in leprosy appears to be associated with greater disability and poor overall perception of general health (van Brakel et al., 2012). In a study by Serpell, who investigated the burden of PHN in 152 patients from Britain aged 50 years or older using BPI, the interference score revealed a substantial impact of pain on the domains of "enjoyment of life", "mood", and "sleep" (Serpell et al., 2014). Phillips and colleagues (Phillips et al., 2014), who studied 66 HIV-infected patients, also reported that patients with painful neuropathy had higher BPI interference scores compared to pain-free neuropathy. The high BPI interference scores found in our study indicate that patients' pain is associated with a negative impact on their day to day living and their quality of life. These findings support the growing evidence that patients with NP have a strong association with the domains of HRQoL (Jensen et al., 2007). This indicates that leprosy NP has produced an additional quality of life burden over that of leprosy itself or leprosy with painless nerve damage. Therefore, there is a pressing need to further investigate specific consequences in patients with leprosy and to assess these HRQoL domains in NP clinical trials.

9.2.5.2 Psychological co-morbidity

NP was found to be the most important factor independently associated with psychological co-morbidity using logistic regression analysis. This finding supports previous observations of leprosy patients from Ethiopia, India and Brazil, in whom the prevalence of psychiatric illness is high and the presence of pain was significantly associated with psychological co-morbidity (Haroun et al., 2012, Lasry-Levy et al., 2011, Reis et al., 2013). This indicates that the concomitant chronic NP pain may exacerbate

existing psychological morbidity and challenges patients' adjustment and coping with their life.

Our findings of increased depressive symptoms in patients with and without neuropathy appear to extend those of our previous study (Haroun et al., 2012), in which we found a linear relationship between pain and psychological distress. Our study controls for the presence of clinical evidence of neuropathy differentiated patients with and without pain. Our study found that even patients with subclinical neuropathy and pain had greater psychological co-morbidity. NP and psychological co-morbidity association was also found in HIV painful neuropathy (Wright et al., 2008, Phillips et al., 2014). These findings support the importance of NP, not only as a physical symptom, but as a possible cause of psychological co-morbidity. A study on HIV-associated neuropathy in South East Asia found that 20% of patients had sensory neuropathy and 36% had depression (Wright et al., 2008). It is possible that association with psychiatric comorbidity could be due to reverse causality, which cannot be fully discarded, since patients with psychological disturbance have been shown to have a decreased threshold for pain (Steer et al., 1993). The consistent and strong association found in this study between NP patients and psychological co-morbidity, suggests the need to find another way of understanding and treating NP. The utility of the biopsychosocial model developed by Novy and colleagues (Novy et al., 1995), which showed growing empirical support and acceptance, would be useful for understanding and treating leprosy NP. In this model, both physical and psychological factors are believed to contribute to the experience of pain. These findings also highlight the importance of an in-depth profiling of NP that includes psychological factors, particularly if tricyclic antidepressants were to be tested for efficacy in treating NP.

9.3 General discussion

Pain is a common problem among leprosy patients in clinical practice particularly in endemic countries. Leprosy-associated NP, which occurs as a result of persistent nerve damage in patients successfully treated with MTD, is now being recognised as an important long-term complication. Our leprosy group at LSHTM has recently shown in India and Ethiopia that 17-20% of treated leprosy patients cured of their infection but left with peripheral nerve damage, have significant NP (Lasry-Levy et al., 2011, Haroun et al., 2012). NP in leprosy leads to impaired quality of life, and increased use of health care, and is associated with more co-morbidity such as anxiety, depression, and sleep disturbances compared to non-NP (Schmidt et al., 2009). For individuals, life with such pain can be disabling even after their disease has been "cured".

The treatment of patients with NP remains a challenge (Finnerup et al., 2010). There are no disease-modifying therapies for NP yet. The current treatments are symptom control based, notably associated with variability of treatment response among patients with pain. For instance, patients with HIV-SN do not respond to pregabalin or amitriptyline treatment whereas many of those with other neuropathies do respond to these drugs. This may be explained by the heterogeneity of pain at clinical presentation, different underlying pain type or patterns and different aetiologies. In leprosy, pain is also heterogeneous with different clinical presentations, and associated with varying pain types and mechanisms. Different types of patients can be distinguished: those with pain associated with reactions "nociceptive pain" and those with NP; overlap of the two categories may occur. Improvement of treatment outcomes requires identification of such differences and the appropriate classification of patients with leprosy pain conditions. Recent developments in the treatment of NP have proposed that the somatosensory characterisations of these patients, with respect to distinguishing the type of pains, is of therapeutic relevance as NP requires a different treatment approach to non-NP (Baron et al., 2010a, Baron et al., 2012, Freeman et al., 2014). Recent developments have also shown that even within one underlying cause, the somatosensory profiles have different subgroups of patients with NP (Baron et al., 2009, Maier et al., 2010). In patients with radiculopathy, Baron and colleagues (Baron et al., 2010b) found that the individual differences in response to NP treatment were due to differences of somatosensory profiling, which may be associated with different underlying pain.

A recent guideline on the assessment of NP by IASP recommends the use of clinical tools such as NP screening questionnaires and QST testing (Haanpaa et al., 2011a). Unlike for NP caused by infectious diseases such as PHN (Pfau et al., 2014) and HIV (Phillips et al., 2014), there has been little work on the assessment of leprosy NP. The impact of NP in leprosy is also not well represented in the literature. The overall aim of this thesis was to accurately characterise patients with and without neuropathy; and patients with and without pain, with particular emphasis on pain and neuropathy. Pain subgroups were chosen as they have commonalities in their clinical pain characteristics.

In this section, the classification of pain in leprosy, the application and utility of the quantitative sensory testing used in this thesis for the somatosensory characterisation of leprosy patients with NP and interpretation of somatosensory profiling, is the focus for general discussion. The strengths and limitations of the study; reflective thoughts of how I would do the work if starting now, implications of the current work, and suggestions for future work, will also be presented.

9.3.1 Classification of pain in leprosy

This study used a recent NP definition and guidelines published by the IASP. However, there is currently no proper systematic classification of NP in leprosy, which led to inconsistency in the thesis. The definitions of the IASP were simple to use at the start of the study, but became difficult when I applied them. For the practical difficulties see the LePaIn flow chart (Figure 8.1). In particular, the definition and classification of neuritis in leprosy was practically difficult. According to the IASP definition, neuritis is inflammation of a nerve, but if this causes nerve damage then the pain, is by definition, neuropathic. In the field of leprosy, neuritis is defined as an acute loss of function (Wagenaar et al., 2012). This loss of function could be without pain, and that is why we regularly use the monofilaments and MRC grading system to identify patients who have got silent neuritis. It is not clear whether this neuritis is an acute neuropathic pain or neuropathic pain of different type. This indicates that a review of the classification of pain in leprosy is needed.

9.3.2 Somatosensory profiling

Quantitative sensory testing is a sophisticated measurement which assesses psychophysical responses to systematic and quantifiable sensory stimuli for the purpose of characterising somatosensory profiling. It can be simply described as follows: an increasing and quantified sensory stimulus is applied to an individual, they are asked with specific questions to report either a perception or pain threshold. Therefore, participants' responses are potentially influenced by psychological components (Backonja et al., 2013, Backonja et al., 2009, Shy et al., 2003). In addition, there are different methods for both the stimulus application (such as mechanical stimuli, electrical stimuli, or contact heat stimuli) and for the response measures, which can be simple measures like pain threshold and intolerance or more complex processes, like temporal summation and conditional pain modulation. Furthermore, demographic factors (age and gender), site of test, environment, training of the instructor and instructions given to participants have an impact on QST measures (Magerl et al., 2010). Hence, standardisation of testing protocol is required to facilitate validity and comparison of QST data between studies. The German Network on NP has developed such a standardised DFNS-QST protocol (Rolke et al., 2006a, Rolke et al., 2006b). It has proved to be a useful tool in identifying the underlying sensory abnormalities in each patient groups and in identifying the differences between the groups in regard to their underlying pain type. The use of QST profiling was important to distinguish groups because the occurrence of nerve damage does not necessarily mean that any pain is neuropathic in origin (Landerholm et al., 2010). The DFNS-QST protocol was employed in this thesis.

In our study, I found that obtaining full DFNS-QST profiles in leprosy patients provided new insights beyond studies that only used thermal testing. Initially I considered whether leprosy patients or group of patients differ in their QST profiles from people who are not experiencing pain, and I found that a variety of differences occurred. The QST findings revealed profound signs of sensory loss through increased thermal detection thresholds (CDT, WDT) and increased mechanical and vibration detection thresholds (MDT, VDT). Also the findings showed a variety of thermal (HPT/CPT) pain stimuli and pricking mechanical pain (MPT, MPS). The QST findings of the differences between leprosy patients and healthy controls were statistically significant, except VDT, WUR and VDT; with patients showing higher sensory loss than pain-free controls. These patterns are the same for patients with and without pain. Thus, it can be seen that I have several QST parameters that distinguish leprosy patients with clinical evidence of neuropathy from patients with no evidence of neuropathy and controls. Another issue that I considered is the possibility to subgroup leprosy patients with pain based on their responses to QST testing. I referred to a development by the DFNS group, who studied 1236 patients with NP due to different diseases and categorised their findings according to sensory loss and/or gain. They found that a minority of patients showed no sensory changes in their profile, but most showed some evidence of sensory perturbation: some showed only loss or negative signs, while some patients showed only gain or the positive signs. The remaining patients showed a combination of loss and gain sensory profile. The authors found that all subtypes are presented within each diagnosis. They concluded that if this reflects different pain mechanisms, a single pain treatment within the diagnosis is not helpful. Fortunately, they found commonalities across pain diagnosis. For instance, some patients with trigeminal neuralgia showed only sensory gain profile, while patients with polyneuropathy showed only sensory loss. Therefore, the authors concluded that these profiles might be better targets for treatment than the diagnosis of the condition. In the present study, I found that the somatosensory profile of leprosy NP could be categorised into two main subgroups. The first one is a patient who shows signs of predominant sensory loss only, where the profile shows signs of sensory loss through increased thermal detection thresholds (CDT, WDT and TSL) and increased mechanical and vibration thresholds (MDT and VDT), but no signs of sensory gain (hyperalgesia and/ or allodynia). The second group is patients with a combination of symptoms and signs. In this subgroup, the profile shows both sensory loss and gain. The pain and temperature are profoundly impaired, but the condition is associated with light mechanical touch pain (dynamic mechanical allodynia) occurred in 5.3% and WUR 15.8%.

Finally, I would like to discuss the implication of these findings along with a recent clinical case report published by Baron and colleagues (Westermann et al., 2012), which I think points towards future work. They report on a patient with bilateral burning and prickling pain in the T9-11 (at-level pain) following spinal cord injury. Pain on both sides was described the same way by the patient; burning, pricking, and severe in nature. On the right side, the QST findings showed a normal sensory profile and cold hyperalgesia, suggesting central NP mechanism. On the left side, there was a loss of thermal and mechanical sensation, suggesting peripheral nerve damage mechanism (deafferenation). There was also more loss of IENFD on the left side compared to the right. The patient was treated with pregabalin, and the result was unilateral pain relief only in the area with remaining sensory function, but not the pain on the left. In spite of

the fact that this a single case report, I think the scenario could also be relevant in leprosy patients with NP. This somatosensory profile for leprosy NP may assist researchers in designing clinical trials for targeting more specific management for these patients. I recommend that leprosy patients with dominant sensory loss profile be treated differently from patients who have combined sensory loss and gain.

9.3.3 Strengths and limitations

The principle strength of the current study is the comprehensiveness of the profiling measures employed to investigate NP in patients with leprosy. In contrast to studies using QST as a stand-alone test for characterisation, this study explored the multiple facets of NP including symptoms and sensory signs, metabolic dysfunction and psychological state. This facilitated the differentiation of subgroups of NP in leprosy pain and provided valuable new insights for treatment.

Limitations to this study related to the recruitment of participants; choice of assessment instruments; communication using patient's pain terminology; and QST-related methodological considerations.

A relatively small number of participants with NP were enrolled, particularly patients with sensory gain. As the ulnar nerve is the most commonly affected by leprosy, this site was deliberately chosen for the QST test of the C8 dermatome. However, this limited the recruitment of patients with NP. Patients with leprosy experience pain in multiple locations. During testing it was found that some patients had painful areas that were not precisely confined to the QST test site (multiple pain areas). For instance, patients with NP over treated skin lesion or skin lesion over the dermatome C8 proved extremely difficult to recruit. Out of 36 clinically examined patients with leprosy and pain, four patients had skin lesion pain over ulnar territory. Of these four patients, only one fulfilled the criteria of our QST site, which may have been insufficient for analysis. QST testing over the affected area in these patients may have been useful. Similarly, it was found that pain associated with light mechanical touch (allodynia) was confined to the ulnar nerve territory, but not QST tested site. In retrospect, including assessment of the entire ulnar nerve territory in our study would have been useful. As the sample size for patients with combined sensory gain and loss was small, any subgroup in the statistical analysis was not necessarily significant, due to loss of power. This meant that the logistic regression model had a very wide CI (95%). The small sample size also limits the conclusions that can be drawn from the impact of NP on health related-quality of life. A larger subgroup of patients with combined sensory gain and loss was necessary for statistical analysis.

A potential limitation also relates to the sampling process used in the study. Since more than 80% of the study cohort were selected from one centre this may have introduced bias and may reduce the generalisability of our results to the larger population of patients with leprosy. Similarly, patients with leprosy reactions were over-represented in our study. Those patients were attending leprosy service at BLP clinic and therefore accessible to our recruitment. This could be a source of recruitment bias. In addition, it was less likely to recruit cured patients who have no ongoing symptoms of nerve damage and will not be attending a clinic. It would have been ideal to recruit patients from different study sites.

In this study the diagnosis of leprosy neuropathy was based on clinical evidence; no additional techniques were used to investigate neuropathy. This study used Semmes-Weinstein monofilaments to evaluate abnormal sensory nerve function. Semmes-Weinstein monofilaments have limited sensitivity and specificity; to reach the definitive diagnosis of neuropathy, an additional measure of abnormal nerve function such as NCS would have contributed to the accurate identification of neuropathy and increased the accuracy of the diagnosis of NP. Measures of abnormal finding by NCS were used only for patients in the subclinical group as part of the TENLEP study (Wagenaar et al., 2012); to carry out more detailed NCSs in such a setting is challenging. In addition, autonomic nervous testing of nerve function was not performed in this study; this might have increased the number of patients with neuropathy.

Another potential limitation related to the health-related quality of life instruments. Recent recommendations in the IASP guidelines concerning NP assessment (Haanpaa et al., 2011a), suggests using a generic HRQoL measure such as medical outcome short survey (SF-36). This study used the BPI measure which while also recommended by the IASP, is a condition-specific instrument appropriate for detection of treatment response, whereas the SF-36 measure is suitable for evaluating the impact of pain on the common elements of health (Brazier et al., 1992).

In this study the diagnosis of psychological disorder was based on assessment by GHQ-12 questionnaire; no additional techniques were used to assess anxiety and depression. In addition, the influence of pain catastrophising on impact of NP is not considered (Phillips et al., 2014). It influences chronic perceptions of one's pain (Lame et al., 2005). Such data may have further supported the interpretation of the impact of leprosy NP on quality of life and allowed for comparison with the quality of life changes in other pain conditions; any specific consequences; and whether or not the findings really contributed to overall health. This study used GHQ-12 to assess symptoms and signs suggestive of depression (i.e. a score of three or more); for more objective diagnosis of anxiety syndromes and depression an additional measure of clinical criteria would have contributed to the accurate identification of psychological status.

Communicating with patients regarding their pain complaints was difficult in this study. The use and interpretation of the word "pain" is dependent on local cultural context and language, which changes from one setting to another. For example, many people in South Africa and India interpret the word "numbness" to mean "nothing" (Haroun, 2014). Additionally, communicating through a third person increases the risk of misunderstanding. To overcome this, communication was limited to a few sentences at a time and detailed examples were used to illustrate meaning. This strategy was developed from prior experience in communicating with leprosy patients through a third person in a study from Ethiopia, which helped to convey the desired message during the current study.

Another potential limitation also related to the fact that patients were using medication for their pain relief, which may contribute to variability in responses obtained and may decrease levels of pain. This may be considered as a confounding factor in the data analysis.

The main limitations of the QST were the onerous technical requirements. The time needed to complete the full DFNS-QST protocol during our study was extensive; it took 30 minutes to complete a single QST test on one ulnar site and one hour for both. In addition, the questionnaires and clinical examination assessment completed prior to QST, required an additional 45-60 minutes.

Other methodological design limitations were related to the QST procedure: the high number of tests, and methods of limits. While QST is used to assess somatosensory function thresholds using 13 parameters, perhaps not all these parameters provide additional information compared with standard assessment of pain patients. If the QST protocol were abbreviated, the duration of the procedure would shorten. Specifically, excluding mechanical pain sensitivity would have made testing easier and faster. The methods of limits algorithm used in this study provided highly reliable threshold results, but took a long time to complete, particularly in patients with long standing disease of the hand. In such situations, the test could take more than an hour per site to complete.

Another limitation of the QST method is that the repeated testing in one area could lead to sensitisation that would confound the QST-Data results. In study from Germany by Grone and colleges (Grone et al., 2012), who investigated the effect of testing order on the results of QST. Twenty healthy subjects were tested twice, 1 week apart with 2 different QST testing orders: the standardized testing order according to the German Research Network on Neuropathic Pain and a modified testing order in which mechanical stimuli were applied before thermal stimuli. The authors found that preceding mild thermal stimulation might lead to a sensitization to mechanical stimuli and thus to mechanical hyperalgesia. Alternative habituation mechanisms in the modified testing order resulting from repeated pinprick stimulation at the beginning should also be debated. I personally, believe that in theory we should do thermal testing second, but since the DFNS-QST protocol has always done in the other order it is best to keep to that.

Another potential limitation also relates to the fact that QST uses psycho-physical methods which require the attention and cooperation of the subject. For instance, the subject's concentration, motivation, reaction time and ability to respond quickly using the hand clicker can influence the results. Hence, the subjective character of the data collected through this method reduces acceptance. In addition, our cohort expectation was high in terms of demonstrating their problem. Participants may exaggerate the response and bias towards a bad outcome in order to get more attention and treatment. However, there is no algorithm of psychophysical testing that can reliably overcome the bias toward showing abnormality found in patients who wish to demonstrate more disability than they have, for whatever reason (Dyck et al., 1998). These observations emphasise the limitation of QST which cannot be considered as a single test to provide full somatosensory profile in patients with NP (Pfau et al., 2012), but it should be thought of as an additional tool to map the area of interest in terms of standard bedside sensory testing (Hansson et al., 2007).

Despite these limitations, our study used stringent criteria for NP, which in turn proved that the diagnosis of NP may be considered optimal. The study still provided an indepth characterisation of somatosensory profile of patients with leprosy NP and view of the impact of pain upon quality of life.

9.3.4 Practical considerations in doing research in resource-limited settings

9.3.4.1 QST battery high technical requirements

Nerve damage is a frequent complication in leprosy patients with and without pain. The DFNS-QST battery provides a comprehensive assessment of this impairment; however, the practicality of using such techniques in resource-limited settings were, prior to this study, unknown. The DFNS-QST protocol and skin biopsy were used to characterise somatosensory profiles in 90 leprosy patients and 52 healthy volunteers in India. The ulnar nerve territory (dermatome C8) was tested. The QST findings were comparable with a study on HIV painful neuropathy done in London (Phillips et al., 2014). This study suggests that the use of DFNS-QST protocol in resource-limited settings is technically feasible. While I was not sure of the DFNS-QST protocol feasibility in resource-limited settings, it appears to be valid because the findings from our study were consistent with the DFNS database and other studies (Rolke et al., 2006a). However, there were some practical considerations with environment, local idiom, and logistics.

The testing environment caused the greatest challenge. For instance, ambient room temperature (around 31.2 °C in Mumbai during August 2012) is critical for the MSA thermal stimulator machine, as well as maintaining temperature distribution across the thermode surface over the tested skin site. In addition, some of devices, such as, pinprick are highly sensitive to dust. Furthermore, the DFNS-QST protocol required patients' concentration, which in turn depends on the size and quietness of the room. Given the high number of patients and associated co-patients in the leprosy clinic, problems with noise were unavoidable.

Although our study was carried out in a well-established centre, the Foundation for Medical Research, the erratic electricity supply still remains a potential limitation to the feasibility of using QST. We encountered frequent loss of electricity supply and power cuts due to overburden of the system and the thermal electrode of the QST device is sensitive to fluctuating electric current. To re-calibrate the device after sudden power cut was time consuming. Another potential challenge to the feasibility was the lack of knowledge of the local idiom. Although the full DFNS-QST protocol was translated and back translated into the local languages (Hindi and Marathi), the need for knowledge of local social conditions is considerable. Training on social conditions (local idiom) may encourage the introduction of novel technology into a new setting.

Finally, logistics and costs must be considered when implementing QST in resourcelimited settings. The rules and regulations for the importing of machinery and the bureaucracy of taxation are logistically extensive. The initial set up cost of approximately \pounds 15 000 is considerable. In addition, the QST kit requires maintenance after being established, adding to the expenses.

This study is the first to present a detailed assessment of the challenges facing the use of the DFNS-QST protocol in a developing setting, and validates the Mumbai centre for the use of QST. The lessons learned with this particular setting of patients emphasise the importance of understanding the local language and customs.

The way forward in minimizing the QST high technology requirements is better knowledge of local-society, and development of resources and tools. Without human resources, willingness to learn and understanding of the benefit, it will be difficult in practice. The identified challenges may be overcome through improved capacity and increased motivation. Potentially the development of battery operated QST, would overcome other challenges.

9.3.4.2 Tissue biopsy and IENFD challenges

Intra-epidermal nerve fibre density (IENFD) is a technique for measuring the endings of small peripheral nerve fibres in the epidermis. It provides anatomical data regarding the sensory nervous system, whereas QST provides functional/physiological data. Both are useful and complementary tools. IENFD has been used for identifying the presence of nerve damage in other peripheral neuropathies (Haanpaa et al., 2011a). IENFD in affected skin is measured by a skin biopsy taken from patients with suspected small fibre neuropathy (Sommer and Lauria, 2007, Holland et al., 1998). Skin biopsy (3mm punch biopsy) is a safe and reliable technique and has therefore become a widely used tool to investigate IENF (Lauria et al., 2005). This study was designed to assess the IENFD in leprosy patients in line with the recent NeuPSIG guidelines on the assessment of NP (Haanpaa et al., 2011a). I hypothesised that skin biopsy examination of patients with NP would show significant loss of intraepidermal fibres in affected sites compared to asymptomatic sites.

A total of 200 biopsy samples were collected: 170 samples from 85 patients and 30 samples from 30 healthy participants. Each sample was divided into 2 pieces. These samples were processed and stored frozen at -80 °c in FMR. While it was planned to transfer the samples to Oxford, UK, the administrative practicality of this process proved difficult within the timeframe of the study. Tissue transfer permission is not granted, and the decision on how to process the biopsy is under review.

9.3.5 If I had to start now, this is what I would do differently

I qualified in using the DFNS-QST protocol application and now I have data from my leprosy study. I had a lot to learn in the beginning, but I quickly learned what I needed, in particular to apply this advanced technology in a resource-limited settings. However, I always felt that if I worked in leprosy neuropathy and pain, I would have started off with an in-depth foundation. I am happy with what I obtained, but if I could start over again, I would probably chose to have a shorter protocol with more patients. It would have been useful to develop a simpler version of the DFNS-QST protocol that could be applied in routine clinical examinations to diagnose small-fibre neuropathies, requiring less psychophysical patient collaboration, having lower cost, maintaining features of high accuracy and rapidity, and with applicability to poor psychomotor performance. Such a tool would be more practical, particularly in resource-limited settings.

In hindsight, the use of the QST protocol according to the DFNS represents a comprehensive protocol of somatosensory profiling and was very useful in leprosy; however, identification of new neurological abnormalities to differentiate patients with and without pain is not yet possible. It would have been useful if to examine other possibilities such as skin biopsy and IENF density.

In hindsight, the use of the BPI measure is appropriate for detection of treatment response in patients with NP. It would have been useful if to use the SF-36 measure for evaluating the impact of pain on HRQoL.

The recruitment of leprosy patients with NP to the study was not representative of all types of pain in those patients (i.e. pain in nerve affected territory and pain in skin lesions). It would have been useful if I had powered the study to the NP patients in each subgroup. This would have given a sample size of 15 patients with NP in each.

I wish I had tested the entire ulnar nerve territory and/or other pain location. This would have given further positive sensory gain findings. Similarly testing the QST parameters in the skin lesions would have been of great value.

9.3.6 What contributions has this study made to our knowledge?

In the literature I reviewed, I found that while the concept of sensory profiling using tools such as QST was well established in NP field, it was not evident in leprosy NP, hence there has been a contribution to knowledge in the later practice through this study.

This is the first study using the DFNS-QST protocol in a resource-limited setting and validates the Mumbai Centre's for the use of the QST. Normative QST data are generated by evaluating somatosensory function in healthy volunteers, a process in which one body area is assessed using the QST measures according to the DFNS protocol. This study contributed to the DFNS reference database with normative data from the dermatome C8 of non-Caucasian population.

To facilitate the identification of leprosy patients with NP, a simple case definition was introduced in this thesis using the DN4 questionnaire and clinical examination, by which the patient's pain could be categorised as NP or non-NP pain. This case definition aimed to identify a leprosy patients with NP in a resource-limited setting. Our findings showed the utility of this simple case definition, and it could be introduced in other leprosy populations.

Furthermore, the QST investigative tool methodology for sensory profiling, is still emerging as a field of research; therefore, new questions investigating practice have come from this research. Firstly, why do leprosy patients with pain not have significantly different QST profiles than leprosy patients without pain? This observation is interesting given recently increased use of QST worldwide. One could argue that the way this study was designed and implemented actually caused this results, but because this study was three-way case controlled with age and gender matched control (i.e. controls were selected correctly from the study population), this is not likely. All methods were carefully planned to improve the study design. In addition, the QST testing was carried out by a trained and qualified candidate according to the DFNS criteria (Geber et al., 2009), ensuring that protocol of QST laboratories was properly used. The second concern of the investigating practice being proposed by thesis, is: how specific are QST measures in identifying phenotypical abnormalities of leprosy patients with neuropathy that is associated with pain? In this context, if investigating somatosensory changes contribute to the presence or absence of pain could also be valuable.

In the current study the QST measures did not identify new abnormalities for patients with and without pain. Similar findings were also obtained by Phillips in HIV-related painful neuropathy (Phillips et al., 2014), unlike other studies in patients with pain, where the QST tool has identified abnormalities. Maier and colleagues studied sensory abnormalities in 1236 patients with NP due to different underlying diseases (Maier et al., 2010). In this large cohort of patients, DFNS-QST measures showed that 92% of all patients with proven neuropathy had at least one sensory abnormality compared with the contralateral unaffected body area or with the reference data obtained from healthy controls. Pfau and colleagues also showed the applicability of the DFNS-QST protocol in identifying abnormalities in PHN (Pfau et al., 2014). In all these studies, age and gender match healthy controls were designed. These studies were therefore similar to ours in study design. It is suggested that a shorter protocol in more leprosy patients may be the best options for leprosy NP. In this context, the current study has shown that investigating leprosy patients using the DFNS-QST protocol was time-consuming and demanding for both-investigators and patients. The protocol needs to be simplified to become a regular screening tool in resource-limited settings. Therefore, there is a pressing need to identify the most sensitive QST measures to determine somatosensory abnormalities for each NP entity in leprosy.

The findings from the current study will contribute to clinical practice in leprosy NP. I demonstrated that, by using the DFNS-QST measures from the healthy controls, the interpretation of sensory findings for patients with NP may be different compared to an un-affected contralateral side, which is usually used as a reference in clinical practice. I have shown that in leprosy patients with unilateral NP, bilateral sensory changes occur too. This observation of bilateral sensory abnormalities in leprosy patients with

unilateral NP is of great importance clinically, because it indicates that the mirror unaffected side should be used carefully as a reference side in sensory examinations in clinical practice (Konopka et al., 2012). In addition, this work validate thermal abnormalities and preserved JPS and vibration in leprosy patients. This may help clinicians to differentiate leprosy from other peripheral neuropathies; if vibration and/or JPS are preserved in patients with peripheral neuropathy, think of leprosy.

Through my doctoral investigation, although I have not found differences between patients with and without pain in terms of their detailed sensory profiles, this tells us that another tool is needed to identify neurological abnormalities in those patients. It could be said that my thesis made a contribution to knowledge about the implications of the current work (i.e. the investigation approach of leprosy NP). Thus, tools such as skin biopsy and its IENF density are now being proposed by this piece of work since it is vital to explore other possible differences. IENF density provides anatomical data regarding the sensory nervous system, whereas QST provides functional and physiological data. Therefore, doing the biopsies is very important, as it might indicate what other tools need to be used in future.

Another contribution from this work that still needs further research is the identification of patients who are at risk of developing pain. This work highlights the question, as temperature abnormalities are a marker of neurological abnormality in pain, why do all leprosy patients not have pain? Furthermore, the thermal abnormality found in this study indicates the need for a field friendly temperature testing, for example apps applications. This will help early detection and treatment of neuropathy in leprosy.

The findings of our profiling measures in leprosy neuropathy are also of relevance for routine clinical use and clinical trials in resource-limited settings. For instance, the selection of sub-groups based on specific QST parameters for the clinical evaluation of drugs will improve trial sensitivity. Moreover, including simplifying sensory and symptom profiles of patient responses to various sensory stimuli such as heat and pressure will empower post-hoc analysis of responders/non-responders, which will then be used to enable efficient prescribing to patients likely to respond to the drug when the intervention is introduced into clinical practice. In addition, knowledge gained in profiling this patient population could also help to determine a mechanism-based therapy for NP.

Finally, the study contributions included development of skills of the health facility staff at collaborating centres. They improved their skills in identifying leprosy patients with NP using clinical examinations and highly specialised techniques. These are generic skills that can be transferred to other settings in India, including non-leprosy NP diseases such as diabetic and HIV-painful neuropathy. In addition, this study may have an impact on policy makers, as there are many governmental and non-governmental leprosy experts in India who can contribute to policy guidelines. I think even highlighting the issues around NP in leprosy would move quickly into priority in treatment. Hence, this work contributed to the academic work in India and policy benefits will follow.

9.3.7 Future perspectives

The DFNS-QST protocol enables a standardised approach to be used when assessing patients with NP. A simple and robust diagnostic tool in identifying neurological abnormalities in leprosy patients with NP can be developed. This would help to answer the question, "why do leprosy patients with pain not have significantly different profiles than leprosy patients without pain?" or "who is at risk of developing leprosy NP?" The tool would help to accurately identify patients for both large epidemiological studies in resource-limited settings and for future clinical trials.

The identification of stratified sub-groups of leprosy patients with NP through this doctoral thesis has established important steps for the future therapeutic efficacy approach. However, findings are not yet sufficient. New therapeutic concepts based on sub-group characteristics of NP leprosy patients with a dominant sensory loss profile and patients who have combined sensory loss and gain need to be developed. Possibly, this sub-grouping approach needs further modification to better assign patients to interventions, but indeed, the two sub-groups should receive different treatments.

Chapter 10 Conclusions and Recommendations

10.1 Conclusions

This study aimed to establish the somatosensory characteristics of leprosy patients with persistent neuropathy, both with and without neuropathic pain. The QST parameters were effective in detecting neuropathy, but were not able to distinguish between patients with and without neuropathic pain. A major finding of this thesis demonstrates that leprosy patients with persistent neuropathy have a unique somatosensory profile compared to other conditions.

10.2 Recommendations

Based on the research findings presented in this thesis and the acknowledged limitations, the following steps are recommended:

- 1. A clear classification of leprosy related neuropathic pain should be developed. Neuropathic pain can present in a number of ways in leprosy. For instance, acute pain may be the presenting symptom in a patient with leprosy who is seeking treatment. In this setting, the pain may be nociceptive in nature, usually due to reactions. However, the type of acute pain from neuritis or other leprosy nerve involvement is not well classified. Acute pain in leprosy can be classified into acute pain with reactions and neuropathic pain. The classification should also involve the newly developed grading system for the assessment of neuropathic pain (Treede et al., 2008). The grading system of neuropathic pain proved to be useful for the identification of neuropathic pain in patients with neck-arm pain (Tampin et al., 2013). Leprosy patients with pain can be classified as having probable, possible, or definitive neuropathic pain. This would add further help to the identification of leprosy related neuropathic pain.
- 2. Understanding of the pathophysiology of leprosy neuropathy and in particular, the unique feature of loss of one $A\beta$ fibre mediated sensory modality (mechanical detection threshold) accompanied by preservation of another (vibration detection threshold) requires further elucidation which will have physiological implications for understanding other conditions as well as the opportunity to develop specific diagnostic tools.

- 3. Future research should investigate the methods of self-reporting pain in determining the prevalence and impact of neuropathic pain in leprosy patients using pain questionnaires. Self-reported pain relies on two factors: the description questionnaire used and cultural barriers to reporting or not reporting pain. Leprosy is global disease, but the same terms cannot be used in different settings.
- 4. Psychological distress was found significantly higher in leprosy patients with neuropathic pain than those with painless neuropathy, as well as the additional quality of life burden that leprosy neuropathic pain has produced over that of leprosy itself or leprosy with painless nerve damage. Future studies should consider these observations for the identification and treatment of neuropathic pain in leprosy.
- 5. Future research should include the development of new tools for identifying leprosy neuropathic pain to inform the appropriate treatment. Given that the full QST battery is beyond the feasibility of most centres in poorer leprosy-endemic countries, a simple and robust diagnostic tool is required. The new tools should include a specific QST parameters such as thermal and vibration testing and requiring less psychophysical patient collaboration, having lower cost, maintaining features of high accuracy and rapidity, and with applicability to poor psychomotor performance. Such tools need to be validated, tested, and deployed in clinical settings. Once established as valid and reproducible, tools for neuropathic pain may be used to identify patients for clinical trials that may be developed in the future.
- 6. A future study should consider a stratified grouping of somatosensory findings: patients with a dominant sensory loss profile and patients who have combined sensory loss and gain, for a future therapeutic efficacy approach. The differentiation between the somatosensory profile of sensory loss and sensory gain could help to predict responses to treatment.

References

- ABBOT, N. C., BECK, J. S., MOSTOFI, S. & WEISS, F. 1996. Sympathetic vasomotor dysfunction in leprosy patients: comparison with electrophysiological measurement and qualitative sensation testing. *Neurosci Lett*, 206, 57-60.
- ASSOCIATION, B. M. 2011. British National Formulary [Online]. Available: <u>http://bnf.org</u>..
- ATTAL, N. 2000. Chronic neuropathic pain: mechanisms and treatment. *Clin J Pain,* 16, S118-30.
- ATTAL, N., FERMANIAN, C., FERMANIAN, J., LANTERI-MINET, M., ALCHAAR, H. & BOUHASSIRA, D. 2008. Neuropathic pain: are there distinct subtypes depending on the aetiology or anatomical lesion? *Pain*, 138, 343-53.
- ATTAL, N., LANTERI-MINET, M., LAURENT, B., FERMANIAN, J. & BOUHASSIRA, D. 2011a. The specific disease burden of neuropathic pain: results of a French nationwide survey. *Pain*, 152, 2836-43.
- ATTAL, N., PERROT, S., FERMANIAN, J. & BOUHASSIRA, D. 2011b. The neuropathic components of chronic low back pain: a prospective multicenter study using the DN4 Questionnaire. *J Pain*, 12, 1080-7.
- BACKONJA, M., WALLACE, M. S., BLONSKY, E. R., CUTLER, B. J., MALAN, P., JR., RAUCK, R., TOBIAS, J. & GROUP, N.-C. S. 2008. NGX-4010, a highconcentration capsaicin patch, for the treatment of postherpetic neuralgia: a randomised, double-blind study. *Lancet Neurol*, 7, 1106-12.
- BACKONJA, M. M., ATTAL, N., BARON, R., BOUHASSIRA, D., DRANGHOLT, M., DYCK,
 P. J., EDWARDS, R. R., FREEMAN, R., GRACELY, R., HAANPAA, M. H.,
 HANSSON, P., HATEM, S. M., KRUMOVA, E. K., JENSEN, T. S., MAIER, C.,
 MICK, G., RICE, A. S., ROLKE, R., TREEDE, R. D., SERRA, J., TOELLE, T.,
 TUGNOLI, V., WALK, D., WALALCE, M. S., WARE, M., YARNITSKY, D. &
 ZIEGLER, D. 2013. Value of quantitative sensory testing in neurological and
 pain disorders: NeuPSIG consensus. *Pain*, 154, 1807-19.
- BACKONJA, M. M. & LAURIA, G. 2010. Taking a peek into pain, from skin to brain with ENFD and QST. *Pain*, 151, 559-60.

- BACKONJA, M. M., WALK, D., EDWARDS, R. R., SEHGAL, N., MOELLER-BERTRAM, T., WASAN, A., IRVING, G., ARGOFF, C. & WALLACE, M. 2009. Quantitative sensory testing in measurement of neuropathic pain phenomena and other sensory abnormalities. *Clin J Pain*, 25, 641-7.
- BALAGON, M. V., GELBER, R. H., ABALOS, R. M. & CELLONA, R. V. 2010. Reactions following completion of 1 and 2 year multidrug therapy (MDT). Am J Trop Med Hyg, 83, 637-44.
- BANDYOPADHYAY, G., SINHA, S., SEN, B. & SEN, G. 1988. Validity of General Health Questionnaire (GHQ-36/GHQ-12) in the psychiatric O.P.D. of a general hospital--a pilot study. Int J Soc Psychiatry, 34, 130-4.
- BARON, R. 2006. Mechanisms of disease: neuropathic pain--a clinical perspective. *Nat Clin Pract Neurol*, **2**, 95-106.
- BARON, R. 2009. Neuropathic pain: a clinical perspective. *Handb Exp Pharmacol*, 3-30.
- BARON, R. 2011. Neuropathic pain-definition, mechanisms diagnosis and management. *Rinsho Shinkeigaku*, 51, 936.
- BARON, R., BINDER, A. & WASNER, G. 2010a. Neuropathic pain: diagnosis, pathophysiological mechanisms, and treatment. *Lancet Neurol*, 9, 807-19.
- BARON, R., FORSTER, M. & BINDER, A. 2012. Subgrouping of patients with neuropathic pain according to pain-related sensory abnormalities: a first step to a stratified treatment approach. *Lancet Neurol*, 11, 999-1005.
- BARON, R., FREYNHAGEN, R., TOLLE, T. R., CLOUTIER, C., LEON, T., MURPHY, T. K., PHILLIPS, K. & INVESTIGATORS, A. 2010b. The efficacy and safety of pregabalin in the treatment of neuropathic pain associated with chronic lumbosacral radiculopathy. *Pain*, 150, 420-7.
- BARON, R., TOLLE, T. R., GOCKEL, U., BROSZ, M. & FREYNHAGEN, R. 2009. A crosssectional cohort survey in 2100 patients with painful diabetic neuropathy and postherpetic neuralgia: Differences in demographic data and sensory symptoms. *Pain*, 146, 34-40.
- BARRETT, K. E., BARMAN, S. M., BOITANO, S. & BROOKS, H. L. 2010. PHYSIOLOGY OF NERVE & MUSCLE CELLS. Ganong's Review of Medical Physiology. 23rd ed.: McGraw-Hill Medical.

- BAUMGARTNER, U., MAGERL, W., KLEIN, T., HOPF, H. C. & TREEDE, R. D. 2002. Neurogenic hyperalgesia versus painful hypoalgesia: two distinct mechanisms of neuropathic pain. *Pain*, 96, 141-51.
- BELL-KROTOSKI, J. 1990. Pocket filaments and specifications for Semmes-Weinstein monofilaments. J Hand Ther, 3, 26-9.
- BENNETT, M. 2001. The LANSS Pain Scale: the Leeds assessment of neuropathic symptoms and signs. *Pain*, 92, 147-57.
- BENNETT, M. I., SMITH, B. H., TORRANCE, N. & POTTER, J. 2005. The S-LANSS score for identifying pain of predominantly neuropathic origin: validation for use in clinical and postal research. *J Pain*, 6, 149-58.
- BHAT, R. & KHANNA, N. 2006. Therapy of paresthesia in leprosy: A comparison of amitriptyline, gabapentin and vitamin B-complex. *64th Annual Meeting of the American-Academy-of-Dermatology.* San Francisco: Amer Acad Dermatol
- BHATIA M.S, CHANDRA R, BHATTACHARYA SN & M., I. 2006. Psychiatric morbidity and pattern of dysfunctions in patients with leprosy. *Indian J Dermatol* 51, 23-5.
- BIRD, S. J., BROWN, M. J., SPINO, C., WATLING, S. & FOYT, H. L. 2006. Value of repeated measures of nerve conduction and quantitative sensory testing in a diabetic neuropathy trial. *Muscle Nerve*, 34, 214-24.
- BIRKE, J. A. & SIMS, D. S. 1986. Plantar sensory threshold in the ulcerative foot. *Lepr Rev*, 57, 261-7.
- BOUHASSIRA, D., ATTAL, N., ALCHAAR, H., BOUREAU, F., BROCHET, B., BRUXELLE, J.,
 CUNIN, G., FERMANIAN, J., GINIES, P., GRUN-OVERDYKING, A., JAFARI-SCHLUEP, H., LANTERI-MINET, M., LAURENT, B., MICK, G., SERRIE, A.,
 VALADE, D. & VICAUT, E. 2005. Comparison of pain syndromes associated with nervous or somatic lesions and development of a new neuropathic pain diagnostic questionnaire (DN4). *Pain*, 114, 29-36.
- BOUHASSIRA, D., ATTAL, N., FERMANIAN, J., ALCHAAR, H., GAUTRON, M., MASQUELIER, E., ROSTAING, S., LANTERI-MINET, M., COLLIN, E., GRISART, J.
 & BOUREAU, F. 2004. Development and validation of the Neuropathic Pain Symptom Inventory. *Pain*, 108, 248-57.

- BOUHASSIRA, D., LANTERI-MINET, M., ATTAL, N., LAURENT, B. & TOUBOUL, C. 2008. Prevalence of chronic pain with neuropathic characteristics in the general population. *Pain*, 136, 380-7.
- BOVE, G. M. & LIGHT, A. R. 1997. The nervi nervorum: Missing link for neuropathic pain? *Pain Forum*, 6, 181-190.
- BOWLING, A. 2005. *Measuring health: a review of quality of life measurement scales*, Oxford.
- BRANDSMA, J. W. 2000. Monitoring motor nerve function in leprosy patients. *Lepr Rev*, 71, 258-67.
- BRANDSMA, J. W., WAGENAAR, I., POST, E., NICHOLLS, P. & RICHARDUS, J. H. 2014. Reliability of clinical nerve function assessment in peripheral neuropathies. *Lepr Rev*, 85, 29-35.
- BRAZIER, J. E., HARPER, R., JONES, N. M., O'CATHAIN, A., THOMAS, K. J., USHERWOOD, T. & WESTLAKE, L. 1992. Validating the SF-36 health survey questionnaire: new outcome measure for primary care. *BMJ*, 305, 160-4.
- BRIDGES, D., THOMPSON, S. W. & RICE, A. S. 2001. Mechanisms of neuropathic pain. *Br J Anaesth*, 87, 12-26.
- BRITTON, W. J. 1998. The management of leprosy reversal reactions. *Lepr Rev*, 69, 225-34.
- BRITTON, W. J. & LOCKWOOD, D. N. 2004. Leprosy. Lancet, 363, 1209-19.
- BRYCESON, A. & PFALTZGRAFF, R. E. 1990. Leprosy, 3rd edition; Edinburgh: Churchill livingstone.
- CALVO, M., DAWES, J. M. & BENNETT, D. L. 2012. The role of the immune system in the generation of neuropathic pain. *Lancet Neurol*, 11, 629-42.
- CAVENAGH, J., GOOD, P. & RAVENSCROFT, P. 2006. Neuropathic pain: are we out of the woods yet? *Internal Medicine Journal*, 36, 251-255.
- CHANTEAU, S., CARTEL, J. L. & ROUX, J. 1992. [Leprosy serology: current status and perspectives]. *Acta Leprol,* 8, 65-70.
- CHEN, S., QU, J. & CHU, T. 2012. Prevalence and characteristics of neuropathic pain in the people affected by leprosy in China. *Lepr Rev*, 83, 195-201.
- CHONG, P. S. & CROS, D. P. 2004. Technology literature review: quantitative sensory testing. *Muscle Nerve*, 29, 734-47.

- CHOO, P. W., GALIL, K., DONAHUE, J. G., WALKER, A. M., SPIEGELMAN, D. & PLATT,
 R. 1997. Risk factors for postherpetic neuralgia. Arch Intern Med, 157, 1217-24.
- CLEELAND, C. 2009. The Brief Pain Inventory. User Guide. The University of Texas M. D. Anderson Cancer Center, Houston, TX, USA.
- CLEELAND, C. S. & RYAN, K. M. 1994. Pain assessment: global use of the Brief Pain Inventory. Ann Acad Med Singapore, 23, 129-38.
- COPLAN, P. M., SCHMADER, K., NIKAS, A., CHAN, I. S., CHOO, P., LEVIN, M. J., JOHNSON, G., BAUER, M., WILLIAMS, H. M., KAPLAN, K. M., GUESS, H. A. & OXMAN, M. N. 2004. Development of a measure of the burden of pain due to herpes zoster and postherpetic neuralgia for prevention trials: adaptation of the brief pain inventory. *J Pain*, 5, 344-56.
- COSTIGAN, M., SCHOLZ, J. & WOOLF, C. J. 2009. Neuropathic pain: a maladaptive response of the nervous system to damage. *Annu Rev Neurosci*, 32, 1-32.
- CRAIG, A. D. 2003. Interoception: the sense of the physiological condition of the body. *Curr Opin Neurobiol*, 13, 500-5.
- CRAWFORD, B., BOUHASSIRA, D., WONG, A. & DUKES, E. 2008. Conceptual adequacy of the neuropathic pain symptom inventory in six countries. *Health Qual Life Outcomes*, 6, 62.
- CROFT, R. 2004. Neuropathic pain in leprosy. *Int J Lepr Other Mycobact Dis,* 72, 171-2.
- CROFT, R. P., NICHOLLS, P. G., STEYERBERG, E. W., RICHARDUS, J. H., CAIRNS, W. & SMITH, S. 2000. A clinical prediction rule for nerve-function impairment in leprosy patients. *Lancet*, 355, 1603-6.
- CROSS, H. 2014. A Delphi exercise to refine the WHO three-point disability grading system for leprosy, and to develop guidelines to promote greater accuracy and reliability of WHO disability recording. *Lepr Rev*, 85, 18-28.
- CRUCCU, G., ANAND, P., ATTAL, N., GARCIA-LARREA, L., HAANPAA, M., JORUM, E., SERRA, J. & JENSEN, T. S. 2004. EFNS guidelines on neuropathic pain assessment. *Eur J Neurol*, 11, 153-62.
- CRUCCU, G. & TRUINI, A. 2009. Tools for assessing neuropathic pain. *PLoS Med*, 6, e1000045.

- DANIEL, H. C., NAREWSKA, J., SERPELL, M., HOGGART, B., JOHNSON, R. & RICE, A. S. 2008. Comparison of psychological and physical function in neuropathic pain and nociceptive pain: implications for cognitive behavioral pain management programs. *Eur J Pain*, 12, 731-41.
- DAUT, R. L., CLEELAND, C. S. & FLANERY, R. C. 1983. Development of the Wisconsin Brief Pain Questionnaire to assess pain in cancer and other diseases. *Pain*, 17, 197-210.
- DAVIES, M., BROPHY, S., WILLIAMS, R. & TAYLOR, A. 2006. The prevalence, severity, and impact of painful diabetic peripheral neuropathy in type 2 diabetes. *Diabetes Care*, 29, 1518-22.
- DAVIS, K. D., TAYLOR, K. S. & ANASTAKIS, D. J. 2011. Nerve injury triggers changes in the brain. *Neuroscientist*, 17, 407-22.
- DEMANT, D. T., LUND, K., VOLLERT, J., MAIER, C., SEGERDAHL, M., FINNERUP, N. B., JENSEN, T. S. & SINDRUP, S. H. 2014. The effect of oxcarbazepine in peripheral neuropathic pain depends on pain phenotype: a randomised, double-blind, placebo-controlled phenotype-stratified study. *Pain*.
- DEVIGILI, G., TUGNOLI, V., PENZA, P., CAMOZZI, F., LOMBARDI, R., MELLI, G., BROGLIO, L., GRANIERI, E. & LAURIA, G. 2008. The diagnostic criteria for small fibre neuropathy: from symptoms to neuropathology. *Brain*, 131, 1912-25.
- DOTH, A. H., HANSSON, P. T., JENSEN, M. P. & TAYLOR, R. S. 2010. The burden of neuropathic pain: a systematic review and meta-analysis of health utilities. *Pain*, 149, 338-44.
- DOTSON, R. M. 1997. Clinical neurophysiology laboratory tests to assess the nociceptive system in humans. *J Clin Neurophysiol*, 14, 32-45.
- DWORKIN, R. H., O'CONNOR, A. B., BACKONJA, M., FARRAR, J. T., FINNERUP, N. B., JENSEN, T. S., KALSO, E. A., LOESER, J. D., MIASKOWSKI, C., NURMIKKO, T. J., PORTENOY, R. K., RICE, A. S., STACEY, B. R., TREEDE, R. D., TURK, D. C. & WALLACE, M. S. 2007. Pharmacologic management of neuropathic pain: evidence-based recommendations. *Pain*, 132, 237-51.
- DWORKIN, R. H., TURK, D. C., FARRAR, J. T., HAYTHORNTHWAITE, J. A., JENSEN, M. P., KATZ, N. P., KERNS, R. D., STUCKI, G., ALLEN, R. R., BELLAMY, N., CARR, D.
B., CHANDLER, J., COWAN, P., DIONNE, R., GALER, B. S., HERTZ, S., JADAD, A. R., KRAMER, L. D., MANNING, D. C., MARTIN, S., MCCORMICK, C. G., MCDERMOTT, M. P., MCGRATH, P., QUESSY, S., RAPPAPORT, B. A., ROBBINS, W., ROBINSON, J. P., ROTHMAN, M., ROYAL, M. A., SIMON, L., STAUFFER, J. W., STEIN, W., TOLLETT, J., WERNICKE, J. & WITTER, J. 2005. Core outcome measures for chronic pain clinical trials: IMMPACT recommendations. *Pain*, 113, 9-19.

- DYCK, P. J., DYCK, P. J., KENNEDY, W. R., KESSERWANI, H., MELANSON, M., OCHOA, J., SHY, M., STEVENS, J. C., SUAREZ, G. A. & O'BRIEN, P. C. 1998. Limitations of quantitative sensory testing when patients are biased toward a bad outcome. *Neurology*, 50, 1213.
- DYCK, P. J., KARNES, J. L., GILLEN, D. A., O'BRIEN, P. C., ZIMMERMAN, I. R. & JOHNSON, D. M. 1990. Comparison of algorithms of testing for use in automated evaluation of sensation. *Neurology*, 40, 1607-13.
- FACER, P., MANN, D., MATHUR, R., PANDYA, S., LADIWALA, U., SINGHAL, B., HONGO, J., SINICROPI, D. V., TERENGHI, G. & ANAND, P. 2000. Do nerve growth factor-related mechanisms contribute to loss of cutaneous nociception in leprosy? *Pain*, 85, 231-8.
- FACER, P., MATHUR, R., PANDYA, S. S., LADIWALA, U., SINGHAL, B. S. & ANAND, P. 1998. Correlation of quantitative tests of nerve and target organ dysfunction with skin immunohistology in leprosy. *Brain*, 121 (Pt 12), 2239-47.
- FARRAR, J. T., YOUNG, J. P., JR., LAMOREAUX, L., WERTH, J. L. & POOLE, R. M. 2001. Clinical importance of changes in chronic pain intensity measured on an 11point numerical pain rating scale. *Pain*, 94, 149-58.
- FIELD, M. J., COX, P. J., STOTT, E., MELROSE, H., OFFORD, J., SU, T. Z., BRAMWELL, S., CORRADINI, L., ENGLAND, S., WINKS, J., KINLOCH, R. A., HENDRICH, J., DOLPHIN, A. C., WEBB, T. & WILLIAMS, D. 2006. Identification of the alpha2delta-1 subunit of voltage-dependent calcium channels as a molecular target for pain mediating the analgesic actions of pregabalin. *Proc Natl Acad Sci U S A*, 103, 17537-42.
- FIELDS, H. L., ROWBOTHAM, M. & BARON, R. 1998. Postherpetic neuralgia: irritable nociceptors and deafferentation. *Neurobiol Dis*, 5, 209-27.

- FINE, P. E. 2008. Leprosy's global statistics--still room for improvement. *Lepr Rev,* 79, 235-8.
- FINNERUP, N. B., JOHANNESEN, I. L., FUGLSANG-FREDERIKSEN, A., BACH, F. W. & JENSEN, T. S. 2003. Sensory function in spinal cord injury patients with and without central pain. *Brain*, 126, 57-70.
- FINNERUP, N. B., OTTO, M., MCQUAY, H. J., JENSEN, T. S. & SINDRUP, S. H. 2005. Algorithm for neuropathic pain treatment: an evidence based proposal. *Pain*, 118, 289-305.
- FINNERUP, N. B., SINDRUP, S. H. & JENSEN, T. S. 2010. The evidence for pharmacological treatment of neuropathic pain. *Pain*, 150, 573-81.
- FISHBAIN, D. A., CUTLER, R., ROSOMOFF, H. L. & ROSOMOFF, R. S. 1997. Chronic pain-associated depression: antecedent or consequence of chronic pain? A review. *Clin J Pain*, 13, 116-37.
- FITE, G. L. 1951. The pathology and pathogenesis of leprosy. *Ann N Y Acad Sci*, 54, 28-33.
- FITZEK, S., BAUMGARTNER, U., FITZEK, C., MAGERL, W., URBAN, P., THOMKE, F., MARX, J., TREEDE, R. D., STOETER, P. & HOPF, H. C. 2001. Mechanisms and predictors of chronic facial pain in lateral medullary infarction. *Ann Neurol*, 49, 493-500.
- FLYNN, R., CHAPMAN, K., IFTINCA, M., ABOUSHOUSHA, R., VARELA, D. & ALTIER, C.
 2014. Targeting the Transient Receptor Potential Vanilloid Type 1 (TRPV1)
 Assembly Domain Attenuates Inflammation-induced Hypersensitivity. J Biol
 Chem, 289, 16675-16687.
- FREEMAN, R., BARON, R., BOUHASSIRA, D., CABRERA, J. & EMIR, B. 2014. Sensory profiles of patients with neuropathic pain based on the neuropathic pain symptoms and signs. *Pain*, 155, 367-76.
- FREYNHAGEN, R., BARON, R., GOCKEL, U. & TOLLE, T. R. 2006a. painDETECT: a new screening questionnaire to identify neuropathic components in patients with back pain. *Curr Med Res Opin*, 22, 1911-20.
- FREYNHAGEN, R., BARON, R., TOLLE, T., STEMMLER, E., GOCKEL, U., STEVENS, M. & MAIER, C. 2006b. Screening of neuropathic pain components in patients with chronic back pain associated with nerve root compression: a

prospective observational pilot study (MIPORT). *Curr Med Res Opin,* 22, 529-37.

- FREYNHAGEN, R., ROLKE, R., BARON, R., TOLLE, T. R., RUTJES, A. K., SCHU, S. & TREEDE, R. D. 2008. Pseudoradicular and radicular low-back pain--a disease continuum rather than different entities? Answers from quantitative sensory testing. *Pain*, 135, 65-74.
- FRUHSTORFER, H., LINDBLOM, U. & SCHMIDT, W. C. 1976. Method for quantitative estimation of thermal thresholds in patients. J Neurol Neurosurg Psychiatry, 39, 1071-5.
- FURUKAWA, T. & GOLDBERG, D. P. 1999. Cultural invariance of likelihood ratios for the General Health Questionnaire. *Lancet*, 353, 561-2.
- GARBINO, J. A., NAAFS, B., SALGADO, M. H., URA, S., VIRMOND MDA, C. & SCHESTATSKY, P. 2011. Association between neuropathic pain and A-waves in leprosy patients with type 1 and 2 reactions. *J Clin Neurophysiol*, 28, 329-32.
- GAUDET, A. D., POPOVICH, P. G. & RAMER, M. S. 2011. Wallerian degeneration: gaining perspective on inflammatory events after peripheral nerve injury. *J Neuroinflammation*, 8, 110.
- GAUTAM, S., NIJHAWAN, M. & KAMAL, P. 1987. Standardisation of hindi version of goldbergs general health questionnaire. *Indian J Psychiatry*, 29, 63-6.
- GEBER, C., SCHERENS, A., PFAU, D., NESTLER, N., ZENZ, M., TOLLE, T., BARON, R., TREEDE, R. D. & MAIER, C. 2009. [Procedure for certification of QST laboratories]. *Schmerz*, 23, 65-9.
- GOLDBERG, D. & WILLIAMS, P. G. 1988. A user's guide to the General Health Questionnaire, Windsor, Berks. : NFER-Nelson, 1988.
- GOLDBERG, D. P. 1972. *The Detection Of Psychiatric Illness By Questionnaire,* London, Oxford University Press.
- GOLDBERG, D. P., GATER, R., SARTORIUS, N., USTUN, T. B., PICCINELLI, M., GUREJE,O. & RUTTER, C. 1997. The validity of two versions of the GHQ in the WHO study of mental illness in general health care. *Psychol Med*, 27, 191-7.
- GOSLING, A. P., CASTRO, L. E., SAADI, L., GOMES, M. K. & CUNHA, A. P. Prevalence and characteristics of neuropathic pain in treated leprosy patients in a

tertiary care reference hospital. 18th International Leprosy Congress 2013 Brussels, Belgium.

- GRAFF-RADFORD, S. B., SHAW, L. R. & NALIBOFF, B. N. 2000. Amitriptyline and fluphenazine in the treatment of postherpetic neuralgia. *Clin J Pain*, 16, 188-92.
- GREENSPAN, J. D. & MCGILLIS, S. L. 1991. Stimulus features relevant to the perception of sharpness and mechanically evoked cutaneous pain. *Somatosens Mot Res*, 8, 137-47.
- GRIFFIN, J. W., MCARTHUR, J. C. & POLYDEFKIS, M. 2001. Assessment of cutaneous innervation by skin biopsies. *Curr Opin Neurol*, 14, 655-9.
- GRONE, E., CRISPIN, A., FLECKENSTEIN, J., IRNICH, D., TREEDE, R. D. & LANG, P. M. 2012. Test order of quantitative sensory testing facilitates mechanical hyperalgesia in healthy volunteers. *J Pain*, 13, 73-80.
- GUINTO, R. S. & RODRIGUEZ, J. 1936. A field study of leprosy in Talisay, Cebu, Philippines. *International Journal of Leprosy* 9, 149–66.
- GUYATT, G. H., FEENY, D. H. & PATRICK, D. L. 1993. Measuring health-related quality of life. *Ann Intern Med*, 118, 622-9.
- HAANPAA, M., ATTAL, N., BACKONJA, M., BARON, R., BENNETT, M., BOUHASSIRA,
 D., CRUCCU, G., HANSSON, P., HAYTHORNTHWAITE, J. A., IANNETTI, G. D.,
 JENSEN, T. S., KAUPPILA, T., NURMIKKO, T. J., RICE, A. S., ROWBOTHAM, M.,
 SERRA, J., SOMMER, C., SMITH, B. H. & TREEDE, R. D. 2010. NeuPSIG
 guidelines on neuropathic pain assessment. *Pain*.
- HAANPAA, M., ATTAL, N., BACKONJA, M., BARON, R., BENNETT, M., BOUHASSIRA,
 D., CRUCCU, G., HANSSON, P., HAYTHORNTHWAITE, J. A., IANNETTI, G. D.,
 JENSEN, T. S., KAUPPILA, T., NURMIKKO, T. J., RICE, A. S., ROWBOTHAM, M.,
 SERRA, J., SOMMER, C., SMITH, B. H. & TREEDE, R. D. 2011a. NeuPSIG
 guidelines on neuropathic pain assessment. *Pain*, 152, 14-27.
- HAANPAA, M., ATTAL, N., BACKONJA, M., BARON, R., BENNETT, M., BOUHASSIRA,
 D., CRUCCU, G., HANSSON, P., HAYTHORNTHWAITE, J. A., IANNETTI, G. D.,
 JENSEN, T. S., KAUPPILA, T., NURMIKKO, T. J., RICE, A. S. C., ROWBOTHAM,
 M., SERRA, J., SOMMER, C., SMITH, B. H. & TREEDE, R. D. 2011b. NeuPSIG
 guidelines on neuropathic pain assessment. *Pain*, 152, 14-27.

- HAANPAA, M., LOCKWOOD, D. N. & HIETAHARJU, A. 2004. Neuropathic pain in leprosy. *Lepr Rev*, 75, 7-18.
- HAANPAA, M. L., BACKONJA, M. M., BENNETT, M. I., BOUHASSIRA, D., CRUCCU, G.,
 HANSSON, P. T., JENSEN, T. S., KAUPPILA, T., RICE, A. S., SMITH, B. H.,
 TREEDE, R. D. & BARON, R. 2009. Assessment of neuropathic pain in primary
 care. Am J Med, 122, S13-21.
- HAGGE, D. A., OBY ROBINSON, S., SCOLLARD, D., MCCORMICK, G. & WILLIAMS, D. L.
 2002. A new model for studying the effects of Mycobacterium leprae on Schwann cell and neuron interactions. *J Infect Dis*, 186, 1283-96.
- HAMMOND, C. J. & KLENERMAN, P. 1988. Protective sensation in the foot in leprosy. *Lepr Rev*, 59, 347-54.
- HANSSON, P. 2002. Neuropathic pain: clinical characteristics and diagnostic workup. *Eur J Pain*, 6 Suppl A, 47-50.
- HANSSON, P., BACKONJA, M. & BOUHASSIRA, D. 2007. Usefulness and limitations of quantitative sensory testing: clinical and research application in neuropathic pain states. *Pain*, 129, 256-9.
- HANSSON, P., LACERENZA, M. & P., M. 2001. Aspects of clinical and experimental neuropathic pain: the clinical perspective. In: Neuropathic pain: pathophysiology and treatment IASP Press.
- HAROUN, O. M. 2014. RE: The use and interpretation of the word "Numbness" in South Africa and India.
- HAROUN, O. M., HIETAHARJU, A., BIZUNEH, E., TESFAYE, F., BRANDSMA, J. W., HAANPAA, M., RICE, A. S. & LOCKWOOD, D. N. 2012. Investigation of neuropathic pain in treated leprosy patients in Ethiopia: a cross-sectional study. *Pain*, 153, 1620-4.
- HAUSMANN, R. 2013. The Global Gender Gap Report. Harvard University.
- HAYTHORNTHWAITE, J. A. & BENRUD-LARSON, L. M. 2000. Psychological aspects of neuropathic pain. *Clin J Pain*, 16, S101-5.
- HELDESTAD, V., LINDER, J., SELLERSJO, L. & NORDH, E. 2010. Reproducibility and influence of test modality order on thermal perception and thermal pain thresholds in quantitative sensory testing. *Clin Neurophysiol*, 121, 1878-85.

- HELGASON, S., PETURSSON, G., GUDMUNDSSON, S. & SIGURDSSON, J. A. 2000. Prevalence of postherpetic neuralgia after a first episode of herpes zoster: prospective study with long term follow up. *BMJ*, 321, 794-6.
- HIETAHARJU, A., CROFT, R., ALAM, R., BIRCH, P., MONG, A. & HAANPAA, M. 2000. Chronic neuropathic pain in treated leprosy. *Lancet*, 356, 1080-1.
- HOLLAND, N. R., CRAWFORD, T. O., HAUER, P., CORNBLATH, D. R., GRIFFIN, J. W. & MCARTHUR, J. C. 1998. Small-fiber sensory neuropathies: clinical course and neuropathology of idiopathic cases. *Ann Neurol*, 44, 47-59.
- IASP 1994. International Association for the Study of Pain Classification of Chronic Pain, Seattle, WA, IASP Press.
- IASP. 2009. International Association for the Study of Pain [Online]. Available: <u>http://www.iasp-pain.org</u>.
- IASP. 2012. International Association for the Study of Pain [Online]. Available: <u>http://www.iasp-pain.org/AM/Template.cfm?Section=Pain_Definitions</u>.
- JAASKELAINEN, S. K., TEERIJOKI-OKSA, T. & FORSSELL, H. 2005. Neurophysiologic and quantitative sensory testing in the diagnosis of trigeminal neuropathy and neuropathic pain. *Pain*, 117, 349-57.
- JAMISON, D. G. 1969. Sensitivity testing as a means of differentiating the various forms of leprosy found in Nigeria. *Lepr Rev*, 40, 17-20.
- JANG, S. W., LIU, X., CHAN, C. B., WEINSHENKER, D., HALL, R. A., XIAO, G. & YE, K. 2009. Amitriptyline is a TrkA and TrkB receptor agonist that promotes TrkA/TrkB heterodimerization and has potent neurotrophic activity. *Chem Biol*, 16, 644-56.
- JENNEKENS, F. G. & JENNEKENS-SCHINKEL, A. 1992. Neurological examination of patients suffering from leprosy: is it worthwhile? *Lepr Rev*, 63, 269-76.
- JENSEN, M. P., CHODROFF, M. J. & DWORKIN, R. H. 2007. The impact of neuropathic pain on health-related quality of life: review and implications. *Neurology*, 68, 1178-82.
- JENSEN, T. S. & BARON, R. 2003. Translation of symptoms and signs into mechanisms in neuropathic pain. *Pain*, 102, 1-8.
- JENSEN, T. S., BARON, R., HAANPAA, M., KALSO, E., LOESER, J. D., RICE, A. S. & TREEDE, R. D. 2011. A new definition of neuropathic pain. *Pain*, 152, 2204-5.

- JENSEN, T. S., GOTTRUP, H., SINDRUP, S. H. & BACH, F. W. 2001. The clinical picture of neuropathic pain. *Eur J Pharmacol*, 429, 1-11.
- JINDAL, K. C., SINGH, G. P., MOHAN, V. & MAHAJAN, B. B. 2013. Psychiatric morbidity among inmates of leprosy homes. *Indian J Psychol Med*, 35, 335-40.
- JOB, C. K. 1989. Nerve damage in leprosy. Int J Lepr Other Mycobact Dis, 57, 532-9.
- KAMATH, S., VACCARO, S. A., REA, T. H. & OCHOA, M. T. 2014. Recognizing and managing the immunologic reactions in leprosy. *J Am Acad Dermatol*.
- KAUTIO, A. L., HAANPAA, M., SAARTO, T. & KALSO, E. 2008. Amitriptyline in the treatment of chemotherapy-induced neuropathic symptoms. J Pain Symptom Manage, 35, 31-9.
- KELLER, S., BANN, C. M., DODD, S. L., SCHEIN, J., MENDOZA, T. R. & CLEELAND, C. S. 2004. Validity of the brief pain inventory for use in documenting the outcomes of patients with noncancer pain. *Clin J Pain*, 20, 309-18.
- KHAMBATI, F. A., SHETTY, V. P., GHATE, S. D. & CAPADIA, G. D. 2009. Sensitivity and specificity of nerve palpation, monofilament testing and voluntary muscle testing in detecting peripheral nerve abnormality, using nerve conduction studies as gold standard; a study in 357 patients. *Lepr Rev*, 80, 34-50.
- KHANOLKAR-YOUNG, S., RAYMENT, N., BRICKELL, P. M., KATZ, D. R., VINAYAKUMAR, S., COLSTON, M. J. & LOCKWOOD, D. N. 1995. Tumour necrosis factor-alpha (TNF-alpha) synthesis is associated with the skin and peripheral nerve pathology of leprosy reversal reactions. *Clin Exp Immunol*, 99, 196-202.
- KHANOLKAR, V. R. 1964. Pathology of leprosy. *In:* COCHRANE, R. G. & DAVEY, T. F. (eds.) *Leprosy in theory and practice.* 2nd ed. Bristol, UK,: John Wright and Sons Ltd.
- KIEBURTZ, K., SIMPSON, D., YIANNOUTSOS, C., MAX, M. B., HALL, C. D., ELLIS, R. J., MARRA, C. M., MCKENDALL, R., SINGER, E., DAL PAN, G. J., CLIFFORD, D. B., TUCKER, T. & COHEN, B. 1998. A randomized trial of amitriptyline and mexiletine for painful neuropathy in HIV infection. AIDS Clinical Trial Group 242 Protocol Team. *Neurology*, 51, 1682-8.
- KIM, H. Y. 2013. Statistical notes for clinical researchers: assessing normal distribution (2) using skewness and kurtosis. *Restor Dent Endod*, 38, 52-4.

- KLEGGETVEIT, I. P. & JORUM, E. 2010. Large and small fiber dysfunction in peripheral nerve injuries with or without spontaneous pain. *J Pain*, 11, 1305-10.
- KLENERMAN, P. & HAMMOND, C. 1988. Vibration sensation in leprosy patches. Int J Lepr Other Mycobact Dis, 56, 466-8.
- KLENERMAN, P., HAMMOND, C., KULKARNI, V. N. & MEHTA, J. M. 1990. Vibration sense and tarsal disintegration. *Indian J Lepr*, 62, 422-8.
- KONOPKA, K. H., HARBERS, M., HOUGHTON, A., KORTEKAAS, R., VAN VLIET, A., TIMMERMAN, W., DEN BOER, J. A., STRUYS, M. M. & VAN WIJHE, M. 2012. Bilateral sensory abnormalities in patients with unilateral neuropathic pain; a quantitative sensory testing (QST) study. *PLoS One*, **7**, e37524.
- KRUMOVA, E. 2010. Quantitative sensory testing: a diagnostic tool for painful neuropathy. *Future Neurology*, 5 721-733.
- KRUTZIK, S. R., OCHOA, M. T., SIELING, P. A., UEMATSU, S., NG, Y. W., LEGASPI, A.,
 LIU, P. T., COLE, S. T., GODOWSKI, P. J., MAEDA, Y., SARNO, E. N., NORGARD,
 M. V., BRENNAN, P. J., AKIRA, S., REA, T. H. & MODLIN, R. L. 2003. Activation
 and regulation of Toll-like receptors 2 and 1 in human leprosy. *Nat Med*, 9, 525-32.
- KUMAR, H. K. 2010. IAL Textbook of Leprosy, Indian.
- LAHIRI, R., RANDHAWA, B. & KRAHENBUHL, J. L. 2010. Infection of mouse macrophages with viable Mycobacterium leprae does not induce apoptosis. *J Infect Dis*, 201, 1736-42.
- LAME, I. E., PETERS, M. L., VLAEYEN, J. W., KLEEF, M. & PATIJN, J. 2005. Quality of life in chronic pain is more associated with beliefs about pain, than with pain intensity. *Eur J Pain*, 9, 15-24.

LANCET 2009. Leprosy as a neurological disease. Lancet Neurol, 8, 217.

- LANDERHOLM, S. H., EKBLOM, A. G. & HANSSON, P. T. 2010. Somatosensory function in patients with and without pain after traumatic peripheral nerve injury. *Eur J Pain*, 14, 847-53.
- LASRY-LEVY, E., HIETAHARJU, A., PAI, V., GANAPATI, R., RICE, A. S., HAANPAA, M. & LOCKWOOD, D. N. 2011. Neuropathic pain and psychological morbidity in

patients with treated leprosy: a cross-sectional prevalence study in Mumbai. *PLoS Negl Trop Dis,* **5,** e981.

- LAURIA, G., CORNBLATH, D. R., JOHANSSON, O., MCARTHUR, J. C., MELLGREN, S. I., NOLANO, M., ROSENBERG, N. & SOMMER, C. 2005. EFNS guidelines on the use of skin biopsy in the diagnosis of peripheral neuropathy. *Eur J Neurol*, 12, 747-58.
- LAURIA, G., HSIEH, S. T., JOHANSSON, O., KENNEDY, W. R., LEGER, J. M., MELLGREN, S. I., NOLANO, M., MERKIES, I. S., POLYDEFKIS, M., SMITH, A. G., SOMMER, C. & VALLS-SOLE, J. 2010. European Federation of Neurological Societies/Peripheral Nerve Society Guideline on the use of skin biopsy in the diagnosis of small fiber neuropathy. Report of a joint task force of the European Federation of Neurological Societies and the Peripheral Nerve Society. *Eur J Neurol*, 17, 903-12, e44-9.
- LAURIA, G., LOMBARDI, R., CAMOZZI, F. & DEVIGILI, G. 2009. Skin biopsy for the diagnosis of peripheral neuropathy. *Histopathology*, 54, 273-85.
- LEEKASSA, R., BIZUNEH, E. & ALEM, A. 2004. Prevalence of mental distress in the outpatient clinic of a specialized leprosy hospital. Addis Ababa, Ethiopia, 2002. *Lepr Rev*, 75, 367-75.
- LEFFLER, A. S. & HANSSON, P. 2008a. Letter to the Editor of Pain on Freynhagen et al.: Pseudoradicular and radicular low-back pain - a disease continuum rather than different entities? Answers from quantitative sensory testing. Pain 2007;135:65-74. *Pain*, 135, 312-3; author reply 313-5.
- LEFFLER, A. S. & HANSSON, P. 2008b. Painful traumatic peripheral partial nerve injury-sensory dysfunction profiles comparing outcomes of bedside examination and quantitative sensory testing. *Eur J Pain*, **12**, 397-402.
- LEVY, L., SHEPARD, C. C. & FASAL, P. 1976. The bactericidal effect of rifampicin on M. leprae in man: a) single doses of 600, 900 and 1200 mg; and b) daily doses of 300 mg. *Int J Lepr Other Mycobact Dis*, 44, 183-7.
- LIGHT, A. R. & PERL, E. R. 1993. Peripheral sensory systems. *In:* DYCK, P. J., THOMAS,
 P. K., GRIFFIN, J. W., LOW, P. A. & PODUSLO, J. F. (eds.) *Peripheral neuropathy.* Philadelphia: Saunders.

- LINDENLAUB, T. & SOMMER, C. 2003. Cytokines in sural nerve biopsies from inflammatory and non-inflammatory neuropathies. *Acta Neuropathol*, 105, 593-602.
- LIST, T., HELKIMO, M. & KARLSSON, R. 1991. Influence of pressure rates on the reliability of a pressure threshold meter. *J Craniomandib Disord*, 5, 173-8.
- LITTLE, D., KHANOLKAR-YOUNG, S., COULTHART, A., SUNEETHA, S. & LOCKWOOD, D. N. 2001. Immunohistochemical analysis of cellular infiltrate and gamma interferon, interleukin-12, and inducible nitric oxide synthase expression in leprosy type 1 (reversal) reactions before and during prednisolone treatment. *Infect Immun*, 69, 3413-7.
- LIU, X., ESCHENFELDER, S., BLENK, K. H., JANIG, W. & HABLER, H. 2000. Spontaneous activity of axotomized afferent neurons after L5 spinal nerve injury in rats. *Pain*, 84, 309-18.
- LOCKWOOD, D. 2002. Leprosy. Clin Evid, 709-20.
- LOCKWOOD, D. N. 1996. The management of erythema nodosum leprosum: current and future options. *Lepr Rev*, 67, 253-9.
- LOCKWOOD, D. N., COLSTON, M. J. & KHANOLKAR-YOUNG, S. R. 2002. The detection of Mycobacterium leprae protein and carbohydrate antigens in skin and nerve from leprosy patients with type 1 (reversal) reactions. *Am J Trop Med Hyg*, 66, 409-15.
- LOCKWOOD, D. N. & REID, A. J. 2001. The diagnosis of leprosy is delayed in the United Kingdom. *QJM*, 94, 207-12.
- LOSETH, S., STALBERG, E., JORDE, R. & MELLGREN, S. I. 2008. Early diabetic neuropathy: thermal thresholds and intraepidermal nerve fibre density in patients with normal nerve conduction studies. *J Neurol*, 255, 1197-202.
- LUND, C., KOSKINEN, M., SUNEETHA, S., LOCKWOOD, D. N., HAANPAA, M., HAAPASALO, H. & HIETAHARJU, A. 2007. Histopathological and clinical findings in leprosy patients with chronic neuropathic pain: a study from Hyderabad, India. *Lepr Rev*, 78, 369-80.
- LYNN, B., SCHUTTERLE, S. & PIERAU, F. K. 1996. The vasodilator component of neurogenic inflammation is caused by a special subclass of heat-sensitive nociceptors in the skin of the pig. *J Physiol*, 494 (Pt 2), 587-93.

- MAGERL, W., KRUMOVA, E. K., BARON, R., TOLLE, T., TREEDE, R. D. & MAIER, C. 2010. Reference data for quantitative sensory testing (QST): refined stratification for age and a novel method for statistical comparison of group data. *Pain*, 151, 598-605.
- MAHAJAN, P. M., JOGAIKAR, D. G. & MEHTA, J. M. 1996. A study of pure neuritic leprosy: clinical experience. *Indian J Lepr*, 68, 137-41.
- MAHN, F., HULLEMANN, P., GOCKEL, U., BROSZ, M., FREYNHAGEN, R., TOLLE, T. R. & BARON, R. 2011. Sensory symptom profiles and co-morbidities in painful radiculopathy. *PLoS ONE*, 6, e18018.
- MAIA, M. V., CUNHA MDA, G. & CUNHA, C. S. 2013. Adverse effects of alternative therapy (minocycline, ofloxacin, and clofazimine) in multibacillary leprosy patients in a recognized health care unit in Manaus, Amazonas, Brazil. *An Bras Dermatol*, 88, 205-10.
- MAIER, C., BARON, R., TOLLE, T. R., BINDER, A., BIRBAUMER, N., BIRKLEIN, F., GIERTHMUHLEN, J., FLOR, H., GEBER, C., HUGE, V., KRUMOVA, E. K., LANDWEHRMEYER, G. B., MAGERL, W., MAIHOFNER, C., RICHTER, H., ROLKE, R., SCHERENS, A., SCHWARZ, A., SOMMER, C., TRONNIER, V., UCEYLER, N., VALET, M., WASNER, G. & TREEDE, R. D. 2010. Quantitative sensory testing in the German Research Network on Neuropathic Pain (DFNS): somatosensory abnormalities in 1236 patients with different neuropathic pain syndromes. *Pain*, 150, 439-50.
- MAKI, D., RAJAB, E., WATSON, P. J. & CRITCHLEY, D. J. 2014. Cross-cultural Translation, Adaptation, and Psychometric Testing of the Roland-Morris Disability Questionnaire Into Modern Standard Arabic. *Spine (Phila Pa 1976)*, 39, E1537-44.
- MALAVIYA, G. N. 2005. Neuropathic pain in leprosy patients. *Int J Lepr Other Mycobact Dis*, 73, 34-5.
- MANEESRIWONGUL, W. & DIXON, J. K. 2004. Instrument translation process: a methods review. *J Adv Nurs*, 48, 175-86.
- MANGANELLI, F., IODICE, V., PROVITERA, V., PISCIOTTA, C., NOLANO, M., PERRETTI,
 A. & SANTORO, L. 2007. Small-fiber involvement in spinobulbar muscular atrophy (Kennedy's disease). *Muscle Nerve*, 36, 816-20.

- MARTIN, C., SOLDERS, G., SONNERBORG, A. & HANSSON, P. 2003. Painful and nonpainful neuropathy in HIV-infected patients: an analysis of somatosensory nerve function. *Eur J Pain*, **7**, 23-31.
- MARTINEZ, V., SZEKELY, B., LEMARIE, J., MARTIN, F., GENTILI, M., BEN AMMAR, S., LEPEINTRE, J. F., GARREAU DE LOUBRESSE, C., CHAUVIN, M., BOUHASSIRA, D. & FLETCHER, D. 2013. The efficacy of a glial inhibitor, minocycline, for preventing persistent pain after lumbar discectomy: a randomized, double-blind, controlled study. *Pain*, 154, 1197-203.
- MASAKI, T., QU, J., CHOLEWA-WACLAW, J., BURR, K., RAAUM, R. & RAMBUKKANA,
 A. 2013. Reprogramming adult Schwann cells to stem cell-like cells by leprosy bacilli promotes dissemination of infection. *Cell*, 152, 51-67.
- MAX, M. B. 1987. Endogenous monoamine analgesic systems: amitriptyline in painful diabetic neuropathy. *Anesth Prog*, 34, 123-7.
- MAX, M. B., CULNANE, M., SCHAFER, S. C., GRACELY, R. H., WALTHER, D. J., SMOLLER, B. & DUBNER, R. 1987. Amitriptyline relieves diabetic neuropathy pain in patients with normal or depressed mood. *Neurology*, 37, 589-96.
- MCCLEANE, G. 2003. Pharmacological management of neuropathic pain. *CNS Drugs*, 17, 1031-43.
- MCKNIGHT, J. 2010. *Clinical testing and prognostic markers for the development of leprosy neuropathy.* PhD, University of London.
- MCMAHON SB, KOLTZENBURG M, TRACEY, I. & TURK, D. C. 2013. Wall & Melzack's Textbook of Pain.
- MEIMA, A., SAUNDERSON, P. R., GEBRE, S., DESTA, K., VAN OORTMARSSEN, G. J. & HABBEMA, J. D. 1999. Factors associated with impairments in new leprosy patients: the AMFES cohort. *Lepr Rev*, 70, 189-203.
- MELZACK, R. 1975. The McGill Pain Questionnaire: major properties and scoring methods. *Pain*, 1, 277-99.
- MELZACK, R. 1987. The short-form McGill Pain Questionnaire. Pain, 30, 191-7.
- MERSKEY, H. & BOGDUK, N. 1994. Classification of Chronic Pain: Descriptions of Chronic Pain Syndromes and Definitions of Pain Terms, Seattle, ISAP Press 222.

MEYER-ROSBERG, K., KVARNSTROM, A., KINNMAN, E., GORDH, T., NORDFORS, L. O.
 & KRISTOFFERSON, A. 2001. Peripheral neuropathic pain--a multidimensional burden for patients. *Eur J Pain*, 5, 379-89.

- MEYER, R. A., RINGKAMP, M., CAMPBELL, J. N. & RAJA, S. N. 2006. Peripheral mechanisms of cutaneous nociception. *In:* MCMAHON SB & KOLTZENBURG M (eds.) *Wall and Melzack's textbook of pain.* Philadelphia: Elsevier.
- MISHRA, B., MUKHERJEE, A., GIRDHAR, A., HUSAIN, S., MALAVIYA, G. N. & GIRDHAR,
 B. K. 1995. Neuritic leprosy: further progression and significance. *Acta Leprol*, 9, 187-94.
- MURTO, C., CHAMMARTIN, F., SCHWARZ, K., DA COSTA, L. M., KAPLAN, C. & HEUKELBACH, J. 2013. Patterns of migration and risks associated with leprosy among migrants in Maranhao, Brazil. *PLoS Negl Trop Dis*, 7, e2422.
- NELSON, E. C. & BERWICK, D. M. 1989. The measurement of health status in clinical practice. *Med Care*, 27, S77-90.
- NG, V., ZANAZZI, G., TIMPL, R., TALTS, J. F., SALZER, J. L., BRENNAN, P. J. & RAMBUKKANA, A. 2000. Role of the cell wall phenolic glycolipid-1 in the peripheral nerve predilection of Mycobacterium leprae. *Cell*, 103, 511-24.
- NICHOLLS, P. G., WIENS, C. & SMITH, W. C. 2003. Delay in presentation in the context of local knowledge and attitude towards leprosy--the results of qualitative fieldwork in Paraguay. *Int J Lepr Other Mycobact Dis*, 71, 198-209.
- NOLANO, M., PROVITERA, V., ESTRANEO, A., SELIM, M. M., CAPORASO, G., STANCANELLI, A., SALTALAMACCHIA, A. M., LANZILLO, B. & SANTORO, L. 2008. Sensory deficit in Parkinson's disease: evidence of a cutaneous denervation. *Brain*, 131, 1903-11.
- NOVY, D. M., NELSON, D. V., FRANCIS, D. J. & TURK, D. C. 1995. Perspectives of chronic pain: an evaluative comparison of restrictive and comprehensive models. *Psychol Bull*, 118, 238-47.
- OAKLANDER, A. L. 2001. The density of remaining nerve endings in human skin with and without postherpetic neuralgia after shingles. *Pain*, 92, 139-45.
- OCHOA, J. L. & YARNITSKY, D. 1993. Mechanical hyperalgesias in neuropathic pain patients: dynamic and static subtypes. *Ann Neurol*, 33, 465-72.

- OLIVEIRA, R. B., MORAES, M. O., OLIVEIRA, E. B., SARNO, E. N., NERY, J. A. & SAMPAIO, E. P. 1999. Neutrophils isolated from leprosy patients release TNF-alpha and exhibit accelerated apoptosis in vitro. *J Leukoc Biol*, 65, 364-71.
- OLIVEIRA, R. B., OCHOA, M. T., SIELING, P. A., REA, T. H., RAMBUKKANA, A., SARNO, E. N. & MODLIN, R. L. 2003. Expression of Toll-like receptor 2 on human Schwann cells: a mechanism of nerve damage in leprosy. *Infect Immun,* 71, 1427-33.
- ORSTAVIK, K., NAMER, B., SCHMIDT, R., SCHMELZ, M., HILLIGES, M., WEIDNER, C., CARR, R. W., HANDWERKER, H., JORUM, E. & TOREBJORK, H. E. 2006. Abnormal function of C-fibers in patients with diabetic neuropathy. *J Neurosci,* 26, 11287-94.
- OSKAM, L., SLIM, E. & BUHRER-SEKULA, S. 2003. Serology: recent developments, strengths, limitations and prospects: a state of the art overview. *Lepr Rev*, 74, 196-205.
- OXFORD. 2009. Oxford University Press: Disability [Online]. Available: <u>http://www.askoxford.com/concise_oed/disability?view=uk</u> [Accessed 10th August 2009].
- PARKASH, O. 2011. Serological detection of leprosy employing Mycobacterium leprae derived serine-rich 45 kDa, ESAT-6, CFP-10 and PGL-I: a compilation of data from studies in Indian populations. *Lepr Rev*, 82, 383-8.
- PATRICK, D. L. & DEYO, R. A. 1989. Generic and disease-specific measures in assessing health status and quality of life. *Med Care*, 27, S217-32.
- PEREZ, C., GALVEZ, R., HUELBES, S., INSAUSTI, J., BOUHASSIRA, D., DIAZ, S. & REJAS, J. 2007. Validity and reliability of the Spanish version of the DN4 (Douleur Neuropathique 4 questions) questionnaire for differential diagnosis of pain syndromes associated to a neuropathic or somatic component. *Health Qual Life Outcomes*, 5, 66.
- PETRIE, A. & SABIN, C. 2009. *Medical Statistics at a Glance*, Oxford, England, Wiley-Blackwell.

PETZKE, F., CLAUW, D. J., AMBROSE, K., KHINE, A. & GRACELY, R. H. 2003. Increased pain sensitivity in fibromyalgia: effects of stimulus type and mode of presentation. *Pain*, 105, 403-13.

PFALTZGRAFF R. E & G, R. 1994. *Leprosy*, Edinburgh, Churchill Livingstone.

- PFAU, D. B., GEBER, C., BIRKLEIN, F. & TREEDE, R. D. 2012. Quantitative sensory testing of neuropathic pain patients: potential mechanistic and therapeutic implications. *Curr Pain Headache Rep*, 16, 199-206.
- PFAU, D. B., KRUMOVA, E. K., TREEDE, R. D., BARON, R., TOELLE, T., BIRKLEIN, F., EICH, W., GEBER, C., GERHARDT, A., WEISS, T., MAGERL, W. & MAIER, C. 2014. Quantitative sensory testing in the German Research Network on Neuropathic Pain (DFNS): reference data for the trunk and application in patients with chronic postherpetic neuralgia. *Pain*, 155, 1002-15.
- PHILLIPS, T. J., BROWN, M., RAMIREZ, J. D., PERKINS, J., WOLDEAMANUEL, Y. W., WILLIAMS, A. C., ORENGO, C., BENNETT, D. L., BODI, I., COX, S., MAIER, C., KRUMOVA, E. K. & RICE, A. S. 2014. Sensory, psychological, and metabolic dysfunction in HIV-associated peripheral neuropathy: A cross-sectional deep profiling study. *Pain*.
- PHILLIPS, T. J., CHERRY, C. L., COX, S., MARSHALL, S. J. & RICE, A. S. 2010. Pharmacological treatment of painful HIV-associated sensory neuropathy: a systematic review and meta-analysis of randomised controlled trials. *PLoS One*, 5, e14433.
- PNA. 1993. Quantitative sensory testing: a consensus report from the Peripheral Neuropathy Association. *Neurology*, 43, 1050-2.
- POCATERRA, L., JAIN, S., REDDY, R., MUZAFFARULLAH, S., TORRES, O., SUNEETHA, S.
 & LOCKWOOD, D. N. 2006. Clinical course of erythema nodosum leprosum: an 11-year cohort study in Hyderabad, India. *Am J Trop Med Hyg*, 74, 868-79.
- POLYDEFKIS, M., YIANNOUTSOS, C. T., COHEN, B. A., HOLLANDER, H., SCHIFITTO, G., CLIFFORD, D. B., SIMPSON, D. M., KATZENSTEIN, D., SHRIVER, S., HAUER, P., BROWN, A., HAIDICH, A. B., MOO, L. & MCARTHUR, J. C. 2002. Reduced intraepidermal nerve fiber density in HIV-associated sensory neuropathy. *Neurology*, 58, 115-9.

- QUATTRINI, C., TAVAKOLI, M., JEZIORSKA, M., KALLINIKOS, P., TESFAYE, S., FINNIGAN, J., MARSHALL, A., BOULTON, A. J., EFRON, N. & MALIK, R. A. 2007. Surrogate markers of small fiber damage in human diabetic neuropathy. *Diabetes*, 56, 2148-54.
- RACINE, M., CASTARLENAS, E., DE LA VEGA, R., TOME-PIRES, C., SOLE, E., MIRO, J., JENSEN, M. P., MOULIN, D. E. & NIELSON, W. R. 2014. Sex Differences in Psychological Response to Pain in Patients with Fibromyalgia Syndrome. *Clin J Pain*.
- RAICHER, I., STUMP, P. R., BACCARELLI, R., MARCIANO, L. H., URA, S., VIRMOND MDA, C., TEIXEIRA, A. L. & DE ANDRADE, D. C. Prevalence and characteristic of leprosy-related neuropathic pain: validation of two tools. 18th International Leprosy Congress, 2013 Brussels, Belgium.
- RAJA, S. N., HAYTHORNTHWAITE, J. A., PAPPAGALLO, M., CLARK, M. R., TRAVISON,
 T. G., SABEEN, S., ROYALL, R. M. & MAX, M. B. 2002. Opioids versus antidepressants in postherpetic neuralgia: a randomized, placebo-controlled trial. *Neurology*, 59, 1015-21.
- RAMBUKKANA, A. 2004. Mycobacterium leprae-induced demyelination: a model for early nerve degeneration. *Curr Opin Immunol*, 16, 511-8.
- RAMBUKKANA, A., SALZER, J. L., YURCHENCO, P. D. & TUOMANEN, E. I. 1997. Neural targeting of Mycobacterium leprae mediated by the G domain of the laminin-alpha2 chain. *Cell*, 88, 811-21.
- RAMBUKKANA, A., ZANAZZI, G., TAPINOS, N. & SALZER, J. L. 2002. Contactdependent demyelination by Mycobacterium leprae in the absence of immune cells. *Science*, 296, 927-31.
- RAMOS, J. M., ALONSO-CASTANEDA, B., ESHETU, D., LEMMA, D., REYES, F., BELINCHON, I. & GORGOLAS, M. 2014. Prevalence and characteristics of neuropathic pain in leprosy patients treated years ago. *Pathog Glob Health*, 108, 186-90.
- RANQUE, B., NGUYEN, V. T., VU, H. T., NGUYEN, T. H., NGUYEN, N. B., PHAM, X. K., SCHURR, E., ABEL, L. & ALCAIS, A. 2007. Age is an important risk factor for onset and sequelae of reversal reactions in Vietnamese patients with leprosy. *Clin Infect Dis*, 44, 33-40.

- RASMUSSEN, P. V., SINDRUP, S. H., JENSEN, T. S. & BACH, F. W. 2004. Symptoms and signs in patients with suspected neuropathic pain. *Pain*, 110, 461-9.
- REIMER, M., HELFERT, S. M. & BARON, R. 2014. Phenotyping neuropathic pain patients: implications for individual therapy and clinical trials. *Curr Opin Support Palliat Care*, 8, 124-9.
- REIS, F. J. J., GOMES, M. K., RODRIGUES, J., GOSLING, A. P., FONTANA, A. P. & CUNHA, A. J. L. 2013. Pain and Its Consequences in Quality of Life: A Study with WHOQOL-Bref in Leprosy Patients with Neuropathic Pain. *ISRN Tropical Medicine*, 2013, 7.
- REYNOLDS, E. H. 2014. The neurology of folic acid deficiency. *Handb Clin Neurol*, 120, 927-43.
- RICHARDUS, J. H., NICHOLLS, P. G., CROFT, R. P., WITHINGTON, S. G. & SMITH, W. C. 2004. Incidence of acute nerve function impairment and reactions in leprosy: a prospective cohort analysis after 5 years of follow-up. *Int J Epidemiol,* 33, 337-43.
- RIDLEY, D. S. & JOPLING, W. H. 1966. Classification of leprosy according to immunity. A five-group system. *Int J Lepr Other Mycobact Dis*, 34, 255-73.
- ROJEWSKA, E., MAKUCH, W., PRZEWLOCKA, B. & MIKA, J. 2014. Minocycline prevents dynorphin-induced neurotoxicity during neuropathic pain in rats. *Neuropharmacology*, 86, 301-10.
- ROLKE, R., BARON, R., MAIER, C., TOLLE, T. R., TREEDE, R. D., BEYER, A., BINDER, A., BIRBAUMER, N., BIRKLEIN, F., BOTEFUR, I. C., BRAUNE, S., FLOR, H., HUGE, V., KLUG, R., LANDWEHRMEYER, G. B., MAGERL, W., MAIHOFNER, C., ROLKO, C., SCHAUB, C., SCHERENS, A., SPRENGER, T., VALET, M. & WASSERKA, B. 2006a. Quantitative sensory testing in the German Research Network on Neuropathic Pain (DFNS): standardized protocol and reference values. *Pain*, 123, 231-43.
- ROLKE, R., MAGERL, W., CAMPBELL, K. A., SCHALBER, C., CASPARI, S., BIRKLEIN, F. & TREEDE, R. D. 2006b. Quantitative sensory testing: a comprehensive protocol for clinical trials. *Eur J Pain*, 10, 77-88.
- SANDKUHLER, J. 2009. Models and mechanisms of hyperalgesia and allodynia. *Physiol Rev*, 89, 707-58.

- SAUER, S. K., BOVE, G. M., AVERBECK, B. & REEH, P. W. 1999. Rat peripheral nerve components release calcitonin gene-related peptide and prostaglandin E2 in response to noxious stimuli: evidence that nervi nervorum are nociceptors. *Neuroscience*, 92, 319-25.
- SAUNDERSON, P., BIZUNEH, E. & LEEKASSA, R. 2008. Neuropathic pain in people treated for multibacillary leprosy more than ten years previously. *Lepr Rev*, 79, 270-6.
- SAUNDERSON, P., GEBRE, S. & BYASS, P. 2000a. ENL reactions in the multibacillary cases of the AMFES cohort in central Ethiopia: incidence and risk factors. *Lepr Rev*, 71, 318-24.
- SAUNDERSON, P., GEBRE, S. & BYASS, P. 2000b. Reversal reactions in the skin lesions of AMFES patients: incidence and risk factors. *Lepr Rev*, 71, 309-17.
- SAUNDERSON, P., GEBRE, S., DESTA, K., BYASS, P. & LOCKWOOD, D. N. 2000c. The pattern of leprosy-related neuropathy in the AMFES patients in Ethiopia: definitions, incidence, risk factors and outcome. *Lepr Rev*, **71**, 285-308.
- SAVE, M. P., SHETTY, V. P., SHETTY, K. T. & ANTIA, N. H. 2004. Alterations in neurofilament protein(s) in human leprous nerves: morphology, immunohistochemistry and Western immunoblot correlative study. *Neuropathol Appl Neurobiol*, 30, 635-50.
- SCHENK, M., KRUTZIK, S. R., SIELING, P. A., LEE, D. J., TELES, R. M., OCHOA, M. T.,
 KOMISOPOULOU, E., SARNO, E. N., REA, T. H., GRAEBER, T. G., KIM, S.,
 CHENG, G. & MODLIN, R. L. 2012. NOD2 triggers an interleukin-32dependent human dendritic cell program in leprosy. *Nat Med*, 18, 555-63.
- SCHERENS, A., MAIER, C., HAUSSLEITER, I. S., SCHWENKREIS, P., VLCKOVA-MORAVCOVA, E., BARON, R. & SOMMER, C. 2009. Painful or painless lower limb dysesthesias are highly predictive of peripheral neuropathy: comparison of different diagnostic modalities. *Eur J Pain*, 13, 711-8.
- SCHMIDT, C. O., SCHWEIKERT, B., WENIG, C. M., SCHMIDT, U., GOCKEL, U., FREYNHAGEN, R., TOLLE, T. R., BARON, R. & KOHLMANN, T. 2009. Modelling the prevalence and cost of back pain with neuropathic components in the general population. *Eur J Pain*, 13, 1030-5.

- SCHOLZ, J. & WOOLF, C. J. 2007. The neuropathic pain triad: neurons, immune cells and glia. *Nat Neurosci*, 10, 1361-8.
- SCHREUDER, P. A. 1998. The occurrence of reactions and impairments in leprosy: experience in the leprosy control program of three provinces in northeastern Thailand, 1987-1995 [correction of 1978-1995]. III. Neural and other impairments. *Int J Lepr Other Mycobact Dis*, 66, 170-81.
- SCOLLARD, D. M. 2000. Endothelial cells and the pathogenesis of lepromatous neuritis:insights from the armadillo model. *Microbes Infect*, 2, 1835-43.
- SCOLLARD, D. M. 2008. The biology of nerve injury in leprosy. *Lepr Rev*, 79, 242-53.
- SCOLLARD, D. M., ADAMS, L. B., GILLIS, T. P., KRAHENBUHL, J. L., TRUMAN, R. W. & WILLIAMS, D. L. 2006. The continuing challenges of leprosy. *Clin Microbiol Rev*, 19, 338-81.
- SCOLLARD, D. M., MCCORMICK, G. & ALLEN, J. L. 1999. Localization of Mycobacterium leprae to endothelial cells of epineurial and perineurial blood vessels and lymphatics. *Am J Pathol*, 154, 1611-20.
- SCOTT, F. T., LEEDHAM-GREEN, M. E., BARRETT-MUIR, W. Y., HAWRAMI, K., GALLAGHER, W. J., JOHNSON, R. & BREUER, J. 2003. A study of shingles and the development of postherpetic neuralgia in East London. J Med Virol, 70 Suppl 1, S24-30.
- SEMMES, J., WEINSTEIN, S. & TEUBER, H. L. 1960. Somatosensory Changes After Penetrating Brain Wounds in Man, Cambridge, Massachusetts, Harvard University Press.
- SENTURK, V., STEWART, R. & SAGDUYU, A. 2007. Screening for mental disorders in leprosy patients: comparing the internal consistency and screening properties of HADS and GHQ-12. *Lepr Rev*, 78, 231-42.
- SERPELL, M., GATER, A., CARROLL, S., ABETZ-WEBB, L., MANNAN, A. & JOHNSON, R. 2014. Burden of post-herpetic neuralgia in a sample of UK residents aged 50 years or older: findings from the Zoster Quality of Life (ZQOL) study. *Health Qual Life Outcomes*, 12, 92.
- SETIA, M. S., SHINDE, S. S., JERAJANI, H. R. & BOIVIN, J. F. 2011. Is there a role for rifampicin, ofloxacin and minocycline (ROM) therapy in the treatment of

leprosy? Systematic review and meta-analysis. *Trop Med Int Health,* 16, 1541-51.

- SHETTY, V. P., ANTIA, N. H. & JACOBS, J. M. 1988. The pathology of early leprous neuropathy. *J Neurol Sci*, 88, 115-31.
- SHETTY, V. P., MEHTA, L. N., ANTIA, N. H. & IRANI, P. F. 1977. Teased fibre study of early nerve lesions in leprosy and in contacts, with electrophysiological correlates. *J Neurol Neurosurg Psychiatry*, 40, 708-11.
- SHLAY, J. C., CHALONER, K., MAX, M. B., FLAWS, B., REICHELDERFER, P., WENTWORTH, D., HILLMAN, S., BRIZZ, B. & COHN, D. L. 1998. Acupuncture and amitriptyline for pain due to HIV-related peripheral neuropathy: a randomized controlled trial. Terry Beirn Community Programs for Clinical Research on AIDS. JAMA, 280, 1590-5.
- SHY, M. E., FROHMAN, E. M., SO, Y. T., AREZZO, J. C., CORNBLATH, D. R., GIULIANI,
 M. J., KINCAID, J. C., OCHOA, J. L., PARRY, G. J., WEIMER, L. H.,
 THERAPEUTICS & TECHNOLOGY ASSESSMENT SUBCOMMITTEE OF THE
 AMERICAN ACADEMY OF, N. 2003. Quantitative sensory testing: report of
 the Therapeutics and Technology Assessment Subcommittee of the
 American Academy of Neurology. *Neurology*, 60, 898-904.
- SIMPSON, D. M., BROWN, S., TOBIAS, J. & GROUP, N.-C. S. 2008. Controlled trial of high-concentration capsaicin patch for treatment of painful HIV neuropathy. *Neurology*, 70, 2305-13.
- SIMPSON, D. M., HAIDICH, A. B., SCHIFITTO, G., YIANNOUTSOS, C. T., GERACI, A. P., MCARTHUR, J. C. & KATZENSTEIN, D. A. 2002. Severity of HIV-associated neuropathy is associated with plasma HIV-1 RNA levels. *AIDS*, 16, 407-12.
- SINDRUP, S. H., OTTO, M., FINNERUP, N. B. & JENSEN, T. S. 2005. Antidepressants in the treatment of neuropathic pain. *Basic Clin Pharmacol Toxicol*, 96, 399-409.
- SMITH, B. H. & TORRANCE, N. 2012. Epidemiology of neuropathic pain and its impact on quality of life. *Curr Pain Headache Rep*, 16, 191-8.
- SMITH, B. H., TORRANCE, N., BENNETT, M. I. & LEE, A. J. 2007. Health and quality of life associated with chronic pain of predominantly neuropathic origin in the community. *Clin J Pain*, 23, 143-9.

- SMITH, W. C., NICHOLLS, P. G., DAS, L., BARKATAKI, P., SUNEETHA, S., SUNEETHA, L., JADHAV, R., SUNDAR RAO, P. S., WILDER-SMITH, E. P., LOCKWOOD, D. N. & VAN BRAKEL, W. H. 2009. Predicting neuropathy and reactions in leprosy at diagnosis and before incident events-results from the INFIR cohort study. *PLoS Negl Trop Dis*, 3, e500.
- SMYTH, K., AFFANDI, J. S., MCARTHUR, J. C., BOWTELL-HARRIS, C., MIJCH, A. M., WATSON, K., COSTELLO, K., WOOLLEY, I. J., PRICE, P., WESSELINGH, S. L. & CHERRY, C. L. 2007. Prevalence of and risk factors for HIV-associated neuropathy in Melbourne, Australia 1993-2006. *HIV Med*, 8, 367-73.
- SOMMER, C. & LAURIA, G. 2007. Skin biopsy in the management of peripheral neuropathy. *Lancet Neurol*, 6, 632-42.
- SORENSEN, L., MOLYNEAUX, L. & YUE, D. K. 2006. The relationship among pain, sensory loss, and small nerve fibers in diabetes. *Diabetes Care*, 29, 883-7.
- SRIRAM, T. G., CHANDRASHEKAR, C. R., ISAAC, M. K. & SHANMUGHAM, V. 1989. The General Health Questionnaire (GHQ). Comparison of the English version and a translated Indian version. *Soc Psychiatry Psychiatr Epidemiol*, 24, 317-20.
- STAUD, R., ROBINSON, M. E., VIERCK, C. J., JR., CANNON, R. C., MAUDERLI, A. P. & PRICE, D. D. 2003. Ratings of experimental pain and pain-related negative affect predict clinical pain in patients with fibromyalgia syndrome. *Pain*, 105, 215-22.
- STEER, R. A., RISSMILLER, D. J., RANIERI, W. F. & BECK, A. T. 1993. Structure of the computer-assisted Beck Anxiety Inventory with psychiatric inpatients. *J Pers Assess*, 60, 532-42.
- STUMP, P. R., BACCARELLI, R., MARCIANO, L. H., LAURIS, J. R., TEIXEIRA, M. J., URA,
 S. & VIRMOND, M. C. 2004. Neuropathic pain in leprosy patients. *Int J Lepr* Other Mycobact Dis, 72, 134-8.
- STUMP, P. R., BACCARELLI, R., MARCIANO, L. H., LAURIS, J. R., TEIXEIRA, M. J., URA,
 S. & VIRMOND, M. C. The DN-4 questionnaire has a higher sensitivity as screening tool for neuropathic pain in leprosy. 14th World Congress on Pain, 2012 Milan. IASP.

- SUNEETHA, L. M., VARDHINI, D., SUNEETHA, S., BALASUBRAMANIAN, A. S., JOB, C.
 K. & SCOLLARD, D. 2001. Biochemical aspects of Mycobacterium leprae binding proteins: a review of their role in pathogenesis. *Int J Lepr Other Mycobact Dis*, 69, 341-8.
- TAMPIN, B., BRIFFA, N. K., GOUCKE, R. & SLATER, H. 2013. Identification of neuropathic pain in patients with neck/upper limb pain: application of a grading system and screening tools. *Pain*, 154, 2813-22.
- TAPINOS, N., OHNISHI, M. & RAMBUKKANA, A. 2006. ErbB2 receptor tyrosine kinase signaling mediates early demyelination induced by leprosy bacilli. *Nat Med*, 12, 961-6.
- THAKKAR, S. & PATEL, S. V. 2014. Clinical profile of leprosy patients: a prospective study. *Indian J Dermatol*, 59, 158-62.
- TORRANCE, N., SMITH, B. H., BENNETT, M. I. & LEE, A. J. 2006. The epidemiology of chronic pain of predominantly neuropathic origin. Results from a general population survey. *J Pain*, **7**, 281-9.
- TREEDE, R. D. & COLE, J. D. 1993. Dissociated secondary hyperalgesia in a subject with a large-fibre sensory neuropathy. *Pain*, 53, 169-74.
- TREEDE, R. D., JENSEN, T. S., CAMPBELL, J. N., CRUCCU, G., DOSTROVSKY, J. O., GRIFFIN, J. W., HANSSON, P., HUGHES, R., NURMIKKO, T. & SERRA, J. 2008. Neuropathic pain: redefinition and a grading system for clinical and research purposes. *Neurology*, 70, 1630-5.
- TREEDE, R. D., ROLKE, R., ANDREWS, K. & MAGERL, W. 2002. Pain elicited by blunt pressure: neurobiological basis and clinical relevance. *Pain*, 98, 235-40.
- TRUMAN, R. W., SINGH, P., SHARMA, R., BUSSO, P., ROUGEMONT, J., PANIZ-MONDOLFI, A., KAPOPOULOU, A., BRISSE, S., SCOLLARD, D. M., GILLIS, T. P.
 & COLE, S. T. 2011. Probable zoonotic leprosy in the southern United States. *N Engl J Med*, 364, 1626-33.
- ÜÇEYLER, N. & SOMMER, C. 2011. Neuropathic Pain Assessment An Overview of Existing Guidelines and Discussion Points for the Future. *European Neurological Review*, 6, 128-131.

- UNAL-CEVIK, I., SARIOGLU-AY, S. & EVCIK, D. 2010. A comparison of the DN4 and LANSS questionnaires in the assessment of neuropathic pain: validity and reliability of the Turkish version of DN4. *J Pain*, 11, 1129-35.
- UNRUH, A. M. 1996. Gender variations in clinical pain experience. *Pain*, 65, 123-67.
- VALK, G. D., DE SONNAVILLE, J. J., VAN HOUTUM, W. H., HEINE, R. J., VAN EIJK, J. T., BOUTER, L. M. & BERTELSMANN, F. W. 1997. The assessment of diabetic polyneuropathy in daily clinical practice: reproducibility and validity of Semmes Weinstein monofilaments examination and clinical neurological examination. *Muscle Nerve*, 20, 116-8.
- VAN BRAKEL, W. H., DE SOLDENHOFF, R. & MCDOUGALL, A. C. 1992. The allocation of leprosy patients into paucibacillary and multibacillary groups for multidrug therapy, taking into account the number of body areas affected by skin, or skin and nerve lesions. *Lepr Rev*, 63, 231-46.
- VAN BRAKEL, W. H., KHAWAS, I. B. & LUCAS, S. B. 1994a. Reactions in leprosy: an epidemiological study of 386 patients in west Nepal. *Lepr Rev*, 65, 190-203.
- VAN BRAKEL, W. H., NICHOLLS, P. G., DAS, L., BARKATAKI, P., MADDALI, P., LOCKWOOD, D. N. & WILDER-SMITH, E. 2005a. The INFIR Cohort Study: assessment of sensory and motor neuropathy in leprosy at baseline. *Lepr Rev*, 76, 277-95.
- VAN BRAKEL, W. H., NICHOLLS, P. G., DAS, L., BARKATAKI, P., SUNEETHA, S. K., JADHAV, R. S., MADDALI, P., LOCKWOOD, D. N., WILDER-SMITH, E. & DESIKAN, K. V. 2005b. The INFIR Cohort Study: investigating prediction, detection and pathogenesis of neuropathy and reactions in leprosy. Methods and baseline results of a cohort of multibacillary leprosy patients in north India. *Lepr Rev*, 76, 14-34.
- VAN BRAKEL, W. H., NICHOLLS, P. G., WILDER-SMITH, E. P., DAS, L., BARKATAKI, P. & LOCKWOOD, D. N. 2008a. Early Diagnosis of Neuropathy in Leprosy-Comparing Diagnostic Tests in a Large Prospective Study (the INFIR Cohort Study). *PLoS Negl Trop Dis*, 2, e212.
- VAN BRAKEL, W. H., NICHOLLS, P. G., WILDER-SMITH, E. P., DAS, L., BARKATAKI, P., LOCKWOOD, D. N. & GROUP, I. S. 2008b. Early diagnosis of neuropathy in

leprosy--comparing diagnostic tests in a large prospective study (the INFIR cohort study). *PLoS Negl Trop Dis*, 2, e212.

- VAN BRAKEL, W. H., SAUNDERSON, P., SHETTY, V., BRANDSMA, J. W., POST, E., JELLEMA, R. & MCKNIGHT, J. 2007. International workshop on neuropathology in leprosy--consensus report. *Lepr Rev*, 78, 416-33.
- VAN BRAKEL, W. H., SHUTE, J., DIXON, J. A. & ARZET, H. 1994b. Evaluation of sensibility in leprosy--comparison of various clinical methods. *Lepr Rev*, 65, 106-21.
- VAN BRAKEL, W. H., SIHOMBING, B., DJARIR, H., BEISE, K., KUSUMAWARDHANI, L., YULIHANE, R., KURNIASARI, I., KASIM, M., KESUMANINGSIH, K. I. & WILDER-SMITH, A. 2012. Disability in people affected by leprosy: the role of impairment, activity, social participation, stigma and discrimination. *Glob Health Action*, 5.
- VAN VEEN, N. H., MEIMA, A. & RICHARDUS, J. H. 2006. The relationship between detection delay and impairment in leprosy control: a comparison of patient cohorts from Bangladesh and Ethiopia. *Lepr Rev*, 77, 356-65.
- VASO, A., ADAHAN, H. M., GJIKA, A., ZAHAJ, S., ZHURDA, T., VYSHKA, G. & DEVOR, M. 2014. Peripheral nervous system origin of phantom limb pain. *Pain*.
- VERDUGO, R. & OCHOA, J. L. 1992. Quantitative somatosensory thermotest. A key method for functional evaluation of small calibre afferent channels. *Brain*, 115 (Pt 3), 893-913.
- VERMA, K. K. & GAUTAM, S. 1994. Effect of rehabilitation on the prevalence of psychiatric morbidity among leprosy patients. *Indian J Psychiatry*, 36, 183-6.
- VETTER, T. R. 2007. A primer on health-related quality of life in chronic pain medicine. *Anesth Analg*, 104, 703-18.
- VILLARROEL, M. F., ORSINI, M. B., GROSSI, M. A. & ANTUNES, C. M. 2007a. Impaired warm and cold perception thresholds in leprosy skin lesions. *Lepr Rev*, 78, 110-21.
- VILLARROEL, M. F., ORSINI, M. B., LIMA, R. C. & ANTUNES, C. M. 2007b. Comparative study of the cutaneous sensation of leprosy-suspected lesions using Semmes-Weinstein monofilaments and quantitative thermal testing. *Lepr Rev*, 78, 102-9.

- VINIK, A. I., SUWANWALAIKORN, S., STANSBERRY, K. B., HOLLAND, M. T., MCNITT, P.
 M. & COLEN, L. E. 1995. Quantitative measurement of cutaneous perception in diabetic neuropathy. *Muscle Nerve*, 18, 574-84.
- VLCKOVA-MORAVCOVA, E., BEDNARIK, J., BELOBRADKOVA, J. & SOMMER, C. 2008a. Small-fibre involvement in diabetic patients with neuropathic foot pain. *Diabet Med*, 25, 692-9.
- VLCKOVA-MORAVCOVA, E., BEDNARIK, J., DUSEK, L., TOYKA, K. V. & SOMMER, C. 2008b. Diagnostic validity of epidermal nerve fiber densities in painful sensory neuropathies. *Muscle Nerve*, 37, 50-60.
- VOOREND, C. G. & POST, E. B. 2013. A systematic review on the epidemiological data of erythema nodosum leprosum, a type 2 leprosy reaction. *PLoS Negl Trop Dis*, 7, e2440.
- VRETHEM, M., BOIVIE, J., ARNQVIST, H., HOLMGREN, H. & LINDSTROM, T. 2002. Painful polyneuropathy in patients with and without diabetes: clinical, neurophysiologic, and quantitative sensory characteristics. *Clin J Pain*, 18, 122-7.
- WAGENAAR, I., BRANDSMA, W., POST, E., VAN BRAKEL, W., LOCKWOOD, D., NICHOLLS, P., SAUNDERSON, P., SMITH, C., WILDER-SMITH, E. & RICHARDUS,
 J. H. 2012. Two randomized controlled clinical trials to study the effectiveness of prednisolone treatment in preventing and restoring clinical nerve function loss in leprosy: the TENLEP study protocols. *BMC Neurol*, 12, 159.
- WALK, D., SEHGAL, N., MOELLER-BERTRAM, T., EDWARDS, R. R., WASAN, A.,
 WALLACE, M., IRVING, G., ARGOFF, C. & BACKONJA, M. M. 2009.
 Quantitative sensory testing and mapping: a review of nonautomated quantitative methods for examination of the patient with neuropathic pain. *Clin J Pain*, 25, 632-40.
- WALKER, S. L. & LOCKWOOD, D. N. 2006. The clinical and immunological features of leprosy. *Br Med Bull*, 77-78, 103-21.
- WALKER, S. L. & LOCKWOOD, D. N. 2008. Leprosy type 1 (reversal) reactions and their management. *Lepr Rev*, 79, 372-86.

- WALKER, S. L., NICHOLLS, P. G., BUTLIN, C. R., NERY, J. A., ROY, H. K., RANGEL, E., SALES, A. M. & LOCKWOOD, D. N. 2008. Development and validation of a severity scale for leprosy type 1 reactions. *PLoS Negl Trop Dis*, 2, e351.
- WALKER, S. L., WATERS, M. F. & LOCKWOOD, D. N. 2007. The role of thalidomide in the management of erythema nodosum leprosum. *Lepr Rev*, 78, 197-215.
- WALLACE, V. C. J. & RICE, A. S. 2008. Applied physiology: neuropathic pain. *In:*WILSON, P. R., WATSON, P. J., HAYTHORNTHWAITE, J. A. & JENSEN, T. S.
 (eds.) *Clinical pain management chronic pain* 2nd ed. London: Hodder Arnold.
- WEIDNER, C., SCHMELZ, M., SCHMIDT, R., HANSSON, B., HANDWERKER, H. O. & TOREBJORK, H. E. 1999. Functional attributes discriminating mechano-insensitive and mechano-responsive C nociceptors in human skin. *J Neurosci*, 19, 10184-90.
- WEINSTEIN, S. 1968. Intensive and extensive aspects of tactile sensitivity as a function of body part, sex, and laterality in the skin senses.
- WEIS, J., KATONA, I., MULLER-NEWEN, G., SOMMER, C., NECULA, G., HENDRICH, C., LUDOLPH, A. C. & SPERFELD, A. D. 2011. Small-fiber neuropathy in patients with ALS. *Neurology*, 76, 2024-9.
- WESTERMANN, A., KRUMOVA, E. K., PENNEKAMP, W., HORCH, C., BARON, R. & MAIER, C. 2012. Different underlying pain mechanisms despite identical pain characteristics: a case report of a patient with spinal cord injury. *Pain*, 153, 1537-40.
- WHO. 1946. WHO definition of Health [Online]. Available: <u>http://www.who.int/about/definition/en/print.html</u>.
- WHO 1948. Constitution of the World Health Organization. Basic documents. Geneva, Switzerland: World Health Organization.
- WHO 1988. WHO Expert Committee on Leprosy. *World Health Organ Tech Rep Ser,* 768, 1-51.
- WHO 1994. *Chemotherapy of leprosy : report of a WHO study group,* Geneva, World Health Organization.
- WHO 2006. Global Strategy for Further Reducing the Leprosy Burden and Sustaining Leprosy Control Activities 2006-2010, Operational Guidelines.

- WHO 2009a. Enhanced Global Strategy for Further Reducing the Disease Burden due to Leprosy (plan period: 2011-2015). New Delhi: WHO Regional Office for South-East Asia.
- WHO. 2009b. International Classification of Functioning, Disability and Health (ICF);
 [Online]. Available: <u>http://www.who.int/classifications/icf/en/</u> [Accessed 10th August 2009].
- WHO 2012. WHO Expert Committee on Leprosy 8th Technical Report 2012. World Health Organ Tech Rep Ser. Geneva, Switzerland. Available at: <u>http://www.searo.who.int/entity/global leprosy programme/publications/</u> <u>8th expert comm 2012.pdf</u>.
- WHO. 2013a. *Country Profile, India* [Online]. Available: http://www.who.int/countries/ind/en/.
- WHO 2013b. Global leprosy situation, 2012; Reported Leprosy cases, as at 30 August, 2013, taken from Wkly Epidemiol Rec, No. 35, 2013, 88, 365–380. <u>http://www.who.int/wer/en/</u>.
- WHO. 2014. Process of translation and adaptation of instruments [Online]. Available:

http://www.who.int/substance abuse/research tools/translation/en/#.

- WILKIE, D. J., HUANG, H. Y., REILLY, N. & CAIN, K. C. 2001. Nociceptive and neuropathic pain in patients with lung cancer: a comparison of pain quality descriptors. J Pain Symptom Manage, 22, 899-910.
- WILLIAMS, L. S., JONES, W. J., SHEN, J., ROBINSON, R. L. & KROENKE, K. 2004. Outcomes of newly referred neurology outpatients with depression and pain. *Neurology*, 63, 674-7.
- WOOLF, C. J. & MANNION, R. J. 1999. Neuropathic pain: aetiology, symptoms, mechanisms, and management. *Lancet*, 353, 1959-64.
- WRIGHT, E., BREW, B., ARAYAWICHANONT, A., ROBERTSON, K.,
 SAMINTHARAPANYA, K., KONGSAENGDAO, S., LIM, M., VONTHANAK, S., LAL,
 L., SARIM, C., HUFFAM, S., LI, P., IMRAN, D., LEWIS, J., LUN, W. H.,
 KAMARULZAMAN, A., TAU, G., ALI, S. T., KISHORE, K., BAIN, M. P., DWYER,
 R., MCCORMACK, G., HELLARD, M., CHERRY, C., MCARTHUR, J. &

WESSELINGH, S. 2008. Neurologic disorders are prevalent in HIV-positive outpatients in the Asia-Pacific region. *Neurology*, **71**, 50-6.

- WU, E. Q., BORTON, J., SAID, G., LE, T. K., MONZ, B., ROSILIO, M. & AVOINET, S.
 2007. Estimated prevalence of peripheral neuropathy and associated pain in adults with diabetes in France. *Curr Med Res Opin*, 23, 2035-42.
- XU, G. Y., HUANG, L. Y. & ZHAO, Z. Q. 2000. Activation of silent mechanoreceptive cat C and Adelta sensory neurons and their substance P expression following peripheral inflammation. *J Physiol*, 528 Pt 2, 339-48.
- YAWALKAR, S. J., MCDOUGALL, A. C., LANGUILLON, J., GHOSH, S., HAJRA, S. K., OPROMOLLA, D. V. & TONELLO, C. J. 1982. Once-monthly rifampicin plus daily dapsone in initial treatment of lepromatous leprosy. *Lancet*, **1**, 1199-202.
- ZELMAN, D. C., GORE, M., DUKES, E., TAI, K. S. & BRANDENBURG, N. 2005. Validation of a modified version of the brief pain inventory for painful diabetic peripheral neuropathy. *J Pain Symptom Manage*, 29, 401-10.
- ZHAO, P., WAXMAN, S. G. & HAINS, B. C. 2006. Sodium channel expression in the ventral posterolateral nucleus of the thalamus after peripheral nerve injury. *Mol Pain*, 2, 27.
- ZHOU, L., KITCH, D. W., EVANS, S. R., HAUER, P., RAMAN, S., EBENEZER, G. J., GERSCHENSON, M., MARRA, C. M., VALCOUR, V., DIAZ-ARRASTIA, R., GOODKIN, K., MILLAR, L., SHRIVER, S., ASMUTH, D. M., CLIFFORD, D. B., SIMPSON, D. M. & MCARTHUR, J. C. 2007. Correlates of epidermal nerve fiber densities in HIV-associated distal sensory polyneuropathy. *Neurology*, 68, 2113-9.

Appendices

Appendix A

Ethical Approvals

Appendix 1. LSHTM Ethical approval

ondon school of Hygle	ene & Tropical Medicine	10	NDON 🚄	10
eppel Street, London W	C1E 7HT	ŝĈ	HOOL of	
Inited Kingdom		ĬŤ	VGIENE	
Switchboard: +44 (0)20	7636 8636	&Т	ROPICAL	1. A
www.lshtm.ac.uk		M	EDICINE	7.95
	Observational / Intervention	ns Research Ethics Con	unittee	
Diana Lockwood				
Professor of Tropical N	ledicine			
CRD/ITD				
Latter				
12 July 2012				
Dear Professor Lockwo	ood,			
Study Title:	The development of new tools for	r the investigation of ne	uropathic pain in	
LSHTM ethics ref:	6181			
Thank you for your let information on the abo	ter of 25 June 2012, responding to t ove research and submitting revised	he Observational Comm documentation.	ittee's request for fur	ther
The further informatio	n has been considered by the Comn	uittee.		
Confirmation of ethic	al opinion			
Confirmation of ethic	al opinion ittee. I am pleased to confirm a favo	urable ethical opinion f	or the above research	on the basis
Confirmation of ethic On behalf of the Comm described in the applic	al opinion ittee, I am pleased to confirm a favo ation form, protocol and supporting	urable ethical opinion f documentation as revi	or the above research sed, subject to the cor	on the basis iditions
Confirmation of ethic On behalf of the Comm described in the applic specified below.	c al opinion iittee, I am pleased to confirm a favo ation form, protocol and supporting	urable ethical opinion f documentation as revi	or the above research sed, subject to the cor	on the basis aditions
Confirmation of ethic On behalf of the Comm described in the applic specified below.	cal opinion iittee, I am pleased to confirm a favo ation form, protocol and supporting nurable opinion	urable ethical opinion f documentation as revi	or the above research sed, subject to the cor	on the basis aditions
Confirmation of ethic On behalf of the Comm described in the applic specified below. Conditions of the fave	cal opinion ittee, I am pleased to confirm a favo ation form, protocol and supporting purable opinion	urable ethical opinion f documentation as revi	or the above research sed, subject to the cor	on the basis ditions
Confirmation of ethic On behalf of the Comm described in the applic specified below. Conditions of the fave Approval is dependent	cal opinion ittee, I am pleased to confirm a favo ation form, protocol and supporting purable opinion : on local ethical approval having be	urable ethical opinion f ; documentation as revi en received, where relev	or the above research sed, subject to the cor yant.	on the basis aditions
Confirmation of ethic On behalf of the Comm described in the applic specified below. Conditions of the fave Approval is dependent Approved documents	al opinion ittee, I am pleased to confirm a favo ation form, protocol and supporting purable opinion : on local ethical approval having be	urable ethical opinion f g documentation as revi en received, where relev	or the above research sed, subject to the cor vant.	on the basis uditions
Confirmation of ethic On behalf of the Comm described in the applic specified below. Conditions of the fave Approval is dependent Approved document: The final list of docum	cal opinion ittee, I am pleased to confirm a favo action form, protocol and supporting ourable opinion : on local ethical approval having be s ents reviewed and approved by the	urable ethical opinion f documentation as revi en received, where relev Committee is as follows	or the above research sed, subject to the cor vant.	on the basis aditions
Confirmation of ethic On behalf of the Comm described in the applic specified below. Conditions of the fave Approval is dependent Approved documents The final list of docum	cal opinion ittee, I am pleased to confirm a favo vation form, protocol and supporting ourable opinion con local ethical approval having be s ents reviewed and approved by the	urable ethical opinion f g documentation as revi en received, where relev Committee is as follows	or the above research sed, subject to the cor vant.	on the basis ditions
Confirmation of ethic On behalf of the Comm described in the applic specified below. Conditions of the fave Approval is dependent Approved documents The final list of docum	cal opinion ittee, I am pleased to confirm a favor vation form, protocol and supporting ourable opinion con local ethical approval having be s ents reviewed and approved by the Document	urable ethical opinion f ; documentation as revi en received, where relev Committee is as follows Version	or the above research sed, subject to the cor vant.	on the basis iditions
Confirmation of ethic On behalf of the Comm described in the applic specified below. Conditions of the fave Approval is dependent Approved documents The final list of docum LSHTM ethics applicat	cal opinion ittee, I am pleased to confirm a favor vation form, protocol and supporting ourable opinion : on local ethical approval having be s ents reviewed and approved by the Document ion	en received, where releved to a solution of the second sec	or the above research sed, subject to the cor vant.	on the basis ditions
Confirmation of ethic On behalf of the Comm described in the applic specified below. Conditions of the fave Approval is dependent Approved documents The final list of docum LSHTM ethics applicat Protocol	cal opinion ittee, I am pleased to confirm a favor vation form, protocol and supporting ourable opinion : on local ethical approval having be s ents reviewed and approved by the Document ion	en received, where releved to a service of the serv	or the above research sed, subject to the cor vant. : Date 21/06/2012	on the basis ditions
Confirmation of ethic On behalf of the Comm described in the applic specified below. Conditions of the fave Approval is dependent Approved documents The final list of docum LSHTM ethics applicat Protocol Information Sheet	cal opinion ittee, I am pleased to confirm a favor vation form, protocol and supporting ourable opinion : on local ethical approval having be s ents reviewed and approved by the Document ion	en received, where releved to the second sec	or the above research sed, subject to the cor vant. : Date 21/06/2012 04/04/2012	on the basis ditions
Confirmation of ethic On behalf of the Comm described in the applic specified below. Conditions of the fave Approval is dependent Approved documents The final list of docum LSHTM ethics applicat Protocol Information Sheet Consent form	cal opinion ittee, I am pleased to confirm a favor vation form, protocol and supporting ourable opinion : on local ethical approval having be s ents reviewed and approved by the Document ion	en received, where releved to the second sec	or the above research sed, subject to the cor vant. : : Date 21/06/2012 04/04/2012 04/04/2012	on the basis ditions
Confirmation of ethic On behalf of the Comm described in the applic specified below. Conditions of the fave Approval is dependent Approved documents The final list of docum LSHTM ethics applicat Protocol Information Sheet Consent form	cal opinion littee, I am pleased to confirm a favo action form, protocol and supporting ourable opinion : on local ethical approval having be s ents reviewed and approved by the Document ion	en received, where relev Committee is as follows Na. Version N3.2 V1.0 V1.0	or the above research sed, subject to the cor vant. : Date 21/06/2012 04/04/2012 04/04/2012	on the basis ditions
Confirmation of ethic On behalf of the Comm described in the applic specified below. Conditions of the fave Approval is dependent Approved documents The final list of docum LSHTM ethics applicat Protocol Information Sheet Consent form After ethical review	ral opinion ittee, I am pleased to confirm a favor iation form, protocol and supporting ourable opinion : on local ethical approval having be 5 ents reviewed and approved by the Document ion	en received, where relev Committee is as follows N/3.2 V1.0 V1.0	or the above research sed, subject to the cor vant. : Date 21/06/2012 04/04/2012 04/04/2012	on the basis aditions
Confirmation of ethic On behalf of the Comm described in the applic specified below. Conditions of the fave Approval is dependent Approved documents The final list of docum LSHTM ethics applicat Protocol Information Sheet Consent form After ethical review Any subsequent chang	eal opinion ittee, I am pleased to confirm a favor iation form, protocol and supporting ourable opinion : on local ethical approval having be s ents reviewed and approved by the Document ion es to the application must be submited in the submited based in	urable ethical opinion f documentation as revi en received, where relev Committee is as follows Version n/a V3.2 V1.0 V1.0 V1.0	or the above research sed, subject to the cor vant. Date 21/06/2012 04/04/2012 04/04/2012 ia an E2 amendment f	on the basis aditions
Confirmation of ethic On behalf of the Comm described in the applic specified below. Conditions of the fave Approval is dependent Approved documents The final list of docum LSHTM ethics applicat Protocol Information Sheet Consent form After ethical review Any subsequent chang studies are also requir	eal opinion ittee, I am pleased to confirm a favore ittee, I am pleased to confirm the submit of the study please patific the committee of ittee, I am pleased to confirm the submit	urable ethical opinion f gocumentation as revi en received, where relev Committee is as follows Version N/a V3.2 V1.0 V1.0 V1.0 tted to the Committee v any serious adverse eva	or the above research sed, subject to the cor vant. : Date 21/06/2012 04/04/2012 04/04/2012 ia an E2 amendment f ents which occur duri	on the basis aditions
Confirmation of ethic On behalf of the Comm described in the applic specified below. Conditions of the fave Approval is dependent Approved documents The final list of docum LSHTM ethics applicat Protocol Information Sheet Consent form After ethical review Any subsequent chang studies are also requir via form E4. At the end	eal opinion ittee, I am pleased to confirm a favor ittee, I am please notify the com ittee, I am please notify the com	aurable ethical opinion f gocumentation as revi en received, where relev Committee is as follows Version N/a V3.2 V1.0 V1.0 V1.0 tted to the Committee v any serious adverse even mittee via form E5.	or the above research sed, subject to the cor vant. Date 21/06/2012 04/04/2012 04/04/2012 ia an E2 amendment f ents which occur duri	on the basis aditions
Confirmation of ethic On behalf of the Comm described in the applic specified below. Conditions of the fave Approval is dependent Approved document: The final list of docum LSHTM ethics applicat Protocol Information Sheet Consent form After ethical review Any subsequent chang studies are also requir via form E4. At the end Yours sincerely,	eal opinion ittee, I am pleased to confirm a favor ittee, I am please notify the com ittee, I am please notify the com	aurable ethical opinion f gocumentation as revi en received, where relev Committee is as follows Version N/a V3.2 V1.0 V1.0 V1.0 tted to the Committee v any serious adverse even mittee via form E5.	or the above research sed, subject to the cor vant. : Date 21/06/2012 04/04/2012 04/04/2012 ia an E2 amendment f ents which occur duri	form. All ng the project
Confirmation of ethic On behalf of the Comm described in the applic specified below. Conditions of the fave Approval is dependent Approved documents The final list of docum LSHTM ethics applicat Protocol Information Sheet Consent form After ethical review Any subsequent chang studies are also requir via form E4. At the end Yours sincerely,	eal opinion ittee, I am pleased to confirm a favor ittee, I am pleased to confirm a	en received, where relev Committee is as follows Version N/a V3.2 V1.0 V1.0 V1.0 tted to the Committee v any serious adverse even mittee via form E5.	or the above research sed, subject to the cor vant. Date 21/06/2012 04/04/2012 04/04/2012 ia an E2 amendment f ents which occur duri	on the basis aditions
Confirmation of ethic On behalf of the Comm described in the applic specified below. Conditions of the fave Approval is dependent Approved documents The final list of docum LSHTM ethics applicat Protocol Information Sheet Consent form After ethical review Any subsequent chang studies are also requir via form E4. At the end Yours sincerely,	eal opinion ittee, I am pleased to confirm a favor ittee, I am pleased to confirm a favor ourable opinion es to the application must be submited to notify the ethics committee of d of the study, please notify the com	en received, where relevent committee is as follows version value va va value value value value	or the above research sed, subject to the cor vant. : : Date 21/06/2012 04/04/2012 04/04/2012 04/04/2012 ia an E2 amendment f ents which occur duri	on the basis aditions
Confirmation of ethic On behalf of the Comm described in the applic specified below. Conditions of the fave Approval is dependent Approved document: The final list of docum LSHTM ethics applicat Protocol Information Sheet Consent form After ethical review Any subsequent chang studies are also requir via form E4. At the end Yours sincerely,	eal opinion ittee, I am pleased to confirm a favor ittee, I am pleased to confirm a favor ittee, I am pleased to confirm a favor ittee, I am pleased to confirm a favor ourable opinion item is on local ethical approval having be s ents reviewed and approved by the Document ion es to the application must be submited of the study, please notify the com	en received, where releved committee is as follows version va v3.2 v1.0 v1.0 v1.0 v1.0 tted to the Committee vi any serious adverse even mittee via form E5.	or the above research sed, subject to the cor vant. : : : : : : : : : : : : : : : : : : :	on the basis aditions
Confirmation of ethic On behalf of the Comm described in the applic specified below. Conditions of the fave Approval is dependent Approved document: The final list of docum LSHTM ethics applicat Protocol Information Sheet Consent form After ethical review Any subsequent chang studies are also requir via form E4. At the end Yours sincerely,	eal opinion ittee, I am pleased to confirm a favor ittee, I am pleased to confirm a favor aution form, protocol and supporting ourable opinion : on local ethical approval having be s ents reviewed and approved by the Document ion es to the application must be submited to notify the ethics committee of d of the study, please notify the com	en received, where relev Committee is as follows Version Na V3.2 V1.0 V1.0 V1.0 tted to the Committee v any serious adverse eve mittee via form E5.	or the above research sed, subject to the cor vant. : : : : : : : : : : : : : : : : : : :	on the basis aditions

Improving health worldwide

Page 1 of 1

11	spendix 2. Imperiar Conege London	
	Imperial College London	Imperial College Research Ethics Committee Imperial College London G02 Sir Alexander Fleming Building South Kensington Campus SW7 2AZ Tel: +44 (0)207 594 1770 Fax: +44 (0) 207 594 1792 researchethicscommittee@imperial.ac.uk.
	03 March 2011	
	Professor Andrew Rice, Section of Anaesthetics, Department of Surgery and Cancer, Imperial College London, Chelsea and Westminster Campus, 369 Fulham Road, London, SW 10 9NH	
	Dear Professor Rice,	
	Study Title: Healthy Volunteer Quantitative Sensor Study	y Testing (QST); Quality Control
	ICREC reference: ICREC_11_2_3	
	CI: Professor Andrew Rice	
	The above study was approved by your HoD on Research Office on the 1 st March 2011.	17 th February 2011 and by the Joint
	Under the Imperial College Research Ethics Comm reviewed by the Joint Research Office and Head of D no significant ethical issues have been identified in the approved without requiring it to go to full committee.	nittee process, a study that has been ivision/Department (or Principal), where he protocol or ethics application, can be
	Documents	
	The documents reviewed were:	
	 ICREC Application form Participant Information Sheet V2 Consent form for healthy volunteers Recruitment Invitation email 	
	Yours sincerely, Internet of the second	
	Gary Roper, Head of Regulatory Compliance, Imperial College London	
	Imperial College of Science, Technology and Medicine	

Appendix 2. Imperial College London Ethical approval

Appendix 3. FMR Ethical approval



Appendix B

Participant Recruitment, Information Sheets and Consent Forms

Healthy volunteers study: Recruitment invitation, email ***This email has been approved by the ICREC Number: 11_2_3*** Dear All, We are seeking volunteers to participate in a study examining the way the skin sense touch, temperature, pressure and pain sensation using a well established technique called Quantitative Sensory Testing (QST). This is essentially a sophisticated clinical examination of the skin and would take about 1.5 hours of your time. Volunteers should be in good general health. If you would like more information or would like to volunteer please let me know. Please feel free to pass this invitation on to colleagues within Imperial Collage or LSHTM who might be interested in taking part. Many thanks Dr Omer Haroun +447411220020 +447536136026

Appendix 4. Healthy volunteer's recruitment invitation letter

Version 1.0 <15.02.11>



Appendix 5. Healthy volunteer's recruitment invitation in Hindi



Appendix 6. Healthy volunteer's recruitment invitation in Marathi


Appendix 7. Healthy volunteer information sheet







Appendix 8. Healthy information sheet – German version

 Abhauf der Untersuchungen Die Untersuchungsdaner beträgt ca.

□ 30 min
 □ 60 min
 □ 90 min
 □ 120 min
 □ 150 min

7 180 min.

Untersucht werden die folgenden Körperareale:

(A) Die quantitative sensorische Testung (QST)

Zu Beginn werden Sie körperlich auf neurologische Symptome hin un ursucht. Darauf folgt eine Untersuchung mittels im Folgenden beschriebener Methoden, um Vergleichsdaten zu Patienten zu erhoben. Dabei findet keine Hautverletzung statt.

Zunächst wird Ihre Warm- und Kaltempfindung sowie thre Wahrnehmung von Kätte- und Hitzeschmerzreizen geprüft. Hierfür wird thron eine ca. 13 cm² große Metallfäche auf die Haut gelegt. Diese hat eine Ausgangstemperatur von 32°C, die in einer festgelegten Reihenfolge ansteligt oder abfällt. Sobalit Sie Temperaturveränderungen oder Schmerzempfindungen un der Auflagofläche wahrnehmen, bitten wir Sie, eine Stop-Taste zu drücken, woraufhin die Metallfläche ihre Ausgangstemperatur wieder erreicht. Dieser Vorgang wiederholt sich insgesamt 2-mal für jedz Temperaturänderung in jedem Testareal. Es folgt die Prüfung Ihrer Wahrnehmungsschwellen für Berührungsreize und Nadetreize, die mittels eines Heares und einer Nadel unterschiedlichen Auflagedrucks auf der Haut geprüft wird. Darsuffolgend wird untersucht, ob hei Ihnen Schmerzen durch Berührung, eine sogenannte Allodynie, bestehen. Hierzu wird mit einem Wattebausch, einem Q-Tip sowie einem standardisierten Pinsel die Haut bestrichen bzw. leichter Druck auf die Haut ausgeübt. In einer weiteren Untersuchung wird ein Einzelreiz mit einer stumpfen Nadel auf die Haut aufgebracht. Im Abstand von zehn Sekunden folgt eine Serie von zehn identischen Nadelreizen innerhalb desselben Hautarcals. Unmittelbar nach dem Einzelreiz sowie erneut im Anschluss an die nachfulgende Reizserie erfolgt eine zusammenfassende Beurteilung der Impfindungsstärke mit Hilfe einer numerischen Skala von 0-100. Zur Untersuchung der Vibrationsschwelle wird eine Stimmgabel auf die Haut über einem Knochenvorsprung (z.B. an den Füßen über Innenknöchel, an den Händen über dem Handgelank, am Kopf seitlich über der Schläfe) platziert. Zur Prüfung der Wahrnehmung der Druckempfindlichkeit der Muskulatur wird über definierten Muskeln ein Druckstimulator aufgesetzt (2.B. an den Füßen über der Fußinneumuskulatur, an den Händen über dem Daumenhallen, am Gesicht über der Schläferregion). Mittels einer aufsteigenden Druckintensität kann die Druckschmerzschwelle über den entsprechenden Muskeln bestiment worden.

(B) Erfassung krankheitsspezifischer Daten mittels Fragehögen

Sie werden gebeten, verschiedere Fragehögen auszufüllen. Diese erfassen Informationen hinsichtlich möglicher Schmerzer, aber auch Informationen zu einer möglichen Beeinträchtigung fhres Alltagslebens und Ihrer allgemeinen Befindlichteit.

4. Datenverarbeitung und Datenschutz

Die im Rahmen dieser Studie erhobenen Daten werden auf einer mit Ihren persönlichen Daten versehenen Krankenakte in der Abteilung für Neurophysiologie im Zentrum für Biomeslizin und Medizintechnik Manchelm gespeichert. Zur Auswertung der erhobenen Daten können diese Daten <u>anonymistert mit einer Probundennummer</u>, d.h. ohne <u>Namensnennung</u>, innerhalb des Deutschen Forschungsverbundes Neuropathischer Schmerz an die beteiligten Institutionen für wissenschaftliche Zwecke weitergeleitet werden. Alle im Rahmen dieser Studie erhobenen Daten unterliegen der ärztlichen Schweigepflicht und werden strikt vertrauflich gemäß dem Datenschutz behandelt.

Ihre Daten sollten für die Durchführung des Projektes langfristig gespeichert worden. Sie können aber jederzeit ohne Angabe von Gränden Ihr Kinverständnis widerrafen und die Lüschung Ihrer Daten verlangen. Ihr Ansprechpartner ist in diesem Fall der die Untersuchung durchführende Mitarheiter, den Sie über die Telefonnummer 0621-3839926 erreichen.

> Henrouchung der Funktion und Bigenschniten des steuratioenschichten Nerverspolente mit Helfe der quaritheitven ensonischen Testene (QSF) im Reimen des Deutschen Eusekungswertneten Nermenstlöcher Schmetz (DPNS) Leitzer/M filt Normphyseilegie, Muliciaische Fahrlächkonstenn den Resecht Kaule Universite Heidelberg (2.10.2207; Werten 1

> > Continue...

5. Versicherungsschutz

Für die heschriebenen Untersuchungen besteht eine Unfall-/Wegeversicherung.

Der Versicherer ist der Badische Gemeinde-Versicherungs-Verhand, Durlacher Allee 56, 76131 Karlsruhe (Telefonnummer 0721 660-0), die Vertragsnummer ist V1087106. Versichert sind Unfälle, von denen Sie als versicherte Person auf den direkten Fahrten von Ihrer Wohnung zu der Untersuchungsstätte und zurtick betroffen worken. Der Versicherungsschutz bestcht ausschließlich während der Fahrten. Er beginnt auf dem Hinweg mit dem Verlassen der Wohnung bzw. auf dem Röckweg mit dem Verlassen der Untersuchungsstätte eind eindet mit der Ankunft an der Untersuchungsstätte bzw. mit der Röckkehr in die Wohnung. Der Versicherungsschutz entfällt, wenn die normale Dauer des Weges verlängert uder der Weg selbst durch rein private und eigenwirtschaftleiche Maßnahmen (z.B. Einkauf, Bestich von Wirtschaften zu Privatzweeken) unterbrochen wird.

Um den Versicherungsschutz im Bedarfsfall nicht zu geführden, sind gewisse Bedingungen zu erfüllen. Bitte berücksichtigen Sie folgende Punkte (Auszug aus den Allgemeinen Unfallversicherungsbedingungen 2001; Punkte 7 und 8 Obliegenheiten des Versicherten):

7. Was ist nach einem Unfall zu beschten (Obliegenheiten)?

Ohne Ihre Mitwidtung und die der versicherten Person können wir unsere Leistung nicht erbringen.

- Nach einem Unfall, der voraussichtlich eine Leistungspflicht herbeiführt, müssen Sie oder die versicherte Person unverzüglich einen Arzt binzuziehen, seine Anordnungen befolgen und uns unterrichten.
 Die von uns übersendte Unfallanzeige müssen Sie oder die versieherte Person wahrheitsgemäß ausfüllen und uns unver-
- Werden Ärzte von uns heat. Fragt, mess sich die versicherte Person auch von diesen untersuchen lassen. Die hotwondigen Kosten einschließlich eines dudurch entstandenen Verdienstausfulls trugen wir.
- Die Ärzte, die die versicherte Person auch aus anderen Anlässen behandelt oder untersucht haben, andere Versicherungsträger und Behörden sind zu ermöchtigen, alle erfordenlichen Auslatinfte zu erteilen.
 Hat der Unfall den Tod zur Folge, ist uns dies innerhalb von 48 Stunden zu melder, auch wenn nus der Unfall sehon
- Hat der Unfall den Tod zur Folge, ist uns dies innerhalb von 48 Stunden zu melder, auch wenn uns der Unfall schon angezeugt war. Uns ist das Recht zu verschaften, gegebenenfalls eine Obduktion durch einen von uns beauftragten Arzt vorzunehmen zu lassen.

Welche Folgen hat die Nichtbeachtung von Obliegenheiten?

Wird eine nach Eintritt eines Unfalls zu arfüllande Obliegenbeit verletzt, verlieren Sie den Versieherungsschutz, es sei derm, Siehaben die Ubliegenheit werder versätzlich noch greb fehrlässig verletzt.

Bei grob fahrlässiger Verletzung behahen Sie hassweit den Versicherungsseheitz, als die Verletzung weder Einfluss auf die Feststellung des Leistungsfahls nech auf die Bemessung der Leistung gelneb hat. Bei vorsitzlieher Verletzung behahen Sie in diesen Föllen den Versicherungsschutz insoweit nur, wenn die Verletzung nicht ge-

Bei vorsitzlicher Verletzung behalten Sie in diesen Füllen den Versicherungsschutz insoweit nur, wenn die Verletzung nicht geeignet war, unsere Interessen erns haft zu hesintrichtigen oder wenn Sie kein erhehliches Verschulden trifft.

Kontaktpersonen für Fragen:

Dr. Doreen Pfau

L.

Telefonnummer: 0621-3839926

Uness dang de Bucklon auf Egencehoften des sonnross oprischen Nersenysterne mit führe der gemnitztiver senser leden Denng ((%)) um Rebres des Doutes on Kornehondes Nerreportieder Kahrinze (DENS) Leronthil für Nangehoefsegie, Matizinsche Fachtlu Version für Progressie-Kaele-Driversität Heikelbag 02.10.2007, Version 1



Appendix 9. Healthy volunteer information sheet – Hindi version







Appendix 10. Healthy volunteer information sheet – Marathi version

Continue...

भाग २

मी जर सहाभाग घेण्याचे ठरवले तर माझे काय होईल ?

अभ्यासात भाग घेण्याचे तुम्हीं ठरवले तर तुमच्या सीयीनुसार अन्वेषकाबरोबर तुमच्या भेटीची वैळ निश्वित केली जाईल. शक्य असल्यास QST मोजणीच्या २४ तास आधी कुठलेही वेदनानाशक औषभ तुम्हीं घेऊ नये असे आम्हीं तुम्हांला सांगू, ह्या कालावधीत जर तुम्हीं वेदनानाशक औषध धेतले तर कृपया संशोधन अन्वेषकाला तरो सांगा.

हात व पाय यांची संवेदनात्मक (sensory) गुणात्मक चाचणी व त्वचा जीवोतक परीक्षा तुमच्या त्वचेतील संवेदनात्मक मञ्जातंतूचे परीक्षण करण्याचा हा तपशीलवार अभ्यास आहे, ज्यातून तापमान, दाव, तीक्ष्णता आणि स्पर्श ह्यातील बदल शोधले जातात.

तापमान बदलणारे म्हणजेच थंड व गरम होणारे खेटे साधन तुमच्या त्वचेवर लावून तापलानातील बदल जाणून घेण्याची तुमची क्षमता मापली जाईल. तुम्हांला कुठलाही बदल जाणवला की सांगा असे तम्हांला सांगण्यात येईल.

हलका त्पर्श जाणून घेण्याची तुमची क्षमता अतिशय सूक्ष्म तंतू वापरून ठरवली जाईल. ह्या वेळी देखील तुम्हांता काही जाणवले की सांगा असे वीचान्यीत येइल .

तीक्ष्णता समजून घेण्यासाठी जिच्यामुळे तुमच्या त्वचेला हानी पोचणार नाही अशे छोटे साधन तुमच्या त्वचेवर लावली जाईल, ज्यावेळी तुम्हांला तीक्ष्णता जाणवेल त्यावेळी सांगा असे तुम्हांला वीचार्ज्यात वेइल.

दबाव टाकण्याप्ताठी बोथट मापक वापरले जाईल. ज्या क्षणी दबाव असुखाकारक होऊ लागेल त्यावेळी तुम्हीं ते सांगा असे तुम्हांना सांगितने जाईल.

जी जापने तुम्हांना थोडी अस्वस्थता आणेल अशी काही मापने देखील वापरली जातील. जर तुम्हाला परिक्षतात पुढे जायचे नसेल तर कृपया आम्हाला सांगा.

३ मिली मीटर छिंद्रक त्वचा जीवोतक वापरून तुमये हात आणि पाय झांची त्वचा जीवोतक चावणी देखील केली जाईल. आम्ही तुम्हांना वैध सामान्य आरोग्य प्रश्न्वावली १२ ची उत्तरे टेण्याची देखील आम्ही तुम्हांना वितंती करू.

सहभाग धेण्यात काय धोके व त्याचे काय तोटे आहेत ?

तुम्हांल आम्हांला अंदाजे दीड तास धावा लागेल हा सर्वात मोठा तोटा आहे. प्रतिकूल काही प्रसंग उद्भवेल झाची शक्यता नाही कारण आम्ही वैथ पद्धती वापरीत आहोत.

सहभाग घेण्याचे फायदे काय आहेत ?

सहभाग घेऊन तुम्हीं आम्हांला निरोगी स्वयंसेवक नियंत्रण माहिती गोळा करण्यासे मदत करणार आहात. आम्हीं ही प्रमाणानुसार संदर्भ माहिती, कुछरोग सहभागी माहिती वरोवर तुलना करू शकू. तुम्हाला थेट फायटा नहीं आहे.

Continue...





Appendix 11. Patient's recruitment in Hindi

रोगभरती निकष आम्हीं शोध घेत आहोत: - कुष्टरोगाचे नवीन निदान झालेले रोगी; किंवा -कुष्ठरोगी उन्होंना हात व पाय झांवर साधारण संवेदना आहेत. (साधारण मोनोफिलामेट्स आणि MRC गुणसंख्या - वेतारुग्नी नाही) किंवा - कुष्टरोगी जयांना प्रतिक्रिया आहेत. कुष्टरोगी ज्यांता हात व पाय ह्यांगध्ये वेदना होतात. t y

Appendix 12. Patient's recruitment in Marathi

Appendix 13. Patient information sheet

Patient Information Sheet _ QST IN LEPROSY Bombay leprosy centre _ BLP Mumbai/India PATIENT INFORMATION SHEET Information about a Research Project Title: the development of new tools for the investigation of neuropathic pain in leprosy: a pilot study in Mumbai, India Background: Leprosy is a curable disease, but even so people with leprosy may develop pain, because leprosy may damage their nerves. Peripheral neuropathy may occur as a result of nerve damage, most often in the feet and hands, and changes the way that skin senses touch, temperature, pressure, and pain. One type of pain affecting leprosy people is called Neuropathic Pain. Objective and methods: Neuropathic pain is a common problem in leprosy. We would like to assess this pain among leprosy patients in India. This study is being undertaken by a researcher at London School of Hygiene and Tropical Medicine. The main objective of the project is to develop a simple tool for the assessment of neuropathic pain in leprosy for future use in larger multi-centre studies. This study may help to treat the problem probably. The Maharasha region of India was selected for this study because the record showed that there is increased number of leprosy patients. Bombay Leprosy Project (BLP) clinic is the leprosy referral centre for the Maharasha region. You have been invited to take part in this study because you may have complications of leprosy. The study will involve an interview and clinical examination assessing your nerve function and your pain symptoms. These methods can tell us about the type of pain that you may be experiencing. Interviews will take place at the BLP clinic and are expected to last approximately two hours and 45 min. Participations: Your participation is entirely voluntary, and should you agree to take part you may withdraw at any time without given a reason. If you do not wish to take part it will not in any way affect your current treatment. Should you agree to participate, we would like to collect some routine information from your hospital notes. We shall ask you to answer validated pain questionnaires. We shall do a very thoroughly examination of your skin and nerves using qualitative sensory test, and if needed will do urine examination. We shall do skin biopsy from your lesion using 3 mm punch skin biopsy. 1 Version 2.0; 08.02.2012

Patient Information Sheet _ QST IN LEPROSY

Bombay leprosy centre _ BLP Mumbai/India

There are no risks to you in participating in this study. The study will not be of any direct benefits to you, but will provide important results as it will assess the neuropathic pain problem among leprosy patients in India. Such information can help doctors in deciding how best to treat pain and it is useful for researchers to understand and manage this chronic pain condition seen in leprosy patients and give an important contribution to care after cure.

Confidentiality:

All information that is collected about you during the course of the research will be kept strictly confidential. Data from this aspect of the study will be kept in an anonymous form and will be available only to the research workers listed below. Initially the information will be stored on secure computers at London School of Hygiene and Tropical Medicine. We will then send this anonymised data to the university computers for analysis and also include it in computer databases which are used to store and analyse data about neuropathy and pain; these are the London Pain Consortium and the German Neuropathic Pain Network databases. Please note that we intend to disseminate our final results to other researcher by publishing it in international journals, but we will not identify any of the participants by name, and therefore, we will maintain your privacy, and anonymity and confidentiality of information will be preserved.

Further information:

The project is funded by the Hospital and Homes of St. Giles (HHSG), a registered charity in UK. It is being conducted by Dr. Omer Haroun of London School of Hygiene & Tropical Medicine, and is supervised by Professor Diana Lockwood (Chief consultant, LSHTM). And collaborators from the BLP and the Foundation for Medical Research (FMR) India, Professor Andrew Rice (Imperial Collage London), Dr David Bennett, Kings College London (KCL) and the German Network on Neuropathic Pain (DFNS); Professor Christoph Maier and Professor Rolf-Detlef Treede.

This study has been reviewed and approved by the Ethical Review Committee in India (IERC), number XXX and LSHTM Research Ethics Committee number XXX; contact person is Professor Andy Hall, email address: <u>ethics@lshtm.ac.uk</u>.

Contact for further information:

Should you have any questions that are not answered here or require any further information or explanation, please contact Dr. Omer Haroun at the address below

Dr. Omer Haroun Clinical Research Department (CRD) Faculty of Infectious and Tropical Diseases London School of Hygiene and Tropical Medicine University of London Room 358, Keppel Street London WC1E 7HT, UK <u>oharoum@hotmail.com</u>

Version 2.0; 08.02.2012

Datient Information Sheet OST IN LEDDOSY	Pembau Janzaru centre - DI D
	Bombay leprosy centre _ BLP Mumbai/India
Tel: + 442076127863; Fax: + 442076374314 Mobile: + 447536136026	
Thank you for taking time to read this information lea the study please read and sign the consent form.	flet. If you think you will take part in
	3
Version 2.0; 08.02.20	012

रुग्ण जानकारी पत्र अन्वेषण अध्ययन के बारे में जानकारी. परियोजनाका नामः कुष्टरोग में होनेवाले तंत्रिकविकृति दर्द को जाँचने के लिए नया साधन तैयार करनाः मुंबई, भारत में मार्गदर्शक जाँच पृष्ठभूमिः (background) कुष्टरोग एक इलाज होनेवाला रोग है, परन्तु जिन्हें कुष्टरोग हुआ है उन्हें कभी कभी दर्द हो सकता है क्योंकि ये रोग नसों को नुकसान पहुँचाता है. नसों के नुकसान के कारण परिधीय (peripheral) तंत्रिक विकृति हाथ और पैरों में निर्माण हो सकती है, और ये त्वचा कि स्पर्श, तापमान, दबाव और दर्द की समझ बदल सकता है, एक प्रकार का दर्द जो कुष्टरोगियों को होता है उसे तंत्रिकविकृति दर्द कहते है. उद्देश्य और ढंगः तंत्रिकविकृति दर्द कृष्टरोग में एक सामान्य समस्या है, हमें ये दर्द भारत के कृष्टरोगियों में जॉचना है. ये अध्ययन लन्दन स्कूल ऑफ हायजीन ॲन्ड ट्रॉपीकल मेडिसीन के एक अन्वेषक ने शुरू किया है. इस अध्ययन का मुख्य उदेश्य कृष्टरोग में होनेवाले तंत्रिक विकृति दर्द को जॉचने के लिए साधारण साधन तैयार करना है जिसका भविष्य में विशाल रूप से उपयोग हो सके. ये अध्ययन शायद इस समस्या के उपचार में मदद करेगा. इस अध्ययन के लिए भारत में महाराष्ट्र प्रदेश चुना गया क्योंकि दस्तावेज (record) दिखाते हैं कि कुष्टरोगों का प्रमाण बढ़ रहा हूं. मुंबई के लिए मुंबई लेपसी प्रोजेक्ट चिकित्सालय कुष्टरोग संप्रेषण केंद्र है. आपको इस अध्ययन में भाग लेने के लिए आमंत्रित किया गया है क्योंकि कुष्टरोग के कारण आपको उलझने हो सकती है. इस अध्ययन में आपकी मुलाकात और रोगविषयक जॉन की जाएगी जिससे आपके नस कृत्य (nerve function) और आपके दर्द के लक्षण, आकलन होंगे. आपको किस प्रकार का दर्द हो रहा है यह ये जॉच हमें बता सकेगी. मुलाकाते और परिक्षण फॉंडेशन फॉर मेडिकल रीसर्च चिकित्सालय में की जायेंगी और लगभग चार घंटे पैतालीस मिनट चलेंगे. सहयोगः आपका सहयोग पूर्ण तरीकेसे स्वैच्छिक है. आप सहमति देने के बाद में भी, कभीभी विना कारण दिए वापस जा सकते है. अगर आप सहभाग नहीं लेना चाहते तो भी आपके चालू इलाज पर कोई असर नहीं पड़ेगा. अगर आप सहआग लेने का निर्णय लेते है तो हम आपके अस्पताल की जानकारी से कुछ सामान्य जानकारी लेंगे. हम आपको दर्द के वैध प्रश्नावली भरने के लिए पूछेंगे. हम आपके त्वचा और नस की पूरी तरह से क्वान्टीटेटीव्ह सेंसरी टेस्ट और मूत्र परीक्षा करेंगे. हम आपकी त्वचा ऊलकों (Biopsy) की जाँच तीन मिलीमीटर के छेदक यंत्र से करेंगे. इस अध्ययन में सहयोग लेने के कोई खतरें नहीं है, ये अध्ययन आपके लिए कोई सीधे फायदेगंद नहीं है, परन्तु आप भारत में जो कुष्टरोग में होनेवाले तंत्रिकविकृति दर्द के मराले का हंल निकालने में मदद करेंगे. इस प्रकार की जानकारी, डाक्टराें को इस दर्द को सबसे अच्छे तरीके का उपचार करने में मदद कर , सकती है. अन्वेषकों के किये ये जानकारी महत्वपूर्ण हैं क्योंकि वह इस दर्द को समझ सकते है और ये स्थायी दर्द जो कुष्टरोगों में देखी जाती है इसके इलाज के चाद के देखभाल में एक महत्वपूर्ण योगदान कर राकते हैं. Continue...

Appendix 14. Patient information sheet – Hindi version

गोपनीयताः

इस अध्ययन के दरभियों आपसे पायी हुई सब जानकारी सखती से गुप्त रखी जाएगी. इस अध्ययन की जानकारी गुप्त रूप में रखी जायेगी और सिर्फ नीचे लिखे गए अन्वेषकों को उपलब्ध होगी. शुरू में ये जानकारी लल्दन स्कूल ओफ़ हायजिन अेन्ड ट्रोपिकल मेडिसीन में सुरक्षित कंप्यूटरों में रखी जायेगी. हम ये गुमलाम जानकारी महावियालय के कंप्यूटरों को और जो कंप्यूटर तंत्रिकविकृति दर्द की जानकारी को रखते है और विक्षेषण करते है उन्हें भेजेंगे: ये लन्दन पेन कंसोटियम और जर्मन न्युरोपथीक पेन नेटवर्क के कंप्यूटर होंगे. कृपया ये ध्यान में रखे कि हमारे अंतिम परिणाम हम दुसरे अन्वेषक को अंतरराष्ट्रीय पत्रिका में प्रकाशित करने के लिए क्षेजेंगे. परन्तु हम किसी भी सहआगी को लाम से नहीं पहचानेंगे और इस तरह हम आपकी गोपलीयता, और गुमनामी सुरक्षित रखेंगे.

अधिक जानकारी:

इस परियोजना का पैसा हॉस्पिटल ॲन्ड होम्स ओफ सेंट.गिल्स (HHSG) जो युनायटेड किंगडम की एक पजीकृत हुई सेवाग्रह है, ये अध्ययन लन्दन स्कूल ऑफ हयगीन हायजिन मेंन्ड ट्रॉपिकल अंडिसीन के डॉ. ओगर हारन कर रहे हैं और इसका निरीक्षण प्राध्यापिका डायना लोक्सूड (मुख्य सलाहकार, LSHTM) कर रही हैं. इस अध्ययन में बॉम्चे लेपसी प्रोजेक्ट और फौन्डेशन फौर मेडिकन रेक्स (FMP), आरल, प्राध्यापक अंन्डरू सईस (ईम्पीरिअल कॉलेज लन्दन), डॉ. डेविड बेनेट, किंग्स कॉलेज लन्दन (KCL) और जर्मन नेटवर्क ऑन न्युरोपथीक पेन (DFNS); प्राध्यापक क्रिस्टोफर मायर और प्राध्यापक सेन्स्र -देतलेफ ट्रीड सहयोगी है.

ये अध्ययन को एथीकल रिव्हू कमिटी इन इंडिया (IERC), अंक XXX और लश्तम रिसर्च कमीटी अंक XXX ते लिरीक्षण और अनुमोदित किया गया है और संपर्क साधने के लिए प्राध्यापक ॲन्डरू राईस, ईमेल पता हैं: <u>ethics@lshcn.ac.uk</u>.

अधिक जानकारी के लिए संपर्कः

अगर आपके किसी प्रश्नों का जवाब आपको नहीं मिला है या आप अधिक जानकारी चाहते हैं, तो कृपया नीचे दी गए पते पर डों. ओमर हारुन से संपर्क कीजिये.

Dr. Omer Haroun Clinical Research Department (CRD) Faculty of Infectious and Tropical Diseases London School of Hygiene and Tropical Medicine University of London Room 358, Keppel Street London WC1E 7HT, UK oharoun@hotmail.com Tel: + 442076127863; Fax: + 442076374314 Mobile: + ¥¥'643583€9€

या

मैत्रेयी निगवेकर - तंबर - ००९१ ९८२०९३६५३४

इस जानकारी पत्र को पढने के लिए वक्त निकालने के लिए शुंक्रिया. अगर आपको इस अध्ययन में माग लेना है तो कृपया सहगति पत्र पढ़िए और उसपर दस्तखत कीजिये.



Appendix 15. Patient information sheet – Marathi version

ह्या संशोधनामध्ये सहभागी होण्यात कोणत्त्याही प्रकारचा तुम्हाला धोका नाही. ह्यातून तुम्हांला कोणत्त्याही प्रकारचा प्रत्यक्ष फायदा नसला तरीही ह्या परिक्षनामुळे पुढच्या कुष्टरोगींना मदद हाउ शकते. ह्या संशोधनामुळे वेदनांवर सर्वात चांगल्या पढतोने कोणते उपचार केले जावेत ह्या संबंधातील निर्णय घेण्यास डॉक्टरांना मदत होऊ शकेल. तसेच संशोधकांस ह्या कुष्ठरोगांमधील दीर्घकालीन (जुनाटchronic) वेदना स्थिती समजून घेण्यास व त्यांना ताब्यात ठेवण्यावरील अभ्यास करण्यास मदत करु शकेल. बरे झाल्यानंतर घेण्यात थेणारी खबरदारी (care) विषयक अभ्यासात हा तुमचा फार महत्त्वचा हातभार (योगदीक्षा) ठरेल.

गोपतीयताः

संशोधन करीत असता मिळालेली तुमच्याविषयक संपूर्ण माहिती ही काटेकोरपणे गुप्त राखली जाईल.

अधिक माहितीः

ध्रा प्रकल्पाला युतायटेड किंगडममधील जाँदणीकृत धर्मादाय, हॉस्पिटल्स ॲण्ड होम्स ऑफ सेंट. गिल्स (HHSG) ह्यांचे आर्थिक पाठबळ लाभले आहे. लंडन स्कूल ऑफ हायजीन ॲण्ड ट्रॉपिकल मेडिसिन येथील डॉ. ओमर हरून हे हा प्रकल्प चालवणार आहेत व प्राध्यापक डायना लॉकवूड सांचे पर्यवेक्षण असेल. आणि BPL आणि फॉडेशन फॉर मेडिकल रिसर्च (FMR) मारत, प्राध्यापक उॉण्डू राईस (इम्पीरेअल कॉलेज लंडन). डॉ. डेव्हिड चेनेट, किंग्स कॉलेज लंडन (KCL), द जर्मन नेटवर्क ऑन न्यूरोपश्चिक पेल (DFNS); प्राध्यापक ख्रिस्तोफ मेयात आणि प्राध्यापक रॉल्फ-डेटलेफ ट्रीड सांचा सहभाग असेल.

ह्या अभ्यासास एथिकल रेव्हू कमिटी इन इंडिया, क्र. XXX आणि LSHTM रिसर्च एथिक्स कमिटी क्र. XXX, झांनी समालोचन (review) करून संमती दिली आहे. संपर्क करिण्यासाठी प्राध्यापक अंण्डी हॉल, इ-मेल पत्ता: <u>ethics@lshtm.ac.uk</u>.

अधिक माहितीसाठी संपर्क साधाःतुमच्या कोणत्याही प्रश्न्तांना येथे उत्तर दिले गेले नसेल किंवा चुम्हांला काही अधिक माहिती वा स्पष्टीकरण हवे असेल तर कृपया खातील पत्यावर डॉ. औमर हरून ह्यांच्याशी संपर्क साधा.

Dr. Omer Haroun Clinical Research Department (CRD) Faculty of Infectious and Tropical Diseases London School of Hygiene and Tropical Medicine University of London Room 358, Keppel Street London WCIE 7HT, UK <u>oharoun@hotmail.com</u> Tel. No. 9819079014

तुम्हीं वेळ काढून हे माहितीपत्रक वाचले झावदल धन्थवाद. ह्या संशोधनात तुम्हीं जर सहभागी होऊ इच्छित असाल तर कृपया संमती पत्रक वाचा व त्यावर सही करा.

A DELEVISION OF A DELEVISIONO OF A DELEV	Impe Lond	rial College Ion
CONSENT FORM I	FOR HEALTHY VO	DLUNTEER
ICERC Number: xxx		
Healthy Volunteer Quantitative Sens	ory Testing (QST)	
Chief investigators: Prof Andrew Study organizer: Dr O Haroun	Rice Professor Diana Lockwood	L
Please initial the boxes next to state	ments you are in agreement with.	
		Please initial Box
 I confirm that I have rea information sheet dated study, and I have had the o have been fully answered. 	d and understood the participant versionfor the above opportunity to ask questions which	
 I understand that my particip withdraw at anytime, without being affected. 	pation is voluntary and I am free to ut my medical care or legal rights	
	is have been discussed with me.	
The compensation regulation		
 The compensation regulation I agree to take part in this res 	earch study	
 The compensation regulation I agree to take part in this res 	earch study Signature	Date
 The compensation regulation I agree to take part in this res 	earch study Signature Signature	Date Date
3. The compensation regulation 4. I agree to take part in this res Name of participant Name of Person taking consent (If not the Principle investigator) Principle Investigator	earch study Signature Signature Signature	Date Date Date Date

Appendix 16. Healthy volunteer consent form

Prof. Dr. med. R.-D. Treede Lehrstuhl für Neurophysiologie Zentrum für Biomedizin und Medizintechnik Mannheim Medizinische Fakultät Mannheim der Universität Heidelberg Deutscher Ludolf-Krehl-Str 13-17 Forschungsverbund 68167 Mannheim Neuropathischer Einverständniserklärung Schmerz Ich, ____, wurde vollständig über Art, Umfang und Bedeutung der klinischen Studie: Untersuchung der Funktion und Eigenschaften des somatosensorischen Nervensystems mit Hilfe der quantitativen sensorischen Testung (QST) im Rahmen des Deutschen Forschungsverbundes Neuropathischer Schmerz (DFNS) aufgeklärt. Ich hatte die Möglichkeit, den aufklärenden Arzt ausführlich zu befragen. Dabei wurden u.a. Studienziel und Studienlänge, studienbedingte Erfordernisse und mögliche Nebenwirkungen der Studienbehandlung besprochen. Die Probandeninformation sowie ein Exemplar der Einverständniserklärung habe ich gelesen und verstanden sowie eine Kopie von beidem erhalten. In diesem Zusammenhang bestehende Fragen wurden besprochen und beantwortet. Ich hatte ausreichend Zeit, mich für oder gegen eine Teilnahme an dieser Studie zu entscheiden. Über die vorliegende Probandenversicherung wurde ich informiert, ein Auszug aus den Allgemeinen Versicherungsbedingungen wurde mir ausgehändigt. Ich bestätige durch meine Unterschrift, dass ich mich mit der vorgenannten Prüfung und ihrer Durchführung einschließlich der dafür notwendigen ärztlichen Untersuchungen einverstanden erkläre. Ich bin mit der im Rahmen der Studie erfolgenden Aufzeichnung von personenbezogenen Untersuchungsdaten, auch in elektronischer Form, und ihrer Weitergabe sowie Verarbeitung in anonymisierter Form im Rahmen des Deutschen Forschungsverbundes Neuropathischer Schmerz (DFNS) entsprechend der Probandeninformation einverstanden. Alle im Rahmen dieser Studie erhobenen Daten unterliegen der ärztlichen Schweigepflicht und werden vertraulich gemäß dem Datenschutz behandelt. Das zugehörige Kapitel "Datenverarbeitung und Datenschutz" (Seite 3 der Probandeninformation zu dieser Studie) habe ich gelesen und stimme dem beschriebenen Vorgehen zu. Ich weiß, dass diese Studie in erster Linie der medizinischen Wissenserweiterung dient und gegebenenfalls keinen persönlichen Vorteil für mich bringen kann. Ebenso weiß ich, dass meine Teilnahme freiwillig ist und ich jederzeit meine Einwilligung ohne Angabe von Gründen widerrufen kann, ohne dass in irgendeiner Weise ein Nachteil für mich entsteht. Ich kann die Untersuchung jederzeit abbrechen. Auch der Arzt kann aufgrund seiner ärztlichen Erfahrung die Prüfung jederzeit beenden. Ort, Datum Unterschrift des Probanden Ort, Datum Unterschrift des Arztes Untersuchung der Funktion und Eigenschaften des somstosensorischen Nervensystems mit Hilfe der quantitativen Testung (QST) im Rahmen des Deutschen Fonchungsverbunden Neuropathischer Schmerz (DFNS) Lehrstuhl für Neurophysiologie, Medizinische Fakultät Maunheim der Ruprecht-Karls-Universität Heidelberg fü orischen Nervensystems mit Hilfe der quantitativen sens 1 g 02.10.2007;

Appendix 17. Healthy volunteer consent form – German version

	निरोगी स्वयंसेवकों के लिए सहम	ते पत्रः			
	नैतिक अनुमोदन अंक:				
	- परियोजना का नामः कुष्टरोग में करनाः मुंबई, भारत में मार्गदर्शक	होनेवाले तंत्रिक विकृति दर्द जाँच	को जॉचने के लिए तथा	साधन तैयार	
	निरोगी स्वयंसेवक क्वान्टीटेटीव्ह र प्रमुख अन्त्वेषकः प्रोफ. डायला लौ	सेंसरी टेस्टींग और चमड़ी उ कवूड, प्रोफ. आन्द्र् राईस अं	लकों की जाँच (blopsy) रेर डॉ.वनजना शेट्टी		
	अल्वेषक का नामः डॉ. ओमर हर	চল			
	कृपया इस भाग में आपके आवाध	तर लिखकर नीचे दस्तखत	कीजिये.		
C	 मैं ये रूहमति देता/देती हूँ वि और मैं ये जानकारी पत्र दिनांग 	फ मैंने पढ़ा है या मुझे अर्थ क यूत ममझती हूँ और मुट्ठो इस अ नाएगी. मुझे ये जानकारी र ना दिलाया गया और मेरे प्र है.	बताया गया है ान्त ध्ययल की एक मन्द्राने के लिए श्रों के उत्तर		
	२) मैं ये समझता/समझती हूँ वि कारण दिए जब चाहे वापसी कानूनी अधिकारों पर कोई अ	5 मेरा तहभाग स्वैच्छिक है ले सकता/सकती हूँ और बे रसर नहीं होगा.	और मैं बिना रेरे चिकित्सक या		
	३) मुझे मुआवजों के नियम तम	नझाए गए है.			
6.5	४) मैं अन्देषण समूह को चमडी	ऊतकों के दो जांच करने वं	गे सहमति देता∕देती हूँ.		
	५) मैं मेरा मांसतंतु फौब्डेशवा फॉन यूनाइटेड विंगाडम में विश्लेषण	र मेडिकल रिसर्च और किंग और बचाए रखने के लिए	रा कोंलेज लन्डन, अनुमति देता∕देती हूँ.		
	६) में इस अन्वेषण अध्ययन में १	भाग लेने के लिए अनुमति	देता/देती हूँ.		
	सहभागी का नाम	दस्तखत	दिनाक		
	सहमति लेनेवाले का नाम	दस्तखत	दिनांक		
	प्रमुख अन्चेषक	दस्तखत	दिनांक		

Appendix 18. Healthy volunteer consent form – Hindi version

Appendix 19. Healthy volunteer consent form – Marathi version

निरोगी स्वयंसेवकांसाठी संमती प्रप	गत्र	
नीतीपूर्ण अनुमती अंकः		
प्रकल्प शीर्षकः कुष्ठरोगात होणार् नव्या साधनांतील प्रगती : मुंबई,	या वेदनांची चेतारुग्णी भारत येथील मार्गदर्शी	(neuropathic) करण्याऱ्या ि अभ्यास
निरोगी स्वयंसेवकांसाठी क्वान्टिटेर्ी	टेव्ह सेन्सरी टेस्टींग ((QST)आणि त्वचा जीवोतक परीक्ष
प्र मुख अन्वेषक: प्रा. डायना लॉकव् संशोधकाचे नाव: डॉ. ओमर हरून	ग्रूड, प्रा. अंण्हू राइस आपि	गे डॉ. वनजा शेट्टी
कृपया तुमची संमती असलेल्या वि	नेवेदनापुढील चौरसांमध	ये तुमची आद्याक्षरे भरा.
१) मी ही खात्री देते/देतो की मी पत्र दिनांकवाच गेले आहे, आणि मला, रुपांतर त्याची प्रतिलिपि मला देण्यात विचार करण्यासाठी, प्रश्न विच आणि माझ्या प्रश्नांना समाध २) मला समजावण्यात आले आहे मी कधीही कारण न देता मा वैद्यकीय आणि कायदेशीर हक 3) मोबदल्याचे नियंत्रण माझ्याबरोग	सहभागी होण्यावाबतन ले आहे, किंवा मला स कळल त आली आहे. मला ह्या वारण्यासाठी संधी दिली गलकारक उत्तरे दिली ग हे की माझा सहभाग ऐ घार घेऊ शकते/शकतो कांवर काही परिणाम ह बर मला समजावले गेले	वे हे माहिती मजावले ले आहे, माहितीवर ो गेली आहे ोली आहेत. चिछक आहे आणि ो व त्यामुळे माझ्या होणार नाहीत. ो आहेत.
४) मी अध्ययन संघला दोन चामर्ड	ोचे जीवोतक परीक्षा करा	ायला अनुमति देते.
५) मी माझं ऊति फौन्देशन ऑफ़ लन्दन,युनायटेड कींगडम, ह्याना ६) मी या संशोधनात भाग घ्यायत	मेडिकल रीसर्च आणि 1 विश्लेषण आणि साठवन्थ त्रा सहमती देते.	किंग्स कॉलेज पासाठी परवानगी देते.
सहभागी व्यक्तीचे नाव	सही	दिनांक
संमती घेणाऱ्या व्यकीचे नाव	सही	दिनांक
प्रमुख अन्चेषक	सही	दिनांक
ह्या संशोधनामध्ये सहभागी होण्य	पासाठी मी माझी अनुग	मती देत आहे.

Appendix 20. Patient consent form

	, i	LONSEINT FOI	KIVI	
Title o pilot s	of Project: The development of tudy in Mumbai, India	new tools for the inves	tigation of neuropathic pa	in in leprosy:
Name	of Researcher: Dr Omer Haro	n		
Chief	investigators: Prof Diana Lockv	vood, Prof A. Rice and D)r Vanaja Shetty	
Please	e complete this section by initi	alling the boxes and th	en signing at the bottom	
Patien	t consent to study protocol			
1.	I confirm that I have read, or	had explained to me, an	nd understand the	
	information sheet dated	versionconce	erning the above study	
	the information ack question	o keep. I have had the o	opportunity to consider	
	the information, ask question	is and nave had these a	nswereu satistactorily.	
2.	I understand that my particip	ation is voluntary and t	hat I am free to	
	withdraw at anytime, withou	t giving any reason, wit	hout my medical care or	
	legal rights being affected			
3.	I understand that relevant se	ctions of any of my mee	dical notes and data	
	collected during the study ma	ay be looked at by respo	onsible individuals from	
	LSHTM or pain research team	 I give permission for t 	hese individuals to	
	access my records.			
4.	l agree with the study team t	hat skin biopsies being	taken.	
5.	I agree for my tissue to be an	alysed and stored at Fo	undation for Medical	
	Research (FMR) India and Kin	g's College London, Lor	idon, UK.	
6.	I agree to take part in this res	earch study.		
	· · · · · · · · · · · · · · · · · · ·	,		
Name	e of participant	Signature	Date	
Name (If not	e of Person taking consent t the Principle investigator)	Signature	Date	
Princi	ple Investigator	Signature	. Date	
	1 copy for participant; 1 copy	for Principle Investigato	or; 1 copy to be kept with	BLP notes

-				die	
	सहमति पत्र				
	परियोजना का नाम: कुष्टरोग मे करना: मुंबई, भारत में मार्गदर्शव	र्ग होनेवाले तंत्रिक विकृति फ जाँच	दर्द को जाँचने के लिए नया साधन तं	नैयार	
	अन्वेषक का नामः डॉ. ओमर ह प्रमुख अन्वेषक: प्रोफ. डायना ल	हरून गौकवूड, प्रोफ. आन्द्रू राईस) और डॉ. वनजना शेट्टी		
	कृपया इस भाग में आपके आद्या	क्षर लिखकर नीचे दस्तख	त कीजिये.		
	रुग्ण सहमति पत्र				
	 मैं ये सहमति देता/देती हूँ वि मैं ये जानकारी पत्र दिनांक समझती हूँ और मुझे इस 3 जानकारी समझने के लिए प्रश्नों के उत्तर संतोषजनक ढं 	के मैंने पढा है या मुझे अ यूतात १ध्ययन की एक प्रतिलिपि और प्रश्न पूछने के लिए उ ग से दिए गए है.	र्थ बताया गया है और न्त समझता/ रखने के लिए दी जाएगी. मुझे ये मौका दिलाया गया और मेरे		
	२) मैं ये समझता / समझती हूँ नि भी वक्त बिना कारण दिए वाप या कानूनी अधिकारों पर कोइ	के मेरा सहभाग स्वैच्छिक गसी ले सकता/सकती हूँ ई असर नहीं होगा.	है और मै किसी और मेरे चिकित्सक		
	३) मैं ये समझता⁄समझती हूँ र्व हुए चिकित्सक व्याख्या के उ LSHTM के व्यक्ति या अन्व अभिगम करने की अनुमति व	गे इस अध्ययन के दरमिर चित भाग और दी हुयी स पेषण समुह देखेंगे. मैं इन देता∕देती हूँ.	पान मेरे इकट्ठे किये ामग्री सिर्फ जिम्मेदार व्यक्तियोंको मेरे अभिलेख		
	४) मैं अन्वेषण समूह को चमड़ी	की जीवित ऊतकों की ज	ाँच करने की सहमति देता/देती हूँ.		
	५) मैं मेरा मांसतंतु फौन्डेशन फो यूनाइटेड किंगडम में विश्लेषप	ॉर मेडिकल रिसर्च और वि ग और बचाए रखने के ति	फ्रेंग्स कॉलेज लन्दन, पेए अनुमति देता/देती हूँ.		
	६) मैं इस अन्वेषण अध्ययन में	भाग लेने के लिए अनुमां	ते देता/देती हूँ.		
	सहभागी का नाम	दस्तखत	दिनांक		
	सहमति लेनेवाले का नाम	दस्तखत	दिनांक		
	प्रमुख अन्वेषक	दस्तखत	दिनांक		
				Continu	ıe

Appendix 21. Patient consent form – Hindi version

मैं ये सहमति देता⁄देती हूँ की	मैंने मेरी तस्वीर लेने की 🛛	अनुमति दी हुई है.	
में समझता/समझती हूँ की इ में ये सामग्री समुचित व्यवसार पुस्तक और दुनियाभर की अ इलेक्ट्रोनिक प्रकाशितों में वित इस कारण मैं ये समझता/स इसमें से सब या कुछ सामग्री साथ इस्तेमाल किये जाएंगे.	स सामग्री को शिक्षासंबंधी ग्री कर्मचारी और शिक्षासंबं न्य रूप या माध्यम में, हर रित किये जाने की अनुमति मझती हूँ की ये सामग्री सा 1 अन्य तस्वीर,चित्र, वीडिय	मूल्य है. धी प्रकाशित ग्रंथ, दैनिक पत्र, अध्ययन - ⁻ तरह के देता/देती हूँ. मान्य जनता को दिखाई जा सकती है. ो टेप, ध्वनीलेखन या अन्य रूप के चित्र के	
मेरी पहचान गुप्त रखने का प	प्रयास किया जाएगा परंतु	इसकी कोई गारन्टी नहीं है.	
सहभागी का नाम	दस्तखत	दिनांक	C
			\mathbf{C}
		1.4	

1 <u>.</u>		1
	संमती प्रपत्र	
	प्रकल्पाचे जायः कुष्ठरोगात होणाऱ्या चेतारुग्णी (neuropathic) वेदनांचा लपास करण्यार्थ। जन्मा साधनातील प्रगती : मुंबई, भारत येथील मार्गदर्शी अभ्यास	
	संशोधकाचे नावः डॉ. ओमर हरून प्रमुख अन्वेषक: प्राध्यापक डायाना लॉकवूड, प्राध्यापक ॲण्डू- राईस आणि डॉ. वनजा-शेट्टी	
	कृपया चौरस अरून प्रणत्र पूर्ण भरा आणि सर्वात शेवटी सही करा.	
	रुग्गाच्या सूळ नौदी अभ्यास करण्यासाठी संमती:	
(\cdot)	?) नी ही खात्री देते/देतो की शी था तपाराणीवद्दल वाचले आहे, विवा जला समजावले गेले आहे, आणि मला हे माहिती पत्र, दिलांक	
	२) मना रामजावण्यात आले आहे की माइत सहभाग ऐच्छिक आहे आणि मी कधीही कारण न देता माचार घेऊ शकते/शकतों द त्यामुळे माझ्या वैधकीय आणि कायदेशीर हक्कांवर काही परिणाम होणार नाहीत.	
Ó	3) जन्ता समजावण्यात आले आहे की माझी वैद्यकीय लॉदणी आणि जमा केलेली माहिती LSHTM च्या जवाबदार व्यक्ति किंवा वेदला संशोधन संघ/पथक वचतील. मी ह्या व्यक्तींना ताझी माहिती वापरण्यासाठी परवालगी देते/देतो.	
	४) मा त्वचेच्या जीवोतक परीक्षेसाठी अनुमति देते/देतो.	
	५) फोन्डेशन ओफ मेडिकल रीसर्च ॲन्ड किंग्स कॉलेज लंडन, युनायटेड किंग्डन, ह्यांना माझ्या ऊति. विश्लेषण आणि साठगण्यासळी मी परवानगी देते/देतो. ६) मी या संशोधनात भाग गेण्यास सहमती देते/देतो.	
1		
		Continue

Appendix 22. Patient consent form - Marathi version

सरभागी व्यक्तीचे नाव	सही	दिनांक	
संमती धेणाऱ्या व्यक्तीचे नाव	सही	दिलांक	
प्रमुख अन्वेषक	सही	दिनांक	
माझे फोटो काढण्यासाठी मी माझी योग्य व्यावसायिक कर्मचारी वर्य लियतकालिके, पाठ्यपुस्तके, तसेच ह्यांमध्ये हे साहित्य प्रकाशित केले माझी अनुमती देत आहे मला हे माहित आहे की त्यामुळे स इतर छायाचिवे, रेखाचित्रे, विहडीयोरे ह्यातील काही भाग वा हे संपूर्ण वा माझे लाव गुप्त राखण्यासाठी एय ताही.	ो परवानगी दिली आहे ह गिण्ट्ये साहित्य दाखवत इतर माध्यमांमध्ये, व ने जाईल व जगभर कुरे वर्वसाधारण लोकांच्या न वर्वसाधारण लोकांच्या न र्यस इमेजेस, ध्वतिमुद्रण परले जाईल. ात्न केले जातील परंतु,	ग़ाला मी येथे अनुमती देत आहे. ने जाईल आणि शैक्षणिक उकाशने, कम्प्यूटस्थी संबंधित इतर प्रकाशने, इही ह्याचे वितरण होईल ह्यासाठी गी जरेत हे साहित्य येऊ शकेल. ; किंवा चित्राचे इतर प्रकार झांमध्ये पूर्ण गोपनीयतेची खात्री देता येणार	0
सहभागी व्यक्तीचे नाव	सही	दिनांक	
			\odot
		1 4	

Appendix C

Patient Screening Forms and Clinical Assessment Forms

Appendix 23. Patients record form (PRF)

LeP	aln:		
Patient Re	cord Form		
Study: (QST_IN_L	EPROSY	
Bombay Leprosy Project Mumbai, India	Clinic		
	Date: Start time:: End time::		
Investigators: Dr. Omer Haroun Dr. V. V Pai Dr. Ashish Khodke Dr. Vanja Shetty Prof. Andrew Rice Prof. Diana Lockwood			

RM 1:	RECRUITMENT SHEET
dvice	on the completion of the PRF
C	plation of the BDC.
com	
1.	Complete PKF in black ink (not pencil)
2.	Do not uso ditto marke
۵. ۵	If the answer is "zero" do not leave the field blank
5	If the answer to a question is unknown, write "NK" or Not Known or "8"
6.	If a requested test has not been done, write "ND" or Not Done
7.	If a question is not applicable, write "NA" or Not Applicable
8.	All dates in this PRF take the format dd/mm/yyyy. Note that 'dd', 'mm' and 'yyyy' shoul
	be filled in with numbers
9.	If a date is partially known (e.g. year only) enter that part which is known and NK for th rest of the information
10	. If a mistake is made please correct in the following way:
	 Cross through with a single straight line
	 Write the correct value clearly above or to the side
	 Initial and date the correction
	 If appropriate, add an explanation

Patient work flow

Summary:

- 1. Patient identified at Bombay Leprosy Clinic
- 2. Patient informed about study, and recruited with consent
- 3. Full history and examination by study investigator
- 4. Patients sent for laboratory investigations
- 5. Patients screened for neuropathy using MFs and MRC
- 6. Patient reviewed by study investigator
- 7. Patient given review date and time

QST_IN_LEPROSY

Patient Record Form:

PI: ___

Page 2

EQuiSTA code number: |__|__|

```
Patient Initials: |__|__|
```

Schedule of events:

Protocol activities and Forms to be completed	Screen days	;		Test day
Patient information and Consent	х			
Screening form	х			
Clinical assessment	х			
Urine analysis		х		
Complete blood count		х		
HbA1c (Blood glucose)		х		
Blood sample for VitB12 level		х		
Blood sample for serology (Syphilis and HIV test)		х		
Neuropathy assessment	х			
Leprosy assessment	х			
Assessment of leprosy reactions	х			
Pain assessment (pain intensity)			х	
Clinical neurological examination		х		
DN4 Questionnaire			х	
Pain detect questionnaire			х	
GHQ-12 Questionnaire				Х
BPI Questionnaire				Х
Quantitative sensory testing (QST)				Х
3 mm skin punch biopsy				Х

Patient screening and registration

	last name	First name
Contact details/Mobile:	Last name	Thischame
Contact details /Landline:		
Province/Region:		
District:		
Location:		
Village:		
Name of the Head of Household:		
	Last name	First name
Please note	e: This section must remain at s	ite
Please note	e: This section must remain at s	ite

	1umber:		Patient Initia	ls:
	S (Screening	CREENING: DAY 0 may last more than o	ne day)	
Consent				
Did the Date of	patient consent to particip Consent (dd/mm/yyyy)	oate in the study? Yes [/	; (1) No (0) [] /]	
Did the Date of	patient consent for sampl Consent (dd/mm/yyyy)	e collection and lab ir [/	nvestigation? Yes (1) No /]	o (0) [_]
Did the Date of	patient consent for storag Consent (dd/mm/yyyy)	e of sample for futur	e use? Yes (1) No (0) [/]	
Demography Date of	Birth (dd/mm/yyyy)	[_ _	. / _ _ / _ _ _	_ _]
OK Age of	the patient (in years): (est	imate)	[]]	1
Gender Data ca	: Male (1) Female (2	:)		
Results	: Positive (1) N	egative (0)		_!!!
History of Sym Please indicate Unknown (8). I duration is less	ptoms on the day of interv if the patient reported a p f yes, complete the duration than 1 month, complete [/	riew: resence of the follow on as the number of r <u>)[0]</u> in "months" colu	ing symptoms as Yes (1 nonths and/or days: wh mn.) No (0) here the
	Description of problem	Yes (1) No (0) unknown (8)	Months	
	Pain			
	Skin changes Reactions			
	No problem			
Concomitant d	isease			
Concomitant d Does the patier	isease 1t have any clinically signifi	cant concomitant dis	eases at baseline?	
Concomitant d Does the patier	isease It have any clinically signifi Yes (1) No pecify	cant concomitant dis o (0) Unknown (8)	eases at baseline?	
Concomitant d Does the patier If yes S	isease It have any clinically signifi Yes (1) No pecify	cant concomitant dis o (0) Unknown (8)	eases at baseline?	
Concomitant d Does the patier If yes S For every ment	isease nt have any clinically signifi Yes (1) No pecify ioned condition requiring t	cant concomitant dis o (0) Unknown (8) reatment, please wri	eases at baseline? - - te the medication.	
Concomitant d Does the patier If yes S For every ment Does the patier	isease It have any clinically signifi Yes (1) No pecify ioned condition requiring t It have alcohol?	cant concomitant dis o (0) Unknown (8) 	eases at baseline? - - te the medication.	
Concomitant d Does the patier If yes S For every ment Does the patier If yes; v	isease nt have any clinically signifi Yes (1) No pecify ioned condition requiring t It have alcohol? what's the type (Beer (1), V	cant concomitant dis (0) Unknown (8) reatment, please wri Yes (1) No (0) Vine (2), Whisky (3), c	eases at baseline? - - te the medication.	
Concomitant d Does the patier If yes S For every ment Does the patier If yes; \ Duratic Freque	isease It have any clinically signifi Yes (1) No pecify ioned condition requiring t It have alcohol? vhat's the type (Beer (1), V n of Alcohol consumptions ncy of Alcohol consumptions	cant concomitant dis (0) Unknown (8) reatment, please wri Yes (1) No (0) Vine (2), Whisky (3), c ::Years n: (Daily (1), weekly (eases at baseline?	
Concomitant d Does the patier If yes S For every ment Does the patier If yes; v Duratio Freque Estimat	isease It have any clinically signifi Yes (1) No pecify ioned condition requiring t It have alcohol? what's the type (Beer (1), V In of Alcohol consumptions ncy of Alcohol consumption Red Alcohol consumption in (1) Period	cant concomitant dis (0) Unknown (8) reatment, please wri Yes (1) No (0) Vine (2), Whisky (3), o ::Years n: (Daily (1), weekly (glasses per weeks:	eases at baseline? - - te the medication. other) 2), occasionally (3)) glass/wk	
Concomitant d Does the patier If yes S For every ment Does the patier If yes; v Duratio Freque Estimat	isease at have any clinically signifi Yes (1) No pecify ioned condition requiring t at have alcohol? what's the type (Beer (1), V who of Alcohol consumptions ncy of Alcohol consumption in a Alcohol consumption in (1)	cant concomitant dis o (0) Unknown (8) rreatment, please wri Yes (1) No (0) Vine (2), Whisky (3), o ::Years n: (Daily (1), weekly (glasses per weeks:	eases at baseline? te the medication. other) 2), occasionally (3)) glass/wk	
Concomitant d Does the patier If yes S For every ment Does the patier If yes; v Duratic Freque Estimat	isease It have any clinically signifi Yes (1) No pecify ioned condition requiring t It have alcohol? what's the type (Beer (1), V in of Alcohol consumptions ncy of Alcohol consumption in p	cant concomitant dis (0) Unknown (8) rreatment, please wri Yes (1) No (0) Vine (2), Whisky (3), c ::Years n: (Daily (1), weekly (glasses per weeks:	eases at baseline? - - te the medication. other) 2), occasionally (3)) glass/wk	
Concomitant d Does the patier If yes S For every ment Does the patier If yes; v Duratio Freque Estimat	isease It have any clinically signifi Yes (1) No pecify ioned condition requiring t It have alcohol? vhat's the type (Beer (1), V In of Alcohol consumptions ncy of Alcohol consumption in p	cant concomitant dis o (0) Unknown (8) rreatment, please wri Yes (1) No (0) Vine (2), Whisky (3), o ::Years n: (Daily (1), weekly (glasses per weeks:	eases at baseline?	
Concomitant d Does the patier If yes S For every ment Does the patier If yes; v Duratic Freque Estimat	isease It have any clinically signifi Yes (1) No pecify ioned condition requiring t It have alcohol? what's the type (Beer (1), V on of Alcohol consumptions ncy of Alcohol consumption in p	cant concomitant dis (0) Unknown (8) reatment, please wri Yes (1) No (0) Vine (2), Whisky (3), c ::Years n: (Daily (1), weekly (glasses per weeks:	eases at baseline? - - te the medication. other) 2), occasionally (3)) glass/wk	
Concomitant d Does the patier If yes S For every ment Does the patier If yes; v Duratio Freque Estimat	isease It have any clinically signifi Yes (1) No pecify ioned condition requiring t It have alcohol? what's the type (Beer (1), V on of Alcohol consumptions ncy of Alcohol consumption is Alcohol consumption in (1) SY Patient Record	cant concomitant dis o (0) Unknown (8) reatment, please wri Yes (1) No (0) Vine (2), Whisky (3), o s:Years n: (Daily (1), weekly (glasses per weeks:	eases at baseline? te the medication. other) 2), occasionally (3)) glass/wk	

5	CREENING: DAY 0			
Clinical assessment Date measurements taken (dd/m	nmm/yyyy) [_ _ / _ _]/ _ _ _ _]			
Weight:	[]•[] Kg			
Height	[_ _ _] cm			
Body Mass Index - BMI	[]_] kg/m ²			
Haematologic Examination				
Date sample taken (dd/mmm/yyyy) [_	_ _ / _ _ _]/ _ _ _]			
Haemoglobin [_•_] g/dL				
Clinical Chemistry Examination				
Date sample taken (dd/mmm/yyyy) [_	_ _ / _ _ _]/ _ _ _]_]			
HbAc1 [•_ _] mmo B12 level [• _] r	ol/L ng/L			
Serological test for syphilis and HIV Date sample taken (dd/mm/yyyy) (Results will be reported separatel	[_ _ / _ _ / _ _]_] iy)			
Urinalysis (Dipstick Test) Date sample taken (dd/mm/yyyy) Please record one of the following Negative (0) + (1) ++ (2) ++ Glucose Urine for pregnancy test: Negative (0) Positive (1 If positive: Duration in month	[] / /] g numeric codes for the result on the +, ++, +++ scale. ++ (3) [] 1) [] hs []			
Last menstrual per Last child birth:	eriod: (dd/mm/yyyy) [_ _ / _ _ / _ _ _ _]_] (dd/mm/yyyy) [/ /]_]			
OCT IN LEBRORY Detices Beer	rd Form: PI: Page 9			
EQuiSTA code number	:1	1	1.1	
---------------------	----	---	-----	--
Edding Lange Lange	•			

Patient Initials: |__|__|

[__]

Screening for neuropathy:

Baseline peripheral neuropathy assessment

	Right		Movement		Left	
Dura#	Imp Y/N	MRC	Wovement	MRC	Imp Y/N	Dura#
			Little finger abduction (ADM)			
			Index finger abduction (IDI)			
			Thumb abduction (ABP)			
			Wrist extension			
			Foot dorsiflexion (DF)			
			Extension big toe (EHL)			
			Flexion big toe (EHL)			
			Toe fanning (TF)			
			Closes eyes (strong and gentle closure)			

Does the patient have motor neuropathy at baseline? Yes (1) No (0)

wononiam	ent asse	Right ci	s (Scoring and defining i	mpairment by monori	lament)		
		Right S	ae	102 12	Leit side		
Site	MED	ULN	人口	in	Site	MED	ULN
1.			(3/3)	10/01	1		
2.			11-20	11153	2		
3.			Utorna	Mort 9	3		
Total			2.81/1.9	B SENT	Total		
Imp (Y/N)			000	000	Imp (Y/N)		
Site	RCN	Ubr	<u>6</u> 80.	.680	Site	RCN	Ubr
1.			$\mathbb{P}(\mathcal{A}_{s})$	$\left \left \left \left \right \right \right \right \right \left \left $	1.		
2.			42 g1 ' 1	(¹) (0,43	2.		
3.			Q #20	d*2,0	3.		
Total			R.O.	- °/11	Total		
Imp (Y/N)			R	, r	Imp (Y/N)		
Site	Post. 1	ibial	- AP	OG	Site	Post. 1	Tibial
1.			and a	E and	1.		
2.			10-1	6-01	2.		
3.			1. 61	151	3.		
4			(°)	1 1	4		
Total					Total		
Imp (Y/N)					Imp (Y/N)		
Site	SURAL	.N			Site	SURAL	. N
1.			AVA	1 VI	1.		
2.			all 3	1011.20	2.		
3.			600	Cert In	3.		
Total			2 45 B		Total		
Imp (Y/N)				L	Imp (Y/N)		
Does the p	atient h atient h	ave sens ave neur	ory neuropathy at base	line? Yes (1) Yes (1)	No (0)	1	[
QST_IN_LE	PROSY	uve neul	Patient Record Form	PI:			Page

Does the patient have pain? If yes, please indicate: 1. Duration of pain symp 2. The pain intensity sco Please mark th A zero (0) mea	Yes (1) No (0) otoms in months? re using 11-point Likert scale he scale below to show how intense y ans no pain, and ten (10) means extrem	[] [] our pain is. me pain.
PAIN SCOR	E 0-10 NUMERICAL	RATING
0-10	Numerical Rating Scale	
0 1 2 3 Pain	4 5 6 7 Moderate Pain	8 9 10 Worst Possible Pain
3. The DN4 score		
Please answer yes or no for each item	of the following four questions.	
Question 1: Door the pain have one or	more of the following characteristics	
 Burning 	Yes (1) No (0)	r []
Painful cold	Yes (1) No (0)	
Electric shocks	Yes (1) No (0)	
Questions 2:		
Is the pain associated with	one or more of the following sympto	ms in the same area?
Tingling	Yes (1) No (0)	
Pins and needles Numbness	Yes (1) No (0) Yes (1) No (0)	
Itching	Yes (1) No (0)	
Question 3:		L
Is the pain located in the a	rea where the physical examination n	nay reveals one or more
of the following character	istics?	
 Hypoethesia to to 	uch Yes (1) No (0)	
 Hypoetnesia to pil Question 4: 	nprick tes (1) NO (U)	
In the painful area, can the	e pain be caused or increased by:	
 Brushing 	Yes (1) No (0)	
YES	= 1 point NO = 0 points	
	Patient's score/10	
How did you find this questionnaire?		r 1
(Very easy (1) / easy (2) / fair (3) / diff	icult (4) / very difficult (5))	
	1	

	SCREENING: DAY 0		
Please co meet all	mplete the boxes with Yes (1) No (0) Not applicable (3) as appropriate. I of the inclusion criteria and none of the exclusion criteria	Patients sho	uld
Inclusion	1		
1. W 2. A	/ritten informed consent to participate ged between 16 and 65 years (inclusive) who are able to comply with the	e protocol	
Exclusio	1		
3. K	nown to be Diabetes Mellitus		
4. H	ave positive serological test for syphilis		
5. Si c u	arum B12 level outside the normal range for age and gender	0 ar mara	
0. H	ho drink excessive amounts of alcohol regularly	to or more	
	5 ,		
lf ai	nswer YES to any of questions 3-7, the patient <u>must NOT</u> be entered into	o the QST st	udy
Investiga	tor signature:		
	Date (dd/mm/yyyy) [/ /]		
la alva			
inclus	ion as:		
	Indicate inclusion number according to the table below:		
Number	Inclusion as		
1.	Leprosy patients with established pain and clinical neuropathy		
2.	Leprosy patients with established pain and no clinical evidence of neur	ropathy	
3.	Leprosy patients with pain free neuropathy		
4.	Leprosy patients with no pain and no clinical evidence of neuropathy		
Chack	list for painful nouronathy:		
The follo	wing answers has to be "vee"	Ver	No
- F	Proved evidence of clinical neuropathy by MFs and MRC.		
- F	Pain with neuropathy-type of location and evidence of neuropathic pa	in _	
t	based on a score of ≥ 4/10 using the DN4 questionnaire		
- F	ain since more than 3 months		
- (Current pain intensity >2 (NRS 0-10)		
Chack	list for painlage nouronathy:		
The follo	wing answers has to be "yes"	Vec	No
- F	Proved evidence of clinical neuropathy by MFs and MRC		
- 1	No pain of any type		
- F - T	Proved evidence of clinical neuropathy by MFs and MRC No pain of any type		
The follo - F	wing answers has to be "yes" Proved evidence of clinical neuropathy by MFs and MRC No pain of any type	Yes	No

Form 1: Registration and	l Demographi	c details of	the patient	
Registration:				
epain patient ID (EquiSTA code num	1ber ###)	(
Registration date (interview date: d	d/mmm/yyyy)	[_ / _ _	_ _]/]	
linic record card number:		[
entre name: BLP (1) FMR (2)				
'HC name: Rhabha Hospital Bandra (1) Hosr	vital (2) Dharavi Urb	an Health Centre	(3) Worli BDD-45 (4)	
orivali (5), BLP Clinic (6), Narmada (Complex Panvel Clini	c (7), other (8))	(s), wom 000-15 (4),	
tart time: [_ _ /]		- (- // (- //		
nd time: [/]				
		_		
Demographic details o	of the patien	t:		
Date of Birth (dd/mmm/yyyy	<i>i</i>)	[_ _ / _ _	. _]/ _ _ _ _]	
OR Age of the patient (in years):	(actimate)		1 1	
Gender: Mai	e (1) Female (2)	[]	
Language: Hindi (1) Mar	athi (2) English (3)		
Religion: Hindu (1)/ Muslim	2)/ Christian (3)/ Jai	n (4)/ Other	_ [_]	
State: Maharashtra (1)/ Utta	ar Pradesh (2)/ Gujar	at (3)/ Other	_ [_]	
Marital status	t (1) Left (2)			
Single	1			
Married	2			
Widowed	3			
Divorced	4			
Illiterate	1		[]	
Primary	2			
Secondary	3			
High school	4			
College and above	5			
Other (specify)		7		
Occupation				
None		1		
Housewite		2		
Farmer / skilled labo	ur	4		
Office worker / Teac	her / Business	5		
Student		6		
Other (specify)		7		
Smoking status		0		
Current smoker	1			
Former smoker	2			
Occasionally	3			
Never smoked	4			
NOT KNOWN	0			

Form 2	: History	of the disease		
History of	the diseas	e		
(Patient first s	mptoms of the	– disease help seeking and the rea	sons behind the delay of stari	ing the
treatment)	,		,,	
Is it new or old	case? New (1) o	old (2)		[]
First s	mptoms:		Yes (1) No (0)	[]
lf yes				
	Skin lesion(s)		Yes (1) No (0)	<u> </u>
	Anaesthesia		Yes (1) No (0)	
	Loss of warms	iensation	Yes (1) No (0) Ves (1) No (0)	
	Loss of muscle	strength	Yes (1) No (0)	
	Symptom of T	1R	Yes (1) No (0)	
	Symptom of T	2R	Yes (1) No (0)	
	Other (specify)	Yes (1) No (0)	
	Skin lesions		Ves (1) No (0)	r 1
	• Type:	Patches (1), Nodules (2), Primary	neuritis (3). Don't know (8)	
	 Location 	on of lesion (Site)		
		Face	Yes (1) No (0)	
		Ears	Yes (1) No (0)	[]
		Upper extremities unilateral	Yes (1) No (0)	[]
		Upper extremities bilateral	Yes (1) No (0)	[_]
		Lower extremities unilateral	Yes (1) No (0)	
		Lower extremities bilateral	Yes (1) No (0)	
		Buttock	Yes (1) No (0)	
		All over	Yes (1) No (0)	
	 Numb 	er of skin lesions		
		SSL	Yes (1) No (0)	
		2 – 5	Yes (1) No (0)	[]
		6-10	Yes (1) No (0)	[_]
		Diffused lesions >10	Yes (1) No (0)	닏
	• 40000	Don't know	8	
	 Appea 	Hypo-pigmented (Pale)	Yes (1) No (0)	r 1
		Reddish lesions (ervthematic)	Yes (1) No (0)	
		Diffuse	Yes (1) No (0)	
		Anaesthetic	Yes (1) No (0)	
Date o	of first symptoms	:: [_ _ / _	_ _ / _ _ _ _	
Did yo	u recognise the	possibility of leprosy? Yes (1) / N	o (0) / don't know (3)	
Date o	f leprosy diagno	sis: [/		
How n	nany months it is	since became aware of the first	sign or symptoms of leprosy	· [_]
M/b	war the first -!	are the patient want to get hele?		r 1
where	Pharmacist (1) Traditional healer (2) Alternat	ive medicine practitioner (3)	
	Private or heal	th service clinic/doctor (4), Lepro	osy hospital or clinic (5),	
	Person affecte	d by leprosy (6), other – specify ([7]]	
OST IN LEPR	OSY Pa	tient Record Form: PI		Page 10

Duration of disease (presumably, if not known) < 6 months 6 months up to 1 year more than 1 year up to 5 years Result of slit skin smear at diagnosis: +ve (1) -ve (0) M	oneer than 5 years
Duration of disease (presumably, if not known) < 6 months 6 months up to 1 year more than 1 year up to 5 years Result of slit skin smear at diagnosis: +ve (1) -ve (0) M	oneer than 5 years
Duration of disease (presumably, if not known) < 6 months 6 months up to 1 year u more than 1 year up to 5 years Result of slit skin smear at diagnosis: +ve (1) -ve (0) M	onger than 5 years
Result of slit skin smear at diagnosis: +ve (1) -ve (0) M Bacterial Index (BI) at diagnosis:	inknown
Dacterial moex (bi) at diagnosis.	NA (3) [] []
Type of leprosy (WHO classification): PB (1) MB (2) Type of leprosy (R.J classification): TT (1)/BT (2)/BB (3)/BL (4)/LL (5)/PN (6)/Indeterminate	[]
(Enter using standard 0/1/2 coding for each area) Right Left [] Eyes [_] [] Hands [_] [] Feet [_]	
Disability at diagnosis Yes (1) M If yes, Complete Disability Record (CDR) at diagnosis:	No (0) []
C = clawed Yes (1) M C = wound or open crack Yes (1) M 	No (0) [] No (0) [] No (0) []
Started MDT: Yes (1) M If yes,	No (0)
MDT treatment: MB-MDT (1) PB-MDT (2) Other Completed MDT: Yes (1) YB-MDT (2) Yes (1) M	. / [] No (0) []
If yes, Date of MDT (completed): Total duration of MDT received: [.] /]] months
Did the patient ever have MDT prescribed in the past but treatment? Yes (1) / No (0) /	he/she did not complete the don't know (3) []
treatment? Yes (1) / No (0) /	don't know (3)

Form 3: History o	f current illness		
Skin changes			
• Time	of onset		
	< 6 months (1), 6 months up to	o 1 yr (2), > 1 yr up to 2 yr	rs (3),
	Longer than 2 years (4), Unkno	own (8)	
 Type 			
	Patches	Yes (1) No (0)	
	Nodules Primanu nouritir	Yes (1) No (0)	
• Locati	on of lesion (Site)	Tes (1) NO (0)	L]
Locut	Face	Yes (1) No (0)	[]
	Ears	Yes (1) No (0)	
	Upper extremities unilateral	Yes (1) No (0)	
	Upper extremities bilateral	Yes (1) No (0)	
	Lower extremities unilateral	Yes (1) No (0)	
	Lower extremities bilateral	Yes (1) No (0)	
	I runk Buttock	Tes (1) NO (0)	
	All over	Yes (1) No (0)	
Numb	er of skin lesions		
	SSL	Yes (1) No (0)	
	2-5	Yes (1) No (0)	
	6-10	Yes (1) No (0)	
	Diffused lesions >10	Yes (1) No (0)	
 Appa 	Don't know	ŏ	
- Apped	Hypo-pigmented	Yes (1) No (0)	[]
	Reddish lesions (erythematic)	Yes (1) No (0)	
	Diffuse	Yes (1) No (0)	
	Anaesthetic	Yes (1) No (0)	
	Not applicable	8	
Active Tatal	skin lesion(s)	Yes (1) No (0)	
Iotal	number of active skin lesions (if i	No, then #=0)	
If yes, area of sensory	loss) NO(0)	
Overs	kin lesion Yes (1) No (0)	
On th	e hands Yes (1) No (0)	
On the	e feet Yes (1) No (0)	
Uncer	tain Yes (1) No (0)	Ц
Did you notice any new loss of	sensation in your hands or feet	during the past month?	
Did you notice any new dryne: Did you notice any new weak	ss of your nand paims or foot soi less in your hands or feet during	es during the past month the past month?	
Did you notice any new weak	ess in your eye during the past r	nonth?	
Did you notice any new sensat	ions of pin and needles or 'insec	ts crawling' in your hands	s or feet
during the past month?			
Did you notice any new pain s	ensations, such as burning or sho	ooting pain in your hands	or feet
during the past month?	dente alta en el la marta de la		
(If a question is answered posi	tively, the patient will be asked v	which limb is affected. Th	is should be
Do you have any pain	around your nerve?	Yes (1)	No (0)
Have you noticed any	painless cuts or blisters? Yes (1)	No (0) [1	
,		-,-, L	

rosy reactions: Is the patient on reaction now? Yes (1) No (Is reaction diagnosed or new? Yes (1) No (If yes, How diagnosed? Skin lesion: Yes (1) No (If yes, Red Yes (1) No (Swollen Yes (1) No (o) [] o) []
Is the patient on reaction now? Yes (1) No (Is reaction diagnosed or new? Yes (1) No (If yes, How diagnosed? Skin lesion: Yes (1) No (If yes, Red Yes (1) No (Swollen Yes (1) No (o) [_] o) [_]
If yes, How diagnosed? Skin lesion: Yes (1) No (If yes, Red Yes (1) No (Swollen Yes (1) No (o, o)
Hyes, How diagnosed? Skin lesion: Yes (1) No (If yes, Red Yes (1) No (Swollen Yes (1) No (D) []
Skin lesion: Yes (1) No (If yes, Red Yes (1) No (Swollen Yes (1) No (D) []
If yes, Red Yes (1) No (Swollen Yes (1) No (-,
Red Yes (1) No (Swollen Yes (1) No (
Swollen Vec (1) No (D) [_]
54001cm 165(1) 140(D) []
Tender Yes (1) No (D) []
Nerve: Yes (1) No (0) [_]
If yes,	
Swollen Yes (1) No (0) [_]
Painful, tende Yes (1) No (0) [_]
reduced func Yes (1) No (D) [_]
New resions appear res (1) No (Fever and malaise Ves (1) No (ע <u>ו</u> ע היי
Tibial tenderness Yes (1) No (0) []
Orchitis Yes (1) No (o) [_]
Oedema Yes (1) No (D) [_]
If yes,	
Face Yes (1) No (0) [_]
Hands Yes (1) No (0) [_]
Feet Yes (1) No (0) [_]
Type of reaction is: TIRs (1) ENL (2) Neuritis (3) unce	rtain (8) []
Date of onset of reactions:	
Current treatment of reaction:	·_····
Steroid (Prednisolone) (1)/ Clofazimine (2)/ Thalidomide	(3) other (7) []
Dose:	., .,
Date of reaction's treatment started: [_ _ / _	_ _]/ _ _ _]_]
Duration of treatment: weeks	
History of reactions:	
Have you had previous reactions? Yes (1)	No (0) []
If yes,	
Number of previous reactions:	
	-





	er:	Patient	Initials:
How long you have P Paraethesia grade on Absent (0) Mild – only a	ins and needles/ parethesia (presi a 4 point scale: ware intermittently; does not limi	umably, if not known): [it activity (1)	. months []
Severe – inca	sleep disturbed and/or activities (i apacitating (3)	nciuding work) diminished	a (2)
Pain treatment (curr	ent and previous treatment):		
Current pain If yes	medication: Yes (1 s, what treatment Drug: Doce:) No (0)	
	How often you take it:		
ls your pain r How much de	elived by medical treatment? Yes oes it relieve your pain?	s (1) No (0)	
	PAIN SCORE 0-10 NUM	MERICAL RATING	
D No. Pan	0-10 Numerical Rat	ting Scale	0 rat Fan
Past pain me If yes	edication: Yes (1 s, what treatment in the past? Drug: Dose:) No (0)	
	Duration of time under pain m	edication (months):	

Are you happy or unhappy? Happy (1) unhappy (0) If you are feeling unhappy or depress what do you think the reasons? Is it due to (you can check more than one): Pain Yes (1) No (0) Leprosy Yes (1) No (0) General life Yes (1) No (0) Uncertain (8) Have you told anyone about the leprosy diagnosis? Yes (1) No (0) Family Yes (1) No (0) Education/work Yes (1) No (0) Education/work Yes (1) No (0) Community Yes (1) No (0) Are you willing to tell? Yes (1) No (0) Gonumity Yes (1) No (0) Are you willing to tell? Yes (1) No (0) Gonumity Yes (1) No (0) Friends Yes (1) No (0) Friends Yes (1) No (0) Gonumity Yes (1) No (0) Meighbours Yes (1) No (0) Gonumuity Yes (1) No (0) Friends Yes (1) No (0)	Problems associated with lepros	v and pain		
If you are feeling unhappy or depress what do you think the reasons? Is it due to (you can check more than one): Pain Yes (1) No (0) Leprosy Yes (1) No (0) General life Yes (1) No (0) Uncertain (8) Have you told anyone about the leprosy diagnosis? Yes (1) No (0) If yes, who already knows? Spouse Yes (1) No (0) Family Yes (1) No (0) Education/work Friends Yes (1) No (0) Education/work Neighbours Yes (1) No (0) Friends Spouse Yes (1) No (0) Friends Community Yes (1) No (0) Friends Spouse Yes (1) No (0) Friends Key who you are willing to tell? Yes (1) No (0) Friends Spouse Yes (1) No (0) Friends Yes (1) No (0) Reighbours Yes (1) No (0) Friends Yes (1) No (0) Friends Spouse Yes (1) No (0) Friends Yes (1)	Are you happy or unhappy?	Happy (1)	unhappy (0)	[]
Is it due to (you can check more than one): Pain Yes (1) No (0) Leprosy Yes (1) No (0) General life Yes (1) No (0) Uncertain (8)	If you are feeling unhappy or dep	ress what do you	think the reasons?	
Pain Yes (1) No (0) Image: second seco	ls it due to (you can check more t	- han one):		
Leprosy Yes (1) No (0) I General life Yes (1) No (0) I Have you told anyone about the leprosy diagnosis? Yes (1) No (0) I Family Yes (1) No (0) I Family Yes (1) No (0) I Education/work Yes (1) No (0) I Friends Yes (1) No (0) I Friends Yes (1) No (0) I Are you willing to tell other? Yes (1) No (0) I Friends Yes (1) No (0) I I Are you willing to tell other? Yes (1) No (0) I Friends Yes (1) No (0) I I Education/work Yes (1) No (0) I I Neighbours Yes (1) No (0) I I Reducation/work Yes (1) No (0) I I Reducation/work Yes (1) No (0) I I Reducation	Pain	Yes (1)	No (0)	
General life Yes (1) No (0) Image: specify the specific term of term	Leprosy	Yes (1)	No (0)	[]
Uncertain (8)	General life	Yes (1)	No (0)	[]
Have you told anyone about the leprosy diagnosis? Yes (1) No (0) If yes, who already knows? Spouse Yes (1) No (0) Family Yes (1) No (0) Education/work Yes (1) No (0) Education/work Yes (1) No (0) Friends Yes (1) No (0) Neighbours Yes (1) No (0) Community Yes (1) No (0) Are you willing to tell other? Yes (1) No (0) Family Yes (1) No (0) Education/work Yes (1) No (0) Education/work Yes (1) No (0) Community Yes (1) No (0) Neighbours Yes (1) No (0) Community Yes (1) No (0) Who you do not want them to know? Spouse Yes (1) No (0) Family Yes (1) No (0) Heighbours Yes (1) No (0) Community Yes (1) No (0) Heighbours Yes (1) No (0) Heighbours Yes (1) No (0) Heighbours Yes (1)	Uncertain	(8)		[]
Spouse Yes (1) No (0) Image: spouse indication indicatina inditeration indicatina inditeration indication inditer	Have you told anyone about the lepr If yes, who already knows?	osy diagnosis?	Yes (1) No (0)	
Family Yes (1) No (0)	Spouse	Yes (1)	No (0)	[]
Education/work Yes (1) No (0)	Family	Yes (1)	No (0)	[]
Friends Yes (1) No (0)	Education/work	Yes (1)	No (0)	[]
Neighbours Yes (1) No (0)	Friends	Yes (1)	No (0)	Ц
Community Yes (1) No (0)	Neighbours	Yes (1)	No (0)	
Are you willing to tell other? Yes (1) No (0)	Community	Yes (1)	No (0)	L,
If yes, who you are willing to tell? Spouse Yes (1) No (0) Family Yes (1) No (0) Education/work Yes (1) No (0) Friends Yes (1) No (0) Neighbours Yes (1) No (0) Community Yes (1) No (0) Who you do not want them to know? Spouse Yes (1) No (0) Family Yes (1) No (0) Image: Spouse Yes (1) No (0) Family Yes (1) No (0) Image: Spouse Yes (1) No (0) Image: Spouse<	Are you willing to tell other?		Yes (1) No (0)	
Spose Yes (1) No (0)	If yes, who you are willing to tell?	Vor (1)	No (0)	r 1
raminy res (1) No (0)	Spouse	Yes (1)	No (0)	
Education work Fes (1) No (0)	Family Education/work	Vec (1)	No (0)	
Neighbours Yes (1) No (0) Image: Community Community Yes (1) No (0) Image: Community Spouse Yes (1) No (0) Image: Community Family Yes (1) No (0) Image: Community Family Yes (1) No (0) Image: Community Friends Yes (1) No (0) Image: Community Friends Yes (1) No (0) Image: Community Neighbours Yes (1) No (0) Image: Community Marriage Yes (1) No (0) Image: Community If yes, problems in/with: Marriage Yes (1) No (0) Image: Community Marriage Yes (1) No (0) Image: Community Yes (1) No (0) Image: Community Keighbours Yes (1) No (0) Image: Community Yes (1) No (0) Imariage: Community Yes (1)	Friends	Yes (1)	No (0)	
Community Yes (1) No (0) Image: Community of the second sec	Neighbours	Yes (1)	No (0)	
Who you do not want them to know? Spouse Yes (1) No (0) Image: Spouse in the second se	Community	Yes (1)	No (0)	
Spouse Yes (1) No (0) Image: Spouse Family Yes (1) No (0) Image: Spouse Image:	Who you do not want them to know	?		
FamilyYes (1)No (0)IEducation/workYes (1)No (0)IFriendsYes (1)No (0)INeighboursYes (1)No (0)ICommunityYes (1)No (0)IHas the diagnosis of leprosy already caused a problem?Yes (1) No (0)IHas the diagnosis of leprosy already caused a problem?Yes (1) No (0)IIf yes, problems in/with:Yes (1)No (0)IEducation/workYes (1)No (0)IFriendsYes (1)No (0)IEducation/workYes (1)No (0)INeighboursYes (1)No (0)ICommunityYes (1)No (0)IMay it cause a problem in the future?Yes (1)No (0)IIf yes, problems in/with:Yes (1)No (0)IMarriageYes (1)No (0)IFamily membersYes (1)No (0)IEducation/workYes (1)No (0)IFriendsYes (1)No (0)IEducation/workYes (1)No (0)IFriendsYes (1)No (0)IReighboursYes (1)No (0)IReighboursYes (1)No (0)ICommunityYes (1)No (0)ICommunityYes (1)No (0)ICommunityYes (1)No (0)ICommunityYes (1)No (0)ICommunityYes (1)No (Spouse	Yes (1)	No (0)	
Education/work Yes (1) No (0)	Family	Yes (1)	No (0)	[]
FriendsYes (1)No (0)INeighboursYes (1)No (0)ICommunityYes (1)No (0)IHas the diagnosis of leprosy already caused a problem?Yes (1) No (0)IIf yes, problems in/with:Yes (1)No (0)IFamily membersYes (1)No (0)IEducation/workYes (1)No (0)IFriendsYes (1)No (0)INeighboursYes (1)No (0)ICommunityYes (1)No (0)IIf yes, problems in/with:Yes (1)No (0)IIf yes, problems in/with:Yes (1)No (0)IIf yes, problems in/with:Yes (1)No (0)IFamily membersYes (1)No (0)IFamily membersYes (1)No (0)IFamily membersYes (1)No (0)IFamily membersYes (1)No (0)IEducation/workYes (1)No (0)IFriendsYes (1)No (0)IFriendsYes (1)No (0)INeighboursYes (1)No (0)ICommunityYes (1)No (0)INeighboursYes (1)No (0)ICommunityYes (1)No (0)ICommunityYes (1)No (0)I	Education/work	Yes (1)	No (0)	[]
Neighbours Yes (1) No (0) Image (1) Community Yes (1) No (0) Image (1) Has the diagnosis of leprosy already caused a problem? Yes (1) No (0) Image (1) If yes, problems in/with: Marriage Yes (1) No (0) Image (1) Family members Yes (1) No (0) Image (1) Yes (1) No (0) Image (1) Family members Yes (1) No (0) Image (1) Yes (1) No (0) Image (1) Education/work Yes (1) No (0) Image (1) Yes (1) No (0) Image (1) Friends Yes (1) No (0) Image (1) Yes (1) No (0) Image (1) May it cause a problem in the future? Yes (1) No (0) Image (1) Yes (1) No (0) Image (1) If yes, problems in/with: Image (1) No (0) Image (1) No (0) Image (1)	Friends	Yes (1)	No (0)	[]
Community Yes (1) No (0) I Has the diagnosis of leprosy already caused a problem? Yes (1) No (0) I If yes, problems in/with: Marriage Yes (1) No (0) I Family members Yes (1) No (0) I I Family members Yes (1) No (0) I I Education/work Yes (1) No (0) I I Friends Yes (1) No (0) I I Neighbours Yes (1) No (0) I I Community Yes (1) No (0) I I May it cause a problem in the future? Yes (1) No (0) I If yes, problems in/with: Marriage Yes (1) No (0) I Family members Yes (1) No (0) I I Family members Yes (1) No (0) I I Education/work Yes (1) No (0) I I Friends Yes (1) No (0) I I Reighbours Yes (1) No (0) I I<	Neighbours	Yes (1)	No (0)	
Has the diagnosis of leprosy already caused a problem? Yes (1) No (0) If yes, problems in/with: Marriage Yes (1) No (0) I Family members Yes (1) No (0) I I Education/work Yes (1) No (0) I I Friends Yes (1) No (0) I I Neighbours Yes (1) No (0) I I Community Yes (1) No (0) I I May it cause a problem in the future? Yes (1) No (0) I If yes, problems in/with: Marriage Yes (1) No (0) I Family members Yes (1) No (0) I I Family members Yes (1) No (0) I I Education/work Yes (1) No (0) I I Friends Yes (1) No (0) I I Neighbours Yes (1) No (0) I I Community Yes (1) No (0) I I	Community	Yes (1)	No (0)	
Marriage Yes (1) No (0)	Has the diagnosis of leprosy already	caused a problem?	Yes (1) No (0)	
Wainage Yes (1) No (0)	Il yes, problems in/with.	Vor (1)	No (0)	r 1
Education/work Yes (1) No (0)	Family members	Yes (1)	No (0)	
Friends Yes (1) No (0)	Education/work	Yes (1)	No (0)	
Neighbours Yes (1) No (0)	Friends	Yes (1)	No (0)	
Community Yes (1) No (0) I May it cause a problem in the future? Yes (1) No (0) Yes (1) No (0) I If yes, problems in/with: Marriage Yes (1) No (0) I Family members Yes (1) No (0) I I Family members Yes (1) No (0) I I Education/work Yes (1) No (0) I I Friends Yes (1) No (0) I I Neighbours Yes (1) No (0) I I Community Yes (1) No (0) I I	Neighbours	Yes (1)	No (0)	
May it cause a problem in the future? Yes (1) No (0) If yes, problems in/with:	Community	Yes (1)	No (0)	
If yes, problems in/with: Marriage Yes (1) No (0) [_] Family members Yes (1) No (0) [_] Education/work Yes (1) No (0) [_] Friends Yes (1) No (0) [_] Neighbours Yes (1) No (0) [_] Community Yes (1) No (0) [_]	May it cause a problem in the future	?	Yes (1) No (0)	[]
Marriage Yes (1) No (0) [_] Family members Yes (1) No (0) [_] Education/work Yes (1) No (0) [_] Friends Yes (1) No (0) [_] Neighbours Yes (1) No (0) [_] Community Yes (1) No (0) [_]	If yes, problems in/with:			
Family members Yes (1) No (0) [_] Education/work Yes (1) No (0) [_] Friends Yes (1) No (0) [_] Neighbours Yes (1) No (0) [_] Community Yes (1) No (0) [_]	Marriage	Yes (1)	No (0)	[]
Education/workYes (1)No (0)FriendsYes (1)No (0)NeighboursYes (1)No (0)CommunityYes (1)No (0)	Family members	Yes (1)	No (0)	[]
Friends Yes (1) No (0) Neighbours Yes (1) No (0) Community Yes (1) No (0)	Education/work	Yes (1)	No (0)	
Neighbours Yes (1) No (0) [] Community Yes (1) No (0) [_]	Friends	Yes (1)	No (0)	
Community Yes (1) No (0)	Neighbours	Yes (1)	No (0)	
	Community	Yes (1)	No (0)	

Form 7: Past n	nedical histor	у		
Past medical history:				
Drug history (past and prese Current drug therap	ent): py:			
Previous drug treat	ment (potentially neurop	athy inducing ager	nts:	
Family history of neuropath	y:	Yes (1)	No (0)	
Data relevant to peripheral	neuropathy from medica	l records if presen	t	

Form 6.C	urrent ci	inical exa	inination	
* Accessmen	at of skin:	Clinical exar	nination:	
Skin lesions: Yes	(1) No	o (0)		[]
lf yes,	(_)	. (0)		·
Nature of le	esion (new (1) / d	old (2))		[]
Number of	lesions:			[]
(SSL (1)/ 2-5	5 (2)/ 6-10 (3)/ >	10 Diffused lesio	ns (4)/ PN (5))	
(Untreated	on (1) / nartially tre	ated (2) / fully to	(S) hatea	
Distribution	of any lesion: (use body chart)	cated (5))	
	Face		Yes (1) No (0)	[]
	lf ye	es: few (1)/ many	y (2)/ not known (3)	Ц
	Ears	an faur (1)/	Yes (1) No (0)	
	Upper extre	enities unilatera	y (2)/ not known (3) al Yes (1) No (0)	
	lfye	es: few (1)/ many	y (2)/ not known (3)	
	Upper extre	emities bilateral	Yes (1) No (0)	
	lf ye	es: few (1)/ many	y (2)/ not known (3)	
	Lower extre	emities unilatera es: few (1)/ man	ai Yes (1) No (0) v (2)/ not known (3)	
	Lower extre	emities bilateral	Yes (1) No (0)	
	lf ye	es: few (1)/ many	y (2)/ not known (3)	
	Trunk		Yes (1) No (0)	
	lf ye	es: few (1)/ many	y (2)/ not known (3)	
	BULLOCK	es: few (1)/ man	res (1) NO (0)	
	All over	est lew (1)/ many	Yes (1) No (0)	
	lf ye	es: few (1)/ many	y (2)/ not known (3)	
ls the lesions symm	etrically distributed	uted?	Yes (1) No (0)	[]
Type of lesions:	Macular		Ver (1) No (0)	r 1
	Papules		Yes (1) No (0)	
	Nodules		Yes (1) No (0)	
	Plagues		Yes (1) No (0)	[]
	Infiltration		Yes (1) No (0)	, [_]
Hypo pigmented les	sion count:	None (1) / s	ome (2) / all (3) / not known (8 ome (2) / all (3) / not known (8	9 L_1
Any shiny skin lesion Any erythrematous	lesions:	None (1) / s	ome (2) / all (3) / not known (8	9 L] 8) []
Are there any reacti	ional signs on the	e skin lesions?	Yes (1) No (0) L
lf yes,				
Oed	lema	Yes (1)	No (0)	
Any anaesthetic lesi	ions:	None (1) / s	ome (2) / all (3) / not known (8	a) [] (a
Sensory examinatio	on over the skin	lesions, MFs gra	dients;	,
0.05gm	0.2gm	2gm		
4gm	10gm	300gm	A ⁰	[]
OST IN LEPROSY	Patient	Record Form:	PI:	Page 19



A. Nerve papation: a. Nerves involved Yes (1) No (0) If No, skip to (nerve function) Name of Enlarged Nerve Right Yes (1) / No (0) / Not done (8) Yes (1) / No (0) / Not done (8) Upper limb (Yes/No) [_] [_] [_] Greater auricular [_] [_] Greater auricular [_] [_] Ulnar nerve [_] [_] Radial Cutaneous [_] [_] Ulnar branch [_] [_])/ Not done (8) [_] [_] [_] [_] [_] [_] [_]	Right Left Yes (1) No (0) [] If No, skip to (nerve function) [] Irged Nerve Right Left Yes (1) / No (0) / Not done (8) Yes (1) / No (0) / Not done (8) [es/No) [] [] Jar []		
a. Nerves involved Yes (1) No (0) If No, skip to (nerve function) Name of Enlarged Nerve Right Yes (1) / No (0) / Not done (8) Upper limb (Yes/No) [_] [_] Greater auricular [_] Ulnar nerve [_] Radial Cutaneous [_] Ulnar branch [_]) / Not done (8) [] [] [] [] [] [] []	a. Nerves involved Yes (1) No (0) If No, skip to (nerve function) Irged Nerve Right Yes (1) / No (0) / Not done (8) Yes (1) / No (0) / Not done (8) Yes (1) / No (0) / Not done (8) Yes (1) / No (0) / Not done (8) Yes (1) / No (0) / Not done (8) Yes (1) / No (0) / Not done (8) Yes (1) / No (0) / Not done (8) Yes (1) / No (0) / Not done (8) Yes (1) / No (0) / Not done (8) Yes (1) / No (0) / Not done (8) Yes (1) / No (0) / Not done (8) Yes (1) / No (0) / Not done (8) Yes (1) / No (0) / Not done (8) Yes (1) / No (0) / Not done (8)	A. Nerve palpation:	r 1
Name of Enlarged Nerve Right Yes (1) / No (0) / Not done (8) Left Yes (1) / No (0) / Not done (8) Upper limb (Yes/No) [_] [_] [_] Greater auricular [_] [_] Ulnar nerve [_] [_] Radial Cutaneous [_] [_] Ulnar branch [_] [_]) / Not done (8) [] [] [] [] [] []	Right Left Yes (1) / No (0) / Not done (8) Yes (1) / No (0) / Not done (8) 'es/No) [] [] Jar []	a. Nerves involved Yes (1) No (0)	
Name of Enlarged Nerve Right Yes (1) / No (0) / Not done (8) Left Yes (1) / No (0) / Not done (8) Upper limb (Yes/No) [_] [_] [_] Greater auricular [_] [_] Ulnar nerve [_] [_] Radial Cutaneous [_] [_] Ulnar branch [_] [_]) / Not done (8)	Right Yes (1) / No (0) / Not done (8) Left Yes (1) / No (0) / Not done (8) 'es/No) [_] [_] [_] ular [_] [_] 0us [_] [_] [_] [_] [_]	If No, skip to (herve function)	
Variate of charged iverve Yes (1) / No (0) / Not done (8) Yes (1) / No (0) / Not done (8) Upper limb (Yes/No) [] [] [] Greater auricular [] [_] Ulnar nerve [] [_] Radial Cutaneous [_] [_] Ulnar branch [_] [_])) / Not done (8) [] [] [] [] [] [] [] [Impediation Yes (1) / No (0) / Not done (8) Yes (1) / No (0) / Not done (8) Yes (1) / No (0) / Not done (8) Yes (1) / No (0) / Not done (8) [] Yes (1) / No (0) / Not done (8) [] [] Yes (1) / No (0) / Not done (8) Yes (1) / No (0) / Not done (8) [] Yes (1) / No (0) / Not done (8) [] [] Yes (1) / No (0) / Not done (8) [] [] Yes (1) / No (0) / Not done (8) [] [] Yes (1) / No (0) / Not done (8) [] [] Yes (1) / No (0) / Not done (8) [] [] Yes (1) / No (0) / Not done (8) [] [] Yes (1) / No (0) / Not done (8) [] [] Yes (1) / No (0) / Not done (8) [] [] Yes (1) / No (0) / Not done (8) [] [] Yes (1) / No (0) / Not done (8) [] [] Yes (1) / No (0) / Not done (8) [] [] Yes (1) / No (0) / Not done (8) [] [] Yes (1) / No (0) / Not done (8) [] [] Yes (1) / No (0) / Not done (8) [] []	Name of Falana Alana Right Left	
Upper limb (Yes/No) [_] [_] [_] Greater auricular [_] [_] Ulnar nerve [_] [_] Radial Cutaneous [_] [_] Ulnar branch [_] [_] Median [_] [_]		[_] [_] [_] ular [_] [_] ous [_] [_]	Name of Enlarged Nerve Yes (1) / No (0) / Not done (8) Yes (1) / No (0) / Not done (8)	;)
Greater auricular [_] [_] Ulnar nerve [_] [_] Radial Cutaneous [_] [_] Ulnar branch [_] [_] Median [_] [_]		ular [_] [_] ous [_] [_] ous [_] [_] [_] [_] [_] [_]	Upper limb (Yes/No) [] []	
Ulnar nerve [_] [_] Radial Cutaneous [_] [_] Ulnar branch [_] [_] Median [_] [_]		ous [_] [_]	Greater auricular [_] [_]	
Radial Cutaneous [_] [_] Ulnar branch [_] [_] Median [_] [_]		ous [_] [_]	Ulnar nerve [_] [_]	
Ulnar branch [_] [_] Median [_]			Radial Cutaneous []	
Median L_J L_J			Ulnar branch	
			Median	
	1 1			
		PS/INDIT LT LT LT LT		
Laterar populed []			Lateral poplitoal	
Sural [] []			Lateral popliteal	
Posterior tibial [] []		zal [_] [_] zal [_] roneal [_] [_]	Lateral popliteal [_] [_] Superficial peroneal [_] [_] Sural	
		c.,, c.,., c.,.,., c.,.,., c.,.,, c.,.,, c.,., c.,., c.,., c.,., c.,., c.,., c.,.	Lateral popliteal [_] [_] Superficial peroneal [_] [_] Sural [_] [_] Posterior tibial [_] [_]	
Right	left []	control control eal [_] roneal [_] [_] [_] il [_] il [_] ilarged nerves, if No, then #=0 [_] right [_] [_] b. Nerve tenderness Yes (1) No (0) If No, skip to (nerve function) [_]	Lateral popliteal [_] [_] Superficial peroneal [_] Sural [_] [_] Posterior tibial [_] Posterior tibial [_] Total lower [_] [_] Number of enlarged nerves, if No, then #=0 [_ _] right [_] left [_ b. Nerve tenderness Yes (1) No (0) If No, skip to (nerve function)	
Name of Tender Nerve Right Yes (1) / No (0) / Not done (8) Left Yes (1) / No (0) / Not done (8)	L] left [] [[control contro control control	Lateral popliteal [_] [_] Superficial peroneal [_] [_] Sural [_] [_] Posterior tibial [_] [_] Posterior tibial [_] [_] Total lower [_] [_] Number of enlarged nerves, if No, then #=0 [_] right [_] b. Nerve tenderness Yes (1) No (0) If No, skip to (nerve function) Yes (1) No (0) Name of Tender Nerve Right Left Yes (1) / No (0) / Not done (8) Yes (1) / No (0) / Not done (8)]
Name of Tender Nerve Right Yes (1) / No (0) / Not done (8) Left Yes (1) / No (0) / Not done (8) Upper limb (Yes/No) [_] [_] [_]	L] left [] [control contro control control	Lateral popliteal [_] [_] Superficial peroneal [_] [_] Sural [_] [_] Sural [_] [_] Posterior tibial [_] [_] Total lower [_] [_] Number of enlarged nerves, if No, then #=0 [_] right [_] b. Nerve tenderness Yes (1) No (0) If No, skip to (nerve function) Yes (1) No (0) Name of Tender Nerve Right Left Yes (1) / No (0) / Not done (8) Yes (1) / No (0) / Not done (8) Upper limb (Yes/No) [_] [_]	_] [_]
Name of Tender Nerve Right Yes (1) / No (0) / Not done (8) Left Yes (1) / No (0) / Not done (8) Upper limb (Yes/No) [] [] [] Greater auricular [] []	L] left [] [Left !) / Not done (8) [] []	c	Lateral popliteal [_] [_] Superficial peroneal [_] [_] Sural [_] [_] Posterior tibial [_] [_] Posterior tibial [_] [_] Total lower [_] [_] Number of enlarged nerves, if No, then #=0 [_] right [_] b. Nerve tenderness Yes (1) No (0) If No, skip to (nerve function) [_] Name of Tender Nerve Right Left Yes (1) / No (0) / Not done (8) Yes (1) / No (0) / Not done (8) Upper limb (Yes/No) [_] [_] Greater auricular [_] [_]	
Name of Tender Nerve Right Yes (1) / No (0) / Not done (8) Left Yes (1) / No (0) / Not done (8) Upper limb (Yes/No) [_] [_] [_] Greater auricular [_] [_] Ulnar nerve [_] [_]	L left [[c	Lateral popliteal [_] [_] Superficial peroneal [_] [_] Sural [_] [_] Sural [_] [_] Posterior tibial [_] [_] Total lower [_] [_] Number of enlarged nerves, if No, then #=0 [_]_] right [_] b. Nerve tenderness Yes (1) No (0) If No, skip to (nerve function) Yes (1) No (0) Name of Tender Nerve Right Left Yes (1) / No (0) / Not done (8) Yes (1) / No (0) / Not done (8) Upper limb (Yes/No) [_] [_] Greater auricular [_] [_] Ulnar nerve [_] [_]	
Name of Tender Nerve Right Yes (1) / No (0) / Not done (8) Left Yes (1) / No (0) / Not done (8) Upper limb (Yes/No) [_] [_] [_] Greater auricular [_] [_] Ulnar nerve [_] [_] Radial Cutaneous [_] [_]	L left [[c	Lateral popliteal [_] [_] Superficial peroneal [_] [_] Sural [_] [_] Sural [_] [_] Posterior tibial [_] [_] Posterior tibial [_] [_] Total lower [_] [_] Number of enlarged nerves, if No, then #=0 [_] right [_] b. Nerve tenderness Yes (1) No (0) right [b. Nerve tenderness Yes (1) No (0) No (0) If No, skip to (nerve function) Yes (1) / No (0) / Not done (8) Yes (1) / No (0) / Not done (8) Upper limb (Yes/No) [[] [] [] Greater auricular [] [] [] Ularat nerve [] [_] [_] Warst hered [_] [_] [_]	
Name of Tender Nerve Right Yes (1) / No (0) / Not done (8) Left Yes (1) / No (0) / Not done (8) Upper limb (Yes/No) [_] [_] [_] Greater auricular [_] [_] Ulnar nerve [_] [_] Radial Cutaneous [_] [_] Ulnar branch [_] [_]	L left [[c	Lateral popliteal [_] [_] Superficial peroneal [_] [_] Sural [_] [_] Sural [_] [_] Posterior tibial [_] [_] Posterior tibial [_] [_] Total lower [_] [_] Number of enlarged nerves, if No, then #=0 [_] right [_] Number of enlarged nerves, if No, then #=0 [_] right [_] Number of enlarged nerves, if No, then #=0 [_] right [_] Number of enlarged nerves, if No, then #=0 [_] right [_]	
Name of Tender Nerve Right Yes (1) / No (0) / Not done (8) Left Yes (1) / No (0) / Not done (8) Upper limb (Yes/No) [_] [_] [_] Greater auricular [_] [_] Ulnar nerve [_] [_] Radial Cutaneous [_] [_] Ulnar branch [_] [_] Median [_] [_]	L	c	Lateral popliteal [_] [_] Superficial peroneal [_] [_] Sural [_] [_] Sural [_] [_] Posterior tibial [_] [_] Total lower [_] [_] Number of enlarged nerves, if No, then #=0 [_] right [_] Number of enlarged nerves, if No, then #=0 [_] right [_] Number of enlarged nerves, if No, then #=0 [_] right [))
Name of Tender Nerve Right Yes (1) / No (0) / Not done (8) Left Yes (1) / No (0) / Not done (8) Upper limb (Yes/No) [_] [_] [_] Greater auricular [_] [_] Ulnar nerve [_] [_] Radial Cutaneous [_] [_] Ulnar branch [_] [_] Median [_] [_] Total upper [_] [_] [_]	Left (c	Lateral popliteal [_] [_] Superficial peroneal [_] [_] Sural [_] [_] Sural [_] [_] Posterior tibial [_] [_] Total lower [_] [_] Number of enlarged nerves, if No, then #=0 [_] right [_] b. Nerve tenderness Yes (1) No (0) If No, skip to (nerve function) Yes (1) No (0) / Not done (8) Vame of Tender Nerve Right Left Yes (1) / No (0) / Not done (8) Yes (1) / No (0) / Not done (8) Upper limb (Yes/No) [_] [_] [_] Greater auricular [_] [_] Ulnar nerve [_] [_] Radial Cutaneous [_] [_] Ulnar branch [_] [_] Median [_] [_] Total upper [_] [_] [_] Ioware limb (Yes/No) [_] [_]	
Name of Tender Nerve Right Yes (1) / No (0) / Not done (8) Left Yes (1) / No (0) / Not done (8) Upper limb (Yes/No) [_] [_] [_] Greater auricular [_] [_] Ulnar nerve [_] [_] Radial Cutaneous [_] [_] Ulnar branch [_] [_] Median [_] [_] Lower limb (Yes/No) [_] [_] Lateral popliteal [_] [_]	L	c	Lateral popliteal [_] [_] [_] Superficial peroneal [_] [_] Sural [_] [_] Posterior tibial [_] [_] Posterior tibial [_] [_] Total lower [_] [_] [_] Number of enlarged nerves, if No, then #=0 [_ _] right [_] left [_ b. Nerve tenderness Yes (1) No (0) If No, skip to (nerve function) Name of Tender Nerve Xes (1) / No (0) / Not done (8) Upper limb (Yes/No) [_] [_] [_] Greater auricular [_] [_] Ulnar nerve [_] [_] Radial Cutaneous [_] [_] Ulnar branch [_] [_] Median [_] [_] Total upper [_] [_] [_] [_] Lower limb (Yes/No) [_] [_] [_] Lateral popliteal [_] [_]	
Name of Tender Nerve Right Yes (1) / No (0) / Not done (8) Left Yes (1) / No (0) / Not done (8) Upper limb (Yes/No) [_] [_] [_] Greater auricular [_] [_] Ulnar nerve [_] [_] Radial Cutaneous [_] [_] Ulnar branch [_] [_] Median [_] [_] Total upper [_] [_] [_] Lower limb (Yes/No) [_] [_] Superficial peroneal [_] [_]	L] left [] Left)/ Not done (8) [_] [_] [_] [_] [_] [_] [_] [_]	comparing the set of the	Lateral popliteal [_] [_] Superficial peroneal [_] [_] Sural [_] [_] Sural [_] [_] Posterior tibial [_] [_] Total lower [_] [_] Number of enlarged nerves, if No, then #=0 [_] right [_] b. Nerve tenderness Yes (1) No (0) If No, skip to (nerve function) Yes (1) No (0) Name of Tender Nerve Right Left Yes (1) / No (0) / Not done (8) Yes (1) / No (0) / Not done (8) Upper limb (Yes/No) [_] [_] Greater auricular [_] [_] Ulnar nerve [_] [_] Median [_] [_] Total upper [_] [_] [_] Lower limb (Yes/No) [_] [_] Lateral popliteal [_] [_] Superficial peroneal [_] [_]	
Name of Tender Nerve Right Yes (1) / No (0) / Not done (8) Left Yes (1) / No (0) / Not done (8) Upper limb (Yes/No) [_] [_] [_] Greater auricular [_] [_] Ulnar nerve [_] [_] Radial Cutaneous [_] [_] Ulnar branch [_] [_] Median [_] [_] Total upper [_] [_] [_] Lower limb (Yes/No) [_] [_] Superficial peroneal [_] [_]	L] left [] Left	comparison comparison <td>Lateral popliteal [_] [_] Superficial peroneal [_] [_] Sural [_] [_] Sural [_] [_] Posterior tibial [_] [_] Total lower [_] [_] Number of enlarged nerves, if No, then #=0 [_] right [_] b. Nerve tenderness Yes (1) No (0) If No, skip to (nerve function) Yes (1) No (0) Name of Tender Nerve Right Left Yes (1) / No (0) / Not done (8) Yes (1) / No (0) / Not done (8) Upper limb (Yes/No) [_] [_] Greater auricular [_] [_] Ulnar nerve [_] [_] Median [_] [_] It upper [_] [_] [_] Lower limb (Yes/No) [_] [_] Lateral popliteal [_] [_] Superficial peroneal [_] [_]</td> <td>_] [_]</td>	Lateral popliteal [_] [_] Superficial peroneal [_] [_] Sural [_] [_] Sural [_] [_] Posterior tibial [_] [_] Total lower [_] [_] Number of enlarged nerves, if No, then #=0 [_] right [_] b. Nerve tenderness Yes (1) No (0) If No, skip to (nerve function) Yes (1) No (0) Name of Tender Nerve Right Left Yes (1) / No (0) / Not done (8) Yes (1) / No (0) / Not done (8) Upper limb (Yes/No) [_] [_] Greater auricular [_] [_] Ulnar nerve [_] [_] Median [_] [_] It upper [_] [_] [_] Lower limb (Yes/No) [_] [_] Lateral popliteal [_] [_] Superficial peroneal [_] [_]	_] [_]
Name of Tender Nerve Right Yes (1) / No (0) / Not done (8) Left Yes (1) / No (0) / Not done (8) Upper limb (Yes/No) [_] [_] [_] Greater auricular [_] [_] Ulnar nerve [_] [_] Radial Cutaneous [_] [_] Ulnar branch [_] [_] Median [_] [_] Total upper [_] [_] [_] Lateral popliteal [_] [_] Superficial peroneal [_] [_] Sural [_] [_]	L] left [] Left [_] [_] [_] [_] [_] [_] [_] [_] [_] [_]	c	Lateral popliteal [_] [_] Superficial peroneal [_] [_] Sural [_] [_] Sural [_] [_] Posterior tibial [_] [_] Total lower [_] [_] Number of enlarged nerves, if No, then #=0 [_]	
Right Left	L] left [_[] Left	control contro control control	teral popliteal [_] [_] uperficial peroneal [_] [_] ural [_] [_] osterior tibial [_] [_] ostal lower [_] [_] umber of enlarged nerves, if No, then #=0 [_] right [_] b. Nerve tenderness Yes (1) No (0) If No, skip to (nerve function) Right Left	_
Name of Tender Nerve Right Yes (1) / No (0) / Not done (8) Left Yes (1) / No (0) / Not done (8) pper limb (Yes/No) [_] [_] [_] reater auricular [_] [_] lnar nerve [_] [_] adial Cutaneous [_] [_]	L left [c	ateral popliteal [_] [_] uperficial peroneal [_] [_] ural [_] [_] osterior tibial [_] [_] otal lower [_] [_] [_] lumber of enlarged nerves, if No, then #=0 [_ _] right [_] left [_] lumber of enlarged nerves, if No, then #=0 [_ _] right [_] left [_] b. Nerve tenderness Yes (1) No (0) If No, skip to (nerve function) Name of Tender Nerve Xes (1) / No (0) / Not done (8) Pper limb (Yes/No) [_] [_] [_] reater auricular [_] [_] Inar nerve [_] [_] adial Cutaneous [_] [_]	_] _[_]
Name of Tender Nerve Right Yes (1) / No (0) / Not done (8) Left Yes (1) / No (0) / Not done (8) Upper limb (Yes/No) [_] [_] [_] Greater auricular [_] [_] Ulnar nerve [_] [_] Radial Cutaneous [_] [_] Ulnar branch [_] [_]	L left [[c	Lateral popliteal [_] [_] Superficial peroneal [_] Sural [_] [_] Posterior tibial [_] [_] Posterior tibial [_] [_] Total lower [_] [_] Number of enlarged nerves, if No, then #=0 [_]_] right [_] left [_ b. Nerve tenderness Yes (1) No (0) If No, skip to (nerve function) Name of Tender Nerve Yes (1) / No (0) / Not done (8) Upper limb (Yes/No) [_] [_] Greater auricular [_] [_] Unar nerve [_] [_] Radial Cutaneous [_] [_] Unar branch [_] [_] Median [_] [_]]
Name of Tender Nerve Right Yes (1) / No (0) / Not done (8) Left Yes (1) / No (0) / Not done (8) Upper limb (Yes/No) [_] [_] [_] Greater auricular [_] [_] Ulnar nerve [_] [_] Radial Cutaneous [_] [_] Ulnar branch [_] [_] Median [_] [_] Total upper [_] [_] [_]	L] left [] Left [_] [_] [_] [_] [_] [_] [_] [_] [_] [_]	c	Lateral popliteal [_] [_] Superficial peroneal [_] [_] Sural [_] [_] Sural [_] [_] Posterior tibial [_] [_] Total lower [_] [_] Number of enlarged nerves, if No, then #=0 [_] right [_] b. Nerve tenderness Yes (1) No (0) If No, skip to (nerve function) Yes (1) No (0) Name of Tender Nerve Right Left Yes (1) / No (0) / Not done (8) Yes (1) / No (0) / Not done (8) Upper limb (Yes/No) [_] [_] Greater auricular [_] [_] Ulnar nerve [_] [_] Radial Cutaneous [_] [_] Ulnar branch [_] [_] Median [_] [_]]
Name of Tender Nerve Right Yes (1) / No (0) / Not done (8) Left Yes (1) / No (0) / Not done (8) Upper limb (Yes/No) [_] [_] [_] Greater auricular [_] [_] Ulnar nerve [_] [_] Radial Cutaneous [_] [_] Ulnar branch [_] [_] Median [_] [_] Total upper [_] [_] [_] Lower limb (Yes/No) [_] [_]	L] left [] Left)/ Not done (8) [_] [_] [_] [_] [_] [_] [_] [_]	c	Lateral popliteal [_] [_] Superficial peroneal [_] [_] Sural [_] [_] Sural [_] [_] Posterior tibial [_] [_] Posterior tibial [_] [_] Total lower [_] [_] Number of enlarged nerves, if No, then #=0 [_] right [_] b. Nerve tenderness Yes (1) No (0) If No, skip to (nerve function) Yes (1) No (0) Name of Tender Nerve Right Left Yes (1) / No (0) / Not done (8) Yes (1) / No (0) / Not done (8) Upper limb (Yes/No) [_] [_] Greater auricular [_] [_] Ulnar nerve [_] [_] Ulnar branch [_] [_] Median [_] [_] Total upper [_] [_] [_] Lower limb (Yes/No) [_] [_]	
Name of Tender Nerve Right Yes (1) / No (0) / Not done (8) Left Yes (1) / No (0) / Not done (8) Upper limb (Yes/No) [_] [_] [_] Greater auricular [_] [_] Ulnar nerve [_] [_] Radial Cutaneous [_] [_] Ulnar branch [_] [_] Median [_] [_] Total upper [_] [_] [_] Lower limb (Yes/No) [_] [_] Lateral popliteal [_] [_]	L	comparing the system of the	Lateral popliteal [_] [_] Superficial peroneal [_] [_] Sural [_] [_] Sural [_] [_] Posterior tibial [_] [_] Posterior tibial [_] [_] Total lower [_] [_] Number of enlarged nerves, if No, then #=0 [_] right [_] b. Nerve tenderness Yes (1) No (0) If No, skip to (nerve function) Yes (1) No (0) Name of Tender Nerve Right Left Yes (1) / No (0) / Not done (8) Yes (1) / No (0) / Not done (8) Upper limb (Yes/No) [_] [_] Greater auricular [_] [_] Ulnar nerve [_] [_] Ulnar branch [_] [_] Ulnar branch [_] [_] Median [_] [_] Lower limb (Yes/No) [_] [_] Lateral popliteal [_] [_]	
Name of Tender Nerve Right Yes (1) / No (0) / Not done (8) Left Yes (1) / No (0) / Not done (8) Upper limb (Yes/No) [_] [_] [_] Greater auricular [_] [_] Ulnar nerve [_] [_] Radial Cutaneous [_] [_] Ulnar branch [_] [_] Median [_] [_] Total upper [_] [_] [_] Lower limb (Yes/No) [_] [_] Superficial peroneal [_] [_]	L	count count <td< td=""><td>Lateral popliteal [_] [_] [_] Superficial peroneal [_] [_] Sural [_] [_] Posterior tibial [_] [_] Posterior tibial [_] [_] Total lower [_] [_] [_] Number of enlarged nerves, if No, then #=0 [_ _] right [_] left [_ b. Nerve tenderness Yes (1) No (0) If No, skip to (nerve function) Name of Tender Nerve Xes (1) / No (0) / Not done (8) Upper limb (Yes/No) [_] [_] [_] Greater auricular [_] [_] Ulnar nerve [_] [_] Radial Cutaneous [_] [_] Ulnar branch [_] [_] Median [_] [_] Lower limb (Yes/No) [_] [_] [_] Lower limb (Yes/No) [_] [_] [_] Lateral popliteal [_] [_] Superficial peroneal [_] [_]</td><td></td></td<>	Lateral popliteal [_] [_] [_] Superficial peroneal [_] [_] Sural [_] [_] Posterior tibial [_] [_] Posterior tibial [_] [_] Total lower [_] [_] [_] Number of enlarged nerves, if No, then #=0 [_ _] right [_] left [_ b. Nerve tenderness Yes (1) No (0) If No, skip to (nerve function) Name of Tender Nerve Xes (1) / No (0) / Not done (8) Upper limb (Yes/No) [_] [_] [_] Greater auricular [_] [_] Ulnar nerve [_] [_] Radial Cutaneous [_] [_] Ulnar branch [_] [_] Median [_] [_] Lower limb (Yes/No) [_] [_] [_] Lower limb (Yes/No) [_] [_] [_] Lateral popliteal [_] [_] Superficial peroneal [_] [_]	
Name of Tender Nerve Right Yes (1) / No (0) / Not done (8) Left Yes (1) / No (0) / Not done (8) Upper limb (Yes/No) [_] [_] [_] Greater auricular [_] [_] Ulnar nerve [_] [_] Radial Cutaneous [_] [_] Ulnar branch [_] [_] Median [_] [_] Total upper [_] [_] [_] Lower limb (Yes/No) [_] [_] Superficial peroneal [_] [_]	L	comparing the second	Lateral popliteal [_] [_] [_] Superficial peroneal [_] [_] Sural [_] [_] Posterior tibial [_] [_] Posterior tibial [_] [_] Total lower [_] [_] [_] Number of enlarged nerves, if No, then #=0 [_ _] right [_] left [_ b. Nerve tenderness Yes (1) No (0) If No, skip to (nerve function) Name of Tender Nerve Xes (1) / No (0) / Not done (8) Upper limb (Yes/No) [_] [_] [_] Greater auricular [_] [_] Ulnar nerve [_] [_] Ulnar nerve [_] [_] Nadial Cutaneous [_] [_] Ulnar branch [_] [_] Median [_] [_] Lower limb (Yes/No) [_] [_] [_] Lateral popliteal [_] [_] Superficial peroneal [_] [_] Sural [_]	
Name of Tender Nerve Right Yes (1) / No (0) / Not done (8) Left Yes (1) / No (0) / Not done (8) Upper limb (Yes/No) [_] [_] [_] Greater auricular [_] [_] Ulnar nerve [_] [_] Radial Cutaneous [_] [_] Ulnar branch [_] [_] Median [_] [_] Total upper [_] [_] [_] Lower limb (Yes/No) [_] [_] Superficial peroneal [_] [_] Sural [_] [_]	L] left [] Left []_ [] [_] [eal	Lateral popliteal [_] [_] [_] Superficial peroneal [_] [_] Sural [_] [_] Posterior tibial [_] [_] Total lower [_] [_] [_] Number of enlarged nerves, if No, then #=0 [_ _] right [_] left [_ b. Nerve tenderness Yes (1) No (0) If No, skip to (nerve function) Name of Tender Nerve Xes (1) / No (0) / Not done (8) Upper limb (Yes/No) [_] [_] [_] Greater auricular [_] [_] Ulnar nerve [_] [_] Ulnar nerve [_] [_] Radial Cutaneous [_] Ulnar branch [_] [_] Median [_] [_] Total upper [_] [_] [_] Lateral popliteal [_] Superficial peroneal [_] Sural [_] Sural [_] Cater tibial [_]	-]
			dian [_] [_]	
edian [] []		<u> </u>	nar branch	
nar branch [_] [_] edian [_]			idial Cutaneous	
adial Cutaneous [_] [_] nar branch [_] [_] edian [_] [_]		ous	nar nerve	
Inar nerve [_] [_] Idial Cutaneous [_] [_] Inar branch [_] [_] edian [_] [_]		ous	reater auricular [_]	
ireater auricular [_] [_] Inar nerve [_] [_] adial Cutaneous [_] [_] Inar branch [_] [_] Iedian [_] [_]		ular [_] [_] ous [_] [_] ous [_] [_] [_] [_] [_]	pper limb (Yes/No) [] []	
Breater auricular [_] [_] JInar nerve [_] [_] Iadial Cutaneous [_] [_] JInar branch [_] [_] Addian [_] [_]		Image: Second	Yes (1) / No (0) / Not done (8) Yes (1) / No (0) / Not done (8)	9
Jpper limb (Yes/No) [] [_] [_] [_] Sreater auricular [_] [_] [_] Ilnar nerve [_] [_] [_] Iadial Cutaneous [_] [_] [_] Ilnar branch [_] [_] [_]		(es/No) [_] [_] <td< td=""><td>Vame of Enlarged Nerve Yes (1) / No (0) / Not done (8) Yes (1) / No (0) / Not done (8)</td><td>a </td></td<>	Vame of Enlarged Nerve Yes (1) / No (0) / Not done (8) Yes (1) / No (0) / Not done (8)	a
Name of Enlarged Nerve No (0) / No (0) / Not done (8) Yes (1) / No (0) / Not done (8) Jpper limb (Yes/No) [_] [_] [_] Greater auricular [_] [_] Jinar nerve [_] [_] Iadial Cutaneous [_] [_] Inar branch [_] [_]) / Not done (8)	arged Nerve No (0) / Not done (8) Yes (1) / No (0) / Not done (8) 'es/No) [_] [_] [_] ular [_] [_]	If No, skip to (nerve function)	
If No, skip to (nerve function) Name of Enlarged Nerve Right Yes (1) / No (0) / Not done (8) Left Yes (1) / No (0) / Not done (8) Jpper limb (Yes/No) [_] [_] [_] Greater auricular [_] [_] JInar nerve [_] [_] Idaial Cutaneous [_] [_] JInar branch [_] [_] Median [_] [_])/ Not done (8) [] [] [] [] [] [] [] [If No, skip to (nerve function) Left Irged Nerve Right Left Yes (1) / No (0) / Not done (8) Yes (1) / No (0) / Not done (8) Ies/No) [] [] Jar []	A. Nerve palpation: a. Nerves involved Yes (1) No (0)	[]
If No, skip to (nerve function) If No, skip to (nerve function) Name of Enlarged Nerve Right Yes (1) / No (0) / Not done (8) Left Yes (1) / No (0) / Not done (8) Upper limb (Yes/No) [_] [_] [_] Greater auricular [_] [_] Ulnar nerve [_] [_] Radial Cutaneous [_] [_] Ulnar branch [_] [_]) / Not done (8) [] [] [] [] [] [] []	If No, skip to (nerve function) Left rged Nerve Right Left Yes (1) / No (0) / Not done (8) Yes (1) / No (0) / Not done (8) 'es/No) [] [] ular [] [] ous [] []	Nerve examination: A. Nerve palpation:	r ۱



- Joint positi	on sense:	Normal (0)/ Abr	ormal (1) []
	Right	Le	eft
	Normal (0)/ Abnormal (1)	Normal (0)/	Abnormal (1)
Upper limb			
Index finger		[_
Lower limb			
Great toe		[]
A. Reflex	25:	Normal (0)/ Abn	ormal (1) []
	Right	Le	eft
	Normal (0) / Absent (1)/ Increased (2)	Normal (0) / Absen	t (1)/ Increased (2)
Upper Limb: Norm	al (0)/ Abnormal (1) []		
Biceps	[[[]
Triceps	[[]
Supinator	[[]
Lower Limb: Norma	al (0)/ Abnormal (1) []	-	
Knee		1	
Ankie Ext.elenter	Descrifferrien of the his too Ves / No		
cxt planter	borsiliexion of the big toe res / No	DOISINEXION OF the	Dig toe res/ No
Current Disability If yes, Current Complete C =	L Hands L Hands L Feet Yes (1) Disability Record (CDR): clawed Y wound or open crack Y	No (0) es (1) No (0) es (1) No (0)	
(<i>p</i>) = =	shortening level Y	es (1) No (0)	
=	shortening level Y	es (1) No (0)	

	DN4 questionn	aire
DN4 completed Date DN4 completed Start time: [/] End time: [/]	(dd/mm/yyyy)	Yes (1) No (0) [[/ _ / _
DN4 score DN4 patient's result	Yes ≥4 (1)	[No ≤3 (0)
PainDETECT completed Date PainDETECT completed Start time: [/] End time: [/]	PainDETECT questio	onnaire Yes (1) No (0) [] [/ _ _ _]
PainDETECT score PainDETECT patient's score	Yes (1)	No (0)
GHQ-12 completed Date GHQ-12 completed	(dd/mm/yyyy)	Yes (1) No (0) [[/ /]] /]
GHQ-12 completed Date GHQ-12 completed Start time: [/] End time: [/] GHQ-12 score	(dd/mm/yyyy)	Yes (1) No (0) [
GHQ-12 completed Date GHQ-12 completed Start time: [/] End time: [/] GHQ-12 score GHQ-12 patient's result Briat	(dd/mm/yyyy) Yes ≥4 (1)	Yes (1) No (0) [[// _ /] [No ≤3 (0) [uestionnaire
GHQ-12 completed Date GHQ-12 completed Start time: [/] End time: [/] GHQ-12 score GHQ-12 patient's result Brief	(dd/mm/yyyy) Yes ≥4 (1) Pain Inventory Qu	Yes (1) No (0) [[/ _ / _ [] No ≤3 (0) [Jestionnaire
GHQ-12 completed Date GHQ-12 completed Start time: [/] End time: [/] GHQ-12 score GHQ-12 patient's result BPI: Date GHQ-12 done Start time: [/] End time: [/]	(dd/mm/yyyy) Yes ≥4 (1) Pain Inventory Qu (dd/mm/yyyy)	Yes (1) No (0) [[/ / No ≤3 (0) [Jestionnaire Yes (1) No (0) [[/ _ /]
GHQ-12 completed Date GHQ-12 completed Start time: [_/ /] End time: [_/ /] GHQ-12 score GHQ-12 patient's result BPI: Date GHQ-12 done Start time: [_/ /] End time: [_/ /] BPI score: BPI patient's result	(dd/mm/yyyy) Yes ≥4 (1) Pain Inventory Qu (dd/mm/yyyy) Yes (1)	Yes (1) No (0) [[/ _ / No ≤3 (0) [Jestionnaire Yes (1) No (0) [[/ _ / [] No (0) [

Date QST testing dome (dd/mm/yyyy)	Form 11: Quanti	tative Sen	sory Testi	ng	
RIGHT Hand C8 Yes (1) No (0)	QST testing Date QST testing done (dd/mm Start time: [/] End time: [/] Site of QST	/үүүү) [_ _ /	Yes (1) ' _ /	No (0) .[]]	
LEFT Hand C8 Yes (1) No (0)	RIGHT Hand C8		Yes (1)	No (0)	[]
RIGHT Foot S1 Yes (1) No (0)	LEFT Hand C8		Yes (1)	No (0)	
LEFT Foot \$3 Yes (1) No (0)	RIGHT Foot S1		Yes (1)	No (0)	
Form 12: Skin biopsy Skin biopsy taken Yes (1) No (0) Date skin biopsy taken (dd/mm/yyyy)	LEFT Foot S1 Number of QST tested sites		Yes (1)	No (0)	
Skin biopsy taken Yes (1) No (0)	Form 12:Skin b	iopsy			
Skin biopsy taken Yes (1) No (0)	Shie bizerutekee		V (1)	N= (0)	. 1
Biopsy site	Date skin biopsy taken (dd/mm Start time: [/ /] End time: [/ /]	/////// [_ _ /	1_ _ / _ _	_ _]	
RIGHT Hand C8 Yes (1) No (0)	Biopsy site				
LEFT Hand C8 Yes (1) No (0) RIGHT Leg Yes (1) No (0) LEFT Leg1 Yes (1) No (0) Number of skin biopsy taken	RIGHT Hand C8		Yes (1)	No (0)	
RIGHT Leg Yes (1) No (0)	LEFT Hand C8		Yes (1)	No (0)	[]
LEFT Leg1 Yes (1) No (0)	RIGHT Leg		Yes (1)	No (0)	
Number of skin biopsy taken	LEFT Leg1		Yes (1)	No (0)	
Date skin biopsy code:	Number of skin biopsy taken Skin biopsy code:				
Date skin biopsy received at KCL: (dd/mm/yyyy) [] Date skin biopsy received at KCL: (dd/mm/yyyy) [] Biopsy report 1: [] Normal IENFD 1 Increased IENFD 2 Decreased IENFD 3 Missing 9 Biopsy report 2: [] Normal IENFD 1 Increased IENFD 2 Decreased IENFD 2 Decreased IENFD 2 Decreased IENFD 2 Decreased IENFD 3 Missing 9 Biopsy report 1: [] Normal IENFD 1 Increased IENFD 2 Decreased IENFD 2 Decreased IENFD 3 Missing 9 Biopsy report 2: [] Normal IENFD 1 Increased IENFD 2 Decreased IENFD 2 Decreased IENFD 3 Missing 9 Diopsy report 2: [] Normal IENFD 1	Date skin biopsy code.		dd/mm/www)		1 1 1 1
Biopsy report 1: Normal IENFD 1 Increased IENFD 2 Decreased IENFD 3 Missing 9 Biopsy report 2: Normal IENFD 1 Increased IENFD 2 Decreased IENFD 3 Missing 9 Biopsy report 1: Normal IENFD 1 Increased IENFD 2 Decreased IENFD 2 Decreased IENFD 3 Missing 9 Biopsy report 2: Normal IENFD 1 Increased IENFD 2 Decreased IENFD 3 Missing 9 Biopsy report 2: Normal IENFD 1 Increased IENFD 2 Decreased IENFD 3 Missing 9 Biopsy report 2: Normal IENFD 1 Increased IENFD 2 Decreased IENFD 3 Missing 9 Decreased IENFD 3 Missing 9 Decreased IENFD 3 Missing 9 Dther findings or remarks:	Date skin biopsy received at KC	L: (e	id/mm/yyyy) [
Normal IENFD 1 Increased IENFD 2 Decreased IENFD 3 Missing 9 Biopsy report 2: [] Normal IENFD 1 Increased IENFD 2 Decreased IENFD 3 Missing 9 Biopsy report 1: [] Normal IENFD 1 Increased IENFD 2 Decreased IENFD 1 Increased IENFD 2 Decreased IENFD 2 Decreased IENFD 2 Decreased IENFD 3 Missing 9 Biopsy report 2: [] Normal IENFD 1 Increased IENFD 2 Decreased IENFD 2 Decreased IENFD 3 Missing 9 Stopsy report 2: [] Normal IENFD 1 Increased IENFD 2 Decreased IENFD 3 Missing 9 Other findings or remarks:	Biopsy report 1:				
Increased IENFD 2 Decreased IENFD 3 Missing 9 Biopsy report 2: [] Normal IENFD 1 Increased IENFD 2 Decreased IENFD 3 Missing 9 Biopsy report 1: [] Normal IENFD 1 Increased IENFD 2 Decreased IENFD 3 Missing 9 Biopsy report 2: [] Normal IENFD 1 Increased IENFD 1 Increased IENFD 2 Decreased IENFD 3 Missing 9 Biopsy report 2: [] Normal IENFD 1 Increased IENFD 2 Decreased IENFD 3 Missing 9 Decreased IENFD 3 Missing 9 Decreased IENFD 1 Increased IENFD 2 Decreased IENFD 3 Missing 9 Decreased IENFD 4 Missing 9 Missing 9 Mi	Normal IENFD	1			
Decreased IENFD 3 Missing 9 Biopsy report 2: [] Normal IENFD 1 Increased IENFD 2 Decreased IENFD 3 Missing 9 Biopsy report 1: [] Normal IENFD 1 Increased IENFD 2 Decreased IENFD 3 Missing 9 Biopsy report 2: [] Normal IENFD 1 Increased IENFD 1 Increased IENFD 2 Decreased IENFD 3 Missing 9 Decreased IENFD 3 Missing 9 Decreased IENFD 2 Decreased IENFD 1 Increased IENFD 2 Decreased IENFD 3 Missing 9 Detreased IENFD 3 Missing 9 Detreased IENFD 3 Missing 9	Increased IENFD	2			
Missing 9 Biopsy report 2: [Decreased IENFD	3			
Normal IENFD 1 Increased IENFD 2 Decreased IENFD 3 Missing 9 Biopsy report 1: [] Normal IENFD 1 Increased IENFD 2 Decreased IENFD 2 Decreased IENFD 3 Missing 9 Biopsy report 2: [] Normal IENFD 1 Increased IENFD 2 Decreased IENFD 1 Increased IENFD 1 Increased IENFD 2 Decreased IENFD 3 Missing 9 Other findings or remarks:	Rionsy report 2:	9			r 1
Increased IENFD 2 Decreased IENFD 3 Missing 9 Biopsy report 1: [] Normal IENFD 1 Increased IENFD 2 Decreased IENFD 3 Missing 9 Biopsy report 2: [] Normal IENFD 1 Increased IENFD 2 Decreased IENFD 3 Missing 9 Dther findings or remarks:	Normal IENFD	1			
Decreased IENFD 3 Missing 9 Biopsy report 1: [] Normal IENFD 1 Increased IENFD 2 Decreased IENFD 3 Missing 9 Biopsy report 2: [] Normal IENFD 1 Increased IENFD 2 Decreased IENFD 2 Decreased IENFD 3 Missing 9 Other findings or remarks:	Increased IENFD	2			
Missing 9 Biopsy report 1: [] Normal IENFD 1 Increased IENFD 2 Decreased IENFD 3 Missing 9 Biopsy report 2: [] Normal IENFD 1 Increased IENFD 2 Decreased IENFD 2 Decreased IENFD 3 Missing 9 Other findings or remarks:	Decreased IENFD	3			
Biopsy report 1: [_] Normal IENFD 1 Increased IENFD 2 Decreased IENFD 3 Missing 9 Biopsy report 2: [_] Normal IENFD 1 Increased IENFD 2 Decreased IENFD 3 Missing 9 Other findings or remarks:	Missing	9			
Normal IENFD 1 Increased IENFD 2 Decreased IENFD 3 Missing 9 Biopsy report 2: [_] Normal IENFD 1 Increased IENFD 2 Decreased IENFD 3 Missing 9 Other findings or remarks:	Biopsy report 1:	-			
Decreased IENFD 2 Decreased IENFD 3 Missing 9 Biopsy report 2: [_] Normal IENFD 1 Increased IENFD 2 Decreased IENFD 3 Missing 9 Other findings or remarks:	Normal IENFD	1			
Missing 9 Biopsy report 2: [_] Normal IENFD 1 Increased IENFD 2 Decreased IENFD 3 Missing 9 Other findings or remarks:	Decreased IENED	2			
Biopsy report 2: [_] Normal IENFD 1 Increased IENFD 2 Decreased IENFD 3 Missing 9 Other findings or remarks:	Missing	9			
Normal IENFD 1 Increased IENFD 2 Decreased IENFD 3 Missing 9 Other findings or remarks:	Biopsy report 2:	2			[]
Increased IENFD 2 Decreased IENFD 3 Missing 9 Other findings or remarks:	Normal IENFD	1			<u> </u>
Decreased IENFD 3 Missing 9 Dther findings or remarks:	Increased IENFD	2			
Missing 9 Dther findings or remarks:	Decreased IENFD	3			
Other findings or remarks:	Missing	9			
	Other findings or remarks:			·	

Res	NR	No	D	e sample taken: Date result received:							
		E	Description	Res	NR						
		2	Hormones								
			T3		70-204ng/dl						
	4.7-6mill/c.mm		T4		3.2-12.6µg/dl						
	13.5-18g/dl		TSH		0.45-4.5µIU/mI						
	42-52%	6	VitB12 level		211-911pgmi						
	78-100fl.	7	Pregnancy test								
	27-31pg	8	HIV-DUO								
	32-36g/dl		Non reactive		0.0-1.0 index						
-	11.5-14%		Keactive		Above 1.0 index						
+	(Normal)	2	General Even								
+		+	Volume		ml						
			Colour		Pale vellow						
		+	Appearance		Clear						
			Deposit		Absent						
			Reaction (Ph)		4.5-8.0						
			Specific gravity		1.01-1.03						
			Chemical Exm:								
	4000-10500/c.m		Albumin		Absent						
	40-80%		Sugar		Absent						
	20-40%		Acetone		Absent						
	2-10%		Bile pigments		Absent						
	1-6%	L	Bile salts		Absent						
	0-2%	<u> </u>	Urobilinogen		Normal						
		I	Nitrates		Negative						
	150-450X1000/c		Micro Exm:		44						
	6-9.5cum		Red blood cells		Absent						
	0.2-0.5%		Pus cells (WBCs)		0.0-5.0/hpf						
	9-1/70 0.15%.mm/hr		Epitheliai celis		0.0-4.0/npj						
	ma/di	+	Cast		Absent						
	mgyar	+	Ddeposits		Absent						
		+	Bacteria		Absent						
	Below 1.0 index		Trichomonas Vag		Absent						
	Above 1.0 index		Yeast cells		Absent						
		78-100fl. 27-31pg 32-36g/dl 11.5-14% (Normal) 4000-10500/c.m 4000-10500/c.m 40-80% 20-40% 2-10% 1-6% 0-2% 150-450X1000/c 6-9.5cum 0.2-0.5% 9-17% 0-15%mm/hr mg/dl Below 1.0 index Above 1.0 index	78-100fl. 7 27-31pg 8 32-36g/dl 11.5-14% (Normal) 9 (Normal) 9 4000-10500/c.m 1 4000-10500/c.m 1 2-10% 1 1-6% 0 0.240% 1 1.50-450X1000/c 1 59-5cum 0.2-0.5% 9-17% 0 0.2105%mm/hr mg/dl Below 1.0 index Above 1.0 index	78-100fl. 7 Pregnancy test 27-31pg 8 HIV-DUO 32-36g/dl Non reactive 11.5-14% Reactive (Normal) 9 Urine Exam. (Normal) 9 Urine Exam. Colour Appearance Deposit Reaction (Ph) Reaction (Ph) Specific gravity Chemical Exm: Chemical Exm: 4000-10500/c.m Albumin 40-80% Sugar 20-40% Acetone 2-10% Bile pigments 1-6% Bile salts 0-2% Urobilinogen 150-450X1000/c Micro Exm: 6-9.5cum Red blood cells 0.2-0.5% Pus cells (WBCs) 9-17% Epithelial cells 0-15%mm/hr Crystals mg/dl Cast Deleosits Bacteria Below 1.0 index Trichomonas Vag Above 1.0 index Yeast cells	78-100fl. 7 Pregnancy test 27-31pg 8 HIV-DUO 32-36g/dl Non reactive 11.5-14% Reactive (Normal) 9 Urine Exam. Volume Colour Appearance Deposit Deposit Reaction (Ph) Specific gravity Chemical Exm: 4000-10500/c.m Albumin 4000-10500/c.m Albumin 4000-10500/c.m Albumin 2-10% Sugar 2-0.40% Acetone 2-10% Bile pigments 1-6% Bile salts 0-2% Urobilinogen 150-450X1000/c Micro Exm: 6-9.5cum Red blood cells 0.2-0.5% Pus cells (WBCs) 9-17% Epithelial cells 0-15%mm/hr Crystals mg/dl Cast Below 1.0 index Trichomonas Vag Above 1.0 index Yeast cells						

Form A: Voluntar	y muscle testing		
ID number	QST_IN_LEPROS	YSIUDY	Date
Manual			
muscle assessme	iit.		
Modified 5-point	MRC scale for muscle strength scorin	g (Grading crite	ria for the VMT)
	HANDS AND FEET		MRC GRADE
Full range of mov	ement		5
Full range of move	ement, reduced resistance		3
Reduced range of	movement, some joint movement		2
Flicker only			1
Full paralysis			0
Nerve Ulnar	Movement Little finger abduction		Muscle/Muscle group abductor digiti minimi
Ulnar	Index finger abduction		1 st dorsal interosseos
Median	Thumb abduction		abductor pollicis brevis
Radial	Wrist extension		Wrist extensors
Lateral popliteal	Foot dorsiflexion		Foot dorsiflexors
Lateral popliteal	Extension big toe		Extensor hallucis longous
Lateral popliteal	Toe fanning		Intrinsic muscle of the foot
If any particular m missing value (9) particular follow-r	nuscle could not be tested (e.g. becaus will be recorded for the nerve score. S up time, a missing value will be record	se of joint stiffne imilarly, if test d led.	ess or previous surgery), a lata are not available for any
Criteria for motor	impairment was any muscle scoring,	4.	
This form modi	fied from: Notes for completing lepro	osy neurological	assessment, developed by
This form mou	Prof Diana Lockwood, Hospital for	Tropical Disease	s, London

Appendix 24. Modified 5 - points MRC scale for muscle strength scoring

Study code:		Study number:	Patient Initi	als:
Form B: Sensor	y testing usin	g Semmes–Weinstein monofilaments (MF)		
Use the nylon i	nonofilamen	ts on the marked sites on hands and feet		
Palms		Soles		
R	ι	R	L	
J.	Ŵ		SD	
Hands-do full ra	ange	Feet – omit 0.05 g	m / 0.2 gm	
Perform the ev	aluation in th	e sequence listed below, and document the	first nylon wł	ich has a
Nylon colour	Approx.	Interpretation	Score	Score
Green	Force (0.05 gm)	Sensation within normal limits for the	Hands 0	Feet -
Blue	(0.2 gm)	hand and foot Diminished light touch sensation in the	1	
Dide	(0.2 gm)	hand with difficulty in the fine tactile discrimination. Within normal limits for the foot	-	
Purple	(2.0 gm)	Diminished protective sensation in the	2	0
Dark Red	(4.0 gm)	Loss of protective sensation of the hand, in some cases for the foot. Usually loss of	3	1
Orange	(10.0 gm)	Definite loss of protective sensation of the foot. Continues to feel deep pressure	4	2
Bright red	(300.0	Able to feel deep pressure and pain	5	3
No response	gm) ()	Loss of deep pressure sensation. Usually	6	6
Missing/unable		does not feel pain.	U	U
to test			-	-
Criterion for se If a patient scor level for all site This form modi	nsory impain res 3 or more s on the hand fied from: Not L	ment for any nerve, the nerve had sensory impair l is 0.05mg; the normal sensation level for al es for completing leprosy neurological assessmu ockwood, Hospital for Tropical Diseases, London	ment. The no I sites on the ent, developed N	rmal sensation foot is 2g. I by Prof. Diana
	1 1 1 1			

Appendix 25. Sensory testing using Semmes-Weinstein monofilaments



Appendix 26. Body chart





Appendix D

Questionnaires

Appendix 28. DN4 Questionnaire

DN4 – QUESTIONNAIRE

To estimate the probability of neuropathic pain, please answer yes or no for each item of the following four questions.

INTERVIEW OF THE PATIENT

QUESTION 1: Does the pain have one or more of the following characteristics? Burning Painful cold	YES	NO
Electric shocks	🖬	•
QUESTION 2:		
Is the pain associated with one or more of the following		
symptoms in the same area?	YES	NO
Tingling	🖬	
Pins and needles	🖬	
Numbness	🖬 👘	
Itching	🖬	

EXAMINATION OF THE PATIENT	
QUESTION 3:	
Is the pain located in an area where the physical examination may reveal one or more of the following characteristics? YES	NO
Hypoesthesia to touch	
Hypoesthesia to pinprick	
QUESTION 4: In the painful area, can the pain be caused or increased by: YES Brushing?	NO
YES = 1 point NO = 0 points Patient's Score:	/10

		Rulient Initia	λ:
		et a lim	
		Stort time	3-
		End fine	
	डी एन ४ - फ़ार्म (DN 4 QUESTIONNAIR	RE)	
	आपके तत्रिकार्यकृति दद का अनुमान लगान क जवाब दीजिये:	िलिए, कृपया अगल चार प्रश्नाक 'हा' या 'ना' म	
	रोगी की मुलाकातः		
	प्रश्न रः		
	वया आपके दर्द की एक या ज्यादा नीचे लिखी	हुई विशेषताएं हैं?	
	हाँ	ँ नहीं	
_	তালন্য		
\sim \sim	दर्दनाक ठंडा		
	दिजली के झटके जैसा 📃		
	TH 2		
	पश्च २ः जन्म अगस्य सर्व जीने किये गव्ह या ज्यादा जय	भाग में मिलना जनना ⁸ 7	
	क्या आपकः दद नाच लिख एक या ज्यादा ल॰ टाँ	त्व स विवय युवसारः वर्ध	
	रू सन्दरनी जैमा		
	जुबाजुवा असा 🛄 🛄		
	सन्न हो जाता		
	खलती जैसा		
	3		
	प्रश्न ३:		
	क्या आपका दुई ऐसे जगहपर है जहाँ शारीरिक	परिक्षण के बाद नाचे लिखे गए एक या ज्यादा ———————————————————————————————————	
0	ददादयतह? हा	नहा	
	स्पश को को अनुभव 🛄		
	पुरान का कन अनुमय 🔝		
	দেশ ধঃ		
	क्या वह दर्दनाक जगहूपर, नोचे लिखी क्रिया व	इर लेपर दर्द शुरू होता या बढ़ता है?	
-	हाँ	नहीं	
	झडना		
	हाँ = १	1 4	
	नहीं = ०		
	रागा का अक =	/10	
आपको	ये प्रश्नावली केंड्रा	CTAIL 2	
(ৰহুল ভার	सान (1)/ आञ्चाल (०)/		
	1 - 1 (10) (- 2) + (- 2)	म (३)/कडीन (4)/बहुत -	চঠীন(চ্র)–
100			

Appendix 29. DN4 Questionnaire – Hindi vesrion

								_
	DN4 प्रश्र्मावली (Question	naire)						
	चेतारुग्णी वेदलेची संभाव्यत मुद्यांचे कृपया हो किंवा न	ता (probability गाही स्वरूपात उ) जाणून घेण्य त्तर द्या.	ासाठी खालील	चार प्रश्र्लांमधील	Ŧ		
	रुग्णाची मुलाखत							
	प्रश्न १: तेटनेची सालील एक किंस	- সংহিত্য নগ্পটা	आहेत का ?					
	વદ્યવા વ્યાભાભ ૨૧૦ વિગ્લ	। आयम लबाग हो	जाहत प्रथम न जाही					
	নক নক							
\sim	वेदनामय थंड							
Q 2	वियुत झटका							
	प्रश्रच २:							
	वेदना होत असलेल्या भाग	ात खातील लक्ष टो	तणे देखील वा नादी	टत आहेत का	?			
	सुंग्या थेणे (tingling)							
	टाचणी आणि सुई							
	वधीर जान							
	७३णाचा तथासणा प्रश्रज्ञ ३:							
\sim	खाती दिलेल्यातील एक व	अधिक लक्षणे	दिसून थेतील	अशा ठिकाणी	वेदना होत आहे	त का ?		
	स्पर्शाची जाणीव कजी							
	टाचणीची टोच कमी							
	प्रश्नन ४:							
	वेदनामय जागी खालील 1	क्रिया केल्यानंतः	र वेदना वाढते ।	व सुरु होते क	17			
	धासणे ?	Ļ	—					
	हो = १ गुण				1 -			
	लाही = ॰ गुण				4			
	रुग्णाचे गुण /१०							
CLAPICA	हा प्रकाव	aft aba	ମ ସହ	201 2				
(ञ्चुपच २	nan (1)/2mm	137		- C-11 - X - Y				
	- A DIVINI	(2)/जल्ब	ज्म (३) / २	চঠীল (4)/खुपच	কঠীল	(5)-L	

Appendix 30. DN4 Questionnaire – Marathi version

Date :	Pallants La	f name		net neene:	
How would you assess	s your pain mins, at it	in manuntif	,	Please mark you	r.
0 1 2 5	4 8 6 7	e + 10			
How alterng was the si	rengest pain during	The part 4 weeks?			2
	4001	e a 10	*	5	1 1
toe strong was the pa	in during the past 4 v	egeneve no steer	: <i> </i>	20 1	
0 1 2 8	4 5 \$ 7	8 5 10	0	10 6	113
Note the picture the	t best describes the	eourse of your		V	
pain :	ersistent pain with			1	
	light Bucketione undetent oaks with a	• ·	1		11
i handa i	Rectos	- L			
	un stazia alberi ela interes tiera		Does your body?	pein radiele io othe yes 🔄 no	r regions of your
	ain attacks with pair atmosp them		Fye	, planes drow the d which the pain rad	inaction in Islas
never 🗌 Do you have a tingit	hardly noticed	elightly	modenizaty	etrongly	very strongly
in light touching (die	neraly nabala staing, a bientist) is 1	signoy It is area painful?		amongry []	
never	hendly noticed	elightly	moderately	etrongly 📃	very strongly
	handly maticand	alightiy	modentaly	alongly 🗌	
ie cold or heat (bath histor	handly noticed	identiy indexes	moderatialy	alrongly	very strongly
Do you cuffer from a	execution of sumin	and in the areas	that you marked?		
	u in Trin Link, 4.4. W	aignty Che Taget, trigge	modentiny		
naver	handly noticed	elightly	moderately	strongly	very strongly
newer -	handly noticed	elightly	moderately	etrongly	very strongly
	×1 =	z2 -	x8=	24 -	x 5 -
				_	
		Total o	inter and a sub-		

Appendix 31. PainDETECT Questionnaire



Appendix 32. PainDETECT Questionnaire - Hindi version



Appendix 33. PainDETECT Questionnaire – Marathi vesrion



Appendix 34. Brief Pain Inventory Questionnaire (BPI)

8.	In the provid	last 24 led? P	hours lease	, how r circle th	much n he one	elief ha percer	ive pair itage tř	n treatm at mos	nents o at show	r med s how	ications much relief
	0% No Relict	10%	20%	30%	40%	50%	60%	70%	80%	90%	100% Complete Relief
9.	Circle	the on ered wi	e num th you	ber tha	at desci	rbes h	ow, dui	ing the	past 2	4 hou	rs, pain has
	A.	Gene	ral Act	ivity		体别的	P. N.				
	0 Does Interfe	not ere	2	3	4	5	6	7	8	9	10 Completely Interferes
	B. 0 Dces	Mood 1 not	2	3	4	5	6	7	8	9	10 Completely Interferes
	C.	Walki	ng Abi	lity	Contraction of the second	RUS			S. E. L.	NOD THE	interretes
	0 Does Interfe	1 not ere	2	3	4	5	6	7	8	g	10 Completely Interferes
	D.	Norm	al Wor	k (inclu	ides bo	oth wor	k outsi	de the i	home a	and ho	ousework)
	0 Does Interf	1 not ere	2	3	4	5	6	7	8	8	10 Completely Interferes
	E.	Relat	ions wi	th othe	er peop	le	6	7	8	0	10
	Does Interf	not ere	2	3	4	D	o	1	0	9	Completely Interferes
	0 Does Interf	Sleer 1 not ere	2	3	4	5	6	7	8	9	10 Completely Interferes
	G.	Enjoy	ment	of life	and the		0	7	0	0	10
	0 Does Interf	1 not ere	2	3	4	5	6	7	8	9	Completely Interferes
				Cx	ooynght 19 Pa A	9' Cherlos in Researc II rights no	s S. Cleela sh Group isorved.	nd, PhU			

Brief Pain Inventory इस चित्रकारी पर, जहाँ आपको दर्द महसूस होता है वहाँ निशान लगाइये. जहाँ आपको सबसे ज्यादा दर्द हो रहा है वहाँ 'X' लिखिए. Fur S hurs १) कृपया जो अंक आपके पिछले हफ्ते का सबसे दर्दनाक अनुभव का वर्णन करता है उस अंक को गोल कीजिये. 8 2 3 8 ц ξ ی ٢ ٩ 80 दर्द नहीं सबसे दर्दनाक २) कृपया जो अंक आपके मध्यम तौर पर होनेवाले दर्द का वर्णन करता है उस अंक को गोल कीजिये. 8 3 3 8 ξ فا ٢ ٩ 80 ч दर्द नहीं सबसे दर्दनाक 3) कृपया जो अंक आपको इस वक्त होनेवाले दर्द का वर्णन करता है उस अंक को गोल कीजिये. १२ Ş ξ 6 ٢ ٩ 80 8 4 दर्द नहीं सबसे दर्दनाक 1 4 Continue...

Appendix 35. Brief Pain Inventory Questionnaire - Hindi version

सरसक्षेप माम में इस्तक्षेप का वर्णन करता हैं जोई करतक्षेप नहीं अंग विष्ठ हेफ्त होनेवाले दर्द की वजहरों आपके काम में इस्तक्षेप का वर्णन करता हैं अंग विष्ठ हेफ्त होनेवाले दर्द की वजहरों आपके काम में इस्तक्षेप का वर्णन करता है अंग विष्ठ हेफ्त होनेवाले दर्द की वजहरों आपके काम में इस्तक्षेप का वर्णन करता है अंग विष्ठ हेफ्त हों ० १ २ 3 ४ ५ ६ ७ ८ ९ १० कोई हस्तक्षेप नहीं पूरी तरह से इस्तक्षेप का ह २ 3 ४ ५ ६ ७ ८ ९ १० कोई हस्तक्षेप नहीं ० १ २ 3 ४ ५ ६ ७ ८ ९ १० कोई हस्तक्षेप नहीं ९ हर २ 3 ४ ५ ६ ७ ८ ९ १० कोई हस्तक्षेप नहीं ० १ २ 3 ४ ५ ६ ७ ८ ९ १० कोई हस्तक्षेप नहीं 9 दुसरे लोगोंके साथ संबंध ९ हर २ ३ ४ ५ ६ ७ ८ ९ १० कोई हस्तक्षेप नहीं को ह र ३ ४ ५ ६ ७ ८ ९ १० कोई हस्तक्षेप नहीं ग्री तरह से हस्तक्षेप										
इस्तक्षेप मापन जोड विद्यके सन्त के दर्ज के का कृपया गौल की जिये. अ) साधारण हलवल ० १ २ 3 ४ ५ ६ ७ ८ ९ १० कोई हस्तक्षेप नहीं २) माधारण हलवल २) मावेदणा २) मावेदणा ० १ २ 3 ४ ५ ६ ७ ८ ९ १० कोई हस्तक्षेप नहीं २) मलने की क्षमता ० १ २ 3 ४ ५ ६ ७ ८ ९ १० कोई हस्तक्षेप नहीं २) मलने की क्षमता ० १ २ 3 ४ ५ ६ ७ ८ ९ १० कोई हस्तक्षेप नहीं ३) सामान्य काम (घर के और बाहर के) ० १ २ 3 ४ ५ ६ ७ ८ ९ १० कोई हस्तक्षेप नहीं २) सुको की क्षमता ० १ २ 3 ४ ५ ६ ७ ८ ९ १० कोई हस्तक्षेप नहीं २) सुको ली के क्षमता ० १ २ 3 ४ ५ ६ ७ ८ ९ १० कोई हस्तक्षेप नहीं ० १ २ 3 ४ ५ ६ ७ ८ ९ १० कोई हस्तक्षेप नहीं ० १ २ 3 ४ ५ ६ ७ ८ ९ १० दुरी तरह से हस्तक्षेप ० १ २ ३ ४ ५ ६ ७ ८ ९ १० कोई हस्तक्षेप नहीं ० १ २ ३ ४ ५ ६ ७ ८ ९ १० दुरी तरह से हस्तक्षेप ०) जिन्दगी से आनंद ० १ २ ३ ४ ५ ६ ७ ८ ९ १० दुरी तरह से हस्तक्षेप ० १ २ ३ ४ ५ ६ ७ ८ ९ १० दुरी तरह से हस्तक्षेप ० १ २ ३ ४ ५ ६ ७ ८ ९ १० कोई हस्तक्षेप नहीं ० १ २ ३ ४ ५ ६ ७ ८ ८ ९ १० कोई हस्तक्षेप नहीं									80 	
मस्तक्षेप मापन जो क्रंक पिछले इफ्से होनेवाले दर्द की वजहरंसे आपके काम में हस्तक्षेप का वर्णन करता है अंक को कृपया गोल कीजिये. अ) सापारण हलचल ० १ २ 3 ४ ५ ६ ७ ८ ९ १० पूरी तरह से हस्तक्षेप व) मनोदशा ० १ २ 3 ४ ५ ६ ७ ८ ९ १० पूरी तरह से हस्तक्षेप ० १ २ 3 ४ ५ ६ ७ ८ ९ १० पूरी तरह से हस्तक्षेप ०) पलने की हामता ० १ २ 3 ४ ५ ६ ७ ८ ९ १० पूरी तरह से हस्तक्षेप ०) पलने की हामता ० १ २ 3 ४ ५ ६ ७ ८ ९ १० पूरी तरह से हस्तक्षेप ह) सामान्य काम (पर के और बाहर के) ० १ २ 3 ४ ५ ६ ७ ८ ९ १० पूरी तरह से हस्तक्षेप ह) नुसरे तानहीं ० १ २ 3 ४ ५ ६ ७ ८ ९ १० पूरी तरह से हस्तक्षेप ह) नुसरे लोगोंक साथ संबंध ० १ २ 3 ४ ५ ६ ७ ८ ९ १० पूरी तरह से हस्तक्षेप ० १ २ ३ ४ ५ ६ ७ ८ ९ १० पूरी तरह से हस्तक्षेप ० १ २ ३ ४ ५ ६ ७ ८ ९ १० पूरी तरह से हस्तक्षेप ० १ २ ३ ४ ५ ६ ७ ८ ९ १० पूरी तरह से हस्तक्षेप गोर्गेह स्तक्षेप नहीं ० १ २ २ ३ ४ ५ ६ ७ ८ ९ १० पूरी तरह से हस्तक्षेप गोर्ग दा र इ स्तक्षेप नहीं ० १ २ ३ ४ ५ ६ ७ ८ ९ १० पूरी तरह से हस्तक्षेप गोर्ग दा र र ने स्तक्षेप नहीं ० १ २ ३ ४ ५ ९ १ ९ ८ ९ १ ९ १ ० १ २ २ ३ ४ ५ ९ १ ९ ८ ९ १ ९ १ ० १ २ २ ३ ४ ९ १ ९ १ ८ ७ १ ९ १ ० १ २ २ ३ ४ ४										
इस्तक्षेप मापन जो अंक विछले इसते होनेवाले दर्द की वजहरों आपके काम में हस्तक्षेप का वर्णन करता है अंक को कृपया गोल कीजिये. अ) साधारण हलचल ० १ २ 3 ४ ५ ६ ७ ८ ९ १० थूरी तरह से हस्तक्षेप य) मनोदशा ० १ २ 3 ४ ५ ६ ७ ८ ९ १० पूरी तरह से हस्तक्षेप य) मनोदशा ० १ २ 3 ४ ५ ६ ७ ८ ९ १० पूरी तरह से हस्तक्षेप ग) चलने की हामता ० १ २ 3 ४ ५ ६ ७ ८ ९ १० पूरी तरह से हस्तक्षेप छ) चलने की हामता ० १ २ 3 ४ ५ ६ ७ ८ ९ १० पूरी तरह से हस्तक्षेप इस्तक्षेप नहीं ० १ २ 3 ४ ५ ६ ७ ८ ९ १० पूरी तरह से हस्तक्षेप इस्तक्षेप नहीं ० १ २ 3 ४ ५ ६ ७ ८ ९ १० पूरी तरह से हस्तक्षेप इस्तक्षेप नहीं ० १ २ 3 ४ ५ ६ ७ ८ ९ १० पूरी तरह से हस्तक्षेप इस्तक्षेप नहीं ० १ २ 3 ४ ५ ६ ७ ८ ९ १० पूरी तरह से हस्तक्षेप एरी तरह से हस्तक्षेप ० १ २ 3 ४ ५ ६ ७ ८ ९ १० पूरी तरह से हस्तक्षेप एरी तरह से हस्तक्षेप ० १ २ 3 ४ ५ ६ ७ ८ ९ १० दे १० पूरी तरह से हस्तक्षेप ० १ २ 3 ४ ५ ६ ७ ८ ९ १० पूरी तरह से हस्तक्षेप ० १ २ 3 ४ ५ ६ ८ ७ ८ ९ १० ते हे हस्तक्षेप ० १ २ 3 ४ ५ ६ ८ ७ ८ ९ १० ते हे हस्तक्षेप ० १ २ ३ ४ ५ ६ ७ ८ ९ १० ते हे हस्तक्षेप ० १ २ ३ ४ ५ ६ ७ ८ ९ १० ते हे हस्तक्षेप										
हस्तक्षेप मापन जो अंक पिछले, इसने होनेवाले दर्द की वजहरों आपके काम में हस्तक्षेप का वर्णन करता है उस अंक को कृपया गोल कीजिये. अ) साधारण हलयल ० १ २ 3 ४ ५ ६ ७ ८ ९ १० कोई हस्तक्षेप नहीं 2) म्रोतेरह वोई हस्तक्षेप नहीं ० १ २ 3 ४ ५ ६ ७ ८ ९ १० कोई हस्तक्षेप नहीं 0 ग वलने की क्षमता ० १ २ 3 ४ ५ ६ ७ ८ ९ १० कोई हस्तक्षेप नहीं 2) सामान्य काम (घर के और बाहर के) 5) सामान्य काम (घर के और बाहर के) 5) सामान्य काम (घर के और बाहर के) 5) हा सामान्य काम (घर के और बाहर के) 7) कोई हस्तक्षेप नहीं 2) दुसरे लोगोंके साथ संबंध फ) नींद फ) जींद 7) जिन्दगी से आनंद 7) जिन्दगी से आनंद										
जो अंक पिछले हफ्ते होनेवाले दर्द की वजहरू आपके काम में हस्तक्षेप का वर्णन करता है उस अंक को कृपया गोल कीजिये. 3) साधारण हलपल • १ २ 3 ४ ५ ६ ७ ८ ९ १० कोई हस्तक्षेप नहीं • १ २ 3 ४ ५ ६ ७ ८ ९ १० कोई हस्तक्षेप नहीं • १ २ 3 ४ ५ ६ ७ ८ ९ १० कोई हस्तक्षेप नहीं • १ २ 3 ४ ५ ६ ७ ८ ९ १० कोई हस्तक्षेप नहीं • १ २ 3 ४ ५ ६ ७ ८ ९ १० कोई हस्तक्षेप नहीं • १ २ 3 ४ ५ ६ ७ ८ ९ १० कोई हस्तक्षेप नहीं • १ २ 3 ४ ५ ६ ७ ८ ९ १० कोई हस्तक्षेप नहीं • १ २ 3 ४ ५ ६ ७ ८ ९ १० कोई हस्तक्षेप नहीं • १ २ 3 ४ ५ ६ ७ ८ ९ १० कोई हस्तक्षेप नहीं • १ २ 3 ४ ५ ६ ७ ८ ९ १० कोई हस्तक्षेप नहीं • १ २ 3 ४ ५ ६ ७ ८ ९ १० कोई हस्तक्षेप नहीं • १ २ 3 ४ ५ ६ ७ ८ ९ १० कोई हस्तक्षेप नहीं • १ २ 3 ४ ५ ६ ७ ८ ९ १० कोई हस्तक्षेप नहीं • १ २ 3 ४ ५ ६ ७ ८ ९ १० कोई हस्तक्षेप नहीं • १ २ 3 ४ ५ ६ ७ ८ ९ १० कोई हस्तक्षेप नहीं • १ २ 3 ४ ५ ६ ७ ८ ९ १० कोई हस्तक्षेप नहीं • १ २ 3 ४ ५ ६ ७ ८ ९ १० कोई हस्तक्षेप नहीं		हस्तक्षेप मापन								
3) साधारण हतचल \circ ? 2 3 $&$ 9 $&$ 0 $&$ 0 $&$ $?$? \circ तोई हस्तक्षेप नहीं \mathcal{U} 0 तरह से हस्तक्षेप a) मनोदशा \circ ? 2 3 $&$ 9 $&$ 0 $&$ 0 $&$? ? \circ तोई हस्तक्षेप नहीं \mathcal{U} 0 तरह से हस्तक्षेप a) चलने की क्षमता \circ ? 2 3 $&$ 9 $&$ 0 $&$ 0 $&$? ? \circ a) चलने की क्षमता \circ ? 2 3 $&$ 9 $&$ 0 $&$ 0 $&$? ? \circ a) चलने की क्षमता \circ ? 2 3 $&$ 9 $&$ 0 $&$ 0 $&$? ? \circ a) चलने की क्षमता \circ ? 2 3 $&$ 9 $&$ 0 $&$ 0 $&$? ? \circ a) चलने की क्षमता \circ ? 2 3 $&$ 9 $&$ 0 $&$ 0 $&$? ? \circ a ? 2 3 $&$ 9 $&$ 0 $&$ 0 $&$? ? \circ a ? 2 3 $&$ 9 $&$ 0 $&$ 0 $&$? ? \circ a ? 2 3 $&$ 9 $&$ 0 $&$ 0 $&$? ? \circ a ? ? 2 3 $&$ 9 $&$ 0 $&$ 0 $&$? ? \circ a ? ? 2 3 $&$ 9 $&$ 0 $&$ 0 $&$? ? \circ a ? ? 2 3 $&$ 9 $&$ 0 $&$ 0 $&$? ? \circ a ? ? 2 3 $&$ 9 $&$ 0 $&$ 0 $&$? ? \circ a ? ? \cdot ? ? \circ a ? ? \cdot ? \cdot ? ? \circ a ? ? \cdot ? ? \circ ? ? ? \circ a ? ? \cdot ? ? \cdot ? ? \cdot ? ? \cdot ? ? ? \cdot a ? ? ? \cdot ? ? ? \cdot ? ? ? \cdot ? ? ? ? ? ? ? ? ? ? ? ? ? ? ? ? ? ? ?		जो अंक पिछले ह उस अंक को कृप	रफ्ते होनेवा या गोल की	ले दर्द क ोजिये.	गे वजहसे	। आपवे	काम	में हस्त	क्षेप का वर्णन करता है	
\circ r		अ) साधारण हलग	प्रल							
a) मनोदशा o १ २ 3 ४ ५ ६ ७ ८ ९ १० कोई हस्तकोप नहीं a) चलने की क्षमता o १ २ 3 ४ ५ ६ ७ ८ ९ १० a) चलने की क्षमता o १ २ 3 ४ ५ ६ ७ ८ ९ १० aोई हस्तकोप नहीं s) सामान्य काम (घर के और बाहर के) o १ २ 3 ४ ५ ६ ७ ८ ९ १० aìई हस्तकोप नहीं g, सुनरे लोगोंके साथ संबंध o १ २ 3 ४ ५ ६ ७ ८ ९ १० aìई हस्तकोप नहीं aìई हस्तकोप नहीं vy, तरह से हस्तकोप aìई हस्तकोप नहीं aìई हस्तकोप नहीं vy, तरह से हस्तकोप aìई हस्तकोप नहीं aì क्रियनहीं aì क्रियनहीं<		० १ २ कोई हस्तक्षेप नह	३ মী	8	ч	Ę	ھا	٢	९ १० पूरी तरह से हस्तक्षेप	
० १ २ 3 ४ ५ ६ ७ ८ ९ १० कोई हस्तक्षेप नहीं एग्रे तरह से हस्तक्षेप ० १ २ 3 ४ ५ ६ ७ ८ ९ १० कोई हस्तक्षेप नहीं ९ १ २ 3 ४ ५ ६ ७ ८ ९ १० कोई हस्तक्षेप नहीं ९ १ २ 3 ४ ५ ६ ७ ८ ९ १० कोई हस्तक्षेप नहीं ९ १ २ 3 ४ ५ ६ ७ ८ ९ १० कोई हस्तक्षेप नहीं ९ १ २ 3 ४ ५ ६ ७ ८ ९ १० कोई हस्तक्षेप नहीं ७ १ २ 3 ४ ५ ६ ७ ८ ९ १० कोई हस्तक्षेप नहीं ७ १ २ 3 ४ ५ ६ ७ ८ ९ १० कोई हस्तक्षेप नहीं ७ १ २ 3 ४ ५ ६ ७ ८ ९ १० कोई हस्तक्षेप नहीं ७ १ २ 3 ४ ५ ६ ७ ८ ९ १० कोई हस्तक्षेप नहीं ७ १ २ 3 ४ ५ ६ ७ ८ ९ १० कोई हस्तक्षेप नहीं ७ १ २ 3 ४ ५ ६ ७ ८ ९ १० कोई हस्तक्षेप नहीं ७ १ २ 3 ४ ५ ६ ७ ८ ९ १० कोई हस्तक्षेप नहीं ७ १ २ 3 ४ ५ ६ ७ ८ ९ १० कोई हस्तक्षेप नहीं ७ १ २ 3 ४ ५ ६ ७ ८ ९ १० कोई हस्तक्षेप नहीं		ब) मनोदशा								
क) चलने की क्षमता ० १ २ 3 ४ ५ ६ ७ ८ ९ १० कोई हस्तक्षेप नहीं इ) सामान्य काम (घर के और बाहर के) ० १ २ 3 ४ ५ ६ ७ ८ ९ १० कोई हस्तक्षेप नहीं १ दुसरे लोगोंके साथ संबंध ० १ २ 3 ४ ५ ६ ७ ८ ९ १० कोई हस्तक्षेप नहीं ० १ २ 3 ४ ५ ६ ७ ८ ९ १० कोई हस्तक्षेप नहीं ० १ २ 3 ४ ५ ६ ७ ८ ९ १० कोई हस्तक्षेप नहीं ७) जीद ० १ २ 3 ४ ५ ६ ७ ८ ९ १० कोई हस्तक्षेप नहीं ० १ २ 3 ४ ५ ६ ७ ८ ९ १० कोई हस्तक्षेप नहीं ० १ २ 3 ४ ५ ६ ७ ८ ९ १० ग) जिन्दगी से आनंद ० १ २ 3 ४ ५ ६ ७ ८ ९ १० कोई हस्तक्षेप नहीं ० १ २ 3 ४ ५ ६ ७ ८ ९ १० कोई हस्तक्षेप नहीं ० १ २ 3 ४ ५ ६ ७ ८ ९ १० कोई हस्तक्षेप नहीं ० १ २ 3 ४ ५ ६ ७ ८ ९ १० कोई हस्तक्षेप नहीं ० १ २ 3 ४ ५ ६ ७ ८ ९ १० कोई हस्तक्षेप नहीं ० १ २ ३ ४ ५ ६ ७ ८ ९ १० कोई हस्तक्षेप नहीं	1	० १ २ कोई हस्तक्षेप नह	३ दी	8	ц	ξ	ال	د	९ १० पूरी तरह से हस्तक्षेप	~-
• १ २ 3 ४ ५ ६ ७ ८ ९ १० कोई हस्तक्षेप नहीं पूरी तरह से हस्तक्षेप ड) सामान्य काम (घर के और बाहर के) • १ २ 3 ४ ५ ६ ७ ८ ९ १० कोई हस्तक्षेप नहीं पूरी तरह से हस्तक्षेप इ) दुसरे लोगोंके साथ संबंध • १ २ 3 ४ ५ ६ ७ ८ ९ १० कोई हस्तक्षेप नहीं पूरी तरह से हस्तक्षेप • १ २ 3 ४ ५ ६ ७ ८ ९ १० कोई हस्तक्षेप नहीं पूरी तरह से हस्तक्षेप ७) नींद पूरी तरह से हस्तक्षेप • १ २ 3 ४ ५ ६ ७ ८ ९ १० कोई हस्तक्षेप नहीं पूरी तरह से हस्तक्षेप ग) जिन्दगी से आनंद • १ २ 3 ४ ५ ६ ७ ८ ९ १० कोई हस्तक्षेप नहीं • १ २ 3 ४ ५ ६ ७ ८ ९ १० कोई हस्तक्षेप नहीं • १ २ 3 ४ ५ ६ ७ ८ ९ १० कोई हस्तक्षेप नहीं		क) चलने की क्ष	मता							
5) सामान्य काम (घर के और बाहर के) ० १ २ 3 ४ ५ ६ ७ ८ ९ १० कोई हस्तक्षेप नहीं २) दुसरे लोगोंके साथ संबंध ० १ २ 3 ४ ५ ६ ७ ८ ९ १० कोई हस्तक्षेप नहीं ५) नींद ० १ २ 3 ४ ५ ६ ७ ८ ९ १० कोई हस्तक्षेप नहीं ५) नींद ० १ २ 3 ४ ५ ६ ७ ८ ९ १० कोई हस्तक्षेप नहीं ५) जींद ० १ २ 3 ४ ५ ६ ७ ८ ९ १० कोई हस्तक्षेप नहीं ५ १ २ ३ ४ ५ ६ ७ ८ ९ १० कोई हस्तक्षेप नहीं ५ १ २ ३ ४ ५ ६ ७ ८ ९ १० कोई हस्तक्षेप नहीं ५ १ २ ३ ४ ५ ६ ७ ८ ९ १० कोई हस्तक्षेप नहीं ५ १ २ ३ ४ ५ ६ ७ ८ ९ १० कोई हस्तक्षेप नहीं ५ १ २ ३ ४ ५ ६ ७ ८ ९ १० कोई हस्तक्षेप नहीं ५ १ २ ३ ४ ५ ६ ७ ८ ९ १० कोई हस्तक्षेप नहीं ५ १ २ ३ ४ ५ ६ ७ ८ ९ १० कोई हस्तक्षेप नहीं ५ १ २ ३ ४ ५ ६ ७ ८ ९ १० कोई हस्तक्षेप नहीं ५ १ २ ३ ४ ५ ६ ७ ८ ९ १० कोई हस्तक्षेप नहीं		० १ २ कोई हस्तक्षेप ना	३ ही	8	ц	Ę	ھا	٢	९ १० पूरी तरह से हस्तक्षेप	
 १२३४५५६७८९१० कोई हस्तक्षेप नहीं दुसरे लोगोंके साथ संबंध १२३४५५६७८९१० कोई हस्तक्षेप नहीं १२३४५६७८९१० पूरी तरह से हस्तक्षेप ५१२३४५५६७८९१० पूरी तरह से हस्तक्षेप ग) जिन्दगी से आनंद १२३४५६७८९१० पूरी तरह से हस्तक्षेप 		ड) सामान्य काम	न (घर के उ	और बाहर	क)					
 इ) दुसरे लोगोंके साथ संबंध १२३४५६७८९१० कोई हस्तक्षेप नहीं प्री तरह से हस्तक्षेप फ) नींद १२३४५६७८९१० कोई हस्तक्षेप नहीं प्री तरह से हस्तक्षेप ग) जिन्दगी से आनंद १२३४५६७८९१० प्री तरह से हस्तक्षेप नहीं प्री तरह से हस्तक्षेप 		० १ २ कोई हस्तक्षेप न	३ ही	8	y	ξ	ھا	٢	९ १० पूरी तरह से हस्तक्षेप	
 १२३४७६७८९१० कोई हस्तक्षेप नहीं ९१२३४७६७८९१० पूरी तरह से हस्तक्षेप ७१२३४७६७८९१० कोई हस्तक्षेप नहीं ग) जिन्दगी से आनंद १२३४७६७८९१० पूरी तरह से हस्तक्षेप 		इ) दुसरे लोगोंके	साथ संबंध	T						\bigcirc
फ) नींद • १ २ ३ ४ ५ ६ ७ ८ ९ १० कोई हस्तक्षेप नहीं ग) ज़िन्दगी से आनंद • १ २ ३ ४ ५ ६ ७ ८ ९ १० कोई हस्तक्षेप नहीं पूरी तरह से हस्तक्षेप		० १ २ कोई हस्तक्षेप न	३ ही	ጸ	ц	Ę	وا	٢	९ १० पूरी तरह से हस्तक्षेप	
 १२३४५६७८९१° कोई हस्तक्षेप नहीं ग) जिन्दगी से आनंद १२३४५६७८९१° १२३४५६७८९१° कोई हस्तक्षेप नहीं 		फ) नींद								
ग) ज़िन्दगी से आनंद ० १ २ ३ ४ ५ ६ ७ ८ ९ १० कोई हस्तक्षेप नहीं पूरी तरह से हस्तक्षेप		० १ २ कोई हस्तक्षेप न	३ हीं	8	ц	ξ	ს	۷	९ १० पूरी तरह से हस्तक्षेप	
० १ २ ३ ४ ५ ६ ७ ८ ९ १० कोई हस्तक्षेप नहीं पूरी तरह से हस्तक्षेप		ग) ज़िन्दगी से	आनंद						1 4	
		० १ २ कोई हस्तक्षेप न	३ ही	8	ч	દ્	فا	٢	९ १० पूरी तरह से हस्तक्षेप	
3 2	A									
-------	---	---								
	वेदनेचे तीव्रता मूल्यमापन (Brief pain inventory)									
	आकृतीमध्ये ज्या जागी तुम्हाला वेदना होत असतील त्या जागा भरा. सर्वात जास्त जिथे वेदना होत असतील तिथे 'X' अशी खुण करा.									
	The loss									
Es. 1	१) गेल्या आठवड्यातील तुमच्या सर्वात असह्य वेदनांना गुणमान देण्यासाठी, कृपया खालील एका आंकड्याला गोल करा.									
	०१२३४५६७८९१०									
	वेदना नाहीत असह्य वेदना									
	२) तुमच्या मध्यम स्वरूपाच्या वेदनांना गुणमान देण्यासाठी, कृपया खालील एका आकड्याला गोल करा.									
	०१२३४७६७८९१०									
	वेदना नाहीत असह्य वेदना									
	३) आत्ता ह्या क्षणाला तुम्हांला होणाऱ्या वेदनांना गुणमान देण्यासाठी, कृपया खालील एका आकड्याला गोल करा.									
	०१२३४५६७८९१०									
	वेदना नाहीत असह्य वेदना									
	Continue	2								

Appendix 36. Brief Pain Inventory Questionnaire – Marathi version

हस्तक्षेप मोजमापन

h.,

जो आकडा, गेल्या आठवड्यातील तुमच्या वेदनांनामुळे तुमच्या कामांमध्ये अडथळा निर्माण झाला हे सांगतो कृपया त्या आकड्याला गोल कराः

भ)	मर्वमाधारण	कामकाज

	0	8	ર	3	8	ц	Ę	یا	٢	९	१०
	बाध	ग नाही								संपूर्ण ब	गाधा
	ब) वृत्त ०	ती १	ર	3	۶	ц	Ę	ឲ	۷	९	१०
r.	बाध	या नाही								संपूर्ण व	मधा
	क) च	ालण्याच	वी क्षम	ता							
	٥	۶	२	ş	8	ц	٤	ს	٢	९	१०
	बाध	धा नाही								संपूर्ण व	बाधा
	ड) रो	जचे का	म (घर	तबाहेरीत	न आणि	घराती	ल कामे)			
	٥	۶	२	3	8	ц	ξ	وا	٢	९	१०
	बाध	धा नाही								संपूर्ण व	बाधा
	ई) इत	तर लोव	नंबरोब	रचे संबं	ध		17				
-	0	۶	२	3	8	ц	Ę	فا	٢	९	१०
	बाष	धा नाही	ſ							संपूर्ण	बाधा
	फ) इ	नोप									
		ه ه	ર	ş	8	ц	દ્	فا	٢	९	80
	बा	धा नाही	r							संपूर्ण	बाधा
	ग) 3	नायुष्यात	नील अ	ानंद						1.1	
	0	8	R	3	8	y,	ξ	ھا	٢	٩	80
	बा	धा नाह	T							संपूर्ण	बाधा

General Health Questionnaire	(GHQ-	12)				
We would like to know how your health has been in general	l, over the	past fe	w week	s.		
Please answer the following questions by circling the number that best applies to you.						
Have you recently,						
	much less than usual	same as usual	more than usual	much more than usual		
Been able to concentrate on whatever you are doing?	C	0	0	0		
Lost much sleep over worry?	o	0	c	с		
Felt that you were playing a useful part in things?	C	0	0	Ċ.		
Felt capable of making decisions about things?	C	D	0	C		
Felt constantly under strain?	C	D	0	С		
Felt that you couldn't overcome your difficulties?	c	Ö	0	С		
Been able to enjoy your normal day-to-day activities?	C	D	0	Ċ.		
Been able to face up to your problems?	С	D	0	c		
Been feeling unhappy and depressed?	C	D	0	C.		
Been losing self-confidence in yourself?	С	D	0	C		
Been thinking of yourself as a worthless person?	C	0	0	Ċ.		
Been feeling reasonably happy, all things considered?	С	0	0	С		

Appendix 37. General Health Questionnaire (GHQ-12)

1					
		•			
	सामान्य सेहत फ़ार्म . (General Health Questionnaire -	12)			
	हम ये जानना चाहते हैं की आपकी सेहत पिछले हफ़्तों में व	कैसी रही	أ ?		
	कृपथा नीचे लिखे प्रश्नों का उत्तर आप पर जो अंक लागू होत	ना है उसे	पर गोला	लगाइये.	
	क्या आपते हाल ही में				
	श) क्या आप जो काम कर रहे है उसपे ध्यान दे पाते है?	सम्बन्ध का स	नान्त्र पे ससन	समान्य से ज्याचा	सामान्य से बहुत त्यादा 🔲
<u>_</u>	२) आप चिंता के कारण सो नहीं पाते?				
	३) अप को लगा की आपकी भूमिका काम मेंं महत्वपूर्ण	<u>*</u> ?□			
	४) आप को लगा की आप चीजोंके बारेमें निर्णय ले सकते है	?			
	५).आप को लगा की आप हमेशा तनाव के नीचे है?				□ .
	E) आप को लगा की आप अपनी कठिनाईयों पर काबू नहीं कर पा रहे हैं?		Ċ		
	 अ वया आप अपने दिनभर के काम में आनंद ले पाते है? 				
	८) क्या आप अपने समस्या का सामना कर पाते हैं?				
$\langle \hat{c} \rangle$	९) क्या आप अप्रसन्न या खिल्न महसूस कर रहे है?				
	१०) क्या आप अपना आत्मविश्वास खो रहे है?				
	११) क्या आप अपने आप को मूल्यहीन समझते है?				
	१२) रॉब चीजों को ध्यान में रखते हुए, क्या आप खुश है?				Ξ
			1 4		
	· ·				

Appendix 38. General Health Questionnaire – Hindi version

	सामान्य आरोग्य प्रश्नावनी (Questionnaire)(GHQ -१२)						
	गेल्या काही आठवड्यांमध्ये तुनचे आरोग्य करो होते हे अम्हां खातील प्रश्न्नातला जो आकडा तुम्हांला तुमच्यासाठी वरोबर वा	ता जाणून टतो त्या	ध्यावयात् आकड्याल	वे आहे. ग वर्तुळ व	करा.		
	हल्सीच्या काळात						
	202	वेश गुरु बझे	टरेसीपरामेन्।	नेहेगोंपेसा अध्य	ে ইইমার্টলা অ্ব	अधिक	
	१) तुम्हीं जे काही करत आहात त्या कृतीवर तुम्हीं तुमचे लक्ष केंद्रित करू शकलात का ?						
	२) चिंतेमुळे तुशची झौप खूप कमी ज़ाली का ?						
	३) गोष्टींमध्ये तुम्हीं उपयोगी भूमिका पार पाडत आहात ?						
	४) विविध गोष्ट्रीबद्दल तुम्ही निर्णय घेऊ शकाल असे तुम्हाला वाटते का?						
	. १) कायम मानसिक तणावाखाली अंसल्यासारखे घाटले का ?				,		
	६) तुमच्या अडचणींमधून तुम्हीं बाहेर पडू शकता नाही आहार असे तुम्हांना वाटले काय ?	T					
-	७) दैनंदिन व्यवसरामध्ये तुम्हीं आनंद घेऊ शकलात का ?						
<u>_</u>	८) तुम्हीं तुमच्या अडचणींना तॉड देऊ शकलात का ?						
	१) तुम्हांला दुःखी व खिल्ल वाटले का ?						
	१०) आत्मविश्वास कमी होत चालला आहे असे तुम्हांला वाटले का ?						
	११) आपण तुच्छ आहोत असे तुम्हांला वाटले का ?						
	१२) सर्व बब्तींवा विचार करता तुम्हांला बर् या पैकी आनंधी वाटले का ?						
	·						

Appendix 39. General Health Questionnaire – Marathi version

Appendix E

DFNS-QST Protocol



Appendix 40. Verbal instructions for performing QST-DFNS protocol

General instructions for the volunteers/patients

"In the following tests, we will explore, using various procedures, how you perceive temperature changes as well as touch and vibration stimuli. In addition, we will examine, from what point on different test stimuli are felt as being painful. The results of these tests are compared to normative data gained from healthy volunteers. To make this possible, these tests are always performed in the same manner. To ensure this, among others, the test instructions will be read to you aloud.

If you have not understood the test instructions, please always feel free to immediately ask for clarification."

a.) Thermal detection thresholds and thermal pain thresholds

Demonstration of the test procedure

"The device placed on your skin is able to either warm or cool the skin. In addition you are given a stop button that enables you to immediately stop the ongoing test stimulus at any time. For every test I will explain to you when to use the stop button.

Please tell me, whether the device on your skin feels warm, cool or neutral.

Please do not look at the computer screen during the test procedures."

- CDT "First we will test your ability to perceive cold sensations. Please press the stop button immediately once you perceive a change in temperature to cool/cooler for the first time. Subsequently, the thermode will warm up again, until it reaches the baseline temperature. This procedure will start in a few seconds."
- WDT "Now we will test your ability to perceive warm sensations. Please press the stop button immediately once you perceive a change in temperature to warm/warmer for the first time. Subsequently, the thermode will cool down again, until it reaches the baseline temperature. This procedure will start in a few seconds."
- TSL "Now we will test how well you are able to discern between successive temperature changes. Please press the stop button immediately once you feel a temperature change towards "warm" or "cold" sensations, and tell us explicitly whether you felt the temperature change as "warm" or "cold". It may well be that some of the temperature changes are felt as "hot" or "painfully hot"."
- CPT "Now we will test as to when you perceive the cooling of the thermode as painful. Your skin will be slowly cooled. At some point in time you will feel a second sensation on top of the usual "cold" sensation. The impression of "cold" will change its quality towards an additional impression of a "burning", "stinging", "drilling" or "aching" sensation. Please press the stop button immediately once you perceive such a change. Please DO NOT wait to press the stop button until the sensation has become unbearably painful. Subsequently, the thermode will warm up again, until it reaches the baseline temperature. This procedure will start in a few seconds."
- HPT "Now we will test as to when you perceive the warming of the thermode as painful. Your skin will be slowly warmed. At some point in time you will feel a second sensation on top of the usual "warm" or "hot" sensation. The impression of "warmth" or "heat" will change its quality towards an additional impression of a "burning", "stinging", "drilling" or "aching" sensation. Please press the stop button immediately once you perceive such a change. Please DO NOT wait to press the stop button until the sensation has become unbearably painful. Subsequently, the thermode will cool down again, until it reaches the baseline temperature. This procedure will start in a few seconds."

QST instructions according to the protocol of the German Research Network on Neuropathic Pain (DFNS) Version 2.1 - 08.07.2010 ^o Chair of Neurophysiology, University Medicine Mannheim; Heidelberg University

Testing within the control and test site

Please tell me, whether the device on your skin feels warm, cool or neutral."

CDT "Just as we have done in the practice round, you will first perceive a cooling of the skin. Please press the stop-button immediately as soon as you first feel a change of temperature to "cool or cooler".

This procedure will be performed a total of 3 times."

- WDT "Please press the stop-button immediately as soon as you first perceive a warming of the skin. Again this procedure will be performed a total of 3 times."
- TSL "Please press the stop-button immediately as soon as you first perceive any kind of temperature change. Please state, whether the sensation was "cold", "warm", "hot" or "painfully hot". This procedure will be performed a total of 6 times."
- CPT "Please press the stop-button immediately as soon as the "cold" sensation changes its quality to an additional sensation of "burning", "stinging", "drilling" or "aching". This procedure will be performed a total of 3 times."
- HPT "Please press the stop-button immediately as soon as the "warm" or "hot" sensation changes its quality to an additional sensation of "burning", "stinging", "drilling" or "aching". This procedure will be performed a total of 3 times."

b.) Tactile detection thresholds

Demonstration of the test procedure

MDT "This is a test of your ability to detect light touch. Please do not look at the skin area we are testing at any time during the test procedures. I will now touch your skin with these thin hairs. Please say "yes" as soon as you perceive a touch sensation"

Testing within the control or test site

MDT "Please say "yes" as soon as you perceive a touch sensation."

QST instructions according to the protocol of the German Research Network on Neuropathic Pain (DFNS) Version 2.1 - 08.07.2010 ^o Chair of Neurophysiology, University Medicine Mannheim; Heidelberg University

c.) Mechanical pain threshold

Familiarization – Demonstration of the test procedure

MPT "This is a test of your ability to perceive "sharp", "pricking" or "stinging" stimuli. Various fine metal rods that are able to exert different pressure intensities will be carefully placed onto your skin.

Please say "sharp" as soon as you no longer perceive only a touching sensation on your skin, but experience an additional "sharp", "pricking" or "stinging" sensation.

Please say "blunt" when you only feel a touching sensation.

Under no circumstances, you should look at the skin area we are testing during the test procedure!"

Testing within the control or test site

- MPT Please say "sharp" as soon as you not only perceive a touching sensation on your skin, but an additional "sharp", "pricking" or "stinging" sensation. Please say "blunt" when you only perceive a touching sensation.
- d.) S/R-(Stimulus/Response) functions: Mechanical pain sensitivity (MPS) and dynamic mechanical allodynia (DMA)

Familiarization – Demonstration of the test procedure

- MPS "As in the test before, blunt fine metal rods will be carefully pressed against your skin with varying pressure.
- DMA In between these punctual stimuli your skin will occasionally be touched by a cotton wisp, a Q-tip, and a brush. Some of these stimuli may be accompanied by a "sharp", "pricking", "stinging" or "burning" sensation. Other stimuli may only be perceived as a touching sensation, others may not be perceived at all.
 - Please rate the painfulness of each stimulus by giving a number between "0" and "100". Any "sharp", "pricking", "stinging" or "burning" sensation should be defined as being painful and given a rating value above "0". You may also use decimals.

"0" meaning: No pain, no "sharp", "pricking", "stinging" or "burning" sensation. "100" meaning: Most intense pain sensation imaginable.

Under no circumstances, you should look at the skin area we are testing during the test procedure!

(Should the subject/patient give a rating of "100", please ask:

Are you sure that this was the most intense pain sensation imaginable for you?")

Testing within the control or test site

MPS "Again, please rate the painfulness of each stimulus by giving a number between "0" and "100".

DMA "0" meaning: No pain, no "sharp", "pricking", "stinging" or "burning" sensation. "100" meaning: Most intense pain sensation imaginable."

(Should the subject/patient give a rating of "100", again please ask:

"Are you sure that this was the most intense pain sensation imaginable for you?"

If applicable: "Then we will no longer use this or any other of the more severe stimuli.")

QST instructions according to the protocol of the German Research Network on Neuropathic Pain (DFNS) Version 2.1 - 08.07.2010 ^o Chair of Neurophysiology, University Medicine Mannheim; Heidelberg University

e.) Wind-up ratio

Familiarization – Demonstration of the test procedure

WUR

"Like in the former test I will now press a single fine metal rod against your skin. Please rate the painfulness of this single stimulus by giving a number between "0" and "100" Any "sharp", "pricking", "stinging" or "burning" sensation should be defined as being painful and given a rating value above "0" You may also use decimals.

"0" meaning: No pain, no "sharp", "pricking", "stinging" or "burning" sensation. "100" meaning: Most intense pain sensation imaginable.

Now I will apply a series of 10 stimulations with the same metal rod at 1 second intervals on your skin. Once the entire series is over, please rate its average painfulness by giving a number between "0" and "100".

"0" again meaning: No pain, no "sharp", "pricking", "stinging" or "burning" sensation. "100" meaning: Most intense pain sensation imaginable."

Testing within the control or test site

WUR "The whole procedure of applying one single stimulus followed by the stimulus series will be carried out within this area and will be repeated 5 times. Please rate again how painful this single stimulus was on a scale of 0 to 100.

"0" meaning: No pain, no "sharp", "pricking", "stinging" or "burning" sensation. "100" meaning: Most intense pain sensation imaginable.

Please rate again how painful the stimulus series was on a scale of 0 to 100."

f.) Vibration detection threshold

Familiarization - Demonstration of the test procedure

VDT "This procedure tests your ability to perceive "vibrations". I will now place this vibrating tuning fork on your skin. Please tell me if you are able to feel the vibrations?

Please immediately say "NOW" as soon as you are no longer able to feel any further vibrations!

You may hear the vibrations of the tuning fork as a sound. Please try to pay attention only to the vibration, and not to the sound. \H

Testing within the control and test site

VDT "A series of 3 consecutive tests will now be carried out within the control and the test site. I will now place this vibrating tuning fork on your skin. Please tell me if you are able to feel the vibrations?

Please immediately say "NOW" as soon as you are no longer able to feel any further vibrations!"

QST instructions according to the protocol of the German Research Network on Neuropathic Pain (DFNS) Version 2.1 - 08.07.2010 ^o Chair of Neurophysiology, University Medicine Mannheim; Heidelberg University

g.) Pressure pain threshold

Familiarization - Demonstration of the test procedure

PPT "This procedure tests your ability to feel pressure pain above muscles.

I will press this pressure measuring device against one of your muscles. Please immediately say "NOW" as soon as the usual sensation of pressure changes towards an additional sensation of "burning", "stinging", "drilling" or "aching"."

Testing within the control and test sites

PPT "Again, I will press this pressure measuring device against one of your muscles. Please immediately say "NOW" as soon as the usual sensation of pressure changes towards an additional sensation of "burning", "stinging", "drilling" or "aching"."

This procedure will be carried out a total of 3 times."

QST instructions according to the protocol of the German Research Network on Neuropathic Pain (DFNS) Version 2.1 - 08.07.2010 ^o Chair of Neurophysiology, University Medicine Mannheim; Heidelberg University

Appendix 41. QST-DFNS protocol – Hindi version

bbe	endix
ene	eral instructions for the volunteers/patients:
मगरे ज्पन देना एरग	ने परिक्षण में हम विविध प्रक्रिया इस्तेमाल करके ये जॉच लेंगे कि आप तापमान में होनेवाले यदल. स्पर्श और उतेजन को किस प्रकार समझते हैं. उसी तरह हम ये जॉच लेंगे कि किस स्तर से आपको विविध परिक्षण उत्तेजन क महसूस होते हैं. इस परिक्षण के अंतिम परिणाम, निरोगी स्वयंसेवकों से इकट्ठा की हुई जानकारी से तुलना की 1. ये मुमकिन करने के लिए ये परिक्षण हमेशा एक ही प्रकार में किया जाएगा. ये सुनिश्वित करने के लिए ये सूचना हो पढकर समझायी जाएगी.
गर	आपको परिक्षण की सूचना समझ में नहीं आप है तो कृपया आप तुरंत या कभी भी स्पष्टीकरण माँग सकते हैं."
l) T	hermal detection thresholds and thermal pain thresholds:
Xem	constration of the test procedure:
आप	की त्वचा पर रखा गया यंत्र, उष्ण या ठंडा होता है. इस यंत्र के साथ आपको एक बंद करने का बटन दिया गया है जो
नाप नि	किसी भी वक्त परिक्षण उतेजन सोकने के लिए इस्तेमाल कर सकते है. मैं आपको समझाऊंगी कि हर परिक्षण के
नेए ;	आपको कब ये बंद करने का बटन इस्तेमाल करना है.
हुपग	। मुझे बताईये कि आपकी त्वचा पर रखा गया यंत्र आपको उष्ण, ठंडा या निष्क्रिय लगता है.
हुपश	ा परिक्षण प्रक्रिया शुरू होने पर कंप्यूटर के परदे को ना देखे."
:DT	- "पहले हम आपकी ठंड संवेदन महसूस करने की क्षमता जॉच लेंगे. कृपया जब आपको-तापमान में पाहिले बार
ज्डा/	अधिक ठंडा बदल महसूस हो तब बंद बटन को तुरंत दबाईबे. इसके पश्चात ये यंत्र सामान्य तापमान पर आ जाएगा.
रे प्रहि	केया कुछ रोकंडों में शुरू होगी."
মিটা	' - "अव हम आपकी उष्ण संवेदन महसूस करने की क्षगता जॉच लेगे. कृपया जब आपको तापमान में पाहिले वार
মেটা	/अधिक उष्ण बदल महसूस हो तब बंद बटन को तुरंत दवाईथे. पश्चात ये यंत्र सामान्य तापमान पर आ जाएगा. ये
মুক্টিম	या कुछ सेकंडों में शुरू होगी."
CSL	- "तापमान में होनेवाले बदल आप कितनी अच्छी तरह से परख सकते है थे हम अब देखेंगे. कृपया जब आपको
राष∓	गन में 'उष्ण' या 'ठण्ड' बदल महसूस हो तव बंद बटन तुरंत दबाईये और हमें स्पष्ट वताईये कि तापमान में 'उष्ण' या
ठण्ड	'बदल महसूस हुआ. कभी कभी आपको ये तापमान गरम या दर्दनाक गरम महसूस हो सकता है."
CPT	- "अब हम ठंडा स्पर्श कब दर्दलाक होता है इसका परिक्षण करेंगे. आपकी त्वचा धीरे धीरे ठंडी की जाएगी. एक
समर	1 ऐता आएगा जब साधारण ठंड संवेदन के ऊपर और संवेदन होगा. ये ठंडा संवेदन बदलकर इसके ऊपर आपकी
जलव	ने, इंक, बेधनीय या वेदना जैसे लगेगा. जब आपको ये बदल महसूस होगा तब कृपया तुरंत बंद बदन दबाईथे.
कृषय	11 ये संवेदन असहतीय दर्दनाक होने तक ना रुके और तुरंत बंद बटन दबाईथे. पश्चात ये युव सामान्य तापमान पर
मार्थ	'- "अब हम उष्ण स्पर्श कब दर्दनाक होता है इसका परिक्षण करेंगे. आपकी त्वचा धीरे धीरे गरम की जाएगी. एक
समय	1 रेसा आएगा जब साधारण उष्ण संवेदन के ऊपर और संवेदन होगी. ये उष्ण संवेदन बदलकर इसके ऊपर आपको
जन्म	ने, डंक, बेधनीय या वेदना जैसे लगेगा. जब आपको ये बदल महसूस होगा तब कृपया तुरंत बद बटन दबाईये.
कृषर	11 ये संवेदना असहतीय दर्दनाक होते तक ना रुके और तुरंत बंद बटन दबाईये. पश्चात ये यंत्र सामात्य तापमान पर
आ उ	गएगा. ये प्रक्रिया कुछ सेकंडों में शुरु होगी."
	8

Tosting within the control and test site:

"कृपया मुझे बताईये कि आपकी त्वचा पर रखा गया यंत्र आपको उष्ण, ठंडा या निष्पक्ष लगता है."

CDT - "जैसे हमने पहले किया है, वैसे आपकी त्वचा पर शुरू में ठंडा स्पर्श होगा. कृपया जब आपको पहले तापमान में ठंडा/अधिक ठंडा बदल महसूस हो तब बंद बटन को तुरंत दबाइंये. ये प्रक्रिया तीन बार की जाएगी."

WDT - "कृपचा जब आपको पहले तापमाने में उष्ण/अधिक उष्ण बदल महसूस हो तब बंद बटन को तुरंत दबाईथे. थे प्रक्रिया तीन बार की जाएगी."

TSL - "कृपया जब आपको तापमान में कोई भी वदल महसूस हो तय वंद बटन तुरंत दवाईचे. कृपया हमें स्पष्ट बताईचे कि तापमान में 'उष्ण', 'ठण्ड', 'गरत' या 'दर्दनाक गरम' बदल महसूस हुआ. ये प्रक्रिया ६ वार की जाएगी."

CPT - "कृपया जब 'ठण्ड' संवेदन के ऊपर आपको जलने, डंक, वैधनीय या वेदना जैसे लगते लगता है तब वंद बटन तुरंत दबाईये. ये प्रक्रिया तीन बार की जाएगी."

HPT - "कृपया जब 'उष्ण' या 'गरम' संवेदन के ऊपर आपको जलने, डंक, वेधनीय या वेदना जैसे लगने लगता है तब बंद बटन तुरंत दवाईये, ये प्रक्रिया तीन बार की जाएगी."

) Tactile detection thresholds: Demonstration of the test procedure:

MDT - "ये परिक्षण आपका हलका स्पर्श जानने की क्षमता के लिए है. कृपया त्वचा के जिस हिस्से पर हम जॉच कर रहे हैं, उत्त हिस्से को परिक्षण के समय ना देखे. मैं अब आपको इन पतले केश से स्पर्श करुंगी. कृपया 'हाँ' बोलिए जिस समय आपको इस स्पर्श की अभिजता होती है."

Testing within the control or test site:

MDT - "जिस समय आपको इस स्पर्श की अभिज्ञता होती है उस समय कृपया 'हाँ' बोलिए ."

C) Mechanical pain threshold: Familiarization - Demonstration of the test procedure:

MPT - "थे परिक्षण आपकी तीच्र, चुभला, और इंक के उत्तेजन को समझने की है. तरह तरह धातू के छोटे भिन आपकी िवचा पर सावधानी से विविध दवाव मात्र में रखे जाएंगे. कृपया 'शार्प' ऐसे बोलिए जब आपको त्वचा पर मामूली स्पर्श महसूस होनेके अतिरिक्त तीच्र, चुभला, और इंक जैसे महसूस होता है.

जब आपको सिर्फ मामूली स्पर्श महसूस होता है तब कृपया 'बलंट' ऐसे कहिये.

किसी भी परिस्थिति में आप के त्वचा के जिस जगह पर परीक्षा हो रही है वहाँ न देखे."

Testing within the control or test site:

MPT - "कृपया 'शार्प' ऐसे बोलिए जब आपको त्वचा पर नामूली स्पर्श महसूस होनेके अतिरिक तीव्र, चुभना, और डंक जैसे महसूस होता है.

कृपया 'ब्लंट' एसे कहिये जब आपको सिर्फ मामूली स्पर्श महसूस होता है."

D) S/R - (Stimulus/Response) Functions: Mechanical pain sensitivity (MPS) and Dynamic mechanical allodynia (DMA): Familiarization - Demonstration of the test procedure: MPS - "पहले परिक्षण के समान, तरह तरह की धातू के छोटे पिन आपके त्वचा पर सावधानी से विविध दबाव मात्र में रखे जाएंगे. DMA - इस परिक्षण के बीच में आपके त्वचा पर रुई, रुई बड (q tip) या ब्रश रखा जाएगा. इस में से कुछ उत्तेजक आपको तीवता, वृशना, जसन और इंक जैसे स्पर्श देंगे. कुछ उतेजक आपको एक हल्का स्पर्श देंगे और कुछ उत्तेजक शूल्य स्पर्श देंगे. कृपया इर उत्तेजक आपको कितना दर्दनाक होता है उसे १ से १०० तक गुणमान दीजिये. किसी भी तीव्रता, चुभन', जलन और डंक जैसे स्पर्श को दर्दनाक कहकर शूल्य के ऊपर गुणमान दीजिये. आप दशमलय पहलि का इस्तेमाल कर सकते है. "॰" का अर्थ दर्दरहित, तोवताहीन, न चुभना, न जलना, और न इंक जैसा स्पर्श होता है. "१००" का अर्थ सबसे अधिक दर्द संवेदन करूपनीथ. किसी भी परिस्थिति में आप जिस जगह पर परीक्षा हो रही है वहाँ न देखे." If the patient/volunteer gives a score of 100, please ask them: "क्या आप लिश्चित है कि ये आपको होनेवाला सबसे अधिक दर्दनाक है?" Testing within the control or test site: MPS - "फिर से, कृपया हर उत्तेजक आपको कितना दर्दनाक होता है उसे १ से १०० तक गुणजान दीजिये. DMA - "॰" का अर्थ दर्दरहित, तीव्रताहील, न चुशाना, ल जलना, और न डंक जैसा स्पर्श होता है. '१००" का अर्थ सबसे अधिक दर्द संवेदन कल्पनीय. If the patient/volunteer gives a score of 100, ploase ask them: "क्या आप निश्चित है कि ये आपको होनेवाला सबसे अधिक दर्द है?" lf applicable: "फिर हम इस तरह कि तीव्र उतेजक का इसतेगाल नहीं करेंगे." -) Wind up ratio: Familiarization - Demonstration of the test procedure: WUR - "पहले परिक्षण की तरह, मैं एक धातु की छोटी पिन आपके त्वचा पर रखूँगी, कृपया ये उत्तेजक आपको कितना दर्दनाक होता है उसे ? से ?०० तक गुपामान दीजिये. किसी भी तीव्रता, चुगाना, जलन और डंक जैसे स्पर्श को दर्दनाक कहकर शून्य के अपर गुणमान दीजिये. आप दशमलव पद्धति का इस्तेमाल कर सकते है . "॰" का अर्थ दर्दरहित, तीव्रताहील, न चुभना, व जलना, और न डंक जैसा स्पर्श होता है. "१००" का अर्थ सबसे अधिक दर्द संवेदन कल्पनीय. अब में एक सेकंड के अंतर पे यही धात की छोटी पिन का उत्तेजक १० बार आपके त्वचा पर स्थूँगी. यह १० उतेजक की किया समास होने पर कृपया इस दर्द को ० से १०० तक मध्यगवर्ती गुणगान दीजिये. "॰" का अर्थ दर्दरहित, तीव्रताहीन, न चुभवा, व जलवा, और व इंक जैसा स्पर्श होता है. "१००" का अर्थ सबसे अधिक दर्द संवेदन कल्पनीय. Continue...

٨	· · · · · · · · · · · · · · · · · · ·
Te:	sting within the control or test site:
WI कृष	UR - "ये पूर्ण एक उत्तेजक की क्रिया और उसके बाद १० उत्तेजक की क्रिया इस भाग में की जाएगी और ५ वार की जाएगी. ग्या इस उत्तेजक आपको कितना दर्दनाक होता है उसे १ से १०० तक गुणमान दीजिये.
"o" "?*	' का अर्थ दर्दरहित, तीव्रताहीन, न चुभना, न जलना, और न इंक जैसा स्पर्श होता है ००" का अर्थ सबसे अधिक दर्द."
कृष	ग्या इस उत्तेजक आपको किलना दर्देनाक होता है उसे १ से १०० तक गुणसान दीजिये.
F) Fa	Vibration detection threshold: miliarization - Demonstration of the test procedure:
VE बत	DT - "ये प्रक्रिया आपकी कंपन मर्यादा समझने के लिए है. अब मैं इस कंपन यंत्र को आपके त्वचा पर रखूँगी. कृपथा गुझे गईए की क्या आपको कंपन महसूस हो रहे हैं?
कृष	पया जिस समय आपको कंपन अहसूस होना बंद होता है तभी तुरंत "अभी" ऐसे कहिथे.
आ	पको शायद कंपन की आवाज सुनाई देगी. कृपया आपका संपूर्ण ध्यान आवाजपर ना रखकर सिर्फ कंपन पर रखिये."
Tr	sting within the control or test site:
vt नुर	OT - "अब ३ परिक्षण एक के बाद एक आपके त्याचा पर किये जायेंगे. अभी मैं इस कंपन यंत्र को आपके त्वाचे पर रखूंगी. कृपथा झे बताईए की क्या आपको कंपन महसूस हो रहे हैं?
कु	पथा जिस समय आपको कंपन महत्त्र होना बंद होता है तभी तुरंत "अभी" ऐसे कड़िये.
G F:) Pressure pain threshold: amiliarization - Demonstration of the test procedure:
P दर क	PT - "ये परिक्षण आएकी नांसपेशी के ऊपर दबाव से दर्द का मापन करेगी. मैं ये दबाव मापन यंत्र आपके एक मांसपेशी पर वाऊँगी. कुपया जिस समय साधारण दबाव जलन, तीवता, बेधनीय, या वेदनामय में बदलता है उस समय तुरंत "अभी" ऐसे दिये.
-	esting within the control or test site:
P ब	PT - "मैं फिर से ये दबाव मापन यंत्र आपके एक मांसपेशी पर दबाऊँगी. कृपया जिस समय साधारण दबाव जलन, तीचता, [धनीय, थ] वेदनामय में बदलता है उस समय तुरंत "अभी" ऐसे कहिये.
ये	क्रिया ३ वार की जाएगी."
	to an
	•

Appendix 42. QST-DFNS protocol - Marathi version

Marathi Version Appendix General instructions for the volunteers/patients: "पुढील परीक्षणांगध्ये, विविध पद्धती वापरून तापमानातील बदल तसेच स्पर्श व कंपन उत्तेजन. तुम्हांला काय प्रकारे जाणवते हे आपण समजून घेऊ. तसेच आपण परीक्षण करू की कुठल्या स्थितीपासून वेगवेगळे उत्तेजक तुम्हांला वेदना देऊ लागतात. ह्या परीक्षणाचे लिकाल लिरोगी स्वयंसेवकांच्या परीक्षणाबरोबर तोलले जातील. हे शक्य होण्यासाठी ह्या परीक्षा जेहेमीच एकाच पढ़तीप्रमाणे केल्या जातील. हे खातरजगा करण्यासाठी परीक्षेच्या सुचना स्पष्ट शब्दांमध्ये तुम्हांला समजवल्या जातील. जर तुम्हांला त्या समजल्या नाहीत तर कृपया मोकळ्या मनाने लगेच आम्हाला विचास कस." A) Thermal detection thresholds and thermal pain thresholds Demonstration of the test: तुंमच्या त्वचेवर ठेवले गेलेले यंत्र हे तुमची त्वचा थंड किंवा उष्ण करेल. त्या यंत्रावरोवरच ही प्रक्रिया, गरज पडल्यास थांबवण्यासाठी तुम्हांला एक बटण दिले गेले आहे. प्रत्येक परीक्षेरााठी, त्या बटणाचा वापर कधी करायचा आहे हे मी तुम्हांला समजावेन. कृपया तुमच्या त्वचेवर ठेवले गेलेले यंत्र तुम्हांला थंड, उष्ण किंवा तटस्थ वाटत आहे. कृपया करण्यूटरच्या पडयाकडे बचू नका." CDT - "पहिल्यांदा आपण तुमची थंड संवेदनाची समर्थतेचे परीक्षण करू. कृपया जेव्हां तुम्हांला तापमानात पहिल्यांदा थंड आणि अती थंड बदल जाणवेल तेव्हा लगेच बंद करण्याचे वटण दावा. पश्यात हे यंत्र पुल्हां साधारण तापमानावर बैईल. ही प्रक्रिया काही सेकंदांमध्ये सुरू होईल." े/DT - "आता आपण तुमची उष्ण संवेदनाची समर्थतेचे परीक्षण करू. कृपया जेव्हां तुम्हांला तापमानत पहिल्यांदा उष्ण आणि अती उष्ण वदल जाणवेल तेव्हा लगेच बंद करण्याचे बटण दाबा. पश्यात हे यंत्र पुन्हा साधारण तापमानावर येईल. ही प्रक्रिया काही सेकंदांमध्ये सुरू होईल." TSL - "ताप्रमानातील बदल तुम्ही किंती चांगल्या प्रकारे ओळखू शकता हे आपण आता बघू, कृपया जेव्हां तुम्हांला तापमानामध्ये थंड किंवा उज्ज बदल जाणचेल, तेव्हां तातडीने बंद बटण दाबा, आणि आम्हांला तापमानातील हा थंड किंवा उष्ण वदल स्पष्टपणे सांगा. कदाचित हे बदल तुम्हाला गरम, किंवा वेदनापूर्ण गरम जाणवतील." Continue...

CPT-"आता आपण थंड बदल कधी वेदलामय होतो, हे तपासुया. तुमची त्वचा हळूहळू शंड करण्यात रेईल. कधीतरी एका क्षणानंतर तो थंड स्पर्श तुम्हांला वेदनामय होऊ लागेल. हा थंड स्पर्श वदलेल आणि तुम्हाला जळजळ, डंख, गिरसिट, किंवा ठणकण्यासारखा जाणवू लागेल. कृपथा हा बदल असहा होईस्तोवर थांबू नका. लगेच बटण दाबा. बटण दावल्यानंतर, ते यत्र पुन्हां साधारण तापसानावर येईल. ही प्रक्रिया काही सेकंदांमध्ये सुरू होईल."

HPT - "आता आपण उष्ण बदल कथी वेदनामय होतो, हे तपासुया. तुमची त्वचा हळूहळू गरम करण्यात येईल. कधीतरी एका क्षणानंतर तो उष्ण स्पर्श तुम्हांला वेदचामय होऊ लागेल. हा उष्ण स्पर्श वदलेल आणि तुम्हाला जळजळ, डख, गिरगिट, किंवा ठणकण्यासारखा जाणवू लागेल. कृपया हा बदल असब होईस्तोवर थांबू जका. लगैच बटण दाबा. बटण दाबल्यानंतर, ते यंत्र पुन्हां साधारण तापगालावर येईल. ही प्रक्रिया काही सेकदामध्ये सुरू होईल."

Testing within the control and test site:

"कृपया तुमच्या त्वचेवर ठेवले गेलेले यंत्र तुम्हांला थंड, उष्ण किंवा तटस्थ वाटत आहे."

CDT - 'जसे आपण आधी वधितले तसे तुम्हांला पहिल्यांदा थंड संवेदना होईल. जेव्हा तुम्हांला तापमानातील बदल थंड किंवा अति थंड वाटतो तेव्हां कृपया लगेच बंद बटण दाबा. ही प्रक्रिया तीन वेळा केली जाईल."

WDT - "जसे आपण आधी बचितले तसे तुम्हांला पहिल्यांदा उष्ण संवेदना होईल. जेव्हा तुम्हांला तापमानातील बदल उष्ण किंवा अति उष्ण वाटतो तेव्हां कृपया लगेच बंद बटण दावा. ही प्रक्रिया तीन वेळा केली जाईल."

TSL - "ज्या क्षणी तापमानामध्ये तुम्हांला बदल जागवू लागेल तेव्हां कृपया लगेच वंद वटण दावा. कृपया ही .म्पवेदना थंड, उध्य, गरम की वेदनापूर्ण गरल होता हे स्पष्ट सांगा. ही प्रक्रिया राहा वेळा केली जाईल."

CPT - "कृपया 'बंद बटण' लगेच दाबा ज्या क्षणी थंड संवेदनाच्या वर तुम्हांला जळजळ, इंख, गिरमिट, किंवा ठणकण्यासारखा जाणवू लागेल. ही प्रक्रिया तीन वेळा केली जाईल."

HPT-'ज्या क्षणी उष्ण संवेदनाच्या वर तुम्हांता जळजळ,डंख,मिरमिट,किंवा ठणकण्यासारखा जाणव् लागेल तेव्हाकृपया बंद बटण लगेच दाबा ही प्रक्रिया तील वेळा केती जाईल."

B) Tactile detection thresholds Demonstration of the test procedure:

MDT - "ही परीक्षा तुमची हलका स्पर्श ओलाखन्याची परीक्षण आहे. कृपया त्वचेचा जो आग तपासला जात आहे तिथे परिक्षेच्या वेळी बधू नका. भी आता ह्या वारीक टाचण्यांनी तुम्हांला स्पर्श करेन. कृपया, ज्यावेळी तुम्हांला तो स्पर्श जाणवेल त्यावेळेस 'हो' असे म्हणा."

Testing within the control and test site:

MDT - "कपया, ज्यावेळी तुग्हांला तो स्पर्श जाणवेल त्यावेळेस 'हो' असे म्हणा."

C) Mechanical pain threshold

Familiarization - Demonstration of the test procedure:

Testing within the control and test site:

D) S/R (Stimulus/Response) Functions: Mechanical pain sensitivity (MPS) and Dynamic mecannical allodynica (DMA)

Familiarization - Demonstration of the test procedure:

MPS - "जसे आधीच्या परिक्षेमधे तसे विविध धातूंच्या छोट्या पिना तुमच्या त्वचेवर विविध दावावाने दायले. जातील."

DMA - "ह्या परिक्षेमधे तुमच्या त्वचेवर, अधूनमधून कापूरा, कापूस वड (q tip), किंवा ब्रश दाबले जाईल. ह्यामधील काही उत्तेजक तुम्हांला तीक्ष्ण, टोवणे, डंख किंव्हा जलजलण्याचा अनुभव देतील. इतर उत्तेजक फक्त हलक्या संवेदनाचा अनुभव देतील आणि काही उत्तेजक शूच्य जाणीव देतील.

कृपया प्रत्येक उत्तेजकाला तुमच्या चेदनेनुसार ० ते १०० मध्ये गुणमान या. कोणतेही तीक्षण, टोचणे, इंच किंव्हा जनजलण्याचा अनुभव हे वेदनादायक असे सांगितले जावे आणि त्याचे गुणमान ० च्या वर दिले जावे. तुम्हीं दशमान पद्धत वापरू शकता.

चा अर्थ वेदनाविरहित, न तीक्षण, न टोचणे, न डंख किंव्हा न जलजलण्याचा असा होतो.

🗇० चा अर्थ कल्पना करू शकू अशा असद्य वेदनेच्या संवेदना असा होतो.

कुठल्याही परिस्थितील तुमच्या त्वचेच्या ज्या जागी हे परीक्षण होत आहे, तिथे वध् नका." (Should the subject/patient give a rating of "100", please ask : 'तुम्हांला खात्री आहे का की झा वेदना तुमच्यासाठी असल आहेल ?')"

Testing within control and test site:

MPS - "आता पुन्हा कृपया प्रत्येक उत्तेजकाला तुमच्या अनुभवानुसार ० ते १०० मध्ये गुणमान खा."

~9MA - "॰ चा अर्थ वेदनाविरहित, न तीक्ष्ण, न टोचणे, न डंख किंव्हा न जलजलण्याचा असा होतो. १०० चा अर्थ कल्पना करू शकू अशा असह्य वेदनेच्या संवेदना असा होतो."

(Should the subject/patient give a rating of "100", please ask : 'तुम्हांला खात्री आहे का की ह्या वेदना तमच्यासाठी असहा आहेत ?')"

If applicable : ('मग झापुढे कुठलेही वेदनादायक उत्तेजक वापरण्यात येणार जाही.')

E) Wind - up ratio Familiarization - Demonstration of the test procedure: WUR - "आधीच्या परीक्षणासारखी मी आता एका धातूची छोटी पिन तुमच्या त्वचेवर ठेवेन. कृपया ह्या एका उत्तेजकाने दिलेल्या वेदनेला ० ते १०० मध्ये गुणमान द्या. कोणतेही तीक्षण, टोचणे,डंख किंवा जळजळणे अनुभव हे वेदनादायक असे सांगितले जावेत आणि त्याचे गुणमान ० च्या वर दिले जावे. तुम्हीं दशमान पद्धत वापरू शकता. चा अर्थ वेदनाविरहित, न तीक्षण, न टोचणे, न इंख किंवा न जळजळणे असा होतो. १०० चा अर्थ कल्पना करू शकू अशा असह्य वेदनेच्या संवेदना असा होतो." आताभी एका सेकंदाच्या अंतरावर,हे उत्तेजक दहा येळा तुमच्या त्वचेवर ठेवेन.ही पूर्ण दहा उत्तेजकांची क्रिया संपल्यावर कपया ह्या वेदनांचे ० ते १०० ह्या दरज्यान मध्यमान गुणमान करा. चा अर्थ वेदनाविरहित, न तीक्ष्ण, न टोचणे, न डंख किंवा न जळजळणे असा होतो. १०० चा अर्थ कल्पना करू शकू अशा असह्य वेदनेच्या संवेदना असा होतो." Testing within the control and test site: WUR - "आता जी आपण एक उत्तेजक आणि दहा उत्तेजकांची क्रिया केली ती ह्या जागी पाच वेळा करण्यात थेईल. कृपया एक उत्तेजक, किली वेदनामय होता ह्याचे ० ते १०० ह्या दरम्यान गुणमान करा. चा अर्थ वेदनाविरहित, न तीक्षण, न टोचणे, न डंख किंवा न जळजळणे असा होतो. १०० चा अर्थ कल्पना करू शकू अशा असंघ वेदनेच्या संवेदना असा होतो." कृपया धातू उत्तेजक दहा वेळा दाबल्यानंतर त्या वेदनांचे ० ते १०० ह्या दरम्यान मध्यमान गुणमान करा." F) Vibration detection threshold Familiarization - Demonstration of the test procedure: VDT - "ही प्रक्रिया तुमची कंपन संवेदना समर्थतेचे परीक्षण करेत. आता मी कंपन यन्त्र तुमच्या त्वचेवर ठेवेन. तुम्हांला कंपन जाणवत आहे किंवा नाही हे तुम्हीं कृपया मला सांगा. कृपया ज्या क्षणाला तुम्हांला कंपन जाणवेजासे होईल त्या क्षणी तुम्हीं 'आता' असे सांगा.तुम्हांला कदाचित कंपनाचा आवाज ऐकू येईल. कृपया तुगचे पूर्ण लक्ष आवाजावर न ठेवता फक्त कंपनावर ठेवा." Continue...

Testing within the control and test site: VDT - "आता ३ लागोपाठ परीक्षा घेतली जातील. आता मी कंपल यन्त्र तुमच्या त्वचेवर ठेवेव. तुम्हांला कंपल जाणवत आहे किया नाही हे तुम्हीं कृपया मला सांगा. कृपया ज्या क्षणाला तुम्हांला कंपन जाणवेनासे होईल तथा क्षणी तुम्हीं 'आता' असे सांगा." G) Pressure pain threshold Familiarization - Demonstration of the test procedure: PPT - "ही परीक्षा तुमच्या स्नायूंवरील दबाव वेदना मर्यादाची चाचणी घेईल. भी दबाव भाषन उपकरण तुमच्या स्नायूंवर दाबेन. कृपया ज्यावेळी साधारण दबावावर. जळजळ,डंख,णिरमिट किंव्हा ठणकणे होऊ लागते त्यावेळी लगेच 'आता' असे सांगा." Testing within the control and test site: PPT - "पुन्हां मी दबाव मापन यंत्र तुमच्या रूगयूंवर दावेन. कृपया ज्यावेळी साधारण दवादावर, जळजळ,इंख,गिरमिट किव्हा ठणकणे होऊ लागते त्यावेळी लगेच 'आता' असे सांगा. ही क्रिया तीन वेळा केली जाइंल."

Appendix 43. Participant's state of health

Questionnaire about the participant's state of health 1. In the past 3 months have you suffered from strong pain, last no yes, on less than 3 days yes, on less than 3 days yes, on more than 3 days a. If yes, what was the reason for your pain (multiple answe trauma operation pain during a flue period pain alcohol consumption ("hangover") injury/ aching muscles after sports other: no yes, less than 3 tablets/ month yes, 4-10 tablets/ month more often a. If yes, what was the reason for your medication (multiple trauma operation yes, less than 3 tablets/ month more often a. If yes, what was the reason for your medication (multiple trauma operation pain during a flue period pain alcohol consumption ("hangover") injury/ aching muscles after sports other: operation pain during a flue period pain alcohol consumption ("hangover")	ng <u>for more than 24 hours</u> ? rs possible): headache back pain stomach ache painful joints neuralgia stomach ache back pain stomach ache back pain stomach ache back pain stomach ache painful joints neuralgia
 In the past 3 months have you suffered from strong pain, last no yes, on less than 3 days yes, on more than 3 days a. If yes, what was the reason for your pain (multiple answe trauma operation pain during a flue period pain alcohol consumption ("hangover") injury/ aching muscles after sports other: 2. In the last 3 months have you taken medication for your pain no yes, less than 3 tablets/ month yes, 4-10 tablets/ month more often a. If yes, what was the reason for your medication (multiple trauma operation pain during a flue period pain alcohol consumption ("hangover") injury/ aching muscles after sports a. If yes, use the reason for your medication (multiple of the strauma) operation pain during a flue period pain alcohol consumption ("hangover") injury/ aching muscles after sports other: alcohol consumption ("hangover") injury/ aching muscles after sports other: 	ng <u>for more than 24 hours</u> ? rs possible): headache back pain stomach ache painful joints neuralgia headache back pain stomach ache painful joints neuralgia
 no yes, on less than 3 days yes, on more than 3 days a. If yes, what was the reason for your pain (multiple answe trauma operation pain during a flue period pain alcohol consumption ("hangover") injury/ aching muscles after sports other:	rs possible): - headache - back pain - stomach ache - painful joints - neuralgia
 yes, on more than 3 days a. If yes, what was the reason for your pain (multiple answe operation operation pain during a flue operiod pain alcohol consumption ("hangover") injury/ aching muscles after sports other:	rs possible): headache back pain stomach ache painful joints neuralgia sometric stomach ache back pain stomach ache painful joints neuralgia
 a. If yes, what was the reason for your pain (multiple answe trauma operation pain during a flue period pain alcohol consumption ("hangover") injury/ aching muscles after sports other: other: ves, less than 3 tablets/ month yes, 4-10 tablets/ month more often a. If yes, what was the reason for your medication (multiple trauma operation pain during a flue period pain alcohol consumption ("hangover") injury/ aching muscles after sports other: other: injury/ aching muscles after sports other: other: operation operation operation alcohol consumption ("hangover") injury/ aching muscles after sports other: other:	rs possible): headache back pain stomach ache painful joints neuralgia answers possible): headache back pain stomach ache painful joints neuralgia
 a. If yes, what was the reason for your medication (multiple a. If yes, what was the reason for your medication (multiple b. If yes, what was the reason for your medication (multiple c. If yes, what was the reason for your medication (multiple c. If yes, what was the reason for your medication (multiple c. If yes, what was the reason for your medication (multiple c. If yes, what was the reason for your medication (multiple c. If yes, what was the reason for your medication (multiple c. If yes, what was the reason for your medication (multiple c. If yes, what was the reason for your medication (multiple c. If yes, what was the reason for your medication (multiple c. If yes, what was the reason for your medication (multiple d. If yes, what was the reason for your medication (multiple d. If yes, what was the reason for your medication (multiple d. If yes, what was the reason for your medication (multiple d. If yes, what was the reason for your medication (multiple d. If yes, what was the reason for your medication (multiple d. If yes, what was the reason for your medication (multiple d. If yes, what was the reason for your medication (multiple d. If yes, what was the reason for your medication (multiple d. If yes, what was the reason for your medication (multiple d. If yes, what was the reason for your medication (multiple d. If yes, what was the reason for your medication (multiple d. If yes, what was the reason for your medication (multiple d. If yes, what was the reason for your medication (multiple d. If yes, what was the reason for your medication (multiple) d. If yes, what was the reason for your medication (multiple) d. If yes, what was the reason for your medication (multiple) d. If yes, what was the reason for your medication (multiple) 	answers possible): beadache painful joints neuralgia answers possible): headache back pain stomach ache painful joints neuralgia
 operation pain during a flue period pain alcohol consumption ("hangover") injury/ aching muscles after sports other:	 back pain stomach ache painful joints neuralgia answers possible): headache back pain stomach ache painful joints neuralgia
 pain during a flue period pain alcohol consumption ("hangover") injury/ aching muscles after sports other:	□ stomach ache □ painful joints □ neuralgia answers possible): □ headache □ back pain □ stomach ache □ painful joints □ neuralgia
 period pain alcohol consumption ("hangover") injury/ aching muscles after sports other:	□ pamful joints □ neuralgia answers possible): □ headache □ back pain □ stomach ache □ painful joints □ neuralgia
 a. If yes, what was the reason for your medication (multiple operation operation operation operation operation of the period pain operation of the period pain operation (multiple other:	answers possible): □ headache □ back pain □ stomach ache □ painful joints □ neuralgia
 other:	answers possible): □ headache □ back pain □ stomach ache □ painful joints □ neuralgia
 2. In the last 3 months have you taken medication for your pain no yes, less than 3 tablets/ month yes, 4-10 tablets/ month a. If yes, what was the reason for your medication (multiple trauma operation pain during a flue period pain alcohol consumption ("hangover") injury/ aching muscles after sports other: 3. Have you ever (more than 3 months ago) suffered from disea 	answers possible): = headache = back pain = stomach ache = painful joints = neuralgia
 no yes, less than 3 tablets/ month yes, 4-10 tablets/ month more often a. If yes, what was the reason for your medication (multiple trauma operation pain during a flue period pain alcohol consumption ("hangover") injury/ aching muscles after sports other:	answers possible): □ headache □ back pain □ stomach ache □ painful joints □ neuralgia
 yes, less than 3 tablets' month yes, 4-10 tablets' month more often a. If yes, what was the reason for your medication (multiple trauma operation pain during a flue period pain alcohol consumption ("hangover") injury/ aching muscles after sports other:	answers possible): □ headache □ back pain □ stomach ache □ painful joints □ neuralgia
 a. If yes, what was the reason for your medication (multiple trauma operation pain during a flue period pain alcohol consumption ("hangover") injury/ aching muscles after sports other:	answers possible): □ headache □ back pain □ stomach ache □ painful joints □ neuralgia
 a. If yes, what was the reason for your medication (multiple trauma operation pain during a flue period pain alcohol consumption ("hangover") injury/ aching muscles after sports other:	answers possible): □ headache □ back pain □ stomach ache □ painful joints □ neuralgia
 trauma operation pain during a flue period pain alcohol consumption ("hangover") injury/ aching muscles after sports other:	□ headache □ back pain □ stomach ache □ painful joints □ neuralgia
 operation pain during a flue period pain alcohol consumption ("hangover") injury/ aching muscles after sports other:	□ back pam □ stomach ache □ painful joints □ neuralgia
 pain during a nue period pain alcohol consumption ("hangover") injury/ aching muscles after sports other:	□ stomach ache □ painful joints □ neuralgia
 alcohol consumption ("hangover") injury/ aching muscles after sports other:	o neuralgia
□ injury/ aching muscles after sports □ other: 3. Have you ever (more than 3 months ago) suffered from disea	
 other: Have you ever (more than 3 months ago) suffered from disea 	
3. Have you ever (more than 3 months ago) suffered from disea	
(law and then 2 months)?	ses accompanied by long lasting pain
no ves	
If yes: h. If yes:	n narribla):
 If yes, what was the reason for your pain (multiple answer trauma 	n headache
operation	□ back pain
□ pain during a flue	🗆 stomach ache
□ period pain	🗆 painful joints
□ aconot consumption ("nangover")	🗆 neuraigia
o other:	
a. How long time ago did this happen?	
□ 3-12 months ago	
1-3 years ago	
□ 3-5 years ago	
□ 5-10 years ago	
a more man to years ago	

Questionnaire for healthy participants _ Template for the investigator Prof. A Rice Imperial College of Science, Technology and Medicine Instructions for the investigator: The grey marked answers in the template are not definit criteria. However, the investigator should check again by an extensive exploration and an er neurological/internistic examination, if the subject can be enrolled in the study as a "healthy 4. Have you ever sought professional help for your pain (from a doctor, holicstic therapist, psychotherapist, physiotherapist or chiropractor etc)? no new no never no	ve exclusion ttensive clinical y subject". psychiatrist/
Questionnaire for healthy participants _ Template for the investigator Prof. A Rice Imperial College of Science, Technology and Medicine Instructions for the investigator: The grey marked answers in the template are not definit criteria. However, the investigator should check again by an extensive exploration and an en- neurological/internistic examination, if the subject can be enrolled in the study as a "healthy 4. Have you ever sought professional help for your pain (from a doctor, holicstic therapist, psychotherapist, physiotherapist or chiropractor etc)? • no • yes 5. In the past 5 years have you had any psychological or psychiatric treatment? • no • yes, only for a short time • yes, for a longer 6. How often have you been signed off work due to pain in the last 3 years? • not applicable (i.e. retired or not actively working) • never • only a couple of days (overall max. 7 days) • more frequently 7. Do you smoke? • no • yes: <u>tow much?</u> • no • yes: <u>tow much?</u> • no at the moment, but I used to smoke: when did you quit? Instructions for the investigator: If the subject smokes more than 39 cigarettes per day, shexcluded. 8. How often and how much do you drink alcohol? • never • regularly, a lot • occasionally, a lot • often, beyond the proper amount	ve exclusion ttensive clinical y subject". psychiatrist/
Instructions for the investigator: The grey marked answers in the template are not definiti criteria. However, the investigator should check again by an extensive exploration and an enerological/internistic examination, if the subject can be enrolled in the study as a "healthy psychotherapist, physiotherapist or chiropractor etc?? a no a yes b no a yes c no a yes a no applicable (i.e. retired or not actively working) a not applicable (i.e. retired or not actively working) a not applicable (i.e. retired or not actively working) a not a power a only a couple of days (overall max. 7 days) a once (duration max. 7 days) a not at the moment, but I used to smoke: when did you quit? a not at the moment, but I used to smoke: when did you quit? c not at the invest	ive exclusion ktensive clinical y subject". psychiatrist/
 4. Have you ever sought professional help for your pain (from a doctor, holicstic therapist, psychotherapist, physiotherapist or chiropractor etc)? no yes 5. In the past 5 years have you had any psychological or psychiatric treatment? no yes, only for a short time yes, for a longer on tapplicable (i.e. retired or not actively working) never only a couple of days (overall max. 7 days) once (duration max. 7 days) more frequently 7. Do you smoke? no yes: in yes: low much? for how many yea no ta the moment, but I used to smoke: when did you quit? Instructions for the investigator: If the subject smokes more than 39 cigarettes per day, sh excluded. 8. How often and how much do you drink alcohol? never often, beyond the proper amount a. Please specify on average how much exactly you drink alcohol: 	psychiatrist/
 5. In the past 5 years have you had any psychological or psychiatric treatment? no yes, only for a short time yes, for a longer o ot applicable (i.e. retired or not actively working) never only a couple of days (overall max. 7 days) once (duration max. 7 days) more frequently 7. Do you smoke? no yes: <u>how much?</u> for how many yea not at the moment, but I used to smoke: when did you quit? for how many yea not at the moment, but I used to smoke: when did you quit? Instructions for the investigator: If the subject smokes more than 39 cigarettes per day, she excluded. 8. How often and how much do you drink alcohol? never regularly, in moderation regularly, a lot often, beyond the proper amount a. Please specify on average how much exactly you drink alcohol:	
 6. How often have you been signed off work due to pain in the last 3 years? not applicable (i.e. retired or not actively working) never only a couple of days (overall max. 7 days) once (duration max. 7 days) more frequently 7. Do you smoke? no yes: <a href="mailto:how much?] for how many yea not at the moment, but I used to smoke: when did you quit? Instructions for the investigator: If the subject smokes more than 39 cigarettes per day, sh excluded. 8. How often and how much do you drink alcohol? never regularly, in moderation occasionally, in moderation often, beyond the proper amount a. Please specify on average how much exactly you drink alcohol: method you consider your behaviour concerning the inteles of medication? 	period
 7. Do you smoke? no yes: how much? for how many yea not at the moment, but I used to smoke: when did you quit? Instructions for the investigator: If the subject smokes more than 39 cigarettes per day, shexcluded. 8. How often and how much do you drink alcohol? never regularly, in moderation occasionally, in moderation occasionally, a lot often, beyond the proper amount a. Please specify on average how much exactly you drink alcohol: 9. How do you consider your behaviour concerning the intake of medication? 	
Instructions for the investigator: If the subject smokes more than 39 cigarettes per day, shexcluded. 8. How often and how much do you drink alcohol? • never • regularly, in moderation • occasionally, in moderation • regularly, a lot • often, beyond the proper amount a. Please specify on average how much exactly you drink alcohol: • Please specify on average how much exactly you drink alcohol:	rs?
 8. How often and how much do you drink alcohol? never regularly, in moderation occasionally, in moderation regularly, a lot often, beyond the proper amount a. Please specify on average how much exactly you drink alcohol:	e/ he should be
a. Please specify on average how much exactly you drink alcohol: How do you consider your behaviour concerning the intake of medication?	
9 How do you consider your babariour concerning the intake of medication?	
 I don't take ever medication I take medication only seldom I take medication often I take too much medication 	
a. Which medication in what dose have you taken in the last 4 weeks regularly or on der prescription, over the counter and homeopathic/ herbal medication)?	nand (on
Healthy Volunteers Study Version 1.0 <	

Continue...

Questionnaire for healthy participants _ Template for the investigator Prof. A Rice Imperial College of Science, Technology and Medicine

Do you suffer from any other illnesses or disorders? In the table below are groups of conditions including examples. Please underline any of these which apply. Then please indicate how much this condition interferes with your everyday life where 0 implies no interference and 3 represents a high degree of interference.

Tumour, cancer	🗆 yes	□ no		
Which condition:	No	Interference	e	High degree
	[0]	[1]	[2]	[3]
Disorder of the nervous system, brain, or spinal cord	🗆 yes	□ no		
eg. Epilepsy, MS, Parkinsons, nerve injury,	No	Interforenc	~	Uigh dagraa
polyneuropathy, spinal cord injury, paralysis, neuralgia,	100	run run	r91	rigii degree
stroke, cranial nerve injury	[0]	[1]	[4]	[5]
Other:				
Disorder of the respiratory system e.g. Asthma, chronic	🗆 yes	□ no		
bronchitis, emphysems, pneumothorax, tuberculosis or				
inflammation of the lungs	No	Interferenc	e	High degree
Other:	[0]	[1]	[2]	[3]
Disorder of the heart or circulatory system e.g.	🗆 yes	□ no		
Coronary heart disease, coronary arrhythmia, thrombosis,	No	Interferenc	P	High degree
embolism, aneurysm, high blood pressure, myocardial	100	[1]	[21	[3]
infarction	[0]	[-]	[~]	[2]
Other:				
Gastrointestinal disorder e.g. Gastric reflux disorder,	□ yes	□ no		
Crom's disease, ulcerative colitis, naemmorrhoids,	No	Interferenc	e	High degree
incontinence of faeces, duodenal ulcers, irritable bowel	[0]	[1]	[2]	[3]
syndrome, incidence of stomach of intestinal naemorrage	1-1	1-1	L-1	-1
Other:				
Disorder of the liver, gall bladder or pancreas e.g.	□ yes			
the liver, call stones, call bladder inflammation	No	Interferenc	e	High degree
neucroatic inflammation	[0]	[1]	[2]	[3]
Other:	••		• •	
Disorder of the bidneys or urinary system e.g. chronic	FT 1/05	C 10		
kidney failure urinary tract inflammation bladder	L yes	2.10		
weakness endometriosis condition following renal colic	No	Interferenc	e	High degree
or kidney stones, sexual disturbance	[0]	[1]	[2]	[3]
Other:				
Metabolic disorder e.g. Disrupted sugar metabolism.	ves	□ no		
hyper- or hypo- glandular function, elevated blood lipid				
values	No	Interferenc	e	High degree
Other:	[0]	[1]	[2]	[3]
Muculoskeletal disorder e.g. Chronic polyarthritis,	🗆 yes	□ no		
rheumatic muscular inflammation, scoliosis, osteoporosis,	No	Interforme		Uich dograa
arthritis in the knee, hip or shoulder	100	r11	r91	rigii degree
Other:	[0]	[1]	[4]	[5]
Mental health disorder e.g. Depression, clinical anxiety,	🗆 yes	□ no		
panic attacks, anorexia, chronic fatigue and exhaustion,	No	Tutorform	-	Uich dograd
addiction or dependence, psychosis	100	run run	r91	rigii degree
Other:	[0]	[1]	[4]	[5]
Other disorders or conditions:	🗆 yes	□ no		
	No	Interferenc		High degree
	100	[11]	r91	riigii üegree
Pick factors of Heamonhilia hapatitis HIV	[V]	[1]	<u>14</u>	[2]
Other:	L yes			
Allergies e.g. Plasters, house dust, domestic/ cleaning	🗆 yes	n no		
products, cosmetics, pollen				

Healthy Volunteers Study

Version 1.0 <15_02_11>



Appendix 44. Recording sheet for QST

Appendix F: Publication and Presentations

Appendix F

Publications and Presentations

Appendix 45. Publications and Presentations

Poster presentations

- Neuropathic pain in treated leprosy patients in Ethiopia: a cross-sectional study. 14th World Congress on Pain, Milan, Italy August 27th – 31st, 2012 Haroun OMO, Hietaharju A, Bizuneh E, Tesfaye F, Brandsma J.W, Haanpää M, Rice A.S.C, Lockwood D.N.J
- Quantitative Sensory Testing Profiles in Leprosy Patients in Mumbai-India.
 4th International Congress on Neuropathic Pain, Toronto, Canada May 23rd 26th, 2013
 Haroun OMO, Pai VV, Shetty V, Pfau D, Bennett DLH, Maier C, Treede R-D, Rice
 ASC, Lockwood D.N.J
- Neuropathic Pain in Leprosy Patients in Mumbai: A case control study. 18th International Leprosy Congress, Brussels, Belgium September 16th – 19th, 2013 Haroun OMO, Khodke AS, Pai V, Shetty V, Pfau D, Bennett D, Maier C, Treede R-D, Rice ASC, Lockwood D.N.J

Accepted talk presentations

 Symptom and quantitative sensory profiling in Leprosy patients with and without neuropathic pain: a case control study.
 29th Biennial Conference of Indian Association of Leprologists on March 28-30, 2014

(cancelled)

• Sensory profiling and nerve damage in Leprosy patients with and without neuropathic pain: insights into mechanisms of disease.

15th World Congress on Pain, Buenos Aires, Argentina, October 6-11, 2014.

Publications

• In process of submitting a manuscript titled "Comparability of detection and pain thresholds in different approaches: Approaches to simplify the DFNS QST protocol"

Appendix G

Others

Independent variables	Proportion in the sample	Cases (%)	Odds ratio & (95% CI)	P. value
Skin lesions 1# symptoms				
No	28 (76.67%)	12 (46.15%)	1	0.05
Yes	62 (68.89%)	14 (53.85%)	0.38 (0.15-1.04)	0.05
Burning 1* symptoms				
No	86 (95.55%)	23 (88.46%)	1	0.04
Yes	4 (44.45%)	3 (11.54%)	1.19 (0.17-2.86)	0.04
Delay in presentation				
up to 6 month	44 (48.89%)	16 (61.59%)	1	
>6 months up to 1 year	11 (12.22%)	6 (23.08%)	1.22 (0.37–3.93)	<0.00
>1 year	35 (38.89%)	4 (15.83%)	0.13 (0.05-0.37)	
Diagnosis				
Newly diagnosed case	28 (31.11%)	2 (7.14%)	1	<0.00
Old case	62 (68.89%)	24 (38.17%)	8.21 (1.62-41.55)	
Duration of disease				
<6 months	11 (12.22%)	4 (36.36%)	1	
6 months up to 1 year	12 (13.33%)	6(50.00%)	1.57 (0.33-9.30)	0.05
>1 year up to 5 years	40 (44.44%)	29 (74.36%)	5.07 (1.22-21.06)	0.05
>5 years	27 (30.00%)	20 (74.07%)	4.99 (1.12-22.41)	
Disability at diagnosis				
Absent	54 (60.00%)	9 (16.67%)	1	
Grade 1	24 (26.67%)	13 (54.17%)	2 (0.62–16.11)	<0.00
Grade 2	12 (13.33%)	29 (82.86%)	3.87 (1.31-11.38)	
On MDT treatment				
No	18 (20.00%)	1 (5.56%)	1	0.004
Yes	72 (20.00%)	25 (34.72%)	9 (1.04-78.31)	0.000

Appendix 46. Univaiate logistic regression – Neuropathic pain

Independent variables	Proportion in the sample	Cases (%)	Odds ratio & (95% CI)	P. value
Release from MDT				
No	44 (48.89 %)	39 (61.90%)	1	<0.00
Yes	46 (51.11 %)	20 (76.92%)	2 (1.71-5.92)	
Previous MDT				
No	65 (72.11 %)	14 (57.69%)	1	0.05
Yes	26 (27.78 %)	11 (42.31%)	2.6 (0.96-7.16)	
Leprosy reactions				
No	64 (71.11 %)	14 (53.85%)	1	0.02
Yes	26 (28.89 %)	20 (46.15%)	3 (1.12-8.39)	
Previous reaction				
No	50 (55.56%)	8 (30.77%)	1	< 0.00
Yes	40 (44.45%)	18 (69.23%)	4 (1.51–12.14)	
No. of previous reactions				
0	50 (55.56%)	8 (30.77)	1	
1-3	35 (38.89%)	17 (65.38%)	1.3 (0.13–13.34)	0.04
>3	5 (5.56%)	1 (3.85%)	4.9 (1.81–13.55)	1
Depression				
Absent	30 (33.71)	2 (7.69%)	1	<0.00
Present	59 (66.29)	24 (92.31%)	9.59 (2.08-44.14)	-0.00

Page 2 of 2

Crude OR& CI	Adjusted OR & CI	P. valu
	,	_
1	1	0.01
9.59 (2.08-44.14)	8 (1.17-33.68)	0.01
1	1	
		0.07
4 (1.51–12.14)	3 (0.92 -9.11)	0.07
4 (1.51–12.14)	3 (0.92 -9.11)	0.07
4 (1.51–12.14)	3 (0.92 -9.11)	
	Crude OR& CI 1 9.59 (2.08-44.14)	Crude OR& CI Adjusted OR & CI 1 1 9.59 (2.08-44.14) 8 (1.17-33.68) 1 1

Appendix 47. Multivariate logistic regression – Neuropathic pain

Page 1 of 1

Independent variables	Proportion in the sample	Proportion with GHQ threshold score >3	Odds ratio & (95% CI)	P. valu
Estimates of association betw	veen GHQ cas	eness and socio	-demographic variabl	les
Sex				
Male	69 (76.67%)	44 (63.77%)	1	
Female	21 (23.33%)	15 (75.00%)	1.70 (0.38–5.31)	0.3
Age group				
<30 years	43 (47.78%)	27 (45.76%)	1	0.7
>30 years	47 (52.22%)	32 (54.24%)	1.19 (0.17-2.86)	- 0.71
Education				
Illiterate	14 (15.56%)	9 (64.29%)	1	
Primary	50 (55.56%)	37 (75.51%)	1.71 (0.48-6.11)	-
Secondary	11 (12.22%)	3 (27.27%)	0.21 (0.04-1.16)	0.04
High secondary	7 (7.78%)	4 (57.14%)	0.74 (0.12-4.73)	
College and above	8 (8.89%)	6 (75.00%)	0.74 (0.24–11.58)	
Marital status				
Married	60 (66.67%)	40 (67.80%)	1	0.67
Unmarried/widowed	30 (33.33%)	19 (63.33%)	1.22 (0.48-3.08)	
Estimates of association betw	veen GHQ cas	eness and clinic	al features	
Delay in presentation				
up to 6 month	44 (48.89%)	15 (62.50%)	1	
>6 months up to 1 year	11 (12.22%)	12 (85.71%)	3.6 (0.65–19.90)	0.00
>1 year	35 (38.89%)	32 (97.50%)	7.6 (2.72–20.90)	1
Diagnosis				
Newly diagnosed case	28 (31.11%)	13 (46.43%)	1	0.00
Old case	62 (68.89%)	46 (75.41%)	3.54 (1.31–9.53)	1

Appendix 48. Univaiate logistic regression – Depression

Independent variables	Proportion in the sample	Proportion with GHQ threshold score >3	Odds ratio & (95% CI)	P. value	
Duration of disease					
<6 months	11 (12.22%)	4 (36.36%)	1		
6 months up to 1 year	12 (13.33%)	6(50.00%)	1.57 (0.33-9.30)	0.05	
>1 year up to 5 years	40 (44.44%)	29 (74.36%)	5.07 (1.22-21.06)		
>5 years	27 (30.00%)	20 (74.07%)	4.99 (1.12-22.41)		
WHO classification					
MB	57 (63.33%)	40 (71.43%)	1	0.18	
PB	33 (36.67%)	19 (57.58)	1.84 (0.73-4.60)		
Disability at diagnosis					
Absent	54 (60.00%)	6 (17.14%)	1	0.00	
Present	36 (40.00%)	29 (82.86%)	3.87 (1.31-11.38)		
On MDT treatment					
No	18 (20.00%)	7 (38.89%)	1	0.00	
Yes	72 (20.00%)	52 (73.24%)	4.3 (1.38–13.44)		
Leprosy reactions					
No	46 (71.11 %)	39 (61.90%)		0.17	
Yes	26 (28.89 %)	20 (76.92%)	2.05 (0.71-5.92)		
Previous reaction					
No	50 (55.56%)	32 (80.00 %)	1	0.03	
Yes	40 (44.45%)	36 (90.00%)	2.25 (1.13-8.21)	1	
No. of previous reactions					
0	50 (55.56%)	28 (56.00%)	1	+	
1-3	35 (38.89%)	27 (79.41%)	3.03 (1.11-8.24)	0.06	
>3	5 (5.56%)	4 (80.00%)	3.14 (0.33–30.19)	-	

Page 2 of 4

Independent variables	Proportion in the sample	Proportion with GHQ threshold score >3	Odds ratio & (95% CI)	P. value
Skin lesion O/E				
No skin lesion	20 (22.22%)	16 (84.21%)	1	0.0
Skin lesion	70 (77.78%)	43 (61.43%)	0.30 (0.08–1.16)	- 0.00
Motor nerve impairment				
Absent	43 (47.78)	23 (53.49%)	1	0.0
Present	47 (52.22)	36 (78.26%)	3.13 (1.20-8.18)	
Ulnar neuropathy				
Absent	48 (53.33%)	25 (52.08%)	1	0.00
Present	42 (46.67%)	34 (82.93%)	4.46 (1.55–12.82)	0.00
Sensory nerve impairment				
Absent	39 (43.33%)	22 (56.41%)		0.0
Present	51 (56.67%)	37 (74.00%)	2.20 (0.88-5.49)	
Clinical evidence of neuropathy				
Absent	36 (40%)	20 (55.56%)	1	0.0
Present	54 (60%)	39 (73.58%)	2.22 (0.88-5.58)	
Disability (current)				
Absent	40 (44.44%)	22 (55.00%)	1	0.0
Present	50 (55.56%)	37 (75.51%)	2.52 (1.00-6.38)	- 0.0
Estimates of association betw	veen GHQ cas	eness and pain		
Current pain				
Absent	54 (60.00%)	29 (54.72%)		0.00
Present	36 (40.00%)	30 (83.33%)	4.14 (1.40–12.24)	0.00
	Page	3 of 4		

Independent variables	Proportion in the sample	Proportion with GHQ threshold score >3	Odds ratio & (95% CI)	P. value	
Burning sensation		1			
Absent	67 (74.44%)	39 (59.09 %)		0.01	
Present	23 (25.56%)	20 (86.96%)	4.62 (1.18–18.03)	0.01	
Electric shocks					
Absent	64 (71.11%)	35 (55.560 %)	1	0.001	
Present	26 (28.89)	24 (92.31%)	9.60 (1.86-49.50)	0.001	
Tingling sensation					
Absent	58 (64.44%)	32 (56.14%)	1	0.000	
Present	32 (35.56%)	36 (84.38%)	4.22 (1.35–13.22)	0.007	
Pins and needles sensation					
Absent	71 (78.89%)	42 (60.00%)		0.01	
Present	19 (21.11%)	17 (89.47%)	5.66 (1.14-28.15)		
Numbness sensation					
Absent	69 (76.67%)	40 (58.82%)	1	0.007	
Present	21 (23.33%)	19 (90.48%)	6.65 (1.33–33.27)	-	
Neuropathic pain					
Absent	64 (71.11%)	35 (55.56%)		0.001	
Present	26 (28.89%)	24 (92.31%)	9.60 (1.86-49.50)	1	
Non-NP pain					
Absent	80 (88.89%)	53 (67.09%)	1	0.66	
Present	10 (11.11%)	6 (60.00%)	1.07 (0.20-5.59)		
Pain					
Absent	54 (60.00%)	29 (54.72%)	1	0.001	
Non-neuropathic pain	10 (11.11%)	6 (60.00%)	1.24 (0.31-4.91)	-	
Neuropathic pain	26 (28.89%)	24 (92.31%)	9.93 (2.12-46.35)	-	

Page 4 of 4
Variables	Crude OR& CI	Adjusted OR & CI	P. value				
Neuropathic pain							
Absent	1	1					
Present	9.60 (1.86–49.50)	6.25 (1.17–33.57)	0.03				
Motor nerve impairment							
Absent	1	1	0.00				
Present	3.13 (1.20-8.18)	4.21 (0.79-22.50)	0.09				
Skin lesion O/E							
Absent	1	1	0.20				
Present	3.43 (0.08–1.16)	0.38 (0.85-1.68)	0.20				
On MDT treatment							
No	1	1	0.21				
Yes	4.30 (1.38–13.44)	2.61 (0.58-11.98)	- 0.21				
Sensory nerve impairment							
Absent	1	1	0.28				
Present	3.20 (0.88–5.49)	0.38 (0.07-2.15)					
Disease duration							
<1 year	1	1	0.36				
>1 year	3.75 (1.32–10.61)	1.80 (0.51-6.42)					
Education							
Literate	1	1	0,51				
Illiterate	1.11 (0.33–3.69)	0.61 (0.15-2.57)					
Disease diagnosis							
New case	1	1	0.83				
Old case	3.54 (1.31–9.23)	0.86 (0.21-3.57)					

Appendix 49. Multivariate logistic regression – Depression

 \Uparrow Multivariate logistic regression of the association between variables and depression outcome

Page 1 of 1

Appendix 50. Pain-related definitions (IASP Taxonomy, 2014)

Pain-related definitions (IASP)

Allodynia: Pain due to a stimulus which does not normally provoke pain.

Analgesia: Absence of pain in response to stimulation which would normally be painful.

Anesthesia dolorosa: Pain in an area or region which is anesthetic.

Causalgia: A syndrome of sustained burning pain, allodynia, and hyperpathia after a traumatic nerve lesion, often combined with vasomotor and sudomotor dysfunction and later trophic changes.

Dysesthesia: An unpleasant abnormal sensation, whether spontaneous or evoked.

Hyperalgesia: An increased response to a stimulus which is normally painful.

Hyperesthesia: Increased sensitivity to stimulation, excluding the special senses.

Hyperpathia: A painful syndrome characterized by an abnormally painful reaction to a stimulus, especially a repetitive stimulus, as well as an increased threshold.

Hypoalgesia: Diminished pain in response to a normally painful stimulus.

Hypoesthesia: Decreased sensitivity to stimulation, excluding the special senses.

Neuralgia: Pain in the distribution of a nerve or nerves.

Neuritis: Inflammation of a nerve or nerves.

Neuropathic Pain: Pain caused by a lesion or disease of the somatosensory nervous system

Neuropathy: A disturbance of function or pathological change in a nerve: in one nerve, mononeuropathy; in several nerves, mononeuropathy multiplex; if diffuse and bilateral, polyneuropathy.

Nociceptor: A high-threshold sensory receptor of the peripheral somatosensory nervous system that is capable of transducing and encoding noxious stimuli.

Nociceptive pain: Pain that arises from actual or threatened damage to non-neucal tissue and is due to the activation of nociceptors.

Noxious stimulus: A stimulus that is damaging or threatens damage to normal tissues.

Pain: An unpleasant sensory and emotional experience associated with actual or potential tissue damage, or described in terms of such damage.

Pain Threshold: The minimum intensity of a stimulus that is perceived as painful.

Pain tolerance level: The maximum intensity of a pain-producing stimulus that a subject is willing to accept in a given situation.

Paresthesia: An abnormal sensation, whether spontaneous or evoked.

Recent advances in Leprosy with Special reference to Neuropathic Pain in Leprosy Safurday 27 October, 10:00 – 12:30 Seminar room, department of Neurology, 4th Floor, main building, J.J Hospital, Mumbai 10:00 Welcome 10:05 – 10:25 Historical prospective of leprosy work conducted at Grant Medical College-JJ Hospital from 1845 Dr Shubhada Pandya Research Officer The Founcation for Medical Research officer The Founcation for Medical Research officer The Founcation for Medical Research (FMR) 10:25 – 10:25 Leprosy Professor Diana Loatwood Longon Medical Research (FMR) 10:25 – 10:25 Leprosy Professor Andrew Rice Importation for Medical Research (FMR) 10:25 – 11:25 Leprosy Professor Andrew Rice Importation Research (FMR) 10:25 – 11:25 Neuropathic pain concept Dr. Ormer Haroun Philo Pandidate, LSHTM 11:25 – 11:40 Tea – Coffice break Dr. Ormer Haroun Philo Pandidate, LSHTM 11:40 – 12:00 Pain in Loprosy: An experience from Philo candidate, LSHTM Dr. S Anadikar Professor and Head, Dept of Neurology, Bombay Hospital 12:20 – 12:30 Discussion Dr. Nerges Mistry Director, FMR	<u>2012 Se</u>	minar:	
Safurday 27 October, 10:00 – 12:30 Seminar room, department of Neurology, 4th Floor, main building, J.J Hospital, Mumbai 10:00 Welcome Dr V. V. Pai Director Bombay Leprosy Project 10:05 - 10:25 Historical prospec live at leprosy work conducted at Grant Medical College-JJ Hospital from 1645 Dr Shubhada Pandya Research Officer The Foundation for Medical Research (FMR) 10:25 - 10:55 Leprosy Professor Diana Lackwood Loncon School of Hygiene and Trapical Medicine (LSHTM) 10:55 - 11:25 Neuropathic pain concept Professor Andrew Rice Imperial College London 11:20 - 11:20 Pain in Laprosy: An experience from Ethiopian neuropathic poin study Dr. Orner Haroun PhD candidate, LSHTM 12:20 - 12:20 Case scenario Dr. S Chadilkar Professor and Head, Dept of Neurology, Bombay Hospital 12:20 - 12:30 Discussion Dr. Nerges Mistry Director, FMR	Recent a to Neuro	idvances in Leprosy with S pathic Pain in Leprosy	Special reference
10:00WelcomeDr V. V. Pai Director Bombay Leprosy Project10:05 - 10:25Historical prospecifive of leprosy work conducted at Grant Medical College-LJ Hospital from 1845Dr Shubhada Pendya Research Officer The Foundation for Medical Research Officer The Foundation for Medical Research (FMR)10:25 - 10:55LeprosyProfessor Diana Lockwood London School of Hygiene and Tropical Medicine (LSHTM)10:55 - 11:25Neuropathic pain conceptProfessor Andrew Rice Imperial College London11:25 - 11:40Tea - Coffice breakDr. Ormer Haroun PhD condidate, LSHTM12:20 - 12:20Case scenarioDr. S Khadilkar Professor and Head, Dept of Neurology, Bombay Hospital12:30Vote of ThanksDr. Nerges Mistry Director, FMR	Saturday 27 Seminar roc 4th Floor, m	7 October, 10:00 – 12:30 om, department of Neurology, ain building, J.J Hospital, Mumbo	яі
10:05 - 10:25Historical prospective of leprosy work conducted at Grant Medical College-JJ Hospital from 1845Dr Shubhada Pandya Research Officer The Foundation for Medical Research (FMR)10:25 - 10:55LeprosyProfessor Diana Lockwood London School of Hygiene and Tropical Medicine (LSHTM)10:55 - 11:25Neuropathic pain conceptProfessor Andrew Rice Imperial College London11:25 - 11:40Tea - Coffee breakDr. Orner Haroun PhD candidate, LSHTM11:40 - 12:00Pain in Loprosy: An experience from Ethiopian neuropathic pain studyDr. S Khadikar Professor and Head, Dept of Neurology, Bombay Hospital12:20 - 12:30DiscussionDr. Nerges Mistry Director, FMR	10:00	Welcome	Dr V. V. Pai Director Bombay Leprosy Project
10:25 - 10:55LeprosyProfessor Diana Lockwood London School of Hygiene and Tropical Medicine (LSHTM)10:55 - 11:25Neuropathic pain conceptProfessor Andrew Rice Imperial College London11:25 - 11:40Tea - Coffae breakDr. Omer Haroun PhD candidate, LSHTM11:40 - 12:00Pain in Loprosy: An experience from Ethiopian neuropathic pain studyDr. Omer Haroun 	10:05 - 10:25	Historical prospec live of leprosy work conducted at Grant Medical College-JJ Hospital from 1845	Dr Shubhada Pandya Research Officer The Foundation for Medical Research (FMR)
10:55 - 11:25Neuropathic pain conceptProfessor Andrew Rice Imperial College London11:25 - 11:40Tea - Coffee break11:40 - 12:00Pain in Leprosy: An experience from Ethiopian neuropathic pain studyDr. Omer Haroun PhD candidate, LSHTM12:00 - 12:20Case scenarioDr. S Khadilkar Professor and Head, Dept of Neurology, Bombay Hospital12:20 - 12:30D'scussion12:30Vote of ThanksDr. Nerges Mistry Director, FMR	10:25 – 10:55	Leprosy	Professor Diana Lockwood London School of Hygiene and Tropical Medicine (LSHTM)
11:25 - 11:40Tea - Coffee break11:40 - 12:00Pain in Loprosy: An experience from Ethiopian neuropathic pain studyDr. Omer Haroun PhD candidate, LSHTM12:00 - 12:20Case scenarioDr. S Khadilkar Professor and Head, Dept of Neurology, Bombay Hospital12:20 - 12:30D'scussion12:30Vote of ThanksDr. Nerges Mistry Director, FMR	10:55 11:25	Neuropathic pain concept	Professor Andrew Rice Imperial College London
11:40 - 12:00Pain in Loprosy: An experience from Ethlopian neuropathic pain studyDr. Omer Haroun PhD candidate, LSHTM12:00 - 12:20Case scenarioDr. S Khadilkar Professor and Head, Dept of 	11:25-11:40	Tea – Coffee break	
12:00 - 12:20Case scenarioDr. S Khadilkar Professor and Head, Dept of Neurology, Bombay Hospital12:20 - 12:30Discussion12:30Vote of ThanksDr. Nerges Mistry Director, FMR	11:40 - 12:00	Pain in Loprosy: An experience from Ethiopian neuropathic pain study	Dr. Omer Haroun PhD candidate, LSHTM
12:20 – 12:30 D'soussion 12:30 Vote of Thanks Dr. Nerges Mistry Director, FMR	12:00 - 12:20	Case scenario	Dr. S Khadilkar Professor and Head, Dept of Neurology, Bombay Hospital
12:30 Vote of Thanks Dr. Nerges Mistry Director, FMR	12:20 - 12:30	Discussion	
	12:30	Vote of Thanks	Dr. Nerges Mistry Director, FMR
		•	

Appendix 51. Neuropathic pain workshop – Mumbai

Appendix 52. DFNS-QST translation - Mumbai



Appendix 53. Studies on amitriptyline for the treatment of neuropathic pain

Condition	on Authors, year Study design No of Intervention I participants (Durati on	Primary outcome	Secondary outcome	Conclusion		
	Max et al. 1987	Double-blind,	29 patients	Amitriptyline	6	VAS pain	HDS, POMS-D and	Amitriptyline was
	(Max, 1987)	randomised placebo crossover study		(average 90mg) vs placebo	weeks	intensity	DACL	superior to placebo
Diabetes Mellitus (DM)	Max <i>et al.</i> 1992 (Max et al., 1992)	Double-blind, randomised placebo crossover study	38 patients	Amitriptyline (average 105mg) vs desipramine (average 111mg) vs placebo	6 weeks	Rating pain intensity	Hamilton Depression Scale (HDS)	Amitriptyline was superior to placebo Similar effects of the amitriptyline & desipramine
	Verthem <i>et al.</i> 1997 (Vrethem et al., 1997)	Double-blind, randomised placebo crossover study	37 patients (33 completed)	Amitriptyline (75mg) vs maprotiline vs placebo	4 weeks	Pain relief scale	Comprehensive Psychological Pain Scale (CPRS)	Amitriptyline was superior to placebo
Post Herpetic Neuralgia (PHN)	Watson <i>et al.</i> 1985 (Watson and Evans, 1985)	Open-label cross- over	15 patients	Amitriptyline vs zimelidine	6 weeks	Pain relief scale	Depression	Amitriptyline was superior to Zimelidine
	Watson <i>et al.</i> 1992 (Watson et al., 1992)	Randomised, double-blind, crossover	35 patients	Amitriptyline vs maprotiline	6 weeks	Pain relief scale	Depression	Amitriptyline was superior to maprotiline
	Watson <i>et al.</i> 1998 (Watson et al., 1998)	Randomised, double-blind, crossover	33 patients (31 finished)	Amitriptyline vs nortriptyline	6 weeks	VAS pain intensity	Mood and satisfaction	Similar effects of the amitriptyline & nortript
	Radford <i>et al.</i> 2000 (Graff- Radford et al., 2000)	Double-blind, placebo-controlled, parallel study	49 patients	Amitriptyline	8 weeks	VAS pain intensity	MPQ and side- effects scale	Amitriptyline was superior to placebo
	Rowbotham et al. 2005 (Rowbotham et al., 2005)	Randomised, double-blind, parallel design	38 patients	Amitriptyline 150 mg vs imipramine 150mg vs fluoxetine 60mg	6 weeks	VAS pain intensity	Pain relief scale, BDI and QST	Similar effects of the three drugs

1

Continue...

Condition	Authors, year	Study design	No of participants	Intervention	Durati on	Primary outcome	Secondary outcome	Conclusion
Phantom limb pain	Robinson et al. 2004 (Robinson et al., 2004)	Randomised, controlled, parallel trial	39 patients	Amitriptyline 10- 125 mg vs placebo	6 weeks	NRS pain intensity	SF-MPQ, BPI interference scale	Similar effects of the amitriptyline and placebo
HIV/AIDS	Kieburtz <i>et al.</i> 1998 (Kieburtz et al., 1998)	Randomised, double-blind, parallel design	145 patients	Amitriptyline 25- 100 mg vs. Mexiletine vs placebo	10 weeks	Gracely Pain Score (GP): The change in pain intensity at baseline and week 10		Neither Amitriptyline nor mexiletine provides significant pain relief in HIV neuropathy
	Shly JC <i>et al.</i> 1998 (Shlay et al., 1998)	Randomised, placebo controlled, parallel design	136 patients	Amitriptyline 25- 75 mg (n=71) vs. Placebo (n=65)	14 weeks	Changes in mean pain score at base line & wk 14	Quality of life assessment	Amitriptyline was not more effective than placebo in relieving pain in HIV neuropathy
Chemother apy induced neuropathy	Kautio <i>et al</i> . 2008	Double-blind, randomised, placebo-controlled, parallel design	44 patients (33 completed the study)	Amitriptyline 50mg vs placebo	8 weeks	Numerical rating pain intensity	Severity of mood, sleep and QOL: QOL, STAI and NPSI	Amitriptyline was superior to placebo
Malignancy neuropathi c pain	Mercadante <i>et al.</i> 2002 (Mercadante et al., 2002)	Double-blind, randomised placebo crossover study	16 patients	Amitriptyline 50mg vs placebo	7 days	NRS pain intensity	Spitzer's QOL index, sleep, mood (NRS)	Amitriptyline was the same as placebo
Central post stroke pain	Leijon <i>et al.</i> 1989 (Leijon and Boivie, 1989)	Randomised placebo crossover study	15 patients	Amitriptyline 75mg vs. placebo	4 weeks	VRS pain intensity	Global rating scale and CPRS	Amitriptyline was superior to placebo
Spinal cord injury pain	Cardenas et al. 2002 (Cardenas et al., 2002)	Randomised, double-blind, parallel design	84 patients	Amitriptyline 125mg vs. placebo	6 week	NRS pain intensity	BPI, short form McGill pain qs.	Similar effects of the amitriptyline & placebo
Post mastectom	Kalso et al. 1996 (Kalso et al., 1996)	Crossover study	15 patients	Amitriptyline 100mg vs. placebo	4 weeks	VAS and VRS pain relief	Anxiety and depression	Amitriptyline was superior to placebo

2

Continue...

References:

CARDENAS, D. D., WARMS, C. A., TURNER, J. A., MARSHALL, H., BROOKE, M. M. & LOESER, J. D. 2002. Efficacy of amitriptyline for relief of pain in spinal cord injury: results of a randomized controlled trial. *Pain*, 96, 365-73.

GRAFF-RADFORD, S. B., SHAW, L. R. & NALIBOFF, B. N. 2000. Amitriptyline and fluphenazine in the treatment of postherpetic neuralgia. *Clin J Pain*, 16, 188-92.

KALSO, E., TASMUTH, T. & NEUVONEN, P. J. 1996. Amitriptyline effectively relieves neuropathic pain following treatment of breast cancer. *Pain*, 64, 293-302.

KIEBURTZ, K., SIMPSON, D., YIANNOUTSOS, C., MAX, M. B., HALL, C. D., ELLIS, R. J., MARRA, C. M., MCKENDALL, R., SINGER, E., DAL PAN, G. J., CLIFFORD, D. B., TUCKER, T. & COHEN, B. 1998. A randomized trial of amitriptyline and mexiletine for painful neuropathy in HIV infection. AIDS Clinical Trial Group 242 Protocol Team. *Neurology*, 51, 1682-8.

LEUON, G. & BOIVIE, J. 1989. Central post-stroke pain--a controlled trial of amitriptyline and carbamazepine. Pain, 36, 27-36.

MAX, M. B. 1987. Endogenous monoamine analgesic systems: amitriptyline in painful diabetic neuropathy. Anesth Prog, 34, 123-7.

MAX, M. B., LYNCH, S. A., MUIR, J., SHOAF, S. E., SMOLLER, B. & DUBNER, R. 1992. Effects of desipramine, amitriptyline, and fluoxetine on pain in diabetic neuropathy. N Engl J Med, 326, 1250-6.

MERCADANTE, S., ARCURI, E., TIRELLI, W., VILLARI, P. & CASUCCIO, A. 2002. Amitriptyline in neuropathic cancer pain in patients on morphine therapy: a randomized placebo-controlled, double-blind crossover study. *Tumori*, 88, 239–42.

ROBINSON, L. R., CZERNIECKI, J. M., EHDE, D. M., EDWARDS, W. T., JUDISH, D. A., GOLDBERG, M. L., CAMPBELL, K. M., SMITH, D. G. & JENSEN, M. P. 2004. Trial of amitriptyline for relief of pain in amputees: results of a randomized controlled study. Arch Phys Med Rehabil, 85, 1-6.

ROWBOTHAM, M. C., REISNER, L. A., DAVIES, P. S. & FIELDS, H. L. 2005. Treatment response in antidepressant-naive postherpetic neuralgia patients: double-blind, randomized trial. J Pain, 6, 741-6.

SHLAY, J. C., CHALONER, K., MAX, M. B., FLAWS, B., REICHELDERFER, P., WENTWORTH, D., HILLMAN, S., BRIZZ, B. & COHN, D. L. 1998. Acupuncture and amitriptyline for pain due to HIV-related peripheral neuropathy: a randomized controlled trial. Terry Beirn Community Programs for Clinical Research on AIDS. JAMA, 280, 1590-5.

VRETHEM, M., BOIVIE, J., ARNQVIST, H., HOLMGREN, H., LINDSTROM, T. & THORELL, L. H. 1997. A comparison a amitriptyline and maprotiline in the treatment of painful polyneuropathy in diabetics and nondiabetics. *Clin J Pain*, 13, 313-23.

WATSON, C. P., CHIPMAN, M., REED, K., EVANS, R. J. & BIRKETT, N. 1992. Amitriptyline versus maprotiline in postherpetic neuralgia: a randomized, double-blind, crossover trial. *Pain*, 48, 29-36.

WATSON, C. P. & EVANS, R. J. 1985. A comparative trial of amitriptyline and zimelidine in postherpetic neuralgia. Pain, 23, 387-94.

WATSON, C. P., VERNICH, L., CHIPMAN, M. & REED, K. 1998. Nortriptyline versus amitriptyline in postherpetic neuralgia: a randomized trial. *Neurology*, 51, 1166-71.

3

	Year																
	2010 2011			2012				2013				2014					
Quarter	4 th	1 st	2 nd	3 rd	4 th	1 st	2 nd	3 rd	4 th	1 st	2 nd	3 rd	4 th	1 st	2 nd	3 rd	4 th
Planning and literature review		<u> </u>	I	.													
Refinement of PHD objectives																	
Study protocols																	
QST training and validation studies																	
Ethical approvals																	
Upgrading exam																	
Recruitment																	
Data Analysis																	
Thesis Writing																	
Submission																	
Viva																	

Appendix 54. Thesis timeline